

# NOTICE

THIS DOCUMENT HAS BEEN REPRODUCED FROM THE BEST COPY FURNISHED US BY THE SPONSORING AGENCY. ALTHOUGH IT IS RECOGNIZED THAT CERTAIN PORTIONS ARE ILLEGIBLE, IT IS BEING RELEASED IN THE INTEREST OF MAKING AVAILABLE AS MUCH INFORMATION AS POSSIBLE.

BACKARD CONTRACTOR DATA

AGARDograph No. 139

# NORTH ATLANTIC TREATY CROANIZATION Advisory group for Aerospace Research and Development (organisation du traite de l'Atlantique Nord)

THEORY AND APPLICATIONS OF WALMAN FILTERING

# Edited by

# C. T. Leondes

# University of California Los Angeles, USA

. .

.

.

This AGARDograph was propared at the request of the Guidance and Control Panel of AGARD - NATU

**)** 

•

**D. WAYNE DAUS** 

n Triestern Calle

·

. .

.

. . .

Published February 1970

# 629.7.058 : 519.61 : 629.78

•

# €

Printed by Technical Editing and Reproduction Ltd Harford House, 7-9 Charlotte Street, London W1P 1HD.

# 11

1 .....

-C1--\_1

in a station of the station



# PREFACE

The theory and applications of Kalman filtering have now been under development for overten years. This general area has achieved a level of maturity and importance to easily justify a thorough treatment of the subject. In view of the particular importance of Kalman filtering techniques to the field of guidance and control the Guidance and Control Panel of NATO-AGARD recommended that such a text be developed under the auspices of NATO-AGARD.

This textbook is the result of that recommendation. The text is organized into three principal parts. The first part examines the theory of Kalman filtering in depth. A sumber of significant new results of fundamental importance are included here. For instance, such questions as existence of Kalman filters under very general conditions, Kalman filtering for Gauss-Markov processes, suboptimal Kalman filtering techniques, and other areas are treated here.

The second part of the text deals with the general area of related topics. Questions of the comparison of Kalman filtering with other approaches such as Bayesian and maximum likelihood estimation, nonlinear filtering, linear and nonlinear smoothing (post-flight data analysis), and other topics are reviewed in depth in this part.

The third part is a very comprehensive review of many of the important applications of Kalman filtering. Although many very specific areas of application are treated in this part, many general principles and bechniques for a very broad range of applications of Kalman filtering will be found here. As a result, the reader should also find this part quite valuable no matter what particular application he might have in mind.

It is a great pleasure to acknowledge the contributions of many individuals who made this text possible. First of all, Professor W.Wrigley's contributions as the first Panel Chairman of the Guidance and Control Panel cannot be praised too highly. His cutatanding efforts in guiding this Panel during its formative stages cannot be praised enough. This book is one of the fruits of his efforts. All the members of the Guidance and Control Panel were solicited for their advice and suggestions, and their help is also greatly appreciated. Colonel L.Sugarman and Nr.Frank Sullivan provided much important support. Nr O.H.Schuck made many important suggestions. Colonel W.Studabaker and Major C.Mount provided invaluable assistance in their roles as Guidance.

Cornelius T. Leondes



CONTENTS;

PREFACE

Page iii

A Regiment of Fre

MAN PAR

St. C. Harrison C.

PART I - THEORY CHAPTER 1 LINEAR ESTIMATION THEORY by H.W. Sorenson and A.R. Stubberud 1 CHAPTER 2 FURTHER COMMENTS ON THE DERIVATION OF KALMAN FILTERS PERIVATION OF THE KALMAN FILTERING EQUATIONS FROM ELEMENTARY Section 1 STATISTICAL PRINCIPLES 43 by P.M.Barham and D.E.Humphries GAUSSIAN ESTIMATES AND KALMAN FILTERING Section II by Y. Genin 51 CHAPTER 3 65 by Stanley F. Schmidt MODELING ERRORS IN KALMAN FILTERS; by T.Nishimura CHAPTER 4 87 SUBOPTIMAL KALMAN FILTER TECHNIQUES CHAPTER 5 by A.S. Stubberud and D.A. Wisser 105

# PART II - RELATED TOPICS

CHAPTER	6	COMPARISON OF KALMAN, BAYESIAN AND MAXIMUM LIKELIHOOD ESTIMATION Techniques by M.W.Sofermon	. 119
CHAPTER	7	NONLINEAR FILTERING AND COMPARISON WITH KALMAN FILTERING.	143
CHAPTER	8	LINEAR BRUOTHING TECHNIQUES (POST-PLIGHT DATA ANALYSIS)," by Nerbert E. Rauch	163
CHAPTER	9	NONLINEAR SMOOTHING TECHNIQUES	183

# PART III - APPLICATIONS

CHAPTER	10	GENERAL QUESTIONS ON KALMAN FILTERING IN NAVIGATION SYSTEMS by Larry D.Brock and George T.Bohmidt	203
CHAPTER	11	> APPLICATION OF KALMAN FILTERING THEORY TO AUGMENTED INERTIAL Navigation systems by J. R. Muddie	231
CH AP TER	12	** APPLICATION OF KALMAN FILTERING TO BARO/INERTIAL HEIGHT SYSTEMS * by P.N.Barham and P.Manville	26
CH APTER	13	APPLICATION OF KALMAN FILTERING TO THE C-5 GUIDANCE AND CONTROL SYSTEM ; by Stanley F. Schmidt, John D. Weinberg and John S. Lukesh	28
CHAPTER	14	APPLICATION OF KALMAN FILTERING TECHNIQUES TO THE APOLLO PROGRAM by Richard H.Battin and Gerald M.Lovine	33
CHAPTER	15	SOME APPLICATIONS OF KALMAN FILTERING IN SPACE GUIDAN 'E - by J.F.Bellantoni	36

et et et et et et

1.1.

المراجع والمراجع وال

CHAPTER 14		Page
Rection 1	> APPLICATION OF BALMAN FILTERING FOR THE ALIGNMENT OF CARRIER ABRCBAFT INVETEAL NAVIGATION SYSTEMS by J.T. Komba	
Section []	NAVIGATION AT REA USING THE INVARIANTS FORM OF KALMAN FILTERING ( by N.B. Bambrook and N. Nalmandaria	483
CHAPTER 17	- WARINE APPLICATIONS OF KALWAN PILTERING by J. Boldavorth and J. Bloja	451
CWAPTER 18	• OPTIMAL USE OF REDUNDANT INFORMATION IN AN INERTIAL NAVIGATION _ = BYRTEN by F.B.Uager and N.OLL	467
CWAPTER 19	APPLICATION OF KALBAN FILTERING TECHNIQUES TO STRAPDOWN System initial alignment	
	my carle b trocker and Leonard Esbina	491
CHAPTER 20	by H. Halabandarin and D. Ozden /	812

Auto

. . . . 1

ر --- ر سر 1 سام رو

# CHAPTER I - LINEAR ESTIMATION THEORY

100000000

1.772.00

-

ALL DE C

30. C ()

the Winstein Wheel

at Sinthe from Siens

W Durinnin Diality

by

H.W.Sorenson\* and A.R.Stubberud<sup>†</sup>

 University of California, San Diego, La Jolla, California, USA

<sup>†</sup> University of California, Irvine, California, USA

1

•	NUMBER		
t	the independent variable, usually referred to as time	الميسيمية بطرهم المراجع	
t <sub>o</sub>	initial time		
I(t)	the residual or innovations process and defined to be $r(t/\tau) \stackrel{h}{\in} z(t) = H(t) \hat{z}(t/\tau)$		
l(t)	addifive white-noise process in the measurement data with mean zero and covariance matrix $R(t)\delta(t-\tau)$		
T(t)	additive white-hoise process in the plant with mean zero and covariance matrix $Q(t)\delta(t+ au)$		
<u>X</u> (t)	the n-dimensional state vector described by a linear differential equation $\frac{1}{2} = F(t) + y$		
<u>î(t/</u> 7)	estimate of the state $\underline{x}(t)$ given the measurement data $\underline{Z}(\tau)$		
<u>]</u> (t/τ)	error in the estimate, $\underline{I}(t/\tau) \stackrel{\Delta}{\Rightarrow} \underline{X}(t) - \underline{\hat{X}}(t/\tau)$		
Z(t)	the noise-free measurement data, $y(t) \stackrel{\Delta}{\to} H(t) \chi(t)$		
<u>z(t)</u>	the measurement data, $\underline{x}(t) \triangleq H(t) \underline{x} + \underline{y}$		
<u>Z(</u> 7)	the measurement data, $Z(\tau)$ for all $t_0 \leq \tau < t$		
C(t)	cross-correlation matrix for plant and measurement noise, $E[\underline{v}(t)\underline{v}^{T}(\tau)] = C(t)\delta(t-\tau)$		
F(\$)	plant matrix, $\frac{1}{2} = F(t) \underline{x} + \underline{w}$		
H(t)	observation matrix, $\underline{z} = H(t)\underline{x} + \underline{y}$		
K(t)	optimal gain matrix for unbiased minimum variance filter		
M(t,7)	observability matrix, $H(t,\tau) \triangleq \int_{t}^{\tau} \Phi^{\tau}(\sigma,t) H^{\tau}(\sigma) R^{-1}(\tau) H(\sigma) \Phi(\sigma,t) d\sigma$		
P(t/7)	covariance of the error in the estimate, $P(t/\tau) \triangleq E[\underline{z}(t/\tau)\underline{z}^{T}(b/\tau)]$		
Q(t)	covariance matrix for the plant noise w(i)		
R(t)	covariance matrix for the measurement noise $\underline{v}(t)$		
Φ(t,T)	transition matrix associated with the plant equation, $\Phi(t,\tau) = F(t)\Phi(t,\tau)$ , $\Phi(\tau,\tau) = I$		
Ÿ(t,↑)	transition matrix associated with the filter dynamics, $\Psi(t,\tau) = [F(t) - K(t)H(t)]\Psi(t,\tau)$ , $\Psi(\tau,\tau) = I$		
I	identity matrix of appropriate dimension		
Q	the quantity ( ) is a vector (or column matrix)		
( ) <sup>T</sup>	the matrix transpose of ( )		
()-1	the matrix inverse of ( )		
( ) <sup>-T</sup>	the matrix inverse of ( ) <sup>T</sup>		
<b>Z</b> (A)	expected value of the random variable a		
8(*)	the Dirac delta function	a na gala 💼 agala sa	
() ⊈[]	the quantity ( ) is defined to be equivalent to [ ]		





## CHAPTER 1 - LINEAR ESTIMATION THEORY

H.W. Sorenson and A.R. Stubberud

#### 1. THE LINEAR ESTIMATION PROBLEM

-11

這些父母的重要的影響。以及改善的影響。他的公司

#### 1.1 Introduction to the Linear Estimation Problem

A large class of estimation problems is concerned with finding an optimal estimate of some quantity (an unknown parameter, a random variable, or a random signal) when a linear function of this quantity corrupted by an additive noise is available for generating the estimate. One of the first studies of this class of problems was performed by Gauss<sup>1</sup> in the early 1800's. In this work he studied least-squares estimates of unknown parameters.

In the early 1940's Wiener<sup>2</sup> and Kolmogorov<sup>3</sup> attacked the class of problems dealing with estimation of random signals. The key result from this work is an integral equation called the Wiener-Hopf equation. The solution of this equation is a weighting function which, when convolved with the corrupted linear measurement, produces an unbiased minimum variance estimate of the random signal. Since the Wiener-Hopf equation can only be solved explicitly for certain special cases of the general problem, this work has only limited practical application. Many generalizations of this work were presented in the 1940's and 1950's but none of these improved its practical applicability<sup>4-7</sup>.

In the 1950's the idea of generating least-squares estimates recursively was introduced. This interest was stimulated by the increased usage of digital computers. These estimates are generated dynamically and the data processing algorithm is either a differential or a difference equation. Carlton<sup>5</sup> asserts that the first work on this subject was done by Follin<sup>9</sup> about 1955. In 1968 Swerling published a Rund Corporation report (which received wider distribution in 1959 in the Journal of the Astronautical Sciences<sup>10</sup>) that presented a recursive filtoring procedure similar to that described shortly thereafter by Kaluan<sup>11</sup>. The latter work is generally considered to have sparked the widespread intowast in the subject and subsequent references to "Kalman filtering".

The paper by Kalman<sup>11</sup> in 1960 introduced a different approach to the problem of Wiener and Kolmogorov for random sequences. In 1961, Kalman and Bucy<sup>12</sup> generalized the results to random processes. Basically, this approach circumvents the problem of solving the Wiener-Hopf integral equation. By recognizing that digital computers are much more effective at solving differential equations than integral equations, Kalman and Bucy transformed the integral equation into an equivalent differential equation. Then, rather than demand an analytical solution for this equation, they recognized that, from a practical standpoint, it is better to put the computational burden on the computer. These results are closely related to those obtained for sequential least-squares, estimation. The practicality of the Kalman approach to the estimation problem has made it immensely popular in servespace applications, such as in navigation and guidance.

It is the intent in this chapter to discuss the fundamental aspects of the unbiased, minimum variance, linear estimation problem. The treatment of the problem as presented here has identifiable roots in the paper by Kalman and Buoy<sup>12</sup> and even more so in the remarkable treatise by Kalman that constitutes Reference 13. However, the presentation describes not only those early and basic results but attempts to present the developments in the years since the seminal papers appeared that either provide insights into the foundations or which add significantly to the theoretical structure. It may be an overstatement to suggest that there are as many derivations of the Kalman filter equations as there are workers in the field. On the other hand it cannot be denied that there are many approaches to the problym and each has its vociferous supporters. It is not practical, nor is it desirable, to attempt to satisfy all tastes. The development here is designed to appeal to intuition and to provide insight that explains those aspects which may not be obvious. In taking this approach, mathematical rigor is sacrificed or details are omitted when it appears that greater clarity can be achieved by ignoring what amounts to be a technical detail that does not affect the famel results.

The basic mathematical model and a more precise definition of the linear estimation problem are given in Section 1.2. Some variations and generalizations of this model and problem are considered in Sections 2 and 3. A detailed summary of the more important results of Sections 2 and 3 is given in Section 4. It is suggested that this section be read after completing Section 1.2 and before going to the detailed discussions of Sections 2 and 3. In this way this summary can be used as a guide through the seeming morass of Kalman-Buoy filter theory.

#### 1.2 The Mathematical Model and Problem Statement

Consider a dynamical system whose state evolution is described by a linear, stochastic, vector differential equation.



Industed y

A DOWN A DOWN





Hand Street and Street

Å = F(t)χ + μ.

where

X is an n-dimensional state vector

F(t) is an nxn matrix whose elements are continuous functions of the independent variable t

is an u-dimensional, gaussian white-noise process with the following statistics

$$E[\underline{w}(t)] = 0 \quad \text{for all } t$$
$$E[\underline{w}(t)w^{T}(\tau)] = Q(t)^{S}(t-\tau) ,$$

where Q(t) is an  $n \times n$  symmetric, non-negative-definite matrix and  $\delta(t - \tau)$  is the Dirac delta function. The initial state  $\underline{r}(t_{\alpha})$  is a random variable with known statistics.

$$E[\underline{\chi}(t_0)] = \underline{\mathbf{R}}_0$$

$$E\{[\underline{\chi}(t_0) - \underline{\mathbf{R}}_0][\underline{\chi}(t_0) - \underline{\mathbf{R}}_0]^{\mathsf{T}}\} \qquad \mathbf{M}_0$$

Also,  $\underline{x}(t_n)$  is independent of  $\underline{w}(t)$ , that is,

$$\mathbb{E}[\underline{w}(t)\underline{x}^{T}(t_{n})] = 0 \quad \text{for all } t.$$

The solution of Equation (1.1) is given by  $1^{4}$ 

$$\underline{\mathbf{x}}(\mathbf{t}) = \Phi(\mathbf{t}, \mathbf{t}_0) \underline{\mathbf{x}}(\mathbf{t}_0) + \int_{\mathbf{t}_0}^{\mathbf{t}} \Phi(\mathbf{t}, \tau) \underline{\mathbf{x}}(\tau) d\tau ,$$

where  $\Phi(t,\tau)$  is the transition matrix and is the solution of the matrix differential equation

$$\frac{d\Phi(t,\tau)}{dt} = F(t)\Phi(t,\tau); \quad \Phi(\tau,\tau) = I \quad \text{for all } \tau \; .$$

The transition matrix has the property that

 $\Phi(\mathbf{t}_k,\mathbf{t}_j)\Phi(\mathbf{t}_j,\mathbf{t}_j) = \Phi(\mathbf{t}_k,\mathbf{t}_j) \quad \text{for all } \mathbf{t}_k,\mathbf{t}_j,\mathbf{t}_j .$ 

This implies that

 $\Phi^{-1}(\mathbf{t_k},\mathbf{t_j}) = \Phi(\mathbf{t_j},\mathbf{t_k}) ,$ 

so that  $\Phi$  is nonsingular for all the  $t_{k}, t_{j}$ . The solution and its properties will be used in the succeeding discussion.

The only information available about the state are m measurements  $\underline{x}(t)$  that are related to  $\underline{x}(t)$  according to

 $\underline{x}(t) = H(t)\underline{x}(t) + \underline{y}(t)$ . (1.2) A

The  $\underline{y}(t)$  is an m-dimensional, gaussian white-noise process with the statistics

$$\mathbf{E}[\underline{\mathbf{y}}(t)] = 0 \quad \text{for all } t$$
$$\mathbf{E}[\underline{\mathbf{y}}(t)\underline{\mathbf{r}}^{T}(\tau)] = \mathbf{R}(t)\delta(t-\tau) .$$

The plant noise y(t) and the measurement noise y(t) are assumed to be independent for most of the discussion, although the generalization to correlated processes is discussed in Section 2.1.3. The symmetric  $m \times m$ matrix R(t) must be assumed to be positive-definite for the straightforward development of the Kalman filter equations. When R(t) is not positive-definite, it is necessary to introduce the special considerations relating to the colored-noise problem discussed in Section 2.2. Also, by a corruption of the language, the matrices Q(t)and R(t) will often be referred to as the noise covariance matrices, although the covariance matrices actually are of infinite magnitude because of the delta function.

The estimation problem associated with the system Equations  $(1.1) \sim (1.2)$  can now be defined.







(1.1)

Unbiased, Minimum Variance, Linear Estimation Problem

Given the linear system described by Equations (1.1) and (1.2), determine an estimate  $\hat{g}(t)\tau$ ) of the state  $\underline{x}(t)$  that is a linear function of all measurement data\*  $z(\sigma)$ ,  $\underline{t}_0 \leq \sigma < \tau$ , and satisfies the following conditions:

(1)  $g(t|\tau)$  is unbiased so that

$$\mathbb{E}[\underline{\hat{z}}(t|\tau)] = \mathbb{E}[\underline{x}(t)].$$

(ii)  $\hat{g}(t|\tau)$  is "best" in the sense that the expected value of the square of the error magnitude is minimized. Thus, the estimate  $\hat{g}(t|\tau)$  is chosen so that

$$\mathbb{E}\left\{\left[\underline{\mathbf{x}}(\mathbf{t}) - \underline{\mathbf{x}}(\mathbf{t}|\boldsymbol{\tau})\right]^{\mathsf{T}}\left[\underline{\mathbf{x}}(\mathbf{t}) - \underline{\mathbf{x}}(\mathbf{t}|\boldsymbol{\tau})\right]\right\} = \min$$

The estimate that is obtained depends upon the amount of data that is available (i.e. as defined by  $\tau$ ) and can be described in terms of three special problems.

**Prediction:** Suppose that the state at a time t is to be estimated from data  $Z(\tau)$  where  $\tau < t$ . Thus, the state  $\chi(t)$  is to be predicted from data obtained at times prior to the time t. This shall be referred to here as the prediction problem.

Filtering: Consider the problem of estimating the  $\underline{x}(t)$  from data  $\underline{z}(t)$  (i.e.  $\tau = t$ ). This shall be referred to as the filtering problem.

Scoothing: Suppose that the state  $\underline{x}(t)$  is to be estimated from data  $\underline{Z}(\tau)$  where  $\tau > t$ . In this case  $\underline{x}(t)$  is estimated from data obtained at times prior to, coincident with, and subsequent to, the time of interest, t. This shall be referred to as the smoothing problem.

Those three special cases of the estimation problem are also referred to as the extrapolation, smoothing and interpolation problems (see, for example, References 2 and 15). The problems have been stated in the order of increasing complexity of solution. The emphasis here is upon the prediction and filtering problems, although the smoothing problem is discussed briefly in Section 2.3. The so-called Kalman-Buoy filter is involved with the first two cases.

#### 2. SOLUTION OF THE LINEAR ESTIMATION PROBLEM

## 2.1 Linear Prediction and Filtering

#### 2.1.1 Heuristic Derivation

A derivation of the Kalman-Bucy equations is presented in this section that appeals primarily to intuitive reasoning rather than mathematical rigor. This approach is taken initially to provide insight into the character of the solution which is shown more rigorously in the following section to provide the unbiased, minimum variance estimate of the state  $\chi(t)$  based on the measurement data Z(t).

Suppose that an estimate of the state  $\underline{\chi}(t_0)$  is available at some time  $t_0$  that is based upon measurement data  $\underline{\chi}(t_0)$ . Let this estimate be denoted as  $\underline{\chi}(t_0|t_0)$ . In Equation (1.1), it is seen that the strip changes in accordance with a linear differential equation with a white-noise forcing function. The process  $\underline{\chi}(t)$  has zero mean and the value at different times is uncorrelated, regardless of the magnitude of the time difference. Thus, it is reasonable to expect, in the absence of additional data, that the behavior of the estimate would be described by

$$\hat{\mathbf{g}} = \Gamma(\mathbf{t}) \hat{\mathbf{g}}, \quad \mathbf{t} > \mathbf{t}_0, \quad (2.1)$$

where  $\hat{g}(t_0|t_0)$  is known. But when measurement data are available for time subsequent to  $t_0$  one can consider the residual (i.e. the difference between the measurement data  $\underline{z}(t)$  and that predicted by the estimate  $H(t)\hat{g}(t|t)$ ).

r(t) = z(t) - H(t)g(t|t) (2.2)

This residual is referred to by Kailath<sup>16</sup> as the innovations process and will be examined in more detail in Section 2.1.2.

The residual can be considered to provide an indication of the error in the estimate  $\hat{\chi}(t|t)$ . Let us assume that this error will be used to modify the estimate provided by Equation (2.1) by introducing an unknown weighting or gain matrix K(t). Using this weighting matrix, assume that the estimate, including all new measurement data, is to have the form

 $\frac{1}{2}(t|t) = F(t)g(t|t) + K(t)[z(t) - H(t)g(t|t)]$  for  $t > t_0$ , (2.3)

• Menceforth, the collection of measurement data  $g(\sigma)$  on an interval  $t_0 \le \sigma \le \tau$  will be denoted as  $Z_i(\tau)$ .



May on grand the

5









where  $\hat{\chi}(t_0|t_0)$  is known. To simplify notation, the arguments of the estimate will be suppressed, so that  $\hat{\chi}$  will be synonymous with  $\hat{\chi}(t|t)$ .

Using the form assumed in Equation (2.3), the matrix K(t) will be chosen so that the variance of the error is minimized as required in the statement of the filtering problem in Section 2.1. Note also that the solution of Equation (2.1) yields the solution of the prediction problem for  $t > t_0$ .

First note that the estimate provided by Equation (2.3) is unbiased if the initial conditions are solected in an appropriate manner.

Equation (2.3) is an unbiased estimator if the initial condition  $f(t_0|t_0)$  satisfies the constraint that

$$\mathbb{E}[\hat{\underline{x}}(t_0|t_0)] = \mathbb{E}[\underline{x}(t_0)] . \tag{2.4}$$

Series an

To verify this, note from Equation (1.1) that

 $E[\frac{1}{2}] = F(t)E[\frac{1}{2}]$ 

and from Equation (2.3) that

 $\mathbb{E}\left[\frac{\dot{k}}{2}\right] = \mathbb{F}(t)\mathbb{E}\left[\frac{\dot{k}}{2}\right] + \mathbb{K}(t)\left\{\mathbb{E}\left[\frac{1}{2}\right] - \mathbb{H}(t)\mathbb{E}\left[\frac{\dot{k}}{2}\right]\right\} .$ 

But from Equation (1.2) one sees that

$$\mathbf{E}[\underline{\mathbf{z}}] = \mathbf{H}(\mathbf{t})\mathbf{E}[\underline{\mathbf{x}}] ,$$

so that

R

$$E[\underline{\dot{x}}] = F(t)E[\underline{\dot{x}}] + K(t)H(t)\{E[\underline{x}] - E[\underline{\dot{x}}]\}$$

Thus, it follows that

$$\frac{u}{dt} \left( \mathbb{E} \left[ \underline{x} - \underline{\hat{x}} \right] \right) = \left\{ \mathbb{P}(t) - \mathbb{K}(t) \mathbb{H}(t) \right\} \mathbb{E} \left[ \underline{x} - \underline{\hat{x}} \right] .$$

This is a homogeneous, linear differential equation in the variable  $E[\underline{x} - \hat{\underline{x}}]$  so its solution has the general form

 $\mathbf{E}[\underline{\mathbf{X}}(t) - \underline{\widehat{\mathbf{X}}}(t)] = \Psi(t, t_0) \mathbf{E}[\underline{\mathbf{X}}(t_0) - \underline{\widehat{\mathbf{X}}}(t_0|t_0)] .$ 

The matrix  $\Psi$  is the transition matrix associated with the filter dynamics and is the solution of the matrix differential equation

$$\Psi(\mathbf{t},\mathbf{t}_{0}) \simeq \left[ \mathbf{F}(\mathbf{t}) - \mathbf{K}(\mathbf{t})\mathbf{H}(\mathbf{t}) \right] \Psi(\mathbf{t},\mathbf{t}_{0}) \quad \Psi(\mathbf{t}_{0},\mathbf{t}_{0}) = \mathbf{I} \; .$$

Thus, if

Define

 $\mathbf{E}[\mathbf{g}(\mathbf{t}_0)] = \mathbf{E}[\mathbf{g}(\mathbf{t}_0|\mathbf{t}_0)] ,$ 

it follows that  $E[g(t)] = E[\hat{g}(t|t)]$ ,

so that the estimate is unbiasod.

Using the form for the estimate assumed in Equation (2.3), consider the reformulated filtering problem. Determine the time-varying matrix K(t) so that  $\mathbb{E}[(\underline{x} - \underline{\hat{x}})^T(\underline{x} - \underline{\hat{x}})]$  is minimized.

Let  $\underline{\tilde{g}}(t|t) \stackrel{\Delta}{=} \underline{g}(t|t) = \underline{\hat{g}}(t|t)$  (2.5) so that  $\mathbb{E}[(\underline{x} - \underline{\hat{x}})^T (\underline{x} - \underline{\hat{x}})] = \mathbb{E}[\underline{\hat{x}}^T \underline{x}]$  $= \text{trace } \mathbb{E}[\underline{\hat{x}} \underline{x}^T]$ .

# $P(t|t) \triangleq E[\underline{f}(t|t)]$ (2.6)

Thus, it is desired that the gain matrix K(t) be chosen to minimize the trace of the error covariance matrix P(t|t).

The rate of change of the error in the estimate is obtained by using Equations (1.1), (1.2), and (2.3).

$$\frac{d}{dt}(\underline{X}) = \frac{d}{dt} \underline{X} - \frac{d}{dt} \underline{\hat{X}}$$
$$= F(t)\underline{X} + \underline{W} - F(t)\underline{\hat{X}} - K(t)[\underline{X} - H(t)\underline{\hat{X}}]$$
$$= [F(t) - K(t)H(t)]\underline{Y} - K(t)\underline{Y} + \underline{W}.$$

The rate of change of the error covariance matrix is given by

$$\frac{d}{dt} (P) = E[\underline{x}\underline{x}^T] + E[\underline{x}\underline{x}^T]$$
(2.8)

where

and analysis in

 $\mathbb{E}\left[\frac{1}{XX}^{T}\right] = \mathbb{E}\left\{\left[\mathbb{P}(t) - K(t)H(t)\right]\frac{1}{XX}^{T} - K(t)\underline{X}^{T} + \underline{w}\underline{X}^{T}\right\}.$ 

But it follows from the assumptions about the noise processes that

and 
$$E[\underline{Y}(t)\underline{X}^{T}(t|t)] = -\frac{1}{2}R(t)K^{T}(t)$$

For example, consider the first of these two relations. The solution of Equation (2.7) has the form

$$\underline{\underline{X}}(t|t) = \Psi(t,t_0)\underline{\underline{X}}(t_0|t_0) - \int_{t_0}^t \Psi(t,\tau)K(\tau)\underline{\underline{V}}(\tau) d\tau + \int_{t_0}^t \Psi(t,\tau)\underline{\underline{W}}(\tau) d\tau .$$

-

Using this, one sees that

$$= \mathbb{E}[\underline{\mathbf{y}}(t)\underline{\mathbf{x}}^{\mathsf{T}}(t|t)] = \mathbb{E}[\underline{\mathbf{y}}(t)\underline{\mathbf{x}}(t_0|t_0)]\Psi^{\mathsf{T}}(t,t_0) - \int_{t_0}^{t} \mathbb{E}[\underline{\mathbf{y}}(t)\underline{\mathbf{x}}^{\mathsf{T}}(\tau)]\Psi^{\mathsf{T}}(t,\tau) d\tau + \int_{t_0}^{t} \mathbb{E}[\underline{\mathbf{y}}(t)\underline{\mathbf{x}}^{\mathsf{T}}(\tau)]\Psi^{\mathsf{T}}(t,\tau) d\tau .$$

-But

$$\mathbb{E}\left[\underline{\mathbf{Y}}(t)\underline{\mathbf{X}}(t_0|t_0)\right] = 0$$
$$\mathbb{E}\left[\underline{\mathbf{Y}}(t)\underline{\mathbf{u}}^{\mathrm{T}}(\tau)\right] = 0,$$

from the basic assumptions, so that

$$\mathbf{E}[\underline{\mathbf{y}}(t)\underline{\mathbf{g}}^{\mathrm{T}}(t)\mathbf{t}] = -\int_{t_0}^{t} \mathbf{R}(\tau)\delta(t-\tau)\mathbf{K}^{\mathrm{T}}(\tau)\Psi^{\mathrm{T}}(t,\tau) d\tau$$

 $= -\frac{1}{t}R(t)K^{T}(t) ,$ 

As a result, the error covariance matrix is found to satisfy the differential equation.

$$\frac{d}{dt} P = \left[ F(t) - K(t)H(t) \right] P + P[F(t) - K(t)H(t)]^{T} + K(t)K(t)K^{T}(t) + Q(t) .$$
(2.9)

It is important to realize that Equation (2.9) describes the behavior of the error covariance matrix for any gain matrix K(t). This relation is next used to derive the K(t) that yields the minimum error variance. Also, note from the definition that P(t|t) is symmetric.

Combine all terms that contain the unknown gain matrix K(t) . Then

 $\dot{P} = F(t)P + PF^{T}(t) + Q(t) + \{K(t)R(t)K^{T}(t) - K(t)H(t)P - PH^{T}(t)K^{T}(t)\}$ .

The terms outside the brackets cannot be affected directly by the choice of the gain matrix K(t), so they will be ignored for the moment. The matrix R(t) has been assumed to be symmetric and positive-definite so it can be factored into the product of a nonsingular matrix S(t) and its transpose  $S^{T}(t)$ ,

 $R(t) \cdot = S(t)S^{T}(t)$ .

Assume the existence of a matrix A(t) such that

 $KRK^{T} - KHP - PH^{T}K^{T} = [KS - A][KS - A]^{T} - AA^{T}$ .



. . .

(2.7)

For this to be valid, choose

A = PHTS"T

where  $[]^{-7}$  denotes the inverse of  $[]^{T}$ . With this choice for A, the differential equation for P becomes

$$\dot{\mathbf{P}} = \mathbf{F}\mathbf{P} + \mathbf{P}\mathbf{K}^{T} + \mathbf{Q} - \mathbf{P}\mathbf{H}^{T}\mathbf{R}^{-1}\mathbf{H}\mathbf{P} + [\mathbf{K} - \mathbf{P}\mathbf{H}^{T}\mathbf{R}^{-1}]\mathbf{R}[\mathbf{K} - \mathbf{P}\mathbf{H}^{T}\mathbf{R}^{-1}]^{T}.$$
(2.10)

Thus the gain matrix enters the quadratic term alone. For a prescribed initial value  $P(t_0)$  and recognition of the fact that

$$\frac{d}{dt}$$
 (trace P) = truce  $\frac{d}{dt}$  (P),

it follows that (trace P) is minimized at each time by choosing K(t) so that  $\dot{P}$  is as small as possible. This is accomplished by choosing K so that the quadratic term in Equation (2.10) is eliminated. Thus, the optimal gain is

$$K(t) = P(t|t)H^{T}(t)B^{-1}(t)$$
 (2.11)

and the error covariance mutrix for the optimal gain is described by

$$\dot{P} = P(t)P + PP^{T}(t) - PH^{T}(t)R^{-1}(t)H(t)P + Q(t) , \qquad (2.12)$$

with prescribed initial condition  $P(t_0)$ . The arguments leading to Equations (2.11) and (2.12) can be made more precise by using variational arguments, as discussed by Athans<sup>17</sup>. The error covariance matrix P(t|t) is determined by solving Equation (2.12). This equation is a matrix Ricatti equation<sup>14</sup> and is discussed in more detail in Section 3.2.

This completes the heuristic derivation of the equations of the Kalman-Buoy filter.

# Summary of Principal Results

The unbiased, minimum variance estimate of the linear system described by Equations (1.1) ~ (1.2) is given as the modution of the system

$$\frac{1}{2}(t) = P(t) \frac{1}{2}(t) + K(t) [\frac{1}{2}(t) - H(t) \frac{1}{2}(t) t)],$$
 (2.3)

where  $\hat{g}(t_0|t_0)$  is selected so that

$$\mathbb{E}[\underline{\hat{z}}(t_0|t_0)] = \mathbb{E}[\underline{x}(t_0)].$$

The optimal gain matrix is given by

$$K(t) = P(t|t)H^{T}(t)R^{-1}(t)$$
 (2.11)

and the error covariance is obtained as the solution of the matrix Ricatti equation

$$\frac{d}{dt} P(t|t) = F(t)P(t|t) + P(t|t)F^{T}(t) + Q(t) - P(t|t)H^{T}(t)R^{-1}(t)H(t)P(t|t) . \qquad (2.12)$$

2.1.2 The Innovations Approach

A more rigorous derivation of the solution of the linear, unbiased, minimum variance filtering problem is presented in this section. The approach follows that of Kailath<sup>16</sup> and has a strong similarity to the derivation presented by Kaiman and Buoy in their original paper<sup>12</sup>.

To begin, the properties of the residual  $\underline{r}(t)$  defined in Equation (2.2) will be examined in more detail. Kallath refers to this process as the innovations process because the residual contains in essence the "new" information contained in the measurement data  $\underline{x}(t)$ .

The process

# $\underline{\mathbf{r}}(t) = \underline{\mathbf{x}}(t) - \mathbf{H}(t) \hat{\underline{\mathbf{x}}}(t|t)$

is a white-noise process with the same statistics as the measurement noise process  $\underline{v}(t)$  when  $\underline{x}(t|t)$  is the minimum variance estimate of  $\underline{x}(t)$  using  $\underline{z}(t)$ .





The plausibility of this statement can be seen by direct calculation. First, verify that g(t) is a whitenoise process.

Let	χ(t)	₽	H(t) <u>Z(t)</u> ,	(2.13)
Then	Ŷ(t t)	=	H(t)ĝ(t t)	* (2. 14)
and	∑(tit)	=	月(t) <u>賞(t</u> )t) ;	(2. 15)
80	Ę(t)	=	$\underline{\mathbf{z}}(t) = \hat{\underline{\mathbf{y}}}(t t)$	
			H/+>9/+ (+> + W/+>	(2.10)

Suppose t > s . Using Equation (2.16), form

 $\mathbb{E}[\underline{r}(t)\underline{r}^{\mathsf{T}}(s)] = \mathbb{E}[\underline{g}(t)\underline{y}^{\mathsf{T}}(s|s)] + \mathbb{E}[\underline{y}(t)\underline{y}^{\mathsf{T}}(s|s)] + \mathbb{E}[\underline{y}(t)\underline{y}^{\mathsf{T}}(s)] + \mathbb{E}[\underline{y}(t)\underline{y}^{\mathsf{T}}(s)] = \mathbb{E}[\underline{y}(t)\underline{y}^{\mathsf{T}}(s)]$ 

Since  $\underline{\mathbf{v}}(\mathbf{t})$  is a white-noise process,

 $\mathbf{E}[\underline{\mathbf{Y}}(\mathbf{t})\underline{\mathbf{y}}^{\mathsf{T}}(\mathbf{s})] = 0$  $\mathbf{E}[\underline{\mathbf{Y}}(\mathbf{t})\underline{\mathbf{y}}^{\mathsf{T}}(\mathbf{s})] = 0.$ 

Consequent ly

and

and

1

 $\mathbb{E}[\underline{r}(t)\underline{r}^{T}(s)] = \mathbb{E}[\underline{y}(t)t)[\underline{y}(s)s) + \underline{y}(s)]^{T}]$ 

# = $\mathbb{E}\{\widehat{g}(t|t)[g(s) - \widehat{g}(s)]^T\}$ .

But it is well-known that the error  $\frac{y}{2}(t)t$  for the minimum variance estimate must be orthogonal to the measurements  $\underline{Z}(s)$  in the sense that

 $\mathbb{E}\left[\frac{1}{2}T(t|t)\mathbb{Z}(s)\right] = 0$ , s < t, (2.17)

This property is proven in Section 2.1.3 and is essentially the Wiener-Hopf equation.

Assuming that  $\frac{2}{3}(t|t)$  is a linear function of the measurement data Z(t) and is the minimum variance estimate, it follows from Equation (2.17) that

$$\mathbb{E}[\mathbf{r}(t)\mathbf{r}^{\mathrm{T}}(s)] = 0 \quad \text{for } t > s$$

A similar argument for t < s shows that

$$\mathbb{E}[\underline{r}(t)\underline{r}^{T}(s)] = 0 \quad \text{for } t < s ,$$

so it only remains to consider  $t \neq s$ . It is necessary to demonstrate that  $E[\underline{r}(t)\underline{r}^{T}(t)]$  is infinite in order to conclude that  $\underline{r}(t)$  is a white-noise process.

$$\left[ \mathbf{r}(t) \mathbf{r}^{\mathsf{T}}(t) \right] = \mathbf{E} \left[ \mathbf{v}(t) \mathbf{v}^{\mathsf{T}}(t) \right] + \mathbf{E} \left[ \mathbf{\overline{y}}(t|t) \mathbf{\overline{y}^{\mathsf{T}}}(t|t) \right] + \mathbf{E} \left[ \mathbf{v}(t) \mathbf{\overline{y}^{\mathsf{T}}}(t|t) \right] + \mathbf{E} \left[ \mathbf{\overline{y}}(t|t) \mathbf{v}^{\mathsf{T}}(t) \right] .$$

But  $\underline{y}(t)$  is defined as the error in the estimate of  $\underline{y}(t)$  for the data  $\underline{x}(s)$ ,  $t_g \leq s < t$  and  $\underline{y}(t)$  is white-noise so

$$\mathbf{E}[\mathbf{y}(\mathbf{t}|\mathbf{t})\mathbf{y}^{\mathrm{T}}(\mathbf{t})] = \mathbf{0} = \mathbf{E}[\mathbf{y}(\mathbf{t})\mathbf{y}^{\mathrm{T}}(\mathbf{t}|\mathbf{t})]$$

# $\mathbb{E}[\underline{r}(t)\underline{r}^{T}(t)] = \mathbb{R}(t)\delta(t-t) + \mathbb{E}[\underline{r}(t|t)\underline{y}^{T}(t|t)] ,$

But the covariance  $\mathbb{E}[\underline{y}(t|t)\underline{y}^{T}(t|t)]$  is finite so it is negligible by comparison with  $R(t)\delta(t-t)$  and one concludes that

$$\mathbb{E}[\underline{r}(t)\underline{r}^{T}(s)] = \mathbb{R}(t)\delta(t-s) .$$

The residual  $\underline{r}(t)$  represents the difference between the new measurement  $\underline{x}(t)$  and the prediction based on all previous data  $\frac{1}{2}(t)t$ . Thus  $\underline{r}(t)$  contains the information being contributed by the  $\underline{x}(t)$ , so  $\underline{r}(t)$  and  $\underline{x}(t)$  can be used interchangeably. Suppose that the optimal estimate is a linear function of the innovations process and has the form











Ant O Hill in Arra D (C) auf





المدانة والاللية

(2.18)

(2.3)

(2.22)

 $\frac{\hat{g}(t|t)}{t} = \int_{t_0}^{t} W(t,s) \underline{r}(s) ds$ 

where W(t,s) is to be chosen so that Equation (2.17) is satisfied. Form

$$\mathbb{E}[\underline{\hat{x}}(t|t)\underline{\Gamma}^{T}(\sigma)] = \int_{t_{0}}^{t} \mathbb{W}(t,s)\mathbb{E}[\underline{\Gamma}(s)\underline{\Gamma}^{T}(\sigma)] ds .$$

Using the orthogonality of the error in the estimate and the measurement data, one obtains

 $\mathbb{E}[\underline{\hat{x}}(t|t)\underline{r}^{\mathsf{T}}(\sigma)] \approx \mathbb{E}[\underline{x}(t)\underline{r}^{\mathsf{T}}(\sigma)] .$ 

From the white-noise property of \_r(s) , it follows that

$$\mathbf{E}[\underline{\mathbf{x}}(t)\underline{\mathbf{x}}^{\mathsf{T}}(\sigma)] = \mathbf{W}(t,\sigma)\mathbf{R}(\sigma) , \quad t_0 \leq \sigma < t . \quad (2.19)$$

Since  $R(\sigma)$  is positive-definite, the weighting matrix can be determined and the estimate is given by

$$\hat{\underline{g}}(t|t) = \int_{t_0}^{t} E[\underline{x}(t)\underline{r}^{\mathsf{T}}(s)]R^{-1}(s)\underline{r}(s) ds . \qquad (2.20)$$

The Kalman-Bucy filter equations presented in the prededing section are obtained from Equation (2.20) by differentiating with respect to t and by using Equation (1.1).

Differentiate Equation (2.20) to obtain

$$\frac{d}{dt} \hat{g}(t|t) = E[\underline{x}(t)\underline{r}^{\mathsf{T}}(t)]R^{-1}(t)\underline{r}(t) + \int_{t_0}^{t} \frac{d}{dt} E[\underline{x}(t)\underline{r}^{\mathsf{T}}(\mathbf{s})]R^{-1}(\mathbf{s})\underline{r}(\mathbf{s}) d\mathbf{s}$$

 $= E[\underline{x}(t)\underline{y}^{T}(t)]R^{-1}(t)\underline{y}(t) + F(t) \int_{t_{0}}^{t} E[\underline{x}(t)\underline{y}^{T}(s)]R^{-1}(s)\underline{y}(s) ds + \int_{t_{0}}^{t} E[\underline{y}(t)\underline{y}^{T}(s)]R^{-1}(s)\underline{y}(s) ds .$ 

But the wecond bern is equal to fit aft that it and the plant noise and residual are independent, so this reduces to

 $\frac{1}{2}(t|t) = P(t)\frac{1}{2}(t|t) + E[\frac{1}{2}(t)\chi^{T}(t)]R^{-1}(t)\chi(t)$ 

Letting

10

# $K(t) \stackrel{\Delta}{=} m[\underline{x}(t)\underline{x}^{T}(t)]R^{-1}(t)$ , (2.21)

the differential equation assumed in Section 2.1.1 is obtained:

$$\frac{1}{2}(t|t) = F(t)\frac{2}{2}(t|t) + K(t)[\frac{1}{2}(t) - H(t)\frac{2}{2}(t|t)]$$

Consider  $\mathbb{E}[\underline{x}(t)\underline{r}^{T}(t)]$  and observe that

 $\mathbb{E}[\underline{x}(t)\underline{r}^{T}(t)] = \mathbb{E}\{\underline{x}(t)[H(t)\underline{x}(t)] + \underline{v}(t)]^{T}\}$ 

 $= E\{[\underline{X}(t|t) + \underline{X}(t|t)]\underline{X}^{T}(t|t)\}H^{T}(t)$ 

 $= P(t|t)H^{T}(t)$ ,

since the estimate and error are orthogonal and  $P(t|t) \stackrel{\Delta}{=} E[\underline{X}(t|t)\underline{X}^{T}(t|t)]$ .

The error covariance matrix is specified by deriving the matrix differential equation that it must satisfy. This is accomplished by differentiating P(t|t) and using the gain established by Equations (2.21) and (2.22). These operations are similar to the discussion of Section 2.1.1 and, as expected, yield the result obtained there. More specifically, P(t|t) is the solution of the matrix Ricatti equation

 $\dot{P} = F(t)P + FF^{T}(t) - PH^{T}(t)R^{-1}(t)H(t)P + Q(t)$ , (2.12)

where it has been shown, using Equations (2.21) and (2.32), that

 $K(t) = P(t|t)H^{T}(t)R^{-1}(t)$  (2.11)

This completes the derivation of the Kalman-Bucy filter by the innovations approach.

There are many other aspects that could be considered in conjunction with the problem discussed in the preceding sections. Some of these will be treated in this section. In Section 2.1.2 the orthogonality of the optimal linear estimate and the associated error was used without proof and this will now be supplied. Also, the generalization to the case in which the plant and measurement noise processes are correlated is discussed, as is the effect of a deterministic forcing function in the plant model (e.g., arising from control system considerations).

# 2.1.3.1 Orthogonal projections and the Wiener-Hopf equation

The orthogonality condition (Eqn (2.17)) used in Section 2.1.2 is a special case of the geometric property of orthogonal projection. Since the estimate is a linear function of the measurement data, it is contained in the linear subspace spanned by these data. If the state vector is not contained in this subspace, it is clear that the error in the estimate will not vanish. In fact, the error will have its smallest mugnitude when the estimate is taken as the orthogonal projection of the state on the subspace spanned by the measurement data. This idea will be made more precise below and then applied to derive the well-known Wiener-Hopf equation and the orthogonality conditions used in the preceding section.

Consider a linear space X such that an inner product  $(\underline{x},\underline{y})$  is defined for any two elements  $\underline{x},\underline{y}$  in X. Define a norm by

 $||\underline{x}|| = \langle \underline{x}, \underline{x} \rangle^{\frac{1}{2}}$ 

Let M be a subspace of X and consider the problem of finding a vector  $\mathcal{Q}$  in M which minimizes  $||\underline{x}-\underline{y}||$  with respect to any  $\underline{y} \in M$ . The solution of this problem, if it exists, is given by the following result.

Orthogonal Projection Lemma:  $\|\underline{x} - \hat{\underline{y}}\|$  is a minimum for all  $\underline{y} \in M$ .

$$||\underline{x}-\underline{y}|| \ge ||\underline{x}-\underline{\hat{y}}||$$
 for all  $\underline{y} \in M$ ,

if and only if  $(\underline{x} - \hat{y})$  is orthogonal to all  $\underline{y} \in M$ 

 $\langle \underline{z} - \hat{\underline{y}}, \underline{y} \rangle = 0$  for all  $\underline{y} \in \mathbb{M}$ . (2.23)

Thus, the yeM can be regarded as the linear combination of the elements spanning M (e.g. the measurement ...data). This estimate yields an error that is orthogonal to all elements  $y \in M$ .

The proof of this result is straightforward. Assume that Equation (2.22) is valid. Then, for any  $y \in N$ ,

$$\begin{split} \|\underline{z} - \underline{y}\|^2 &= \|(\underline{z} - \underline{\hat{y}}) + (\underline{\hat{y}} - \underline{y})\|^2 \\ & \cdot &= \|\underline{z} - \underline{\hat{y}}\|^2 + 2\langle \underline{x} - \underline{\hat{y}}, \ \underline{\hat{y}} - \underline{y} \rangle + \|\underline{\hat{y}} - \underline{y}\|^2 \end{split}$$

But  $(\hat{y} - y) \in \mathbb{N}$ ; so, by Equation (2.23), the middle term vanishes and

with equality if and only if  $y = \hat{y}$ .

To complete the proof, assume that  $\frac{1}{2}$  minimizes  $||\underline{x} - \underline{y}||$  for all  $\underline{y} \in M$  and assume that there exists some  $\underline{y}_i \in M$  such that

$$\langle \underline{\mathbf{z}} - \underline{\hat{\mathbf{y}}}, \underline{\hat{\mathbf{z}}} \rangle = \alpha \neq 0.$$
  
$$\| \underline{\mathbf{x}} - \underline{\hat{\mathbf{y}}} - \beta \underline{\mathbf{y}}_1 \|^2 = \| \underline{\mathbf{z}} - \underline{\hat{\mathbf{y}}} \|^2 - 2\alpha\beta + \beta^2 \| \underline{\mathbf{y}}_1 \|^2.$$

Then

But, by appropriate choice of  $\beta$ , it is possible to make the last two terms negative, thereby contradicting the minimality of 9.

The orthogonal projection lemma can be used to derive the Wiener-Hopf equation and associated orthogonality conditions for the estimation problem described in Section 1.2. Let X be the n-dimensional state space and define the inner product of two elements in X to be

For convenience, it will be assumed that the variables all have zero mean. The norm of  $\underline{z} \in X$  is

 $||\mathbf{x}||^2 = \langle \mathbf{x}, \mathbf{x} \rangle = \mathbb{E}[\mathbf{x}^T \mathbf{x}] = \text{trace } \mathbb{E}[\mathbf{x}\mathbf{x}^T]$ .

11



A PROPERTY AND

and deside and the second second

and

From the orthogonal projection lemma, the trace of the error covariance matrix is minimized if

$$(\underline{x}(t) - \underline{\hat{x}}(t|t), \underline{x}(s)) = E[\underline{x}^{T}(t|t)\underline{x}(s)] = 0$$
, for  $t_{0} \leq s < t$ . (2.24)

Similarly, this lemma is used to establish Equation (2.17) and the other orthogonality conditions used in Section 2.1.2.

To derive the Wiener-Hopf equation for the nonstationary linear system of this presentation, suppose that an estimate of the state  $\underline{x}(t)$  is to be a linear function of the measurement data  $\underline{z}(t)$  having the form

$$\frac{\hat{\chi}}{t}(t|t) = \int_{t_0}^{t} W(t,s) \chi(s) \, ds$$
 (2.25)

Since the components of  $\hat{\underline{x}}(t|t)$  can be chosen independently of each other, it suffices to chose  $\hat{\underline{x}}(t|t)$  so that

 $E\{[x(t) - \hat{x}(t|t)]x^{T}(s)\} = 0, \quad t_{n} \le s \le t.$ 

Using Equation (2.25), one obtains the Wiener-Hopf equation

$$\mathbb{E}[\underline{x}(t)\underline{z}^{T}(s)] = \int_{t_{0}}^{t} \mathbb{W}(t,\sigma)\mathbb{E}[\underline{z}(\sigma)\underline{z}^{T}(s)] d\sigma \quad \text{for } t_{0} \leq s \leq t . \qquad (2.26)$$

2.1.3.2 Correlation between plant and measurement noise processes

It has been assumed heretofore that

$$\mathbb{E}[\underline{w}(t)\underline{y}^{T}(s)] = 0 \quad \text{for all } t, s.$$

In this section, consider the more general case in which

$$\mathbf{E}[\underline{\mathbf{y}}(t)\underline{\mathbf{y}}^{\mathrm{T}}(s)] = C(t)\delta(t-s) \quad \text{for all } t, s. \qquad (2.27)$$

Relative to the development of Section 2.1.1, note that the effect of the correlation appears first in the derivation of Equation (2.9). One sees, in this case, that

 $\mathbf{E}[\underline{\mathbf{y}}(t)\underline{\mathbf{x}}^{\mathrm{T}}(t|t)] = \frac{1}{2}\mathbf{R}(t)\mathbf{K}^{\mathrm{T}}(t) + \frac{1}{2}\mathbf{C}^{\mathrm{T}}(t)$ 

 $\mathbb{E}\left[w(t)\overline{t}^{T}(t|t)\right] = \frac{1}{2}Q(t) - \frac{1}{2}C(t)K^{T}(t) .$ 

Consequently, Equation (2.9) is found to be modified and the differential equation describing the error covariance matrix becomes

$$\frac{d}{dt} \mathbf{P} = [\mathbf{P}(t) - \mathbf{K}(t)\mathbf{H}(t)]\mathbf{P} \leftrightarrow \mathbf{P}[\mathbf{F}(t) - \mathbf{K}(t)\mathbf{H}(t)]^{T} + \mathbf{Q}(t) + \mathbf{K}(t)\mathbf{R}(t)\mathbf{K}^{T}(t) - \mathbf{K}(t)\mathbf{C}^{T}(t) - \mathbf{C}(t)\mathbf{K}^{T}(t)$$

$$= \mathbf{F}(t)\mathbf{P} + \mathbf{P}\mathbf{F}^{T}(t) + \mathbf{Q}(t) + \{\mathbf{K}(t)\mathbf{R}(t)\mathbf{K}^{T}(t) - \mathbf{K}(t)[\mathbf{H}(t)\mathbf{P} + \mathbf{C}^{T}(t)] - [\mathbf{P}\mathbf{H}^{T}(t) + \mathbf{C}(t)]\mathbf{K}^{T}(t)\} . \qquad (2.28)$$

The optimal gain matrix is found to be

 $K(t) = [P(t|t)H^{T}(t) + C(t)]R^{-1}(t)$  (2.29)

and the optimal error covariance is described by

 $\mathbf{P} = \pi - \mathbf{P}(t)\mathbf{P} + \mathbf{P}\mathbf{F}^{T}(t) = \left[\mathbf{P}\mathbf{H}^{T}(t) + \mathbf{C}(t)\right]\mathbf{R}^{-1}\left[\mathbf{C}^{T}(t) + \mathbf{H}(t)\mathbf{P}\right] + \mathbf{Q}(t) \quad .$ (2.30)

يحاط عاريه العربي المريحة ومدير ماريحة والمريحة والمريحة معاريحة معارية المريحة والمريحة والمريحة والمريحة والمواقعة والمواج

The same result is obtained using the innovations approach where Equation (2.22) is modified to account for the correlation.

2.1.3.3 Deterministic plant forcing function

Suppose that the plant model is altered to

$$\dot{z} = F(t)z + g(t) + T$$
, (2)

where  $\underline{d}$  is a known function. The only modification required to the filter equations in this case is to note that the prediction must account for  $\underline{d}(t)$ . Thus, the estimate is given by



31)

(2.32)

 $\frac{1}{2} = F(t)S + d(t) + K(t)[z(t) - H(t)S]$ 

where K(t) and P(tit) are unchanged from Equations (2.11) and (2.12).

2.1.3.4 Conditional mean and minimum variance estimates

The minimum variance estimate has an interesting interpretation in terms of the conditional density of the state  $\chi(t)$  given the measurement data  $\chi(\tau)$ . This relation is described in the following lemma.

Lemma: Suppose that a random variable  $\underline{x}$  is to be estimated from measurement data  $\underline{z}$  and suppose that  $\underline{x}$  and  $\underline{z}$  have the joint probability density function  $p(\underline{x},\underline{z})$ . An estimate  $\hat{\underline{x}}$  is to be determined from the data  $\underline{z}$  so that

 $E[(\chi - \hat{\chi})^{T} (\chi - \hat{\chi})] = \min \min .$ 

Thon, the minimum variance estimate 2 is

*Proof:* Write  $\mathbb{E}[(\underline{x}-\underline{z})^T(\underline{x}-\underline{z})]$  in terms of the conditional density, using the identity

 $\mathbb{E}\left[\left(\mathbf{x}-\hat{\mathbf{x}}\right)^{T}\left(\mathbf{x}-\hat{\mathbf{x}}\right)|\mathbf{Z}\right] = \hat{\mathbf{x}}^{T}\hat{\mathbf{x}} - 2\hat{\mathbf{x}}^{T}\mathbb{E}\left[\mathbf{x}|\mathbf{Z}\right] + \mathbb{E}\left[\mathbf{x}^{T}\mathbf{x}|\mathbf{Z}\right]$ 

$$\mathbf{E}[(\mathbf{x} - \hat{\mathbf{x}})^{\mathrm{T}}(\mathbf{x} - \hat{\mathbf{x}})] = \mathbf{E}\{\mathbf{E}[(\mathbf{x} - \hat{\mathbf{x}})^{\mathrm{T}}(\mathbf{x} - \hat{\mathbf{x}}) | \mathbf{z}]\}.$$

But

$$= (\hat{z} - \kappa[x|Z])^{T} (\hat{s} - \kappa[x|Z]) + \kappa[x^{T}z|Z] - \kappa^{T} [x|Z] \kappa[x|Z]$$
(2.34)

By definition this quantity is positive; so, to minimize  $\mathbb{E}[(\underline{x}-\underline{x})^T(\underline{x}-\underline{x})]$ , it is sufficient to minimize Equation (2.34). Only the first term involves  $\underline{x}$  and it is quadratic so that the smallest value it can assume is zero. This obtains when

so Equation (2.33) is proven.

Note also that the conditional mean provides an unbiased estimate of  $\underline{x}$ . This follows easily by recognizing that

$$\mathbb{D}[\underline{\hat{x}}] = \mathbb{E}\{\mathbb{E}[\underline{x}|\underline{z}]\} = \mathbb{E}[\underline{x}].$$

For the linear system (Eqns (1,1) - (1,2)) and the gaussian character of the initial state and the noise processes, it follows that  $\chi(t)$  is gaussian for any time t. Further, it can be shown that the conditional density of the state  $\chi(t)$ , given the data  $\chi(t)$ , is gaussian with mean value and covariance described by Equations (2.3), (2.11), and (2.12). Thus, the Kaiman-Bucy estimate describes the behavior of the mean and covariance of the conditional density function. As a result of this fact and the lemma proved above, the linear Kaiman-Bucy equations provide the best estimate, even when nonlinear estimators are considered. Probabilistic aspects of the estimation problem are considered in depth in Chapter 6.

#### 2.1.4 The Time-Discrete Problem

The time-discrete linear filtering and prediction problem<sup>11</sup> is included for the sake of completeness and results are stated without proof. More details can be found in References 19 and 20 or Chapter 6 of this book. One can use arguments that are very similar to those used in the preceding sections and the results contain few surprises. The principal difference resides in the fact that the measurement noise covariance matrix does not have to be positive-definite, although a related matrix must have this property. No attempt will be made here to treat the time-discrete problem exhaustively. Most results for the time-continuous case can be modified without difficulty to apply to this problem.

Consider a system whose state is described by a linear difference equation

$$\underline{\mathbf{x}}_{\mathbf{k}} = \Phi_{\mathbf{k}, \mathbf{k}-1} \underline{\mathbf{x}}_{\mathbf{k}-1} + \underline{\mathbf{x}}_{\mathbf{k}-1}, \quad \mathbf{k} = 1, 2, \dots$$
(2.35)

and which is observed through measurement data  $z_k$  obtained at discrete instants of time  $t_k$ . These data are assumed to be linearly related to the state according to

$$\mathbf{z}_{\mathbf{k}} = \mathbf{H}_{\mathbf{k}}\mathbf{x}_{\mathbf{k}} + \mathbf{y}_{\mathbf{k}} \,. \tag{2.36}$$

The noise sequences nie assumed to be gaussian and uncorrelated between sampling instants (i.e. white-noise sequences) with zero means and covariances:







$$\begin{split} \mathbf{E} \left[ \underline{\mathbf{y}}_{\mathbf{k}} \underline{\mathbf{y}}_{\mathbf{j}}^{\mathsf{T}} \right] &= \mathbf{Q}_{\mathbf{k}} \delta_{\mathbf{k} \mathbf{j}} \\ \mathbf{E} \left[ \underline{\mathbf{y}}_{\mathbf{k}} \underline{\mathbf{y}}_{\mathbf{j}}^{\mathsf{T}} \right] &= \mathbf{R}_{\mathbf{k}} \delta_{\mathbf{k} \mathbf{j}} \\ \mathbf{E} \left[ \underline{\mathbf{y}}_{\mathbf{k}} \underline{\mathbf{y}}_{\mathbf{j}}^{\mathsf{T}} \right] &= \mathbf{C}_{\mathbf{k}} \delta_{\mathbf{k} \mathbf{j}} \end{split}$$

The  $\chi_k$  and  $\chi_k$  are assumed to be uncorrelated with the initial state  $\chi_0$  , which is a gaussian random variable with

# $\mathbb{E}[X_0] = X_0$

For this system one can prove that the unbiased, min.  $\cdot$  variance estimate of  $\underline{x}_k$ , given the data  $\underline{Z}(k)$ , is described by the following system. The estimate of the state  $x_k$  is

$$\hat{\mathbf{x}}_{\mathbf{k}}' = \hat{\mathbf{q}}_{\mathbf{k},\mathbf{k}-1} \hat{\mathbf{x}}_{\mathbf{k}-1} + \mathbf{A}_{\mathbf{k}-1} [\mathbf{z}_{\mathbf{k}-1} - \mathbf{H}_{\mathbf{k}-1} \hat{\mathbf{x}}_{\mathbf{k}-1}']$$
(2.37)

before the measurement  $\underline{x}_k$  is processed. The estimate when  $\underline{x}_k$  is included is obtained by modifying the pre-dicted estimate  $\frac{d}{dx}_k$  according to

$$\frac{1}{2k} = \frac{1}{2k} + K_{\rm L} \left[ \frac{1}{2k} - H_{\rm L} \frac{2}{2k} \right] , \qquad (2.38)$$

The optimal gains  $A_{k-1}$  and  $K_k$  are given by

$$A_{k-1} = C_{k-1} \left[ H_{k-1} P_{k-1}^{\prime} H_{k-1}^{\prime} + R_{k-1} \right]^{-1}$$
(2.39a)

$$K_{L} = P_{L}^{\prime} H_{L}^{T} [H_{L} P_{L}^{\prime} H_{L}^{T} + R_{L}]^{-1}$$
, (2.39b)

where the predicted error covariance matrix is described by

$$P_{k}^{T} = \Phi_{k,k-1}P_{k-1}\Phi_{k,k-1}^{T} + Q_{k-1} - \tilde{\Phi}_{k,k-1}K_{k-1}C_{k-1}^{T} - \\ - C_{k-1}K_{k-1}^{T}\Phi_{k,k-1}^{T} - A_{k-1}[H_{k-1}P_{k-1}^{T}H_{k-1}^{T} + R_{k-1}]^{-1}A_{k-1}^{T}$$
(2.40)

and the current error covariance matrix is

$$\mathbf{P}_{\mathbf{k}} = \mathbf{P}_{\mathbf{k}}' - \mathbf{K}_{\mathbf{k}}\mathbf{H}_{\mathbf{k}}\mathbf{P}_{\mathbf{k}}' . \tag{2.41}$$

When the plant and measurement noise are uncorrelated, the matrix  $C_{\rm b}$  vanishes and causes  $A_{\rm b}$  to vanish and the covariance matrices reduce to

$$\mathbf{P}_{k}^{\prime} = \Phi_{k,k-1} \mathbf{P}_{k-1} \Phi_{k,k-1}^{T} + \mathbf{Q}_{k-1}$$
(2.42)

$$P_{k} = P_{k}' + K_{k}H_{k}P_{k}' . \qquad (2.43)$$

Note that the gain matrix requires the inversion of  $[H_k P_k' H_k^T + R_k]$  instead of the measurement covariance matrix  $R_k$ , as occurs in the time-continuous case. However, one can derive a form for  $K_k$  that is more similar to Section 2.1.1 when  $P_k'$  and  $R_k$  are positive-definite by making use of the following lemma.

Matrix Inversion Lemma: Suppose the nxn matrix B and the mxm matrix R are positive-definite and let H be an arbitrary mxn matrix. If A is given by

> A = B - BHT [HBHT + R] "1 HB . (2.44)

then the inverse of A is

and

$$A^{-1} = B^{-1} + H^{T} R^{-1} H . \qquad (2.45)$$

The proof follows by multiplication of A and  $A^{-1}$  .

The error covariance matrix can be rewritten, by substituting Equation (2.43) into Equation (2.43),

 $\mathbf{P}_{\mathbf{k}} = \mathbf{P}_{\mathbf{k}}^{\prime} - \mathbf{P}_{\mathbf{k}}^{\prime} \mathbf{H}_{\mathbf{k}}^{\mathsf{T}} [\mathbf{H}_{\mathbf{k}}^{\mathsf{T}} \mathbf{P}_{\mathbf{k}}^{\mathsf{T}} \mathbf{H}_{\mathbf{k}}^{\mathsf{T}} + \mathbf{R}_{\mathbf{k}}^{\mathsf{T}}]^{-1} \mathbf{H}_{\mathbf{k}}^{\mathsf{T}} \mathbf{P}_{\mathbf{k}}^{\prime} \ .$ (2.46)







14

(2.47)

Using the lemma, one obtains

$$P_{k}^{-1} = (P_{k}^{\prime})^{-1} + H_{k}^{T}R_{k}^{-1}H_{k}$$

The gain matrix can also be modified:

$$K_{\mu} = (P_{\mu} P_{\mu}^{-1}) P_{\mu}^{\prime} H_{\mu}^{T} R_{\mu}^{-1} [H_{\mu} P_{\mu}^{\prime} H_{\mu}^{T} R_{\mu}^{-1} + 1]^{-1}$$

But, from Equation (2.43),

80

$$P_{k}^{-1}P_{k}' = I + H_{k}^{T}R_{k}^{-1}H_{k}P_{k}',$$

$$K_{k} = P_{k}\left[I + H_{k}^{T}R_{k}^{-1}H_{k}P_{k}'\right]H_{k}^{T}R_{k}^{-1}\left\{I + H_{k}P_{k}'H_{k}^{T}R_{k}^{-1}\right\}^{-1}$$

$$= P_{k}H_{k}^{T}R_{k}^{-1},$$
(2.48)

Equations (2.47) and (2.48) provide an alternative form when  $P'_k$  and  $R'_k$  are positive-definite. Equation (2.47) has the disadvantage that an n×n matrix must be inverted, rather than the m×m matrix that must be inverted in Equation (2.42) (usually, m < n).

## 2.2 The Colored-Noise Problem

2.2.1 Colored-Noise, Shaping Filters, and State Vector Augmentation

It has been assumed that the plant and measurement noise processes are gaussian and white. This assumption is not always satisfied in practice, so it is desirable to consider linear systems in which the noise exhibits correlation between different instants of time (i.e. the noise is "colored"). This problem is successfully approached when the noise can be described by a shaping filter<sup>21</sup>.

Definition: Consider a gaussian random process  $\underline{n}(t)$  that has zero mean and whose second-order correlations are given by

$$\mathbb{E}[\underline{n}(t)\underline{n}^{T}(\tau)] = D(t,\tau) \quad (2.49)$$

A linear dynamical system driven by a gaussian white-noise process whose output has the same statistical characteristics as  $\underline{n}(t)$  is called a shaping filter.

Thus, by introducing a shaping filter, many random processes n(t) can be described by

where A(t) and the statistics of the white-noise  $\underline{w}(t)$  are chosen so that  $\underline{n}(t)$  has the prescribed statistics. The problem of determining shering filters for random processes will not be treated here, but the filter development for those processes describable by Equation (2.50) will be given below.

Consider the following linear system (analogous to Equations (1.1) and (1.2)), in which colored noise exists in the plant and measurement systems. To distinguish between white and colored-noise components, consider the following partitioned system.

$$\begin{bmatrix} \mathbf{X}_1 \\ \mathbf{X}_2 \end{bmatrix} = \begin{bmatrix} \mathbf{F}_1(\mathbf{t}) & \mathbf{F}_2(\mathbf{t}) \\ \mathbf{F}_3(\mathbf{t}) & \mathbf{F}_4(\mathbf{t}) \end{bmatrix} \begin{bmatrix} \mathbf{X}_1 \\ \mathbf{X}_2 \end{bmatrix} + \begin{bmatrix} \mathbf{Y}_1 \\ \mathbf{X}_2 \end{bmatrix}$$
(2.51)

where mi is a white-noise process

🛃 is a colored-noise process

where

and Mg is a white-noise process.

The measurement data are described by

$$\begin{bmatrix} \mathbf{x}_{1} \\ \mathbf{x}_{2} \end{bmatrix} = \begin{bmatrix} \mathbf{H}_{1}(\mathbf{t}) & \mathbf{H}_{2}(\mathbf{t}) \\ \mathbf{H}_{3}(\mathbf{t}) & \mathbf{H}_{4}(\mathbf{t}) \end{bmatrix} \begin{bmatrix} \mathbf{x}_{1} \\ \mathbf{x}_{2} \end{bmatrix} + \begin{bmatrix} \mathbf{y}_{1} \\ \mathbf{x}_{2} \end{bmatrix}, \qquad (2.53)$$



1 Joenny



(2.52)

where y, is a white-noise process

where

$$\dot{\xi}_{a} = B(t)\xi_{a} + g_{a}$$

and y, is a white-noise process.

The problem described by Equations (2.51) = (2.54) can be reformulated by state vector augmentation to obtain a system in which the state variables  $z_1$  and  $z_2$  are combined with the colored-noise variables  $\xi_1$  and  $\xi_2$ to define a system in which only white-noise appears explicitly. This system is

[i]		F <sub>1</sub> (t)	$F_2(t)$	Û	• ]		"H1	
Å2	_	F,(t)	F <sub>4</sub> (t)	I	o	X2	O	
121	*	0	D	A(t)	o	<u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u></u>	₩.,	(2.058)
22		0	o	0	B(t)	5,		

with measurement data

$$\begin{bmatrix} \underline{\mathbf{z}}_{1} \\ \underline{\mathbf{z}}_{8} \end{bmatrix} \begin{bmatrix} \mathbf{H}_{1}(\mathbf{t}) & \mathbf{H}_{8}(\mathbf{t}) & \mathbf{0} & \mathbf{0} \\ \mathbf{H}_{3}(\mathbf{t}) & \mathbf{H}_{4}(\mathbf{t}) & \mathbf{0} & \mathbf{I} \end{bmatrix} \begin{bmatrix} \underline{\mathbf{z}}_{1} \\ \underline{\mathbf{z}}_{2} \\ \mathbf{\underline{x}}_{1} \\ \mathbf{\underline{x}}_{1} \end{bmatrix} \begin{bmatrix} \underline{\mathbf{y}}_{1} \\ \mathbf{0} \end{bmatrix}.$$
(2.56a)

or more succinctly

with similar definitions for F(t), H(t),  $\underline{W}$ , and  $\underline{V}$ . Thus, it appears that the problem has been reformulated to the model given initially by Equations (1.1) and (1.2), so that the solutions already obtained can be applied. Unfortunately, this is not true because the zero components appearing in the measurement noise vector  $\underline{\mathbf{y}}$  prevent the covariance matrix  $\mathbf{R}(\mathbf{t})$  from being positive-definite. This property is required in order to form the optimal gain matrix K(t), since R<sup>-1</sup>(t) is required. Thus, shaping filters and state vector sugmontation permit the plant equation to be rewritten in a manner that is compatible with our previous results (i.s. the plant covariance matrix Q does not have to be inverted) but additional consideration must be given to colored measurement noise.

≙ [x[,x],€[,€]]

# 2.2.2 Solution for Colored Measurement Noise

The components of the measurement vector which contain only colored-noise provide the source of the difficulty in applying the filter equations developed earlier. This problem was first discussed by  $\cos^{24}$  and Bryson and Johansen<sup>23</sup> and has more recently been considered by  $Bucy^{24}$ , Stear and Stubberud<sup>24</sup>, and Sarachik<sup>24</sup>. To develop a solution for this case, assume that the model has the form

$$\underline{1} = F(t)\underline{1} + \underline{w}$$
 (2.57)

and

 $\begin{bmatrix} \mathbf{H}_1 \\ \mathbf{H}_2 \end{bmatrix} = \begin{bmatrix} \mathbf{H}_1(\mathbf{t}) \\ \mathbf{H}_2(\mathbf{t}) \end{bmatrix} \mathbf{X} + \begin{bmatrix} \mathbf{Y} \\ \mathbf{0} \end{bmatrix} ,$ 

is an (n + p) dimensional state vector whore I

> is an m-dimensional measurement vector containing additive white-noise **Z**1

is a p-dimensional noise-free measurement vector. ā.

This model is motivated by Equations (2.55) and (2.56), although F(t),  $H_1(t)$ , and  $H_2(t)$  will not be required to have the form given there. The white-noise  $\underline{y}$  is assumed to have a positive-definite covariance matrix R(t) and the  $H_1$  and  $H_2$  are assumed to have maximal rank.

**MANYA** 



(2.54)

(2.58)

· 17

the manipulation by

B. C. Mark In Marin (C)

There are several aspects of this problem that require mention. First observe that one might solve the problem formally by differentiating the measurement data  $z_p$ :

$$\dot{Z}_{2} = H_{2}\dot{\xi} + \dot{H}_{2}\dot{\chi}$$
  
=  $(H_{2}F + \dot{H}_{2})\chi + H_{2}\chi$  (2.50)

If  $H_{g}QH_{1}^{T}$  (i.e. the covariance matrix of the noise vootor  $H_{g}W_{1}$  is positive-definite, then  $\frac{1}{2}_{2}$  can be treated as the measurement data in place of  $Z_{2}$  since they are equivalent if the initial conditions on  $\chi$  are selected appropriately. Now the Kalman-Bucy results can be applied using the measurement data  $Z_{1}$  and  $\frac{1}{2}_{2}$ . If  $H_{g}QH_{2}^{T}$ has rank r < p, then one can transform the  $\frac{1}{2}_{2}$  into a vector that is separated into components either with or without white-noise. The noise-free components can be differentiated again in an strompt to introduce whitenoise. The procedure of separation and differentiation can be repeated until a set of p measurements containing white-noise are generated to replace  $Z_{2}$ . If it proves to be imponsible to introduce white-noise into every component, then these "perfect" measurements can be dealt with an described by Bryson and Johansen<sup>23</sup>.

Note, however, that the noise-free measurements in Equation (2.58) provide perfect knowledge of p variables, so that one would expect that the filter equations would only involve a variables instead of the (n + p) appearing in the state vector of Equation (2.57). This permits a desirable relaxation of computational requirements. Furthermore, it is undesirable in practice to differentiate data, so these two disadvantages motivate one to develop other means for dealing with colored measurement noise.

Using the definition of  $\underline{\mathbf{g}}_{j}$  and the assumption that  $\mathbf{H}_{j}$  has maximal rank, it is possible to define an  $(\mathbf{n}-\mathbf{p})\times\mathbf{n}$  matrix  $\mathbf{H}_{j}$  such that a nonsingular matrix  $\mathbf{T}(\mathbf{t})$ , with  $\mathbf{t} \geq \mathbf{t}_{0}$ , can be formed.

The H<sub>5</sub> must be selected so that

and

 $H_g H_3^T = 0$  $H_a H_4^T = 1.$ 

The task of choosing H, is not as difficult as it may at first appear; at least it is trivially accomplished for an important class of systems that are discussed later in this section.

Let the inverse of T be defined as

$$T^{-1}(t) = [J_{1}(t) J_{2}(t)], \quad (2.61)$$

where it can easily be shown that

 $J_{g} = H_{2}^{T} (H_{g} H_{g}^{T})^{-1}$  $J_{3} = H_{3}^{T}$ 

when the constraints on H, are invoked.

Define new state variables with subvectors g, and 🔮 such that

$$\begin{bmatrix} \underline{\mathbf{x}}_2 \\ \underline{\mathbf{x}}_1 \end{bmatrix} \stackrel{\circ}{=} \begin{bmatrix} \mathbf{H}_2 \\ \mathbf{H}_3 \end{bmatrix} \mathbf{x} = \mathbf{T} \mathbf{x} . \qquad (2.62)$$

Since T has an inverse, the  $\underline{x}$  can be recovered,

$$\mathbf{I} = J_2 \mathbf{I}_2 + J_3 \mathbf{I}_2 \,. \tag{2.63}$$

But  $\underline{x}_{i}$  is known measurement data, so the problem of obtaining the minimum variance estimate of  $\underline{x}$  reduces to that of estimating  $\underline{c}$ . As a result, the order of the filter that is required reduces, as expected, from (n+p) to n. After the estimate of  $\underline{c}$  is determined, the estimate of the state  $\underline{x}$  is obtained from

$$\frac{2}{3}(t|t) = J_1(t)Z_2(t) + J_2(t)\hat{C}(t|t)$$
 (2.64)

Consider the problem of estimating the vector 🔮 . From Equation (2.62), it is seen that

so a differential equation describing 🛫 is obtained by differentiating and substituting Equation (2.57):

$$\xi^{2} = \hat{H}_{3} \chi + \hat{H}_{3} \chi$$

$$= (\hat{H}_{3} + \hat{H}_{3}F)\chi + \hat{H}_{3} \chi$$

$$= [(\hat{H}_{3} + \hat{H}_{3}F)J_{3}]\xi^{2} + (\hat{H}_{3} + \hat{H}_{3}F)J_{3}\chi_{3} + \hat{H}_{3} \chi \qquad (2.68)$$

The term  $(\dot{H}_3 + H_3F)J_2Z_2$  is a known forcing function so it can be treated in the fashion described in Section 2.1.3.3.

The measurement data  $z_1$  can also be expressed in terms of  $\leq z_1$ 

$$\underline{z}_{1} = H_{1}J_{2}\overline{z}_{2} + H_{1}J_{3}\underline{\zeta} + \underline{v} .$$

$$\underline{z} \stackrel{\Delta}{=} \underline{z}_{1} - H_{1}J_{2}\overline{z}_{2} ; \qquad (2.60)$$

$$\underline{g} = (H_{1}J_{3})\underline{\zeta} + \underline{v} . \qquad (2.67)$$

Let

1.197.2

2

Equations (2.65), (2.66), and (2.67) form a system to which the proviously derived filter equations can be applied, since  $\underline{\mathbf{y}}$  is assumed to have a positive-definite covariance matrix. One point remains to be defined. Ordinarily, statistics are preacribed for  $\underline{\mathbf{x}}(t_0)$ , but in this application it is necessary to specify statistics for  $\underline{\mathbf{x}}(t_0)$ . To accomplish this, e that the measurement  $\underline{\mathbf{x}}(t_0)$  is known; so it can be used to estimate  $\underline{\mathbf{x}}(t_0)$  and, through Equation (2.6. to obtain initial conditions for  $\underline{\mathbf{x}}(t_0)$ . Using the remults for time-discrete filtering presented in Sec ion 2.1.4, one sees that the statistics of  $\underline{\mathbf{x}}(t_0)$ , based on the *a priori* statistics for  $\underline{\mathbf{x}}(t_0)$  and the noise-free measurement data  $\underline{\mathbf{x}}_2(t_0)$ , are

$$\begin{aligned} \frac{\mathcal{L}}{2}(t_0) &= H_3(t_0)\underline{R}(t_0)t_0) \\ &= H_3(t_0)P(t_0)H_2^T(t_0)[H_1(t_0)P(t_0)H_2^T(t_0)]^{-1}\underline{Z}_2(t_0)], \end{aligned}$$
(2.68)

with error covariance matrix

$$\Pi(t_{o}) = H_{a}(t_{o}) \left[ P(t_{o}) - K(t_{o}) H_{a}(t_{o}) P(t_{o}) \right] H_{a}^{T}(t_{o}) , \qquad (2.69)$$

where the gain matrix is given by

$$K(t_{o}) = P(t_{o})H_{a}^{T}(t_{o})[H_{a}(t_{o})P(t_{o})H_{a}^{T}(t_{o})]^{-1}$$
(2.70)

Naturally, it is necessary to assume that the inverse of  $H_2(t_0) P(t_0)H_2^T(t_0)$  exists for these relations to be valid. Since  $H_2$  has already been assumed to have maximal rank, it is sufficient to require that  $P(t_0)$  be positive-definite, although this is not a mecessary condition.

Using the initial conditions (Eqns (2.68) and (2.69)), the minimum variance estimate of  $\xi(t)$  based on the data  $(g_1 - H_1J_2 Z_2)$  is obtained by applying Equations (2.32), (2.11) and (2.12) (or Equations (2.32), (2.29) and (2.30) if the plant and measurement noises are correlated). There are no differentiations of measurement data required in this approach and the order of the filter has been reduced from (n + p) to n. Of course, this solution requires that there exist some measurements containing white-noise. If the vector  $\underline{x}_1$  in Equation (2.58) does not exist, additional manipulations must be introduced, as now discussed.

Suppose that the plant is described by Equation (2.57) but that the measurements are entirely noise-free (i.e. there is no white-noise in the data) and described by

 $\underline{x}_{g} = H_{g}(t)\underline{x}, \qquad (2.71)$ 

where the subscript : has been retained to be consistent with Equation (2.58). To obtain a form that can be treated by the above procedure, differentiate  $z_2$  to obtain

$$\dot{\mathbf{x}}_2 = \mathbf{H}_2 \dot{\mathbf{x}} + \mathbf{H}_2 \mathbf{x}$$
  
=  $(\mathbf{H}_2 \mathbf{F} + \dot{\mathbf{H}}_2) \mathbf{x} + \mathbf{H}_2 \mathbf{x}$ . (2.72)

To simplify the discussion, it shall be assumed that the covariance  $H_2 QH_2^T$  of the noise  $H_2 w$  is positivedefinite. The differentiated data  $\dot{z}_2$  can now be treated as additional measurement data, so that the complete system has the form

<u>i</u> = F(t)x + ¥

$$\mathbf{z}_1 \stackrel{\text{def}}{=} \dot{\mathbf{z}}_2 = [\mathbf{H}_2(\mathbf{t})\mathbf{F}(\mathbf{t}) + \dot{\mathbf{H}}_2(\mathbf{t})]\mathbf{X} + \mathbf{H}_2(\mathbf{t})\mathbf{W}$$

 $\mathbf{\bar{g}}_{2} = \mathbf{H}_{2}(\mathbf{t})\mathbf{\underline{x}}$ 

and now the preceding results can be applied without additional modification.

From Equation (2.66), the measurement data used in the filtor will be

$$\mathbf{E} = \mathbf{\dot{\mathbf{x}}}_2 - [\mathbf{H}_2\mathbf{F} + \mathbf{\dot{\mathbf{H}}}_2]\mathbf{J}_2\mathbf{z}_2 , \qquad (3.73)$$

where J, is defined in Equation (2.81). By Equation (2.67), the 👔 is represented in terms of 🔮 as

$$\mathbf{z} = [\mathbf{H}_{g}\mathbf{r} + \dot{\mathbf{H}}_{g}]\mathbf{J}_{g}\mathbf{\xi} + \mathbf{H}_{g}\mathbf{y}$$
 (2.74)

The filter state f is still described by Equation (2.65). The plant and measurement noises are correlated with covariance matrix

$$O(t)\delta(t-\tau) \stackrel{\text{\tiny (1)}}{=} \mathbb{E}[H_{g}(t)\underline{w}(t)\underline{w}^{T}(\tau)H_{g}^{T}(\tau)]$$

$$O(t)\delta(t-\tau) = H_{g}(t)H_{g}^{T}(t)\delta(t-\tau) , \qquad (2.75)$$

The initial conditions are still described by Equations (2.68) and (2.69) and Equations (2.32), (2.29) and (2.30) must be used for the filtering. Note also that the plant and measurement noise covariances for this system are  $H_g G I_3^T$  and  $H_g G I_4^T$ , respectively.

From Equation (2.32), the estimate of  $\xi$  is given by

$$\frac{1}{2} = 0J_{3}\hat{c} + 0J_{2}Z_{2} + K[Z - JJ_{3}\hat{c}]$$
, (2.76)

where

The g given by Equation (2.74) appears to indicate that a differentiation  $\frac{1}{2}$  is required in order to obtain an estimate of  $\leq$ . This requirement can be circumvented computationally by resorting to the following artifice. Define

so that

 $\dot{\xi}^{+} = \dot{\xi} - K\dot{g}_{g} - \dot{K}g_{g}$  $= G_{g}J_{g}\dot{\xi} + GJ_{g}g_{g} + K[g - JJ_{g}\dot{\xi}] - K\dot{g}_{g} - \dot{K}g_{g} .$ 

Using Equation (2.73), this reduces to

and does not require the differentiation of data. The gain matrix X must be differentiated, however.

There is a special case<sup>34,23</sup> that is sufficiently important that it will be considered here and will complete the discussion of colored measurement noise. In the following the notation will be changed somewhat from that used elsewhere in this section, in order to deal not with the sugmented state implied by Equation (2.57) but with a vector that is more closely skin to the state of the physical system. Let the system be described by

$$\dot{z} = F(t) \underline{z} + \underline{z}$$
 (2.78)

$$\vec{z} = H(t)\vec{z} + \vec{u}$$
, (2.79)

where n is a zero mean colored-noise process described by a shaping filter.

$$i' = A(t)\underline{u} + \underline{v}$$
, (2.80)

and with initial covariance  $E[\underline{n}(t_0)\underline{n}^T(t_0)] = N$  where  $\underline{y}$  has a positive-definite covariance matrix. Assume, also, that  $\underline{y}$  and  $\underline{y}$  are uncorroluted. This is essentially the problem considered immediately above, except that  $\underline{x}$  is an n-Gimensional state unaugmented by measurement noise variables. To put this system in a form compatible with Equations (2.57) and (2.71), rewrite Equations (3.78) = (2.80) as

 $\begin{bmatrix} \dot{x} \\ \dot{n} \end{bmatrix} = \begin{bmatrix} F(t) & 0 \\ 0 & A(t) \end{bmatrix} \begin{bmatrix} \underline{x} \\ \underline{n} \end{bmatrix} + \begin{bmatrix} \underline{y} \\ \underline{y} \end{bmatrix}, \qquad (2.81)$  $E\left\{ \begin{bmatrix} \underline{y} \\ \underline{y} \end{bmatrix} \begin{bmatrix} \underline{y}^{T} \underline{y}^{T} \end{bmatrix} \right\} = \begin{bmatrix} Q & C \\ C^{T} & R \end{bmatrix} \delta(t-\tau)$  $\underline{x} = \begin{bmatrix} H(t) I \end{bmatrix} \begin{bmatrix} \underline{x} \\ \underline{n} \end{bmatrix}. \qquad (2.82)$ 

where

and

For this problem the definition of the matrix  $H_{j}$  required in the transformation T of Equation (2.60) can be defined explicitly as

 $H_{1} = \{I \ 0\},\$ 

so that transformation T is

$$T(t) = \begin{bmatrix} H & I \\ I & 0 \end{bmatrix} \stackrel{A}{=} \begin{bmatrix} H_2 \\ H_3 \end{bmatrix}$$
(2.83)  
$$T^{-1}(t) = \begin{bmatrix} 0 & I \\ I & -H \end{bmatrix} \stackrel{A}{=} \begin{bmatrix} J_2 & J_3 \end{bmatrix}$$
(2.84)

With this transformation, the filtering state  $\xi$  is

ć = 1,

so that  $\underline{x}$  is to be estimated directly. Obviously  $\underline{x}$  is described by Equation (2.78), although it can be verified that Equation (2.65) reduces to this equation by substituting appropriately. The measurement given by Equation (2.66) is seen to be

$$\frac{1}{2} - A(t) = [HP - AH + \dot{H}] + H + \dot{H} + \dot{Y}.$$
 (2.85)

The estimate for  $\underline{x}$  can be written explicitly in the following manner.

 $\dot{\hat{x}} = F(t)\hat{x} + K(t)[\dot{z} - A(t)z - J(t)\hat{x}], \qquad (2.86)$ 

where

The initial condition for  $f(t_n)$  is obtained from Equation (2.88) and is found to be

$$\frac{1}{2}(t_0) = M(t_0) H^{T}(t_0) [H(t_0) M(t_0) H^{T}(t_0) + N(t_0)]^{-1} \underline{Z}(t_0) , \qquad (2.87)$$

where  $M(t_n)$  is the covariance of the initial state  $\underline{x}(t_n)$  .

The gain matrix K(t) is given by

$$K(t) = [P(t)J^{T}(t) + q(t)H^{T}(t)][H(t)Q(t)H^{T}(t) + R(t)]^{-1}$$
(2.88)

and the error covariance matrix is

$$\dot{P} = F(t)P + F^{T}(t)P + Q(t) - K(t)[H(t)Q(t)H^{T}(t) + R(t)]K^{T}(t) . \qquad (2.89)$$

The i simi conditions are determined from Equation (2.69) to be

$$P(t_{0}) = M(t_{0}) - M(t_{0})R^{T}(t_{0})[H(t_{0})M(t_{0})R^{T}(t_{0}) + N(t_{0})]^{-1}H(t_{0})M(t_{0}).$$
(2.90)

These results reduce to those published by Bucy<sup>24</sup> and by Stear and Stubberud<sup>23</sup> after minor changes in the model and notation are introduced.

# 2.3 Linear Smoothing

In this section the problem of finding the linear, unbiased, minimum variance estimate of  $\chi(t)$ , given the data  $\underline{Z}(\tau)$ ,  $\tau > t$ , is considered. The system defined by Equations (1.1) - (1.2) is treated and the innovations approach of Section 2.1.2 is used to derive the solution. The development follows Kailath27.

Based on the results of auction 2.1.2, suppose that the smoothed estimate is given by

$$\underline{\hat{g}}(t|\tau) = \int_{t_0}^{\tau} W_{g}(t,s) \underline{r}(s) ds , \qquad (2.91)$$

where  $\pi_{a}$  is chosen so that the error in the estimate is orthogonal to the innovations process  $\underline{r}(s)$ 

$$\mathbf{E}\left\{\left[\underline{\mathbf{x}}(\mathbf{t}) - \underline{\mathbf{\hat{x}}}(\mathbf{t}|\tau)\right], \underline{\mathbf{\hat{r}}}(\sigma)\right\} = \mathbf{0}, \quad \mathbf{t}_{\mathbf{0}} \leq \sigma < \tau.$$

From the orthogonality property, one obtains the condition that

$$E[\underline{\chi}(t)\underline{\Gamma}^{T}(\sigma)] = \int_{t_{0}}^{\tau} W_{g}(t,s)E[\underline{\chi}(s)\underline{\Gamma}^{T}(\sigma)] ds$$
  
=.  $W_{g}(t,\sigma)R(\sigma)$ . (2.92)

But this implies that

$$\frac{1}{2}(t|\tau) = \int_{t_0}^{\tau} \mathbb{E}[\underline{x}(t)\underline{r}^{\mathrm{T}}(s)]\mathbf{R}^{-1}(s)\underline{r}(s) ds$$
$$= \int_{t_0}^{t} \mathbb{E}[\underline{x}(t)\underline{r}^{\mathrm{T}}(s)]\mathbf{R}^{-1}(s)\underline{r}(s) ds + \int_{t}^{\tau} \mathbb{E}[\underline{x}(t)\underline{r}^{\mathrm{T}}(s)]\mathbf{R}^{-1}(s)\underline{r}(s) ds . \qquad (2.93)$$

The first term can be recognized as the filtered estimate; so Equation (2.93) reduces to

$$\underline{\hat{g}}(t|\tau) = \underline{\hat{g}}(t|t) + \int_{t}^{T} \mathbb{E}[\underline{x}(t)\underline{r}^{T}(s)] \mathbb{R}^{-1}(s)\underline{r}(s) ds \qquad (2.94)$$

Now consider  $E[\underline{x}(t)\underline{r}^{T}(s)]$ . From Equation (2.92), and assuming that  $\sigma > t$ ,

$$W_{\mathbf{x}}(\mathbf{t},\sigma)\mathbf{R}(\sigma) = \mathbf{E}[\mathbf{x}(\mathbf{t})\mathbf{r}^{\mathrm{T}}(\sigma)]$$

- $= \mathbf{E}[\underline{\mathbf{x}}(\mathbf{t})\underline{\mathbf{x}}^{\mathsf{T}}(\boldsymbol{\sigma}|\boldsymbol{\sigma})\mathbf{H}^{\mathsf{T}}(\boldsymbol{\sigma}) + \underline{\mathbf{x}}(\mathbf{t})\underline{\mathbf{y}}^{\mathsf{T}}(\boldsymbol{\sigma})]$
- =  $\mathbf{E}[\mathbf{x}(\mathbf{t})\mathbf{x}^{T}(\sigma|\sigma)]\mathbf{H}^{T}(\sigma)$
- =  $\mathbb{E}\left[\left[\frac{1}{2}(t|t) + \frac{1}{2}(t|t)\right] \frac{1}{2}^{T}(\sigma|\sigma)\right] H^{T}(\sigma)$
- $= \mathbf{P}(\mathbf{t},\sigma)\mathbf{H}^{\mathsf{T}}(\sigma) , \quad \sigma > \mathbf{t} ,$

where

# $\mathbf{P}(\mathbf{t},\sigma) \stackrel{\Delta}{=} \mathbf{E}[\underline{\mathbf{x}}(\mathbf{t}|\mathbf{t})\underline{\mathbf{x}}^{\mathrm{T}}(\sigma|\sigma)] .$

Using this definition, the smoothed estimate becomes

盇

$$\underline{\hat{g}}(t|\tau) = \underline{\hat{g}}(t|t) + \int_{t}^{\tau} P(t,s) H^{T}(s) \underline{r}(s) \underline{r}(s) ds . \qquad (2.98)$$

This shows that  $\hat{\mathbf{x}}(t|\tau)$  is the linear combination of the filtered estimate  $\hat{\mathbf{x}}(t|t)$  and a correction term that contains the data not included in  $\hat{i}(t | \hat{f})$  .

The error covariance matrix  $P(t|\tau) \stackrel{B}{\to} E[\underline{x}(t|\tau)\underline{x}^T(t|\tau)]$  can be defined in terms of the filtering error covariance ance matrix P(t|t). Observe that the error in the smoothed estimate is

 $\underline{\vec{X}}(t|\tau) = \frac{\mathbf{a}_{t}}{\mathbf{B}_{t}} \underbrace{\vec{a}}_{t}(t|t) = \int_{t}^{\tau} \mathbf{P}(t,s) \mathbf{H}^{T}(s) \mathbf{R}^{-1}(s) \underline{\mathbf{r}}(s) ds .$ 

Since  $\underline{I}(t|\tau)$  fis orthogonal to the residual  $\underline{r}(s)$ , it is clear that

$$\frac{\mathcal{I}}{\mathcal{I}}\left[\left[\underline{\mathcal{I}}(t|\tau)\underline{\mathbf{X}}^{\mathsf{T}}(t|\tau)\right] = \mathbb{E}\left[\underline{\mathbf{X}}(t|t)\underline{\mathbf{X}}^{\mathsf{T}}(t|t)\right] - \int_{t}^{\mathsf{T}} \mathbb{P}(t,s)\mathbf{H}^{\mathsf{T}}(s)\mathbf{R}^{-1}(s)\mathbb{E}\left[\underline{\mathbf{r}}(s)\underline{\mathbf{X}}^{\mathsf{T}}(t|t)\right] ds$$

 $\mathbf{P}(t|\tau) = \mathbf{P}(t|t) - \int_t^{\tau} \mathbf{P}(t,s) \mathbf{H}^{\mathsf{T}}(s) \mathbf{R}^{-1}(s) \mathbf{H}(s) \mathbf{P}(s,t) \, \mathrm{d}s \, .$ 

(2.97)

(2.95)



The integrand is non-negative-definite, so the effect of the additional data in the smoothed estimate is to reduce the error in the filtered estimate.

The correlation matrix P(t,s) is easily determined. As defined, it is known that

$$P(t,s) \cong E[\underline{\sharp}(t|t)\underline{\sharp}^{T}(s|s)], \quad s > t$$

But g(tit) is the solution of

 $\frac{1}{2} = [F(t) - K(t)H(t)]\frac{1}{2} + \underline{w}(t) - K(t)\underline{v}(t) :$ 

80

$$\underline{X}(\mathtt{s}|\mathtt{s}) \simeq \Psi(\mathtt{s},\mathtt{t})\underline{X}(\mathtt{t}|\mathtt{t}) + \Psi(\mathtt{s},\mathtt{t}) \int_{\mathtt{t}}^{\mathtt{s}} \Psi(\mathtt{t},\sigma) [\underline{w}(\sigma) - K(\sigma)\underline{y}(\sigma)] \ \mathrm{d}\sigma \ ,$$

where  $\Psi(\mathbf{s},t)$  is the fundamental solution obtained from the matrix differential equation

$$\frac{d}{ds}\Psi(s,t) = [F(s) - K(s)H(s)]\Psi(s,t) , \quad \Psi(t,t) = I ,$$

Using this solution, one obtains

$$P(t,s) = P(t|t)\Psi^{T}(s,t) .$$

The smoothed estimate becomes

$$g(t|\tau) = g(t|t) + P(t|t) \int_{-\infty}^{\infty} \Psi^{T}(s,t) H^{T}(s) R^{-1}(s) g(s) ds$$
 (2.98)

and

 $P(t|\tau) \triangleq E[\underline{I}(t|\tau)\underline{I}^T(t|\tau)]$ 

$$= P(t|t) - P(t|t) \int_{0}^{t} \Psi^{T}(s,t) H^{T}(s) R^{-1}(s) H(s) \Psi(s,t) ds P(t|t)$$

$$P(t|t) = P(t|t)M_{n}(t,\tau)P(t|t) , \qquad (2.99)$$

where

$$\mathtt{M}_{\mathbf{s}}(\mathbf{t},\tau) = \int_{\mathbf{t}}^{\tau} \Psi^{\mathsf{T}}(\mathbf{s},\mathbf{t}) \mathtt{H}^{\mathsf{T}}(\mathbf{s}) \mathtt{R}^{-1}(\mathbf{s}) \mathtt{H}(\mathbf{s}) \Psi(\mathbf{s},\mathbf{t}) \ \mathrm{d}\mathbf{s} \ .$$

The matrix  $M_g(t,\tau)$  is similar to the observability matrix introduced in Section 3.1. Equations (2.98) and (2.99) are the general smoothing equations. Three classes of smoothing problems have been discussed in the literature<sup>24</sup>:

- (i) Fixed-interval smoothing<sup>29</sup>: The initial time  $t_0$  and the final time  $\tau$  are fixed.
- (ii) Pixed-point smoothing<sup>10</sup>: The time t for which a smoothed estimate is determined is fixed, while the amount of data increases (i.e.  $\tau$  increases).
- (iii) Fixed-lag smoothing<sup>28</sup>: The time for which a smoothed estimate is determined is a fixed amount  $\triangle$  behind the most recent data occurring at time  $\tau$ .

These results can be derived?? from the general results and will not be discussed here.

#### 3. OBSERVABILITY AND THE BEHAVIOR OF THE ERROR COVARIANCE MATRIX

In this section two important concepts related to the linear estimation problem are discussed. The first of these is observability<sup>13,31</sup>, that ir, the property of a system which permits estimation of its state. The second concept is the stability of the estimate as defined by the behavior of the error covariance<sup>13,32</sup>.

## 3.1 Observability of Deterministic Systems

Consider a system described by Equations (1.1) and (1.2). In contrast to Section 2, it is assumed at this point that  $\underline{w}(t)$  is a deterministic signal and that  $\underline{v}(t)$  is identically zero. A sinte  $\underline{\chi}(t)$  of the resultant deterministic system is called observable if from the input  $\underline{w}(\tau)$ ,  $t_o \leq \tau \leq t$ , and the output  $\underline{\zeta}(t)$ ,  $\underline{\chi}(t)$ 

can be completely determined. If all states  $\chi(t)$  corresponding to all admissible Z(t) are observable, the system is called *completely observable*.

The solution of Equation (1.1) is given by

$$\underline{\mathbf{x}}(\mathbf{t}) = \Phi(\mathbf{t}, \mathbf{t}_0) \underline{\mathbf{x}}(\mathbf{t}_0) + \int_{\mathbf{t}_0}^{\mathbf{t}} \Phi(\mathbf{t}, \tau) \underline{\mathbf{x}}(\tau) d\tau , \qquad (3.1)$$

where  $\Phi(t,\tau)$  is the transition matrix for Equation (1.1),

$$\frac{d\Phi(t,\tau)}{dt} = F(t)\Phi(t,\tau); \quad \Phi(\tau,\tau) = I. \quad (3.2)$$

The  $\underline{\mathbf{x}}(\tau)$ ,  $\mathbf{t}_0 \leq \tau \leq \mathbf{t}$ , and  $\Phi(\mathbf{t},\tau)$  completely define the second term appearing in Equation (3.1). Apparently if  $\underline{\mathbf{x}}(\mathbf{t}_0)$  can be completely determined from knowledge of  $Z(\mathbf{t})$ , then  $\underline{\mathbf{x}}(\mathbf{t})$  can be completely determined, that is, the state  $\underline{\mathbf{x}}(\mathbf{t})$  will be observable. If this is true for an arbitrary state, then the system is completely observable.

# 3.1.1 Observability Criterion

South States

Combining Equation (3.1) and Equation (1.2) with  $\underline{y}(t)$  set to zero, the observation vector  $\underline{z}(t)$  is given by

$$\underline{\mathbf{g}}(\mathbf{t}) = \mathbf{H}(\mathbf{t})\underline{\mathbf{x}}(\mathbf{t}) = \int_{\mathbf{t}_0}^{\mathbf{t}} \mathbf{H}(\mathbf{t})\Phi(\mathbf{t},\tau)\underline{\mathbf{x}}(\tau) \, d\tau + \mathbf{H}(\mathbf{t})\Phi(\mathbf{t},\mathbf{t}_0)\underline{\mathbf{x}}(\mathbf{t}_0) \, . \tag{3.3}$$

Knowledge of  $\underline{Z}(t)$  and  $\underline{y}(\tau)$ ,  $t_0 \le \tau \le t$ , completely determines  $\Delta z(t) = z(t) - \int^t H(t) \Phi(t, \tau) w(\tau)$ 

$$g(t) = \underline{g}(t) - \int_{t_0}^{t} H(t) \Phi(t, \tau) \underline{w}(\tau) d\tau$$
  
=  $H(t) \Phi(t, t_0) \underline{x}(t_0)$ ;  $t_0 \leq \tau \leq t$ . (3.4)

Thus it is sufficient to deal with  $\Delta g(\tau)$ ,  $t_0 \leq \tau \leq t$ , alone and seek to determine under what conditions  $g(t_0)$  can be completely determined from  $\Delta g(\tau)$ ,  $t_0 \leq \tau \leq t$ . These conditions are then conditions for the observability of the state g(t). If these conditions do not depend on  $g(t_0)$  or on  $\Delta g(\tau)$ ,  $t_0 \leq \tau \leq t$ , they are also conditions for complete observability of the system.

Now consider a linear function of the form

$$\int_{t_0}^{t} \mathbf{u}^{\mathrm{T}}(\tau) \Delta \underline{\mathbf{z}}(\tau) \, \mathrm{d}\tau \,, \qquad (3.5)$$

where  $s(\tau)$  is a matrix of piecewise continuous functions. The system is completely observable if, for some  $t > t_0$ , an  $s(\tau)$  exists such that this function equals  $\underline{x}(t_0)$  for arbitrary  $\underline{x}(t_0)$ . Since  $\Delta \underline{x}(\tau)$  is linear in  $\underline{x}(t_0)$ , only linear functions need be considered.

#### Theorem 3.1

A necessary and sufficient condition that a system defined by Equations (1.1) and (1.2), with  $\underline{u}(t)$  a deterministic function and  $\underline{v}(t)$  identically zero, be completely observable is that the matrix

$$M(t_0, t) = \int_{t_0}^{t} \Phi^{T}(\pi, t_0) H^{T}(\tau) H(\tau) \Phi(\tau, t_0) d\tau \qquad (3.6)$$

be positive-definite for some,  $t > t_n$  .

Proof of Sufficiency

Assume that  $M(t_0, t)$  is positive-definite and thus has an inverse. In Equation (3.5) let

$$B(\tau) = H(\tau) \Phi(\tau, t_{n}) M^{-1}(t_{n}, t);$$
 (3.7)

then

$$\int_{0}^{L} \mathbf{M}^{-1}(\mathbf{t}_{0}, \mathbf{t}) \, \Phi^{T}(\tau, \mathbf{t}_{0}) \, \mathbf{H}^{T}(\tau) \, \Delta \mathbf{g}(\tau) \, d\tau$$

$$= \mathbf{N}^{-1}(\mathbf{t}_{0}, \mathbf{t}) \, \int_{\mathbf{t}_{0}}^{L} \Phi^{T}(\tau, \mathbf{t}_{0}) \, \mathbf{H}^{T}(\tau) \, \mathbf{H}(\tau) \, \Phi(\tau, \mathbf{t}_{0}) \, d\tau \, \mathbf{g}(\mathbf{t}_{0})$$

$$= \mathbf{g}(\mathbf{t}_{0}) \, . \qquad (3.8)$$

Since the condition is independent of  $\underline{x}(t_n)$ , the system is completely observable.

-9**6**1 A SHIPPIN ter C. akliga bijen

# Proof of Necessity (by Contradiction)

Suppose the system is completely observable, but that  $M(t_{g}, t)$  is singular and hence not positive-definite. This implies that a non-zero constant vector p exists such that

$$\mathbf{p}^{\mathrm{T}}\mathsf{M}(\mathbf{t}_{0},\mathbf{t})\mathbf{p} = \mathbf{p}^{\mathrm{T}} \int_{\mathbf{t}_{0}}^{\mathbf{t}} \Phi^{\mathrm{T}}(\tau,\mathbf{t}_{0}) \mathrm{H}^{\mathrm{T}}(\tau) \mathrm{H}(\tau) \Phi(\tau,\mathbf{t}_{n}) \mathrm{d}\tau \mathrm{p}$$

$$= \int_{\mathbf{t}_{0}}^{\mathbf{t}} [\mathrm{H}(\tau) \Phi(\tau,\mathbf{t}_{0}) \mathbf{p}]^{\mathrm{T}} [\mathrm{H}(\tau) \Phi(\tau,\mathbf{t}_{0}) \mathbf{p}] \mathrm{d}\tau$$

$$= \mathbf{0} . \qquad (3.9)$$

Since  $H(\tau)\Phi(\tau, t_0)p$  is a continuous function for  $t_0 \le \tau \le t$ , it must be identically zero in order that Equation (3.9) be satisfied.

Now, if the system is completely observable, then for  $\chi(t_0) = p$  there exists an  $s^{*}(\tau)$ ,  $t_0 \le \tau \le t$ , such that

$$\int_{t_0}^{t} \mathbf{s}^{\mathbf{s}^{\mathbf{t}}}(\tau) \Delta \underline{\mathbf{z}}(\tau) d\tau = \int_{t_0}^{t} \mathbf{s}^{\mathbf{s}^{\mathbf{t}}}(\tau) H(\tau) \Phi(\tau, t_0) \underline{\mathbf{p}} d\tau = \underline{\mathbf{p}} . \qquad (3.10)$$

However, since  $H(\tau)\Phi(\tau,t_0)p = 0$ ,  $t_0 \le \tau \le t$ , the last equality cannot hold and the system is not completely observable.

When H(t) and F(t) are constant matrices, the complete observability criterion can be reduced to an algebraic criterion.

#### Theorem 3.2

Automotion - and the second of the second

A necessary and sufficient condition that a system defined by Equations (1.1) and (1.2), with  $\underline{w}(t)$  a deterministic function,  $\underline{w}(t)$  identically zero, and H(t) and F(t) constant matrices, by completely observable is that the matrix (dimension  $n \times mn$ )

$$\mathbf{U} = [\mathbf{H}^{T}, \mathbf{F}^{T} \mathbf{H}^{T}, \dots, (\mathbf{F}^{T})^{n-1} \mathbf{H}^{T}]$$
(3.11)

have rank n (the dimension of the state  $g(t_o)$ ).

**Proof of Sufficiency (by Contradiction)** 

Since the system is time-invariant, t, may be set equal to zero without loss of generality. Also

$$\Phi(\mathbf{t},\mathbf{0}) = \mathbf{e}^{\mathbf{p}\mathbf{t}}, \qquad (3,12)$$

Let  $h_1$ ,  $h_2$ , ...,  $h_n$  be the columns of  $H^T$ . Assume that U has rank n but that the system is not completely observable, that is, M(0,t) is singular. If M(0,t) is singular then a non-zero <u>p</u> exists such that

$$\mathbf{p}^{\mathsf{T} \mathbf{e}^{\mathsf{p} \cdot \mathsf{t}}} \mathbf{h}_{i} = 0; \quad 0 \leq \tau \leq \mathsf{t}, \quad i = 1, 2, \dots, m.$$
 (3.13)

Now differentiate each of the m Equations (3.13) j times, j = 0, 1, 2, ..., n-1, thus generating

$$\mathbf{p}^{T}(\mathbf{F}^{T})^{j} \mathbf{e}^{\mathbf{p} \cdot \mathbf{c}} \mathbf{h}_{1} = 0$$
;  $\mathbf{i} = 1, 2, ..., \mathbf{m}$  and  $\mathbf{j} = 0, 1, ..., n-1$ . (3.14)

Now Equations (3.14) must hold for all t > 0, since U is constant. In the limit as t approaches zero, Equations (3.14) become

$$\mathbf{p}^{\mathbf{T}}(\mathbf{r}^{\mathbf{T}})^{\mathbf{j}}\mathbf{h}_{1} = 0$$
;  $\mathbf{i} = 1, 2, \dots, \mathbf{m}$  and  $\mathbf{j} = 0, 1, \dots, n-1$ . (3.15)

If the rank of U is n , then  $\mathbf{p} = \mathbf{0}$  , which contradicts the assumption that  $\mathbf{p} \neq \mathbf{0}$  .

#### **Proof of Necessity**

Assume the system is completely observable but that the rank of U is less than n. There then exists a non-zero <u>p</u> which satisfies Equation (3.15). Now form

$$\mathbf{p}^{\mathsf{T}_{\mathbf{0}}\mathbf{P}^{\mathsf{T}_{\mathsf{C}}}\mathsf{H}_{i}} = \mathbf{p}^{\mathsf{T}} \sum_{j=0}^{\infty} (\mathbf{P}^{\mathsf{T}_{\mathsf{C}}})^{j} / j + \mathbf{h}_{i}$$
$$= \mathbf{p}^{\mathsf{T}} \sum_{j=0}^{n-1} \alpha_{j} (\mathbf{P}^{\mathsf{T}_{\mathsf{C}}})^{j} \mathbf{h}_{i}, \quad i = 1, 2, \dots, \mathbf{u}, \qquad (3.16)$$

بالمستعظم والمستعم والمعالية والمستعم والمستعم والمستعم والمستعم والمستع والمستع والمستع والمستع والمستع والمست





State Million Children

where the last equality follows from the Cuyley-Hamilton theorem. It follows from Equation (3.16) that

$$p^{T}M(0,t)p = 0$$
 (3.17)

for some  $p \neq 0$ , thus contradicting the assumption that the system is completely observable.

#### 3.1.2 Observability and Least-Squares Estimates

Consider again a system described by Equations (1.1) and (1.2). In this case it is assumed that  $\underline{w}(t)$  is identically zero and  $\underline{y}(t)$  is a vector white-noise with covariance matrix  $R(t)\delta(t-\tau)$ , where R(t) is positive-definite. Now the observation vector can be written in the form

$$\vec{z}(t) = H(t)\Phi(t, t_0) \underline{x}(t_0) + \underline{y}(t) . \qquad (3.18)$$

This system is called *completely observable* if, for every  $t_0$  and every  $\underline{\chi}(t_0)$ , there exists a t > 0 such that an unbiased estimate  $\hat{\chi}(t_0)$ , which is a linear function of  $\underline{\chi}(t)$ , can be constructed. The general criterion for complete observability is given in the following theorem.

#### Theorem 3.3

Land and the second

A necessary and sufficient condition that a system defined by Equations (1.1) and (1.2), with  $\Psi(t)$  identically zero and  $\Psi(t)$  a white-noise with autocorrelation matrix  $R(t)\delta(t-\tau)$ , R(t) positive-definite, be completely observable in that the matrix

$$M(t_{0},t) = \int_{t_{0}}^{t} \Phi^{T}(\tau,t) H^{T}(\tau) R^{-1}(\tau) H(\tau) \Phi(\tau,t) d\tau \qquad (3.19)$$

be positive-definite for some  $t > t_{\sigma}$  .

The proof of this theorem parallels that of Theorem 3.1.

If the system is time-invariant, that is, H(t) and F(t) arc constant, then the complete observability oritorion of Theorem 3.3 can be simplified to the algebraic criterion of the following theorem.

#### Theorem 3.4

A necessary and sufficient condition that the system defined in Theorem 3.3 be completely observable is that the matrix

$$\mathbf{U} = [\mathbf{H}^{\mathrm{T}}, \mathbf{F}^{\mathrm{T}} \mathbf{H}^{\mathrm{T}}, \dots, (\mathbf{F}^{\mathrm{T}})^{n-1} \mathbf{H}^{\mathrm{T}}]$$
(3.20)

have rank n (the dimension of the state  $\underline{x}(t)$ ).

The proof of this theorem purallels that of Theorem 3.2 and is not given.

#### 3.2 The Matrix-Ricatti Equation

# 3.2.1 General Solution

In Section 2 it was shown that a matrix differential equation of the Ricatti type is of central importance in the linear estimation problem. In this section the analytic solution of this type of equation is discussed<sup>10</sup>.

Consider the general matrix-Ricatti equation of the form

$$\frac{1}{2}(t) = W(t) A^{T}(t) + A(t) W(t) + W(t) B(t) W(t) + C(t),$$
(3.21)

with the initial condition

 $\Psi(t_n) = \Psi_n$ 

where Wo is a non-negative-definite matrix.

The matrices A(t), B(t), and C(t) are non-matrices of continuous functions. Further, B(t) and C(t) are assumed to be non-negative-definite for all  $t \ge t_0$ . Now consider the pair of linear matrix differential equations

 $\hat{Y}(t) = A(t)Y(t) + C(t)Z(t); \quad Y(t_0) = W_0$ 

 $\hat{Z}(t) = -B(t)Y(t) - A^{T}(t)Z(t); \quad Z(t_{0}) = I.$  (3.22)





XOI:



Now if Z(t) is nonsingular for all  $t \ge t_0$ , this equation is equivalent to Equation (3.21), and

 $W(t) = Y(t)Z^{-1}(t)$  (3.27)

Note also that

$$W(t_0) = Y(t_0)Z^{-1}(t_0) = W_0I = W_0;$$
 (3.28)

thus the initial condition is also satisfied.

To show that Z(t) is nonsingular, it is noted that

$$\dot{Z}(t) = -B(t)^{Y}(t) - A^{T}(t)Z(t)$$
  
=  $[-A^{T}(t) - B(t)W(t)]Z(t)$ . (3.29)

Since  $Z(t_0) = I$ , then Z(t) is a transition matrix and hence nonsingular.

It was seen in Section 2 that there are two forms of the error covariance equation. The first of these is

 $\dot{P}(t|t) = P(t|t) [P(t) - K(t) H(t)]^{T} + [P(t) - K(t) H(t)] P(t|t) + K(t) R(t) K^{T}(t) + Q(t)$ (3.30)

 $P(t_a) = P_a$ .

Comparing Equation (3.30) with Equation (3.21) the following equivalent set of equations is obtained by a comparison with Equations (3.22);

 $\frac{1}{2}(t) = \left[ F(t) - K(t) H(t) \right] Y(t) + \left[ K(t) R(t) K^{T}(t) + Q(t) \right] Z(t) ; Y(t_{0}) = P_{0} \\
\frac{1}{2}(t) = -\left[ F(t) - K(t) H(t) \right]^{T} Z(t) ; Z(t_{0}) = I.$ (3.31)

Nov

$$Z(t) = \Psi^{T}(t_{0}, t) , \qquad (3.32)$$

where  $\Psi(t, t_{s})$  is a transition matrix describing the filter dynamics and satisfying

$$\frac{d\Psi(t,t_0)}{dt} = [\Psi(t) - K(t)H(t)]\Psi(t,t_0) \qquad (3.33)$$

$$\Psi(t_0,t_0) = I$$

Y(t) can now be written as

$$\Psi(t) = \Psi(t, t_0) P_0 + \int_{t_n}^{t} \Psi(t, \tau) K(\tau) R(\tau) R^{T}(\tau) + Q(\tau) \Psi^{T}(t_0, \tau) d\tau .$$
(3.34)

Finally,

 $P(t|t) = Y(t)Z^{-1}(t)$ 

 $= \Psi(t, t_0) \left[ P_0 + \int_{t_0}^{t} \Psi(t_0, \tau) \right] \left[ K(\tau) R(\tau) K^{T}(\tau) + Q(\tau) \right] \Psi^{T}(t, \tau) d\tau \right] \Psi^{T}(t, t_0) .$  (3.35)



San Winn Van Wart



The second form of the error covariance equation is obtained from Equation (3.30) by letting

 $K(t) = P(t|t)H^{T}(t)R^{-1}(t) .$ 

The resulting equation is

$$\dot{P}(t|t) = P(t|t)F^{T}(t) + F(t)P(t|t) - P(t|t)H^{T}(t)R^{-1}(t)H(t)P(t|t) + Q(t)$$
  
 $P(t_{0}) = P_{0}$ .

The set of equivalent linear equations is

Now lot

$$\Theta(\mathbf{t}, \mathbf{t}_0) = \begin{bmatrix} \Theta_{11}(\mathbf{t}, \mathbf{t}_0) & \Theta_{12}(\mathbf{t}, \mathbf{t}_0) \\ \Theta_{21}(\mathbf{t}, \mathbf{t}_0) & \Theta_{22}(\mathbf{t}, \mathbf{t}_0) \end{bmatrix}$$
(3.39)

be the transition matrix corresponding to the matrix

$$\begin{bmatrix} \mathbf{F}(t) & \mathbf{Q}(t) \\ \cdot \mathbf{H}^{T}(t) \mathbf{R}^{-1}(t) \mathbf{H}(t) & -\mathbf{F}^{T}(t) \end{bmatrix}$$
(3.40)

Therefore

$$Y(t) = \Theta_{11}(t, t_0)P_0 + \Theta_{12}(t, t_0)$$
(3.41)  
$$Z(t) = \Theta_{11}(t, t_0)P_0 + \Theta_{12}(t, t_0)$$
(3.42)

and LinaUy

$$P(t|t) = Y(t)Z^{-1}(t)$$

$$= \left[\Theta_{1}(t, t_{0})P_{0} + \Theta_{12}(t, t_{0})\right]\left[\Theta_{21}(t, t_{0})P_{1} + \Theta_{22}(t, t_{0})\right]^{-1}.$$
(3.43)

We now consider two special cases.

Gase 1: Let H = 0, so that

$$\dot{P} = PP^{T}(t) + F(t)P + Q(t)$$
 (3.44)

In this case the Ricatti Equation (3.37) reduces to a linear matrix equation whose solution can be obtained using the general formalism. From Equation (3.38),

$$\dot{Z} = -\mathbf{F}^{T}(\mathbf{t})\mathbf{Z}$$
,  $\mathbf{Z}(\mathbf{t}_{n}) = \mathbf{I}$ .

 $\Phi(t_0, t_0) = I$ 

But this equation is the adjoint of

φ́ = ₹(t)Φ,

80

$$Z(t) = \Theta_{aa}(t, t_0) Z(t_0) = \Phi^T(t_0, t) . \qquad (3.45)$$

This implies that.

$$\theta_{zi}(t,t_0) = 0$$
  
$$\theta_{zg}(t,t_0) = \Phi^{T}(t_0,t) .$$

Using Equation (3.45), one has

 $\mathbf{\hat{T}} = \mathbf{F}(t)\mathbf{Y} + \mathbf{\hat{q}}(t)\boldsymbol{\Phi}^{\mathrm{T}}(t_{\mathrm{o}},t) ,$ 



D'élén a bién

37

(3.36)

(3.37)

so it follows that

 $\cdot \quad Y(t) = \Phi(t, t_0) Y(t_0) + \Phi(t, t_0) \int_{t_0}^{t} \Phi(t_0, \tau) Q(\tau) \Phi^{\mathsf{T}}(t_0, \tau) d\tau$ 

which implies that

$$\begin{split} \theta_{11}(t,t_0) &= \Phi(t,t_0) \\ \theta_{10}(t,t_0) &= \Phi(t,t_0) \int_{t_0}^{t} \Phi(t_0,\tau) Q(\tau) \Phi^{T}(t_0,\tau) d\tau \; . \end{split}$$

Bubstituting these into Equation (3.43), the solution of the linear matrix differential equation is found to be

$$\mathbf{P}(\mathbf{t}|\mathbf{t}) = \Phi(\mathbf{t}, \mathbf{t}_0) \mathbf{P}(\mathbf{t}_0) \Phi^{\mathsf{T}}(\mathbf{t}, \mathbf{t}_0) + \Phi(\mathbf{t}, \mathbf{t}_0) \left\{ \int_{\mathbf{t}_0}^{\mathbf{t}} \Phi(\mathbf{t}_0, \tau) \mathbf{Q}(\tau) \Phi^{\mathsf{T}}(\mathbf{t}_0, \tau) \, \mathrm{d}\tau \right\} \Phi^{\mathsf{T}}(\mathbf{t}, \mathbf{t}_0) . \tag{3.47}$$

Case 2: Let Q = 0 , so that

 $\dot{\mathbf{P}} = \mathbf{P}\mathbf{P}^{T}(t) + \mathbf{F}(t)\mathbf{P} - \mathbf{P}\mathbf{H}^{T}(t)\mathbf{R}^{-1}(t)\mathbf{H}(t)\mathbf{P} ,$ 

In this case the variance equation has no forcing term and one can consider

f = F(t)Y.

This bas the solution

$$\mathbf{Y}(\mathbf{t}) = \Phi(\mathbf{t}, \mathbf{t}_n) \mathbf{P}(\mathbf{t}_n)$$

and implies that

 $\Theta_{ii}(t,t_0) = \Phi(t,t_0)$  $\Theta_{ii}(t,t_0) = 0.$ 

Using Equation (3.49), the equation for 2 becomes

 $\dot{\mathbf{z}} = -\mathbf{F}^{T}(t)\mathbf{z} + \mathbf{H}^{T}(t)\mathbf{R}^{-1}(t)\mathbf{H}(t)\Phi(t,t_{0})\mathbf{P}_{0}$ .

This has the solution

$$\mathbf{Z}(\mathbf{t}) = \Phi^{\mathsf{T}}(\mathbf{t}_{o}, \mathbf{t}) + \Phi^{\mathsf{T}}(\mathbf{t}_{o}, \mathbf{t}) \int_{\mathbf{t}_{o}}^{\mathbf{R}} \Phi^{\mathsf{T}}(\tau, \mathbf{t}_{o}) \mathbf{H}^{\mathsf{T}}(\tau) \mathbf{R}^{-1}(\tau) \mathbf{H}(\tau) \Phi(\tau, \mathbf{t}_{o}) d\tau \mathbf{P}_{o} , \qquad (1)$$

so that

$$\begin{split} \theta_{zi}(t,t_0) &= \Phi^{T}(t_0,t) \int_{t_0}^{t} \Phi^{T}(\tau,t_0) H^{T}(\tau) R^{-1} H(\tau) \Phi(\tau,t_0) d\tau \\ \theta_{zz}(t,t_0) &= \Phi^{T}(t_0,t) . \end{split}$$

The solution of Equation (3.48), using Equations (3.49) and (3.50) in Equation (3.43), is seen to be

$$P(t|t) = \Phi(t, t_0) P_0 \left[ \Phi^T(t_0, t) + \Phi^T(t_0, t) \int_{t_0}^{t} \Phi^T(\tau, t_0) H^T(\tau) R^{-1}(\tau) H(\tau) \Psi(\tau, t_0) d\tau P_0 \right]^{-1}$$

$$= \Phi(t, t_0) P_0 \left[ I + \int_{t_0}^{t_0} \Phi^T(\tau, t_0) H^T(\tau) R^{-1}(\tau) H(\tau) \Phi(\tau, t_0) d\tau P_0 \right]^{-1} \Phi^T(t, t_0)$$
(3.81)

But observe that the integral term is identically equal to the observability matrix  $M(t_0,t)$  defined in Equation (3.19). This suggests the important role played by observability considerations in establishing the behavior of the error of the minimum variance estimator

# 3.2.2 Solution for the Stationary Case

Suppose in Equation (3.37) that the coofficient matrices F, H, R, and Q are constant. In this case, one can obtain more insight into the character of the general solution and can define the steady-state solution. Form

 $\mathbf{N} \stackrel{\Delta}{=} \begin{bmatrix} \mathbf{P} & \mathbf{Q} \\ \\ \mathbf{H}^{\mathbf{T}} \mathbf{R}^{-1} \mathbf{H} & -\mathbf{P}^{\mathbf{T}} \end{bmatrix} , \qquad (3.52)$ 

(3.49)

(3.48)

(3.50)







(3.46)

where N represents the matrix of coefficients implied in the system (3.38). It has been shown<sup>33</sup> that the eigenvalues of N are roul and that they are symmetric relative to the origin. For this discussion suppose that the eigenvalues are distinct and let  $\Lambda$  be the n×n diagonal matrix containing the eigenvalues that are positive. Then, there exists a transformation T such that

 $D \stackrel{\Delta}{=} \begin{bmatrix} \Lambda & \mathbf{0} \\ \mathbf{0} & -\cdot \Lambda \end{bmatrix}$   $= \mathbf{T}^{\mathbf{1}} \mathbf{N} \mathbf{T}, \qquad (3.53)$ 

where

$$\mathbf{T} \stackrel{\Delta}{=} \begin{bmatrix} \mathbf{T}_{11} & \mathbf{T}_{12} \\ \mathbf{T}_{21} & \mathbf{T}_{22} \end{bmatrix} .$$

It can be shown that

 $\mathbf{T}^{-1} \stackrel{\Delta}{=} \begin{bmatrix} \mathbf{T}_{12}^{\mathrm{T}} & -\mathbf{T}_{12}^{\mathrm{T}} \\ -\mathbf{T}_{21}^{\mathrm{T}} & \mathbf{T}_{11}^{\mathrm{T}} \end{bmatrix} .$ (3.54)

Define matrices  $\Omega$  and  $\Sigma$  such that

 $\begin{bmatrix} \mathbf{Y} \\ \mathbf{z} \end{bmatrix} = \begin{bmatrix} \mathbf{T}_{11} & \mathbf{T}_{12} \\ \mathbf{T}_{21} & \mathbf{T}_{22} \end{bmatrix} \begin{bmatrix} \Omega \\ \Sigma \end{bmatrix}$ (3.65)

and

$$\begin{bmatrix} \dot{\mathbf{Y}} \\ \dot{\mathbf{z}} \end{bmatrix} = \begin{bmatrix} \mathbf{T}_{11} & \mathbf{T}_{12} \\ \mathbf{T}_{21} & \mathbf{T}_{22} \end{bmatrix} \begin{bmatrix} \dot{\boldsymbol{\Omega}} \\ \dot{\boldsymbol{\Sigma}} \end{bmatrix}.$$

From Equation (3.53), it follows that

$$\begin{bmatrix} \dot{\Omega} \\ \dot{\Sigma} \end{bmatrix} = \begin{bmatrix} \Lambda & 0 \\ 0 & -\Lambda \end{bmatrix} \begin{bmatrix} \Omega \\ \Sigma \end{bmatrix}.$$
(3.56)

Since this is a linear, constant coefficient system, the solution of Equation (3.36) is

$$\begin{bmatrix} \Omega(t) \\ \Sigma(t) \end{bmatrix} = \begin{bmatrix} e^{\Lambda t} & 0 \\ 0 & e^{-\Lambda t} \end{bmatrix} \begin{bmatrix} \Omega(0) \\ \Sigma(0) \end{bmatrix}, \qquad (3.57)$$

so that Z and Y are



The transition matrix  $\Theta(t)$  for this system is

$$\Theta(t) = T \begin{bmatrix} e^{At} & 0 \\ 0 & e^{-At} \end{bmatrix} T^{-1}$$

$$= \begin{bmatrix} T_{11}e^{At}T_{22}^{T} - T_{12}e^{-At}T_{21}^{T} & T_{12}e^{-At}T_{11}^{T} - T_{11}e^{At}T_{12}^{T} \\ T_{21}e^{At}T_{22}^{T} - T_{22}e^{-At}T_{21}^{T} & T_{22}e^{-At}T_{11}^{T} - T_{21}e^{At}T_{12}^{T} \end{bmatrix}$$

$$(3.59)$$

Substitution of Equation (3.59) into Equation (3.43) yields the solution P(t) .

$$P(t|t) = [T_{12}e^{-\Lambda t}T_{11}^{T} - T_{11}e^{\Lambda t}T_{12}^{T} + (T_{11}e^{\Lambda t}T_{22}^{T} - T_{12}e^{-\Lambda t}T_{21}^{T})P_{0}]^{*} \times \\ \times [T_{22}e^{-\Lambda t}T_{11}^{T} - T_{21}e^{\Lambda t}T_{12}^{T} + (T_{21}e^{\Lambda t}T_{22}^{T} - T_{22}e^{-\Lambda t}T_{21}^{T})P_{0}]^{-1} .$$
(3.60)


ELECTRONIC CONTRACTOR

E States a

Consider the steady-state value of P as  $t \to \infty$ . The terms involving  $e^{-At}$  will vanish so

$$\begin{array}{rcl} & & & & & & \\ & & & & & \\ & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & \\ & & & & & \\ & & & & & & \\ & & & & &$$

Equation (3.60) has a computational disadvantage, since only the term  $e^{At}$  appears as t increases without bound. It is difficult to retain accuracy as t becomes large because of the prosence of  $e^{A^{\pm}}$  in both Y(t)and  $Z^{-1}(t)$ . This problem can be circumvented by expressing the initial value problem (Equ (3.3B)) as a two-point boundary-value problem<sup>14</sup>. Equation (3.57) can be rewritten as

_ປເທ	-	•-Vt	0	Ω(t)	
_Σ(t)	E	0	e-At	Σ(0)	•

It has been shown that

## Y(t) = P(tit)Z(t) Y(0) = P(0)Z(0).

Using Equation (3.55), it can be seen that

$$(0) = P(0) [T_{HL}\Omega(0) + T_{HL}\Sigma(0)]$$

 $Y(0) = T_{11}\Omega(0) + T_{12}\Sigma(0)$ 

and

so that

and

 $0 = [T_{11} - P(0)T_{21}]\Omega(0) + [T_{12} - P(0)T_{22}]\Sigma(0) ,$ 

$$\Sigma(0) = -[T_{12} - P(0)T_{22}]^{-1}[T_{11} - P_0T_{21}]\Omega(0) \stackrel{\Delta}{=} R\Omega(0) .$$

Using this,  $\Omega(t)$  and  $\Sigma(t)$  can be related by

$$E(t) = e^{\Lambda t} \Sigma(0) = e^{-\Lambda t} R \Omega(0) = e^{-\Lambda t} R e^{-\Lambda t} \Omega(t)$$
(3.62)

Thus, one finds that

These relations imply that

and

 $\Upsilon(t) = T_{11}\Omega(t) + T_{12}\Sigma(t) = [T_{11} + T_{12} e^{-\Lambda t} Re^{-\Lambda t}]\Omega(t)$  $Z(t) = [T_{s1} + T_{ss} e^{-At} Rs^{-At}] \Omega(t)$  $P(t|t) = [T_{11} + T_{12} e^{-\Lambda t} Re^{-\Lambda t}] [T_{21} + T_{22} e^{-\Lambda t} Re^{-\Lambda t}]^{-1}$ (3.63)

In this form only negative exponentials appear and the steady-state value  $P_m$  expressed by Equation (3.61) follows immediately. Thus Equation (3.53) appears to have computational advantages.

### 3.3 Bounds for the Error Covariance Matrix

The error covariance matrix P(t) of the Kalman-Buoy filter provides the primary statistical measure of the behavior of the estimate. Since the P(t|t) does not depend upon measurement data, it can be examined to determine the theoretical effectiveness of a particular measurement process for a given dynamical system. In this section upper and lower hounds for the error opvariance matrix are derived 20,32 that point out the intrinsically different roles played by the plant and measurement noise processes. The bounds indicate the importance of the deterministic observability considerations of Section 3.1 and are derived using some of the properties of the matrix-Ricatti equation solution discussed in Section 3.2.

### 3.3.1 Some General Bounds

Consider the system defined by Equations (1.1) and (1.2). This system can be reinterpreted as the sum of two simpler systems, which allows the effects of the measurement and plant processes to be examined separately and thereby obtain insight into their influence.















(3.67)

(3.68)

Define System P as a system with plant noise but no measurement noise,

 $\dot{z}^{p} = F(t) z^{p} + y, \quad \underline{x}^{p}(t_{0}) = 0$  (3.64)

$$g^{p} = H(t)g^{p}$$
, (3.65)

and define System M to have measurement noise but no plant neise

$$\underline{k}^{m} = F(t)\underline{x}^{m}, \quad \underline{x}^{m}(t_{0}) = \underline{x}(t_{0}) \qquad (3.66)$$

 $\underline{\mathbf{x}}^{\mathsf{m}} = \mathbf{H}(\mathbf{t}) \, \underline{\mathbf{x}}^{\mathsf{m}} + \underline{\mathbf{y}} \, .$ 

Note that

and that

 $= F(t)[x^p + x^n] + y,$ 

 $\underline{\mathbf{X}}(\mathbf{t}_{0}) = \underline{\mathbf{X}}^{\mathbf{H}}(\mathbf{t}_{0}) + \underline{\mathbf{X}}^{\mathbf{P}}(\mathbf{t}_{0})$ 

 $\underline{\hat{x}}^{p} + \underline{\hat{x}}^{m} = \frac{d}{dt} [\underline{x}^{p} + \underline{x}^{m}]$ 

so that

アン・ビー ようの

$$\underline{x}^{p}(t) + \underline{x}^{n}(t) = \underline{x}(t)$$

Also, it follows that

$$g(t) = g^{D} + g^{m}$$
, (3.69)

The minimum variance estimate of  $\underline{x}(t)$ , given  $\underline{Z}(t)$ , has been shown to be

ł

$$\frac{1}{2} = F(t)\frac{1}{2} + K(t)[\frac{1}{2} - H(t)\frac{1}{2}],$$
 (3.70)

where the optimal gain K(t) is defined by Equations (2.11) and (2.12). Considering Systems P and M, define linear estimates of  $\chi^p$  and  $\chi^n$ , using the optimal gain K(t) of Equation (3.70), as

$$f_{1} = F(t)g'' + K(t)[g^{0} - H(t)g'']$$
 (3.71)

$$\frac{1}{2}^{\mu} = F(t) \underline{z}^{\mu} + K(t) [\underline{z}^{\mu} - H(t) \underline{z}^{\mu}]$$
 (3.72)

Note that  $\frac{d}{d}^{\mu}$  and  $\frac{d}{d}^{\mu}$  are not the minimum variance estimators for  $\frac{d}{d}^{\mu}$  and  $\frac{d}{d}^{\mu}$ . However, it follows without difficulty that the minimum variance estimate of  $\frac{d}{d}$  is given by

$$\underline{s} = \underline{s}^{\mu} + \underline{s}^{\mu}$$
, (3.73)

Write the error in the estimate as

$$\begin{split} \mathbf{I} &= \mathbf{x} - \mathbf{\hat{x}} \\ &\approx (\mathbf{x}^{D} + \mathbf{x}^{n}) - (\mathbf{\hat{x}}^{m} + \mathbf{\hat{x}}^{n}) \\ &\approx (\mathbf{x}^{D} - \mathbf{\hat{x}}^{m}) + (\mathbf{x}^{n} - \mathbf{\hat{x}}^{n}) . \end{split}$$
(3.74)

Using Equation (3.74), the error covariance matrix' P(t|t) can be expressed as

### P(t|t) & #{[[(t) -2(t|t)][x(t) -2(t|t)]"}

 $= \mathbb{E}\left[\left(\underline{z}^{\mathsf{p}} - \underline{g}^{\mathsf{m}}\right)\left(\underline{z}^{\mathsf{p}} - \underline{g}^{\mathsf{m}}\right)^{\mathsf{T}}\right] + \mathbb{E}\left[\left(\underline{z}^{\mathsf{p}} - \underline{g}^{\mathsf{m}}\right)\left(\underline{z}^{\mathsf{m}} - \underline{g}^{\mathsf{m}}\right)^{\mathsf{T}}\right]$  $+ \mathbb{E}\left[\left(\underline{z}^{\mathsf{m}} - \underline{g}^{\mathsf{m}}\right)\left(\underline{z}^{\mathsf{p}} - \underline{g}^{\mathsf{m}}\right)^{\mathsf{T}}\right] + \mathbb{E}\left[\left(\underline{z}^{\mathsf{m}} - \underline{g}^{\mathsf{m}}\right)\left(\underline{z}^{\mathsf{m}} - \underline{g}^{\mathsf{m}}\right)^{\mathsf{T}}\right].$ 

But

$$\mathbb{E}\left[\left(\underline{x}^{p} - \underline{x}^{n}\right)\left(\underline{x}^{n} - \underline{x}^{\mu}\right)^{T}\right] = 0$$

since  $\underline{x}^{p}(t_{n}) = 0$  and  $\underline{x}(t)$  and  $\underline{x}(t)$  are uncorrelated white-noise processes; so

$$P(t|t) = E[(x^{\mu} - x^{\mu})(x^{\mu} - x^{\mu})^{T}] + E[(x^{\mu} - x^{\mu})(x^{\mu} - x^{\mu})^{T}], \qquad (3.75)$$

It is possible to state general bounds for P(t|t). Suppose that  $P^{m}(t|t)$  and  $P^{D}(t|t)$  represent the minimum variance error covariance matrices for Eystems M and P, respectively. Then, since  $\frac{2^{m}}{2}$  and  $\frac{2^{m}}{2}$  are not the minimum variance estimates for these mystems, it must certainly follow from Equation (3.75) that

- 44 1 + 4 inju

l nie i na hinnie 🖬 dati 🕯

$$P(t|t) \ge P^{0}(t|t) + P^{0}(t|t) , \qquad (3.76)$$

where the matrix notation  $A \ge B$  implies that the matrix A-B is non-negative-definite.

On the other hand, suppose that some gain other than K(t) is used to obtain estimates of both  $\underline{x}^{p}(t)$  and  $\underline{x}^{m}(t)$ . That is, a suboptimal gain  $K^{m}(t)$  is used in both subsystems. Then, from the definition of K(t) it must be true that

$$P(t|t) \leq P^{0,u}(t|t) + P^{mu}(t|t)$$
, (3.77)

where  $P^{pn}$  and  $P^{mn}$  represent the error covariance matrices associated with the suboptimal gain  $K^{n}(t)$ .

3.3.2 Unforced Dynamical Plant with Noisy Measurement Data

Consider the system described by

 $\dot{\mathbf{x}} = \mathbf{F}(t)\mathbf{z}$ ,  $\mathbf{x}(t_0) = \mathbf{x}_0$  (3.78)

$$\underline{\mathbf{z}} = \mathbf{H}(\mathbf{t})\underline{\mathbf{x}} + \underline{\mathbf{y}}, \tag{3.79}$$

This system is identical with System M but the superscripts have been eliminated to simplify the notation. The minimum variance estimate of  $\chi(t)$ , given Z(t), has been shown to be given by

$$= F(t) \frac{2}{3} + K(t) \left[ \frac{1}{2} - H(t) \frac{2}{3} \right], \qquad (3.80)$$

where

$$K(t) = P(t|t)H^{T}(t)R^{-1}(t)$$
 (3.61)

and

so that

$$\hat{P} = P(t)P + PP^{T}(t) - PH^{T}(t)R^{-1}(t)H(t)P . \qquad (3.62)$$

But in Section 3.2, it was shown (i.e. Equation (3.51)) that the solution of the matrix-Ricatti equation in this case is

$$P(t|t) = \Phi(t, t_0) [P^{-i} + M(t_0, t)]^{-i} \Phi^{T}(t, t_0), \qquad (3.83)$$

where  $M(t_x,t)$  is the observability matrix (see Equation (3.19)) and is defined to be

$$\mathbf{M}(\mathbf{t}_{0},\mathbf{t}) = \int_{\mathbf{t}_{0}}^{\mathbf{t}} \Phi^{T}(\tau,\mathbf{t}_{0}) \mathbf{H}^{T}(\tau) \mathbf{R}^{-1}(\tau) \mathbf{H}(\tau) \Phi(\tau,\mathbf{t}_{0}) d\tau \qquad (3.84)$$

and  $\phi(t, t_{c})$  is the transition matrix associated with the dynamical system (Eqn (3.78)).

From Equation (3.83), one sees that the behavior of the observability matrix essentially determines the behavior of P(t|t). One can show under appropriate conditions that  $M(t_0,t)$  is utrictly increasing with t (i.e.  $M(t_0,t_0) - M(t_0,t_1)$  is positive-definite for some  $t_g > t_1$ ).

It is clear from the definition that

$$\delta \mathbf{M}(\mathbf{t}_1, \mathbf{t}_2) \stackrel{\Delta}{=} \mathbf{M}(\mathbf{t}_0, \mathbf{t}_2) - \mathbf{M}(\mathbf{t}_0, \mathbf{t}_1) = \int_{\mathbf{t}_1}^{\mathbf{t}_2} \Phi^{\mathsf{T}}(\tau, \mathbf{t}_0) \mathbf{H}^{\mathsf{T}}(\tau) \mathbf{R}^{-1}(\tau) \mathbf{H}(\tau) \Phi(\tau, \mathbf{t}_0) d\tau$$
$$\delta \mathbf{M}(\mathbf{t}_1, \mathbf{t}_2) \ge 0 .$$

To consider the asymptotic behavior of  $M(t_0, t)$ , and therefore P(t|t), it is convenient to introduce the concept of *q-observability*.

Definition: A system is said to be *q*-observable for  $q \ge 0$  on an interval  $(t_0, t_N)$ , where  $t_N$  may be infinite, if and only if  $\delta M(t_j, t_k)$  is positive-definite for every  $t_j$ ,  $t_k$  such that  $t_N \ge t_k \ge t_j \ge t_0$  and  $t_k - t_j \ge q$ .

q-observability is a slight generalization of complete observability and is introduced to insure that the observability matrix  $M(t_0, t)$  is strictly increasing over intervals of duration q. For stationary systems there is no difference, so that this concept is useful only for time-varying systems. In this case, q-observability insures that the system is completely observable for every interval of duration q.

If a system is q-observable, then

 $M(t_o,t) > M(t_o,t-q) , \qquad t-q \ge t_o .$ 







Arvina Diete sit mirte Danille

(3.85)

(3,87)

Let the eigenvalues  $M(t_0,t)$  and  $M(t_0,t-q)$  be  $(\lambda_1^t > \lambda_2^t > \lambda_n^t)$  and  $(\lambda_1^{t-q} > \lambda_2^{t-q} > \dots > \lambda_n^{t-q})$ , respectively. It follows that

 $\lambda_1^t > \lambda_1^{t-q} ; \qquad \lambda_2^t > \lambda_2^{t-q} ; \qquad \lambda_n^t > t_n^{t-q} .$ 

Thus,  $M(t_0, t)$  is strictly increasing in the sense that all eigenvalues are strictly increasing as  $t \to \infty$ , slithough some may remain constant for intervals of length long than q.

For a symmetric matrix, the largest eigenvalue serves as a matrix norm (i.e. the spectral norm), so one can write

 $\|M(t_n,t)\| = \lambda_1^t$ ,

 $\lim_{t \to \infty} \|\mathbf{M}(\mathbf{t}_o, \mathbf{t})\| = \mathbf{m}$ 

so that one sees that

$$\lim_{t \to 0} \|M^{-1}(t_0, t)\| = \lim_{t \to 0} \frac{1}{\lambda_0^{-1}} = 0.$$
 (3.80)

and

Note that  $N^{-1}(t_0, t)$  exists if the system is completely observable for t. But Equation (3.66) implies that  $M^{-1}(t_0, t)$  converges to the zero matrix, since its norm vanishes.

Assuming q-observability, one sees from Equation (3.63) that the error covariance matrix P(t|t) is given approximately by

$$\mathbf{P}(\mathbf{t}|\mathbf{t}) \simeq \bar{\Phi}(\mathbf{t}, \mathbf{t}_a) \mathbf{M}^{-1}(\mathbf{t}_a, \mathbf{t}) \bar{\Phi}^{-T}(\mathbf{t}, \mathbf{t}_a)$$

for t sufficiently large. Since  $P_0^{-1}$  is positive-definite, it follows that

$$\begin{split} \|P(t|t)\| &= \|\Phi(t,t_0)[P_0^{-1} + M(t_0,t)]^{-1}\Phi^T(t,t_0)\| \\ &\leq \|\Phi(t,t_0)\|^2 \| \|P_0^{-1} + M(t_0,t)\}^{-1}\| \\ &\leq \|\Phi(t,t_0)\|^2 \| \|M^{-1}(t_0,t)\| . \end{split}$$

As a result of Equation (3.87), one concludes the following:

The error covariance matrix P(t) for the q-observable system (Eqns (3.78) – (3.79)) vanishes as  $t \to \infty$  if  $\|\|\mathbf{M}^{-1}(t, t_0)\|$  converges to zero faster than  $\|\Phi(t, t_0)\|^2$  increases. If the plant (Eqn (3.78)) is stable, then this is cortainly true.

This conclusion indicates, for a large class of systems, that the effect of measurement noise is eliminated by filtering over a sufficiently large period of time. Reference to Equation (3.81) also indicates that the gain matrix K(t) vanishes as the error covariance matrix tends to zero. As a result, the estimate  $\hat{g}$  tends to be characterized by a homogeneous, linear differential equation identical to the plant equation (Equ (3.79)). This above that the most recent measurement data have a decreasing influence on the estimate. While this behavior is understandable from an entirely theoretical viewpoint it should be observed that, unless Equations (3.78) and (3.79) aqually constitute an exact model of the system, the convergence of the P(tlt) can lead to unduly optimistic measures of the error in the estimate and can ultimately lead to filter "divergence" and nonsensical results<sup>34,37</sup>.

3.3.3 Noisy Dynamical Plant and Noise-Free Measurement Data

Consider a system containing no measurement noise

والمرجوع والمرجوع والمرجوع والمرجوع والمتحم والمتحم والمحمو والمحمو والمحمو والمحمو والمحمو والمحمو والمحمو والمحمو

 $\dot{\underline{x}} = F(t)\underline{x} + \underline{w}, \quad \underline{x}(t_0) = 0$  (3.88)

x = H(t)x. (3.89)

It should first be noted that Equations (3.88) and (3.89) are essentially identical with the system (Equations (2.40) and (2.53)) treated in Section 2.2.2. This case was discussed there, since the Kalman-Bucy equations can not be applied directly to this system. Instead, the filter is reduced to dimension (n-m) and is based on the system described by Equations (2.55), (2.56), and (2.58). Certainly, it is still possible to discuss the error covariance matrix P(t) associated with the error in the stimate of  $\chi(t)$  but, as a consequence of the considerations of Section 2.2, the P(t) can have, at most, wark n-m.

The nxn error covariance matrix P(t|t) can never vanish identically as long as the plant covariance matrix Q(t) is non-zero. This is an immediate consequence of the discussion in Section 3.2.1 relating to





Equation (3.30). In Equation (3.30) the matrix K(t) is arbitrary and the associated solution is given by Equation (3.35). The measurement noise covariance matrix R(t) can be set equal to zero (i.e. no measurement noise) in these equations. It is obvious that the term

$$\int_{t_0}^t \Psi(t_0,\tau) Q(\tau) \Psi^{\mathsf{T}}(t_0,\tau) \, \mathrm{d}\tau$$

is non-negative definite and never vanishes. The  $\Psi(t,t_0)$  is a transition matrix, so it is always non-singular. Therefore the matrix P(t)t) described by Equation (3.33) with R(t) sot equal to zero never vanishes for non-zero Q(t) . Thus, in contrast with the discussion of the preceding section, the error covariance matrix P(tit) associated with Equations (3.88) and (3.89) vanishes only under very special conditions, and is never positive-definite. Also it becomes apparent that the plant noise covariance matrix can serve to prevent the error covariance matrix from becoming too small in any principal direction.

### 3.3.4 Error Bounds

It was demonstrated in Section 3.3.1 that the error covariance matrix can be bounded by

$$P^{p}(t|t) + P^{n}(t|t) \leq P(t|t) \leq P^{n}(t|t) + P^{n}(t|t)$$
 (3.90)

But in Section 3.3.1 the  $P^{m}(t|t)$  was expressed more precisely by Equation (3.83). If the gain  $K^{m}(t)$  defined by Equations (3.81) and (3.82) were used in both the M and P systems, then Equation (3.83) can be also used in the upper bound and Equation (3.90) becomes

$$\Phi(\mathbf{t},\mathbf{t}_{n})[\mathbf{P}_{n}^{-1} + \mathbf{M}(\mathbf{t}_{n},\mathbf{t})]^{-1}\Phi^{T}(\mathbf{t},\mathbf{t}_{n}) + \mathbf{P}^{p}(\mathbf{t}|\mathbf{t}) < \mathbf{P}^{\mathsf{D}\mathsf{R}}(\mathbf{t}|\mathbf{t}) + \Phi(\mathbf{t},\mathbf{t}_{n})[\mathbf{P}_{n}^{-1} + \mathbf{M}(\mathbf{t}_{n},\mathbf{t})]^{-1}\Phi^{T}(\mathbf{t},\mathbf{t}_{n})$$
(3.91)

The lower bound can be relaxed by eliminating  $P^{p}(t|t)$  , so that

$$P(t|t) \ge \Phi(t, t_n) \left[ P_n^{-1} \ge M(t_n, t) \right]^{-1} \Phi^{T}(t, t_n) .$$
(3.92)

It is shown in Chapter 6 that this bound is essentially the Cramer-Rao inequality.

Consider the error that one has if there were no filtering for system P . Then

$$\underline{x}^{\mathbf{p}}(\mathbf{t}) = \int_{\mathbf{t}_{a}}^{\mathbf{t}} \Phi(\mathbf{t}, \tau) \underline{w}(\tau) \ \mathrm{d}\tau \ .$$

Assuming that w has zero mean, and assuming that no measurements are processed, the best estimate is

Then, it follows that

$$P^{\mathcal{P}_0}(t|t) \mathbb{E}[(\underline{x}-\underline{x})(\underline{x}-\underline{x})^T] = \int_{t_0}^{t} \Phi(t,\tau) \mathbb{Q}(\tau) \Phi^T(t,\tau) d\tau . \qquad (3.93)$$

But the matrix integral has been referred to as a stochastic controllability matrix by Kalman<sup>13</sup>. Let

$$\mathbf{W}(\mathbf{t}_{0},\mathbf{t}) \stackrel{\Delta}{=} \int_{\mathbf{t}_{n}}^{\mathbf{t}} \Phi(\mathbf{t},\tau) \mathbf{Q}(\tau) \Phi^{\mathsf{T}}(\mathbf{t},\tau) \, \mathrm{d}\tau \quad . \tag{3.94}$$

Note that Equation (3.93) is identical with Equation (3.47) of Section 3.2.1. Using Equation (3.93), the upper bound becomes

$$P(t|t) \leq \Phi(t, t_n) \left[ P_n^{-1} + M(t_n, t) \right]^{-1} \Phi^T(t, t_n) + W(t_n, t) .$$
(3.95)

Equations (3.92) and (3.95) provide more specific upper and lower bounds for the minimum variance error covariance matrix. From the discussion of Section 3.3.2, it can be seen that the upper and lower bounds are defined by  $W(t_0,t)$  and  $P^D(t|t)$  as t increases, since the  $P^m(t|t)$  tends to vanish for q-observable systems.

Kalvan has derived bounds for the error covariance matrix which bear some resemblance to those presented above<sup>13</sup>. It was shown that

$$P(t|t) \leq \Phi(t, \tau^*) N^{-1}(\tau^*, t) \Phi^{\Gamma}(t, \tau^*) + \Psi(\tau^*, t)$$
(3.96)

where  $\tau^*$  defines the time at which the system is completely observable and completely controllable (i.e.  $M(\tau^{*},t)$  and  $W(\tau^{*},t)$  are positive-definite). Equation (3.96) is very similar to Equation (3.95). Kalman's lower bound is

### $[\#^{-1}(\tau^{*},t) + \Phi^{T}(\tau^{*},t)M(\tau^{*},t)\Phi(\tau^{*},t)]^{-1} \leq P(t|t)$

This matrix will vanish for a q-observable and q-controllable system and so dues not appear to contain as much information as Equation (3.92).

A linear dynamical system with transition matrix  $\Psi(t,t_n)$  is said to be uniformly asymptotically stable if

 $\|\Psi(t,t_n)\| \leq \alpha e^{\beta(t-t_0)} \quad \text{for all } t \geq t_n ,$ 

where  $\alpha$ ,  $\beta > 0$ . It is possible to prove that, if the system (Equations (1.1) and (1.2)) is completely controllable and completely observable, then the filter is uniformly asymptotically stable. That is, the norm of the transition matrix obtained as the solution of the matrix differential equation

 $\Psi(t,t_0) = [F(t) - K(t)H(t)]\Psi(t,t_0), \qquad \Psi(t_0,t_0) = I$ 

satisfies the inequality stated above. The proof of this result is omitted and the reader is directed to Kalman<sup>13</sup>.

### 3.4 Behavior of Error Covariance for Stationary Systems

To obtain more insight into the behavior of the error covariance matrix, it is informative to consider stationary systems. In this case, the asymptotic behavior of the error covariance can be established more specifically and the upper bound can be investigated more thoroughly.

Consider the stationary lines; system

$$\mathbf{Z} = \mathbf{R}\mathbf{X} + \mathbf{Y}, \qquad (3.99)$$

where F and H are constant matrices. Assume also that the plant and measurement noise covariance matrices. Q and R are constant.

As has been shown above, the minimum variance error covariance matrix

$$P = FP + PF^{T} - PH^{T}R^{-1}HP + Q, \quad P(t_{0}) = P_{0}. \quad (3.100)$$

Let  $P_{\infty}$  be the steady-state value of Equation (3.100), as discussed in Section 3.2.2. This matrix can be determined from Equation (3.61) and is the solution of the algebraic equation

$$0 = FP_m + P_m F^T - P_m H^T R^{-1} HP_m + Q_{...}$$
(3.101)

Use P<sub>m</sub> to form a suboptimal gain for filter Application

$$K_{e} = P_{e} H^{T} R^{-1}$$
 (3.102)

This gain can be shown to be the gain obtained for the "classical" Wiener filter. The error covariance matrix for this suboptimal gain is obtained, using Equation (2.9) or Equation (3.30).

$$\dot{\mathbf{P}}_{g} = (\mathbf{F} - \mathbf{K}_{g} \mathbf{H}) \mathbf{P}_{g} + \mathbf{P}_{g} (\mathbf{F} - \mathbf{K}_{g} \mathbf{H})^{T} + (\mathbf{K}_{g} \mathbf{R} \mathbf{K}_{g}^{T} + \mathbf{Q}) , \quad \mathbf{P}_{g} (\mathbf{t}_{0}) = \mathbf{P} (\mathbf{t}_{0}) . \quad (3.103).$$

This is a linear matrix equation of the form treated in Section 3.2.1 with the steady-state value  $P_{m}$ . The fact that  $P_{m}(\infty) = P_{m}$  will be confirmed below.

 $\delta P(t|t) \stackrel{\Delta}{=} P(t|t) - P_{ac}, \qquad (3.104)$ 

where P(t|t) is the minimum variance error covariance matrix. Also, define

$$\delta P_{-}(t|t) \stackrel{Q}{=} P_{-}(t|t) - P_{-} \qquad (3.105)$$

and let  $\delta P$  and  $\delta P_{\mu}$  be assigned the initial conditions

Lat

$$\delta P(t_{o}) = \delta P_{u}(t_{o}) = P_{o} - P_{u} \triangleq \delta_{o} . \qquad (3.106)$$

Since the system is stationary, the initial time  $t_0$  can be assumed to be zero without loss of generality. Further, assume that

P. > P. .

This is a natural assumption and states that the initial uncertainty is greater than the steady-state error.



35

(3.97)

while hardentfur out finder





1.

and

It follows from Equations (3.104), (3.101), and (3.100) that

$$\delta \dot{\mathbf{P}} = (\mathbf{F} - \mathbf{K}_{\mathbf{H}}\mathbf{I})\delta \mathbf{P} + \delta \mathbf{P}(\mathbf{F} - \mathbf{K}_{\mathbf{H}}\mathbf{H})^{\mathrm{T}} - \delta \mathbf{P}\mathbf{H}^{\mathrm{T}}\mathbf{R}^{-1}\mathbf{H}\delta \mathbf{P}$$
(3.107)

But this is a matrix-Ricatti equation with no forcing term. This was shown in Section 3.2.1 to have the solution

$$\delta P(t|t) = \Psi(t, t_0) \delta_0 [I + N(t_0, t) \delta_0]^{-1} \Psi^T(t, t_0) , \qquad (3.108)$$

where  $\Psi$  is the transition matrix associated with the linear system with coefficient matrix ( $F \sim K_{\rm g}H$ ). But  $\delta P(t|t)$  vanishes as t tends to infinity if the system is completely observable, thereby confirming that  $P_{\rm co}$  is the steady-state solution of Equation (3.100).

One can also obtain, from Equations (3.105), (3.101), and (3.103),

$$\mathbf{SP}_{\mathbf{w}} = (\mathbf{F} - \mathbf{K}_{\mathbf{w}}\mathbf{H})\mathbf{SP}_{\mathbf{w}} + \mathbf{SP}_{\mathbf{w}}(\mathbf{F} - \mathbf{K}_{\mathbf{w}}\mathbf{H})^{\mathsf{T}}$$
(3.109)

This is a homogeneous linear matrix equation and, from Equation (3.4'), it follows that  $\delta P_{\rm m}$  tends to vanish if  $(F-K_{\rm m}H)$  represents a stable system. This is indiced the case for an observable and controllable system, so that the steady-state value of  $P_{\rm m}$  is established to be  $P_{\rm m}$ . Thus, the asymptotic behavior for the suboptimal gain  $K_{\rm m}$  must be very similar to that for the minimum variance gain.

It is true, from Equation (3.47), that

$$\delta P_{\bullet}(t|t) = e^{At} \delta_{\bullet} e^{A^{T}t} , \qquad (3,110)$$

since the transition matrix  $\Psi(t,0)$  associated with

$$\Psi(t,0) = A\Psi(t,0)$$

$$\Psi(t,0) = e^{At}$$

One also finds, waing Equation (3, 51), that

$$\delta P(t|t) = e^{At} \left[ \delta_{1}^{-1} + M(0, t) \right]^{-1} e^{A^{T}t} , \qquad (3, 111)$$

The difference between  $P_{ij}(t|t)$  and P(t|t) can be established using Equations (3.110) and (3.111). First, note that

$$\delta P(t|t) = e^{At} [I + \delta_0 M(0, t)]^{-1} \delta_0 e^{A^T t}$$
$$= e^{At} [I + \delta_0 M(0, t)]^{-1} e^{-At} \delta P_{W}(t|t) \qquad (3.112)$$

Rearranging terms, one has

$$P_{\Psi}(t|t) - P(t|t) = \delta P_{\Psi}(t|t) - \delta P(t|t) = e^{At} \delta_{0} M(0,t) e^{-At} \delta P(t|t)$$

$$\delta P(t|t) = o^{A^{*}} N^{-1} (0, t) \delta_{0}^{-1} o^{-A^{*}} [P_{u}(t|t) - P(t|t)]$$
 (3.113)

Equating Equations (3.111) and (3.113), it follows that

$$\mathbf{P}_{*}(t|t) - \mathbf{P}(t|t) = \mathbf{A}^{t} \delta_{*} \mathbf{M}(0,t) [\delta_{*}^{-1} + \mathbf{M}(0,t)]^{-1} \mathbf{e}^{\mathbf{A}^{T}t}$$
(3.114)

Equation (3.114) provides a relation for the difference between the error covariance of the suboptimal filter and the minimum variance filter. An inequality is easily derived from Equation (3.114) that can be used to describe the behavior of the error covariance. Let  $\lambda_1$  and  $\lambda_n$  denote the largest and smallest eigenvalues of A. (Note that  $\lambda_1$  and  $\lambda_n$  are negative since A represents a stable system.) Then the spectral norm of the transition matrix is

-1 +

$$\|\mathbf{e}^{\mathbf{At}}\| = \mathbf{e}^{\lambda_{1} \mathbf{t}} \tag{3.115}$$

$$\|e^{-AC}\| = e^{-n^2}$$
, (3.116)



and it follows that

$$\begin{split} \| \mathbf{M}(\mathbf{0}, \mathbf{t}) \| &= \| \int_{0}^{t} \mathbf{e}^{\mathbf{A}^{T} \mathbf{1}} \mathbf{H}^{T} \mathbf{R}^{-1} \mathbf{H} \mathbf{e}^{\mathbf{A} \mathbf{\tau}} \, d\tau \| \\ &\leq \int_{0}^{t} \| \mathbf{e}^{\mathbf{A}^{T} \mathbf{\tau}} \| \| \mathbf{H}^{T} \mathbf{R}^{-1} \mathbf{H} \| \| \mathbf{e}^{\mathbf{A} \mathbf{\tau}} \| \, d\tau \\ &= \| \mathbf{H}^{T} \mathbf{R}^{-1} \mathbf{H} \| \int_{0}^{t} \mathbf{e}^{t \lambda_{1} \mathbf{\tau}} \, d\tau \\ &= \frac{\| \mathbf{H}^{T} \mathbf{R}^{-1} \mathbf{H} \|}{2 \lambda_{1}} \left( \mathbf{e}^{t \lambda_{1} \mathbf{t}} - 1 \right) \\ &= \frac{\| \mathbf{H}^{T} \mathbf{R}^{-1} \mathbf{H} \|}{2 |\lambda_{1}|} \left( 1 - \mathbf{e}^{2 \lambda_{1} t} \right) \,. \end{split}$$
(3. 117)

One also sees that

$$\left\| \left[ \delta_{-1}^{-1} + M(0, t) \right]^{-1} \right\| \le \left\| \delta_{-1} \right\| . \tag{3.118}$$

Using Equations (3.115) through (3.118) in (3.114), it is found that

$$\|P_{W}(t|t) - P(t|t)\| \leq e^{2\lambda_{1}t} \|\delta_{0}\|^{2} \frac{\|H^{T}R^{-1}H\|}{2|\lambda_{1}|} \left(1 - e^{2\lambda_{1}t}\right). \qquad (3.119)$$

The time-dependent factor reaches a maximum which is determined by considering

$$\frac{d}{dt} \left[ e^{z\lambda_1 t} \left( 1 - e^{z\lambda_1 t} \right) \right] = 0.$$

$$2e^{z\lambda_1 t} = 1.$$

This implies that

so that the maximum occurs at

$$t = \frac{1}{2\lambda_1} \log_{0} \frac{1}{2}.$$

The maximum is easily found to be  $\frac{1}{4}$ , so that the maximum difference is

$$\|P_{\mathbf{y}}(t|t) - P(t|t)\| \leq \frac{\|\delta_{\mathbf{0}}\|^2 \|\mathbf{H}^T \mathbf{R}^{-1} \mathbf{H}\|}{8|\lambda|} \quad \text{for all } t. \qquad (3.120)$$

The difference between the suboptimal and optimal error covariances is seen to depend upon the initial uncertainty  $\delta_0$ , the signal-to-noise ratio as described by  $||\mathbf{H}^T \mathbf{R}^{-1} \mathbf{H}||$  and the smallest time-constant  $|\lambda_1|$  of the filter dynamics. As one should expect, the larger the initial uncertainty or the better the signal-to-noise ratio, the greater the error introduced by the suboptimal gain. On the other hand, if the filter system has a large time constant, the suboptimal gain does not necessarily cause a significant deterioration of filter performance.

To summarize, the determination of the steady-state value of the optimal filter permits the definition of a suboptimal gain (i.e. the gain of the Wiener filter) which in many cases may provide satisfactory filter performance. A bound on the norm of the difference between the optimal and suboptimal error covariance matrices is determined which involves the intuitively obvicus elements of the system which determine filter performance. More extensive discussion of this special problem is provided by Singer<sup>35</sup>.

### 4. CONCLUSION

In this concluding section the intent is to state the principal results and the highlights of Sections 2 and 3 in order to allow the casual reader to sidestep the morass of details found there and to provide a support for the more interested reader with which he may another himself as he slides into morass.

estion 1.2 provides a statement of the mathematical model in terms of Equations (1.1) and (1.2) and then actively the Unbiased Minimum Variance, Linear Estimation Problem. These aspects shall not be repeated here.







### Section 2.1: Linear Prediction and Filtering

Section 2.1 contains the solution of the problem stated in Section 1.2 in terms of the Kalman-Budy filter equations. This is first accomplished in Section 2.1, using arguments which, hopofully, appeal to intuition and provide some understanding of the basic character of the solution. This development is put on a more rigorous basis in Section 2.1.2, wherein the properties of the measurement residual process

$$\mathbf{r}(\mathbf{t}) \stackrel{\text{def}}{=} \mathbf{z}(\mathbf{t}) - \mathbf{H}(\mathbf{t}) \mathbf{\hat{z}}(\mathbf{t} | \mathbf{t}) \tag{4.1}$$

are examined and then used to derive the filter equations. Thus, in the first two paragraphs, it is shown in two ways that the solution of the linear filtering and rediction problem stated in Section 1.2 is provided by the system

$$\hat{g}(t|t) = P(t)\hat{g}(t|t) + K(t)Y(t),$$
 (4.2)

where

$$\dot{\underline{x}}(t_0|t_0) = \mathbb{E}[\underline{x}(t_0)]$$

The optimal gain K(t) is given by

$$K(t) = P(t|t)H^{T}(t)R^{1}(t)$$
 (4.3)

where the error covariance matrix

$$\mathbb{P}(\mathbf{t}|\mathbf{t}) \stackrel{\text{\tiny{def}}}{=} \mathbb{E}\{[\mathbf{x}(\mathbf{t}) - \mathbf{\hat{x}}(\mathbf{t}|\mathbf{t})][\mathbf{x}(\mathbf{t}) - \mathbf{\hat{x}}(\mathbf{t}|\mathbf{t})]^T\}$$
(4.4)

is described by the matrix-Ricatti equation

$$\frac{d}{dt} P = F(t)P + PF^{T}(t) - PH^{T}(t)R^{-1}(t)H(t)P + Q(t) , \qquad (4.5)$$

where

$$P(t_0|t_0) =$$

Some additional aspects of the filtering problem are considered in Section 2.1.3. First, the concept of orthogonal projections is introduced and it is proven that the estimate that is formed as a linear function of data minimizes the mean-square error if, and only if, the error in the estimate is orthogonal to the data. This result is used to derive the Wiener-Hopf equation for the system (Equations (1.1) and (1.2)).

Next, the original problem is generalized to allow cross-correlation C(t) between the plant and measurement noise processes to be non-zero. It is shown that the correlation causes the gain matrix K(t) to become

$$K(t) = [P(t|t)|I^{T}(t) + C(t)]R^{-1}(t)$$
(4.6)

and the error covariance matrix is modified to become

$$\dot{P} = F(t)P - PP^{T}(t) - K(t)[C^{T}(t) + H(t)P] + Q(t)$$
(4.7)

The effect of a deterministic forcing function  $\underline{d}(t)$  in the plant is considered and it is shown that this term requires a trivial change to the estimate. Equation (4.2) is modified to

$$\underline{z} = F(t)\underline{z} + \underline{i}(t) + K(t)\underline{r}(t)$$
, (4.8)

Finally Section 2.1.3 is concluded by proving that the conditional mean  $\mathbb{E}[\underline{x}(t)|\underline{Z}(\tau)]$  provides the unbiased, minimum variance estimate for a random variable  $\underline{x}(t)$ . This result is useful in giving a probabilistic interpretation of minimum variance estimates. For the linear problem described in Section 1.2, it follows from this property that the unbiased minimum variance estimate of  $\underline{x}(t)$  for the system (Equations (1.1) and (1.2)) is actually a linear estimate, thereby confirming that the linear Kalman-Buoy filter provides optimal estimate even when nonlinear estimates are considered.

The time-discrete system equivalent to the continuous-time model described by Equations (1.1) and (1.2) is introduced in Section 2.1.4 and the filter equations for this system are stated without proof.

### Section 2.2: The Colored-Noise Problem

In Section 2.2, the white-noise assumptions of Section 1.2 are relaxed to allow the plant and measurement noise processes to exhibit corellation between different times (i.e. the processes are colored). The restriction that the process, say g(t), can be described by a linear shaping filter

$$\dot{\mathbf{n}} = A(\mathbf{t})\mathbf{n} + \underline{\mathbf{v}}(\mathbf{t}) \tag{4.9}$$

ANNAL AND AND AND A SIA



In Soction 2.2.2 the solution for the colored measurement noise problem is determined by noting that the treatment of data without white-noise is accomplished considering a transformation of the state that allows the dimension of the filter equations to be reduced. Several special cases are treated that have appeared in the literature.

to the state. It is seen that this approach is satisfactory for colored plant noise but is inadequate when

### Section 2.3: Linear Smoothing

To complete the discussion of linear estimation theory, the linear smoothing problem is solved using the innovations (or residual) process  $\underline{r}(t)$  introduced in Section 2.1.2. It is shown that the estimate of  $\underline{x}(t)$  given the data  $Z(\tau)$ ,  $\tau > t$ , has the general form

$$\underline{\hat{x}}(t|\tau) = \underline{\hat{x}}(t|t) + P(t|t) \int_{-\infty}^{\infty} \Phi^{T}(s,t) H^{T}(s) R^{-1}(s) r(s) ds , \qquad (4.10)$$

with associated error covariance matrix described by

there are measurement data which do not contain white-noise.

$$P(t|\tau) \triangleq E\{[x(t) - \hat{x}(t|\tau)][x(t) - \hat{x}(t|\tau)]^{T}\} = P(t|t) - P(t|t)M_{-}(t,\tau)P(t|t), \quad (4,11)$$

where

61

$$\mathbf{M}_{\mathbf{s}}(\mathbf{t}, \tau) = \int_{\mathbf{t}}^{\tau} \Psi^{\mathrm{T}}(\mathbf{s}, \mathbf{t}) \mathbf{H}^{\mathrm{T}}(\mathbf{s}) \mathbf{R}^{-1}(\mathbf{s}) \mathbf{H}(\mathbf{s}) \Psi(\mathbf{s}, \mathbf{t}) \, \mathrm{d}\mathbf{s} \, .$$

Equations (4.10) and (4.11) represent the general solution of the smoothing problem. Three classes of smoothing problems, characterized by the time t at which the estimate is to be determined and the interval  $\tau$  for which data exists, are stated and the solution for each is given.

### Section 3.1: Observability of Deterministic System

The concept of observability is introduced in terms of a deterministic linear system obtained from Equations (1.1) and (1.2) by considering the plant noise  $\underline{y}(t)$  as known and the measurement noise  $\underline{y}(t)$  to be identically zero. If all states  $\underline{y}(t)$  of the deterministic system can be determined from knowledge of the input  $\underline{y}(\tau)$ .  $t_0 \leq \tau \leq t$  and the output  $\underline{Z}(t)$ , then the system is completely observable. It is shown that the system is completely observable if, and only if, the observability matrix

$$M(t_0,t) = \int_{t_0}^t \Phi^{T}(\tau,t_0) H^{T}(\tau) R^{-1}(\tau) H(\tau) \Phi(\tau,t_0) d\tau \qquad (4.12)$$

is positive-definite for some  $t > t_n$  .

When the linear system is stationary it is shown that the observability criterion can be expressed in terms of the n x mn matrix

$$U = [H^{T}F^{T}H^{T}, \dots, (F^{T})^{n-1}H^{T}], \qquad (4.13)$$

The system is completely observable if this matrix has rank n .

In Section 3.2 it is shown that the least-squares estimate of an arbitrary initial state exists if the system is completely observable. That is, a least-squares estimator exists if, and only if, the observability matrix is positive-definite.

### Section 3.2: The Matrix-Ricatti Equation

The error covariance matrix was found in Section 2 to be described by a matrix-Ricatti equation. This nonlinear matrix differential equation is investigated in this section and it is shown that it can be rewritten by a system of linear matrix differential equations. Thus, the matrix-Ricatti equation

$$\dot{\mathbf{P}} = \mathbf{P}(\mathbf{t})\mathbf{P} + \mathbf{P}\mathbf{F}^{\mathrm{T}}(\mathbf{t}) - \mathbf{P}\mathbf{H}^{\mathrm{T}}(\mathbf{t})\mathbf{R}^{-1}(\mathbf{t})\mathbf{H}(\mathbf{t})\mathbf{P} + \mathbf{Q}(\mathbf{t}); \qquad \mathbf{P}(\mathbf{t}_{n}) = \mathbf{P}_{n}$$
(4.14)

oan be written as

$$\hat{\mathbf{X}} = \mathbf{F}(\mathbf{t})\mathbf{Y} + \mathbf{Q}(\mathbf{t})\mathbf{Z}; \quad \mathbf{Y}(\mathbf{t}_0) = \mathbf{P}_0$$
 (4.15)

$$\dot{Z} = H^{T}(t)R^{-1}(t)H(t)Y - F^{T}(t)Z; Z(t_{n}) = I.$$
 (4.16)

The solution of Equation (4.14) is obtained from the solutions of Equations (4.15) and (4.16) an

Nore explicit solutions can be obtained when either H(t) or Q(t) vanish. These solutions are derived. Also, additional insights are possible into the character of Equation (4.14) when the system is stationary. This special case is discussed in some detail.

### Section 3.3: Bounds for the Error Covariance Matrix

The error covariance matrix provides the primary statistical measure of the performance of the estimator. This statistic is actually independent of the measurement data so it can be investigated before the Kalman-Bucy filter equations are used to process data for a given system. Upper and lower bounds for this matrix are established in this system in such a manner that the fundamentally different roles of the plant noise and the measurement noise processes are clearly displayed. It is shown that P(t|t) is bounded below by

$$P(t|t) \ge \Phi(t, t_n) [P_n^{-1} + N(t_n, t)]^{-1} \Phi^{T}(t, t_n) + P^{P}(t|t) , \qquad (4.18)$$

where P<sup>P</sup>(tit) is a non-negative-definite (nover positive-definite) matrix and the first term vanishes as t becomes large for q-observable systems. (This term is defined in Section 3.3.) The upper bound is given by

$$P(t|t) \leq \Phi(t,t_n) [P_n^{-1} + M(t_n,t)]^{-1} \Phi^T(t,t_n) + W(t_n,t) , \qquad (4.19)$$

where the stochastic controllability matrix  $W(t_0,t)$  is defined to be

$$W(t_0,t) \stackrel{\Delta}{=} \int_{t_0}^{t} \Phi(t,\tau) Q(\tau) \Phi^{T}(t,\tau) d\tau$$

### Section 3.4: Behavior of the Error Covariance for Stutionary Systems

The error dovariance matrix for stationary systems can be studied in more detail using the steady-value lim  $P(t|t) = P_0$ . A suboptimal gain  $K_{\rm W}$  equivalent to the Wiener filter gain can be defined that is useful in examining the influences that affect the relative performance of the optimal and Wiener filters. Letting  $P_{\rm W}(t|t)$ denote the error covariance matrix of a filter using the constant Wiener gain, it is shown that the difference is bounded by

$$\left\| \mathbb{P}_{g}(t|t) - \mathbb{P}(t|t) \right\| \leq \frac{\|S_{0}\|^{2} \|\mathbf{H}^{T} \mathbf{K}^{T} \mathbf{H}\|}{8|\lambda_{1}|} \quad \text{for all } t.$$
 (4.20)

where  $\delta_0 = F(t_0) - F_{z_0}$ , and where  $||H^T R^{-1} H||$  represents the signal-to-noise ratio of the measurement system and  $|\lambda_1|$  is the smallest time constant of the filter dynamics.

This concludes a summary of the presentation found in Sections 2 and 3. Many topics have been omitted from this discussion and other topics have been dealt with somewhat superficially. Many of these omissions and superficialities are given their deserved airing in subsequent chapters and their variety and number provide ab indication of the breadth, depth, and importance of unbiasod, minimum variance linear estimation theory in engineering theory and application.



(4.17)

WIND TANK DAN

### REFERENCES

1. Gauss, K.F. Theory of the Motion of the Heavenly Bodies Moving about the Sun in Conic Sections. Dover, New York, 1963. 2. Wiener, N. The Extrapolation, Interpolation and Smoothing of Stationary Time Series. Wiley. New York, 1949. 3. Kolmogorov, A.N. Interpolution und Extrupolation von Stationären Zufälligen Folgen. Bulletin of the Academy of Sciences, USSR, Mathematical Series, Vol.5, 1941, pp.3-14. 4 Bode, H.W. A Simplified Derivation of Linear Least-Squares Smoothing and Prediction Theory. Shannon, C.E. Proceedings, Institute of Radio Engineers, Vol. 38, 1950, pp. 417-425. 5. Zadoh, L.A. An Extension of Wiener's Theory of Prediction. Journal of Applied Physics. Regazzini, J.R. Vol.21, 1950, np. 645-655. 6. Shinbrot M. Optimization of Time-Varying Linear Systems with Nonstationary Inputs. Transactions, American Society of Mechanical Engineers, Vol. 80, 1958, pp. 457-462. 7. Haich, H. On the Optimum Synthesis of Multipole Control Systems in the Wiener Sense. Leondes, C.T. Institute of Radio Engineers, Transactions on Automatic Control, Vol. AC-4, No. 2, Nov. 1959, pp. 18-29. 8. Carlton, A.G. Linear Estimation in Stochastic Processes. The Johns Hopkins University, Applied Physics Laboratory, Bumblebee Series Report 311, Baltimore, Maryland, 1962. 9. Carlton, A.G. Recent Developments in Fixed and Adaptive Filtering. AGARDograph 21, 1956. Follin, J.W., Jr. 10. Swerling, P. First-Order Error Propagation in a Stage-Wise Smoothing Procedure for Satellite Observations. Journal of the Astronautical Sciences, Vol. 8, 1989, pp. 46-52. A New Approach to Linear Filtering and Prediction Problems. Journal of Basiu 11. Kalman, R.E. Engineering, Vol. 82D, 1960, pp. 95-108. 12. Kalman, R.E. New Results in Linear Filtering and Prediction Theory. Journal of Basic Engineer-Bucy, R.S. ing, Vol. 83D, 1961, pp. 95-108. 13. Kalman, R.E. New Methods in Wiener Filtering Theory. Proceedings of the First Symposium of Engineering Application of Random Function Theory and Probability. Wiley, New York, 1983, pp. 270-388. 14. Coddington, E.A. Theory of Ordinary Differential Equations. McGraw-Hill, 1955. Levinson. N. 15. Buoy. R.S. Filtering for Stochastic Processes with Applications to Guidance. Intermoience Joseph, P.D. Publishers, New York, 1968. 18. Kmilath, T. An Innovations Approach to Least-Squares Estimation. Part I: Linear Filtering in Additive White-Noise. Institute of Electrical and Electronic Engineers, Transantions on Automatic Control, Vol. AC-13, No. 8, Dec. 1968, pp. 646-655. 17. Athans, M. A Direct Derivation of the Optimal Linear Filter Using the Maximum Principle. Tso. E. Institute of Electrical and Electronic Engineers, Transactions on Automatic Control. Vol.AC-12, No. 6, Dec. 1967, pp. 690-698. 1B. Roid. W.T. A Matrix Differential Equation of Ricatti Type. American Journal of Mathematics, Vol.68, 1946, pp. 237-246. 19. Sorenson, H.W. Kalman Filtering Techniques. Chapter 5 of "Advances in Control Systems", Vol.3, C.T.Leondes, Editor, Academic Press, New York, 1966. Controllability and Observability of Linear, Stochastic, Time-Discrete Control 20. Borenson, H.W. Systems. Chapter 2 of "Advances in Control Systems", Vol. 6, C.T. Leondes, Editor, Academic Press, New York, 1968. 21. Stear, E.B. Synthesis of Shaping Filters for Nonstationary Stochastic Processes and Their Use. University of California, Report 61-50, Los Angeles, California, 1961. Letimation of State Variables via Dynamic Programming. 1964 JACC Proceedings, June 1067, pp. 376-381.







22. Cox. H.

P

J.

		·
42		
23.	Bryson, A.E. Johansen, D.E.	Linear Filtering for Time-Varying Systems Using Measurements Containing Colored- Noise. Institute of Electrical and Electronic Engineers. Transactions on Auto- matic Control, Vol.AC-10, Jan.1985, pp.4-10.
24.	Bucy, R.S.	Optimal Filtering for Colored-Noize. Journal of Mathematical Analysis and its Applications, Vol.20, Oct.1967, pp.1-8.
25.	Stear, E.B. Stubberud, A.R.	Optimal Filtering for Gauss-Markov Noise. International Journal of Control, Vol.8, No.2, 1908, pp.123-130.
26.	Sarachik, P.E.	State Estimation from Measurements with Correlated Noise without Using Differen- tiators. To appear in Institute of Electrical and Electronic Engineers, Trans- actions on Automatic Control.
27.	Kuilath, T. Frost, P.	An Innovation Approach to Least-Squares Estimation. Part II: Linear Smoothing in Addition White-Noise. Institute of Wiectrical and Electronic Engineers, Trans- actions on Automatic Control, Vol.AC-13, No.8, Dec. 1968, pp. 655-660.
28.	Meditch, J.S.	On Optimal Linear Smoothing Theory. Journal of Information and Control, Vol. 10, 1967, pp. 598-615.
29.	Bryson, A.S. Frazier, M.	Smoothing for Linear and Nonlinear Dynamic Systems. Auronautics Systems Division, Wright-Pattermon AFB, Ohio, Technical Report ASD-TDR-63-119, February 1969.
<b>3</b> 0.	Neditch, J.S.	On Optimal Fixed-Point Linear Smoothing. International Journal of Control, Vol.6. 1967, pp.189-199.
31.	Kalman, R.E.	Mathematical Description of Linear Dynamical Systems. Journal BIAM Control, Series A, Vol. 1, No. 2, 1963, pp.152-192.
32.	Sorerson, H.W.	On the Error Behavior in Linear Minimum Variance Estimation Problems. Institute of Electrical and Electronic Engineers. Transactions on Automatic Control. Vol.AC-12, No. 5, 1967, pp.557-562.
83.	0' Donne 11, J. J.	Asymptotic Solution of the Matrix Ricatti Equation of Optimal Control. Proceedings, 4th Annual Allerton Conference on Circuit and System Theory, pp.577-586.
34.	Yaughn, D.B.	A Negative Exponential Solution for the Matrix Ricatti Squation. Thetitute of Electrical and Electronic Engineers, Transactions on Automatic Control, Vol.AC-14, No.1, Feb. 1960, pp. 72-75.
35.	Singer, R.A.	The Design and Synthesis of Linear Multivariable Systems with Application to State Estimation. Stanford Electropics Laboratory Report 6303-8, Stanford, 1968.
36.	Denham, W.F. Pires, S.	Sequential Estimation when Measurement Function Nonlinearity is Computable to Measurement Error. AIAA Journm1, Vol.4, No.6, June 1966, pp.1071-1076.
87.	Schlee, F.H. et ml.	Divergence in the Kalman Filter. AIAA Journal, Vol.5, No.6, June 1967, pp. 1114-1120.

+4



....

half Classics

al Concentration (Child

<u>R</u>i

### CHAPTER 2 - FURTHER COMMENTS ON THE DERIVATION OF KALMAN FILTERS

### BECTION I - DERIVATION OF THE KALMAN FILTERING EQUATIONS FROM ELEMENTARY STATISTICAL PRINCIPLES

**MARKEN** 

P.M.Barham and D.E.Humphries

hv

### Avionics Department, Royal Aircraft Establishment, Faraborough, Nampahire, England









(2.4)

CHAPTER 2 - FURTHER COMMENTS ON THE DERIVATION OF KALMAN FILTERS

### SECTION I - DERIVATION OF THE KALMAN FILTERING EQUATIONS FROM Elementary statistical principles

### P.M. Barham and D.E. Humphries

### 1. INTRODUCTION

いたとうない。「などうちょう」である。

In less than a decade Kalman filtering has become an established technique within the resources of the Aerospace systems designer. Unfortunately the fact that it is an advanced statistical concept, mainly described by mathematicians on the assumption that the reader is familiar with advanced statistical theory, has too often had the effect that the practical engineer has accepted a set of equations while feeling insufficiently qualified to attempt a full appreciation of their significance. The sim in this shaptor is to provide a simple, though rigorous, derivation of the Kalman filtering equations in a form which should appeal to the less mathematical reader. For this firston more steps have been included in the derivation of the equations than would be necessary for those fully a list with statistical methods.

hu necessary statistical theory is Coeloped in Section 2 as an extension of familiar and elementary sampling theory. Whereas a knowledge of simple more in notation and samipulation is unavoidable for the development and application of the practical multi-dimensional filtyr, the theory is first developed for a simple single-dimension case which leads directly from the concepts of mean value and standard deviation.

### R. OPTIMUM COMBINATION OF INDEPENDENT ESTIMATES

The Nalman filtering process consists of combining two independent estimates of a variable to form a weighted mean. The weighting factor is chosen to yield a mean with minimum variance and hence maximum probability. One of the estimates is derived by updating a previous best estimate in accordance with the known equations of motion and the other estimate is obtained from a measurement. The form of the required weighting factor is derived bolow, first for a single-dimension case and secondly for the general multi-dimensional form required for the Kalman filter.

### 2.1 The Single-Dimension Case

Let  $x_1$  and  $x_2$  is two independent estimates of a quantity x with variances  $\sigma_1^2$  and  $\sigma_2^2$  respectively. It is required to combine these estimates to form a weighted mean corresponding to the overall best estimate  $(\hat{x})$  where "best estimate" means the minimum variance estimate.

Now the general form of the weighted mean of  $x_1$  and  $x_2$  is

$$X = (1 - w)x_1 + wx_2$$
 (2.1)

Thus the expected or mean value of X [written E(X)] is given by

$$E(X) = (1 - w) E(X_1) + wE(X_2) , \qquad (2, 2)$$

By definition the variance of a quantity x is

$$E[{x - E(x)}^{2}],$$
 (2.3)

Hence the variance  $(\sigma^2)$  of X is given by

$$\sigma^{2} = \mathbb{E}[\{\overline{x} - \mathbb{E}(\overline{x})\}^{2}] = \mathbb{E}[\{(1 - w)x_{1} + wx_{2} - (1 - w)\mathbb{E}(x_{1}) - u\mathbb{E}(x_{2})\}^{2}]$$
  
=  $\mathbb{E}[(1 - w)^{2}(x_{1} - \mathbb{E}(x_{1}))^{2} + w^{2}(x_{2} - \mathbb{E}(x_{2}))^{2} - 2w(1 - w)(x_{1} - \mathbb{E}(x_{1}))(x_{2} - \mathbb{E}(x_{2}))]$   
=  $(1 - w)^{2}\mathbb{E}[\{x_{1} - \mathbb{E}(x_{1})\}^{2}] + w^{2}\mathbb{E}[\{x_{2} - \mathbb{E}(x_{2})\}^{2}]$ 

or since

$$\mathbb{E}[\{x_1 - \mathbb{E}(x_1)\} \{x_2 - \mathbb{E}(x_2)\}] = 0,$$

as  $x_1$  and  $x_2$  are independent estimates so that  $(x_1 - E(x_1))$  and  $(x_2 - E(x_2))$  are uncorrelated. To determine the value of w for which  $\sigma^2$  is minimum, Equation (2.4) may be partially differentiated with respect to w. Thus

 $\frac{\partial \sigma^2}{\partial n} = -2(1-n)\sigma_1^2 + 2n\sigma_2^2 = 0$ , **PRECEDING** PAGE BLANK

-#6

 $\sigma^2 = (1-w)^2 \sigma_1^2 + w^2 \sigma_2^2$ 

and

or

giving the optimum value \* of weighting factor as

$$\Phi = \frac{\sigma_1^2}{\sigma_1^2 + \sigma_2^2}.$$
 (2.5)

Substitution in Equation (2.1) gives

$$\Sigma = \frac{\sigma_s^2 x_1 + \sigma_s^2 x_2}{\sigma_1^2 + \sigma_s^2} \quad . \tag{2.6}$$

and in Equation (2.4) gives the variance of R as

$$\hat{\sigma}^{\pm} = \frac{\sigma_1^{\pm} \sigma_2^{\pm}}{\sigma_1^{\pm} + \sigma_2^{\pm}}, \qquad (2.7)$$

If  $x_1$  is considered to be a measurement used to improve an updated esimilate  $x_1$ , Equations (2.4) and (2.7) can be put into the more convenient forms

 $\hat{\mathbf{x}} = \mathbf{x}_1 - \hat{\mathbf{x}}(\mathbf{x}_1 - \mathbf{x}_2)$  (2.8)

$$\hat{\sigma}^{\mathfrak{g}} = \sigma^{\mathfrak{g}}(1-\theta) , \qquad (2.9)$$

which more clearly show how the estimate and its variance are "improved" by the measurement in readiness for further updating. Equation (2.8) size shows the close analogy to negative feedback.

### 2.1 The Multi-Dimensional Case

Let  $X_1$  and  $X_2$  be two  $n \times 1$  matrices representing two independent estimates (i.e. with uncorrelated errors) of an n-dimensional vector quantity X.  $P_1$  and  $P_2$  are two  $n \times n$  matrices representing the variances of  $X_1$  and  $X_2$  respectively.

In general the weighted mean of  $X_1$  and  $X_2$  is given by

$$X = (I - W)X_1 + WX_2$$
  
 $\overline{X} = X_1 - W(X_1 - X_2)$ , (2.10)

where W is an arbitrary  $n \times n$  weighting matrix and X is the unit matrix. The best estimate of X (denoted by X) will be given by (2.10) when W is chosen so that the variance of  $\overline{X}$  is a minimum in the sense considered below.

in many practical cases, however, the two estimates are not of equal dimensions, one of them being some function (N, say) of the individual elements of X (for example, a measurement relating to some only of the elements of X). In general M may be an max n restangular matrix operating on X to give an m-dimensional estimate Y .

The general problem therefore is that of forming an optimum estimate  $(\hat{X})$  of X from one estimate  $(X_1$  with variance P') of X and an estimate  $(Y_2$  with variance V) of Y(=MX). If we lot the weighting matrix in (2.10) be W = KM, where K is another arbitrary weighting matrix, then

$$X = X_1 - KM(X_1 - X_2)$$
  
=  $X_1 - K(MX_1 - Y_2)$   
=  $(I - KM)X_1 + KY_2$ . (2.11)

By definition the variance F of X is given by

 $\overline{P} = \mathbb{E}\left\{ [\overline{X} - \mathbb{E}(\overline{X})] [\overline{X} - \mathbb{E}(\overline{X})]^T \right\}$ 

and similar expressions may be written for P' and V, the variances of  $X_1$  and  $Y_2$  respectively, where, as in Section 2.1. E denotes the expected or mean value. Substituting for X from Equation (2.11) we have

$$\mathbf{i} = \mathbf{E}\left[\left(\mathbf{I} - \mathbf{K}\mathbf{M}\right)\mathbf{X}_1 + \mathbf{K}\mathbf{X}_2 + \left(\mathbf{I} - \mathbf{K}\mathbf{M}\right)\mathbf{R}\left(\mathbf{X}_1\right) - \mathbf{K}\mathbf{E}\left(\mathbf{Y}_2\right)\right]\left[\left(\mathbf{I} - \mathbf{K}\mathbf{M}\right)\mathbf{X}_1 + \mathbf{K}\mathbf{Y}_2 - \left(\mathbf{I} - \mathbf{K}\mathbf{M}\right)\mathbf{E}\left(\mathbf{X}_1\right) - \mathbf{K}\mathbf{E}\left(\mathbf{Y}_2\right)\right]^{\frac{1}{2}}\right]$$

 $= (\mathbf{I} - \mathbf{K}\mathbf{H})\mathbf{E}\left[\left[\mathbf{X}_1 - \mathbf{E}(\mathbf{X}_1)\right]\left[\mathbf{X}_1 - \mathbf{E}(\mathbf{X}_1)\right]^{\mathrm{T}}\right] (\mathbf{I} - \mathbf{K}\mathbf{H})^{\mathrm{T}} + \mathbf{K}\mathbf{E}\left[\left[\mathbf{Y}_2 - \mathbf{E}(\mathbf{Y}_2)\right]\left[\mathbf{Y}_2 - \mathbf{E}(\mathbf{Y}_2)\right]^{\mathrm{T}}\right]\mathbf{K}^{\mathrm{T}},$ 

since  $X_1$  and  $Y_2$  are uncorrelated. Thus

 $\vec{P} = (I - KM)P'(I - KM)^T + KVK^T$  (2.12)

It is now required to find the value of K which minimizes  $\vec{P}$  in the sense that each diagonal element of  $\vec{P}$  shall be a minimum variance. Expanding Equation (2.12) gives

 $\mathbf{P} = (\mathbf{I} - \mathbf{K}\mathbf{M})\mathbf{P}'(\mathbf{I} - \mathbf{K}\mathbf{N})^{\mathrm{T}} + \mathbf{K}\mathbf{V}\mathbf{K}^{\mathrm{T}}$ 

- $= P' + (KM)P'(KM)^{T} (KM)P' P'(KM)^{T} + KVK^{T}$
- $= P' + K(MP'M^{T} + V)K^{T} K(MP') (MP')^{T}K^{T}$







since  $(AB)^T = B^T A^T$  and P' is a symmetrical matrix so that  $P'^T = P'$ . Now consider

- $(KS A)(KB A)^{T} = K(SS^{T})K^{T} KSA^{T} AS^{T}K^{T} + AA^{T}$ 
  - $= K(SS^{T})K^{T} K(SA^{T}) (SA^{T})^{T}K^{T} + AA^{T}$
  - $= -\mathbf{K} \left( \mathbf{M} \mathbf{P}' \mathbf{M}^{\mathrm{T}} + \mathbf{V} \right) \mathbf{K}^{\mathrm{T}} \mathbf{K} \left( \mathbf{M} \mathbf{P}' \right) \left( \mathbf{M} \mathbf{P}' \right)^{\mathrm{T}} \mathbf{K}^{\mathrm{T}} + \mathbf{A} \mathbf{A}^{\mathrm{T}} ,$

iľ

$$BS^{T} = MP'M^{T} + V$$

and

whence  $\overline{\mathbf{F}} = \mathbf{P}' + (\mathbf{K}\mathbf{S} - \mathbf{A})(\mathbf{K}\mathbf{S} - \mathbf{A})^{\mathrm{T}} - \mathbf{A}\mathbf{A}^{\mathrm{T}}$  $= \mathbf{P}' + (\mathbf{K}\mathbf{S} - \mathbf{A})(\mathbf{K}\mathbf{S} - \mathbf{A})^{\mathrm{T}} - \mathbf{P}'\mathbf{M}^{\mathrm{T}}(\mathbf{M}\mathbf{P}'\mathbf{M}^{\mathrm{T}} + \mathbf{V})^{-1}\mathbf{M}\mathbf{P}' \quad (2.13)$ 

The only term in this expression for  $\tilde{P}$  which is dependent on K is  $(KS-A)(KS-A)^{T}$ . As this is of the form  $BB^{T}$ , the diagonal elements consist of the sums of squares of the individual elements of (KS-A), and, as such, cannot have negative values. Thus each diagonal element of  $\tilde{P}$  shall be a minimum variance if K takes the value which makes (KS-A) = 0; that is, if

$$\mathbf{K} = \mathbf{P}'\mathbf{M}^{\mathrm{T}} \left[\mathbf{M}\mathbf{P}'\mathbf{M}^{\mathrm{T}} + \mathbf{V}\right]^{-1} . \tag{2.14}$$

Substituting in Equation (2.13) with (KS - A) = 0 gives

as the variance for the best estimate of X which, from Equation (2.11), becomes

$$\ddot{X} = X_1 - X(MX_1 - Y_p)$$
, (2.16)

where K takes the value given in Equation (2.14).

### 3. REPRESENTATION OF PHYSICAL PROCESSES

Classical physics tells us that, in the absence of external discurbance, the future behaviour of a system may be derived from its present state by application of the known equations of motion. It is now realised that, due to the quantum mature of fundamental physical processes, this is only effectively true in the marco case when the quantum noise can be ignored and only the statistical mean need be considered. Makere this is not the case the expected or mean state of the system may be predicted by the equations of motion, but imposed upon this will be a random contribution due to quantum events, whose probability distribution only will be known.

Similar considerations apply to any external disturbance. Some of these are predictable and can, if desired, be included as parameters within our chosen system; this will also apply to any bias which may be present in otherwise unpredictable disturbances. It is thus usen that the behaviour of any physical system may be considered to consist of one part which is precisely predictable from the known equations of motion and a second part which is random about zero mean, but whose probability distribution may be detormined. Fuch a process, in which the statistics of the future behaviour of a system are purely a function of its state at a particular time and the statistics structure of the perturbing process, is termed a Markov Process. It is clear that, by a suitable choice of the variables included, any physical process can be represented in this way.

Assuming a Markov process of this kind, its true state  $X_{k+1}$  at time  $t_{k+1}$  may be represented as

$$X_{k+1} + \Phi_k X_k + O_k U_k , \qquad (3.1)$$

where  $X_k$  was its true state at time  $t_k$ . Here  $X_k$  and  $X_{k+1}$  are  $n \times 1$  matrices representing the n parameters necessary to define the state of the system (termed the state vector).

 $\Phi_{\rm g}$  is an  $n \times n$  matrix representing the known dynamics or equations of motion of the system (termed the transition matrix).

 $U_k$  represents the random inputs (with zero means) to the system. In general, if there are l sources of random input,  $U_k$  will be an  $l \times 1$  matrix and is known as plant noise.

 $\mathbf{G}_{\mathbf{k}}$  is an  $\mathbf{n} \times l$  matrix representing the effects of the l elements of noise on the n parameters of the system.

The choice of elements to include in the state vector is, in practice, somewhat arbitrary. Although theoretically  $X_k$  should include all the systematic elements and  $U_k$  all the random elements affecting the system, this itself is a choice gapendent upon the state of the art. Many effects which were considered random or unpredictable a century ago arg how well understood processes. On the other hand some influences, although fully understood and predictable, may be so remote from the system under consideration that their effects may conveniently be included as part of the random input.

The representation of a Markov process by Equation (3.1) does not necessitate a stationary random process. The form of  $\Phi_k$  and  $G_k$  and the statistical structure of  $U_k$  (i.e. its variance  $q_k$ ) may vary from step to step. Indeed the requirement is not so much that they should be constant even within a single step, provided that the



ADAT LANDARDANCE FERT



Prene Die ein pieris Dittille

48 .

integrated effect shall be known for each individual step. In most cases, of course, the step size will be chosen to ensure that the transition matrix, the variance of the plant noise and its effect on the system may all be considered constant. Finally is should be noted that, as the step size approaches the zero limit, Equation (3.1) collapses to the more familiar differential form.

### 4. DERIVATION OF THE KALMAN FILTER

It is necessary to consider three quite distinct sets of equations in order to devise the Kalman filter. First we must represent the true state of the system in order to be able to study its statistical structure. Secondly, we must consider how we may forecast the state of the system from a previous best estimate and what variance this forecast will have. Finally we must determine a best estimate of the updated state, together with its variance, by suitably combining its forecast value with a measurement.

In this and subsequent sections true values will be represented by non-accented characters such as  $X_k$ , a foreonst value which has not been subjected to a measurement will be represented by a prime (e.g.  $X'_{k+1}$ ) and a best estimate by a hat (e.g.  $\hat{X}'_{k+1}$ ). Where a measurement is available this last will be obtained by combining the forecast with the measurement. In the absence of a measurement the forecast value will be the best estimate.

### 4.1 The True State Equations

We have already shown in Section 3 that the true state  $X_{k+1}$  at time  $t_{k+1}$  is derived from its state  $X_k$  at time  $t_k$  by the equation

$$X_{k+1} = \Phi_k X_k + G_k U_k . \tag{4.1}$$

Here  $\Phi_k$ , the transition matrix, is known, as is  $G_k$  representing the effect of a noise input on the elements of the state vector. We know that statistically  $U_k$  has a zero mean and variance  $A_k$ , but we have no knowledge of the actual noise contribution  $U_k$  to this particular step,

Similarly, when we make a measurement, the quantity measured will be some function  $M_{k+1}$  of the individual elements of the state vector, but the actual reading will also have a noise contribution  $N_{k+1}$ . Thus the actual quantity determined by the measurement will be in the form

$$(_{k+1} = M_{k+1}X_{k+1} + N_{k+1})$$
 (4.2)

 $M_{k+1}$  is, in general, an ... xn matrix which operates on the n elements of  $X_{k+1}$  to give the  $m \times 1$  matrix representing the m elements of the measurement  $Y_{k+1}$ . In most cases  $M_{k+1}$  will be a simple row matrix yielding a single parameter for measurement and frequently all elements of the row except one will be zeros, so that the effect of  $M_{k+1}$  will be to isolate a single element of  $X_{k+1}$ . N<sub>k+1</sub> is the "measurement noise" and has zero mean with variance  $R_{k+1}$ , but the actual contribution to a particular measurement is unknown.

We are then able to describe the true behaviour of the system by Equation (4.1) and of the measurement process by (4.2).

### 4.2 The Forecusting Process

If the error in  $\hat{X}_k$  (the best estimate of the true state  $X_k$ ) is  $<_k$ , then

 $\hat{\mathbf{X}}_{\mathbf{k}} = \mathbf{X}_{\mathbf{k}} + \mathbf{\varepsilon}_{\mathbf{k}}$ 

Since the plant noise  $U_k$  has zero mean, it is easily seen that the best forecast is simply

$$\begin{aligned} \mathbf{X}'_{k+1} &= \Phi_k \hat{\mathbf{X}}_k \\ &= \Phi_k \mathbf{X}_k + \Phi_k \boldsymbol{\varepsilon}_k \\ &= \mathbf{X}_{k+1} - \Phi_k \mathbf{U}_k + \Phi_k \boldsymbol{\varepsilon}_k . \end{aligned} \tag{4.3}$$

Now  $\epsilon_k$ , the error of  $\hat{X}_k$ , will have the same variance as  $\hat{X}_k$  itself (i.e.  $P_k$ ) and hence the variance of  $\Phi_k \epsilon_k$  is  $\Phi_k P_k \Phi_k^T$ . Similarly the variance of  $G \cup_k$  is  $G_k Q_k Q_k^T$ , where  $Q_k$  is the variance of  $U_k$ . Thus, since  $X_{k+1}$  is a true value, the error of  $X'_{k+1}$  is vepresented by the last two terms of Equation (4.3). Binds these two error terms (original error and error increment) are independent, the variance of  $X'_{k+1}$  is given by

$$\mathbf{P}_{k+1} = \Phi_k \mathbf{P}_k \Phi_k^T + \mathbf{0}_k \mathbf{0}_k \mathbf{0}_k^T . \tag{4.4}$$

### 4.3 The Measurement Process

The measurement process consists of a determination of  $Y_{k+1}$ , which is defined in Equation (4.2), and comparing this with a forecast  $(X'_{k+1})$  in order to obtain a best estimate  $(X'_{k+1})$ . Now using Equations (2.14), (2.16) and (2.16) the best estimate of  $X'_{k+1}$  will be

$$\hat{X}_{k+1} = X'_{k+1} - K_{k+1} \left( M_{k+1} X'_{k+1} - Y_{k+1} \right) , \qquad (4.6)$$



AND PRIMAR DALLA

there  

$$K_{k+1} = P'_{k+1}M^T_{k+1}[M_{k+1}P'_{k+1}M^T_{k+1} + R_{k+1}]^{-1} \qquad (4.6)$$
and  $R_{k+1}$  is the variance of  $Y_{k+1}$ .  
The variance of  $\hat{X}_{k+1}$  is given by  

$$P_{k+1} = P'_{k+1} - K_{k+1}M_{k+1}P'_{k+1} \qquad (4.7)$$
5. CONCLUSION  
If the true state of a Markov process and a measurement are defined by the equations  

$$X_{k+1} = \Phi_k X_k + \Phi_k U_k \qquad (5.1)$$
where  $X_k$  is the state vector at time  $t_k$   
 $\Phi_k$  is a transition matrix representing the equation of motion  
 $U_k$  is the plant noise with variance  $\Phi_k$   
 $\Phi_k$  represents the offect of the noise input on the state vector and  
 $X_{k+1} = M_{k-1}X_{k-1} + M_{k-1}$ . (5.2)

where  $Y_{k+1}$  is the quantity measured

 $\mathbf{M}_{k+1}$  represents its relationship with  $\mathbf{X}_{k+1}$ 

 $\mathbf{G}_{k+1}$  is the measurement noise with variance  $\mathbf{R}_{k+1}$  .

then the best estimate  $\hat{X}_k$  of the system and its variance  $P_k$  can be updated by the equations

$$\begin{aligned} \mathbf{X}_{k+1}' &= \Phi_k \hat{\mathbf{X}}_k & (5.3) \\ \mathbf{P}_{k+1}' &= \Phi_k \mathbf{P}_k \Phi_k^T + \Phi_k \mathbf{Q}_k \mathbf{Q}_k^T & (5.4) \\ \mathbf{K}_{k+1} &= \mathbf{P}_{k+1}' \mathbf{M}_{k+1}^T \left[ \mathbf{M}_{k+1} \mathbf{P}_{k+1}' \mathbf{M}_{k+1}^T + \mathbf{R}_{k+1} \right]^{-1} & (5.5) \\ \hat{\mathbf{X}}_{k+1} &= \mathbf{X}_{k+1}' - \mathbf{K}_{k+1} & (\mathbf{M}_{k+1} \mathbf{X}_{k+1}' - \mathbf{X}_{k+1}) & (5.6) \end{aligned}$$

and

ACCOUNTS ACCOUNTS ACCOUNTS ACCOUNTS

 $P_{k+1} = P'_{k+1} - K_{k+1}M_{k+1}P'_{k+1} .$  (5.7)

### ACKNOWLEDGEMENT

The authors wish to record their debt to Mr D.E.Williams of Mathematics Department, Royal Aircraft Establishment, for his helpful criticism of this work.



# CHAPTER 2 - FURTHER COMMENTS ON THE DERIVATION OF KALMAN FILTERS Section II - Gaussian Estimates and Kalman Filtering

by

Y. Genin\*

Applied Mathematics Group MBLE Research Laboratories

# \* 2 avenue van Becelaere, Brussels 17, Belgium

Con Tribulation

liter and the

PRECEDING PAGE BLANK

	NUTATION	
e <sub>i</sub>	generalized noise vector random variable corrupting the generalized measurement $\mathbf{z^i}$ of the state $\mathbf{x_i}$	
e¦	generalized noise vector random variable corrupting the generalized measurement $\mathbf{z}^{i-1}$ of the state $\mathbf{x}_i$	
f <sub>i</sub>	deterministic vector forcing function in discrete, linear dynamical system at $t_i$	
مj	eigenvector corresponding to sigenvalue $\lambda_{j}$ of matrix M	
tj	i <sup>th</sup> sampling time	
۲	random noise vector variable corrupting the measurement $z$ of the constant state $x$	Here Stand And Physics and a second
v <sub>k</sub>	$k^{th}$ random vector variable formed by partioning v in subvectors	
v <sub>i</sub>	white noise sequence corrupting measurement $z_i$ at $t_i$	
<b>vi</b>	generalized random vector variable formed by grouping the $v_i$ for i running from 1 up to i in one vector	
W <sub>i-1</sub>	white noise sequence affecting discrete, linear dynamical system at $t_{i-1}$	
w <sup>1-1</sup>	generalized random vector variable formed by grouping the $w_{1-1}$ for i running from 1 up to i in one vector	
X	constant state vector	
Ŷĸ	estimate of x based upon the k first measurement subvectors $z_{\underline{k}}$	
x <sub>i</sub>	state of discrete, linear dynamical system at t <sub>i</sub>	
ri	generalized state vector formed by grouping the $x_i$ for i running from 1 up to i in one vector	
\$ <u>1</u>	estimate of $x_i$ based upon $z^1$	
îí	estimate of $x_i$ based upon $x^{i-i}$ (extrapolation of $\hat{x}_{i-1}$ )	
<b>y</b> i	suxiliary vector defined by $\hat{x}_i = P_i y_i$	
<b>7</b> 1	extrapolation of $y_{i-1}$ from $t_{i-1}$ to $t_i$	
8	measurement vector of the state x	
"t	k <sup>th</sup> ssurement subvector formed by partitioning z in subvectors	
= <sub>i</sub>	measurement vector of the state $x_1$ at time $t_1$	
s <sup>1</sup>	generalized measurement vector formed by grouping the $z_i$ for $i$ running from 1 up to $i$ in one vector	
c <sub>i</sub>	covariance matrix of $\bullet_i$ , $E(\bullet_i \bullet_i^T)$	side and a second s
¢,	covariance matrix of $e_1^{i}$ , $E[(e_1^{i})(e_1^{i})^T]$	
Di	gein matrix for accumulative filter	
₽ <sup>1</sup>	auxiliary matrix defining $e^i$ as function of $w^{i-1}$ and $v^i$ , $e_i = -H^i P^i w^{i-1} + v^i$	
H	observation matrix of state x	
H.	submatrix of H defined by $\mathbf{z}_k = \mathbf{H}_k \mathbf{x}_k + \mathbf{v}_k$	ALLESSE STRATESTAN Dechter
н <sub>i</sub>	observation matrix of state $x_i$ at $t_i$	
H <sup>1</sup> .	generalized observation matrix	
K <sub>k</sub>	gain matrix for Kalman filter	
M.	non-hegative definite matrix	nervenil Dies anning Diel in
P	covariance matrix of the error $(\hat{\mathbf{x}} - \mathbf{x})$	

4

Pk	covariance matrix of the error $(\hat{x}_k - x)$
Pi	covariance matrix of the error $(\hat{x}_i - x_i)$
Pí	covariance matrix of the error $(\hat{x}'_i - x_i)$
e <u>i - i</u>	covariance matrix of $w_{i-1}$
Q <sup>i-1</sup>	Covariance matrix of w <sup>1-1</sup>
R	Govariance matrix of v
Rk	Govariance matrix of v <sub>k</sub>
R <sub>i</sub>	Covariance matrix or v <sub>i</sub>
R <sup>1</sup>	Covariance matrix of v <sup>1</sup>
W	positive definite weighting matrix
λj	j <sup>th</sup> eigenvalue of M
Φ <sub>1,1-1</sub>	state transition matrix from $t_{i-1}$ to $t_i$
φi	generalized state transition matrix
I	identity matrix
0,	zero vector
0	Rero matrix .
O,I	transpose of ()
0-1	inverse of ()
· <b>≡</b> ()	mean welue of ()
8 <sub>ki</sub>	Kronecker delta
	summation symbol for j running from 1 up to n

6.3

a sa alan gala



Marsha de

I. S. Bernt

### CHAPTER 2 - FURTHER COMMENTS ON THE DERIVATION OF KALMAN FILTERS

### SECTION II - GAUSSYAN ESTIMATES AND KALMAN FILTERING

### Y Cenin

### 1. INTROBUCTION

The mim of this contribution is twofold. First, it will give the reader, who is familiar with mean squares techniques, sufficient background to gain easy access to Kalman filter theory. To this end, the intimate connection between the Kalman and the Gaussian estimation theory will be pointed out: more precisely, it will be shown that the application of results obtained by Gauss, at least in essence, to discrete stochastic linear processes leads to the Kalman filter.

The first part recalls the main points of the Gauss theory<sup>6</sup>, as slightly extended by Plackett<sup>12</sup>. The core of this section is a restatement of two classical theorems, which are of primary importance in modern linear filtering techniques: the first constitutes the cornerstone of recursive filtering, while the second offers its dual form, i.e. accumulative filtering.

The second part is devoted to the application of these theorems to discrete stochastic processes and shows how the Kalman filter and its dual form fit naturally in the framework provided by Gauss,

### 2. THE GAUSSIAN LEAST SQUARES THEORY

### 2.1 Statement of the Problem

-Consider the fallowing equation of measurements:

where x is a N-dimensional vector of measurements, x a n-dimensional constant state vector to be determined (n < N), H a N × n matrix of maximal rank and v a N-dimensional vector of errors with zero mean and positive definite covariance matrix R.

The problem is to find an estimate  $\hat{x}$  of the unknown state x as a linear combination of the measurements x ,

**X** = Az .

so that the estimate  $\hat{\mathbf{x}}$  be unbiased and have the smallest variance for each of its elements.

### 2.2 A Deterministic Problem

Let us consider a particular problem, in which Equation (1) holds<sup>6</sup>. The z vector is known and is assumed to take the value E. Let us temporarily disregard probabilistic and statistical considerations and seek a "good" approximation X of the state x, in the form of a linear combination of the known measurements Z. If we arbitrarily decide to measure the quality of the approximation by adopting the usual mean squares criterion

$$(\Xi - H_X)^T \Psi(\pi - H_X) = minimum$$

with W a positive definite weighting matrix, the best approximation is readily found to be

 $\mathbf{x} = (\mathbf{H}^{\mathrm{T}}\mathbf{W}\mathbf{H})^{-1}\mathbf{H}^{\mathrm{T}}\mathbf{W}\mathbf{x}$ 

which uplyes this simple deterministic problem.

2.3 Theorem 1

10-10940404 - 10040404 - 12040404 - 12040404 - 12040404

Coming back to our main problem (Section 2.1), we shall now prove the following fundamental theorem. The unbiased estimate  $\hat{x}$ , which is a linear combination of the measurements z and has the smallest variance for each of its components, is given by

 $\hat{\mathbf{x}} = (\mathbf{H}^{\mathrm{T}}\mathbf{R}^{-1}\mathbf{H})^{-1}\mathbf{H}^{\mathrm{T}}\mathbf{R}^{-1}\mathbf{z} .$ 

### (5)

# PRECEDING PAGE BLANK



-----

i to odu

(1)

(2)

(3)

(4)

It has the same form as Equation (4), the solution of the deterministic problem in which the weighting matrix W is set equal to  $R^{-1}$ .

The estimate is readily proved to be unbiased, for its mean value

 $E(\hat{x}) = (H^T R^{-1} H)^{-1} H^T R^{-1} E(z)$ 

may be written in view of Equation (1),

 $E(\hat{x}) = (H^{T}R^{-1}H)^{-1}H^{T}R^{-1}[Hx + E(y)]$ 

= x.

The covariance matrix P of the estimate 2 is defined by

 $\mathbf{P} = \mathbf{E}[(\hat{\mathbf{x}} - \mathbf{x})(\hat{\mathbf{x}} - \mathbf{x})^{\mathrm{T}}]$ 

which, in this case, has the form

$$P = (H^{T}R^{-1}H)^{-1}H^{T}R^{-1}B(vv^{T})R^{-1}H(H^{T}R^{-1}H)^{-1}$$

$$= (H^{T}R^{-1}H)^{-1}$$

Before proving that each diagonal element of the P matrix is minimized by Equation (5), let us first establish the following important lemem. Lemma. Let M be ony n x n non-negative definite matrix. Then the unbiased estimate R, which is a linear combination of the measurements x and minimizes the quadratic form

$$\mathbb{E}\left[\left(\hat{\mathbf{x}}-\mathbf{x}\right)^{\mathrm{T}}\mathsf{N}(\hat{\mathbf{x}}-\mathbf{x})\right] \tag{8}$$

is again given by Equation (5).

Suppose this proposition were false and let  $\hat{\mathbf{x}} = \mathbf{E}\mathbf{x}$ , the solution with a matrix B different from  $(\mathbf{H}^T\mathbf{N}^{-1}\mathbf{H})^{-1}\mathbf{H}^T\mathbf{x}^{-1}$ . Without restriction, M may be assumed to be symmetric, for (8) is a homogeneous quadratic form. If it has multiple eigenvalues, it may be submitted to the Gauss-Schmidt orthogonalization procedure<sup>2</sup> so that, in any case, (8) may be written

$$\sum_{j=1}^{n} \lambda_j \alpha_j^{T} \mathbb{E} \left[ (\hat{\mathbf{x}} - \mathbf{x}) (\hat{\mathbf{x}} - \mathbf{x})^T \right] \alpha_j , \qquad (B)$$

with  $q_j$  the sigenvector corresponsing to the sigenvalue  $\lambda_j$  of the matrix M. Note that the sigenvalues  $\lambda_j$  are all real and non-negative, for M is a non-negative definite real symmetric matrix. On the other hand, since the estimate  $\hat{X}$  is unbiased, we have

BH = I, (10)

with I the identity matrix. The quadratic form (9) is therefore equivalent to

 $\sum_{j=1}^{n} \lambda_j q_j^{T} RRB^{T} q_j ,$ 

which may be transformed into

$$\sum_{j=1}^{B} \lambda_{j} q_{j}^{T} BRB^{T} q_{j} = \sum_{j=1}^{B} \lambda_{j} q_{j}^{T} (H^{T} R^{-1} H)^{-1} q_{j} + \sum_{j=1}^{B} \lambda_{j} q_{j}^{T} (B - (H^{T} R^{-1} H)^{-1} H^{T} R^{-1}) R[B - (H^{T} R^{-1} H)^{-1} H^{T} R^{-1}]^{T} q_{j}$$
(11)

in view of Equation (10) and the identity

$$\mathbf{B} = (\mathbf{H}^{T}\mathbf{R}^{-1}\mathbf{H})^{-1}\mathbf{H}^{T}\mathbf{R}^{-1} + [\mathbf{B} - (\mathbf{H}^{T}\mathbf{R}^{-1}\mathbf{H})^{-1}\mathbf{H}^{T}\mathbf{R}^{-1}], \qquad (12)$$

Since in the summation (11) each term is non-negative, we are left with

يبيه بيري والمرجع والم

$$\mathbf{B} = (\mathbf{H}^{\mathsf{T}} \mathbf{R}^{-1} \mathbf{H})^{-1} \mathbf{H}^{\mathsf{T}} \mathbf{R}^{-1}$$
(13)

material and the state of the provided of the second

which proves the lemma.

A direct consequence of the lemma is that the estimate Equation (5) has the smallest variance for each of its elements, as can be seen by choosing all elements of M equal to zero, except for a unit element anywhere on the diagonal.

# **PROTING**

(6)

(7)

### 3.4 Theorem 2 (Recursive Estimation)

Suppose now a partition of the measurements vector  $\mathbf{z}$  in two subvectors  $\mathbf{z}_{k-1}$ ,  $\mathbf{z}_k$ , Partitioning v and H accordingly, Equation (1) may be written

$$\begin{array}{c} \mathbf{m}_{k-1} & \approx & \mathbf{H}_{k-1}\mathbf{x} + \mathbf{v}_{k-1} \\ \mathbf{x}_{k} & = & \mathbf{H}_{k}\mathbf{x} + \mathbf{v}_{k} \end{array}$$
 (14)

and we assume the set  $\mathbf{v}_{k-1}$  to be uncorrelated with the set  $\mathbf{v}_k$  , so that R has the form

$$\mathbf{R} = \begin{bmatrix} \mathbf{R}_{\mathbf{k}-1} & \mathbf{0} \\ \mathbf{0} & \mathbf{R}_{\mathbf{k}} \end{bmatrix}$$
(15)

Let  $\hat{x}_{k-1}$  and  $P_{k-1}$  be the minimum variance unbiased estimate and its associated covariance matrix, defined on the subset  $x_{k-1}$  only. Then the minimum variance unbiased estimate  $\hat{x}_k$  defined on the set of measurements  $[x_{k-1}, x_k]$  may be obtained without reprocessing the subset  $x_{k-1}$  and is given by

$$\hat{x}_{k} = \hat{x}_{k-1} + K_{k} (x_{k} - H_{k} \hat{x}_{k-1}),$$
 (16)

with K<sub>k</sub> a gain matrix defined by

$$K_{k} = P_{k-1}H_{k}^{T}[H_{k}P_{k-1}H_{k}^{T} + R_{k}]^{-1}, \qquad (17)$$

while the covariance matrix  $\mathbf{P}_{\mathbf{k}}$  is obtained by .

$$\mathbf{P}_{\mathbf{k}} = (\mathbf{I} - \mathbf{K}_{\mathbf{k}} \mathbf{H}_{\mathbf{k}}) \mathbf{P}_{\mathbf{k}-1}, \qquad (18)$$

To demonstrate the above relations, note that, in view of Equation (7), the inverse covariance matrix is

$$P_{k}^{-1} = H^{T}R^{-1}H$$

$$= H_{k-1}^{T}R_{k-1}^{-1}H_{k-1} + H_{k}^{T}R_{k}^{-1}H_{k}$$

$$= P_{k-1}^{-1} + H_{k}^{T}R_{k}^{-1}H_{k}$$
(19)

This expression has a classical form in matrix algebra, known as the Frobenius form, so that the inverse may be readily obtained:

$$P_{k} = P_{k-1} - P_{k-1} H_{k}^{T} [H_{k} P_{k-1} H_{k}^{T} + R_{k}]^{-1} H_{k} P_{k-1}$$

$$= (I - K_{k} H_{k}) P_{k-1} , \qquad (20)$$

On the other hand, in view of Equation (5), the new estimate  $\theta_k$  has the form

$$\hat{x}_{k} = P_{k}H^{T}R^{-1}x$$

$$= P_{k}[H_{k-1}^{T}R_{k-1}^{-1}x_{k-1} + H_{k}^{T}R_{k}^{-1}x_{k}] \qquad (21)$$

Combination of Equations (20) and (21) yields the result

$$\hat{\mathbf{x}}_{\mathbf{k}} = (\mathbf{I} - \mathbf{x}_{\mathbf{k}} \mathbf{H}_{\mathbf{k}}) \mathbf{P}_{\mathbf{k}-1} [\mathbf{H}_{\mathbf{k}-1}^{T} \mathbf{R}_{\mathbf{k}-1}^{-1} \mathbf{P}_{\mathbf{k}-1} + i\mathbf{H}_{\mathbf{k}}^{T} \mathbf{R}_{\mathbf{k}}^{-1} \mathbf{z}_{\mathbf{k}}]$$

$$= (\mathbf{I} - \mathbf{x}_{\mathbf{k}} \mathbf{H}_{\mathbf{k}}) \hat{\mathbf{x}}_{\mathbf{k}-1} + (\mathbf{I} - \mathbf{K}_{\mathbf{k}} \mathbf{H}_{\mathbf{k}}) \mathbf{P}_{\mathbf{k}-1} \mathbf{H}_{\mathbf{k}}^{T} \mathbf{R}_{\mathbf{k}}^{-1} \mathbf{z}_{\mathbf{k}}$$

$$= (\mathbf{I} - \mathbf{K}_{\mathbf{k}} \mathbf{H}_{\mathbf{k}}) \hat{\mathbf{x}}_{\mathbf{k}-1} + \mathbf{P}_{\mathbf{k}-1} \mathbf{H}_{\mathbf{k}}^{T} \mathbf{R}_{\mathbf{k}}^{-1} \mathbf{z}_{\mathbf{k}} -$$

$$- \mathbf{P}_{\mathbf{k}-1} \mathbf{H}_{\mathbf{k}}^{T} [\mathbf{H}_{\mathbf{k}} \mathbf{P}_{\mathbf{k}-1} \mathbf{H}_{\mathbf{k}}^{T} + \mathbf{R}_{\mathbf{k}}]^{-1} [\mathbf{H}_{\mathbf{k}} \mathbf{P}_{\mathbf{k}-1} \mathbf{H}_{\mathbf{k}}^{T} + \mathbf{R}_{\mathbf{k}}] \mathbf{R}_{\mathbf{k}}^{-1} \mathbf{z}_{\mathbf{k}} +$$

$$+ \mathbf{P}_{\mathbf{k}-1} \mathbf{H}_{\mathbf{k}}^{T} [\mathbf{H}_{\mathbf{k}} \mathbf{P}_{\mathbf{k}-1} \mathbf{H}_{\mathbf{k}}^{T} + \mathbf{R}_{\mathbf{k}}]^{-1} \mathbf{z}_{\mathbf{k}}$$

$$= \hat{\mathbf{x}}_{\mathbf{k}-1} + \mathbf{K}_{\mathbf{k}} (\mathbf{z}_{\mathbf{k}} - \mathbf{H}_{\mathbf{k}} \hat{\mathbf{x}}_{\mathbf{k}-1}) , \qquad (22)$$

which completes the proof of the theorem.

The generalization to an arbitrary number of subsets of measurements  $z_k$  uncorrelated with each other is obvious.



### 2.5 Theorem 3 (Accumulative Estimation)

With the same definitions as above for  $z_{k-1}$ ,  $z_{k-1}$ ,  $R_k$ ,  $R_{k-1}$ ,  $R_k$ ,  $\hat{x}_{k-1}$  and  $P_{k-1}$ , let us now denote by  $\hat{x}_k$  the minimum variance unbiased estimate defined on the subset  $z_k$  only, and let  $P_k$  be the corresponding covariance matrix; then, the minimum variance unbiased estimate  $\hat{x}$  defined on the two subsets  $[z_{k-1}, z_k]$  is directly obtained by

$$\hat{\mathbf{x}} = \mathbf{p} \left[ \mathbf{p}_{1}^{-1}, \hat{\mathbf{x}}_{k-1} + \mathbf{p}_{k}^{-1} \hat{\mathbf{x}}_{k} \right] , \qquad (23)$$

with a covariance matrix P computed from the relation

$$P^{-1} = P_{k-1}^{-1} + P_k^{-1} . \tag{24}$$

The proof of this theorem is a direct consequence of Equations (5) and (7); in fact, we have

 $f = PH^TR^{-1}s$ 

$$P^{-1} = H^{T}R^{-1}H$$

$$= H^{T}_{k-1}R^{-1}_{k-1}H_{k-1} + H^{T}_{k}R^{-1}_{k}H_{k}$$

$$= P^{-1}_{k-1} + P^{-1}_{k}$$
(25)

and

$$= P[H_{k-1}^{T} R_{k-1}^{-1} z_{k-1} + H_{k}^{T} R_{k}^{-1} z_{k}]$$
  
=  $P[P_{k-1}^{T} R_{k-1} + P_{k}^{-1} R_{k}]$  (26)

The generalization to an arbitrary number of subsets of measurements uncorrelated with each other is again obvious.

### 2.6 Observability

From Equation (7) the existence condition for a minimum variance unbiased estimate is readily deduced: the unknown state vector is observable if, and only if, the matrix H has maximal rank.

### 8.7 Comments

Theorem 2 and its dual form (Theorem 3) have received numerous applications in the field of satellite orbit determination, especially Reference 4; the filtering procedures of Swerling<sup>14</sup>, Claus<sup>5</sup> and Battin<sup>1, 2</sup> are direct consequences of them. Theorem 2 is of primary importance in modern filtering theory, for it constitutes the corneratone of the Kalman recursive filter.

It can be proved<sup>11</sup> that, for normal processes. linear least squares estimation yields the same solution as non-linear least squares estimation. Moreover, in that case, the least squares estimate turns out to be identical to the maximum likelihood estimate, which requires to maximize the conditional probability density of measuring s, assuming x.

The minimum variance unbiased estimate is often called Gaussian estimate. Markov estimate, for obvious reasons in view of Theorem 2 or weighten least squares estimate, underlining the occurrence of the covariance matrix  $\mathbb{R}^{n_1}$  in the quadratic form (9).

Let us finally exphasize that the minimum variance unbiased estimate requires no a priori information concerning the error distribution function.

### 3. THE KALWAN-BUCY FILTERING THEORY

### 3.1 The Kalman Booursive Filter

It will be shown in this section that the Kalman filter, almost simultaneously derived by Kalman and Bucy<sup>2</sup>, Battin<sup>1</sup> and Bwerling<sup>14</sup> is a direct consequence of Theorems 1 and 2, when applied to first-order discrete linear systems.

Consider an n-dimensional non-constant state vector, taking the value  $x_i$  at time  $t_i$  and obeying the following first-order discrete linear equation (i = 1, ..., N):

 $x_{i} = \Phi_{i,i-1}x_{i-1} + w_{i-1}$ 

(27)



Sund Characterit Califor

5A

(31)

(32)

where  $w_{i-1}$  is a white noise random vector sequence with zero bean and positive definite covariance matrix  $Q_{i-1}$ 

$$\mathbf{E}(\mathbf{w}_{i-1}) = 0$$
$$\mathbf{E}(\mathbf{w}_{i-1}\mathbf{w}_{i-1}^T) = \mathbf{Q}_{i-1}$$
$$\mathbf{E}(\mathbf{w}_{i}\mathbf{w}_{i}^T) = \delta_{ki}\mathbf{Q}_{i}$$

while  $\Phi_{i,i-i}$  is the system state transition matrix from time  $t_{i-1}$  to time  $t_i$ . At each time  $t_i$ , the state vector  $x_i$  is observed through the measurement equation

 $\mathbf{z}_{i} = \mathbf{H}_{i}\mathbf{x}_{i} + \mathbf{v}_{i} , \qquad (28)$ 

with the same definitions for  $x_i$ ,  $H_i$  and  $v_i$  as in Theorem 2. It is further assumed that the two white noise random sequences  $v_i$  and  $w_{i-1}$  are uncorrelated.

Then, the problem may be formulated as follows. Find from the measurements  $(x_1, x_{i-1}, \ldots, x_i)$  the minimum variance unblased estimate  $\hat{x}_i$  (covariance matrix  $P_i$ ) of the state  $x_i$ , depending linearly on the measurements, assuming the minimum variance unblased estimate  $\hat{x}_{i-1}$  (covariance matrix  $P_{i-1}$ ) to be known from the measurements  $(x_{i-1}, x_{i-2}, \ldots, x_i)$ .

In order to apply the theorems of the preceeding sections to the problem at hand, let us introduce the vectors  $x^i$ ,  $x^i$ ,  $w^{i+i}$ ,  $v^i$  defined by the following recurrence relations:

x <sup>i</sup>	7	<b>a</b> j <b>a</b> 1-1 <b>a</b> 1-1	g <sup>1</sup>	2	[.*] [*]-]
"i-1	2		v <sup>i</sup>	=	$\begin{bmatrix} \mathbf{v}_1 \\ \mathbf{v}^1 & \mathbf{i} \end{bmatrix}  .$

If we similarly define the matrices  $\ensuremath{\,Q^{i\,\sim\,i}}$  ,  $\ensuremath{R^{i}}$  ,  $\ensuremath{H^{i}}$  ,  $\ensuremath{\Phi^{i}}$  ,  $\ensuremath{P^{i}}$  ,

$$g^{L-1} = g^{L} (w^{L-1}) (w^{L-1})^{T}$$

$$R^{1} = g^{L} (v^{L}) (v^{L})^{T}$$

$$H^{1} = \begin{bmatrix} H_{1} & 0 \\ 0 & H^{L-1} \end{bmatrix}$$

$$\phi^{L} = \begin{bmatrix} \frac{1}{\phi^{L-1}\phi^{-1}} & 0 \\ 0 & H^{L-1} \end{bmatrix}$$

$$p^{L} = \begin{bmatrix} 0 & 1 \\ \phi^{L-1}\phi^{-1} & 0 \\ 0 & H^{L-1} \end{bmatrix}$$

the whole set of Equations (27) and (28) for i running from 1 up to i, may then be globally written

$$\mathbf{x}^{1} = \mathbf{H}^{1} \Phi^{1} \mathbf{x}_{1} - \mathbf{H}^{1} \mathbf{F}^{1} \mathbf{x}^{1-1} + \mathbf{v}^{1}, \qquad (29)$$

or equivalently

San Share States

$$a^{1} = H^{1}\Phi^{1}\pi_{i} + e_{i} , \qquad (30)$$

with wi the random vector variable

with mero mean and Jovariance matrix  $C_i$  :

$$\mathbf{C}_{i} = \mathbf{H}^{i} \mathbf{P}^{i} \mathbf{Q}^{i-1} (\mathbf{H}^{i} \mathbf{P}^{i})^{T} + \mathbf{R}^{i} .$$

\* The matrix  $\Phi_{i, 1-1}$  is assumed to have an inverse  $\Phi_{i, 1-1}^{-1}$  .

Note that Equation (30) has the form of Equation (1) so that, by Theorem 1, the estimate  $\hat{x}_{i-1}$  may be written

$$\hat{\mathbf{x}}_{i+1} = \mathbf{p}_{i+1} (\mathbf{R}^{1-1} \boldsymbol{\Phi}^{i+1})^{\mathrm{T}} \mathbf{C}_{i+1}^{-1} \mathbf{x}^{i+1}$$
(33)

$$\mathbf{P}_{1-1} = [(\mathbf{H}^{1-1}\Phi^{1-1})^{\mathsf{T}}\mathbf{C}_{1-1}^{-1}(\mathbf{H}^{1-1}\Phi^{1-1})]^{-1}, \qquad (34)$$

in view of Equations (5) and (7).

Before calculating  $\hat{x}_i$ , let us first compute the minimum variance unbiased estimate  $\hat{x}'_i$  (covariance matrix  $P'_i$ ) of the state  $x_i$ , based upon the measurements  $(x_{i-1}, x_{i-2}, \dots, x_i)$  only.

Introducing (27) in the generalized measurement equation

$$\mathbf{z}^{1-1} = \mathbf{H}^{1-1} \mathbf{x}_{1-1} + \mathbf{e}_{1-1}$$
, (35)

one obtains

$$\mathbf{z}^{i-1} = \mathbf{H}^{i-1} \Phi^{i-1} \Phi^{-1}_{i,1-1} \mathbf{x}_{i} + \mathbf{e}_{i}^{\prime} , \qquad (30)$$

where of is a new random vector variable:

$$\mathbf{e}'_{1} = \mathbf{e}_{1-1} - \mathbf{H}^{1-1} \mathbf{\hat{e}}_{1, 1-1}^{-1} \mathbf{\hat{e}}_{1, 1-1}^{-1} \mathbf{w}_{1-1} , \qquad (37)$$

with zero mean and covariance matrix  $O_i^{\prime}$  :

$$C_{i}^{i} = C_{i-1} + H^{i-1} \Phi^{i-1} \Phi^{-1}_{i,i-1} Q_{i-1} (1)^{i-1} \Phi^{i-1}_{i,i-1})^{T}$$

As a Probenius form is recognized in the above equation,  $(O_1')^{-1}$  is known, i.e.

$$(\mathbf{0}_{i}^{\prime})^{-1} = \mathbf{C}_{i-1}^{-1} - \mathbf{C}_{i-1}^{-1} \mathbf{H}^{i-1} \Phi_{i,1-1}^{-1} \left( (\Phi_{i,1-1}^{-1})^{T} \mathbf{P}_{i-1}^{-1} \Phi_{i,1-1}^{-1} + \mathbf{Q}_{i-1}^{-1} \right)^{-1} (\mathbf{H}^{i-1} \Phi_{i,1-1}^{-1})^{T} \mathbf{C}_{i-1}^{-1}$$
(38)

so that the covariance matrix  $P_1^{\prime}$  , which may be written, in view of Theorem 1.

$$(\mathbf{P}'_{i})^{-1} \simeq (\mathbf{H}^{i-1}\phi^{i-1}\Phi^{-1}_{i,i-1})^{\mathsf{T}}(\mathbf{C}'_{i})^{-1}(\mathbf{H}^{i-1}\phi^{i-1}\Phi^{i-1}_{i,i-1}) , \qquad (39)$$

reduces to

2012

$$(\mathbf{P}_{i}^{i})^{-1} = (\Phi_{i,i-1}^{-1})^{T} \mathbf{P}_{i-1}^{-1} \Phi_{i,i-1}^{-1} \cdots (\Phi_{i,i-1}^{-1})^{T} \mathbf{P}_{i-1}^{-1} \Phi_{i,i-1}^{-1}] (\Phi_{i,i-1}^{-1})^{T} \mathbf{P}_{i-1}^{-1} \Phi_{i,i-1}^{-1} + \mathbf{u}_{i-1}^{-1}]^{-1} (\Phi_{i,i-1}^{-1})^{T} \mathbf{P}_{i-1}^{-1} \Phi_{i,i-1}^{-1}$$

$$= [\Phi_{i,i-1}^{-1} \mathbf{P}_{i-1}^{-1} \Phi_{i,i-1}^{T} + \mathbf{Q}_{i-1}]^{-1} ,$$

$$(40)$$

as can be easily verified, using Equations (34) and (38). Thus,  $P_1'$  is equal to

$$P'_{i} = \Phi_{i,i-i} P_{i-i} \Phi_{i,i-i} + Q_{i-i}$$
 (41)

A similar manipulation of  $\hat{x}'_1$ , which by Theorem 1 is defined to be  $\hat{x}'_1 = P'_1(H^{1-1}\Phi^{1-1}\Phi^{-1}_{1,1-1})^T(C'_1)^{-1}g^{1-1}$ . (42)

leads via Frustions (33), (34), (38) and (41) to the result

$$\hat{\pi}'_{i} = \Phi_{i,i-1} \hat{\pi}_{i-1} , \qquad (43)$$

We are now in position to simply derive the Kalman filter equations, for all the conditions are satisfied to apply Theorem 2 with

> $P_{k} = R_{1}$   $P_{k-1} = P_{1}^{\prime}$   $\hat{S}_{k-1} = \hat{S}_{1}^{\prime}$   $H_{k} = H_{1}$  $R_{k} = R_{1}$ ,





**}** 

Training in continue (Cardina

the second s

. 61

so that the estimate  $\hat{x}_i$  is immediately given by Equations (18), (17) and (18):

$$\hat{x}_1 = \hat{x}_1 + K_1(\hat{x}_1 - \hat{H}_1\hat{x}_1)$$
 (44)

$$K_{i} = P(H_{i}^{T}(H, P'_{i})H_{i}^{T} + R_{i})^{-1}$$
(45)

$$P_1 = (I - K_1 H_1) P_1'$$
 (46)

Equations (41), (43), (44), (45) and (46) constitute the Kalman filter. The physical meaning of these equations is evident: Equations (41) and (43) extrapolate the estimate  $\hat{x}_{i-1}$  from time  $t_{i-1}$  to time  $t_i$  through the state transition equation (27) while Equations (44), (45) and (46) update this extrapolated estimate on account of the new available measurements  $x_i$ .

The dynamical system model (27) may be readily extended to include deterministic forcing terms  $f_{i-1}$ :

$$\mathbf{x}_{i} = \Phi_{i,i-1} \mathbf{x}_{i-1} + f_{i-1} + \mathbf{w}_{i+1} . \tag{47}$$

This plearly affects the extrapolated estimate  $\hat{x}_1'$  only and Equation (43) must be replaced by

$$\hat{x}_{i} = \Phi_{i,i-1} \hat{x}_{i-1} + f_{i-1} , \qquad (48)$$

the other filtering equations remaining unchanged.

The Kalman filter is cupable of other important extensions: although out of the scope of the present contribution let us quote the simultaneous estimation of imprecisely known parameters, entering the dynamical system linearly<sup>13</sup>, the replacement of whits noise sequences for  $w_{i_{1}-1}$ ,  $v_{i_{1}}$  by wide-sense Markov sequences<sup>13</sup> or by sequences correlated with respect to each other<sup>10</sup>, stochastic optimization, and so on ...

### 3.2 The Accumulative Filter

It has been shown in the preceding section, that the Kalman filter is a direct application of Theorem 2 to firstorder discrete linear systems; it is therefore natural to ask whether Theorem 3 may be applied to the same problem<sup>7,8</sup>.

Consider Equations (27) and (28) of Section 3.1 with the same definitions and assumptions and let us introduce a n-dimensional vector  $y_1$  such that at any bise  $t_1$  the following relation holds:

$$\hat{\mathbf{x}}_{i} = \mathbf{P}_{i} \mathbf{y}_{i} , \qquad (49)$$

It is easily verified from Theorems 1 and 2 that such a vector may always be defined. Furthermore, suppose that the vector  $\mathbf{y}_{i-1}$  and the inverse covariance matrix  $\mathbf{P}_{i-1}^{-1}$  are available at time  $\mathbf{t}_{i-1}$  and let  $\mathbf{S}_i$  be a matrix defined by

$$\mathbf{f}_{i} = (\Phi_{i,i-1}^{-1})^{T} \mathsf{P}_{i-1}^{-1} \Phi_{i,i-1}^{-1} , \qquad (50)$$

Equations (40) and (41) may then be written

$$P_1' = B_1^{-1} + Q_{1-1}$$
(51)

$$(P_i^{\prime})^{-1} = D_i S_i$$
, (52)

with  $D_i$  a matrix given by the relation

$$D_{i} = I - S_{i} [S_{i} + Q_{i-1}^{-1}]^{-1}$$
(58)

Similarly, the extrapolated  $y_{i-1}$  vector, i.e.  $y'_i$  , can be written

$$\mathbf{y}_{i}^{*} = \mathbf{D}_{i} (\Phi_{i, i-1}^{*})^{T} \mathbf{y}_{i-1} , \qquad (54)$$

in view of Equation (43).

Use of Theorem 3 yields the new vector  $y_i$  at time  $t_{i,i}$ 

$$y_i = y_i^{\prime} + H_i^{\prime} R_i^{-1} z_i$$
, (55)

and the inverse estimate covariance matrix

$$P_{1}^{-1} = (P_{1}^{+1} + H_{1}^{T}R_{1}^{+1}H_{1}^{-1}$$

- (3m - 192 - 194 - 1944)

(56)

und Tri duirenti Tuller

The estimate  $\hat{x}_i$  is then available via

 $\hat{\mathbf{x}}_i = \mathbf{P}_i \mathbf{y}_i$ .

Equations (50), (52), (53), (54), (55) and (56) constitute the accumulative filter equations for the problem at hand. It is important, to note that the calculation of the estimate  $\hat{X}_1$  is made independently of the accumulative filter: it means that any error in the inversion of  $\mathbb{P}_1^{-1}$  does not affect the succeeding estimates. For the general case, however, the accumulative filter is not a practical tool to solve the problem, in view of the n + n matrix inversion required in Equation (53). The attuation is completely different in the absence of plant noise; in such cases ( $w_{i-1} = 0$  for any i), the  $M_1$  matrix reduces to the identity matrix so that the accumulative filter equations simply become

$$\theta_{i} = (\Phi_{i,i-1}^{*})^{T} \Psi_{i-1}^{*} \Phi_{i,i-1}$$
(50)

$$(\mathbf{P}_i')^{-1} = \mathbf{s}_i \tag{52}'$$

$$\mathbf{M}_{\mathbf{i}} = \mathbf{I} \tag{53}'$$

$$\mathbf{A}_{l}^{T} = (\mathbf{A}_{l}^{T})^{T} \cdot \mathbf{1}_{l}^{T} \cdot \mathbf{A}_{l}^{T-1}$$
(94).

$$\mathbf{x}^{\mathbf{i}} = \mathbf{x}^{\mathbf{i}} + \mathbf{H}^{\mathbf{i}}\mathbf{H}^{\mathbf{i}}_{\mathbf{i}}, \mathbf{z}^{\mathbf{i}}$$
(22)

$$(\mathbf{P}_{i})^{-1} = (\mathbf{P}_{i})^{-1} + \mathbf{H}_{i}^{T}\mathbf{R}_{i}^{-1}\mathbf{H}_{i}$$
 (56)

Any numerical matrix inversion has completely disappeared in the scheme, which is not the case in the Kalman filter: the computation procedure is made particularly stable and this may be of primary importance when wenkly observable systems are dealt with.

### 3.3 Observability Condition

In view of Theorem 1 and Equation (29), an observability condition is satily obtained: the state  $x_i$  is observable if, and only if, the matrix  $(H^i \Phi^i)^{\top} U_i^{-1} (H^i \Phi^i)$  has an inverse or, more compactly, if and only if the matrix  $H^i \Phi^i$  has a generalized inverse  $(C_i$  is positive definite).

### 4. CONCLUSION

The modern redursive filtering theory has been shown to promeeu from an idea originally due to Gauss, at least in essence. More precisely, simple theorems arising from the minimum variance estimation of a constant state, linearly depending on the measurements, have been applied to discrete first-order stochastic linear systems, by the way of a restatement of the problem. As a result, the Kaiman filter and its dual form have been obtained by simple algebraic manipulations.



(57)

the state of the second

and Barris and a

### REFERENCES

1. Battin, R.H.

A Statistical Optimizing Navigation Procedure for Space Flight. American Rocket Bodiety Journal, Vol.32, 1962, pp.1681-1682.

2. Battin, R.H. Astronautical Guidance. McGraw-Hill, New York, 1937.

3. Bellman, B. Introduction to Matrix Analysis. McGraw-Hill, New York, 1960.

4. Blackmann, R.E. Methods of Orbit Refinement. Bell System Technical Journal, Vol.43, 1965, pp. 885-909.

- B. Claus, A.J. Orbit Determination for Communication Satellites from Angular Data Only. American Rocket Society 17th Annual Meeting, Los Angelos, 1962.
- 6. GRUBB, K.F. Theory of the Motion of the Henvenly Bodies Moving About the Sun in Conic Sections. Dover, New York, 1903.

7. Genin, Y. A Note on Linear Minimum Variance Fytimation Problems. Institute of Electrical and Electronic Engineers, Transactions on Automatic Control, Vol.AO-13, 1968, p.103.

 Genin, Y. Gaussian Estimates and Modern Linear Discrete Filtering Methods. Revue A. Vol.13, 1968, pp. 70-78.

- 9. Kalman, R.E. New Results in Linear Filtering and Prediction Theory. Journal of Basic Engineering, Bucy, R.S. Vol.83D, 1961, pp.95-108.
- 10. Kalman, R.E. Fundamental Study of Adaptive Control Systems. Technical Report ABD-TR 61, Vol.1, (NASA N 62-15355).

11. Papoulis, A. Probability, Hundon Variables and Stochastic Processes. McGraw-Hill, New York, 1965.

12. Romenbroek, H.H. On the Connection Between Discrete Linear Filters and Same Formulae of Gauss. Reprints of AFRA Congress, Paper 1, Paris, May 4-7, 1965.

Press, 1983.

13, Sorenson, H.W.

14. Swerling, P.

CONTRACTOR OF CONTRACTOR

A Proposed Stagewise Differential Correction Provedure for Satellite Tracking and Prediction. Rand Corporation, Paper P-1292, 1958.

Kalman Filtering Techniques. Advances in Control Systems, Vol.3, Chapter 5, Academic













1130 Bister Stan Bich

### CHAPTER 3 - COMPUTATIONAL TECHNIQUES IN KALMAN FILTERING

155 Dec -----

C MANNEAR

NE . The State

0

### by

Stauley F.Sohmidt

Analytical Mechanics Associates, Inc. 260 Eheridan Avenue Palo Alto, California, USA

# PRECEDING PAGE BLANK

### CHAPTER 3 - COMPUTATIONAL TECHNIQUES IN KALMAN FILTERING

### Stanley F. Schmidt

### 1. INTRODUCTION

In the application of Kalman filtering theory, the mathematical formulation of the problem and computational techniques utilized may depend hearily on the computational capabilities available. This interface is particularly strong in real time, acrospace applications wherein considerations of weight, power, and reliability generally require the selection of techniques which minimize onboard computational requirements.

The discussion of this chapter will emphasize computing techniques for Kalman filtering in these onboard type problems. The techniques are applicable but not necessarily essential when large general purpose computing facilities are available.

A note in possing is that the Kalmun filter is a special data processing technique. It was originally introduced<sup>1, 2</sup> for the onboard application because of its reduction in computational requirements over other methods available at the time.

Numerous investigators<sup>3-4</sup> have reported various problems with and "fixes" for Kalman filters. The problems in general fall into the following categories:

- (a) Loss of positive definiteness in the covariance matrix resulting from numerical errors.
- (b) Improper mathematical model, leading to a divergence of the estimate from measurements.
- (c) Nonlinear phenomena generally aggravated by a poor selection of the starting estimate.

of these problems, (a) has probably been experienced to some degree by almost everyone working in the field. Reasonably efficient (from computational considerations) solutions for this problem based on square root algorithms will be given in this chapter.

For Item (b), techniques exist for compensating for errors in the mathematical model. The tachniques, although useful, still leave a good deal of "cut and try" in finding suitable solutions. Some relevant material on this subject will be covered in this chapter.

Item (c) will not be covered here. The mechanization equations described use linearization about the current best estimate of state. This factor and the occasional requirement of specially designed starting calculations (based on the raw data) have removed nonlinear problems in the practical applications with which the author is familiar.

### 2. MATHEMATICAL FORMULATION AND DEFINITIONS

One of the first problems which must be addressed in applying Kalman filtering theory is the mathematical formulation of the problem. This section will define the overall filtering problem from a practical viewpoint.

### 2.1 Problem Statement and Discussion

(i) A true set of dynamical equations describing the dynamic behavior of the system is believed to be represented by a set of vector differential equations

Ŷ

=	F(X,C,U,t) .	(2.1	.)

One therefore develops a set of equations for the system of the form

### $\dot{X} = F(X, C, U, t) ,$

### where

ないないない。「ないない」のないない。「ないないない」のないない。「ないないないない」ではないないないない。「ないないないない」のないないない。「ないないないない」のないないない。「ないないない」のないない

Ň

X = & vector describing the fundamental state variables

C = a vector describing constants used in the system equations

### PRECEDING PAGE BLANK

The term "fundamental state variables" is used for those time-varying quantities whose initial conditions must be specified in solving (2.2) (e.g., the position and velocity vectors of an orbiting spucecraft).



67









(2.2)

- " U = a vector of forcing functions acting on the dynamic system
  - t = the independent variable, time.

(ii) A vector. Y, of measurements is available which is believed related to the fundamental state variables in the form

 $\overline{\mathbf{Y}} = \mathbf{G}(\mathbf{\hat{\lambda}}, \overline{\mathbf{\hat{V}}}, \mathbf{T}) + \mathbf{\ddot{G}}(\mathbf{t})$ (2.3)

A model of these measurements is therefore developed of the form

Y = O(X, V, t) + q(t) ,(2.4)

whore

- V = a vector describing assumed constants and/or time-varying states in the measurement model
- q = the assumed random error in the measurement.

Before proceeding further in the definitions some discussion of the notation and meaning of the phrases "believed to he" and "assumed to be" is in order.

"Believed to be" is defined as meaning: in conformance with such scientific knowledge as is possessed at the present time.

"Assumed to be" is used in the context of known approximation, such as truncations of infinite series representations, and so on.

(iii) An initial estimate of the state vector at a time to may be given, or is assumed, as

~		•	
X(t <sub>o</sub> )	= )	۲.	•

(iv) Other initial conditions and time-varying forcing functions which are assumed are\*

ĉ(to)	Ŧ	ĉ	(2.6)
Ŷ(t <sub>a</sub> )	=	Ŷ。	(2.7)

(2.5)

$$\hat{\mathbf{U}}_{n}(\mathbf{t}) = \hat{\mathbf{U}}_{n} . \tag{2.8}$$

(v) The errors in the previously defined quantities are defined by

$$\mathbf{E}(\mathbf{X}_0 - \mathbf{X}_0) = \mathbf{E}(\mathbf{X}_0) = 0 \tag{2.9}$$

 $\mathbf{E}(\mathbf{\tilde{x}}_{n}\mathbf{\tilde{x}}_{n}^{T}) = \text{covariance matrix} = \mathbf{P}_{n}$ (2.10)

£[q(t)] = 0 (2.11)

$$\mathbf{E}(\mathbf{q}\mathbf{q}^{\mathsf{T}}) = \mathbf{Q}(\mathbf{t}) \tag{2.12}$$

 $\mathbf{E}(\mathbf{C}-\hat{\mathbf{C}}_{n}) = \mathbf{\Sigma}(\mathbf{\overline{0}}_{n}) = \mathbf{0}$ (2.13)

 $\mathbf{E}(\tilde{\mathbf{c}}_{0}\tilde{\mathbf{c}}_{0}^{\dagger}) = \mathbf{P}_{\mathbf{c}_{0}}$ (2.14)

 $\mathbf{E}[\mathbf{U}(\mathbf{t}) - \hat{\mathbf{U}}_{0}(\mathbf{t})] = \mathbf{E}[\hat{\mathbf{U}}(\mathbf{t})] = \mathbf{0}$ (2.15)

$$\mathbf{E}(\mathbf{\tilde{u}})^{\mathrm{T}} = \mathbf{P}_{\mathbf{u}_{\mathrm{n}}}(\mathbf{t}) \tag{2.16}$$

 $\mathbf{E}(\mathbf{V}-\hat{\mathbf{V}})_{0} = \mathbf{E}(\overline{\mathbf{V}}) = 0$ (2.17)

 $\mathbf{E}(\mathbf{\tilde{v}}\mathbf{\tilde{v}}^{\mathrm{T}}) = \mathbf{P}_{\mathbf{v}_{\mathrm{fl}}}$ (2, 18)

The use of the expected value operator E in Equations (2.9) to (2.18) requires some discussion. Nominal usage of the operator implies an ensemble average. For example, if the errors in  $\hat{X}$  are considered as injection errors of a space vehicle, the ensemble represents an infinite number of launch vehicles of the same type launched from the same location. If the basic causes of injection errors are random in this ensemble and it is meaningful for the problem, then (2.9) is a true statement.

In general one cannot define a meaningful ensemble for quantities such as errors in the estimate of the vectors like C and V. Hence one really should consider the initial values as estimates and the covariance matrix as a confidence level. This is, in fact; how one uses the quantities and the problem of defining the ensemble is avoided. Equations (2.3), (2.7) and probably (2.15) are not true statements in the strict statistical sense.

As a result of limited knowledge and known approximation in computations, one may desire to consider a portion of U(t) to a random forcing function. This will be discussed later in this chapter as a possible means of dynamic model compensation. 2.5 Problem Objectives

The objective of filtering (or data processing) considered herein is to find:

Algorithms for use in a given computer which process the measurement data and provide an estimate of the state variables which is sufficiently accurate for real time command and control purposes.

Additional factors which must be considered in this problem include:

- (a) Computer speed.
- (b) Computer memory availability.
- (c) Computational errors.
- (d) Measuremont data availability.
- (e) Development time schedule.

This problem statement contains some of the practical considerations which arise in most applications of Kalman filtering. Theory is used to provide approaches for finding a practical solution. By no means will this solution be optimal in the theoretical sense. This is because exact mathematical models of all quantities, as well as practical performance indices which consider all the factors, cannot be defined.

Subsequent discussion will assume that preliminary considerations of computer speed and memory have led to the requirement of data processing algorithms of a "one pass" nature. That is, once measurements at any time point have been processed they must be discarded. This assumption is made to restrict the scope of subsequent discussion to modified forms of the Kalman filter.

A practical solution of the filtering problem in any specific example is found by the use of existing theory and practical experience to define starting algorithms. Simulation is then used for further validation and/or modifications. Final validation and/or modifications then takes place in tests of the real system.

Before introducing the basic algorithms, some additional definitions are required. If the gradient of (2.2) is taken with respect to X. C., and U., a set of time-varying linear differential equations is obtained\*:

 $\dot{x} = P(t)x(t) + B(t)c + D(t)u(t)$ , (2.19)

HAAAR BE

1 🗩 🕂 🕅 🖓 🕐

144 th

where

 $B(t) = \nabla_{\mathbf{C}} \mathbb{P}(\mathbf{X}, \mathbf{C}, \mathbf{U}, t) | \sigma(t) = \hat{\mathbf{c}}(t)$ 

 $\mathbf{F}(t) = \nabla_{\mathbf{X}} \mathbf{F}(\mathbf{X}, \mathbf{C}, \mathbf{U}, t) |_{\mathbf{X}(t)} = \hat{\mathbf{X}}(t)$ 

 $D(t) = \nabla_{U} F(X, C, U, t) |_{U(t)} = \hat{U}(t)$ 

Since (2.19) is linear, the general solution may be written in the form

$$\mathbf{x}(t) = \Phi(t;t_0)\mathbf{x}(t_0) + \Phi_{c}(t;t_0)a + \int_{t_0}^{t} \Phi(t;\tau)D(\tau)u(\tau) d\tau , \qquad (2.20)$$

The transition matrix  $\Phi(t;t_{a})$  and the sensitivity to constant forces or control can be found by solving

$$\dot{\Phi} = F(t)\Phi$$
,  $\dot{\Phi}(t_0; t_0) = I$  (2.21)

$$\Phi_{c} = F(t)\Phi_{c} + B(t)(I), \quad \Phi_{c}(t_{0};t_{0}) = 0.$$
 (2.22)

The integral term of (2.20) may be solved by approximating u(t) as a constant for small time increments,  $\Delta$ :

 $\hat{\Phi}_{u} = \mathbf{F}(\mathbf{t})\hat{\Phi}_{u} + \mathbf{D}(\mathbf{t})\mathbf{I}, \quad \hat{\Phi}_{u}(\mathbf{t}_{a} + \mathbf{n}\Delta; \mathbf{t}_{a} + \mathbf{n}\Delta) = \mathbf{0}. \quad (2.23)$ 

It is usually convenient to consider an augmented state vector of the form

# $\hat{Z} = \begin{pmatrix} \hat{X} \\ \hat{O} \\ \hat{U}_{q} \end{pmatrix} - \text{fundamental state variables}$ $\hat{z} = \begin{pmatrix} \hat{X} \\ \hat{O} \\ \hat{U}_{q} \end{pmatrix} - \text{constant terms in equations of motion}$

In this instance the sugmented state deviation obeys

$$\mathbf{s}(t) = \Phi_{\mathbf{x}}(t;t_{u})\mathbf{s}(t_{u}) + \int_{t_{u}}^{t} \Phi_{\mathbf{x}}(t;\tau)\mathbf{D}(\tau)\mathbf{u}_{u}(\tau) d\tau \qquad (2.24)$$

if linear assumptions are valid.

• Lower case letters are used to denote amail deviations from the corresponding upper case value. That is, a vector Z is non-ministered as  $Z = \hat{Z} + z$ .

The gradient of the measurement equation (2, 4), is taken with respect to X and V to find linear equations for the ith measurement\*

 $y_1 = h_1 x + g_1 v + q_1$ (2.25)

where

70

$$\mathbf{h}_{i} = \nabla_{\mathbf{X}} \mathbf{G}(\mathbf{X}, \mathbf{V}, \mathbf{t}_{i}) |_{\mathbf{X}(\mathbf{t})} = \hat{\mathbf{X}}(\mathbf{t})$$
(2.26)

$$\mathbf{s}_{i} = \nabla_{\mathbf{y}} \mathbf{G}(\mathbf{X}, \mathbf{V}, \mathbf{t}_{i}) |_{\mathbf{V} = \mathbf{0}} . \tag{2.27}$$

One may desire to relate the vector of measurements to the vector,  $z_0$ , at a fixed epoch. This may be done by using (2.24)<sup>†</sup> and (2.25) to obtain

$$y = Hz_a + Gv + q$$
. (2.28)

In (2.28) the vector y may consist of measurements at different time points. The vector  $z_0$  can be at a fixed epoch. This fact can be used to obtain an algorithm for processing small batches of data or for a simplified form of data compression<sup>10</sup>. It is generally desirable to consider a further augmentation of the state to include the measurement parameters, V . Then (2,28) would be written in the equivalent form

$$\mathbf{y} = (\mathbf{H} \mathbf{G}) \begin{pmatrix} \mathbf{z}_0 \\ \mathbf{y} \end{pmatrix} + \mathbf{q} \stackrel{\wedge}{=} \mathbf{H} \mathbf{z}_0 + \mathbf{q} .$$
 (2.29)

In the case where it is meaningful or necessary, one may write the equation for updating the covariance matrix P\_(t) in time

$$P_{g}(t) = \Phi_{g}(t;t_{0})P_{g}(t_{0}) \Phi_{g}^{T}(t;t_{0}) + R , \qquad (2.30)$$

The matrix R in (2.30) gives the added uncertainty caused by random forcing functions in the time interval  $t = t_n$ .

### 2.3 Kalman Pilter Equations

A modified form of the Kalman filter for nonlinear systems is given here for processing the measurements in a sequential manner. These algorithms are readily derivable from Kalman's original equations for the discrete filter". The algorithm is conveniently stated in two parts:

(1) Between the 1 and 1+1 measurement times,

$$\hat{Z}(t_{i+1}) = \hat{Z}(t_i) + \int_{t_i}^{t_{i+1}} \hat{Z}(\tau) d\tau \qquad (2.31)$$

$$\dot{\Phi}_{\mathbf{g}} = \left( \nabla \hat{\mathbf{z}} \Big|_{\mathbf{z} = \hat{\mathbf{z}}} \right) \Phi_{\mathbf{z}}, \quad \Phi_{\mathbf{z}}(\mathbf{t}_{1}; \mathbf{t}_{1}) = \mathbf{I}.$$
(2.32)

(ii) At measurement times t<sub>i+1</sub>,

$$P_{\mathbf{x}}(t_{i+1}) = \Phi_{\mathbf{x}} P_{\mathbf{x}}(t_i) \Phi_{\mathbf{x}}^{T} + \mathbf{R} \stackrel{\Delta}{=} \mathbf{P}_{\mathbf{h}}$$
(2.33)

$$\hat{z}_{h} = \hat{z}_{h} + K(\hat{x} - \hat{\hat{x}})$$
(2.34)

$$P_{n} = P_{n} - KHP_{n} \qquad (2.35)$$

$$\mathbf{K} = \mathbf{P}_{\mathbf{k}}\mathbf{H}^{\mathbf{T}}(\mathbf{H}\mathbf{P}_{\mathbf{k}}\mathbf{H}^{\mathbf{T}} + \mathbf{Q})^{-1}$$
(2.36)

 $\hat{\mathbf{Y}} = \mathbf{G}(\hat{\mathbf{X}}, \hat{\mathbf{V}}, \mathbf{t}_{i+1}) \approx \text{computed measurement.}$ 

Equation (2.31) is a simple statement of integrating the system dynamic equations (2.2). The Z notation is used to imply a time update of all those components of the state which are assumed non-constant. The transition matrix can be calculated by integrating the variational equations implied in (2.32) with the appropriate initial conditions. At the measurement time  $t_{i+1}$ , the covariance matrix  $P_2(t_{i+1})$  is calculated by (2.33). Equations (2.34), (2.35), and (2.36) are then used to calculate the new estimate of state,  $2_a$ , and the new covariance matrix  $P_a$  . These new values result from including the measurements Y at time  $t_{1+1}$  in both the est 4 .4 and the covariance matrix.

Note that if the influence of random forcing functions is approximated by the use of (2.23), then (2.33) should be updated at  $\triangle$  time intervals between the measurements.

\* The random error in measurement is assumed to be small and therefore considered to give a small ; andom deviation from a noiswless measurement.

The effects of the random variation of u(t) are assumed negligible in writing this expression. Otherwise, a time correlated error must be added to equation (2.28).



11 Derennyth




### 3. SQUARE ROOT FORMULATIONS

As already mentioned, numerical errors frequently cause a problem when the covariance matrix equations (2.33) and (2.35) are used. The usual symptoms of the problem are negative diagonal terms occuring in the covariance matrix after processing data. The problem can occur after the first measurement in some instances  $^{10}$ . It is more of a "nuisance" than of a serious type, as one usually can find a remedy by adding or increasing R of (2.33). This is one of the techniques discussed in Section 4 for model error compensation. The other technique given in Section 4 also can be used as a remedy for this numerical problem.

This numerical difficulty can also be removed by square root methods. The square root methods also provide a significant improvement in numerical accuracy of the covariance matrix. Since the covariance matrix is the weighting factor containing the effects of all past measurements, improving the accuracy of this matrix can be important in some problems.

The square root of P is defined as W and satisfies the relationship

$$WW^{T} = P = \sum_{i=1}^{D} W_{i} W_{i}^{T},$$
 (3.1)

where

 $w_i = i^{th}$  column vector of the matrix W

n = the number of columns of W, which will be defined as the dimension of P.

With this definition, if P is positive definite, the column vectors  $w_i$  are linearly independent. It is generally desirable to force the column vectors of W to be linearly independent. Then when P is positive semi-definite some column vector or vectors will be null. The rank of P is equal to the number of non-zero column vectors of W in this formulation.

In the case of the square root matrix one needs computation algorithms for

- (a) starting the problem (e.g., given  $P_0$  find  $W_0$ ),
- (b) propagating # in time (square root of (2.33)),

(c) modifying W for including measurements (square root of (2.35)).

Subsequent material will summarize algorithms for these purposes.

### 3.1 Initialization of the Square Root Matrix

One obvious way of specifying the initial square root matrix  $W_0$  is to define the column vectors of  $W_0$ . One should note that when the columns of  $W_0$  are made linearly independent, each column specifies an independent error source. All error sources lie in different directions in the state space.

An alternate problem is that  $P_0$  is given and a suitable square root matrix  $W_0$  is to be defined. For this usage and from the preceding discussion it is seen that the desired algorithm will construct a set of column vectors of the matrix  $W_0$  which are linearly independent. Consider the application of the following equation for  $B_1 = P_0$ :

$$B_{k+1} = B_k - \frac{B_k V_k V_k^{TB} B_k}{V_k^{TB} V_k} .$$
 (3.2)

In (3.2)  $v_k$  is an arbitrary (non-zero) column vector. Note that the quantity  $v_k^T B_{k+1} v_k$  is zero. In other words, (3.2) removes all errors of  $P_0$  in the direction of the vector  $v_k$ . If  $v_k$  is chosen such that its  $k^{th}$  element is unity and the remaining elements are zero, the  $k^{th}$  row and column from  $B_k$  are removed. By letting  $k = 1, \ldots, n$ , then  $B_{n+1} = 0$ . The quantity n is the dimension of  $P_0$  (the number of state variables).

Therefore, by defining the column vectors of W, by

$$\mathbf{w}_{k} = \begin{cases} \frac{\mathbf{B}_{k}\mathbf{v}_{k}}{\sqrt{(\mathbf{v}_{k}^{T}\mathbf{B}_{k}\mathbf{v}_{k})}} & \text{if } \mathbf{v}_{k}^{T}\mathbf{B}_{k}\mathbf{v}_{k} \neq 0\\ 0 & \text{if } \mathbf{v}_{k}^{T}\mathbf{B}_{k}\mathbf{v}_{k} = 0 \end{cases} \text{ for } \mathbf{k} = 1, \dots, n, \qquad (3.3)$$

the resulting  $W_{\alpha}$  matrix will be of lower triangular form. From filtering considerations one can view the algorithm as the equivalent of making in perfect observations of the state vector which reduce the error in all components to zero ( $B_{n+1}=0$ ). The error vector removed after each step of the algorithm is stored so that the matrix  $P_{\alpha}$  can be recovered from  $W_{\alpha}$ : This algorithm is given in a slightly different form in Reference 11.

71

Relation of Fill

2019-2222

The algorithm provides a means of calculating a lower triangular matrix. T such that

where, in (3.4),

### B = a positive semi-definite matrix.

Another use of the algorithm is in finding a sequence of observations whose random errors are uncorrelated from a sequence whose errors are correlated. For example, if

> $\begin{pmatrix} y_1 \\ y_2 \end{pmatrix} = \begin{pmatrix} h_1 \\ h_2 \end{pmatrix} x + \begin{pmatrix} q_1 \\ q_2 \end{pmatrix}$ (3.5)

and

$$\mathbf{E}\begin{pmatrix}\mathbf{q}_1\\\mathbf{q}_2\end{pmatrix}(\mathbf{q}_1\mathbf{q}_2) = \begin{pmatrix}\sigma_1^2 & \rho\sigma_1\sigma_2\\\rho\sigma_1\sigma_2 & \sigma_2^2\end{pmatrix} = \mathbf{Q}, \qquad (3.6)$$

then the observations given by

$$\begin{pmatrix} \mathbf{x}_1 \\ \mathbf{x}_2 \end{pmatrix} = \begin{bmatrix} \mathbf{y}_1 \\ \mathbf{a}_1 \mathbf{y}_1 + \mathbf{y}_2 \end{bmatrix} = \begin{bmatrix} \mathbf{h}_1 \\ \mathbf{a}_1 \mathbf{h}_1 + \mathbf{h}_2 \end{bmatrix} \mathbf{x} + \begin{bmatrix} \mathbf{q}_1 \\ \mathbf{a}_1 \mathbf{q}_1 + \mathbf{q}_2 \end{bmatrix}$$
(3.7)

have uncorrelated random errors if

$$\mathbf{a}_1 = -\rho \frac{\sigma_1}{\sigma_1} \,. \tag{3.8}$$

In this instance

$$\mathbf{T} = \begin{pmatrix} \mathbf{i} & \mathbf{0} \\ \rho \sigma_{\mathbf{g}} & \mathbf{i} \end{pmatrix}$$
(3.9)

$$\boldsymbol{\omega} \mathbf{r}^{\mathsf{T}} = \begin{bmatrix} \sigma_{1}^{\mathsf{g}} & \mathbf{0} \\ \mathbf{0} & (\mathbf{1} - \rho^{\mathsf{g}}) \sigma_{\mathsf{g}}^{\mathsf{g}} \end{bmatrix}, \qquad (3.10)$$

Since

the measurement  $\mathbf{x}_i$  is the same as  $\mathbf{y}_i$  and its random error is  $\sigma_i^2$ . The measurement  $\mathbf{x}_p$  is a linear combination of  $\mathbf{y}_1$  and  $\mathbf{y}_2$  so chosen that its random error is uncorrelated with the random error of  $\mathbf{x}_1$ . The variance of the random error in  $\mathbf{x}_2$  approaches zero as the correlation factor  $\rho$  approaches unity.

### 3.2 Measurement Update of the Square Root Matrix

For a single (scalar) measurement, (2.35) may also be written in square root form as\*

W.W. = P.

$$\Psi_{\mathbf{a}} = \Psi_{\mathbf{b}} \left[ \mathbf{I} - \frac{\Psi_{\mathbf{b}}^{2} H^{T} H \Psi_{\mathbf{b}}}{\mathbf{a}} \right], \qquad (3.11)$$

where

and

$$\mathbf{a} = (\mathbf{H}\mathbf{W}_{\mathbf{b}}\mathbf{W}_{\mathbf{b}}^{\mathsf{T}}\mathbf{H}^{\mathsf{T}} + \mathbf{Q})\left(1 + \sqrt{\frac{\mathbf{Q}}{\mathbf{H}\mathbf{W}_{\mathbf{b}}\mathbf{W}_{\mathbf{b}}^{\mathsf{T}}\mathbf{H}^{\mathsf{T}}} + \mathbf{Q}\right)$$

Multiple measurements (at a given time point) may be treated as sequential scalar measurements if their random errors are uncorrelated. The previous material illustrated how an uncorrelated sequence may always be defined; hence the square root formulation of (3.11) is general. Other formulations for multiple measurements are also available<sup>12</sup>.

\* This algorithm is given in Reference 12 and will be referred to as Potter's method.

### 3.3 Time Update of the Square Root Matrix

Two methods for updating the square root matrix in time for random forcing functions are given. Both assume that the matrix R of Equation (2.33) is available in factored form such that

$$R = \sum_{i=1}^{4} s_i s_i^{T} = 55^{T} , \qquad (3.12)$$

where

s, are linearly independent vectors

j = number of random forcing functions.

3.3.1 Method No.1

A square root mutrix time undate algorithm can be developed by noting that

$$P^{-1}(t) = \left[ \Phi P \Phi^{T} + \sum_{i=1}^{j} s_{i} s_{i}^{T} \right]^{-1}$$
 (3.13)

As a result of the well-known matrix indentity

$$[B+tt^{T}]^{-1} = B^{-1} - B^{-1}t(t^{T}B^{-1}t+1)^{-1}t^{T}B^{-1}, \qquad (3.14)$$

and Potter's square root method for equations like the right-hand side of (3.14), one method for obtaining the time update of the square root matrix is readily developed. The steps are as follows:

- (i) Define CC =  $[\Phi W(t_n)]^{-1}$ .
- (ii) Operate j times on the matrix C , using Equation (3.11) with  $H = s_1$ , i = 1, j and Q = 1. Let the result be  $C_j$ .

(iii) Then

 $W(t) = 0_1^{-1}$ .

The author is not awars of any extensive use of this technique, as a result of the obvious drawback of requiring two matrix inversions for every time update. Reference 14 refers to this technique as Potter's method.

3.3.2 Method No.2

An alternate technique which does not require inversion can be developed in a relatively simple manner from the material presented thus far.

Examination of Equation (3.2) illustrates that it may be factored as follows

$$A_{k+1}A_{k+1}^{T} = M_{k}A_{k}A_{k}^{T}M_{k}^{T} = B_{k+1}, \qquad (3.15)$$

where

$$\mathbf{M}_{\mathbf{k}} = \begin{bmatrix} \mathbf{I} - \frac{\mathbf{A}_{\mathbf{k}} (\mathbf{A}_{\mathbf{k}}^{\mathsf{T}} \mathbf{v}_{\mathbf{k}} \mathbf{v}_{\mathbf{k}}^{\mathsf{T}})}{\mathbf{v}_{\mathbf{k}}^{\mathsf{T}} \mathbf{A}_{\mathbf{k}} \mathbf{A}_{\mathbf{k}}^{\mathsf{T}} \mathbf{v}_{\mathbf{k}}} \end{bmatrix} .$$
(3.16)

Hence, if the matrix A is reduced using

$$M_{k+1} = M_k A_k$$
,  $k = 1, ..., n$ , (3.17)

and the W matrix of Equation (3.3) is stored, a lower triangular matrix in W will result. Furthermore, this lower triangular matrix has the property that

$$\mathbf{W}^{\mathrm{T}} = \mathbf{A}_{1} \mathbf{A}_{1}^{\mathrm{T}} \,. \tag{3.18}$$

It should therefore be obvious that, if

the resulting square root matrix W is the desired square root matrix which includes the effects of the random forcing functions. This method was developed by the author<sup>18</sup> but appears to be equivalent in practics to the technique outlined in Reference 14.

The special choice of vectors (the k<sup>th</sup> element unity and all other elements zero) makes the matrix  $M_k$  very simple to calculate. Furthermore, in (3.17) each step of the algorithm makes the k<sup>th</sup> row of the matrix  $A_{k+1}$  zero. As a result the number of operations is not as large at it may appear at first glance.

73





D Mar Han Old

To illustrate the simplicity of the algorithm, a simple example will be carried through each step. Let

$$\Phi W = \begin{pmatrix} 1 & 0 & 0 \\ 1 & 1 & 0 \\ 1 & 1 & 1 \end{pmatrix}, \quad S \neq I . \tag{3.20}$$

Then

$$A_{1} = \begin{pmatrix} 1 & 0 & 0 & 1 & 0 & 0 \\ 1 & 1 & 0 & 0 & 1 & 0 \\ 1 & 1 & 1 & 0 & 0 & 1 \end{pmatrix} \text{ and } V_{1} = \begin{pmatrix} 1 \\ 0 \\ 0 \end{pmatrix}$$

$$\mathbf{A}_{\mathbf{i}}^{\mathsf{T}}\mathbf{v}_{\mathbf{i}} \begin{pmatrix} \mathbf{i} \\ \mathbf{0} \\ \mathbf{0} \\ \mathbf{1} \\ \mathbf{0} \\ \mathbf{0} \end{pmatrix}.$$

$$\begin{aligned} \mathbf{x}_{1}^{\mathbf{u}} &= \begin{pmatrix} 2\\1\\1 \end{pmatrix} = A_{1}A_{1}^{\mathsf{T}}\mathbf{v}_{1} \\ \mathbf{w}_{1} &= \begin{bmatrix} \mathbf{I} - \begin{pmatrix} 2\\1\\1 \end{pmatrix} \frac{(1 \ 0 \ 0)}{2} \end{bmatrix} = \begin{bmatrix} 0 & 0 & 0\\-1/2 & 1 & 0\\-1/2 & 0 & 1 \end{bmatrix} \\ \mathbf{x}_{2} &= \begin{bmatrix} 0 & 0 & 0\\-1/2 & 1 & 0\\-1/2 & 0 & 1 \end{bmatrix} \begin{bmatrix} 1 & 0 & 0 & 1 & 0 & 0\\1 & 1 & 0 & 0 & 1 & 0\\1 & 1 & 1 & 0 & 0 & 1 \end{bmatrix} \\ &= \begin{bmatrix} 0 & 0 & 0 & 0 & 0 & 0\\1/2 & 1 & 0 & -1/2 & 1 & 0\\1/2 & 1 & 1 & -1/2 & 0 & 1 \end{bmatrix} \\ \mathbf{w}_{1} &= \begin{pmatrix} 2/2\\1/2 \end{pmatrix} = \text{first column of } \Psi(t) \end{aligned}$$

$$\mathbf{w}_{2}^{-\mathbf{B}} = \begin{pmatrix} 0 & 0 & 0 & 0 & 0 & 0 \\ 1/2 & 1 & 0 & -1/2 & 1 & 0 \\ 1/2 & 1 & 1 & -1/2 & 0 & 1 \end{pmatrix} \begin{bmatrix} 1/\frac{2}{1} \\ 0 \\ -1/2 \\ 1 \\ 0 \end{bmatrix} = \begin{pmatrix} 0 \\ 2.5 \\ 1.5 \end{pmatrix}$$
$$\mathbf{M}_{2} = \begin{bmatrix} 0 & 0 & 0 \\ 1 - 2.5 & \frac{(0 & 1 & 0)}{2.5} \\ 1.5 & 2.5 \end{bmatrix} = \begin{bmatrix} 1 & 0 & 0 \\ 0 & 0 & 0 \\ 0 & -\frac{1.5}{2.5} & 1 \\ \frac{-1.5}{2.5} & 1 \end{bmatrix}$$
$$\mathbf{A}_{3} = \begin{bmatrix} 1 & 0 & 5 \\ 0 & 0 & 0 \\ 0 & -0.5 & 1 \\ 0 & -0.5 & 1 \end{bmatrix} \begin{bmatrix} 0 & 0 & 0 & 0 & 0 \\ 0.5 & 1 & 0 & -0.5 & 1 & 0 \\ 0.5 & 1 & 1 & -0.5 & 0 & 1 \end{bmatrix}$$

(3. 21)



Define



The final answer is

TAX (ATOM) STRATES, STRATES, STRATES, STRATES, I

$$W(t) = \begin{pmatrix} 2/\sqrt{2} & 0 & 0 \\ 1/\sqrt{2} & \sqrt{2.5} & 0 \\ 1/\sqrt{2} & 1.5/\sqrt{2.5} & \sqrt{2.6} \end{pmatrix} .$$
(3.22)

As a check one may compute  $A_1 A_1^T$  from Equation (3.21) and compare the result to  $W(t)W^T(t)$  of Equation (3.22).

$$A_{1}A_{1}^{T} = \begin{bmatrix} I & 0 & 0 \\ 1 & 1 & 0 \\ 2 & 1 & 1 \end{bmatrix} \begin{bmatrix} I & 1 & 1 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{bmatrix} + \begin{bmatrix} I & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{bmatrix}$$
$$= \begin{bmatrix} 2 & 1 & 1 \\ 1 & 3 & 2 \\ 1 & 2 & 4 \end{bmatrix}$$
$$(t) W^{T}(t) = \begin{bmatrix} 2/\sqrt{2} & 0 & 0 \\ 1/\sqrt{2} & \sqrt{2.5} & 0 \\ 1/\sqrt{2} & 1.5/\sqrt{2.5} & \sqrt{2.6} \end{bmatrix} \begin{bmatrix} 2/\sqrt{2} & 1/\sqrt{2} & 1/\sqrt{2} \\ 0 & \sqrt{2.5} & 1.5/\sqrt{2.5} \\ 0 & 0 & \sqrt{2.6} \end{bmatrix}$$
$$= \begin{bmatrix} 2 & 1 & 1 \\ 1 & 3 & 2 \\ 1 & 2 & 4 \end{bmatrix}, \text{ which checks.}$$

As noted from the example, the following two facts about the method are significant:

(1) The dimension of  $M_k$  and  $A_k$  reduces (effectively) by one at each step.

(ii) The final solution for W(t) is a lower triangular matrix, that is, the elements above the diagonal of W(t) are zero. This fact can be used to reduce the number of calculations for the measurement update for the first measurement processed after the time update algorithm.

Both these facts result from the specific choice of the  $v_k$  vectors.

3.4 Recommended Algorithms for Square Root Implementations

Time Update Equations to a Measurement

Let  $A_1 = (\overline{Q} \overline{n} + S)$ .



AND THE PARTY OF THE PARTY

in the state of th



(3.23)

Calculate the column vectors of W(t) in accordance with the n-scep algorithm k = 1, ..., n where  $v_k(l) = 0$ .  $l \neq K$ , and  $v_k(k) = 1$ .

$$\mathbf{u} = \mathbf{A}_{\mathbf{k}} \mathbf{A}_{\mathbf{k}}^{\mathsf{T}} \mathbf{v}_{\mathbf{k}}$$
(3.24)

$$\mathbf{M}_{\mathbf{k}} = \mathbf{1} - \frac{\mathbf{\overline{w}}_{\mathbf{k}} \mathbf{v}_{\mathbf{k}}^{\mathsf{T}}}{\mathbf{\overline{w}}_{\mathbf{k}}(\mathbf{k})}$$
(3.25)

$$A_{k+1} = M_k A_k \tag{3.26}$$

$$w_{k} = W_{k} / \langle W_{k}(k) \rangle = k^{th} \text{ column of } W(t) . \qquad (3.27)$$

Measurement Update (for a Single Neasurement)

Compute the vectors

$$\mathbf{a} = (\mathbf{v}^{\mathrm{T}}\mathbf{v} + \mathbf{Q})\left(1 + \sqrt{\left[\frac{\mathbf{Q}}{\mathbf{v}^{\mathrm{T}}\mathbf{v} + \mathbf{Q}}\right]}\right). \tag{3.30}$$

Modify W in accordance with

### 3.5 Comparative Computation Requirements

Although the proposed method should enhance the accuracy it may be costly in machine time. The proposed method is compared here with the more conventional method using the covariance matrix (Equations (2.33), (2.35) and (2.36).

The comparison data given assume that none of the matrices or vectors have zeros in prescribed locations. Also P is assumed an n x n matrix and S an H x n matrix.

### Number of Operations for a Time Update

(i) Conventional Method\*†



(ii) Square Root Method

	M & D	A & S	Square Root
Equation (3.23)	n³	n³	
Equation (3.24)	$\sum_{i=1}^{n} (n+m)(n+1-i)$	$\sum_{i=1}^{n} (n+i)(n+1-i)$	· 0
Equation (3.25)	$\sum_{i=1}^{n} (n-i)$	0	·0
Equation (3.26)	$\sum_{i=1}^{n-1} (n+m)(n-1)^2$	$\sum_{i=1}^{n-1} (n+n) (n-1)^2$	0
Equation (3.27)	$\sum_{n=1}^{n} (n-1)$	0	Û

• M & D means multiplications and/or divisions.

\* A & # means additions and/or subtractions.







تفلاف تقاشعهم والأفاق أأشعل

As an example shaume that  $n \approx 10$  and m = 1. The use of the tabulated equations for number of operations gives the following:

	Equivalent M & D	M & D	A & S	Square Roots
Conventional Method	2550	2100	2200	0
Square Root	5837	4830	4785	10

### The column imbeled "Equivalent M & D" assumes

(i) Square roots take 5 times as long as multiplications.

(ii) Additions take 1/b of the time of multiplications.

For this example, the square root method takes twice as much computer time as the conventional method. If the advantage of symmetry of the covariance matrix for the conventional method was forced, then it would be about 3 times as fast as the modified square root method.

Number of Operations for a Measurement Update (Scalar Observation)

(1) Conventional Method

200022777 P

	M&D	Ads
Equations (2.35) and (2.36)	3n <sup>z</sup> + n	$2n^2 + n + 1$

(11) Square Root Nethod

	If Triangul	Not Triangular		
	M & D	A&S	MED	A&S
Equation (3.28)	$\sum_{i=1}^n n+1=i$	$\sum_{i=1}^{n} n+1=i$	n²	<b>n</b> <sup>‡</sup>
Equation (3.29)	$\sum_{i=1}^n n+1=1$	$\sum_{i=1}^n n+1=1$	n <sup>e</sup>	n <sup>g</sup>
Equation (3.30)	n + 2	2	n + 2	2
Equation (3.31)	n# + n	n <sup>2</sup>	<b>n</b> <sup>2</sup> + n	n²

One square root is required in Equation (3.30).

The number of operations in the various columns indicate whether or not W is triangular after a time update. This advantage can only be used for the first weasurement processed at a given time point.

For comparative data assume that n = 10 and a single measurement is processed.

	Equivalent M & D	M&D	AÆS	Square Root
Conventional	352	310	211	0
Square root	387	322	302	1
Square root (triangular)	287	232	212	1

The "Equivalent N & D" column is computed using the wame relationships as stated for the previous case. The square root method has a slight sdwantage for the first measurement processed if W is triangular. However, this advantage is lost for subsequent measurements.



n-sugar Din rus minis Dahiti

If one takes advantage of symmetry for the conventional method and other obvious improvoments in techniques (for example forming the vector PHT/(HPHT+Q) instead of dividing n<sup>2</sup> times), then the conventional method will be about twice as fast for the example cited.

### 3.5.1 Influence of Coding Techniques

Coding can significantly alter the provious computational time estimates. For example, if the square root method were coded as

> $W_{n} = W \left[ I - \frac{W^{T}H^{T}HW}{n} \right]$ (3.32)

and the conventional method as

#### $P_{n} = [I - KH] [P] [I - KH]^{T} + KQK^{T}$ (3.33)

where

 $\mathbf{K} = \mathbf{P}\mathbf{H}^{\mathrm{T}}(\mathbf{H}\mathbf{P}\mathbf{H}^{\mathrm{T}} + \mathbf{Q})^{-1} ,$ 

then the comparison is as follows for a single weasurement. No advantage of W being triangular is used.

	M & D	A&S	Square Root
Conventional	2m <sup>3</sup> + 3n <sup>2</sup> + 3n	$2n^3 + 2n_1^2 + 2n + 1$	0
(n = 10)	2330	2221	0
Square root	$n^3 + 2n^3 + m + 1$	$n^3 + n^2 + 2n + 1$	1
(n = 10)	1211	1122	1

### Several conclusions are evident:

(i) One should not code the square root algorithm using Equation (3.32) if speed of computations is of any importance. There are no obvious advantages of (3.32) for accuracy either, so it is definitely not recommended. The previously outlined procedure, where no matrix multiplications are used, is decidedly superior.

(ii) Equation (3.33) is extremely poor from speed considerations for the conventional method. It requires at least 10 times as long as (2.35) for the example of  $n \approx 10$ . This speed difference will get even worse with large values of n . Equation (3.33) has the advantage that the covariance matrix P should remain positive definite. However, there are numerous other ways of doing this, so that the use of (3.33) is not recommended.

(iii) Proper coding for both techniques should slways make the conventional method superior with respect to computation time.

#### 3.6 Formulation Using the Information Matrix

One may readily obtain a set of filtering equations where the inverse of the covariance matrix (the information matrix) is used. The equations are as follows.

3.6.1 Time Update

Let	p <sup>-1</sup> ≙ ∧ ;	(3.34)
then	$p^{-1}(t) = (\Phi \Phi^T + R)^{-1} = \Lambda(t)$	(3.35)
defining	$\Phi^{-iT}p^{-i}\Phi = R_{i} = \Phi^{-iT} \Lambda \Phi^{-i}$	(3.36)

defining

and

$$= \sum_{i=1}^{J} s_i s_{\bar{i}}^{-1} . \qquad (3.37)$$

(3.38)

#### $B_{k+1} = B_k - B_k s_i (s_1^2 B_k s_i + 1)^{-1} s_1^2 B_k$ , k = 1, j Then the j-step equation,

. R

#### $\mathbf{P}^{-1}(\mathbf{t}) = \mathbf{B}_{\mathbf{t}+1} = \Lambda(\mathbf{t}),$ (3.39) yields the result that

Note that the adjoint equations

$$[\dot{\Phi}^{-1}]^{T} = -\dot{\mu}^{T}(t) \{\dot{\Phi}^{-1}\}^{T} \dot{\Phi}^{-1}^{T}(t_{o}; t_{o}) = I, \qquad (3.40)$$

may be used in place of the transition matrix equations (2.32). This avoids any requirement for matrix inversion. It should therefore be obvious that, if the square root matrix C is defined such that

 $co^{T} = \Lambda$ (3.41)

A B PASSING AN

then the square root matrix C may be updated in time using Potter's factor station of (3.38). This method was given before as part of Method No. 1 for the time update of the square root of the covariance matrix. The quantity was inverted before, using Potter's algorithm. Here, no inversions are required, however, since the information matrix formulation is used.

### 3.6.2 Measurement Update

In this instance the scuare root of an eduction of the form

 $\Lambda_{\perp} = \Lambda + H^{T}Q^{-1}H$ 

is to be calculated. The measurements are defined by

and

and the second sec

y = Hz + q E(qq<sup>†</sup>) = Q.

Quite obviously this is the identical problem to the time update of the square root of the covariance matrix; hunce, if one defines

$$A_{1} = \left[C^{\frac{1}{2}} H^{T} \sqrt{(q^{-1})}\right], \qquad (3.43)$$

and uses equations (3.24) to (3.27), the columns of the new lower triangular matrix C are given by the we of Equation (3.27).

#### 3.6.3 State Change at Measurements

It is well known that the gain K of Equation (2.34) may also be written in the form

The C obtained after including measurements is such that

$$[CO^{T}] = P_{-}^{-1},$$
 (3.45)

if C has no mero column vectors. The matrix C will be of full rank if the problem is initialized with a posibive definite covariance matrix. This initialization is not necessary in the inverse formulation, so the gain R of Equation (3,44) should be written as

. .

$$\mathbf{K} = [\mathbf{C}\mathbf{0}^{\mathsf{T}}]^{\mathsf{H}}\mathbf{I}^{\mathsf{Q}^{-1}}, \qquad (3.46)$$

In (3,46) the (#) symbol reads the pseudo-inverse. The fact that an a priori information matrix is not required with this formulation can be an advantage in some problems.

As a result of the mannor of calculating C , the rank of the information matrix is available;

rank = n - number of diagonal terms of C which are identically zero.

Hance, if K is calculated by (3.46), the knowledge of when the pseudo-inverse is required is available. Also, since G is lower triangular, inversion of C directly (when it is full rank) is recommended. The recommended manner of computing the state change in

$$\mathbf{K}(\mathbf{Y} - \hat{\mathbf{Y}}) = (\mathbf{D}^{+1}(\mathbf{H}^{T}\mathbf{Q}^{-1}(\mathbf{Y} - \hat{\mathbf{Y}}))), \qquad (3.47)$$

Calculations are to be carried out in (3.47) starting with innermost bracketed terms. In (3.47)

otherwise.

 $D^{-1} = [F^T F]^{-1} O$ (3.48)

The matrix F in (3.48) is defined by shifting all non-zero columns of C to the left to form a lower dimension matrix. In (3.48)  $D^{-1}$  is the square root of the resudo-inverse of the information matrix.

#### 3.6.4 Discussion

The inverse formulation given here is approximately the same as that of Reference 14. The method given uses Potter's equation for the time update, which is believed superior to the method of Reference 14 from computational considerations. Reference 15 gives this measurement update square root technique for purposes of improving numerical accuracy of the weighted least square type filter. In this latter instance the estimated states are held at a fixed spoch and all the measurements during a given interval are processed in a "batch" scheme. That is, the state is changed only as a result of all the measurements in the batch.

Reference 14 contains several examples which illustrate the superior scourscy of the square root formulation. An example is as follows.

79

(3.42)

1.46.10 164.9

Lot

$$= \begin{bmatrix} 1 & 1 \\ j & 1 + q \end{bmatrix},$$

(3.49)

where  $\bullet$  is some small number. If  $CC^T$  is formed, then

$$\mathbf{CC}^{\mathrm{T}} = \begin{bmatrix} 2 & 2 + \mathbf{e} \\ 2 + \mathbf{e} & 2 + 2\mathbf{e} + \mathbf{g}^{\mathrm{T}} \end{bmatrix}$$
(3.50)

The vector outer product given by

$$\begin{pmatrix} 2/2 \\ (2+e^{1}/\sqrt{2}) \end{pmatrix} \begin{pmatrix} (2/\sqrt{2} + e^{1})/\sqrt{2} \end{pmatrix} = \begin{bmatrix} 2 & 2+e \\ 2+e & 2+2e + \frac{e^{2}}{2} \end{bmatrix}$$
(3.51)

is also formed. Comparing (3.50) and (3.51) shows that the only way of distinguishing rank (2) in (3.50) is the underlined  $e^2$  as opposed to the  $e^2/2$  of (3.51). If an 8 docimal digit computer is considered, then quite obviously when C is squared, (3.50), rank is lost if e is smaller than about  $4 \times 10^4$ . If C is reduced to lower triangular form, using previous algorithms, then

$$\mathbf{C}^{i} = \begin{bmatrix} 2/\sqrt{2} & \mathbf{0} \\ (2+\mathbf{e})/\sqrt{2} & \mathbf{e}/\sqrt{2} \end{bmatrix}$$

C

An examination of the algorithm shows that the relative size of  $\bullet$  to unity is important, rather than  $\bullet^2$ . Numerical rank would not be lost here unless  $\bullet$  is less than approximately 1 × 10<sup>-7</sup> for an 8 decimal dirit computer.

Another point of importance is that, once C has full rank, it appears impossible to lose rank numerically by adding very accurate observations. This is not true of the information matrix formulation.

### 4. TECHNIQUES FOR COMPENSATION OF MODELING BRRORS

The statement of the problem of Section 2 was made in a manner to emphasize the fact that muny types of mathematical modeling errors can exist. As was mentioned in the real time onboard problem, one desires to minimize computational requirements. Hence, the modeling problem is further aggravated by the desire to keep calculations and word lengths to a minimum. As a result;

- (a) The number of state variables is minimized by omitting as many error sources as possible.
- (b) Closed form approximations are used as much as possible to minimize computer speed requirements.
- (c) The minimum word length may be sought for reducing computer costs, size, and weight.
- (d) Fixed point arithmetic is domirable from computer complexity considerations.

### The overall accuracy attainable depends on

- (A) The frequency and accuracy of, and the state space spanned by, the observations.
- (B) The accuracy of the models used in fitting the data.
- (C) The magnitude and spectrum of random forcing functions acting on the system.

The interrelationships between computational requirement considerations and attainable scouracy are very complex in practical problems.

In this section techniques for dynamic model compensation in Kalman filters are addressed. The techniques presented will tend to "look" the satimate of the state to the most recent measurements and thus prevent a divergence of the estimate from the measurements. This divergence occurs when one attempts to fit duta over relatively long time arcs with a poor dynamic model.

### 4.1 Orbit Determination Example

As an example of model error, an orbit dotermination problem where position and velocity of the spacecraft are the only assumed state variables is considered. If such a constraint is imposed on the number of state variables, then it is easy to show that model errors generally make it impossible to find a solution for the position and velocity which fits the actual data.

In this instance

$$\hat{\mathbf{X}} = \mathbf{F}(\hat{\mathbf{X}}, \mathbf{t}) \tag{4.1}$$

 $\hat{Y} = Q(\hat{X}, t)$ . (4.2)



(4.3)

The problem of finding the estimate of the initial state,  $\hat{X}_{0}$  , which minimizes the loss function

 $\mathbf{L} = (\mathbf{Y} - \hat{\mathbf{Y}})^{\mathsf{T}} \mathbf{Q}^{-1} (\mathbf{Y} - \hat{\mathbf{Y}})$ 

is considered.

Equation (4.1) imposes a constraint on the time history of the solution for  $\hat{X}$  if  $\hat{X}_0$  is given. Equation (4.2) gives the functional relationship for computing the observation from the estimate of  $\hat{X}$ .

The real measurements obey reality rather than (4.1) and (4.2). Hence, if one uses Equations (4.1) and (4.2), a value of  $\hat{X}_0$  for which the residuals  $Y - \hat{Y}$  are small and random about zero mean, over all time, implies there are no errors in the mathematical models.

Since a weighted least squares fit is used in this example, one can hypothesize the typical residual behavior of the solution for errors in (4.1) and (4.2). A pictorial representation is shown in Figure 1. The term "short are" means, for example, the data during a single pass of a ground station for a low altitude satellite.

As shown in Figure 1, for a short arc, dynamic model errors (errors in Equation (4.1)) will not prevent a solution for  $\hat{X}_0$  where residuals are essentially random with zero mean. This can be seen by considering a Taylor series solution of (4.1) Let  $\hat{X}_0$  represent the three position components at the beginning of a data arc. Then

$$\hat{X}(t-t_{0}) = \hat{X}_{0} \ \hat{X}_{0}(t-t_{0}) + \hat{X}_{0} \ \left(\frac{t-t_{0}}{2!}\right)^{2} + \dots \qquad (4.4)$$

In the equation,  $\hat{X}_0$  and  $\hat{X}_0$  are free to be estimated, and a nearly perfect fit can be assured as long as  $(t-t_0)$  is small, regardless of the constraining effect of the dynamic model.

The underlined terms of the infinite series (Equation (4.4)) are dependent on the dynamic model; hence, errors in the dynamic model will become effective over long time arcs. One therefore will not be able to find a value of  $\hat{K}_0$  and  $\hat{K}_0$  which produces small residual errors, for example, over a multiple pass. Dynamic model errors for data over long time arcs should cause effects of the type shown in Figure 1(b).

Measurement errors such as biases will also have only a small effect, for short time arcs, as shown in Figure 1(c). The error in the estimate of state will include the bias caused by the measurement model error in this instance. Only when measurements of many different types are processed showld biased offset residuals occur for short data arcs. This latter results from the fact that no single values of  $\tilde{\chi}_0$  and  $\tilde{\chi}_0$  can be selected which fit multiple measurements (more than 6) in the presence of biases or other measurement model errors. A measurement model error, however, is, in general, bounded. Hence, if a long data are over many stations is visualized, measurement model errors would be expected to cause residuals as shown in Figure 1(d). Since the solution is constrained to obey the dynamic model, measurement model errors will have less und less effect on orbit determination accuracy as more data is processed. In these examples the errors are essumed to be small (not mistakes). Typical measurement model errors are station location errors, timing errors and biases.

The real problem, of course, includes both measurement and dynamic model errors. Dynamic errors may have both periodic and secular effects on the residuals. Hence Figure 1 can only indicate trends to be expected. Also, in the general situation, what is weatt by short and multiple pass is not readily defined.

Figure 2 depicts general trends of the effects of measurement and dynamic model errors on state estimation accuracy. The figure is intended to indicate that dynamic errors cause a ever-increasing state estimation error as more data points, over a large time arc, are included. Mensurement errors have the opposite effect.

One desires to have a compensation technique which gives the best balance between these two effects. This will be called dynamic model compensation and will be discussed in the next section. One should recognize, however, that both measurement and dynamic model errors exist and the resulting effect of the dynamic compensation can cause the measurement errors to be dominant.

#### 4.2 Dynamic Nodel Compensation

والمراجع والمراجع

As indicated in the previous section, some means of preventing the state estimation errors from growing indefinitely is necessary. It should be reasonably obvious that one way of preventing this growth is to gradually reduce the influence of past measurements in determining the estimate from current measurements. Assume that totally false measurements (outliers) caused by malfunctions can be removed. The remaining measurements can then be tristed to give the theoretical value based on the correct mode; plus an error term. It is reasonably 1 welly that upper hounds may be placed on the measurement error. That is, the error which is caused, for example, by calibration, atmospheric effects, receiver noise, and so forth, cannot be larger than some specified number. If this is the case, then the error in the computed measurement should be no worse than this upper bound.





are only made to give some reasons why a reasonable degree of trust in the basic measurements must be made to develop a compensation scheme.

Two different approaches for weighting out the influence of past data in the determination of the current estimate are as follows:

(i) Increase the *a priori* covariance matrix (or decrease the information matrix). Arguments for usi this approach are as follows: Equations of motion which are invalid have been used to update the estimate of state. One should therefore increase the *a priori* covariance matrix in accordance with the errors involved in the timeupdating of the estimate. The difficulties in using this approach lie in defining the real error sources. Their formulation can also become extremely complex. Hence, for practical usages, it is perhaps better to may that pseudo-errors are introduced to cause an increase in the *a priori* covariance matrix. These pseudo-errors can be of two types:

- (a) random forcing functions,
- (b) errors attributed to inaccuracies of constants in the equations of motion.

(ii) Overweight the most recent data. In this approach it is also recognized that the *a priori* covariance matrix may be overly optimistic. The matrix, however, is not modified on the basis of adding the effects produced by pseudo-errors in the dynamic model. Instead, a non-optimal filter algorithm is adopted which straches a greater significance to the recent observations than the optimal filter does. The *a posteriori* covariance matrix is modified to conform with the non-optimal algorithm.

Either of these approaches has many variations to suit any specific problem. Although the philosophy is different, both provide a means of developing a dynamic model compensation technique.

### 4.2.1 Pseudo-Random Forcing Function Approach

In the equation for time update of the covariance matrix, (2,33), the quantity R represents an added growth in error caused by random forcing functions. Since dynamic model or computation errors would cause an additional error to exist, the natural fix is to increase R. As already montioned, if

$$\mathbf{R} = \sum_{i=1}^{J} \mathbf{s}_i \sigma_i^2 \mathbf{s}_i^T , \qquad (4.5)$$

then the direction vectors  $s_1$  and the variances  $\sigma_1^2$  are to be selected for compensation of the dynamic model errors.

In the real time onboard problem it is likely that a far more accurate model can be analytically defined than is possible to calculate. An approach for defining the errors is to use a general purpose computer to calculate the difference between the assumed "exact" solution and the approximate solution as computed by a simulated onboard computer. This difference (or error) data is calculated for a number of initial conditions and time durations. The quantities  $s_1$  and  $\sigma_1^2$  are then adjusted to give a reasonable approximation of the error growth. Such a simulation is likely essential in validating the approximations and obsard computational techniques. As a result this approach should not require a large additional effort. The alternate approach of defining  $s_1$  and  $\sigma_1^2$  (as well as j) by "cut and try" experiments with the real system is definitely not recommended for complex systems.

Model errors and computational errors generally cause bias type errors in the estimate. The error sources may be modeled and their effects included in the filter in a more direct manner. The theory for such a technique is given in Reference 16. For measurement type error cources this technique is reasonably efficient. For dynamic errors, however, it requires extensive calculations and does not necessarily give large reductions in state estimation error over the pseudo-random approach<sup>3</sup>.

### 4.2.2 Direct Overweighting of Most Recent Data

The pseudo-random forcing approach gives a higher weighting to the more recent measurements by causing an additive covariance matrix growth between measurements. There is no reason why one should not consider modifications of the filter algorithm for achieving the overweighting in a direct manner.

one such modification is shown in (4.6). This modification is for sequential processing of the observations (one at a time).

$$\hat{X}_{\mathbf{n}} = \hat{X}_{\mathbf{b}} + (\mathbf{P}\mathbf{H}^{\mathsf{T}} + \mathbf{H}^{\mathsf{T}} \mathbf{Q} \epsilon / \mathbf{H} \mathbf{H}^{\mathsf{T}}) (\mathbf{Y} - \hat{\mathbf{Y}}_{\mathbf{b}}) / (\mathbf{H} \mathbf{P} \mathbf{H}^{\mathsf{T}} + \mathbf{Q}) .$$
(4.6)

The scalar ∈ in Equation (4.6) is a control gain.

والمراجع والمراجع

If linearity prevails, then  $H\bar{X}_{a} = \bar{Y}_{a}$  = the computed measurement after the observation. Multiplication of (4.8) by H gives

1.00

$$H\hat{X}_{a} = H\hat{X}_{b} + \left(\frac{H\Gamma H^{T} + Q\epsilon}{H\Gamma H^{T} + Q}\right) (Y - \hat{Y}_{b}) , \qquad (4.7)$$















Note in (4.7) that, for  $\epsilon = 1$ , the estimate of the observation is equal to the observation. Hence, the error in this component of the estimate of state is no worse than the error in the measurement. For  $\epsilon = 0$  the optimal filter is obtained.

This modification was suggested by considering the form for the estimate of state using a single measurement. That is, if

$$y = Hx + q, \qquad (4.8)$$

then, using the pseudo-inverse to calculate the estimate.

$$\mathbf{\hat{t}} = \mathbf{H}^{\mathbf{f}} \mathbf{y} = \frac{\mathbf{H}^{\mathsf{T}} \mathbf{y}}{\mathbf{H}^{\mathsf{T}}}.$$
 (4.9)

As may be noted in Equation (4.6), part of the change in  $X_a$  is proportional to Equation (4.9).

The covariance matrix of errors in  $\hat{X}_{a}$  is found by taking

 $\mathbf{E}(\mathbf{X} - \hat{\mathbf{X}}_{\mathbf{n}}) (\mathbf{X} - \hat{\mathbf{X}}_{\mathbf{n}})^{\mathrm{T}} = \mathbf{P}_{\mathbf{n}} \ .$ 

This operation applied to (4.6) gives

$$P_{a} = P_{b} - P_{b}H^{T}HP_{b}/(HP_{b}H^{T} + Q) + \frac{\epsilon^{2}Q^{2}H^{T}H/[(HP_{b}H^{T} + Q)(HH^{T})^{2}]}{(4.10)}$$

The underlined term of (4.10) shows the additive error (above optimal filtering) caused by the control gain  $\epsilon$  .

Consider the following simple example, using this non-optimal filter.

	Actual Model			Assumed Model			
¥.	÷	0.1 = a scalar		£	ż	0	
3	=	x		\$	=	Î.	
		•					

Let Q = 1.

Ĩ.

P(0+)	=	Q	(after	first	measurement)	
X(0+)	=	10	(after	first	measurement)	
\$(0+)	=	x(0+)				

Figure 3 shows the estimate of state after the observations at two second intervals are included. The estimate between observations remains constant.

The error in the estimate for the Kalman filter grows indefinitely with time. The growth arises because x(t) was assumed to be a constant, while in reality x(t) is equal to a constant plus a time-dependent term. The modified filter  $(\epsilon \neq 0)$  has an error growth between measurements. However, if  $\epsilon = 1$ , the error in f after the measurement is included is no greater than the measurements. For  $\epsilon < 1$ , the error in estimate, after each measurement is processed, has a bias offset from the measurement value. Hence the use of this non-optimal filter tends to look the estimate of state to the recent observations.

Equivalent results to those given in Figure 3 can be obtained by introducing a pseudo-random forcing function. This can resulty be seen, since random forcing functions cause an added growth to P between measurements. Prior to including the  $n^{th}$  measurement for this example.

$$P_{n}(-) = P_{n-1} + R,$$

if R were made extremely large compared to Q, then the results corresponding to  $\epsilon = 1$  of Figure 3 would be obtained.

4.2.3 Discussion

Two techniques for dynamic model error compensation have been briefly described. In the applications with which the author is familiar either approach will prevent the so-called divergence<sup>3</sup>. The bad features of requiring "out and try" are common to both techniques.

One should note that both techniques can be used to prevent loss of positive definiteness of the covariance matrix. For this application the overweighting-of-most-recent-data approach has been easier (from the author's experience) to use.

Other alternatives for model compensation, such as described in References 7 and 8, have not been covered. These alternatives, although applicable, lead to reasonably complex calculations.

10:

t

### 5. CONCLUDING REMARKS

The over-all development of "software" for real time onboard applications of Kalman flitering theory is a very complex subject. This chapter described approaches and algorithms for solving several of the problems which occur in these type applications as well as applications where large general purpose computers are available. There are, however, many other problem areas which must be addressed and resolved in the development of "appropriate" software.

One of the most difficult problems is the definition of the appropriate mathematical models. These models must be sufficiently complex to satisfy the estimation accuracy specifications yet simple enough for the onboard computer. Cleverness in the selection of coordinate systems and fundamental state variables can lead to simple formulations. Approaches which lead to these simplifications are however more of an "art" than a science.

The filter algorithms considered in this chapter were of the "disorete" type. There are instances where continuous filter theory may load to simpler formulations. Propagation of the covariance matrix between observations in problems where the transition matrix is not required is one possibility<sup>17</sup>.

For high data rate problems the filter equations derived from either the discrete or continuous theory may be too complex for the real time computer. Data averaging or data compression techniques appear to offer a good compromise for haudiing such problems<sup>10</sup>. In applying such techniques some information is lost. In addition, improper formulation or inadequate validation of approximations can lead to unwanted bias type errors in the estimate<sup>18</sup>. The "software" dosigner should therefore exercise a good deal of caution in the application of such techniques.

From the material presented it is obvious that many practical solutions of filtering problems exist. It is also obvious that the theory available will not provide a uniquely optimal solution when all factors are considered,

















Detrinter O





### REFERENCES

Contraction and a second second

100-100-

1.	Smith, G.L. et ml.	Application of Statistical Filter Theory to the Optimal Estimation of Position and Velocity On Board a Circumlunar Vehicle. NASA TR R-135, 1962.
2.	Schmidt, 5.F.	State Space Techniques Applied to the Design of a Space Navigation System. JACC, New York University, 1982.
3.	Schlee, F.H. et al.	Divergence in the Kulman Filter. AIAA Journal, Vol.5 1967, pp.1114-1120.
4.	Schmidt, S.F.	Estimution of State with Acceptable Accuracy Constraints. Report 67-4, Analytical Mechanica Asmociates, January 1967.
5.	Schmidt, S.F.	Compensation for Modelling Errors in Orbit Determination Problems. Report 67-16, Analytical Machanics Associates, November, 1907.
6.	Pines, S. et al.	Modifications of the Minimum Variance Program for the Processing of Real Data. Analytical Mechanics Associates Technical Report, October 1964,
7.	Jazwinski, A.H.	Limited Memory Optimal Filtering. Report 67-11, Analytical Mechanics Associates, August, 1967.
8.	Jazwinski, A.H. Bailie, A.E.	Adaptive Filtering. Report 67-6, Analytical Mechanics Associates, March 1967.
9.	Kaluan, R.E.	A New Approach to Linear Filtering and Prediction Problems. Journal of Basic Engineering, March 1960, pp.33-45.
. 10.	Sohmidt, S.F. et al.	Case Study of Kalman Filtering in the C-5 Aircraft Navigation System. Institute of Electrical and Electronic Engineers Case Studies Seminar, Ann Arbor, Michigan. June 1968
11.	Faddeev, D.K. Fadeeva, V.N.	Computational Methods of Linear Algebra. Freeman, Ban Francisco. 1963, pp.144-147.
12.	Battin, R.H.	Astronautical Guidance. McGraw-Kill, 1964, pp. 339-340.
13.	Bellentoni, J.F. Dodge, K.W.	A Square Root Formulation of the Kalman-Schmidt Filter. AIAA Journal, Vol.5 1967, pp. 1309-1314.
14.	Dyer, P. McReynolds, S.R.	Computational Accuracy of Square Root Filtering. Space Programs Summary 37-54, Vol.III, pp 15-19, for period October 1 to November 30, 1968, Jet Propulsion Lateratory, December 31, 1968.
15.	Schmidt, S.F.	Final Report for Mission Analysis Guidance Study. Analytical Mechanics Associates Report 68-23, January 1969.
18.	Schuidt, S.F.	Application of State Space Methods to Navigation Problems. "Advances in Control Systems", Volume 3, Academic Press, 1968.
17.	Gunckel, T.L.	Guidance and Control of Reentry and Aerospace Vehicles. "Advances in Control Systems", Volume 3, Academic Press, 1966, pp.20-21.

ASSAN DE PASSAR DAAR



in the second second

\_\_\_\_**\_**\_\_\_



12222020 - 104603669 - 13722200 - 12222623

53







Fig. 2 State estimate error trends resulting from model errors



Fig.3 Example results using model compensation in a Kalman filter









an ann an Aonaichte Anns an Aonaichte Anns

ROLLING

At Lines

MAN MARSH

State in Maria

I WA P X PA

.

### CHAPTER 4 - MODELING ERRORS IN KALMAN FILTERS

by

...

### T. Nishimura

## Jet Propulsion Laboratory, California Institute of Technology, Pasadena, California, USA

.

. .

•

. .

87

•



## CHAPTER 4 - MODELING ERRORS IN KALMAN FILTERS

T.Nishimura

### 1. INTRODUCTION

(

In recent same the Kalman filter<sup>1,2</sup> has been extensively used in such applications as the tracking of missiles or planes and the orbit determination of spacecraft. One of the problems arising in these applications is that very often a precise knowledge of the *a priori* statistics of initial conditions and of the noise model (process noise and observation noise), as well as those of system models, are not available, while it is essential for the design of optimal fifters.

For example, in the orbit determination problem of spacecraft in deep space, the observation is usually supplied in the form of Doppler, counted Doppler or range data. These data are subject to oscillator instability, disturbances in the ionomphere, receiver noise, and guantization noise of the counter, which together constitute the observation noise.

Also the spacecraft undergoes, during its long journey to the planet (s.g., about 200 days for a typical Mars mission), various unknown disturbances originating from solar pressure, impacts of meteorites, and fuel leakage from valves. It is a rather difficult task to determine the statistics of any one of these noise sources. Besides the uncertainty of the injection conditions of the spacecraft, the coordinates after the mideourse manuer may enter into the filter design and influence the gain of the filter strongly during the initial period of estimation.

Errors are inevitable in assigning a priori covariance matrices of large dimensions, because of lack of sufficient experience or of incapability of analyzing complex correlations among parameters.

Furthermore lack of precise knowledge of system models is a problem which practising engineers frequently encounter in designing filters. This matter is also closely related to the problem of identification, which is mother major topic in control theory and its applications.

In this chapter attention is focused on the analysis of the effect of errors in these a priori statistics and mystem models on the performance of the resulting suboptimal filters. Both continuous systems and discrete systems are analyzed and an effort is made to find the upper and lower bound for the error covariances of these suboptimal filters.

Throughout this chapter it is assumed that the systems are linear and that the stuchastic variables have gaussian distributions.

### A. ANALYSIS FOR CONTINUOUS SYSTEMS

The basic process is described by a first-order differential equation in vector form,

 $\frac{d\mathbf{x}(t)}{dt} = \mathbf{F}(t)\mathbf{x}(t) + \mathbf{G}(t)\mathbf{w}(t) . \tag{1}$ 

The observation is

### y(t) = H(t)x(t) + u(t),

where x(t) is an  $n_x$  vector of states, with

- $\mathbf{E}[\mathbf{x}(0)] = \mathbf{0} \ .$
- y(t) is an ng vector of observations.
- w(t) is an n. vactor of stochastic inputs to the process with .

#[w(t)] = 0

### $\mathbb{E}[\mathbb{I}(t)\mathbb{I}'(\tau)] = \mathbb{Q}(t)\delta(t-\tau) .$

where  $\delta(t)$  is the Dirac delta function.

n(t) is an n<sub>n</sub> vector of the observation noise with PRECEDING PAGE BLANK

(2)

(3)

(4)



F(t), G(t), H(t) are  $n_x \times n_x$ ,  $n_x \times n_y$ ,  $n_y \times n_x$  matrices respectively.

E[] is an exposted value operator on stochastic variables. It is also assumed that the process noise w and observation noise n have no corrolation to each other,

 $\mathbb{E}[w(t)n'(\tau)] = 0.$  (7)

Then the optimal estimator  $x^{*}(t)$  of x(t) which minimizes

E[||x\*-x||2] ,

having the observation y(t) from t = 0 to t, is described by the differential equation<sup>2</sup>

$$\frac{dx^{*}(t)}{dt} = F(t)x^{*}(t) + K(t)[y(t) - H(t)x^{*}(t)], \qquad (8)$$

where

where

 $K(t) = P(t)H'(t)R^{-1}(t)$ .

It is assumed that R(t) is positive definite for  $t \ge 0$ .

Then the dovariance matrix P(t) is defined by

$$P(t) \triangleq E[(x^{*}(t) - x(t))(x^{*}(t) - x(t))']$$

and it is obtained as a solution of a matrix Ricosti equation

$$\frac{dP(t)}{dt} = F(t)P(t) + P(t)F'(t) - P(t)H'(t)R^{-1}(t)H(t)P(t) + Q(t)Q(t)Q'(t) . \qquad (11)$$

The initial conditions for Equations (8) and (11) are respectively

$$x^{*}(0) = 0$$
 (12)

$$P(0) = B[x(0)x'(0)], \qquad (13)$$

### 3." ASSUMPTIONS AND MATHEMATICAL DERIVATIONS

. The optimal estimator described in the previous section is based on correct information for initial conditions, noise covariances and coefficient matrices. Suppose the estimator design is based on incorrect information with respect to these quantities, namely,

- (i) the incorrect  $P_{\alpha}(0)$  rather than the correct P(0) (a priori covariance of states),
- (ii) the incorrect  $Q_{ij}(t)$  rather than the correct Q(t) (covariance of the process noise),
- (iii) the incorrect  $R_{\alpha}(t)$  rather than the correct R(t) (covariance of the observation noise),
- (iv) the incorrect  $P_{ci}(t)$  rather than the correct P(t) (process matrix)
- (v) the incorrect  $G_{o}(t)$  rather than the correct G(t) (coefficient matrix of the process noise),
- (vi) the incorrect H<sub>0</sub>(t) rather than the correct H(t) (observation matrix). The resulting matrix is no longer than an optimal one, but becomes suboptimal.

This suboptimal estimator is denoted  $x_{n}^{*}(t)$  and it is described by

$$\frac{d\mathbf{x}_{\mathbf{B}}^{*}(t)}{dt} = \mathbf{P}_{\mathbf{0}}(t) \mathbf{x}_{\mathbf{B}}^{*}(t) + \mathbf{K}_{\mathbf{0}}(t) \left[ \mathbf{y}(t) - \mathbf{H}_{\mathbf{0}}(t) \mathbf{x}_{\mathbf{B}}^{*}(t) \right], \qquad (14)$$

$$x_{\pm}^{\pm}(0) = 0$$
 (15)

$$K_{o}(t) = P_{o}(t)H_{o}'(t)R_{o}^{-1}(t) , \qquad (19)$$

and the calculated covariance  $P_0(t)$  is computed by the same Riccati equation as Equation (11), but using the ideorrect model specified by (i) - (vi).

$$\frac{d\mathbf{P}_{a}(t)}{dt} = \mathbf{P}_{a}(t)\mathbf{P}_{a}(t) + \mathbf{P}_{a}(t)\mathbf{P}_{a}'(t) - \mathbf{P}_{a}(t)\mathbf{H}_{a}'(t)\mathbf{R}_{a}^{-1}(t)\mathbf{H}_{a}(t)\mathbf{P}_{a}(t) + \mathbf{G}_{a}(t)\mathbf{Q}_{a}'(t)\mathbf{Q}_{a}'(t) .$$
(17)

90

(9)

(10)



(23)

The actual covariance  $P_{a}(t)$  is defined as the error covariance associated with the suboptimal estimator of Equation (14).

where

with

$$P_{n}(t) \triangleq E[(x_{n}^{*}(t) - x(t))(x_{n}^{*}(t) - x(t))'], \qquad (18)$$

This is the covariance expected on the estimator when there is insufficient information on the design parameters. It is the main objective of this section to derive equations describing this  $P_{i}(t)$  .

### For this purpose it is easier to derive a differential equation for $P_{a}(t)$ .

tt

Thus differentiation of  $P_n(t)$  of Equation (18), and a change in the order of the differentiating operator and the expected value operator, yield

$$\dot{P}(t) = E[(\dot{x}^{\dagger}(t) - \dot{x}(t))(x^{\dagger}(t) - x(t))'] + E[(x^{\dagger}(t) - x(t))(\dot{x}^{\dagger}(t) - \dot{x}(t))'], \quad (19)$$

However, from Equations (1) and (14),

 $\dot{x}^{\bullet}_{a}(t) - \dot{x}^{\bullet}(t) = (F_{a}(t) - K_{a}(t)H_{a}(t))(\dot{x}^{\bullet}_{a}(t) - x(t)) + \Delta F(t)x(t) - K_{a}(t)\Delta H(t)x(t) + K_{a}(t)n(t) - O(t)w(t) ,$ (20)

$$\Delta \mathbf{F}(\mathbf{t}) = \mathbf{F}(\mathbf{t}) - \mathbf{F}(\mathbf{t}) \tag{21}$$

$$\Delta H(t) = H_n(t) - H(t) . \tag{22}$$

Also, x(t) is obtained from Equation (1) as

$$x(t) = U(t,0)x(0) + \int_0^t U(t,s)G(s)w(s) ds ,$$

where U(t,s) is defined by

$$\frac{\partial U(t, m)}{\partial t} = F(t)U(t, m), \qquad (24)$$

$$(\mathbf{s},\mathbf{s}) = \mathbf{I}, \quad \mathbf{t} \ge \mathbf{s} \ge \mathbf{0}, \quad (25)$$

and I is an identity matrix.

Furthermore  $x_{n}^{+}(t)$  is derived from Equation (14) as

$$\mathbf{x}_{\mathbf{a}}^{\mathbf{b}}(t) = \int_{0}^{t} \mathbf{V}_{\mathbf{a}}(t, \mathbf{s}) \mathbf{K}_{\mathbf{a}}(\mathbf{s}) \mathbf{y}(\mathbf{s}) \, \mathrm{d} \mathbf{s} \,, \qquad (26)$$

where  $V_n(t, s)$  is defined by

$$\frac{\partial V_{g}(t, \mathbf{s})}{\partial t} = (F_{g}(t) - K_{g}(t)H_{g}(t))V_{g}(t, \mathbf{s}), \quad t \ge \mathbf{s} \ge 0.$$
(27)

When  $\dot{x}^{*}_{\mathbf{x}}(t)$  and  $x^{*}_{\mathbf{x}}(t)$  are substituted into Equation (19), together with  $\dot{x}(t)$  of Equation (20) and its solution x(t) in Equation (23), paying attention to the fact that w(t) and n(t) are uncorrelated white noises, the following three differential equations are derived.

$$\frac{dP_{A}(t)}{dt} = (F_{0}(t) - K_{0}(t)H_{0}(t))P_{A}(t) + P_{A}(t)(F_{0}(t) - K_{0}(t)H_{0}(t))' + (\Delta F(t) - K_{0}(t)\Delta H(t))\Lambda(t) + \Lambda'(t)(\Delta F(t) - K_{0}(t)\Delta H(t))' + (K_{0}(t)R(t)K_{0}(t) + Q(t)Q(t)Q'(t)$$
(28)

$$\frac{d\Lambda(t)}{dt} = \mathbf{F}(t)\Lambda(t) + \Lambda(t)(\mathbf{F}_{0}(t) - \mathbf{K}_{0}(t)\mathbf{H}_{0}(t))' + \mathbf{P}_{\mathbf{g}}(t)(\Delta\mathbf{F}(t) - \mathbf{K}_{0}(t)\Delta\mathbf{H}(t))' - \mathbf{G}(t)\mathbf{Q}(t)\mathbf{G}'(t)$$
(29)

$$\frac{P_{x}(t)}{dt} = P(t)P_{x}(t) + P_{x}(t)P'(t) + O(t)Q(t)O'(t) , \qquad (30)$$







where  $\Lambda(t)$  and  $P_{x}(t)$  are defined by

$$\Lambda(t) \stackrel{\Delta}{=} \mathbb{E}[\mathbf{x}(t)(\mathbf{x}_{\mathbf{x}}^{\mathsf{h}}(t) - \mathbf{x}(t))'] \qquad (31)$$

$$P_{x}(t) \stackrel{f}{=} \mathbb{E}[x(t)x'(t)],$$
 (32)

The initial conditions for these differential equations are respectively given by

 $\mathbf{P}_{\mathbf{g}}(0) = \mathbf{P}(0)$ 

$$\Lambda(0) = -P(0) \tag{34}$$

$$P_{1}(0) = P(0)$$
. (35)

### 4. ERROR BOUNDS OF SUBOPTIMAL FILTERS (CONTINUOUS CASE)

When both process matrix F and observation matrix H are known correctly, the results in the previous section can be simplified considerably. In fact, only the first differential equation needs to be solved?

$$\frac{dP_{a}(t)}{dt} = (F(t) - K_{o}(t)H(t))P_{a}(t) + P_{a}(t)(F(t) - K_{o}(t)H(t))' + K_{u}(t)R(t)K_{o}'(t) + G(t)Q(t)Q'(t) , \qquad (36)$$

with

$$K_{0}(t) = P_{0}(t)H'(t)R_{0}^{-1}(t)$$
 (37)

$$\mathbf{P}_{\mathbf{A}}(0) = \mathbf{P}(0) ,$$

Let  $\mathbb{F}_{da}(t)$  be the difference between the computed covariance  $P_{d}(t)$  and the actual covariance  $P_{a}(t)$ , then

$$E_{n_{a}}(t) = P_{n}(t) - P_{n}(t)$$
 (39)

V,(t,=)@(\*)∆Q(=)G'(=)V,

Substituting this into Equation (36) yields the following differential equation of  $E_{os}(t)$ , with the aid of Equation (17):

$$\mathbf{E}_{OB}(t) = (\mathbf{F}(t) - \mathbf{K}_{O}(t)\mathbf{H}(t))\mathbf{E}_{OB}(t) + \mathbf{R}_{OB}(t)(\mathbf{F}(t) - \mathbf{K}_{O}(t)\mathbf{H}(t))' + \mathbf{K}_{O}(t)\Delta\mathbf{R}(t)\mathbf{K}_{O}'(t) + \mathbf{G}(t)\Delta\mathbf{Q}(t)\mathbf{Q}(t))' + \mathbf{K}_{O}(t)\mathbf{K}_{O}'(t) + \mathbf{G}(t)\Delta\mathbf{Q}(t)\mathbf{Q}(t) + \mathbf{K}_{O}(t) + \mathbf{K}_{O}(t)\mathbf{K}_{O}'(t) + \mathbf{K}_{O}(t)\mathbf{K}_{O}(t)\mathbf{K}_{O}'(t) + \mathbf{K}_$$

$$\Delta \mathbf{q}(\mathbf{t}) \stackrel{i=1}{\rightarrow} \mathbf{q}_{\mathbf{q}}(\mathbf{t}) - \mathbf{q}(\mathbf{t})$$

Since Equation (39) is a linear differential equation, an explicit analytic solution can be derived.

As observed from the Equation (43),  $E_{OB}(t)$  is a sum of real symmetric matrices, so that if is semilpositive definite provided every term in the right-hand side of Equation (43) is semi-positive definite. Because of the specific (symmetric) configuration of these terms, every one of them will be respectively applied to the definite if every matrix at the conter of the respective term, namely  $E_{OB}(0)$ ,  $\Delta R(s)$ , and  $\Delta Q(s)$  for  $t \ge s \ge 0$ , we semi-positive definite.

((ti)) da 2

Basad on this discussion, the following theorem can be derived?.

 $\mathbf{E}_{on}(t) = V_{o}(t, 0) \mathbf{E}_{on}(0) V_{o}'(t, 0) + \int_{0}^{t} V_{\mathbf{a}}(\underline{t}, a) K_{o}(\underline{s}) \Delta \mathbf{R}_{\mathbf{a}}(\mathbf{s}) K_{o}'(a) V$ 

Theorem 1

 $E_{_{OR}}(t) \geqslant 0$  ; hence  $P_{_{O}}(t) \geqslant P_{_{R}}(t)$  for  $t \geqslant 0$  if the condition C-I is satisfied.

The implication of the inequality sign (plus equality sign), e.g.,  $P_{g}(t) \ge P_{g}(t)$ , in that the difference matrix  $P_{g}(t) - P_{g}(t)$  is semi-positive definite.

(33)

(38)

, 11 m - Hall Mart Mart Mart Lag Land Kang Kang Martin Karatan Martin Karat

لاستفسدت العذفة الاتر

a sure a sure and a sure

**	iqa (	14.11	1.000				
	<u>.</u>			<u>_</u>	1.1		
· ·	٠. '		<u>ان ا</u>		<u>_</u>		

Therefore, an upper bound for the variances of the suboptimal estimator  $x_{n}^{*}(t)$  can be set, and it is equal to the diagonal components of the calculated covariance  $P_{0}(t)$  when the condition C-I is satisfied. The lower bound of these variances is, of course, zero. So let  $p_{0ii}(t)$  and  $p_{aii}(t)$  by respective diagonal components of  $P_{0}(t)$  and  $P_{a}(t)$ ; then

$$p_{nii}(t) \ge p_{nii}(t) \ge 0. \tag{44}$$

Even though the designer does not know exactly the *a priori* statistics, he can expect that his suboptimal esimator will properly behave within the specified range provided the conservative condition C-I is satisfied.

Though it is of less practical importance, the following Corollary is derived from Equation (43).

### Corollary 1

.

 $E_{nn}(t) \leq 0$ ; hence  $P_n(t) \leq P_n(t)$  for  $t \geq 0$  if the condition C-II is satisfied.

C-II:  $\mathbf{E}_{ch}(0) \leq 0$ ,  $\Delta \mathbf{Q}(t) \leq 0$ , and  $\Delta \mathbf{R}(t) \leq 0$  or, equivalently,  $\mathbf{P}_{c}(0) \leq \mathbf{P}_{h}(0)$ ,  $\mathbf{Q}_{c}(t) \leq \mathbf{Q}(t)$ , and  $\mathbf{R}_{c}(t) \leq \mathbf{R}(t)$  for  $t \geq 0$ .

Furthermore, differential equations associated with the other two error matrices,  $E_{a0}(t)$  and  $E_{ov}(t)$ , which are defined as

$$E_{a0}(t) = P_{a}(t) - P(t)$$
 (45)

$$\mathbf{E}_{no}(t) = \mathbf{P}_{n}(t) - \mathbf{P}(t) , \qquad (46)$$

will be derived.

First  $\dot{E}_{g,0}(t)$  can be obtained as a difference between  $\dot{P}_g(t)$  and  $\dot{P}(t)$  given by Equations (36) and (11), respectively:

$$\hat{\mathbf{E}}_{n}(t) = (F(t) - K_{n}(t)H(t))E_{nn}(t) + E_{nn}(t)(F(t) - K_{n}(t)H(t))' +$$

+ 
$$(K_{-}(t)R(t) - P(t)H'(t))R(t)^{-1}(K_{-}(t)R(t) - P(t)H'(t))'$$
, (47)

When a similar discussion leading to Theorem 1 is applied to Equation (37) it may be concluded that  $E_{\rm mo}(t)$  is always semi-positive definite for all  $t \ge 0$ , because R(t) is positive definite by assumption and  $E_{\rm mo}(0)$  is semi-positive definite, as deduced from the definition of P(0). This is described in the following theorem.

Theorem 2

$$\mathbf{E}_{n}(t) \ge 0$$
; hence  $\mathbf{P}_{n}(t) \ge \mathbf{P}(t)$  for  $t \ge 0$ .

This result can naturally be expected because P(t) is the ministum variance by definition.

Similarly the differential equation for  $\mathbf{E}_{g0}(t)$  is derived by subtracting  $\dot{\mathbf{P}}(t)$  of Equation (11) from  $\dot{\mathbf{P}}_{g}(t)$  of Equation (17).

 $\mathbf{E}_{ab}(t) = \{\mathbf{F}(t) - \mathbf{K}_{a}(t)\mathbf{H}(t)\}\mathbf{E}_{ab}(t) + \mathbf{E}_{ab}(t)\{\mathbf{F}(t) - \mathbf{K}_{a}(t)\mathbf{H}(t)\}' +$ 

+  $E_{ab}(t)H'(t)R_{a}^{-1}(t)H(t)E_{ab}(t) + U(t)\Delta Q(t)G'(t) +$ 

+ 
$$P(t)H'(t)(R(t)(\Delta R(t))^{-1}H(t) + R(t)^{-1}H(t)P(t)$$
 for  $\Delta R > 0$ .

Again, it is clear from this equation that  $E_{00}(t)$  is semi-positive definite if C-I is satisfied. When  $\Delta R = 0$  the same conclusion can be proved by taking a limit  $\Delta R \to 0$ .

Thus the following theorem is obtained.

Theorem 3

 $\mathbf{E}_{no} \ge 0$ ; hence  $\mathbf{P}_{o}(t) \ge \mathbf{P}(t)$  for  $t \ge 0$ , if C-I is satisfied.

#### 5. ANALYSIS FOR DISCRETE SYSTEMS

In this section the same technique is applied to discrete systems and similar results are derived. Symbols are defined in the same manner as in the continuous systems, and similar assumptions are made concerning modeling errors and noise statistics.



(48)

with

2.22.0

The process equation and observation equations are respectivel.	y .
$\mathbf{x}(\mathbf{k}+\mathbf{i}) = \Phi(\mathbf{k})\mathbf{x}(\mathbf{k}) + 0(\mathbf{k})\mathbf{x}(\mathbf{k}) + 0($	k)w(k) (49)
y(k) = H(k)x(k) + n(k)	). (80)
Then the optimal estimate $x^{+}(k+1)$ having the information $Y(k)$	= $[y(0), y(1),, y(k)]$ is given by <sup>1</sup>
$x^{*}(k+1) = \Phi(k)x^{*}(k) + K(k)(y(k))$	- H(k)x*(k)) , (51)
where $K(k) = \Phi(k)P(k)H'(k)[H(k)P(k)H'$	(k) + B(k)] <sup>-1</sup> (52)
$\mathbf{x}^{\bigstar}(0) = 0 \ .$	(83)

The covariance matrix P(k) is dufined by

$$P(k) \stackrel{\Delta}{=} \mathbb{E}[(x^{+}(k) - x(k))(x^{+}(k) - x(k))]^{2}]$$
(54)

and it is governed by the nonlinear difference equation

$$P(k+1) = (\Phi(k) - K(k)H(k))P(k)(\Phi(k) - K(k)H(k))' + K(k)R(k)K(k)' + U(k)Q(k)U'(k) ,$$
 (55)

with 
$$P(0) = E[x(0)x'(0)]$$
, (56)

When the incorrect models which are the counterparts in discrete systems of those described in (1) - (vi) are used, the resulting suboptimal estimator  $x_{\pm}^{+}(k)$  is computed by

> $x_{a}^{+}(k+1) = \Phi_{c}(k) x_{a}^{+}(k) + K_{c}(k) (y(k) - H_{c}(k) x_{a}^{+}(k))$ , (57)

$$K_{a}(k) = \Phi_{a}(k)P_{a}(k)H_{a}'(k)[H_{a}(k)P_{a}(k)H_{b}'(k) + R_{a}(k)]^{+1}$$
(58)

$$x_{\pm}^{*}(0) = 0$$
. (59)

The calculated covariance  $P_{cr}(k)$  is

$$\mathbf{P}_{\sigma}(\mathbf{k}+\mathbf{1}) \simeq \left(\Phi_{\sigma}(\mathbf{k}) - \mathbf{X}_{\sigma}(\mathbf{k})\mathbf{H}_{\sigma}(\mathbf{k})\right)\mathbf{P}_{\sigma}(\mathbf{k})\left(\Phi_{\sigma}(\mathbf{k}) - \mathbf{K}_{\sigma}(\mathbf{k})\mathbf{H}_{\sigma}(\mathbf{k})\right)' + \mathbf{K}_{\sigma}(\mathbf{k})\mathbf{R}_{\sigma}(\mathbf{k})\mathbf{K}'(\mathbf{k}) + \mathbf{G}_{\sigma}(\mathbf{k})\mathbf{Q}_{\sigma}(\mathbf{k})\mathbf{G}_{\sigma}'(\mathbf{k}) . \tag{60}$$

The actual covariance associated with this suboptimal estimator  $x^{\bullet}_{n}(k)$  is defined as

$$P_{x}(k) \triangleq m[(x_{x}^{*}(k) - x(k))(x_{x}^{*}(k) - x(k))], \qquad (61)$$

The recurrence equations describing this  $P_{\mu}(k)$  are derived in a similar manner to the continuous case<sup>10</sup>.

$$\begin{split} \mathbf{P}_{\mathbf{R}}(\mathbf{k}+\mathbf{1}) &= (\Phi_{\mathbf{0}}(\mathbf{k}) - \mathbf{K}_{\mathbf{0}}(\mathbf{k})\mathbf{H}_{\mathbf{0}}(\mathbf{k}))\mathbf{P}_{\mathbf{R}}(\mathbf{k})(\Phi_{\mathbf{0}}(\mathbf{k}) - \mathbf{K}_{\mathbf{0}}(\mathbf{k})\mathbf{H}_{\mathbf{0}}(\mathbf{k}))^{\prime} + \\ &+ (\Delta\Phi(\mathbf{k}) - \mathbf{K}_{\mathbf{0}}(\mathbf{k})\Delta\mathbf{H}(\mathbf{k}))\Lambda(\mathbf{k})(\Phi_{\mathbf{0}}(\mathbf{k}) - \mathbf{K}_{\mathbf{0}}(\mathbf{k})\mathbf{H}_{\mathbf{0}}(\mathbf{k}))^{\prime} + \\ &+ (\Phi_{\mathbf{0}}(\mathbf{k}) - \mathbf{K}_{\mathbf{0}}(\mathbf{k})\mathbf{H}_{\mathbf{0}}(\mathbf{k}))\Lambda^{\prime}(\mathbf{k})(\Delta\Phi(\mathbf{k}) - \mathbf{K}_{\mathbf{0}}(\mathbf{k})\Delta\mathbf{H}(\mathbf{k}))^{\prime} + \\ &+ (\Delta\Phi(\mathbf{k}) - \mathbf{K}_{\mathbf{0}}(\mathbf{k})\Delta\mathbf{H}(\mathbf{k}))\mathbf{P}_{\mathbf{x}}(\mathbf{k})(\Delta\Phi(\mathbf{k}) - \mathbf{K}_{\mathbf{0}}(\mathbf{k})\Delta\mathbf{H}(\mathbf{k}))^{\prime} + \\ &+ \mathbf{K}_{\mathbf{0}}(\mathbf{k})\mathbf{K}_{\mathbf{0}}^{\prime}(\mathbf{k}) + \mathbf{G}(\mathbf{k})\mathbf{Q}(\mathbf{k})\mathbf{G}^{\prime}(\mathbf{k}) \ . \end{split}$$

 $\Lambda(\mathbf{k}+\mathbf{1}) = \Phi(\mathbf{k})\Lambda(\mathbf{k})\left(\Phi_{\mathbf{q}}(\mathbf{k}) - \mathbf{K}_{\mathbf{q}}(\mathbf{k})\mathbf{H}_{\mathbf{q}}(\mathbf{k})\right)' + \Phi(\mathbf{k})\mathbf{P}_{\mathbf{g}}(\mathbf{k})\left(\Delta\Phi(\mathbf{k}) - \mathbf{K}_{\mathbf{p}}(\mathbf{k})\Delta\mathbf{H}(\mathbf{k})\right)' - \mathbf{Q}(\mathbf{k})\mathbf{Q}(\mathbf{k})\mathbf{Q}'(\mathbf{k})$ 

$$\mathbf{P}_{\mathbf{x}}(\mathbf{k}+\mathbf{1}) = \Phi(\mathbf{k})\mathbf{P}_{\mathbf{x}}(\mathbf{k})\Phi'(\mathbf{k}) + \mathbf{Q}(\mathbf{k})\mathbf{Q}(\mathbf{k})\mathbf{Q}'(\mathbf{k}) ,$$

where  $\Lambda(\mathbf{k})$  and  $\mathbf{P}_{\mathbf{x}}(\mathbf{k})$  are defined by

$$\Lambda(\mathbf{k}) \triangleq \mathbf{E}[\mathbf{x}(\mathbf{k})(\mathbf{x}_{\mathbf{k}}^{*}(\mathbf{k}) - \mathbf{x}(\mathbf{k}))']$$
(65)

$$P_{\pi}(k) \triangleq \mathbb{E}[\pi(k)\pi'(k)] . \tag{66}$$

The initial conditions for these recurrence equations are respectively given by

$$\Lambda(0) = -P(0)$$
 (68)



(62)

(64)

#### 6. ERROR BOUNDS OF SUBOPTIMAL FILTERS (DISCRETE CASE)

When the process transition matrix  $\hat{\Psi}(k)$  and the observation matrix H(k) are perfectly known, only the first recurrence equation, Equation (A), needs to be solved in order to find  $P_{a}(k)$ . Namely<sup>4-6</sup>,<sup>6</sup>

 $P_{n}(k+1) = (\Phi(k) - K_{n}(k)H(k))P_{n}(k)(\Phi(k) - K_{n}(k)H(k))' + K_{n}(k)R(k)K_{n}'(k) + O(k)Q(k)G'(k) .$ (70)

Then the difference matrix  $F_{ch}(k+1)$  between  $P_{h}(k+1)$  and  $P_{c}(k+1)$  of Equations (53) and (58) respectively becomes

$$\mathbf{E}_{d\mathbf{k}}(\mathbf{k}+1) = (\Phi(\mathbf{k}) - \mathbf{K}_{d}(\mathbf{k})\mathbf{H}(\mathbf{k}))\mathbf{E}_{d\mathbf{k}}(\mathbf{k})(\Phi(\mathbf{k}) - \mathbf{K}_{d}(\mathbf{k})\mathbf{H}(\mathbf{k}))' + \mathbf{K}_{d}(\mathbf{k})\Delta\mathbf{R}(\mathbf{k})\mathbf{K}_{d}'(\mathbf{k}) + \mathbf{G}(\mathbf{k})\mathbf{G}(\mathbf{k})\mathbf{G}'(\mathbf{k}), \qquad (71)$$

where

$$E_{n_{n}}(k) = P_{n}(k) - P_{n}(k)$$
 (72)

Following the same discussions used in the continuous case as well as the induction, the following theorem can be derived for discrete systems.

#### Theorem 4

 $\mathbb{E}_{on}(k) \ge 0$ ; hence  $\mathbb{P}_{o}(k) \ge \mathbb{P}_{n}(k)$  for  $k \ge 0$  if the condition C-III is satisfied.

C-III: 
$$E_{cm}(0) \ge 0$$
,  $\Delta q(k) \ge 0$  and  $\Delta R(k) \ge 0$  or equivalently  $P_{q}(0) \ge P_{m}(0)$ ,  $Q_{q}(k) \ge Q(k)$  and  $R_{q}(k) \ge R(k)$  for  $k \ge 0$ .

Also the counterpart of Corollary 1 is derived, which yields the lower bound of  $P_{\rm m}(k)$  .

### Corollary 2

 $E_{on}(k) \leq 0$ ; hence  $P_{o}(k) \leq P_{n}(k)$  if the condition C-III is satisfied.

In the case of the other two differences,

$$\mathbf{E}_{nn}(\mathbf{k}) = \mathbf{P}_{n}(\mathbf{k}) - \mathbf{P}(\mathbf{k}) \tag{13}$$

$$E_{n0}(k) = P_n(k) - P(k)$$
, (74)

results similar to the continuous case can be proved.

First Equations (60) and (70) are substituted into Equation (73) and after certain manipulation of matrices the following matrix form can be derived.

 $\mathbf{E}_{a0}(k+1) = (\Phi(k) - K_{0}(k)H(k))\mathbf{E}_{00}(k)(\Phi(k) - K_{0}(k)H(k))' + (K_{0}(k) - K(k))B(k)(K_{0}(k) - K(k))', \quad (7\delta)$ 

$$B(\mathbf{k}) = H(\mathbf{k})P(\mathbf{k})H'(\mathbf{k}) + R(\mathbf{k})$$

where

$$B_{\mu}(k) = H_{\mu}(k)P_{\mu}(k)H_{\mu}'(k) + R_{\mu}(k)$$
 (77)

In this derivation the following relation is useful:

$$K_{-}(k) = K(k) + (\Phi(k) - K(k)H(k))E_{nn}(k)H'(k)E_{nn}^{n-1}(k) - K(k)\Delta K(k)E_{n-1}^{n-1}(k) .$$
(78)

In view of Equation (73), the following theorem is derived.

Theorem 5

$$\mathbb{E}_{k}(\mathbf{k}) \ge 0$$
; hence  $\mathbb{P}_{k}(\mathbf{k}) \ge \mathbb{P}(\mathbf{k})$  for  $\mathbf{k} \ge 0$ .

This is a natural conclusion because P(k) is the optimum covariance by definition.

For the third difference matrix  $\mathbf{E}_{no}(\mathbf{k})$  the following relation is obtained:

 $\mathbf{E}_{oo}(\mathbf{k}+1) = (\mathbf{\Phi}(\mathbf{k}) - \mathbf{K}_{o}(\mathbf{k})\mathbf{H}(\mathbf{k}))\mathbf{E}_{oo}(\mathbf{k})(\mathbf{\Phi}(\mathbf{k}) - \mathbf{K}_{o}(\mathbf{k})\mathbf{H}(\mathbf{k}))' +$ 

+ 
$$(K_{\alpha}(k) - K(k))S(k)(K_{\alpha}(k) - K(k))' + K_{\alpha}(k)\Delta R(k)K_{\alpha}'(k) + Q(k)\Delta Q(k)Q'(k)$$
.

Then the following theorem is derived.

Theorem 6

 $B_{n,n}(k) \geqslant 0$  , hence  $P_n(k) \geqslant P(k)$  for  $k \geqslant 0$  , if C-III is satisfied.

85

(70)

(70)

### 7. EXAMPLES

Two examples demonstrate the theoretical analysis of this chapter. The first example is concerned with the modeling errors in the a priori statistics and the second example with the system modeling errors.

### Example 1

A spacecraft is cruising with a constant speed along a straight line and information is supplied by the range data, which are contaminated by white noise having the spectral density  $\Phi_{\mu}$  and zero mean.

Let  $\mathbf{x}_1$  and  $\mathbf{x}_2$  be deviations of speed and position of the spacecraft from the standard trajectory, respectively. Then the process equation becomes

$$k_{1}(t) = 0$$
 (80)

$$\dot{x}_{2}(t) = x_{1}(t)$$
 (81)

The observation equation is

.

 $y(t) = x_n(t) + n(t)$ . (82)

Therefore

$$F(t) = \begin{bmatrix} 0 & 0 \\ 1 & 0 \end{bmatrix}$$

$$(83)$$

$$f(t) = \begin{bmatrix} 0 & 1 \end{bmatrix}$$

$$(83)$$

$$Q(t) = 0$$
 (85)

$$R(t) = \Phi_r . \tag{86}$$

The a priori covariance is chosen as

$$\mathbf{P}(\mathbf{0}) = \begin{bmatrix} \mathbf{p}_{11}(\mathbf{0}) & \mathbf{0} \\ \mathbf{0} & \mathbf{p}_{22}(\mathbf{0}) \end{bmatrix}.$$
(B7)

Then the covariance P(t) of the optimal estimator is derived from Equation (11).

$$P(t) = \frac{1}{Z(t)} \begin{bmatrix} p_{11}(t) \langle 1 + p_{22}(0) t/\Phi_{r} \rangle, & p_{11}(0) t (1 + p_{22}(0) t/2\Phi_{r}) \\ p_{11}(0) t (1 + p_{22}(0) t/2\Phi_{r}), & p_{22}(0) + p_{11}(0) t^{2} + \frac{p_{11}(0)p_{22}(0) t^{3}}{3\Phi_{r}} \end{bmatrix}$$
(88)

where

$$\mathbf{Z}(t) = \mathbf{1} + \left[ \mathbf{p}_{22}(0) + \mathbf{p}_{11}(0)t^2/3 + \mathbf{p}_{11}(0)\mathbf{p}_{22}(0)t^2/12\Phi_{\mathbf{r}} \right] t/\Phi_{\mathbf{r}} .$$
 (89)

Suppose that the incorrect model actually used in the design of the suboptimal estimator is given by

 $e_{gall}(t) = e_{11}^0(t) + e_{11}^R(t)$ ,

$$P_{g}(0) = \begin{bmatrix} p_{g11}(0) & 0 \\ 0 & p_{g22}(0) \end{bmatrix}$$
  
$$= \begin{bmatrix} p_{11}(0) + e_{11}(0) & 0 \\ 0 & p_{22}(0) + e_{22}(0) \end{bmatrix}$$
(90)

 $\Phi_{\mathbf{r}\mathbf{c}} = \Phi_{\mathbf{r}} + \Delta \mathbf{R} \, .$ 

Then the diagonal components of  $E_{os}(t)$  are computed by Equation (36),

where

$$\mathbf{e}_{11}^{0}(t) = \frac{1}{Z_{0}(t)^{2}} \left[ \mathbf{e}_{11}(0) \left( 1 + \mathbf{p}_{022}(0) t/\Phi_{r0} \right)^{2} + \mathbf{e}_{22}(0) \mathbf{p}_{011}^{2}(0) t^{4}/4\Phi_{r0}^{2} \right]$$

(91)

(92)

(93)

A CONTRACTOR

$$\mathbf{e}_{11}^{R}(t) = \frac{p_{c_{11}}^{2}(0)\Delta Rt^{3}}{12\Phi_{ro}^{4}Z_{0}(t)^{2}} \left[ 4\Phi_{rc}^{2} + 2p_{J22}(0)\Phi_{ro}t + p_{c22}^{2}(0)t^{2} \right]$$
(94)

and

$$Z_{\sigma}(t) = 1 + \left[ \psi_{\sigma 22}(0) + \psi_{\sigma 11}(0) t^{2}/3 + \psi_{\sigma 11}(0) \psi_{\sigma 22}(0) t^{3}/12 \Phi_{rc} \right] t / \Phi_{rc} .$$
(95)

A1 80

$$\mathbf{e}_{0R22}(t) = \mathbf{e}_{22}^{0}(t) + \mathbf{e}_{22}^{R}(t), \qquad (96)$$

where

$$\mathbf{e}_{22}^{0}(t) = \frac{1}{Z_{0}(t)^{2}} \left[ \mathbf{e}_{11}(0) \left( 1 + \mathbf{p}_{022}(0) t/2 \Phi_{rc} \right)^{2} t^{2} + \mathbf{e}_{22}(0) \left( 1 - \mathbf{p}_{011}(0) t^{3}/6 \Phi_{rc} \right)^{2} \right]$$
(97)

$$\mathbf{e}_{22}^{\mathbf{R}}(t) = \frac{\Delta \mathbf{R}t}{\Phi_{rc}^{2} \mathbf{z}_{c}(t)^{2}} \left[ \left( \mathbf{p}_{\sigma 22}(0) + \mathbf{p}_{c 11}(0)t^{2}/2 + \mathbf{p}_{c 11}(0)\mathbf{p}_{c 22}(0)t^{2}/12\Phi_{rc}\right)^{2} + \frac{\mathbf{p}_{311}(0)^{2}t^{4}}{12} \left( 1 + \frac{\mathbf{p}_{c 22}(0)t}{2\Phi_{rc}} \right)^{2} \right].$$
(98)

In Figure 1 the optimal variance  $p_{22}(t)$  and actual variance  $p_{022}(t)$  of position of the spacecraft are depicted using  $p_{011}(0)$  as a parameter. The computation of the optimal variance  $p_{22}(t)$  is based on the true model.

The suboptimal filter is designed in such a way that

$$p_{22}(0) > p_{22}(0)$$

$$\Phi_{r_{i}} > \Phi_{r}$$

and using  $p_{0,1,1}(0)$  as a parameter.

The figure indicates that the variance of the suboptimal filter is quite sensitive to variation of  $p_{0,1}(0)$ , i.e., the incorrect initial speed variance. Case (a) is when an excessively large a priori uncertainty of speed was employed, i.e.,  $p_{0,11}(0) = 10p_{11}(0)$ . On the other hand, case (e) is when the a priori value taken was smaller than the true value, namely,  $p_{0,11}(0) = \frac{1}{p_{11}(0)}$ . For both cases significant overshoots of the variance are observed. This is because the gain  $K_0(1)$  was ill-conditioned for both extreme cases. In other words, sufficient weight had not been assigned to the information during the initial period so that the station did not track the spacecraft properly. Case (e) especially demonstrates how the estimator can behave poorly when an optimistic selection is made on the a priori covariance.

In Figure 2 the calculated variances  $p_{cli}(t)$  used in this case are plotted for the same parameters.

Figure 3 is the case when the variance of initial position  $p_{\sigma,\sigma,2}(0)$  is changed as a parameter. It can be observed that the suboptimal filter is not so sensitive to the initial uncertainty of position as to that of speed. However, case (d) reveals a degraded performance of the filter when a smaller value is picked up for the positional uncertainty than the true value.

$$p_{022}(0) = \frac{1}{10}p_{22}(0)$$

is Figure 4 the incorrect information  $\Phi_{ro}$  of the power spectral density of the observation noise is employed as a parameter.

The suboptimal filter behaves very poorly for  $\Phi_{ro}$ , which is either very large (case (a);  $\Phi_{ro} = 10 \ \Phi_r$ ) or very small (case (d);  $\Phi_{ro} = \frac{1}{2} \Phi_r$ ) compared to the true  $\Phi_r$ .

Figure 5 is one example of variances of speed of the spacecraft, corresponding to case (e) of Figure 3.

#### Excuple 2

An exponentially correlated signal is processed by a sequential continuous detector which is contaminated by additive white noise. Let x be the signal, then

$$i(t) = -\beta x(t) + w \tag{99}$$

$$y(t) = x(t) + n$$
, (100)

where w and n are uncorrelated white noise with zero mean, having power spectral density q and r respectively.

It is assumed that the variance  $\sigma_{\rm X}^2$  of signal is known, but its correlation time  $\tau$ (= 1/ $\beta$ ) is not exactly known a priori. Hence an approximate value  $\tau_{\rm g}$ (= 1/ $\beta_{\rm g}$ ) is employed in designing the filter.

Thus

$$\beta_{\mu} = \beta + \Delta \beta$$
.

Also it is assumed that a sufficient time is assigned for tracking compared with the correlation time of the signal, so that only steady-state solutions of Riccati equations which are the limiting values as  $t \to \infty$  are investigated.

The purpose of the subsequent analysis is to find the effect on the filter performance of deviation of the assigned correlation time from its true value. The Riccuti equation of the optimal variance p(t) associated with the estimator  $x^*(t)$  becomes

$$\dot{p}(t) = -2\beta p(t) - \frac{p^2(t)}{r} + q.$$
 (102)

At the steady state  $\dot{p} \rightarrow 0$  ; hence the steady state solution  $|p_{g_{\rm s}}|$  is computed

$$\mathbf{p}_{\mu} = -\beta \mathbf{r} + (\beta^2 \mathbf{r}^2 + \mathbf{r} \mathbf{q})^{1/2}$$
(103)

where

$$q = 2\beta \sigma_r^2 , \qquad (104)$$

On the other hand, the actual filter design is based on incorrect knowledge of the correlation time (or  $\beta_{a}$ ). Thus the steady-state value  $p_{ab}$  of the calculated variance is given by

$$\mathbf{p}_{cn} = -\beta_n \mathbf{r} + (\beta_n^2 \mathbf{r}^2 + \mathbf{r} \mathbf{q}_n)^{1/2} \tag{105}$$

$$\mathbf{q}_{\mathbf{g}} = 2\beta_{\mathbf{g}}\sigma_{\mathbf{x}}^{\mathbf{g}}, \qquad (106)$$

The actual variance  $p_{g}(t)$  associated with the suboptimal estimator  $x_{g}^{\bullet}(t)$  which is designed using this  $p_{g}(t)$  is determined from the following three differential equations:

$$\dot{\mathbf{p}}_{a}(t) = -2[\beta_{c} + p_{c}(t)/r]\mathbf{v}_{a}(t) - 2\Delta\beta\lambda(t) + p_{c}^{2}(t)/r + q \qquad (107)$$

$$\dot{\lambda}(t) = -\left[\beta + \beta_{0} + p_{0}(t)/r\right]\lambda(t) - \Delta\beta p_{0}(t) - q$$
(108)

$$\dot{p}_{x}(t) = -2\beta p_{x}(t) + q.$$
 (109)

The steady-state solutions of these equations are computed and, after some manipulation, the steady state value  $p_{as}$  is found as

$$\mathbf{p}_{u\delta} = \mathbf{p}_{c\delta} - \frac{\mathbf{r} \mathbf{p}_{c\delta} \mathbf{r}_{x}^{u} \Delta \beta}{(\mathbf{p}_{c\delta} + \mathbf{r} \beta_{c}) [\mathbf{p}_{c\delta} + \mathbf{r} (\beta + \beta_{c})]} . \tag{110}$$

This  $p_{as}$  becomes a minimum at  $\Delta\beta = 0$  and it is equal to the optimal value  $p_{a}$ . It is noticed, however, that as the assigned correlation time becomes where  $p_{as}$  approaches  $\sigma_{x}^{2}$ .

$$\beta_{0} \rightarrow \infty$$
,  $p_{CR} \rightarrow \sigma_{\rm X}^{2}$ ,  $p_{\rm RR} \rightarrow \sigma_{\rm X}^{2}$ .

Also as the assigned correlation time becomes longer, again  $p_{ab}$  approaches  $\sigma_{a}^{2}$  .

$$\beta_{\rm c} \rightarrow 0$$
,  $p_{\rm cs} \rightarrow 0$ ,  $p_{\rm as} \rightarrow \sigma_{\rm x}^2$ .

In both extreme cases, the filter is so ill-conditioned that the estimation of the signal is not performed at all. Hence any deviation of the assigned correlation time from its true value in either direction will result in an increase of the actual variance. In Figure 6,  $p_{\rm RS}$  is plotted against  $\Delta\beta/\beta$  for unit value of r and  $\sigma_{\rm g}^2$  respectively.

The ratio between the maximum deviation of  $p_{ab}$  and the optimal value  $p_{a}$  is computed as

 $\alpha = \frac{\sigma_x^2 - p_x}{p_x} = \frac{1}{2} \left( -1 + \left[ 1 + 2 \left( \frac{\sigma_x^2}{\beta_r} \right) \right]^{1/2} \right). \tag{111}$ 

This  $\alpha$  provides the range of the probable deviations of the suboptimal filter performance from that of the optimal one.

(101)

### 8. CONCLUSION

The algorithms for evaluating the effect of ervors due to moduling errors in the Kalman filter have been presented in this chapter for both continuous and discrete systems

Also, the error bound of the Kalman filter has been studied when the incorrect a priori statistics on initial conditions as well as on roise models are employed. The conservative design criterion expressed in Theorems 1 and 4 guarantees that the suboptimal filter satisfying it remains within the specified range over the estimation period. Also, the formulae of Equations (28) - (31) for continuous systems and of Equations (62) - (64) for discrete systems supply the necessary information to evaluate the effect of errors qualitatively for parametric studies. Such parametric investigations are very important to find out to what extent conservative assignment of a priori statistics and noise models can be made, because large covariances of initial coordinates and noises tend to increase the covariance of estimates and eventually to slow down its convergence, thus deteriorating the sensitivity of the filter.

The first example of this chapter well demonstrates the importance of preflight parametric studies when estimations are to be carried out in rather short periods. An optimistic selection of the *a priori* statistics (smaller values of  $P_c(0)$ ,  $Q_0$ , and  $R_c$  than true values) is especially dangerous because it prevents the estimator from having a proper gain K(t) during the initial period of estimation (case (e) of Figure 1, case (d) of Figure 3) and case (d) of Figure 4). On the other hand, it has been observed in examples that an excessively conservative obtice may also be harmful, because it frequently results in a large offset of suboptimal covariances from the optimal ones at the end of the estimation period (case (a) of Figures 1, 3, and 4).

The second example is intended to study the influence of correlation time on the suboptimal filter performance. This is, again, an important problem in space missions because it is often difficult to obtain the exact values of correlation time of stochastic variables such as fluctuations of solar pressure or of low-thrust-engine power. Therefore it is essential to carry out a sensitivity study of the filter as in this exemple.

### ACKNOWLEDGMENT

This chapter represents one phase of research carried out at the Jet Propulsion Laboratory, Californ's Institute of Technology, under NASA Contract NAS 7-100.

### REFERENCES

1. Kalman, R.E.	A New Approach to Linear Filtering and Prediction Theory. Journal of Basic Engineering, Transactions of the American Society of Mechanical Engineers, Vol.82D, March 1960, pp.35-45.	
2. Kalman, R.E. Bucy, R.S.	New Results in Linear Filtering and Prediction Theory. Journal of Basic Engineering. Transactions of the American Society of Mechanical Engineers, Vol.83D, March 1962, pp.95-108.	K.
3. Soong, T.E.	On a priori Statistics in Minimum Variance Estimation Problems. Journal of Basic Engineering, Transactions of the American Society of Mechanical Engineers, Vol.87D, March 1965, pp. 109-112.	
6. Nishimura, T.	On the a priori Information in Sequential Estimation Problems. Institute of Electrical and Electronic Engineers, Transactions on Automatic Control, Vol.AC-11, April 1966, pp. 197-204.	
5.	Correction to and Extension of "On the a priori Information in Sequential Estimation Problems". Institute of Electrical and Electronic Engineers, Transactions on Automatic Control (Correspondence), Vol.AC-12, February 1967, p.123.	
6, Heffes, J.	The Effect of Erroneous Models on the Kalman Filter Response. Institute of Electrical and Electronic Engineers, Transactions on Automatic Control, Vol.AC-11, July 1966, yp. 541-543.	
7. Mishimurs, T.	Error Bounds of Continuous Kalman Filters and the Application to Orbit Determination Problems. Institute of Electrical and Electronic Engineers, Transactions on Automatic Control, Vol.AC-12, June 1987, pp.268-275.	
8. Griffin, R.E. Sage, A.P.	Large and Small Scale Sensitivity Analysis of Optimum Estimation Algorithms. Institute of Electrical and Electronic Engineers, Transactions on Automatic Control, Vol.AC-13, August 1968, pp.320-329.	















9. Sawarage, Y. Katayama, T.

10. Nishimura, T.

On the Performance Loss in Sequential Estimation and the Related Problem of Duality. Transactions of the Society of Instrument and Control Engineers of Japan, Vol.4, No.3, September 1968, pp. 248-265.



Sensitivity Analysis of Noise Correlation Time. Space Program Summary No.37-57, Vol.II, April 1969, Jet Propulsion Laboratory, California Institute of Technology.









Set O PLINE D



Ċ,





Fig. 2 Computed variance of position of a spacecraft with range data for various initial suboptimal speed variances corresponding to Figure 1

ill becreater 1" A C MANNA







Fig. 4 Actual and optimal variances of position of a spacecraft with range data for various suboptizal noise opectral densities





Fig. 5 Computed, actual, and optimal variances of speed of a spacecraft with range data corresponding with case (a) of Figure 1



Fig.6 Actual variance of suboptimal filter

1. 1 1 100 ALC: Y' 

Karuna Dukosa Pina Diski ini

### CHAPTER 5 - SUBOPTIMAL KALNAN FILTER TECHNIQUES

Å,

1

by

.

A.R. Stubberud\* and D.A. Wismer\*

\*Bohool of Engineering University of California, Irvine, California, USA

<sup>†</sup>School of Engineering and Applied Science University of California, Lus Angeles, USA

# PRECEDING PAGE BLANK

A Cartan A Chan

105

### CHAPTER 5 - SUBOPTIMAL KALMAN FILTER TECHNIQUES

A.R. Stubberud and D.A. Wismer

### 1. INTRODUCTION

The application of Kalman filter theory<sup>1</sup> requires the definition of a linear mathematical model describing the system for which the application is intended. In many cases a highly complex model must be used to accurately describe the system. Usually in these cases only a few of the state variables are of primary interest, the rest morely enhancing the description of the system. Since the computational burden associated with the implementation of a Kalman filter increases significantly with the dimension of the system model, designers are motivated to seek simplifications to the required filtering equations which do not result in a large performance degradation in the estimation of state vector components of principal interest. The resulting filters are called suboptimal filters.

One successful technique investigated by Joseph<sup>2</sup>, Meditch<sup>3</sup> and Pen<sup>2</sup>coest<sup>4</sup> for simplifying filter computations involves partitioning the state vector into strongly coupled subsystems, employing Kalman's optimal filtering algorithm for the state vectors of these lower-dimensional systems and reconstructing an estimate of the original system from the lower-dimensional estimates. Another approach, introduced by Acki<sup>4</sup> and Huddle<sup>4</sup> and related to observer theory of Luenberger<sup>7</sup>, employs the concept of linear aggregation of states to achieve a concise representation of data, thereby reducing the dimension of the estimator. Important work on the effects on the Kalman filter performance of error in the knowledge of the covariance matrices describing the initial conditions of the system state vector, and the white observation and disturbance noise vectors has been contributed by Meffes<sup>8</sup> and Nishimura<sup>13</sup>.

The techniques for generating suboptimal filters can be roughly divided into two classes. In the first, the number of equations defining the filter is the same as for the optimal filter, but the number of equations necessary for computing the filter gain is reduced by partitioning the system. In the second, the number of state variables defining the filter is reduced by aggregation. This also reduces the dimensionality of the covariance matrix and hence reduces the computation required to generate the filter gain.

These two classes of suboptimal techniques are discussed in the next two sections.

### 2. DISCRETE TIME SUBOPTIMAL FILTERS EMPLOYING PEDUCED FILTER GAIN COMPUTATION

The computational burden associated with a minimum variance, unbiased linear filter is strongly dependent on the number of simultaneous difference equations which define the filter and the filter gain. If the state of an n-dimensional system is being estimated, a total of u difference equations define the filter and a total of u(u + 1)/2 difference equations must be solved to determine the optimal filter pain. In this section a particular class of suboptimal filters is discussed. The property of these filters is that the order of the equations defining the filter is the same as the order of the optimal filter. Mopefully, the total number of equations can be reduced considerably without the performance of the filter being significantly different from that of the optimal filter.

#### 2,1 Problem Formulation

Consider a dynamic system defined by the linear stochastic difference equation

$$\mathbf{x}_{\mathbf{k},1} = \phi_{\mathbf{k}}\mathbf{x}_{\mathbf{k}} + \mathbf{u}_{\mathbf{k}}, \quad \mathbf{k} = 0, \ 1, \ 2, \dots,$$
 (2.1)

where  $x_k$  is the n-dimensional "tate vector"

- $u_k$  is an n-dimensional zero mean white noise sequence with covariance matrix  $Q_k \delta_{ik}$
- $\phi_{\mathbf{k}}$  is the n x n dimensional state transition matrix

 $\pi_a$ , the fuitial condition, is a zero mean random variable, independent of  $u_k$  , with covariance matrix  $P^*$  .

The system is assumed to have the observable output

$$\mathbf{y}_{\mathbf{k}} = \mathbf{H}_{\mathbf{k}}\mathbf{x}_{\mathbf{k}} + \mathbf{v}_{\mathbf{k}} \,, \tag{2.2}$$

where yk is the m-dimensional observation vector

PRECEDING PAGE BLANK

107

I LIP PRIMA

THE NAVING AN

Differ ting

 $\mathbf{v}_k$  is an m-dimensional zero mean white noise sequence, independent of both  $\mathbf{u}_k$  and  $\mathbf{x}_n$ , with covariance matrix  $R_k \delta_{ik}$ 

M<sub>u</sub> is an m×n matrix defining m linearly independent observable combinations of the state variables. The problem is to generate a linear estimator for the state  $x_k$  which makes use of the observations  $y_j$ , j=0, 1....k . to generate the estimate.

Consider the class of linear unbiased filters defined by the stochastic difference equation

$$\mathbf{\hat{x}}_{k+1} = \bar{\mathbf{x}}_{k+1} + K_{k+1} [\mathbf{y}_{k+1} - M_{k+1} \bar{\mathbf{x}}_{k+1}], \qquad (2.3)$$

where  $\tilde{x}_{k+1}$  is defined by

$$\mathbf{\bar{x}}_{k+1} = \phi_k \mathbf{x}_k , \qquad (2.4)$$

with initial condition  $\hat{x}_0 = 0$ . Here  $\hat{x}_{k+1}$  is an unbiased estimate of  $x_{k+1}$  and  $K_{k+1}$  is an arbitrary n < m gain matrix. The estimation error associated with this estimator is defined as

$$\mathbf{x} = \mathbf{x}_{\mathbf{k}} - \mathbf{x}_{\mathbf{k}} \tag{2.5}$$

and the covariance matrix of this error is defined by

$$\mathbf{\tilde{P}}_{\mathbf{k}} \approx \mathbf{E}[\hat{\mathbf{c}}_{\mathbf{k}}\hat{\mathbf{c}}_{\mathbf{k}}^{\dagger}] . \tag{2.6}$$

This matrix (specifically the elements on the main diagonal) is a measure of the effectiveness of the filter and it satisfies the matrix equations

$$\mathbf{P}_{\mathbf{k}+\mathbf{i}} = \boldsymbol{\phi}_{\mathbf{k}} \mathbf{\bar{P}}_{\mathbf{k}} \boldsymbol{\phi}_{\mathbf{k}}^{\mathrm{T}} + \mathbf{Q}_{\mathbf{k}} \tag{2.7}$$

$$\mathbf{P}_{k} = [\mathbf{I} - \mathbf{K}_{k}\mathbf{M}_{k}]\mathbf{P}_{k}[\mathbf{I} - \mathbf{K}_{k}\mathbf{M}_{k}]^{T} + \mathbf{K}_{k}\mathbf{R}_{k}\mathbf{K}_{k}^{T}, \qquad (2.8)$$

with the initial condition  $\vec{P}_n = P^*$  , where

$$\mathbf{P}_{\mathbf{k}} = \mathbb{E}[(\mathbf{x}_{\mathbf{k}} - \tilde{\mathbf{x}}_{\mathbf{k}})(\mathbf{x}_{\mathbf{k}} - \tilde{\mathbf{y}}_{\mathbf{k}})^{\mathrm{T}}] = \mathbb{E}[\tilde{\mathbf{e}}_{\mathbf{k}} \tilde{\mathbf{e}}_{\mathbf{k}}^{\mathrm{T}}]. \qquad (2.9)$$

Note that, since  $\bar{x}_k$  is the best estimate of  $x_k$  based on  $y_0$ ,  $y_1, \dots, y_{k-1}$ , it is generally a poorer estimate than  $x_k$ , which includes  $y_k$  also.  $\bar{P}_k$  is thus the true measure of the filter performance and  $P_k$  is simply an intermediate matrix which it is convenient to carry along. Equations (2.3) and (2.4) which define the class of unbiased linear filters are valid for any gain  $K_{k+1}$ , and Equations (2.7) and (2.8) define the variances of the errors for any filter having this form. Thus the effectiveness of any suboptimal gain can be evaluated from these general variance equations.

A special member of this class of filters results when the gain is chosen as

$$\mathbf{K}_{k+1} = \mathbf{P}_{k+1} \mathbf{M}_{k+1}^{T} [\mathbf{M}_{k+1} \mathbf{P}_{k+1} \mathbf{M}_{k+1}^{T} + \mathbf{R}_{k+1}]^{-1} .$$
(2.10)

The filter is then a minimum variance unblased filter. In this case Equation (2.8) reduces to

$$\mathbf{P}_{\mathbf{k}} = \left[\mathbf{I} - \mathbf{K}_{\mathbf{k}} \mathbf{M}_{\mathbf{k}}\right] \mathbf{P}_{\mathbf{k}}, \qquad (2.8a)$$

The computational burden of this optimal filter includes the solution of the n difference equations defined by Equations (2.3) and (2.4) plus the solution of the n(n+1)/2 difference equations defined by Equations (2.7) and (2.8) to calculate the optimal gain  $K_k$ . (The symmetry of the  $n^2$  difference equations requires that only n(n+1)/2 need to be solved.)

In the following sub-sections some methods for choosing suboptimal gains  $K_{k+1}$  are discussed.

#### 2.2 Constant Gains

In terms of a reduction in computation, an obvious choice for K, is a constant; that is,

$$K_k = K$$
 for all k.

Probably, the most common choice of a constant gain is the steady-state value (if one exists) of the optimal gain matrix. Consider the case of a time-invariant system, that is, the case where

$$\phi_{\mathbf{k}} = \phi$$
,  $\mathbf{M}_{\mathbf{k}} = \mathbf{M}$ ,  $\mathbf{Q}_{\mathbf{k}} = \mathbf{Q}$ ,  $\mathbf{R}_{\mathbf{k}} = \mathbf{R}$  for all  $\mathbf{k}$ .

The steady-state gain for this system (if it exists) is determined from Equations (2.7), (2.8), and (2.10) by setting

 $P_k = P_i$ ,  $P_k = P_i$ ,  $K_k = K$  for all k.

The result is a set of algebrian equations

P =	: ¢₽¢	5 <sup>4</sup> + Q		(2. 11)
			 _	

 $\vec{P} = [I - KM]P[I - KM]^{T} + KRK^{T}$ (2.12)(2.13)

 $\mathbf{K} = \mathbf{P}\mathbf{N}^{\mathsf{T}}[\mathbf{M}\mathbf{P}\mathbf{N}^{\mathsf{T}} + \mathbf{R}]^{-1} .$ 

The error variance equation for this case becomes

 $\mathbf{P}_{\mathbf{k}+1} = [\mathbf{I} - \mathbf{K}\mathbf{M}]\phi \mathbf{P}_{\mathbf{k}}\phi^{\mathsf{T}}[\mathbf{I} - \mathbf{K}\mathbf{M}]^{\mathsf{T}} + [\mathbf{I} - \mathbf{K}\mathbf{M}]\mathbf{Q}[\mathbf{I} - \mathbf{K}\mathbf{M}]^{\mathsf{T}} + \mathbf{K}\mathbf{R}\mathbf{K}^{\mathsf{T}}, \qquad (2.14)$ 

It should be noted that the filter generated in this way is the classical Wiener filter for discrete time systems, which is suboptimal until steady-state conditions are reached.

### 2.3 A Suboptimal Technique Based On Partitioning

In this section a method is demonstrated which is based on partitioning of the system state vector into several sub-state vectors. The states of the subsystems thus generated are estimated individually and the resulting estimates are combined linearly to generate the total estimate. Since the number of variance equations varies approximately as the square of the dimension of the system state the number of variance equations used to compute the optimal gains is significantly reduced.

Consider a dynamic system defined by Equation (2.1). Joseph<sup>2</sup> described a general method for partitioning the state vector of the system into a set of i subsystem state vectors. The i subsystem state vectors are defined by

$$\hat{x}_{k}^{j} = D_{j} x_{k}^{j}, \quad j = 1, 2, ..., 1$$
, (2.15)

where  $D_j$  is an  $n_j \times n$  matrix of rank  $n_j$ . The state vector  $x_k$  is recovered from the subsystem state vectors by the relationship

$$\mathbf{x}_{\mathbf{k}} = \sum_{j=1}^{L} A_{j} \sum_{\mathbf{k}}^{j} = \sum_{j=1}^{L} A_{j} D_{j} \mathbf{x}_{\mathbf{k}}$$
 (2.16)

where A<sub>i</sub> is an n×n<sub>i</sub> matrix.

Suppose the state vector  $\mathbf{x}_k$  has been partitioned into i sub-state vectors

5.	E E.	with respective dimensions
n, ,	n <sub>2</sub> , n <sub>1</sub>	and suppose that individual estimates
21 5 k ,	ŝ ŝ.	have been generated from the observations
<b>y</b> 1, 1	= 0, 1,, k.	

Then the estimate of the state vector is given by

$$\hat{x}_{k} = \sum_{j=1}^{k} A_{j} \frac{c_{j}}{k}$$
 (2.17)

This estimate is then propagated by the state transition matrix, thus forming

π

 $\widetilde{\mathbf{x}}_{\mathbf{k}+1} = \boldsymbol{\phi}_{\mathbf{k}} \mathbf{\hat{x}}_{\mathbf{k}}$ 

The propagated sub-state estimate of  $\mathcal{J}_{k+1}^{j}$  is then formed by

$$\tilde{\mathbf{x}}_{k+1} = \mathbf{D}_{j} \tilde{\mathbf{x}}_{k+1} . \tag{2.19}$$

Consider now a partitioning of the observation vector defined in Equation (2.2). The vector  $\mathbf{y}_{\mathbf{k}}$  is partitioned into i observation vectors

$$\gamma_{\mathbf{k}}^{\mathbf{j}} = \mathbf{E}_{\mathbf{j}}\mathbf{y}_{\mathbf{k}}, \quad \mathbf{j} = 1, 2, \dots, \mathbf{i},$$
 (2.20)

where E<sub>j</sub> is an u<sub>j</sub> × n matrix. The estimate of  $\zeta_{k+1}^{j}$  is now updated by the equation

$$\hat{S}_{k+1}^{j} = \hat{S}_{k+1}^{j} + \kappa_{k+1}^{j} [\eta_{k+1}^{j} - \tilde{\eta}_{k+1}^{j}], \qquad (2.21)$$

where

$$\begin{aligned} & +1 - \overline{\eta}_{k+1}^{j} &= \mathbb{E}_{j} \mathbf{y}_{k+1} - \mathbb{E}_{j} \overline{\mathbf{y}}_{k+1} \\ & = \mathbb{E}_{j} \mathbf{y}_{k+1} - \mathbb{E}_{j} \mathbf{w}_{k+1} \widetilde{\mathbf{x}}_{k+1} \\ & = \mathbb{E}_{j} \left[ \mathbf{y}_{k+1} - \mathbf{w}_{k+1} \widetilde{\mathbf{x}}_{k+1} \right]. \end{aligned}$$

$$(2.22)$$
The updated state estimate is then given by

$$\begin{aligned} \mathbf{x}_{k+1} &= \sum_{j=1}^{L} A_j \widehat{\mathbf{x}}_{k+1}^{j} \\ &= \sum_{j=1}^{L} A_j \widehat{\mathbf{x}}_{k+1}^{j} + \sum_{j=1}^{L} A_j \mathbf{K}_j [\gamma_{k+1}^j - \tilde{\mathbf{n}}_{k+1}^j] \\ &= \widetilde{\mathbf{x}}_{k+1} + \sum_{j=1}^{L} A_j \mathbf{K}_{k+1}^j \mathbf{E}_j [\mathbf{y}_{k+1} - \mathbf{M}_{k+1} \widetilde{\mathbf{x}}_{k+1}] . \end{aligned}$$

$$(2.23)$$

Comparing Equations (2.3) and (2.23), it is apparent that this represents a member of the class of unbiased filters defined by Equation (2.3). Here the gain is

$$K_{k+1} = \sum_{j=1}^{L} A_j K_{k+1}^{j} E_j . \qquad (2.24)$$

Note that  $D_{j}$  ,  $A_{j}$  , and  $E_{j}$  can all be made a function of the time index k so that

$$K_{k+1} = \sum_{j=1}^{k} A_{k+1}^{j} K_{k+1}^{j} E_{k+1}^{j} . \qquad (2.24n)$$

This generalization allows the subaystems to be changed if the system has several modes of operation which make it mandatory, convenient, or more effective to change the system partitions for the several modes of operation.

Equation (2.23) indicates that the individual estimates of the subsystem states need not be generated. It is seen that, for given  $A_j$  and  $E_j$ , only the  $K_{k+1}^i$  need to be generated and then  $K_{k+1}$  can be generated by Equation (2.24). This  $K_{k+1}$  can be used directly in Equation (2.23) to generate the estimate  $\hat{x}_{k+1}$ .

The next step in this suboptimal method is the generation of the set of equations used to compute the gains  $k_{j}^{i}$ . First, an error equation is generated. Subtracting Equation (2.4) from Equation (2.1) generates the error equation

$$\tilde{\epsilon}_{k+1} = \phi_k \hat{\epsilon}_k + u_k \,. \tag{2.25}$$

It follows immediately that

$$\mathcal{R}_{k+1}^{j} = \mathcal{L}_{k+1}^{j} - \mathcal{L}_{k+1}^{j} = D_{j} \mathbf{x}_{k+1} - D_{j} \mathbf{x}_{k+1}$$

$$= D_j \tilde{\epsilon}_{k+1} = D_j \phi_k \tilde{\epsilon}_k + D_j u_k . \qquad (2.26)$$

$$\hat{\epsilon}_{j}^{j} = D_{j}\hat{\epsilon}_{k} . \qquad (2.27)$$

and therefore  $\hat{\epsilon}_k = D_j^* \hat{\epsilon}_k^j + z$ , (2.28)

where z is a vector in the null space of D<sub>j</sub> and D<sub>j</sub><sup>\*</sup> is the pseudo-inverse of  $v_j$ .

If D<sub>i</sub> has maximal rank, then

$$D_{i}^{\dagger} = D_{i}^{T} (D_{i} D_{i}^{T})^{-1}$$
(2.29)

is a suitable pseudo-inverse.

Substituting Equation (2.28) into Equation (2.26) yields

 $\tilde{\epsilon}_{k+1}^{j} = D_{j}\phi_{k}D_{j}^{+}\hat{\epsilon}_{k}^{j} + D_{j}\phi_{k}z + D_{j}u_{k}$ 

$$= \phi_{k}^{j} \hat{\epsilon}_{k}^{j} + D_{j} \phi_{k} z + D_{j} u_{k} , \qquad (2.30)$$

$$\phi_{1}^{\dagger} = D_{1}\phi_{2}D_{1}^{\dagger}$$
 (2.31)

At this point an approximation is made. It is assumed that the subsystems are uncoupled. This allows the gains for each subsystem to be calculated individually. The absence of coupling between the subsystems implies that  $\mathbf{z} = \mathbf{0}$ . In general this is not true and the resulting gains are suboptimal. Under this assumption Equation (2.30) becomes

$$\epsilon_{k+1}^{2} = \phi_{k}^{2} \epsilon_{k}^{2} + D_{j} u_{k} . \qquad (2.32)$$

A variance equation is now generated from Equation (2.32) as

$$\mathbf{E}[\hat{\mathbf{z}}_{k+1}^{j}(\hat{\mathbf{z}}_{k+1}^{j})^{T}] = \phi_{k}^{j} \mathbf{E}[\hat{\mathbf{z}}_{k}^{j}(\hat{\mathbf{z}}_{k}^{j})^{T}] (\phi_{k}^{j})^{T} + D_{j} \mathbf{E}[u_{k} u_{k}^{T}] D_{j}^{T}. \qquad (2.33)$$

spere

Now





Charles Charles and Solar State



where

$$\mathbf{q}_{\mathbf{k}}^{\mathbf{j}} = \mathbf{D}_{\mathbf{j}} \mathbf{Q}_{\mathbf{k}} \mathbf{D}_{\mathbf{j}}^{\mathbf{j}} \,. \tag{2.37}$$

111

Now consider the error equation

$$\epsilon_{L,1}^{\dagger} = \epsilon_{L,1}^{\dagger} - \epsilon_{L,1}^{\dagger}$$
 (2.38)

Substituting from Equation (2.21)

$$\begin{aligned} \vec{\xi}_{k+1}^{I} &= \xi_{k+1}^{J} - \xi_{k+1}^{J} - \kappa_{k+1}^{J} [\vec{\gamma}_{k+1}^{J} - \vec{\gamma}_{k+1}^{J}] \\ &= \xi_{k+1}^{J} - \kappa_{k+1}^{J} [\vec{E}_{j} \vec{y}_{k+1} - \vec{E}_{j} \vec{y}_{k+1}] \\ &= \xi_{k+1}^{J} - \kappa_{k+1}^{J} [\vec{E}_{j} \vec{M}_{k+1} \vec{x}_{k+1} + \vec{E}_{j} \vec{v}_{k+1} - \vec{E}_{j} \vec{M}_{k+1} \vec{x}_{k+1}] \\ &= \vec{z}_{k+1}^{J} - \kappa_{k+1}^{J} [\vec{E}_{j} \vec{M}_{k+1} \vec{x}_{k+1} + \vec{E}_{i} \vec{v}_{k+1}] . \end{aligned}$$

$$(2.39)$$

Since

$$\mathbf{x}_{k+1} = \mathbf{D}_{j} \mathbf{x}_{k+1},$$
 (2.40)

then

$$\vec{z}_{k+1} = D_j^* \vec{z}_{k+1}^* + \pi_1^*, \qquad (2.41)$$

where  $x_i$  is in the null space of  $D_j$ . Again the coupling is ignored by assuming  $x_i = 0$ . Equation (2.39) becones ••

$$\mathbf{e}_{k+1}^{j} = [\mathbf{I} - \mathbf{R}_{k+1}^{j} \mathbf{E}_{j} \mathbf{N}_{k+1} \mathbf{D}_{j}^{j}] \mathbf{e}_{k+1}^{j} - \mathbf{R}_{k+1}^{j} \mathbf{E}_{j} \mathbf{v}_{k+1}$$
(2.42)

The corresponding variance equation is given by

 $\mathbb{E}[\hat{\boldsymbol{e}}_{k+1}^{j}(\hat{\boldsymbol{e}}_{k+1}^{j})^{T}] = \mathbb{P}_{k+1}^{j}$ 

$$= [\mathbf{I} - \mathbf{K}_{k+1}^{j} \mathbf{E}_{j} \mathbf{M}_{k+1} \mathbf{D}_{j}^{\dagger}] \mathbf{P}_{k+1}^{j} [\mathbf{I} - \mathbf{K}_{k+1}^{j} \mathbf{E}_{j} \mathbf{M}_{k+1} \mathbf{D}_{j}^{\dagger}]^{T} + \mathbf{K}_{k+1}^{j} \mathbf{E}_{j} \mathbf{R}_{k+1} \mathbf{E}_{j}^{T} (\mathbf{K}_{k+1}^{j})^{T}, \qquad (2, 43)$$

with initial condition  $\mathbb{P}_0^j = D_j \mathbb{P}_0 D_j^T$  when  $\mathbb{P}_0 = \mathbb{E}[x_0 x_0^T]$ . The gain which minimizes  $\mathbb{P}_{k+1}^j$  is then given by

$$K_{k+1}^{j} = P_{k+1}^{j} (M_{k+1}^{j})^{T} (M_{k+1}^{j} P_{k+1}^{j} (M_{k+1}^{j})^{T} + R_{k+1}^{j})^{-1} , \qquad (2,44)$$

where

$$\mathbf{M}_{k+1}^{j} = \mathbf{E}_{j}\mathbf{M}_{k+1}\mathbf{D}_{j}^{\dagger} \tag{2.45}$$

$$\mathbf{R}_{\mathbf{k}+1}^{\mathbf{j}} = \mathbf{E}_{\mathbf{j}}\mathbf{R}_{\mathbf{k}+1}\mathbf{E}_{\mathbf{j}}^{\mathsf{T}} . \tag{2.46}$$

The basic equations for generating  $K_{k+1}^j$  are Equations (2.36), (2.43) and (2.44). Having calculated  $K_{k+1}^j$ , j = 1, 2, ..., i, the total gain is calculated from Equation (2.24) and the estimate of  $x_k$  is given by the difference equation (2.23).

As a simple example which illustrates only the computational advantages of this method, consider the case of a system state with nice state variables (n = 9) and suppose it has been partitioned into three subsystems, each with three state variables. The optimal filter requires the solution of a total of (9)(10)/2 = 45 simultaneous variance equations. The suboptimal filter requires the solution of 3[(3)(4)/2] = 18 simultaneous equations.

Obviously there is no unique choice for the partition of a specific system. Pentecost' discusses a rationale for partitioning a system.

One general rule for partitioning a system is that each state variable should be assigned, through a choice of the matrices D<sub>j</sub> , to a subsystem which assures it is strongly corrulated to an observation. Heuristically, using a suboptimal filter is equivalent to making the system less observable. The correlation between a state and the observations is a measure of the observability of the state and should be kept as large as possible when choosing a partition. Several observations will generally be associated with a single subsystem.

A second partitioning rule is that strongly correlated states should be assigned to the same subsystem since, in a general partition, the correlation between states in different subsystems is not taken into account. Any partition should be chosen to minimize this loss of correlation. One measure of the correlations between state variables can be obtained from the error covariance  $\tilde{P}_k$  for the optimal filter. The correlation matrix  $C_k$  with elements





٩,

E,

ŝ

$$\mathbf{c}_{1j}^{\mathbf{k}} = \frac{\mathbf{p}_{1j}^{\mathbf{k}}}{\mathbf{v}\mathbf{p}_{1j}^{\mathbf{k}}\mathbf{p}_{1j}^{\mathbf{k}}} \, .$$

where  $p_{1j}^k$  is an element of  $\tilde{P}_k$ . The terms  $c_{1j}^k$ ,  $i \neq j$ , being a measure of state variable correlation, can be used to group state variables into the same subsystem. Note that it is possible that one state variable will belong to more than one subsystem.

In order to properly design a suboptimal filter as described in this section, it is appropriate to perform a full scale simulation of the optimal filter covariance equations and suboptimal filter covariance equations. Iterating on the choice of subsystems should lead to a suitable filter design. In most aerospace applications extensive simulations are usually the rule; therefore this technique is well suited to these applications.

#### 3. SUBOPTIMAL FILTERS EMPLOYING STATE REDUCTION

An alternate approach to that cited in Section 2 assumes that unavoidable errors will occur in the formulation of a mathematical model of the system. With this premise, a primary system model is formulated including only those state vector components of dominant special interest to the designer. The other components and their interactions are considered as part of a secondary subsystem. When this is done, the state vector of interest has reduced dimension and allows a reduction in the computational burden required for implementing a Kalman filter based on the primary system model. In many practical problems<sup>9-11</sup> the structure of the plant is such that the state vector components which are ignored by such a procedure are independent sources of correlated disturbance or additive observation noise. Thus, the secondary system may be considered to be a model of colored noise in addition to the usual white noise sources assumed in the primary subsystem. This section considers the performance desardation incurred by basing the filter design on the primary system only. The matrix differential equation governing the error in the covariance matrix of the primary system is expressed explicitly in terms of the meglected state components, i.e. the secondary system. The expression for performance loss incurred with any proposed filter allows the designer to determine directly the utility of simple models in filter synthesis. Error

#### 3.1 Problem Formulation

Consider a dynamic system which can be represented by two linear stochastic difference equations of the form

$$\mathbf{x}_{k+1}^{1} = \phi_{k} \mathbf{x}_{k}^{1} + C_{k} \mathbf{x}_{k}^{2} + \mathbf{u}_{k}^{1}$$
(3.1)

$$\mathbf{x}_{k+1}^2 = \Lambda_k \mathbf{x}_k^2 + \mathbf{u}_k^2 , \qquad (3.2)$$

where

- $x_k^i$  is an  $n_i$ -dimensional state vector composed of those components of a refined linear mathematical model which are dominant or of special interest (the state of the primary subsystem).
- $\mathbf{x}_k^2$  is an  $n_2$ -dimensional vector composed of those system components which appear as time-correlated disturbances or additive observation noise in a refined model (the state of the secondary subsystem)
- $u_{k}^{1}$  is an  $n_{1}$ -dimensional zero-mean white noise sequence with covariance matrix  $Q_{k}^{1}\delta_{1k}$

 $u_k^2$  is an  $n_2$ -dimensional zero-mean white noise sequence with covariance matrix  $Q_k^2 \delta_{jk}$ 

Jk.

- $\phi_{\mathbf{k}}$  is the state transition matrix for the primary subsystem, an  $n_1 \times n_1$  dimension matrix
- $C_k$  is a matrix which couples the effect of the secondary state vector into the primary system, an  $n_1 \times n_2$  dimensional matrix

 $\Lambda_{\mathbf{k}}$  is the state transition matrix for the secondary subsystem, an  $n_2 \times n_2$  dimensional matrix.

The system is assumed to have the observable outputs

$$= M_k x_k + v_k , \qquad (3.3)$$

where

- $\mathbf{y}_k$  is an m-dimensional observation vector
- $v_k$  is an m-dimensional zero-mean white noise sequence which is independent of  $u_k^1$  and  $u_k^2$  and has covariance matrix  $B_k \delta_{jk}$
- $x_k$  is an  $n = (n_1 + n_2)$ -dimensional vector composed of  $x_k^1$  and  $x_k^2$
- $M_k = [M_{1k}; M_{2k}]$  is an m × n dimensional matrix defining m linearly independent observable combinations of system state variables.  $M_{1k}$  is an m × n<sub>1</sub> dimensional matrix and  $M_{2k}$  is an m × n<sub>2</sub> dimensional matrix.

To reduce computational requirements for implementing an estimate of the primary subsystem state vector  $\mathbf{x}_k^i$ , the filter design is based on the simplified model

$$\mathbf{\bar{x}}_{k+1} = \phi_k \mathbf{\bar{x}}_k + \mathbf{\bar{u}}_k \tag{3.4}$$

$$\overline{\mathbf{y}}_{\mathbf{k}} = \mathbf{M}_{\mathbf{i}\mathbf{k}}\overline{\mathbf{x}}_{\mathbf{k}} + \overline{\mathbf{v}}_{\mathbf{k}}, \qquad (3.5)$$

(2.47)



namati in a na hainna in shailina Anna an hainna in shailina Anna an hainna in shailina

where

THE PROPERTY OF THE

19142223

1986569251

- $\mathbf{T}_{\mathbf{k}}$  is an  $\mathbf{n}_1$ -dimensional state vector
- **y**k is an m-dimensional observation vector
- $\mathbf{u}_k$  is an  $\mathbf{n}_i$ -dimensional zero-mean white noise sequence with covariance  $\mathbf{\bar{Q}}_k \delta_{ik}$
- $\Psi_k$  is an m-dimensional zero-mean white noise sequence with covariance matrix  $R\delta_{ik}$

It is now of interest to determine the error of the estimate of the state  $x_{\pm}^{1}$  where the Kalman filtering algorithm is applied to the simplified model defined by Equations (3.4) and (3.5).

In order to calculate the performance loss from optimal incurred by this simplification, we could compute the optimal estimate and compare it with the estimate based on the simplified model. However the optimal estimate requires the solution of an  $n \times n$  dimensional matrix Ricatti equation which we wanted to avoid by using the suboptimal approach. It is possible to determine a portion of the performance degradation with little additional computation. This portion is the effect of the secondary state variables on the estimate when the suboptimal Kalman gain is used. Although this is not the totak performance loss due to the suboptimal estimation it does give an indication of this effect.

The linear filter for the entire system is given by

$$\hat{x}_{k+1} = A_k \hat{x}_k + K_k [y_k - M_k \hat{x}_k], \qquad (3.6)$$

where

$$A_{k} = \begin{bmatrix} \phi_{k} & C_{k} \\ 0 & \lambda_{k} \end{bmatrix}$$
$$K_{k} = \begin{bmatrix} K_{1k} \\ K_{2k} \end{bmatrix} \text{ is an } n \times n \text{ arbitrary gain matrix, } K_{1k}$$

being nixm dimensional and Kak

#### being n, xm dimensional.

If the designer assumes the simplified form of the plant as a basis for synthesiming a filter for the state vector nomponents that are of primary interest, the gain matrix becauses

 $\mathbf{K}_{\mathbf{k}} = \begin{bmatrix} \mathbf{K}_{1\mathbf{k}} \\ \mathbf{0} \end{bmatrix}.$ 

Now defining the error of the setimate as

and

we can determine the covariance matrix for the filter. However, since the filter gain is based on the primary system only,  $\hat{x}_k^2 = 0$  and  $\bar{x}_k^2 \simeq x_k^2$ . Thus, by appropriate substitution into (3.7), we get

 $\begin{array}{cccc} \mathbf{X}_k^1 &=& \mathbf{x}_k^1 - \mathbf{\hat{x}}_k^1 \\ \mathbf{X}_k^2 &=& \mathbf{x}_k^2 - \mathbf{\hat{x}}_k^2 \end{array} , \end{array}$ 

$$\mathbf{I}_{k+1}^{i} = \left[\phi_{k} - \mathbf{X}_{ik}\mathbf{H}_{ik}\right]\mathbf{X}_{k}^{i} + \overline{\mathbf{C}}_{k} + \mathbf{X}_{k}^{2} + \mathbf{u}_{k}^{i} - \mathbf{X}_{ik}\mathbf{v}_{k}, \qquad (3.8)$$

where

# $\vec{C}_k = C_k - K_{ik} M_{ik}$ .

The covariance matrix for the overall system can now be written in terms of the subsystem elements. Defining

 $\rho_{ij}^{\mathbf{k}} = \mathbf{E}[(\mathbf{x}_{\mathbf{k}}^{i})(\mathbf{x}_{\mathbf{k}}^{j})^{\mathsf{T}}] ,$ 

where the dimensions of  $\rho_{i1}^k$ ,  $\rho_{12}^k$ ,  $\rho_{22}^k$  are respectively  $n_1 \times n_1$ ,  $n_1 \times n_2$ , and  $n_2 \times n_1$  and where  $\rho_{12}^k \simeq (\rho_{21}^k)^T$  Performing the indicated operations yields

$$\rho_{11}^{k+1} = \mathbf{B}_{k} \rho_{1}^{k} \mathbf{B}_{k}^{T} + \mathbf{B}_{k} \rho_{12}^{L} \tilde{\mathbf{C}}_{k}^{T} + \tilde{\mathbf{C}}_{k} \rho_{21}^{k} \mathbf{B}_{k}^{L} + \tilde{\mathbf{C}}_{k} \rho_{22}^{k} \tilde{\mathbf{C}}_{k}^{T} + \mathbf{Q}_{k}^{1} + \mathbf{K}_{1k} \mathbf{R}_{k} \mathbf{K}_{1k}^{T}$$
(3.9a)

$$\rho_{1s}^{\mathbf{k}+1} = \mathbf{B}_{\mathbf{k}} \rho_{1s}^{\mathbf{k}} \Lambda_{\mathbf{k}}^{\mathbf{r}} + \overline{\mathbf{C}}_{\mathbf{k}} \rho_{2s}^{\mathbf{k}} \Lambda_{\mathbf{k}}^{\mathbf{r}}$$
(3.9b)

$$\rho_{aa}^{k+1} = \Lambda_{k} \rho_{a}^{k} \Lambda_{k}^{T} + Q_{k}^{2}, \qquad (3.90)$$

والمحاج والمحاج

where

B<sub>k</sub> =  $\phi_k = K_{1k} H_{1k}$  .

Note that the nature of the coupling between  $\rho_{22}^{k+1}$ ,  $\rho_{12}^{k+1}$ , and  $\rho_{11}^{k+1}$  permits the explicit solution for each of these matrices in a serial manner.

In order to evaluate the degradation in performance due to the effect of the secondary subsystem, it is necessary to determine the covariance matrix based on the simplified model. This is easily determined from  $\rho_{11}^{k+1}$  by setting  $\overline{C}_k$  to zero. Defining

 $\rho^{k} = \mathbb{E}[(\tilde{\mathbf{x}}_{k})(\tilde{\mathbf{x}}_{k})],$ 









113

(3.7)

we obtain the recurrence relation for the simplified system as

$$\rho^{\mathbf{k}+1} = \mathbf{B}_{\mathbf{k}} \rho^{\mathbf{k}} \mathbf{B}_{\mathbf{k}}^{\mathsf{T}} + \bar{\mathbf{Q}}_{\mathbf{k}} + \mathbf{K}_{1\mathbf{k}} \bar{\mathbf{R}}_{\mathbf{k}} \mathbf{K}_{1\mathbf{k}}^{\mathsf{T}} . \tag{3.10}$$

Now defining the error introduced into the primary subsystem by the secondary subsystem as

 $\mathbf{E}_{\mathbf{k}} = \rho_{111}^{\mathbf{k}} - \rho^{\mathbf{k}} \, .$ 

yields a recurrence calculation for the error matrix; namely

$$\mathbf{E}_{\mathbf{k}+1} = \mathbf{B}_{\mathbf{k}} \mathbf{E}_{\mathbf{k}} \mathbf{B}_{\mathbf{k}}^{\mathsf{T}} + \mathbf{B}_{\mathbf{k}} \boldsymbol{\rho}_{10}^{\mathsf{K}} \mathbf{\overline{C}}_{\mathbf{k}}^{\mathsf{T}} + \mathbf{\overline{C}}_{\mathbf{k}} \boldsymbol{\rho}_{10}^{\mathsf{K}} \mathbf{\overline{C}}_{\mathbf{k}}^{\mathsf{T}} + \Delta \mathbf{Q}_{\mathbf{k}} + \mathbf{K}_{1\mathbf{k}} \Delta \mathbf{R}_{\mathbf{k}} \mathbf{K}_{1\mathbf{k}}^{\mathsf{T}} , \qquad (3.11)$$

where

 $\Delta \mathbf{Q}_{\mathbf{k}} = \mathbf{Q}_{\mathbf{k}}^{1} - \mathbf{\widetilde{Q}}_{\mathbf{k}}$  $\Delta \mathbf{R}_{\mathbf{k}} = \mathbf{R}_{\mathbf{k}} - \mathbf{\widetilde{R}}_{\mathbf{k}},$ 

and where the matrices  $\rho_{12}^k$  and  $\rho_{22}^k$  are obtained recursively as above. The  $\Delta Q_k$  and  $\Delta R_k$  matrices represent the error in modeling the white noise sequences for the simplified system. Thus the error matrix  $E^k$  can be computed simultaneously with the simplified filter equations in order to indicate the error incurred by omitting the secondary subsystem variables from the primary model.

The above analysis holds for any arbitrary gain matrix  $K_{1k}$ . However if the optimal Kalman filter is employed, the gain matrix is chosen as

$$K_{1k} = \rho^{k} M_{1k}^{T} [M_{1k} \rho^{k} M_{1k}^{T} + \bar{R}_{k}]^{-1} . \qquad (3.12)$$

# 3.2 Results for the Continuous Case

In the case where the system is described by linear differential equations instead of difference equations similar results are obtained. Thus if the system is given as

$$\dot{x}_{1} = \Lambda(t)x_{1} + C(t)x_{2} + u_{1}$$

$$\dot{x}_{2} = \Lambda(t)x_{2} + u_{2}$$
(3.13)

and

# y(t) = h(t)x(t) + y(t),

where the vectors and matrices are defined to be analogous to the discrete case and if the simplified system is given by

the corresponding covariance equations are

$$\dot{\rho}_{11} = B\rho_{11} + \rho_{11}B^{T} + \bar{C}\rho_{21} + \rho_{12}\bar{C}^{T} + Q_{1} + K_{1}RK_{1}^{T}. \qquad (3.15a)$$

$$\phi_{12} = B \rho_{12} + \bar{0} \rho_{22} + \rho_{12} \Lambda^{\mathrm{T}}$$
 (3.15b)

$$z_z = \Lambda \rho_{zz} + \rho_{zz} \Lambda^z + \mathbf{Q}_z , \qquad (3.15c)$$

where

# $B = A_1(t) - K_1(t)M_1(t)$ $\overline{O} = O(t) - K_1(t)M_2(t) .$

# Similarly the covariance matrix for the simplified system is given by

$$\dot{\rho} = B\rho + \rho B^{T} + \bar{Q} + K_{1}\bar{R}K_{1}^{T}.$$

In this case the differential equation describing the degradation in the system performance due to the existence of unmodeled state variables is given by

$$\mathbf{\dot{E}} = \mathbf{B}\mathbf{E} + \mathbf{E}\mathbf{S}^{\mathrm{T}} + \mathbf{\ddot{C}}\rho_{21} + \rho_{12}\mathbf{\ddot{C}}^{\mathrm{T}} + \Delta\mathbf{Q} + \mathbf{K}_{1}\Delta\mathbf{R}\mathbf{K}_{1}^{\mathrm{T}}, \qquad (3.17)$$

where

$$\mathbf{E}(0) = \rho_{12}(0) - \rho(0) ,$$

which equals zero if the initial conditions for these covariance matrices coincide. In the continuous case the optimal estimate based on the simplified model is implemented when the gain matrix is chosen as

 $\mathbf{K}_1 = \boldsymbol{\rho} \mathbf{M}_1^{\mathrm{T}} \mathbf{R}^{-1} \ .$ 

# 3.3 The Steady-State Case

An analytic closed form expression for the steady-state values of performance loss can be obtained directly from the above differential equations. Temporarily assuming that the design corresponding to the simplified model



i ateal

(3. 16)

in stable, three matrix equations are obtained by setting

 $\dot{\mathbf{k}} = \dot{\boldsymbol{\rho}}_{12} = \dot{\boldsymbol{\mu}}_{22} = 0.$ 

Employing the well-known solution of these equations<sup>12</sup>, the following sequence of expressions permits the determination of the steady-state performance loss E due to the modeling errors.

$$v_{gg} = \int_{0}^{\infty} exp[\Lambda^{t}] q_{g} exp[\Lambda^{t}t] dt \qquad (3.18a)$$

$$\rho_{12} = \int_{0}^{\infty} \exp(Bt) \partial \rho_{22} \exp[\Lambda^{T}t] dt \qquad (3.18b)$$

$$\mathbf{E} = \int_{0}^{\mathbf{v}} \exp\left[\mathbf{B}\mathbf{t}\right] \left[\tilde{\mathbf{C}} \rho_{12}^{\mathsf{T}} + \mathbf{K}_{1} \Delta \mathbf{R} \mathbf{K}_{1}^{\mathsf{T}} + \Delta \mathbf{Q} + \rho_{12} \tilde{\mathbf{C}}^{\mathsf{T}}\right] \exp\left[\mathbf{B}^{\mathsf{T}}\mathbf{t}\right] \, \mathrm{d}\mathbf{t} \, . \tag{3.180}$$

# 3.4 The Case of Random Sampling

In many applications of Kalman filtering, the filter states are updated randomly in time. This situation may be caused by computational allocation in the system computer, subsystem malfunctions, or when measurements are made aperiodically<sup>8-10</sup>. The "optimal" gain for this type of filter is zero except at the update time when it is given by

$$\mathbf{K}_{1} = \rho \mathbf{M}_{1}^{T} [\mathbf{M}_{1} \rho \mathbf{M}_{1}^{T} + \mathbf{\bar{R}}]^{-1}$$

This is nearly the same optimal gain as the discrete time filter, except that in this case the covariance matrices are propagated by the following differential equations obtained by metting  $K_1 = 0$  in (3.15b) (3.15c) and (3.17) for the continuous case. Thus, between the random update times, we have

$$\mathbf{E} = \mathbf{B}\mathbf{E} + \mathbf{E}\mathbf{B}^{\mathrm{T}} + \mathbf{C}\boldsymbol{\rho}_{g_{1}} + \boldsymbol{\rho}_{1g}\mathbf{C}^{\mathrm{T}} + \Delta\mathbf{Q}$$
(3.19a)

$$\dot{\rho}_{12} = B\rho_{12} + \rho_{12}\Lambda^{T} + C\rho_{27}$$
(3.19b)

$$\dot{\rho}_{gg} = \Lambda \rho_{gg} + \rho_{gg} \Lambda^{T} + Q_{g} \qquad (3.19c)$$

At the time an update in the estimated state of the plant occurs, the old value  $\hat{x}_1^-$  (a priori estimate) is replaced by a new value  $\hat{x}_1^+$  (a posteriori estimate). The a posteriori estimate is obtained as

$$\mathbf{k}_{1}^{*} = \hat{\mathbf{x}}_{1}^{*} + \mathbf{K}_{1} \left[ \mathbf{M}_{1} \mathbf{x}_{1} + \mathbf{v} + \mathbf{M}_{2} \mathbf{x}_{2} - \mathbf{M}_{1} \mathbf{x}_{1}^{*} \right] . \tag{3.20}$$

Since no estimate of the state vector x, is generated,

 $1^*_{,} = 1^-_{,} = 0$ .

Defining the instantaneous estimation errors as

$\mathbf{x}_1^* = \mathbf{x}_1 - \mathbf{\hat{x}}_1^*$	ı
$\mathbf{x}_{i} = \mathbf{x}_{i} - \hat{\mathbf{x}}_{i}$	
$\mathbf{X}_{2} = \mathbf{X}_{2} ,$	
$\mathbf{X}_{1}^{L} = \mathbf{X}_{1}^{-} + \mathbf{K}_{1} [\mathbf{M}_{1} \mathbf{X}_{1}^{-} + \mathbf{M}_{2} \mathbf{X}_{2} + \mathbf{v}]$	(3. 21)

we obtain

 $\overline{\mathbf{x}}_{2}^{\dagger} = \overline{\mathbf{x}}_{2}^{\dagger} . \tag{3.22}$ 

Using these expressions we can obtain expressions for updating the covariance matrices corresponding to the a priori and a posteriori estimates in a straightforward manner. Thus, defining

 $\rho_{ij}^* = \mathbb{E}[(\mathbf{X}_i^*) (\mathbf{X}_j^*)^T]$  $\rho_{ij}^* = \mathbb{E}[(\mathbf{X}_i^*) (\mathbf{X}_j^*)^T]$ 

and yields

 $\rho_{11}^{*} = [I - K_1 M_1] \rho_{11}^{-} [I - K_1 M_1]^{T} + K_1 R K_1^{T} + [K_1 M_2] \rho_{21}^{-} [I - K_1 M_1]^{T} + [I - K_1 M_1] \rho_{12}^{-} [K_1 M_2] \rho_{22}^{-} [K_1 M_2]^{T}$   $\rho_{12}^{*} = [I - K_1 M_1] \rho_{12}^{-} + K_1 M_2 \rho_{22}^{-}$   $\rho_{22}^{*} = \rho_{22}^{*} ,$ 

where I is the identity matrix of dimension  $n_1$  .

The corresponding equation for updating the covariance matrix of the simplified system is  $\rho^{+} = [I - K_1 M_2] \rho^{+} [I - K_1 M_1]^T + K_1 R K_1^T.$ 



allers by a

(3. 234)

(3. 23b)

(3. 230)

(3.24)

į,

Finally the equation for updating the performance loss defined as

$$\mathbf{E} = \rho_{11} - \rho$$

is obtained from (3.23a) and (3.24) as

$$\mathbf{k}^{*} = (\mathbf{I} - \mathbf{K}_{1}\mathbf{M}_{2})\mathbf{k}^{T}(\mathbf{I} - \mathbf{K}_{1}\mathbf{M}_{1})^{T} + \mathbf{K}_{1}\wedge\mathbf{R}\mathbf{K}_{1}^{T} + [\mathbf{K}_{1}\mathbf{M}_{2}]\rho_{21}^{T}[\mathbf{I} - \mathbf{K}_{1}\mathbf{M}_{1}]^{T} + [\mathbf{I} - \mathbf{K}_{1}\mathbf{M}_{1}]\rho_{12}^{T}[\mathbf{K}_{1}\mathbf{M}_{2}]\rho_{22}^{T}[\mathbf{K}_{1}\mathbf{M}_{2}]^{T} + (\mathbf{K}_{1}\mathbf{M}_{2})\rho_{22}^{T}[\mathbf{K}_{1}\mathbf{M}_{2}]^{T} + (\mathbf{K}_{1}\mathbf{M}_{2})\rho_{22}^{T}[\mathbf{K}_{1}\mathbf{M}_{2}]^{T} + (\mathbf{K}_{2}\mathbf{M}_{2})\rho_{22}^{T}[\mathbf{K}_{2}\mathbf{M}_{2}]^{T} + (\mathbf{K}_{2}\mathbf{M}_{2})\rho_{22}^{T}]$$

# 4. CONCLUSIONS

In this chapter the suboptimal filtering problem is discussed. Roughly, the suboptimal filtering techniques can be broken into two general categories. The first category has the property that the order of the suboptimal filter is the same as that of the optimal filter, but the number of equations necessary to generate the filter gain is reduced. One way of performing this reduction is by partitioning the system and ignoring any correlation which exists between the various subsystems thus formed. Generally an extensive simulation is necessary to minimize the degredation in filter performance caused by the suboptimality. The savings in computation can, however, be very substantial if the system can be finely partitioned.

The second category of suboptimal filters has the property that the filter design is based on a lower dimensional state vector than the higher dimensional model which defines the system state. As a result the number of covariance squations is also reduced. An estimate of the acst important elements of the higher dimensional model can be obtained along with a differential equation describing the performance loss incurred by the model simplification. The filtering and performance loss equations can be solved simultareously to obtain an indication of actual performance.



We I Barrow & M

A Contraction Contraction

Mar - Mar

r i te 🗩 est a

1094033)13

# REFERENCES

ŝ

THE REPORT OF THE PARTY OF

an 1.5228528 2254528 24100

Ľ

			all and the second second by
1.	Kalman, R.E.	A New Application to Linear Filtering and Prediction Problems. Transactions, Amorican Society of Mechanical Englances, Journal of Busic Engineering, Vol.82, 1960; pp.35-45.	
2.	Joneph, P.D.	Suboptimal Linear Filtering. Space Technology Laboratory Inc., Technical Report, December 1963.	
3.	Meditch, J.S.	Suboptimal Linear Filtering for Continuous Dynamic Process. Aerospace Corporation, Technical Report, July 1964.	The second s
4.	Pentecost, E.E.	Synthesis of Computationally Effective Sequential Linear Estimation PhD disperta- tion, Department of Engineering, University of California, Los Angoles, May 1908.	
5.	Heffes, H.	The Effect of Erroneous Models on the Kalman Filter Response. Institute of Electrical and Electronic Engineers, Transactions on Automatic Control (Short, Papers), Vol.AC-11, July 1968, pp.541-543.	
6.	Aoki, M.	Control of Large Scale Dynamic Systems by Aggregation, Preprints, Joint Automatic Control Conference (Philadelphia, Pa., 1967), pp.178-186.	
7.	Luenberger, D.G.	Observing the State of a Linear System. Institute of Electrical and Electronic Engineers, Transactions on Military Electronics, Vol.MIL-8, April 1964, pp.74-80.	
8.	Aoki, M. Huddle, J.R.	Estimation of the State Vector of a Linear Stochastic System with a Constrained Estimator. Institute of Electrical and Electronic Engineers, Transactions on Automatic Control (Short Papers), Vol.AC-12, August 1967, pp.432-433.	
9.	Banbrook, H.W. et al.	Final Engineering Report, LOHAN C/D Navigation System Integration. Chapter 8. Contract AF-33(657;-14389. January 1966.	
10.	Huddle, J.R. at al.	Final Engineering Report, Doppler-Inertail LORAN C/D Navigation System Integration. Appendix I, Contract AF-33(618)67C-1184, October 1967.	
11.	Richman, J. Friedland, B.	Design of Optimum Mixer Filter for Aircraft Navigation Systems. Presented at the ION Conference, Dayton, Obio, May 1967.	
12.	Bellman, R.	Introduction to Matrix Analysis, McGraw-Hill, New York, 1960.	
13.	Nishiaura, T.	Error Bounds of Continuous Kalman Filters and the Application to Orbit Determination Problems. Institute of Electrical and Electronic Engineers, Transactions on Automatic Control, Vol.AC-12, June 1967, pp.268-275.	
14.	Larson, R.E.	Optimum Combined Control and Estimation for Partially Controlied Systems. Pro- ceedings, Fourth Alierton Conference on Circuit and Bystem Theory, University of Illincis, Urgana, October 1966.	
15.	Widnall, W.	On the Design of Nearly Optimal Linear Time-Varying Sampled Data Stochastic Controllers. PhD, Department of Aeronautics and Astronautics, Massachumetts Institute of Technology, Cambridge, Massachumetts, September 1986.	
16.	Martin, F.H.	Closed Loop Near-Optimum Steering for a Class of Space Missions. BoD, Report T-413, Instrumentation Laboratory, Massachusetts Institute of Technology, Cambridge, Massachusetts, 1965.	
17.	Johansen, D.E.	Optimal Control of Linear Systems with Complexity Constraints. PhD, Harvard University, Cambridge, Massachumetts, 1984.	
18.	Aoki, M.	Determination of Tuning Factor in a Discrete Observation Data Innorporation Scheme. Litton Industries Inc., Guidance and Control Division Publication No.7651, Woodland Hills, California, April 1968.	and the second
19.	Pearson, J.D.	Dynamic Decomposition Techniques. In "Optimization Mathods for Large Boale Systems", D.A. Wismer (Editor), McGraw-Hill, New York, (in press).	
20.	Huddle, J.R. Wiemer, D.A.	Degradation of Linear Filter Performance Due to Modeling Error. Institute of Electrical and Electronic Engineers, Transactions on Automatic Control, Vol.AC-13, No.4, August 1968.	
21.	Schmidt, S.F.	Estimation of State with Acceptable Accuracy Constraints - Analytical Machanica	



# CHAPTER 6 - COMPARISON OF KALMAN, BAYESIAN AND MAXIMUM LIKELIHOOD ESTIMATION TECHNIQUES

#### byr .

H. W. Sorenson

# University of California at San Diego La Jolla, Galifornia 92037, USA

· · ·

di Oliosinta

.

# PRECEDING PAGE BLANK

# •

# CHAPTER 6 - COMPARISON OF KALMAN, BAYESIAN AND MAXIMUM LIKELIHOOD Estimation techniques

#### H.W.Sorenson

#### 1. PROBABILISTIC APPROACHES TO ESTIMATION THEORY

There are many different approaches to the inferential problem of estimating parameters or states from observed data. Some of these have been discussed in other chapters (e.g., Chapter 1). In this chapter the problem is approached by dealing with the probability density functions describing the state and measurement variables. This general probabilistic formulation is referred to as the Bayesian approach<sup>1-3</sup> and it provides a framework within which many other approaches can be submaned. Using the Bryosian formulation, one arrives naturally at the maximum a posteriori and maximum likelihood estimation procedures<sup>4</sup>. It is also seen that deterministic least-squares can be reinterpreted as maximum a posteriori estimation theory. In Unapter 1 the Kalman filter equations have been derived as the solution of the unbiased, minimum variance estimation problem. It is pointed out here that the minimum variance oriterion is just one of many that could be chosen which all yield the same estimate for linear, guageing systems.

The estimation problem for time-discrete, nonlinear, stochastic westems is formulated in Section 1.2. Within the Rayssian framework, a general treatment of nonlinear systems can be developed in a straightforward manner. As indicated in Sections 2 and 3, it is necessary to introduce additional restrictions in order to obtain practical solutions, but it is felt that the general formulation and subsequent simplifications provide desirable insights into the monlinear estimation problem.

# 1.1 Deterministic Least-squares Estimation

As a preliminary to the probabilistic discussion of estimation theory that is found in Sections 2 and 3, the problem of estimating parameters or states from measurement data is first treated using deterministic leastsquares. This formulation leads to a minimization problem whose solution is nontrivial. It is pointed out in Section 3 that the deterministic least-squares approach is equivalent to the maximum a posteriori approach when the appropriate assumptions are introduced.

To simplify the discussion, first consider the problem of estimating an unknown parameter  $\underline{x}$  from measurement data  $\underline{x}_i (i=1,2,..,N)$ , where the collection of data will be denoted as  $\underline{Z}_N$ . Suppose that the data  $\underline{Z}_N$  are obtained at discrete instants of time  $\underline{t}_i (i=1,2,...,N)$  and that the  $\underline{x}_i$  is an n-dimensional vector representing m independent measurements. These data contain unknown errors  $\underline{Y}_i$  and are related to the parameters according to

$$\underline{a}_{i} = \underline{b}_{i}(\mathbf{x}) + \underline{y}_{i}; \quad \mathbf{i} = 1, 2, \dots, N.$$
 (1.1)

where the  $h_i$  are known functions of the n-dimensional vector of parameter x.

In the classical least-squares procedure, one treats (1.1) in a deterministic fashion and finds the value of g that minimizes the sum of the squares of the error. For example, combine the N measurement vectors into one eN vector (or, suppose N=1) and denote it as

$$\mathbf{X}_{N} = \mathbf{h}(\mathbf{z}) + \mathbf{Y}, \qquad (1.2)$$

where  $Z_{\mu}$ , b and  $\underline{Y}$  have obvious definitions. Then, choose the  $\underline{x}$  so that

 $\mathbf{L} = \left[ \mathbf{Z}_{N} - \mathbf{b}(\mathbf{x}) \right]^{T} \left[ \mathbf{Z}_{N} - \mathbf{b}(\mathbf{x}) \right]$ (1.3)

is minimized. Since there are no other constraints, a necessary condition for L to be minimized is

$$-\left(\frac{\partial \mathbf{h}}{\partial \mathbf{x}}\right)^{\mathsf{T}} \mathbf{Z}_{\mathsf{H}} + \left(\frac{\partial \mathbf{h}}{\partial \mathbf{x}}\right)^{\mathsf{T}} \mathbf{h}(\mathbf{x}) = 0 .$$
 (1.4)

# PRECEDING PAGE BLANK

• The following convention is used. The partial derivative of a scalar function h of an n-dirensional vector variable <u>n</u> is denoted by  $\frac{\partial b}{\partial \underline{a}}$  and is a lan matrix whose elements are  $\frac{\partial b}{\partial \underline{a}_1}$ ,  $\frac{\partial b}{\partial \underline{a}_2}$ ,...,  $\frac{\partial b}{\partial \underline{a}_n}$ . If an m-dimensional vector function <u>b</u> is differentialed, then  $\frac{\partial b}{\partial \underline{a}}$  is an matrix with the ij<sup>th</sup> element equal to  $\frac{\partial b}{\partial \underline{a}_j}$ . Also, the second derivative of ancalar function <u>b</u> is written as  $\frac{\partial}{\partial (\partial b}/\partial \underline{a})^T/\partial \underline{a}$ , and is an analysis whose ij<sup>th</sup> element is  $\frac{\partial^2 b}{\partial x_i \partial x_i}$ .



It is not generally possible to solve (1,4) explicitly for  $\chi$ . However, if (1,2) is linear so that

<u>9</u>µ = H, '

$$\mathbf{Z}_{N} = \mathbf{H}_{\underline{X}} + \underline{Y} , \qquad (1.5)$$

where H is an mN×n matrix, then

so that (1.4) becomes

$$(\mathbf{H}^{\mathrm{T}}\mathbf{H})\mathbf{x} = \mathbf{H}^{\mathrm{T}}\mathbf{Z}_{\mathrm{H}}, \qquad (1.6)$$

Denoting the solution of (1.6) by  $\boldsymbol{\sharp}$ , the least-squares estimate, supposing that (H<sup>T</sup>H) has an inverse, is seen to be

$$\pm = (H^{T}H)^{-1} H^{T}Z_{N}$$
 (1.7)

Observe in (1.7) that an  $n \times n$  matrix must be inverted and that all of the measurement data are processed. The positive-definiteness of (H<sup>T</sup>H) is equivalent to the observability of the system (1.8) as introduced in Chapter 1. It is possible to generate least-squares estimates recursively as new data are obtained and thereby avoid reprocessing old data<sup>3</sup>.

Suppose  $\underline{f}_{k-1}$  is the estimate of  $\underline{x}$  based on the data  $\underline{x}_1$ ,  $\underline{x}_2$ ,...,  $\underline{x}_{k-1}$ . Let  $\underline{x}_{k-1}$  denote the set of all these data and suppose that new data  $\underline{x}_k$  are obtained. Let

$$\mathbf{L}_{\underline{k}} = \sum_{j=1}^{\underline{k}} \left[ \mathbf{g}_{j} - \mathbf{H}_{j} \mathbf{z} \right]^{T} \left[ \mathbf{g}_{j} - \mathbf{H}_{j} \mathbf{z} \right] . \tag{1.8}$$

Note that (1.8) is identical with (1.3), except that the individual samples are written explicitly and the mystem is assumed to be described by a linear version of (1.1).

$$\begin{split} \mathbf{L}_{k} &= \sum_{i=1}^{k-1} \left[ \mathbf{u}_{i} - \mathbf{H}_{ik} \right]^{T} \left[ \mathbf{z}_{i} - \mathbf{H}_{ik} \right] + \left[ \mathbf{u}_{k} - \mathbf{H}_{kk} \right]^{T} \left[ \mathbf{u}_{k} - \mathbf{H}_{kk} \right] \\ &= \mathbf{L}_{k-1} + \left[ \mathbf{u}_{k} - \mathbf{H}_{kk} \right]^{T} \left[ \mathbf{u}_{k} - \mathbf{H}_{kk} \right] \; . \end{split}$$

To determine the estimate that minimizes Lig , form

$$\frac{\partial \tilde{x}}{\partial r^{k}} = 0 = \frac{\partial \tilde{x}}{\partial r^{k-1}} - 3 \Xi_{1}^{k} H^{k} + 3 \Xi_{2} H_{1}^{k} H^{k}$$

Ht

H**k-**≀ ≙ .

or, letting

this condition becomes

$$\cdot 2(\mathbf{Z}_{k-1})^{T} \mathbf{H}^{k-1} + 2\mathbf{g}_{k}^{T} (\mathbf{H}^{k-1})^{T} \mathbf{H}^{k-1} - 2\mathbf{g}_{k}^{T} \mathbf{H}_{k} + 2\mathbf{g}_{k}^{T} \mathbf{H}_{k}^{T} \mathbf{H}_{k} = 0$$

Thus, one obtains

Guided by the intuition provided by Chapter 1, let

Then (1.9) becomes

$$P_{k}^{-1}g_{k} = P_{k-1}^{-1}g_{k-1} + H_{k}^{T}g_{k}$$



MA PRESIDENT DASK

teren Trinspieren Erstern

(1.10)

(1.11)

(1.12a)

Accounting that Phi has an inverse out obtains

$$N_{k} = P_{k}P_{k-1}^{-1}\hat{g}_{k-1} + P_{k}H_{k}^{T}\hat{g}_{k-1}$$

But it follows, using the Matrix Inversion L. n.º (see Section 2.1.4 of Chapter 1), that

 $P_{k}P_{k+1}^{-1} = (e_{k+1}^{-1} + H_{k}^{T}H_{k})^{-1} P_{k+1}^{-1}$  $= \mathbf{I} - \mathbf{P}_{\mathbf{k}_{1},\mathbf{k}_{2}} [\mathbf{i}_{\mathbf{k}_{1},\mathbf{k}_{2}}, \mathbf{0}_{1}^{T} + \mathbf{I}]^{-1} \mathbf{H}_{\mathbf{k}_{2}}$ 1 I ~ Kullu I  $\mathbf{K}_{\mathbf{k}} = \mathbf{P}_{\mathbf{k}-1}\mathbf{H}_{\mathbf{k}}^{\mathrm{T}} \left[\mathbf{H}_{\mathbf{k}}\mathbf{P}_{\mathbf{k}-1}\mathbf{H}_{\mathbf{k}}^{\mathrm{T}} + \mathbf{I}\right]^{-1}$ 

where

and

As shown in Chapter 1, it follows that

$$K_{\mathbf{k}} = P_{\mathbf{k}} \mathbf{H}_{\mathbf{k}}^{\mathrm{T}}, \qquad (1.12b)$$

so (1.11) reduces to

$$\mathbf{\hat{z}}_{k} = \mathbf{\hat{z}}_{k-1} + \mathbf{K}_{k} [\mathbf{z}_{k} - \mathbf{H}_{k} \mathbf{\hat{z}}_{k-1}], \qquad (1.13)$$

There is a generalization that is particularly useful. Suppose that certain measurement errors are to be given more weight than others. Then, the cost function can be written as a weighted loast-squares sun,

$$\mathbf{L}_{\mathbf{R}} = \sum_{i=1}^{\mathbf{L}} \left[ \mathbf{z}_{i} - \mathbf{H}_{i} \mathbf{z} \right] \mathbf{R}_{i}^{-1} \left[ \mathbf{z}_{i} - \mathbf{H}_{i} \mathbf{z} \right] , \qquad (1.14)$$

It follows, by forming  $\partial L_{\mu}/\partial \chi$  , that

$$\frac{\partial \mathbf{L}_{\mathbf{k}}}{\partial \mathbf{x}_{\mathbf{k}}} = \mathbf{0} = \frac{\partial \mathbf{L}_{\mathbf{k}-1}}{\partial \mathbf{x}} - 2\mathbf{z}_{\mathbf{k}}^{\mathsf{T}} \mathbf{H}_{\mathbf{k}} \mathbf{R}_{\mathbf{k}}^{-1} + 2\mathbf{z}_{\mathbf{k}}^{\mathsf{T}} \mathbf{H}_{\mathbf{k}}^{\mathsf{T}} \mathbf{R}_{\mathbf{k}}^{-1} \mathbf{H}_{\mathbf{k}}$$

Thus, the weighting matrix  $R_{\rm k}^{-1}$  modifies only the matrices  $P_{\rm k}$  and  $K_{\rm k}$  ,

$$\mathbf{P}_{k} = \left[\mathbf{P}_{k+1}^{*1} + \mathbf{H}_{k}^{*} \mathbf{R}_{k}^{*1} \mathbf{H}_{k}\right]^{-1}$$
(1.15)

$$K_k = P_{k-1}H_k^T [H_k P_{k-1}H_k^T + R_k]^{-1}$$
 (1.16a)

$$= P_k H_k^{-1}$$
, (1.16b)

Clearly, deterministic least-squares estimates are closely related to the Kalman filter estimates. However, no probabilistic interpretations have been attempted. As will be seen below, the problem can be couched in a framework that provides a probabilistic interpretation of least-squares as the maximum a posteriori or most probable estimate.

Thus far, the parameters a have been treated as constants. Another way of considering these parameters is to view them as state variables described by the plant equation

Ak = 1k-1 -

The extension to a more general model for the plant is not especially difficult. Suppose that the plant is described by

$$\mathbf{x}_k = \mathbf{x}_k(\mathbf{x}_{k-1}) + \mathbf{x}_{k-1}; \quad k = 1, 2, \dots$$
 (1, 17)

and the measurements involve only the current state

$$\mathbf{x}_{k} = \mathbf{b}_{k}(\mathbf{x}_{k}) + \mathbf{y}_{k}; \quad k = 1, 2, \dots$$
 (1.18)

Equation (1.17) represents a constraint on the system. The  $\underline{w}_{k-1}$  represent uncertainties or errors in the model that cannot otherwise be accounted for.  $\underline{w}_{k-1}$  will be considered as parameters that are chosen to minimize the estimation error.

Consider the problem of choosing the sequences  $X_k$  and  $X_{k-1}$  to minimize the mean-square error

$$\mathbf{L}_{\mathbf{k}} = [\mathbf{\hat{g}}_{0} - \mathbf{n}_{0}]^{T} \mathbf{P}_{0}^{-1} [\mathbf{\hat{g}}_{0} - \mathbf{n}_{0}] + \sum_{i=1}^{k} \{[\mathbf{x}_{i} - \underline{h}_{i}(\mathbf{x}_{i})]^{T} \mathbf{R}_{i}^{-1} [\mathbf{x}_{i} - \underline{h}_{i}(\mathbf{x}_{i})] + \underline{\mathbf{w}}_{i-1}^{T} \mathbf{Q}_{i-1}^{-1} \underline{\mathbf{w}}_{i-1}] \}, \qquad (1.16)$$





subject to the constraint (J. 17). The first term in (1.19), is included to account for some initial estimate of the state  $\underline{x}_0$ . Weighting matrices  $\mathbf{P}_0^{-1}$ ,  $\mathbf{R}_1^{-1}$ ,  $\mathbf{Q}_1^{-1}$  are included for generality.

This discussion has been included to set the stage for later discussion. As should be anticipated, the solution of this general problem is difficult. However, when (1.17) and (1.18) are linear, one obtains the Kalman filter equations. This aspect is dealt with again in Section 3.3.

# 1.8 General Estimation Problem for Time-Discrete Stochastic

Systems

It is possible to approach the linear estimation problem in a variety of ways. Too often, however, the treatment of the linear problem does not indicate the manner in which one can apply the technique to nonlinear problems, nor are the difficulties presented by nonlinear systems made apparent. It is the intent in this chapter to describe a nonlinear estimation problem and to discuss two of the methods used to attack this problem. After the general character of the approaches is defined, the discussion is specialized to linear systems to obtain the Kalman filter equations. Time-discrete systems are considered in order to simplify the mathematics and thoreby swoid the difficulties that have led to the development of the stochastic calculues of ito<sup>6</sup> und of Stratenovich<sup>7</sup> for time-continuous systems.

Consider a dynamical system described by a nonlinear difference equation in which the state evolves according to

$$\mathbf{X}_{k} = \mathbf{f}_{k}(\mathbf{X}_{k-1}, \mathbf{W}_{k-1}) . \tag{1.20}$$

The state has an initial state  $\underline{x}_0$  which is a random variable with a known probability density function<sup>6</sup>, say  $p(\underline{x}_0)$ . The  $\underline{y}_{\underline{k}-1}$  represents a sample of a random sequence with known probability density function. Throughout this discussion, it is assumed that the  $\underline{w}_1$  are independent between the sampling times and, thereby, constitute a white-noise sequence. Thus

$$\mathbf{p}(\overline{\mathbf{M}}^0, \overline{\mathbf{M}}^1, \dots, \overline{\mathbf{M}}^F) = \mathbf{b}(\overline{\mathbf{M}}^0) \mathbf{b}(\overline{\mathbf{M}}^T) \dots \mathbf{b}(\overline{\mathbf{M}}^F)$$

The state is observed through related measurement data  $\mathbf{z}_k$  described by

$$\underline{x}_k = \underline{h}_k(\underline{x}_k, \underline{y}_k); \quad k = 1, 2, \dots,$$
 (1.21)

11.111.11

The  $\underline{\gamma}_k$  represent a sumple from a white-noise sequence for which the probability density function is known and given by

$$p(\underline{\mathbf{y}}_1, \underline{\mathbf{y}}_2, \dots, \underline{\mathbf{y}}_k) = p(\underline{\mathbf{y}}_1)p(\underline{\mathbf{y}}_2), \dots p(\underline{\mathbf{y}}_k) .$$

The plant and measurement noise sequences are assumed to be independent of each other and of the initial state. This assumption could be eliminated without significant conceptual difficulty but the notation becomes more complicated.

The filtering problem<sup>†</sup> of estimating  $\underline{x}_k$  from measurement data  $\underline{x}_k$  will be considered almost exclusively. The solution of the prediction problem is a trivial result. As shown in Chapter 1, the solution of the smoothing problem is obtained from the filtering solution, so it is not discussed here.

The problem has been cast in a probabilistic mold that reflects the uncertainty in the dynamic model and the measurements that commonly exist in physical systems. Admittedly, this formulation might be considered to be artificial, since one is unlikely to have a good knowledge of the density functions involved, so the uncertainties appear to have been compounded by their introduction. Setting this objection aside, it will be seen that the classical lenst-squares approach is imbedded in this formulation, so that one does not obscure the problem by these considerations. In fact, one is led to the conclusion that the least-squares philosophy is enriched and deepened by recenting the problem in this manner and that new insights are obtained thereby.

The Bayesian approach provides the theoretical structure within which a variety of approaches can be considered. In Section 2 the general results of this method are presented and the Kalman filter equations are derived. The maximum likelihood and maximum a posteriori estimation procedures are discussed in Section 3. The Cramer-Rac inequality is discussed and the Fisher information matrix is related to the error covariance matrix of the Kalman filter.

• The notation that is used follows Fel'dbaum<sup>1</sup> and has the disadvantage that the argument of the function serves a dual purpose. It is used to name the function, as here, and is also treated as a variable name (e.g., it is treated as the variable of integration). The meaning should be clear from the context.

\* The reader is directed to Chapter 1 for a definition of the terms relating to the different aspects of the estimation problem.

(2.5)

# 8. THE BAYESIAN APPROACH

In the "Bayesian approach" to the filtering problem one is concerned first of all with the determination of the a posteriori density function  $p(\underline{x}_k|Z_k)$ . This density function provides all the information required for the solution of this problem. This statement will be discussed in more detail below. The following four aspects of conditional density functions will be used frequently<sup>8</sup>.

(i) For random variables  $\underline{a}$  and  $\underline{b}$  with joint probability density function  $p(\underline{a}, \underline{b})$ , the conditional density of  $\underline{a}$ , given  $\underline{b}$ , is defined as

$$p(\underline{\mathbf{s}}|\underline{\mathbf{b}}) \stackrel{\Delta}{=} \frac{p(\underline{\mathbf{s}},\underline{\mathbf{b}})}{p(\underline{\mathbf{b}})} . \tag{3.1}$$

(ii) For random variables g, b, and g, it follows from (2.1) that

 $\mathbf{p}(\underline{a},\underline{b}|\underline{c}) = \mathbf{p}(\underline{b}|\underline{c})\mathbf{p}(n|\underline{b},\underline{c}) . \tag{2.2}$ 

This is known as the chain rule.

(iii) Further, it can be seen from (2.2) and the properties of density functions that\*

$$\mathbf{p}(\mathbf{g}|\mathbf{b}) = \int \mathbf{p}(\mathbf{g}|\mathbf{b},\mathbf{g})\mathbf{p}(\mathbf{b}|\mathbf{g}) \, d\mathbf{b} \,. \tag{2.3}$$

This is an integrated form of the chain rule and is essentially the Chapman-Kolmogorov equation.

(iv) Equation (2.1) also implies that

$$\mathbf{p}(\underline{\mathbf{a}}|\underline{\mathbf{b}}) = \frac{\mathbf{p}(\underline{\mathbf{b}}|\underline{\mathbf{a}})\mathbf{p}(\underline{\mathbf{a}})}{\mathbf{p}(\underline{\mathbf{b}})} . \tag{2.4}$$

This relation is known as Bayes's rule and is the source for the term used to describe the approach in this section.

#### 2.1 Performance Criteria Considerations

Equations (1.20) and (1.21) and the concomitant density functions provide the information required for the Bayesian formulation of the problem. Before going to the development of the *a posteriori* density relations, first note that the solution of the unbiased, minimum variance estimation problem, as posed in Chapter 1, is obtained if the *a posteriori* density of the state conditioned on all available measurement data is known. This is demonstrated in the following statement.

## Theorem 2.1

Suppose that a random variable  $\chi$  is to be estimated from measurement data Z and suppose that  $\chi$  and Z have the joint probability density function  $p(\underline{x}, \underline{z})$ . The unbiased estimate of  $\underline{x}$  based on the data  $\underline{z}$  that yields the minimum error variance  $\mathbb{E}[(\underline{x}-\underline{x})^T(\underline{x}-\underline{x})] = \min \max$ 

is given by

\$ = E[x | Ζ].

Proof: First, write the error variance in terms of the conditional density, using the identity

 $E[(x-g)^{T}(x-g)] = E\{E[(x-g)^{T}(x-g)]\},\$ 

Clearly, to minimize the error variance it is sufficient to minimize the conditional expectation. Thus, consider

$$\mathbb{E}\left[\left(\underline{x} - \underline{\hat{x}}\right)^{\mathsf{T}}\left(\underline{x} - \underline{\hat{x}}\right)|\underline{Z}\right] = \underline{\hat{x}}^{\mathsf{T}}\underline{\hat{x}} - 2\underline{\hat{x}}^{\mathsf{T}}\underline{\hat{x}}[\underline{Z}] + \mathbb{E}\left[\underline{x}^{\mathsf{T}}\underline{x}\right]\underline{Z}\right]$$
$$= (\underline{\hat{x}} - \mathbb{E}\left[\underline{x}\right]\underline{Z}\right)^{\mathsf{T}}\left(\underline{\hat{x}} - \mathbb{E}\left[\underline{x}\right]\underline{Z}\right] + \mathbb{E}\left[\underline{x}^{\mathsf{T}}\underline{x}\right]\underline{Z}\right] - (\mathbb{E}\left[\underline{x}\right]\underline{Z}\right)^{\mathsf{T}}\mathbb{E}\left[\underline{x}\right]\underline{Z}\right]. \qquad (2.6)$$

Only the first term in (2.6) involves 2. It is quadratic, so that the smallest value it can assume is zero. This obtains when

 $\mathbf{1} = \mathbf{E}[\mathbf{x}|\mathbf{Z}] \ .$ 

The f given by (2.5) is unbianed, as is seen easily by observing that

 $\mathbb{E}[\underline{x}] = \mathbb{E}\{\mathbb{E}[\underline{x}|\underline{z}]\} = \mathbb{E}[\underline{x}].$ 

" The single integral sign in (2.3) and the differential db are used to indicate integrations involving vector variables. Then more than one vector is involved, the differential will be written as d(a,b,c,...).



val 🔁 Hill 7 Hill Primer 🔁 Balli

This completes the proof and demonstrates that the first moment of  $p(\underline{x}|\underline{z})$  provides the unbiased, minimum variance estimate of  $\underline{x}$  .

The minimum error variance criterion is only one of many criteria that could be selected. It has been frequently stated that it is used because it is more tractuble analytically and not always because it is the most appropriate for the problem. This implies that the minimum variance estimates represent a very specialized class. However, Shorman' domonstrated that the results obtained for this criterion actually apply to a broader class of cost functions that includes many other meaningful criteria.

Consider a general error criterion  $L(\underline{x})$  where  $\underline{x} \stackrel{o}{\cong} (\underline{x} - \underline{x})$  and the L has the following characteristics:

(i) L is symmetric so that

$$\mathbf{L}(\underline{\tilde{\mathbf{x}}}) = \mathbf{L}(-\underline{\tilde{\mathbf{x}}}) \quad (2.7)$$

(ii) L is convex so that

$$L[\lambda \underline{x}_1 + (1 \cdot \lambda) \underline{x}_2] \leq \lambda L(\underline{x}_1) + (1 - \lambda) L(\underline{x}_2) , \quad \lambda \ge 0 .$$
(2.8)

This class certainly includes the minimum variance criterion

 $L_{mv} = \tilde{\underline{x}}^{\dagger} \tilde{\underline{x}}$ 

and the absolute error criterion

 $L_{abs} = |\vec{x}|$ .

Suppose that the x is to be estimated from data Z. Then the following result is applicable.

Theorem 2.2:

126

If the conditional density  $p(\underline{x}|\underline{z})$  is symmetric about its mean value, then the estimate  $\underline{z}$  that minimizes any cost function L in this class is identical with the estimate  $S_{mv}$  obtained with the minimum variance criterion.

Proof: It is desired that the estimate  $\frac{2}{5}$  that minimizes the expected value of  $L(\tilde{x})$  be determined. First, note that ....t. . A. 1-1

$$\mathbb{E}[\mathbf{L}(\underline{\mathbf{X}})|\underline{\mathbf{Z}}] = \mathbb{E}[\mathbf{L}(\underline{\mathbf{X}} - \underline{\mathbf{X}})|\underline{\mathbf{Z}}] = \mathbb{E}[\mathbf{L}(\underline{\mathbf{X}} - \underline{\mathbf{X}})|\underline{\mathbf{Z}}],$$

- A -

where the symmetry of L has been used.

Let

Then, observing that

$$\Sigma[L(\underline{x}-\underline{x})|\underline{Z}] = \int_{-\infty}^{\infty} L(\underline{x}-\underline{x})p(\underline{x}|\underline{Z}) d\underline{x} ,$$

and using the definition of  $\leq$ , this becomes

$$\mathbf{E}[\mathbf{L}(\underline{\mathbf{X}}-\underline{\mathbf{X}})|\underline{Z}] = \int_{-\infty}^{\infty} \mathbf{L}(\underline{\boldsymbol{\varepsilon}}+\underline{\mathbf{X}}_{\mathbf{B}\mathbf{V}}-\underline{\mathbf{X}})\mathbf{p}(\underline{\boldsymbol{\varepsilon}}|\underline{Z}) \, d\underline{\boldsymbol{\varepsilon}} \, .$$

But it has been assumed that  $p(\underline{x} | \underline{z})$  is symmetric about the mean, so this implies that  $p(\underline{\epsilon} | \underline{Z}) = p(-\underline{\epsilon} | \underline{Z})$ .

Thus one obtains

$$\mathbf{E}[\mathbf{L}(\underline{x} - \underline{\hat{x}}) | \underline{Z}] = \int_{-\infty}^{\infty} \mathbf{L}(-\underline{\hat{x}} + \underline{\hat{x}}_{\mathbf{HV}} - \underline{\hat{x}}) p(\underline{\hat{x}} | \underline{Z}), d\underline{\hat{x}}$$
$$= \int_{-\infty}^{\infty} \mathbf{L}(\underline{\hat{x}} - \underline{x}_{\mathbf{HV}} + \underline{\hat{x}}) p(\underline{\hat{x}} | \underline{Z}), d\underline{\hat{x}}$$
$$= \int_{-\infty}^{\infty} \mathbf{L}(\underline{\hat{x}} - \underline{\hat{x}}_{\mathbf{HV}} - \underline{\hat{x}}) p(\underline{\hat{x}} | \underline{Z}), d\underline{\hat{x}}$$

Using these identities, it follows that

 $\mathbb{E}[L(\underline{x} - \underline{\hat{x}}) | \underline{z}] = \frac{1}{2} \mathbb{E}\{L(\underline{c} + (\underline{\hat{x}} - \underline{\hat{x}}_{wv})) | \underline{z}\} + \frac{1}{2} \mathbb{E}\{L(\underline{c} - (\underline{\hat{x}} - \underline{\hat{x}}_{wv})) | \underline{z}\} .$ 









But L is convex so

 $\mathbb{E}[L(\underline{x} - \underline{\hat{x}}) | \underline{Z}] \ge \mathbb{E}\{L[\frac{1}{2}(\underline{\varepsilon} + (\underline{\hat{x}} - \underline{x}_{mv})) + \frac{1}{2}(\underline{\varepsilon} - (\underline{\hat{x}} - \underline{x}_{mv}))] | \underline{Z}\} = \mathbb{E}\{L(\underline{\varepsilon}) | \underline{Z}\},$ 

From the definition of  $\xi$ , it can be seen that equality occurs if  $\xi = \xi_{mv}$ .

Observe that, if L is required to be strictly conver, the  $\frac{1}{2}$  is unique and must equal  $\frac{2}{2m_V}$ .

This result can be extended to non-convux cost functions by considering the following statement.

Theorem 2.3

の主要が見たいというとなると、「ないたみない」

のないとうないので、「ないない」というないです。

Suppose that L is symmetric and non-decreasing and that  $p(\underline{x}|\underline{Z})$  is symmetric about the mean. Suppose also that  $p(\underline{x}|\underline{Z})$  is unimodal and satisfies the conditions

$$\lim_{X \to \infty} L(\underline{X})p(\underline{X}|\underline{Z}) = 0.$$

Then the 1 that minimizes a cost function L in this class is identical to the minimum variance estimate  $2_{m_V}$  .

The proof is similar to that of Theorem 2.2 and is ouitted.

This theorem permits a variaty of non-convex cost functions to be considered. For example it is sometimes more meaningful to weigh equally all errors larger than a certain magnitude, since any errors larger than the prescribed limit may be undesirable. Thus, one could consider a modified minimum variance criterion in which

$$\mathbf{L}(\underline{\underline{x}}) = \left\{ \begin{array}{c} [\underline{\underline{x}}^T \underline{\underline{y}}] \\ k^2 \end{array} \right\}, \qquad |\underline{\underline{x}}| \leq k \\ |\underline{\underline{x}}| > k \\ \end{array}$$

Another example would be the uniform cost function in which errors within a certain magnitude are accepted without cost and all other errors are weighted equally. This is described by the following function

$$\mathbf{L}(\underline{\mathbf{X}}) = \begin{cases} \mathbf{0} , & |\underline{\mathbf{X}}| < \frac{\mathbf{k}}{2} \\ 1 , & |\underline{\mathbf{X}}| > \frac{\mathbf{k}}{2} \end{cases}$$

This cost criterion is closely connected to the maximum a posteriori density. Consider the expected cost for this criterion,

$$\mathbb{E}[\mathbf{L}(\underline{x})] = \mathbb{E}\{\mathbb{E}[\mathbf{L}(\underline{x})] | \underline{z}]\}$$

$$= \int_{-\infty}^{\infty} \left\{ \int_{-\infty}^{\infty} L(\underline{\widetilde{x}}) p(\underline{x} | \underline{z}) d\underline{x} \right\} p(\underline{z}) d\underline{z}$$
$$= \int_{-\infty}^{\infty} \left\{ 1 - \frac{\widehat{x}_{unf} + k/2}{\underline{y}_{unf} + k/2} p(\underline{x} | \underline{z}) d\underline{x} \right\} p(\underline{z}) d\underline{z} ,$$

where  $f_{unf}$  is the estimate associated with this criterion. The cost is minimized by maximizing the inner integral. For small enough values of k, the best choice is essentially the maximum value of the *a posteriori* density. Estimates of this type are discussed in Section 3. The theorem implies that the maximum *a posteriori* estimates are equivalent to the minimum variance estimates when the appropriate conditions are satisfied.

## 2.2 The a posteriori Density Function

The a posteriori density function can be seen from the preceding discussion to provide all of the information required to determine estimates for any cost functions. One of the principal advantages of the Kalman filter equations is their recursive character, which enables new measurements to be processed without reprocessing older data. Thus, consider the problem of determining the filtering density in a recursive fashion. The following theorem provides the desired relations.

#### Theorem 2.4

For the system (1.20) - (1.21), the a posteriori density function  $p(\mathbf{x}_k | \mathbf{z}_k)$  evolves according to

$$\mathbf{p}(\underline{\mathbf{z}}_{\mathbf{k}} | \underline{\mathbf{z}}_{\mathbf{k}}) = \frac{\mathbf{p}(\underline{\mathbf{z}}_{\mathbf{k}} | \underline{\mathbf{z}}_{\mathbf{k}-1}) \mathbf{p}(\underline{\mathbf{z}}_{\mathbf{k}} | \underline{\mathbf{z}}_{\mathbf{k}})}{\mathbf{p}(\underline{\mathbf{z}}_{\mathbf{k}} | \underline{\mathbf{z}}_{\mathbf{k}-1})}$$



(2.9)







A CALL STOR

where the normalizing constant is

 $D(\mathbf{Z}_k | \mathbf{Z}_{k-1}) = \int D(\mathbf{X}_k | \mathbf{Z}_{k-1}) D(\mathbf{Z}_k | \mathbf{X}_k) d\mathbf{X}_k .$ 

The prediction density  $p(\underline{x}_k | \underline{z}_{k+1})$  is described by

$$(\underline{x}_{k}|\underline{z}_{k-1}) = \int p(\underline{x}_{k}|\underline{x}_{k-1}) p(\underline{x}_{k-1}|\underline{z}_{k-1}) d\underline{x}_{k-1}$$

The initial condition  $p(\underline{x}_0 | \underline{x}_0)$  is given by

$$p(\underline{x}_0 | \underline{x}_0) = \frac{p(\underline{x}_0 | \underline{x}_0) p(\underline{x}_0)}{p(\underline{x}_0)} .$$

Proof: First, note that the initial condition follows immediately from Bayes's rule (2.4).

Consider arbitrary k . From the chain rule (2.2), one sees that

 $p(\underline{x}_k,\underline{z}_k|\underline{z}_{k-1}) \neq p(\underline{x}_k|\underline{z}_k)p(\underline{x}_k|\underline{z}_{k+1});$ 

so that

$$p(\underline{\mathbf{x}}_{k}|\underline{\mathbf{Z}}_{k}) = \frac{p(\underline{\mathbf{x}}_{k}, \underline{\mathbf{x}}_{k}|\underline{\mathbf{Z}}_{k-1})}{p(\underline{\mathbf{z}}_{k}|\underline{\mathbf{Z}}_{k-1})}$$

The chain rule also enables one to write

$$\mathbf{p}(\underline{\mathbf{x}}_k,\underline{\mathbf{z}}_k|\underline{\mathbf{Z}}_{k-1}) = \mathbf{p}(\underline{\mathbf{x}}_k|\underline{\mathbf{x}}_k,\underline{\mathbf{Z}}_{k-1})\mathbf{p}(\underline{\mathbf{x}}_k|\underline{\mathbf{X}}_{k-1})$$

which can be simplified to

$$p(\underline{\mathbf{z}}_k, \underline{\mathbf{z}}_k | \underline{\mathbf{Z}}_{k-1}) = p(\underline{\mathbf{z}}_k | \underline{\mathbf{x}}_k) p(\underline{\mathbf{x}}_k | \underline{\mathbf{Z}}_{k-1})$$

since  $\underline{x}_k$ , given  $\underline{x}_k$ , is independent of  $\underline{Z}_{k-1}$ . Equating the two relations for  $p(\underline{x}_k, \underline{z}_k | \underline{Z}_{k-1})$ , one obtains

$$\mathbf{p}(\underline{\mathbf{z}}_{k} | \underline{\mathbf{z}}_{k}) = \frac{\mathbf{p}(\underline{\mathbf{z}}_{k} | \underline{\mathbf{z}}_{k-1}) \mathbf{p}(\underline{\mathbf{z}}_{k} | \underline{\mathbf{\lambda}}_{k})}{\mathbf{p}(\underline{\mathbf{z}}_{k} | \underline{\mathbf{z}}_{k-1})}$$

which proves (2.9).

The normalizing constant is determined immediately by using the condition that

$$p(\underline{x}_k | \underline{z}_k) d\underline{x}_k = 1.$$

The proof of (2.10) follows from the integrated chain rule (2.3). Note that

$$p(\mathbf{x}_{k}|\mathbf{Z}_{k-1}) = \int p(\mathbf{x}_{k}|\mathbf{x}_{k-1},\mathbf{Z}_{k-1})p(\mathbf{x}_{k-1}|\mathbf{Z}_{k-1}) d\mathbf{x}_{k-1}$$

which reduces to

and

 $p(\underline{x}_{k}|\underline{Z}_{k-1}) = \int p(\underline{x}_{k}|\underline{x}_{k-1}) p(\underline{x}_{k-1}|\underline{Z}_{k-1}) d\underline{x}_{k-1}$ 

thereby proving (2.10) and completing the proof of the lemma.

Several characteristics of (2.9) and (2.10) require discussion. First, note that the density  $p(\underline{z}_k | \underline{x}_k)$  in (2.9) is defined by the measurement model (1.21) and the prescribed density function for the measurement noise. Similarly, the density  $p(\underline{x}_k | \underline{x}_{k-1})$  appearing in (2.10) is defined by the plant Equation (1.20) and the density function prescribed for the plant noise. Thus, theoretically, one knows these two functions. The initial a posteriori density  $p(\underline{x}_k | \underline{z}_k)$  is known from a priori information so it is possible to determine the a posteriori density  $p(\underline{x}_k | \underline{z}_k)$  for any subsequent value of k.

In a practical sense several difficulties hinder greatly or prevent  $p(\underline{a}_k^{\dagger}\underline{z}_k)$  from being determined.

(3) The plant and measurement equations,  $\underline{f}_k$  and  $\underline{h}_k$ , may inhibit the determination of the  $p(\underline{s}_k | \underline{s}_k)$  and  $p(\underline{s}_k | \underline{s}_{k-1})$ . To avoid this difficulty, it is frequently assumed that the plant and measurement noise are additive. Then the plant and measurement equations take the form

$$\mathbf{x}_{k} = \mathbf{x}_{k}(\mathbf{x}_{k+1}) + \mathbf{x}_{k+1} \tag{2.11}$$

$$\mathbf{z}_{\mathbf{k}}^{*} = \mathbf{\underline{u}}_{\mathbf{k}}(\mathbf{z}_{\mathbf{k}}) + \mathbf{\underline{v}}_{\mathbf{k}} \quad (2.12)$$



(2, 10)

The state

in the Oak

It is also common to assume that these noise sequences are gaussian, with densities

$$p(\underline{w}_{k}) = k_{w} \exp\{-\frac{1}{2} \underline{w}_{k}^{T} Q_{k}^{T} \underline{w}_{k}\}$$
(2.13)

$$p(\underline{y}_{k}) = k_{y} \exp\{-\frac{1}{2} y_{k}^{T} R_{k}^{-1} y_{k}^{T} \}, \qquad (2.14)$$

Using (2.11) - (2.14), it follows that

$$\mathbf{p}(\underline{\mathbf{x}}_{k}|\underline{\mathbf{x}}_{k-1}) = \mathbf{k}_{w} \exp\left(-\frac{1}{2} \left[ (\underline{\mathbf{x}}_{k} - \underline{\mathbf{f}}_{k}(\underline{\mathbf{x}}_{k+1}))^{T} \mathbf{q}_{k-1}^{-1} [(\underline{\mathbf{x}}_{k} - \mathbf{f}_{k}(\underline{\mathbf{x}}_{k+1}))] \right]$$
(2.15)

$$p(\underline{\mathbf{x}}_{\mathbf{k}}|\underline{\mathbf{x}}_{\mathbf{k}}) = \mathbf{k}_{\mathbf{v}} \exp\left\{-\frac{1}{2} \left[\mathbf{z}_{\mathbf{k}} - \mathbf{h}(\underline{\mathbf{x}}_{\mathbf{k}})\right]^{T} \mathbf{R}_{\mathbf{k}}^{-1} \left[\mathbf{z}_{\mathbf{k}} - \mathbf{h}_{\mathbf{k}}(\underline{\mathbf{x}}_{\mathbf{k}})\right]\right\}.$$
(2.16)

This shows that the difficulty in determining these conditional densities is avoided by assuming additive noise.

- (ii) The integration required in (2.10) cannot generally be accomplished in a closed form. The principal exception occurs when the plant and measurement equations are linear and the initial state and the nuise sequences are gaussian.
- (iii) Equation (2.9) requires the multiplication of two functions so, if they are known, the  $p(\underline{x}_k | \underline{x}_k)$  can at least be determined to within a multiplicative constant. It is impossible in most instances to compute moments or expected values of particular quantities (e.g. cost functions) in a closed form, so that the estimation problem can not actually be solved through knowledge of the *c* posteriori density function. Again the major exception occurs when the system is linear and gaussian. This case is treated later in this section where it is seen that the Kalman filter equations describe the mean and covariance of the gaussian a posteriori density function.

To circumvent the difficulties described in items (ii) and (iii), methods for approximating the density function have been proposed<sup>3, 10</sup>. These aspects are beyond the scope of this chapter.

The recursive a posteriori density function relations for time-discrete systems have their analog for timecontinuous systems. Stratonovich<sup>11</sup> first derived a partial differential equation to describe the evolution of p(g(t)|Z(t)). Subsequently, Kushner<sup>12</sup> and, then Eucy<sup>13</sup> modified these results consistent with the Ito stochastic calculus. These equations for the *a posteriori* density are very difficult to solve. Linear, gaussian systems are again the principal case for which solutions can be obtained. Fisher<sup>14</sup> has attempted to obtain approximate solutions.

2.3 Linear, Gaussian Systems

Suppose that the plant and measurement data are described by linear equations so that the system is

$$\underline{x}_{k} = \Psi_{k, k-1} \underline{x}_{k-1} + \underline{y}_{k-1}$$
(2.17)

$$\underline{\mathbf{z}}_{\mathbf{k}} = \mathbf{H}_{\mathbf{k}} \underline{\mathbf{z}}_{\mathbf{k}} + \underline{\mathbf{Y}}_{\mathbf{k}} \qquad (2.18)$$

where the initial state is a gaussian random variable with density function

$$p(\underline{x}_0) = k_0 \exp\{-\frac{1}{2} (\underline{x}_0 - \underline{s})^T \, \underline{u}_0^{-1} (\underline{x}_0 - \underline{s})\}$$
(2.19)

and the plant and measurement noise sequences are gaussian white-noise with density functions defined by (2.11) and (2.14).

For this linear, gaussian system the a posteriori density function is characterized by the following result.

Theorem 2.5

The a posteriori density  $p(\underline{x}_k | \underline{z}_k)$  for the system (2.17), (2.18) is gaussian

$$p(\underline{\mathbf{z}}_{\underline{k}} | \underline{\mathbf{z}}_{\underline{k}}) = [(2\pi)^{n} | \underline{\mathbf{P}}_{\underline{k}} |]^{-1/2} \exp\{-\frac{1}{2} (\underline{\mathbf{x}}_{\underline{k}} - \underline{\mathbf{z}}_{\underline{k}})^{T} \underline{\mathbf{P}}_{\underline{k}}^{-1} (\underline{\mathbf{x}}_{\underline{k}} - \underline{\mathbf{z}}_{\underline{k}})\}, \qquad (2.20)$$

with mean value given by

$$\mathbf{\hat{z}}_{k} = \mathbf{\hat{z}}_{k} + \mathbf{x}_{k} (\mathbf{z}_{k} - \mathbf{H}_{k} \mathbf{\hat{z}}_{k}^{\prime}) . \qquad (2.21)$$

The  $\frac{2}{k}$  represents the mean value of the *a posteriori* density  $p(\underline{x}_k | \underline{Z}_{k-1})$  and is

$$\mathbf{\hat{x}}_{k}' = \mathbf{E}[\mathbf{x}_{k} | \mathbf{x}_{k-1}] = \mathbf{\Phi}_{k, k-1} \mathbf{\hat{x}}_{k-1} . \tag{2.22}$$

The matrix  $P'_k$  is the covariance of  $p(\mathbf{x}_k | \mathbf{Z}_{k-1})$  and is

 $\mathbf{P}_{\mathbf{k}}' = \mathbf{E} \left[ \left( \underline{\mathbf{z}}_{\mathbf{k}} - \underline{\mathbf{z}}_{\mathbf{k}}' \right) \left( \underline{\mathbf{z}}_{\mathbf{k}} - \underline{\mathbf{z}}_{\mathbf{k}}' \right)^{T} | \underline{\mathbf{z}}_{\mathbf{k}-1} \right] = \Phi_{\mathbf{k}, \mathbf{k}-1} \mathbf{P}_{\mathbf{k}-1} \Phi_{\mathbf{k}, \mathbf{k}-1}^{T} + \mathbf{Q}_{\mathbf{k}-1} .$ (2.23)

14 PANE 79

K. is a gain matrix defined by

$$K_{k} = P_{k}^{\prime} H_{k}^{T} (H_{k} P_{k}^{\prime} H_{k}^{T} + R_{k})^{-1}$$
(2.24)

The covariance matrix of  $p(\underline{x}_k | \underline{Z}_k)$  is defined as  $\mathbb{P}_k$  and is given by

$$\mathbf{P}_{k} = \mathbf{E}[(\mathbf{x}_{k} - \mathbf{\hat{x}}_{k})(\mathbf{x}_{k} - \mathbf{\hat{x}}_{k})^{T} | \mathbf{Z}_{k}] = \mathbf{P}_{k}' - \mathbf{K}_{k} \mathbf{H}_{k} \mathbf{P}_{k}'. \qquad (2.26)$$

Assuming the first measurement occurs at  $t_0$ , the initial density is  $p(\chi_0)$ ; so

$$\hat{\mathbf{z}}_{0} = \mathbf{s} + \mathbf{K}_{0} (\mathbf{z}_{0} - \mathbf{H}_{0} \mathbf{s}) , \qquad (2.26)$$

whore

and

$$\mathbf{P}_{o} = \mathbf{M}_{o} - \mathbf{K}_{o} \mathbf{H}_{o} \mathbf{M}_{o} \qquad (2.27)$$

Note: The equations describing the conditional mean and covariance are identical with the Kalman filter equations. This is not unexpected, since it has been shown that the conditional mean provides the minimum variance estimate. Thus in the linear case the Bayesian approach yields the results presented in Chapter 1. It should also be noted that the covariance of the conditional density is independent of the measurement data. As a result the conditional covariance is identical with the covariance of the error in the estimate. That is,

 $\mathbf{K}_{n} = \mathbf{M}_{n}\mathbf{H}_{n}^{\mathsf{T}}(\mathbf{H}_{n}\mathbf{M}_{n}\mathbf{H}_{n}^{\mathsf{T}} + \mathbf{R}_{n})^{-1}$ 

$$\mathbb{P}_{k} = \mathbb{E}\left[\left(\underline{x}_{k} - \underline{x}_{k}\right)\left(\underline{x}_{k} - \underline{x}_{k}\right)^{T}\right] \mathbb{Z}_{k}\right] = \mathbb{E}\left[\left(\underline{x}_{k} - \underline{x}_{k}\right)\left(\underline{x}_{k} - \underline{x}_{k}\right)^{T}\right]$$

It is interesting that the proof<sup>3</sup> of the theorem is most easily accomplianed by resorting to the use of obsracteristic functions. This approach is taken here. Certainly, direct evaluation of the general recursion relations (2.9) and (2.10) will provide an equivalent result, but it is interesting to note that it is the result stated in Section 2.1.4 of Chapter 1 as an alternative form. This occurs because the density functions involve the inverse of the covariance matrix; so one obtains  $P_k^{-1}$  ruther than  $P_k$  when proceeding directly from (2.9) and (2.10). On the other hand characteristic functions require the covariance matrix; so one obtains  $P_k^{-1}$  ruther than  $P_k$  when proceeding directly from (2.9) and (2.10). On the other hand characteristic functions require the covariance matrices and not their inverses, so this appears to have several advantages. Kalman pointed out that it is more satisfying to deal with the characteristic function since then  $M_0$ ,  $Q_k$ ,  $R_k$  need not be assumed to be positive-definite.

Before proceeding with the proof some characteristic function relations shall be stated.

(i) The characteristic function  $\phi$  and the probability density function p associated with a random variable  $\chi$  form a Fourier transform pair

 $\phi(\underline{\mathbf{g}}) \stackrel{\Delta}{=} \mathbb{E}[\exp(i\underline{\mathbf{g}}^{\mathsf{T}}\underline{\mathbf{g}})] = \int \exp(i\underline{\mathbf{g}}^{\mathsf{T}}\underline{\mathbf{g}}) p(\underline{\mathbf{g}}) d\underline{\mathbf{g}}$ (2.28)

$$p(\underline{x}) \stackrel{\text{\tiny{destring}}}{=} (2\pi)^{-n} \int \exp(-i\underline{x}^T \underline{x}) \phi(\underline{x}) d\underline{x} , \qquad (2.29)$$

(ii) It is also useful to recognize that

$$(2\pi)^{-n} \int \exp\left[i\underline{p}^{T}\underline{x}\right] d\underline{x} = \delta(\underline{n}) , \qquad (2.30)$$

where  $\delta(\cdot)$  is the Dirac delta function.

(iii) It can be proven that

$$\int_{-\infty}^{\infty} \exp\left[n^{T}g - g^{T}Ag\right] dg = \pi^{n/2} |A|^{-1/2} \exp\left[\frac{1}{4} g^{T} A^{-1}g\right]$$
(2.31)

for any complex h and positive-definite A .

From the definitions it follows that the characteristic function for  $\underline{x}_0$  is

$$\phi(\underline{s}_{m}) \simeq \exp\left(i\underline{s}_{m}^{T} \underline{s}_{m} - \frac{1}{2} \underline{s}_{m}^{TH} \underline{s}_{m}\right), \qquad (2, 32)$$

For  $p(\underline{x}_k | \underline{x}_k)$  and  $p(\underline{x}_{k+1} | \underline{x}_k)$  the characteristic functions are

$$b(\underline{\mathbf{s}}_{\mathbf{y}}) = \exp\{\lim_{\mathbf{k}} \frac{\mathbf{T}}{\mathbf{k}_{\mathbf{y}}} + \frac{1}{\mathbf{s}}_{\mathbf{y}}^{\mathbf{R}} + \frac{1}{\mathbf{s}}_{\mathbf{x}}^{\mathbf{R}} + \frac{1}{\mathbf{s}}_{\mathbf{x}}^{$$

$$p(\underline{s}_{\mu}) = \exp\{i\underline{s}_{\mu}^{T} \Phi_{\mu+1, \mu} \underline{s}_{\mu} - \frac{1}{2} \underline{s}_{\mu}^{T} \Theta_{\mu} \underline{s}_{\mu}\}, \qquad (2.34)$$

To accomplish the proof, first observe that the characteristic function of  $p(\underline{x}_k | \underline{x}_k)$  is

 $\phi(\underline{\mathbf{z}}_{k}) = \int \exp(i\underline{\mathbf{z}}_{k}^{\mathsf{T}}\underline{\mathbf{z}}_{k})\rho(\underline{\mathbf{z}}_{k}|\underline{\mathbf{z}}_{k}) d\underline{\mathbf{z}}_{k}$ 

$$= \frac{1}{(2\pi)^{n+n}p(\underline{a}_{k}|\underline{Z}_{k-1})} \int \exp\left[-i(\underline{a}_{k/k-1} - \underline{a}_{k})^{T} \underline{a}_{k} - i\underline{u}_{v}^{T}\underline{a}_{k}\right] \phi(\underline{a}_{k/k-1})\phi(\underline{a}_{v}) d(\underline{a}_{k},\underline{a}_{v},\underline{a}_{k/k-1}) \quad (2.35)$$





(2.36)

Note also that the characteristic function of  $p(\underline{x}_k | \underline{x}_{k-1})$  is

$$\begin{split} \phi(\underline{a}_{V/k-1}) &= \int \exp\left[i\underline{a}_{k/k-1}^{T}\underline{\lambda}_{k}\right] p_{i}(\underline{a}_{k}|\underline{Z}_{k-1}) \ d\underline{\lambda}_{k} \\ &= \frac{1}{(2\pi)^{2n}} \int \exp\left[-i\left(\underline{a}_{W} - \underline{a}_{k/k-1}\right)^{T} \underline{\lambda}_{k} - i\underline{a}_{k-1}^{T}\underline{\lambda}_{k-1}\right] \times \\ &\times \phi(\underline{a}_{W}) \phi(\underline{a}_{k-1}) d(\underline{Z}_{V-1}, \underline{\lambda}_{k}, \underline{a}_{W}, \underline{a}_{k-1}) \end{split}$$

Using the above relations, the proof follows in a straightforward manner.

Proof: The initial conditions (2.26) and (2.27) will be established. Use (2.32) and (2.33) in (2.35). Then

$$\begin{aligned} \phi(\underline{s}_0) &= \frac{1}{(2\pi)^{\mathbf{n}+\mathbf{m}}p(\underline{z}_0)} \int \exp\left[-\mathbf{i}(\underline{s}_{\mathbf{m}} - \underline{s}_0 - \mathbf{H}_0^{\mathsf{T}}\underline{s}_{\mathbf{v}})^{\mathsf{T}} \underline{x}_0 - \right. \\ &- \mathbf{i}\underline{s}_{\mathbf{v}}^{\mathsf{T}}\underline{s}_0 + \mathbf{i}\underline{s}_{\mathbf{m}}^{\mathsf{T}}\underline{s}_0 - \frac{1}{2}\underline{s}_{\mathbf{m}}^{\mathsf{T}}\mathbf{R}_0\underline{s}_{\mathbf{v}}\right] \, \mathbf{d}(\underline{s}_{\mathbf{m}},\underline{s}_{\mathbf{v}},\underline{x}_0) \end{aligned}$$

Integrate with respect to In and use (2.30). Then one obtains

$$\begin{split} \phi(\underline{\mathbf{x}}_{0}) &= \frac{1}{(2\pi)^{\mathbf{H}}_{\mathbf{D}}(\underline{\mathbf{z}}_{0})} \int S(\mathbf{H}_{0}^{T} \underline{\mathbf{x}}_{\mathbf{v}} + \underline{\mathbf{z}}_{0} - \underline{\mathbf{x}}_{\mathbf{u}}) \exp\left[-i\underline{\mathbf{z}}_{\mathbf{v}}^{T} \underline{\mathbf{z}}_{0} + i\underline{\mathbf{z}}_{\mathbf{u}}^{T} \underline{\mathbf{x}}_{0} - \frac{1}{2} \underline{\mathbf{z}}_{\mathbf{u}}^{T} \mathbf{M}_{0} \underline{\mathbf{z}}_{\mathbf{u}} - \frac{1}{2} \mathbf{z}_{\mathbf{u}}^{T} \mathbf{M}_{0} \underline{\mathbf{z}}_{\mathbf{u}} + \frac{1}{2} \mathbf{z}_{\mathbf{u}}^{T} \mathbf{u}_{0} \mathbf{z}_{\mathbf{u}} \right] \mathbf{u} \left( \underline{\mathbf{z}}_{\mathbf{u}}, \underline{\mathbf{z}}_{\mathbf{v}} \right) \; . \end{split}$$

Because of the delts function integration with respect to  $\underline{r}_m$  is trivial, so this reduces to

$$\begin{aligned} \varphi(\underline{\mathbf{n}}_{0}) &= \frac{1}{(2\pi)^{\mathbf{M}} p(\underline{\mathbf{n}}_{0})} \exp[\underline{\mathbf{i}} \underline{\mathbf{n}}_{0}^{\mathsf{T}} \pm \underline{\mathbf{i}}_{0}^{\mathsf{T}} \mathbf{M}_{0} \underline{\mathbf{n}}_{0}] \exp[\underline{\mathbf{n}}_{V}^{\mathsf{T}} (-\mathbf{i} (\underline{\mathbf{n}}_{0} - \mathbf{H}_{0} \underline{\mathbf{n}}) - \mathbf{H}_{0} \mathbf{M}_{0} \underline{\mathbf{n}}_{0}] - \\ &- \frac{1}{2} \underline{\mathbf{n}}_{V}^{\mathsf{T}} (\mathbf{H}_{0} \mathbf{M}_{0} \mathbf{H}_{0}^{\mathsf{T}} + \mathbf{h}_{0}) \underline{\mathbf{n}}_{V} \rangle d\underline{\mathbf{n}}_{V} . \end{aligned}$$

Using (2.31) and evaluating  $p(g_0)$  , it follows that

$$\varphi(\underline{a}_{0}) = \exp\{i\underline{a}_{0}^{T}[\underline{a}_{0} + K_{0}(\underline{a}_{0} - H_{0}\underline{a})] - \frac{1}{2}\underline{a}_{0}^{T}[M_{0} - K_{0}H_{0}M_{0}]\underline{a}_{0}\}, \qquad (2.37)$$

But this is the characteristic function of a gaussian variable with mean and covariance described by (2.28) and (2.27).

To verify (2.22) and (2.23), assume that the lemma is true for  $t_{k-1}$  and form  $\phi(\underline{a}_{k/k-1})$ , using (2.36). This follows in a straightforward manner as does the proof of (2.21), (2.24), and (2.25), using (2.36). The details are unitted.

The use of characteristic functions has eliminated the requirement that the *a priori* distributions have nonsingular covariance matrices. In addition, the development leads directly to the equations presented in the theorem which have come to be known as the Kalman filter equations. As an alternative, it is possible to derive the density relations directly from (2.9) and (2.10). The direct application of these equations leads to an equivalent set of equations under the assumption that all of the *a priori* covariance matrices are nonsingular but which have some important differences. The density is, of course, still described by (2.20) and the mean values given by (2.21) and (2.22) are unchanged, as is the covariance  $P'_k$  described by (2.23). However, the covariance and gain take the following form

$$\mathbf{P}_{k}^{-1} = (\mathbf{P}_{k}^{\prime})^{-1} + \mathbf{H}_{k}^{T} \mathbf{R}_{k}^{-1} \mathbf{H}_{k}$$
(2.38)

$$\mathbf{K}_{\mathbf{k}} = \mathbf{P}_{\mathbf{k}} \mathbf{H}_{\mathbf{k}}^{\mathsf{T}} \mathbf{R}_{\mathbf{k}}^{\mathsf{T}} \,. \tag{2.39}$$

Of importance here is the observation that the  $n \times n$  matrix  $P_k$  must be inverted, rather than the  $m \times m$  matrix  $(H_k P_k^{\dagger} H_k^{\dagger} + R_k)$  appearing in (2.24). Since m may be significantly smaller than n, this is an important computational consideration. It is shown in Section 2.4 of Chapter 1 that the two representations are related through the use of a matrix inversion lemma.









# 3. MAXIMUM A POSTERIORI AND MAXIMUM LINELIHOOD ESTIMATES

In the preceding section, recursive relations for the a posteriori density function were developed which led in a natural way to the recursive Kalman filter equations when the system was assumed to be linear and gaussian. In this section the density functions will not be written in a recursive fashion and the considerations will lead to the maximum a posteriori estimution procedure. When the system is assumed to be linear, the Kalman equations are obtained again, as must be the case because of the Theorem 2.4. This discussion can still be considered to be Bayesian since the results are obtained using Bayes's rule to describe the a posteriori density. Again, the density relations will be non-recursive and this aspect constitutes the primary difference from the discussion of Section 2.

# 3.1 The Maximum a posteriori Estimation Problem

Consider the collection of all states  $\underline{X}_k$  (i.e.  $\underline{x}_0, \underline{y}_1, \dots, \underline{x}_k$ ) and all measurement data  $\underline{Z}_k$  which are described in general by (1.20) and (1.21) and suppose that  $\underline{X}_k$  and  $\underline{Z}_k$  have the joint density  $p(\underline{X}_k, \underline{Z}_k)$ . Then, by Bayes's rule (2.4), it follows that

$$\nu(\underline{X}_{k}|\underline{Z}_{k}) = \frac{\nu(\underline{X}_{k},\underline{Z}_{k})}{\nu(\underline{Z}_{k})}$$
(3.1)

$$= \frac{p(\mathbf{Z}_k | \mathbf{X}_k) p(\mathbf{X}_k)}{p(\mathbf{Z}_k)} , \qquad (3.2)$$

But the measurement noise is independent between samples, so that

$$\mathfrak{p}(\mathbf{\overline{z}}^{\mathbf{k}}|\mathbf{\overline{z}}^{\mathbf{k}}) = \mathfrak{p}(\mathbf{\overline{z}}^{\mathbf{0}}|\mathbf{\overline{z}}^{\mathbf{0}})\mathfrak{p}(\mathbf{\overline{z}}^{\mathbf{1}}|\mathbf{\overline{z}}^{\mathbf{1}}) \cdots \mathfrak{p}(\mathbf{\overline{z}}^{\mathbf{k}}|\mathbf{\overline{z}}^{\mathbf{k}}) = \frac{1}{2} \mathfrak{p}(\mathbf{\overline{z}}^{\mathbf{1}}|\mathbf{\overline{z}}^{\mathbf{1}})$$
(3.3)

and (3, 2) becomes

$$p(\underline{\mathbf{x}}_{k}|\mathbf{z}_{k}) = \frac{p(\underline{\mathbf{x}}_{k})}{p(\underline{\mathbf{x}}_{k})}$$

$$(3.4)$$

Also, the definition of the plant and plant noise allows one to write

$$p(\underline{x}_{k}) = p(\underline{x}_{k}|\underline{x}_{k-1})p(\underline{x}_{k-1})$$

$$= p(\underline{x}_{k}|\underline{x}_{k-1})p(\underline{x}_{k-1}|\underline{x}_{k-2}) \dots p(\underline{x}_{1}|\underline{x}_{0})p(\underline{x}_{0})$$

$$= p(\underline{x}_{0}) \frac{1}{|\underline{x}_{1}|} p(\underline{x}_{1}|\underline{x}_{1-1}) , \qquad (3.5)$$

so that (3.4) can be rewritten as

$$p(\underline{x}_{k}|\underline{z}_{k}) = \frac{p(\underline{x}_{0}) \left[ \frac{|\underline{k}|}{|\underline{x}|} p(\underline{x}_{1}|\underline{x}_{1-1}) \frac{|\underline{k}|}{|\underline{x}|} p(\underline{x}_{1}|\underline{x}_{1}) \right]}{p(\underline{z}_{k})}$$
(3.6)

As pointed out in Section 2, the densities  $p(\underline{x}_i | \underline{x}_{i-1})$  and  $p(\underline{x}_i | \underline{x}_i)$  are defined by the plant and measurement noise characteristics and the equations describing the two systems. If the noise sequences enter nonlinearly, it can be very difficult to define these conditional densities in a tractable form. To circumvent this problem, suppose that the noise enters additively and is gaussian, as described by (2.11) - (2.14). Then Equation (2.15) and Equation (2.16) apply and (3.6) can be written more explicitly. In particular, it is soon that

$$P(\mathbf{Z}_{\mathbf{k}}|\mathbf{X}_{\mathbf{k}}) = C_{\mathbf{v}} \exp\left\{-\frac{1}{2}\sum_{i=0}^{k} \left[\mathbf{Z}_{i} - \mathbf{h}_{i}(\mathbf{x}_{i})\right]^{\gamma} \mathbf{x}_{\lambda}^{-1} \left[\mathbf{Z}_{i} - \mathbf{h}_{i}(\mathbf{x}_{i})\right]\right\}$$
$$\stackrel{2}{=} C_{\mathbf{v}} \exp\left\{-\frac{1}{2}\sum_{i=0}^{k} \left\|\mathbf{Z}_{i} - \mathbf{h}_{i}(\mathbf{x}_{i})\right\|_{\mathbf{x}_{i}^{-1}}^{2}\right\}, \qquad (3.7)$$

$$p(\underline{x}_{k}) = p(\underline{x}_{0}) C_{w} \exp\left\{-\frac{1}{2} \sum_{i=1}^{k} \| \underline{x}_{i} - \underline{f}_{i}(\underline{x}_{i-1}) \|_{q_{k-1}^{-1}}^{2} \right\}.$$
 (3.8)



Asymptic that  $\underline{x}_0$  is gaussian, (3.6) becomes

$$p(\mathbf{x}_{k}|\mathbf{z}_{k}) = 0 \exp\left\{-\frac{1}{k}\left[\|\mathbf{x}_{0} - \mathbf{a}\|_{\mathbf{w}_{0}^{-1}}^{2} + \sum_{i=1}^{k} \|\mathbf{x}_{i} - \mathbf{x}_{i}(\mathbf{x}_{i-1})\|_{\mathbf{u}_{i-1}^{-1}}^{2} + \sum_{i=0}^{k} \|\mathbf{x}_{i} - \mathbf{h}_{i}(\mathbf{x}_{i})\|_{\mathbf{u}_{i-1}^{-1}}^{2}\right]\right\}, \quad (3.9)$$

## where O is a normalizing constant.



$$\mathbf{L}_{\mathbf{k}} = \frac{1}{2} \left\| \mathbf{x}_{0} - \mathbf{g} \right\|_{\mathbf{H}_{0}^{-1}}^{2} + \frac{1}{2} \sum_{i=1}^{2} \left\| \mathbf{x}_{i} - \mathbf{f}_{i}(\mathbf{x}_{i-1}) \right\|_{\mathbf{q}_{i-1}^{-1}}^{2} + \frac{1}{2} \sum_{i=0}^{2} \left\| \mathbf{x}_{i} - \mathbf{h}_{i}(\mathbf{g}_{i}) \right\|_{\mathbf{H}_{1}^{-1}}^{2}; \quad (3.10)$$

80

ALANS PROPERTY

ĺ.

ŝ

# $L_k = \log_{e} C - \log_{e} p(\underline{X}_k | \underline{Z}_k)$ .

The  $L_{\rm R}$  is to be minimized through the selection of the estimates. Equation (3.10) can be put into a form that is similar to that for optimal control problems so that the theory developed to solve those problems can be applied.

As stated, the estimation problem has been reduced to a deterministic minimization problem. Define variables  $y_{i-1}$  so that the plant can be considered as

$$k = f_k(x_{k-1}) + u_{k-1}$$
 (3.11)

This equation is similar to (2.11) but the  $y_{k-1}$  have been introduced to emphasize that, unlike the  $y_{k-1}$ , they are not random variables. Now consider the estimation problem as the following.

Choose the sequences  $\underline{X}_k$  and  $\underline{U}_{k-1}$  so that the cost function

$$u_{N} = \frac{1}{2} \left\| z_{0} - \mathbf{n} \right\|_{\mathbf{M}_{0}^{-1}}^{2} = \frac{1}{2} \sum_{i=0}^{N} \left\| z_{i} - \underline{h}_{i}(\mathbf{x}_{i}) \right\|_{\mathbf{M}_{1}^{-1}}^{2} + \frac{1}{2} \sum_{i=0}^{N-1} \left\| u_{i} \right\|_{\mathbf{u}_{1}^{-1}}$$
(3.12)

is minimized subject to the constraint

$$\underline{x}_{k} = \underline{f}_{k}(\underline{x}_{k-1}) + \underline{y}_{k-1}, \quad k = 1, 2, \dots, N-1,$$
 (3.13)

One can attend t to solve this problem using the mathematical formalism of cytimal control theory. It is not generally possible for arbitrary functions  $f_k$  and  $h_k$  but can be achieved when the system is linear. The general problem is beyond the scope of this chapter but the linear problem will be considered below.

It should be observed that the problem that has been powed is identical to the deterministic least-squares problem formulated in Section 1. This indicates that this deterministic problem has been imbedded in a probabilistic framework in which the errors and uncertainties have been assumed to be gaussian random variables and white-noise sequences. Thus, the deterministic and probabilistic problems are not fundamentally different although the language and analysis procedures are very dissimilar. This formulation has been suggested by  $Cox^{16}$ for time-discrete systems.  $Cox^{16}$  also considered time-continuous systems in which the summations of (3, 12) are replaced by integrations and the difference Equation (3, 13) is replaced by a differential equation. Datchmendy and Bridhar<sup>17</sup> also considered the nonlinear estimation problem in this manner and derived a system of equations that are similar to those of Cox. More recently, Mortensen<sup>16</sup> has considered the time-continuous problem and dura and Henrikson<sup>19</sup> have considered the geveralized least-squares procedure of Chapter 1 and have obtaind and extended the results of Detchmendy and Sridhar.

#### 3.3 Maximum Likelihood Estimation Theory

The maximum likelihood procedure that has been widely used is closely related to the maximum a posteriori estimates of the preceding section. To discuss the difference, consider Equation (3,2). The denominator  $p(Z_k)$  is a normalizing constant and can be ignored. In the maximum a posteriori procedure, the estimate is determined that maximizes  $p(X_k | Z_k)$ . Suppose that the a priori information relating to the  $X_k$ , as described by  $p(X_k)$ . does not suggest that any state is more likely than any other. Then, the  $p(X_k)$  will be a uniform density and the maximization of  $p(X_k | Z_k)$  will actually be determined by the maximization of  $p(Z_k | Z_k)$ . The values of  $X_k$  which are found to maximize  $p(Z_k | X_k)$  are referred to as maximum lkclihood estimates and  $p(Z_k | X_k)$  is referred to as the likelihood function.

Maximum likelihood estimates have several properties which are desirable and have therefore seen widespread application. For the discussion immediately below, some of these aspects are considered and will be related to some of the properties of the Kalman filter. For a more complete discussion the reader is referred to Cramer<sup>30</sup> or van Trees<sup>9</sup>.









The covariance of the error in may estimate can be bounded below through a relatively straightforward application of the Schwarz inequality. This bound is usually referred to as the Cramer-Rao inequality. In the following the Cramer-Rao inequality is proved for unbiased (see van Trees<sup>4</sup> for a treatment for biased estimates) estimates of an unknown vector parameter and then for a random parameter. Suppose that it is desired to estimate an unknown parameter  $\chi$  from measurement data  $\chi_N$ . The parameter and the data are assumed to be related according to

$$\mathbf{z}_{k} = \mathbf{h}_{k}(\mathbf{x}, \mathbf{Y}_{k})$$
,  $\mathbf{k} = 1, 2, \dots, N$ , (3.14)

where the noise  $\underline{x}_k$  has a known distribution. As a result, the conditional density  $p(Z_N | \underline{x})$  can be determined.

# Theorem 3.1 (Gramer-Ruo Inequality)

If  $\underline{s}$  is any unbiased estimate of  $\underline{x}$  based on the measurement data  $\underline{z}_N$ , then the conditional covariance of the error in the estimate given  $\underline{x}$  is bounded below by the inverse, assuming it exists, of the Fisher information matrix J.

$$\mathbb{E}[(\underline{x} - \underline{x})(\underline{x} - \underline{x})^{T}]\underline{x}] \ge J^{-1}, \qquad (3.10)$$

where

$$= E\left\{ \begin{bmatrix} \frac{\partial}{\partial \mathbf{x}} & \log_{\mathbf{0}} \mathbf{p} \left( \mathbf{Z}_{N} \mid \mathbf{x} \right) \end{bmatrix}^{\mathsf{T}} \begin{bmatrix} \frac{\partial}{\partial \mathbf{x}} & \log_{\mathbf{0}} \mathbf{p} \left( \mathbf{Z}_{N} \mid \mathbf{x} \right) \end{bmatrix} \mid \mathbf{x} \right\}$$
(3.16)

$$= -\Pi \left\{ \frac{\partial}{\partial \mathbf{x}} \left[ \frac{\partial}{\partial \mathbf{x}} \left[ \frac{\partial}{\partial \mathbf{x}} \left[ \mathbf{x}_{\mathbf{N}} \right]^{\mathsf{T}} \right] \mathbf{x} \right\} \right\}$$
(3.17)

Equality holds in (3.15) if, and only if,

$$\frac{\partial \mathbf{x}}{\partial t} \log_{\mathbf{e}} p(\mathbf{Z}_{N} | \mathbf{x}) = k(\mathbf{x}) (\mathbf{x} - \mathbf{x})^{T}$$
(3.18)

It is also assumed that

and

exist and are absolutely integrable.

Proof: First, consider that

$$\mathbb{E}\left[\left(\underline{x} - \underline{x}\right) | \underline{x}\right] = \int (\underline{x} - \underline{x}) p(\underline{z}_{N} | \underline{x}) d\underline{z}_{N} = 0,$$

since  $\underline{f}$  is unbiased. Differentiate both sides with respect to  $\underline{x}$  to obtain

$$\mathbf{o} = \int \frac{\partial}{\partial \mathbf{x}} \left[ (\mathbf{g} - \mathbf{x}) \mathbf{p} (\mathbf{Z}_{N} | \mathbf{x}) d\mathbf{Z}_{N} - \left[ -\int \mathbf{p} (\mathbf{Z}_{N} | \mathbf{x}) d\mathbf{Z}_{N} \right] \mathbf{I} + \int (\mathbf{g} - \mathbf{x}) \frac{\partial \mathbf{p} (\mathbf{Z}_{N} | \mathbf{x})}{\partial \mathbf{x}} d\mathbf{Z}_{N} .$$

But observe that

$$\frac{\partial p(\mathbf{Z}_{N} \mid \mathbf{X})}{\partial \mathbf{g}} = \begin{bmatrix} \frac{\partial}{\partial \mathbf{X}} & \log \mathbf{g} \\ \frac{\partial}{\partial \mathbf{x}} & \log \mathbf{g} \\ \frac{\partial}{\partial \mathbf{x}} & \log \mathbf{g} \end{bmatrix} p(\mathbf{Z}_{N} \mid \mathbf{X}) ,$$

so one obtains

$$\mathbf{I} = \int (\mathbf{\hat{z}} - \mathbf{x}) \left[ \frac{\partial}{\partial \mathbf{x}} \log_{\mathbf{p}} p(\mathbf{Z}_{N} | \mathbf{x}) \right] p(\mathbf{Z}_{N} | \mathbf{x}) d\mathbf{Z}_{N}$$
  
$$= \int \left\{ (\mathbf{\hat{z}} - \mathbf{x}) \left[ p(\mathbf{Z}_{N} | \mathbf{x}) \right]^{1/2} \right\} \left\{ \left[ \frac{\partial}{\partial \mathbf{x}} \log_{\mathbf{p}} p(\mathbf{Z}_{N} | \mathbf{x}) \right] \left[ p(\mathbf{Z}_{N} | \mathbf{x}) \right]^{1/2} \right\} d\mathbf{Z}_{N} ...$$

Applying the Schwarz inequality, it follows that

 $\mathbf{I} \leq \left( \int (\underline{\mathbf{I}} - \underline{\mathbf{x}}) (\underline{\mathbf{x}} - \underline{\mathbf{x}})^{\mathrm{T}} p (\mathbf{Z}_{\mathrm{N}} | \underline{\mathbf{x}}) \, \mathrm{d} \mathbf{Z}_{\mathrm{N}} \right) \, \mathbf{J} \, ,$ 

The matrix inequality A C B means that (B-A) 13 non-negative-definite.

unia 🕐 un sen niniti 🗨 al à bai

(3.19)

where J is defined by (D.16). If  $J^{-1}$  exists, then (3.15) is proven. It remains to domonstrate the second form for the Fisher information matrix. Equality holds in (3.19) if and only if

$$\frac{\partial \mathbf{x}}{\partial \mathbf{y}} \log^{2} \mathbf{b}(\mathbf{x}^{\mathsf{N}} | \mathbf{x}) = \mathbf{y}(\mathbf{x})(\mathbf{x} - \mathbf{x})$$

By definition, it is true that

 $\int p(\mathbf{Z}_{N}|\mathbf{X}) d\mathbf{Z}_{N} = 1.$ 

Differentiate with respect to g . Then one obtains

$$\int \frac{\partial}{\partial x} p(Z_N | x) dZ_N = \int \left[ \frac{\partial}{\partial x} \log_{\mathbf{p}} p(Z_N | x) \right] p(Z_N | x) dZ_N = 0.$$

Differentiating a second time with respect to  $\underline{x}$ , it is found that

$$\frac{\delta}{2} \int \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \right] \right]^{2} \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \right] \right]^{2} \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \right] \left[ \frac{\delta}{2} \right] \right]^{2} \right] \right]^{2} \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \right] \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \right] \left[ \frac{\delta}{2} \right] \left[ \frac{\delta}{2} \right] \left[ \frac{\delta}{2} \right] \right]^{2} \right] \right]^{2} \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \right] \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \right] \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \right] \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \right] \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \left[ \frac{\delta}{2} \right] \left[ \frac{\delta}{2} \left[ \frac{\delta}{2}$$

or

$$\mathbf{0} = \int \frac{\partial}{\partial \mathbf{x}} \frac{\partial}{\partial \mathbf{x}} \log_{\mathbf{0}} \mathbf{p}(\mathbf{Z}_{\mathsf{N}} | \mathbf{x}) \frac{\partial}{\partial \mathbf{x}} (\mathbf{Z}_{\mathsf{N}} | \mathbf{x}) \frac{\partial}{\partial \mathbf{x}} \frac{\partial}{\partial \mathbf{x}} \log_{\mathbf{0}} \mathbf{p}(\mathbf{Z}_{\mathsf{N}} | \mathbf{x}) \frac$$

Equation (7.17) follows immediately and the proof is complete. Note that the conditions on the partial derivatives are required so that the interchanges of differentiation and integration are valid.

This theorem and the Fisher information matrix can be applied to the linear filtering problem and some interesting relationships are obtained. Consider the following examples.

Example 1: Suppose that the measurement data are linearly related to the parameter  $\chi$  .

1

$$H_{k} = H_{k} + \chi_{k}$$
, (3.20)

and assume that the noise is gaussian and samples are independent,

$$\mathbf{p}(\mathbf{X}_{\mathbf{k}}) = \mathbf{k}_{\mathbf{y}} \exp\{\frac{1}{2} \mathbf{X}_{\mathbf{k}}^{T} \mathbf{R}_{\mathbf{k}}^{T} \mathbf{X}_{\mathbf{k}}\},$$

Thus, the conditional density  $r(Z_N | \underline{z})$  is

$$p(\underline{Z}_{N}|\underline{x}) = k \exp\left\{-\frac{1}{2} \sum_{i=0}^{N} (\underline{z}_{i} - H_{i}\underline{x})^{T} X_{i}^{-1} (\underline{z}_{i} - H_{i}\underline{x})\right\},$$

where k is the appropriate normalization constant. It follows that

$$\log_{\bullet} p(\mathbf{Z}_{\mathsf{N}} | \mathbf{x}) = \log_{\bullet} \mathbf{k} - \frac{1}{2} \sum_{i=0}^{\mathsf{N}} (\mathbf{z}_{i} - \mathbf{H}_{i} \mathbf{z})^{\mathsf{T}} \mathbf{R}_{i}^{-1} (\mathbf{z}_{i} - \mathbf{H}_{i} \mathbf{z});$$

80

$$\frac{\partial}{\partial \mathbf{z}} \log_{\mathbf{p}} p(\mathbf{z}_{\mathsf{N}} | \mathbf{x}) = \sum_{i=0}^{\mathsf{N}} (\mathbf{z}_{i} - \mathbf{H}_{i}\mathbf{x})^{\mathsf{T}} \mathbf{R}_{i}^{-1} \mathbf{H}_{i}$$

and

$$\frac{\partial}{\partial x} \left[ \frac{\partial}{\partial x} \right]^{T} = - \sum_{i=0}^{N} H_{i}^{T} R_{i}^{-i} H_{i}.$$

Using (3.17), the Fisher information matrix is found to be

 $J = \sum_{i=0}^{N} H_{i}^{T} R_{i}^{-1} H_{i}$ 

and the covariance of the error in any unbiased estimate of x is bounded by

 $\mathbb{E}\left[\left(\underline{\mathbf{x}}-\underline{\mathbf{x}}\right)\left(\underline{\mathbf{x}}-\underline{\mathbf{x}}\right)^{\mathsf{T}}\right] \geq \left[\sum_{i=0}^{\mathsf{N}} \mathbf{H}_{i}^{\mathsf{T}}\mathbf{R}_{i}^{-1}\mathbf{H}_{i}\right]^{-1}$ 

if J<sup>-1</sup> exists.

(3.22)

(3.23)

(3. 21)

0 17835555

Example 2: As a minor variation on the preceding example, suppose that the parameter g is time-varying but astisfies the constraint that

$$\mathbf{x}_{k} = \Psi_{k,k-1} \mathbf{x}_{k-1} + \mathbf{k} = 1, 2, ..., N$$
. (3.24)

The monsurements are described by

$$\underline{a}_{\mathbf{k}} = \mathbf{H}_{\mathbf{k}} \underline{\mathbf{x}}_{\mathbf{k}} + \underline{\mathbf{y}}_{\mathbf{k}} \,. \tag{3.25}$$

Solving (3.24) for the initial state  $\chi_0$  , one has

 $\Delta_k = \Phi_{k, \, u \, \overline{\lambda}_0} , \qquad (3.26)$ 

where

Assuming that  $\Phi_{k_1,k_{-1}}$  is nonsingular for all k , the measurement equation becomes

$$\mathbf{z}_{\mathbf{k}} = \mathbf{H}_{\mathbf{k}} \mathbf{\Phi}_{\mathbf{k},0} \mathbf{x}_{0} + \mathbf{y}_{\mathbf{k}} \,. \tag{3.27}$$

Equation (3.27) has essentially been reduced to the same form as (3.20), so it follows that the Fisher information matrix is

 $\Phi_{k,0} = \Phi_{k,k-1}\Phi_{k-1} + e_{1,0}$ 

$$J = \sum_{i=0}^{H} \Phi_{i,0}^{i} H_{i}^{i} R_{i}^{-1} H_{i} \hat{\varphi}_{i,0}$$
(3.28)

and the error covariance is bounded by

$$\mathbb{E}\left[\left(\underline{\mathbf{x}}-\underline{\mathbf{x}}\right)\left(\underline{\mathbf{x}}-\underline{\mathbf{x}}\right)^{\mathrm{T}}\right] \ge \left[\sum_{n=0}^{N} \Phi_{1,n}^{\mathrm{T}} H_{1}^{-1} H_{1} \Phi_{1,n}^{-1}\right]^{-1} .$$

$$(3.29)$$

This example is important because it shows that the Fisher information matrix is essentially identical with the observability matrix that was introduced in Section 3.1 of Chapter 1. Furthermore, the Gramer-Rao inequality (3.29) provides a lower bound that is very similar to the bound presented in Section 3.3 of Chapter 1.

Any estimate for which equality holds in (3.15) is said to be an *efficient* estimator. It has been seen that equality occurs if and only if

$$\frac{\partial x}{\partial x} \log^2 p(\underline{Z}^N | \underline{x}) = [\underline{x} - \underline{x}] k(\underline{x}) ,$$

It is easily shown that, if this condition can be satisfied, it can be accomplished by a maximum likelihood estimate.

The maximization of  $p(Z_N|g)$  can be replaced by the maximization of  $\log_p(Z_N|g)$ . Then a necessary condition that an estimate  $f_{m1}$  maximize  $\log_p(Z_N|g)$  is

$$\frac{\partial}{\partial x} \log_{\mathbf{p}} p(\mathbf{Z}_{N}|\mathbf{z})|_{\mathbf{x}=\mathbf{x}_{ml}} = 0 = (\mathbf{x}-\mathbf{x})\mathbf{k}(\mathbf{x})|_{\mathbf{x}=\mathbf{x}_{ml}}.$$
(3.30)

Thus, for the last relation to equal zero, either

 $\mathbf{1} = \mathbf{1}_{nl} \tag{3.31}$ 

or

Condition (3.31) is selected since it provides an estimate that depends upon the data; so, if an efficient estimate exists, it is a maximum likelihood estimate. If an efficient estimate does not exist, then there is no measure of the accuracy of the estimate. This constitutes a major drawback of this approach or maximum a *posteriori* estimates. Thus there may be unbiased estimates which yield a "smaller" error covariance when there is no efficient estimate.

k(1.) H 0.

In stating Theorem 3.1, it was assumed that an unknown parameter  $\underline{x}$  was to be treated. If  $\underline{x}$  is treated as a random variable, a similar result can be arrived which yields insight into the existence of efficient estimators. This development follows van Trees".

# Theorem 3.2

Suppose  $\underline{x}$  and  $\underline{Z}_{N}$  have joint density  $p(\underline{x},\underline{Z}_{N})$  and let  $\underline{x}$  be any unbiased estimate of  $\underline{x}$  based on the measurement data  $\underline{Z}_{N}$ . Then, assuming  $L^{-1}$  exists, the covariance of the error in the estimate is bounded below

 $\mathbb{E}[(\underline{x} - \underline{x}) (\underline{x} - \underline{x})^T] \ge L^{-1}, \qquad (3.32)$ 



i 137

(3.33)

 $L = \left\{ \begin{bmatrix} \frac{\partial}{\partial \underline{x}} & \log_{\bullet} p(\underline{x}, \underline{z}_{N}) \end{bmatrix}^{T} \begin{bmatrix} \frac{\partial}{\partial \underline{x}} & \log_{\bullet} p(\underline{x}, \underline{z}_{N}) \end{bmatrix} \right\}$ 

$$\pi = \mathbf{E} \left\{ \frac{\partial}{\partial \mathbf{x}} \begin{bmatrix} \partial \\ \partial \mathbf{x} \end{bmatrix} \mathbf{1} \mathbf{o} \mathbf{g}_{\mathbf{y}} p \left( \mathbf{x}, \mathbf{Z}_{\mathbf{N}} \right) \end{bmatrix}^{\mathsf{T}} \right\}.$$
(3.34)

Equality holds if and only if

$$\frac{\partial \mathbf{x}}{\partial t} \log_{\mathbf{e}} \mathbf{p}(\mathbf{x}, \mathbf{Z}_{N}) = \mathbf{k}(\mathbf{x}, -\mathbf{x})^{\mathrm{T}}. \qquad (3.35)$$

The first and second partial derivatives

$$\frac{\partial \mathbf{p}}{\partial \mathbf{x}} \left( \underline{x}, \underline{z}_{\mathsf{N}} \right) \quad \text{and} \quad \frac{\partial}{\partial \underline{x}} \left[ \frac{\partial}{\partial \underline{x}} \mathbf{p} \left( \underline{x}, \underline{z}_{\mathsf{N}} \right) \right]$$

are assumed to exist and are absolutely integrable with respect to  $\underline{x}$  and  $\underline{z}_{N}$ . Also, it is assumed that

$$\lim_{X \to \infty} \underline{b}(\underline{x}) \underline{v}(\underline{x}) = 0 \tag{3.36}$$

$$\lim_{x \to 0} g(x) p(x) = 0, \qquad (3.37)$$

$$\frac{1}{2} \left[ \frac{1}{2} - \frac{1}{2} \right] p\left( \frac{1}{2} \right] \left[ \frac{1}{2} - \frac{1}{2} \right] p\left( \frac{1}{2} \right] \left[ \frac{1}{2} \right] d\mathbb{Z}_{H} , \qquad (3.38)$$

Proof: The proof is similar to that of Theorem 3.1, although expectations with respect to g must be considered also. First form p(x)b(x) and differentiate with respect to x. This yields

$$\frac{\mathrm{d}}{\mathrm{d} \mathbf{z}} \, \boldsymbol{p}(\mathbf{x}) \, \underline{\mathbf{h}}(\mathbf{x}) \; = \; - \int \boldsymbol{p}(\mathbf{x}, \mathbf{Z}_N) \, \mathrm{d} \mathbf{Z}_N \, \mathbf{1} \, + \int (\mathbf{x} - \mathbf{x}) \, \frac{\partial}{\partial \mathbf{x}} \, \boldsymbol{p}(\mathbf{x}, \mathbf{Z}_N) \, \mathrm{d} \mathbf{Z}_N \; .$$

Integrate with respect to 1 and invoke conditions (3.30) and (3.37) to obtain

$$\mathbf{D} = \mathbf{H} \mathbf{I} + \int (\mathbf{I} - \underline{\mathbf{x}}) \left[ \frac{\partial}{\partial \mathbf{x}} \log_{\mathbf{x}} \mathbf{p}(\mathbf{x}, \mathbf{Z}_{\mathsf{N}}) \right] \mathbf{p}(\mathbf{x}, \mathbf{Z}_{\mathsf{N}}) d(\mathbf{x}, \mathbf{Z}_{\mathsf{N}}) \ .$$

Application of the Schwarz inequality yields

$$\mathbf{I} \leq \left[\mathbf{E}\left(\left(\underline{\mathbf{x}} - \underline{\mathbf{x}}\right)\left(\underline{\mathbf{x}} - \underline{\mathbf{x}}\right)^{\mathsf{T}}\right)\right] \mathbf{L}_{\mathsf{T}}$$

where L is given by (3.33) and equality holds if and only if (3.35) is astisfied. Note that the averaging with respect to both  $\underline{x}$  and  $\underline{z}_{n}$  implies that, unlike the condition of Theorem 3.1. k is a constant independent of both z and Zn

The remainder of the proof is identical with that of Theorem 3.1.

Two results relating to the effect and consequence of treating g as a random parameter are worth noting. First, the matrix L as given by (3.34) can be rewritten as

$$L = J - E \left\{ \frac{\partial E}{\partial x} \left[ \frac{\partial E}{\partial x} \log_{\phi} p(E) \right]^{T} \right\}.$$
(3.39)

where J is defined in Theorem 3.1. Thus, the probabilistic description of x enters independently of the noise etatistics. This will be discussed further below.

The condition (3.35) for equality in (3.32) involves the constant k . Note that the condition can be written

$$\frac{\partial \mathbf{r}}{\partial \mathbf{x}} \left[ \log_{\mathbf{n}} p(\underline{\mathbf{x}} | \underline{\mathbf{z}}_{\mathbf{N}}) \right] \approx k(\underline{\mathbf{x}} - \underline{\mathbf{x}})^{\mathsf{T}} .$$

Integrate with respect to g and take the antilog to obtain

$$\mathbf{p}(\underline{\mathbf{x}}|\underline{\mathbf{Z}}_{\mathbf{N}}) = \exp\left[-\frac{\mathbf{k}}{2}\,\underline{\mathbf{x}}^{\mathsf{T}}\underline{\mathbf{x}} + \mathbf{k}\underline{\mathbf{x}}^{\mathsf{T}}\underline{\mathbf{x}} + \mathbf{o}\right] \,. \tag{3.40}$$

But this implies that the a posteriori density  $p(\underline{x}|Z_n)$  must be gaussian for an efficient estimate to exist.

It follows easily that a maximum a posteriori estimate will be efficient if an efficient estimate exists. However, since  $p(\underline{x};\underline{Z}_N)$  must be gaussian in this case, the minimum variance estimator yields the same result, so it is also efficient.













where

where

## Consider these results through the following example.

Example 3: Consider the system described in Example 2 with the additional requirement that  $\underline{x}_0$  is a gaussian random variable with density function

$$p(\underline{x}_{0}) = k_{0} \exp\{-\frac{1}{2}(\underline{x}_{0} - \underline{a})^{T} M_{0}^{-1}(\underline{x}_{0} - \underline{a})\}, \qquad (3.41)$$

Then, the information matrix L , as given by (3.38), is

$$\mathbf{L} = \mathbf{J} + \frac{1}{2} \mathbf{E} \left\{ \frac{\partial}{\partial \mathbf{x}} \left[ \frac{\partial}{\partial \mathbf{x}} \left[ \frac{\partial}{\partial \mathbf{x}} \left[ \mathbf{x}_0 - \mathbf{k} \right]^T \mathbf{M}_0^{-1} \left( \mathbf{x}_0 - \mathbf{k} \right) \right] \right\}$$
$$= \sum_{i=1}^{N} \Phi_{i,i}^T \mathbf{H}_i^{-1} \mathbf{H}_i^{-1} \mathbf{H}_i^{-1} \mathbf{h}_i^{-1}$$

and the error covariance is bounded below by

$$\mathbb{E} \left[ \left( \underline{\mathbf{x}}_{0} - \underline{\mathbf{x}}_{0} \right) \left( \underline{\mathbf{x}}_{0} - \underline{\mathbf{x}}_{0} \right)^{\mathsf{T}} \right] \geq \left[ \mathbb{M}_{0}^{-1} + \sum_{i=1}^{\mathsf{H}} \Phi_{i,0}^{\mathsf{T}} \mathbb{H}_{i}^{\mathsf{T}} \mathbb{H}_{i}^{-1} \mathbb{H}_{i} \Phi_{i,0} \right]^{-1}.$$

But this is essentially the bound discussed in Section 3.3 of Chapter 1. The right-hand side represents the error covariance given by the Kalman filter for this problem and provides another vorification that the minimum variance estimator is efficient for this system. To obtain precise agreement, note that the estimate of the terminal state  $\underline{x}_N$  is given by

$$\mathbf{\hat{z}}_{N} = \Phi_{N,o} \mathbf{\hat{z}}_{o}$$
.

It follows that the error covariance associated with  $f_{\rm N}$  is

$$\mathbb{E}[(\underline{x}_{\mathsf{N}}-\underline{x}_{\mathsf{N}})(\underline{x}_{\mathsf{N}}-\underline{x}_{\mathsf{N}})^{\mathsf{T}}] = \Phi_{\mathsf{H},0}\mathbb{E}[(\underline{x}_{\mathsf{O}}-\underline{x}_{\mathsf{O}})(\underline{x}_{\mathsf{O}}-\underline{x}_{\mathsf{O}})^{\mathsf{T}}] \Psi_{\mathsf{H},0}^{\mathsf{T}}.$$

. Using this relation, one obtains the result given in Section 3.3.2 of Chapter 1.

This completes the discussion of error bounds using the Cramer-Rao inequality and its generalizations.

## 3.3 Maximum a posteriori Estimates for Linear Systems

Suppose that the plant and messurement equations are linear, so that the maximum a posteriori estimation problem becomes that of choosing the sequences  $\underline{X}_N$  to minimize the cost function

$$\mathbf{L}_{\mathsf{N}} = \frac{1}{2} \left\| \mathbf{x}_{0} - \mathbf{g} \right\|_{\mathbf{H}_{0}^{-1}}^{2} + \frac{1}{2} \sum_{i=0}^{\mathsf{H}_{1}} \left\| \mathbf{x}_{i} - \mathbf{H}_{i} \mathbf{x}_{i} \right\|_{\mathbf{H}_{1}^{-1}}^{2} + \frac{1}{2} \sum_{i=0}^{\mathsf{H}_{-1}} \left\| \mathbf{u}_{i} \right\|_{\mathbf{H}_{1}^{-1}}^{2}, \qquad (3.42)$$

subject to the constraint

$$\underline{\mathbf{x}}_{\mathbf{k}} = \Phi_{\mathbf{k}, \mathbf{k}-1} \underline{\mathbf{x}}_{\mathbf{k}-1} + \underline{\mathbf{u}}_{\mathbf{k}-1}$$
(3.43)

The minimization is accouplished inductively in the following manner. Consider the cost function after the first measurement has been obtained.

$$L_{0} = \frac{1}{2} \left\| \frac{x_{0}}{x_{0}} - \frac{x_{0}}{x_{0}} \right\|_{H_{0}^{-1}}^{2} + \frac{1}{2} \left\| \frac{x_{0}}{x_{0}} - \frac{x_{0}}{x_{0}} \right\|_{H_{0}^{-1}}^{2}$$
(3.44)

As will be seen, this simple problem contains within it a major portion of the problem, since a recursive solution is to be obtained.

. Choose the  $\underline{x}_0$  that minimizes  $L_0$  and call it  $\underline{x}_0$ . A necessary and sufficient condition for this problem is that

$$\left(\frac{\partial L_0}{\partial \underline{x}_0}\right)^T = \mathbf{0} = \mathbf{M}_0^{-1} \left(\frac{\mathbf{g}}{\mathbf{x}_0} - \underline{\mathbf{x}}\right) - \mathbf{H}_0^T \mathbf{x}_0^{-1} \left(\underline{\mathbf{x}}_0 - \mathbf{H}_0 \underline{\mathbf{x}}_0\right) , \qquad (3.40)$$

Solve for 10 to obtain

$$|\mathbf{M}_{0}^{**} + \mathbf{H}_{0}^{*}\mathbf{R}_{0}^{*}\mathbf{H}_{0}|_{\mathbf{Z}_{0}}^{*} = \mathbf{M}_{0}^{**}\mathbf{B} + \mathbf{H}_{0}^{*}\mathbf{R}_{0}^{**}\mathbf{B}_{0} \qquad (3.46)$$

$$_{0} = (M_{0}^{-1} + H_{0}^{T}R_{0}^{-1}H_{0})^{-1} [M_{0}^{-1} + H_{0}^{T}R_{0}^{-1} H_{0}^{-1}].$$
(3.47)

or Let

1

: \ { \

$$P_{0} \stackrel{f}{=} (M_{0}^{-1} + H_{0}^{T} K_{0}^{-1} H_{0})^{-1}$$
(3.48a)

=  $M_0 - M_0 H_0^T (H_0 M_0 H_0^T + B_0)^{-1} H_0 M_0$ , (3.48b)

using the matrix inversion lemma.





/9 48al

Using (3,48b), (3,47) becomes

$$\mathbf{\hat{x}}_{0} = \mathbf{\underline{x}} + \mathbf{k}_{0} [\mathbf{\underline{x}}_{0} - \mathbf{H}_{0} \mathbf{\underline{x}}], \qquad (3.49)$$

$$\mathbf{K}_{0} \triangleq \mathbf{M}_{0}\mathbf{H}_{0}^{\mathsf{T}}(\mathbf{H}_{0}\mathbf{M}_{0}\mathbf{H}_{0}^{\mathsf{T}} + \mathbf{R}_{0})^{-1} \qquad (3.50)$$

Thus the maximum a posteriori estimate of  $\underline{x}_0$ , given  $\underline{x}$  and  $\underline{x}_0$ , is described by (3.49). As must be true, this agrees with the estimate derived in Soction 2 using the Bayesian recursion relations.

Suppose now that there are two measurements and the cost function  $L_1$  is given by

$$\mathbf{L}_{1} = \frac{1}{2} \| \mathbf{z}_{0} - \mathbf{g} \|_{\mathbf{W}_{0}^{-1}}^{2} + \frac{1}{2} \sum_{l=0}^{2} \| \mathbf{z}_{1} - \mathbf{H}_{1} \mathbf{z}_{1} \|_{\mathbf{h}_{1}^{-1}}^{2} + \frac{1}{2} \| \mathbf{u}_{0} \|_{\mathbf{H}_{0}^{-1}}^{2}, \qquad (3.51)$$

where the  $\underline{x}_0$  ,  $\underline{x}_1$  and  $\underline{u}_0$  are related by

 $\mathbf{X}_{i} = \Phi_{i, 0} \mathbf{X}_{0} + \mathbf{y}_{0} \, .$ 

The estimate  $\mathbf{1}_0$  that was derived above will be different than that obtained by minimizing (3.51), since two measurements are involved. In fact, the estimate of  $\underline{x}_0$  that is derived by minimizing (3.51) is a smoothed estimate rather than the filtered estimate given by (3.49). To obtain the desired recursion relation, consider  $\underline{L}_0$  and express it relative to  $\underline{1}_0$ .

$$\mathbf{L}_{0} = \frac{1}{2} \left\| \mathbf{z}_{0} - \mathbf{a} \right\|_{\mathbf{H}_{0}^{-1}}^{2} + \frac{1}{2} \left\| \mathbf{z}_{0} - \mathbf{H}_{0} \mathbf{z}_{0} \right\|_{\mathbf{H}_{0}^{-1}}^{2}.$$
(3.52)

Expand in a Taylor series, which is given exactly by

$$\mathbf{L}_{0} = \frac{1}{2} \left\| \mathbf{\hat{z}}_{0} - \mathbf{\hat{z}} \right\|_{\mathbf{H}_{0}^{-1}}^{2} + \frac{1}{2} \left\| \mathbf{z}_{0} - \mathbf{H}_{0} \mathbf{\hat{z}}_{0} \right\|_{\mathbf{H}_{0}^{-1}}^{2} + \frac{\partial \mathbf{L}_{0}}{\partial \mathbf{\hat{z}}_{0}} \left\| (\mathbf{z}_{0} - \mathbf{\hat{z}}_{0}) + \frac{\partial \mathbf{L}_{0}}{\partial \mathbf{z}_{0}} \right\|_{\mathbf{H}_{0}^{-2}}^{2} \mathbf{\hat{z}}_{0}$$

$$+ \frac{1}{2} \left\langle \underline{\mathbf{a}}_{0} - \underline{\mathbf{a}}_{0} \right\rangle^{T} \left( \frac{\partial^{2} \underline{\mathbf{b}}_{0}}{\partial \underline{\mathbf{x}}_{0}^{2}} \right) \Big|_{\underline{\mathbf{a}}_{0} = \underline{\mathbf{x}}_{0}} \left\langle \underline{\mathbf{a}}_{0} - \underline{\mathbf{x}}_{0} \right\rangle . \tag{3.53}$$

where

10000000 B33

VILLEVER PR

 $\frac{\partial^{2} L_{0}}{\partial \mathbf{x}_{0}^{2}} = \frac{\partial}{\partial \mathbf{x}} \left[ \mathbf{M}_{0}^{-1} (\mathbf{x}_{0} - \mathbf{A}) - \mathbf{H}_{0}^{T} \mathbf{R}_{0}^{-1} (\mathbf{x}_{0} - \mathbf{H}_{0} \mathbf{x}_{0}) \right] \\
= \mathbf{M}_{0}^{-1} + \mathbf{H}_{0}^{T} \mathbf{R}_{0}^{-1} \mathbf{H}_{0}; \\
\mathbf{L}_{0} = \frac{1}{2} \left\| \mathbf{x}_{0} - \mathbf{A} \right\|_{\mathbf{M}_{0}^{-1}}^{2} + \frac{1}{2} \left\| \mathbf{x}_{0} - \mathbf{H}_{0} \mathbf{x}_{0} \right\|_{\mathbf{R}_{0}^{-1}}^{2} + \frac{1}{2} \left\| \mathbf{x}_{0} - \mathbf{g}_{0} \right\|_{\mathbf{P}_{0}^{-1}}^{2} \\
\frac{\Delta}{2} \mathbf{C}_{0} + \frac{1}{2} \left\| \mathbf{x}_{0} - \mathbf{x}_{0} \right\|_{\mathbf{P}_{0}^{-1}}^{2}, \quad (3.54)$ 

where  $O_{\alpha}$  is independent of  $\underline{x}_{\alpha}$  and therefore will not have an influence on subsequent discussion.

The L<sub>i</sub> is now given by

$$\mathbf{L}_{1} = \mathbf{C}_{0} + \frac{1}{2} \left\| \mathbf{x}_{0} - \mathbf{\hat{x}}_{0} \right\|_{\mathbf{F}_{0}^{-1}}^{2} + \left\| \mathbf{x}_{1} - \mathbf{H}_{1}\mathbf{x}_{1} \right\|_{\mathbf{F}_{1}^{-1}}^{2} + \frac{1}{2} \left\| \mathbf{u}_{0} \right\|_{\mathbf{F}_{0}^{-1}}^{2}$$
(3.55)

To eliminate  $\underline{x}_0$ , introduce the plant constraint. Assuming that the inverse of  $\Phi_{1,0}$  exists and is  $\Phi_{0,1}$ . L<sub>1</sub> becomes

 $\mathbf{L}_{1} \stackrel{\text{\tiny de}}{=} \mathbf{C}_{0} + \frac{1}{2} \left\| \Phi_{0, 1} \left( \mathbf{X}_{1} - \mathbf{U}_{0} \right) - \mathbf{X}_{0} \right\|_{\mathbf{F}_{0}^{-1}}^{2} + \left\| \mathbf{X}_{1} - \mathbf{H}_{1} \mathbf{X}_{1} \right\|_{\mathbf{R}_{1}^{-1}}^{2} + \frac{1}{2} \left\| \mathbf{U}_{0} \right\|_{\mathbf{Q}_{0}^{-1}}^{2} .$ 

Now, choose the  $\underline{u}_0$  to minimize  $L_1$  .

$$\begin{pmatrix} \partial t_{-1} \\ \partial \underline{y}_{0} \end{pmatrix}^{T} = 0 = -\Phi_{0,1}^{T} P_{0}^{-1} [\Phi_{0,1}(\underline{x}_{1} - \underline{y}_{0}) - \underline{x}_{0}] + Q_{0}^{-1} \underline{y}_{0} .$$
 (3.56)

Solving for  $\underline{g}_0$  , one obtains

$$l_{o} = [I - \Phi_{i_{1},0} P_{0} \Phi_{i_{1},0}^{T} (\Phi_{i_{1},0} P_{0} \Phi_{i_{1},0}^{T} + u_{0})]^{-1} [\chi_{i} - \Phi_{i_{1},0} \chi_{0}] .$$





(3.57)

and

Let

 $\mathbf{P}_1' \stackrel{\Delta}{=} \Phi_{1,0} \mathbf{P}_0 \Phi_{1,0}^{\mathsf{T}} + \mathbf{Q}_0$ 

and note that

$$\Phi_{1,0} P_0 \Phi_{1,0}^T P_0^{\prime - 1} = I - Q_0 P_0^{\prime - 1} .$$

It follows from (3.57) and (3.58) that

$$\frac{1}{4} \| \Phi_{0,1}(\underline{x}_{1} - \underline{y}_{0}) - \underline{x}_{0} \|_{\underline{p}_{0}^{-1}}^{2} + \frac{1}{4} \| \underline{y}_{0} \|_{\underline{q}_{0}^{-1}}^{2} = \frac{1}{4} \| \underline{x}_{1} - \Phi_{1,0}\underline{x}_{0} - \underline{y}_{0} \|_{\underline{\Phi}_{0,1}^{-1}}^{2} + \frac{1}{4} \| \underline{x}_{1} - \Phi_{1,0}\underline{x}_{0} \|_{\underline{p}_{1}^{\prime-1}}^{2} = \frac{1}{4} \| \underline{x}_{1} - \Phi_{1,0}\underline{x}_{0} \|_{\underline{p}_{1}^{\prime-1}}^{2} + \frac{1}{4} \| \underline{x}_{1} - \Phi_{1,0}\underline{x}_{0} \|_{\underline{p}_{1}^{\prime-1}}^{2} = \frac{1}{4} \| \underline{x}_{1} - \Phi_{1,0}\underline{x}_{0} \|_{\underline{p}_{1}^{\prime-1}}^{2} + \frac{1}{4} \| \underline{x}_{1} - \Phi_{1,0}\underline{x}_{0} \|_{\underline{p}_{1}^{\prime-1}}^{2} = \frac{1}{4} \| \underline{x}_{1} - \Phi_{1,0}\underline{x}_{0} \|_{\underline{p}_{1}^{\prime-1}}^{2} + \frac{1}{4} \| \underline{x}_{1} - \Phi_{1,0}\underline{x}_{0} \|_{\underline{p}_{1}^{\prime-1}}^{2} = \frac{1}{4} \| \underline{x}_{1} - \underline{x}_{1,0}\underline{x}_{0} \|_{\underline{p}_{1}^{\prime-1}}^{2} + \frac{1}{4} \| \underline{x}_{1,0} \| \underline{x}_$$

With this relation, the cost function assumes the form

$$\mathbf{L}_{1} = \mathbf{C}_{0} + \frac{1}{2} \left\| \mathbf{X}_{1} - \Phi_{1,0} \underline{\mathbf{g}}_{0} \right\|_{\mathbf{P}_{1}^{\prime-1}}^{2} + \frac{1}{2} \left\| \mathbf{g}_{1} - \mathbf{H}_{1} \mathbf{X}_{1} \right\|_{\mathbf{R}_{1}^{-1}}^{2}.$$
(3.60)

Note that, in the absence of a new measurement, the best predicted estimate  $f_1'$  is

$$\underline{a}_{1}^{\prime} = \Phi_{1,0} \underline{a}_{0} , \qquad (3.61)$$

which corresponds with the anticipated result.

It can be seen that (3.60) has essentially the same form as the  $L_0$  described by (3.44), since  $C_0$  does not depend upon  $\chi_1$ . Thus minimization of  $L_1$  with respect to  $\chi_1$  will yield the result that was obtained above, except for the notation change. In particular, it follows that

$$\mathbf{\hat{g}}_{1} = \Phi_{1,0}\mathbf{\hat{g}}_{0} + K_{1}\left[\mathbf{\hat{g}}_{1} - H_{1}\Phi_{1,0}\mathbf{\hat{g}}_{0}\right], \qquad (3.62)$$

$$K_{1} = P_{1}^{\prime}H_{1}^{\prime}P_{1}^{\prime}H_{1}^{\prime}+R_{1}^{\prime})^{-1} . \qquad (3.63)$$

Also, the cost  $L_1$  is seen to be given by

$$\mathbf{L}_{1} = \mathbf{C}_{1} + \| \mathbf{x}_{1} - \mathbf{x}_{1}^{*} \|_{\mathbf{F}_{1}^{-1}}^{2}, \qquad (3.64)$$

where

C K

$$= (\mathbf{P}_{1}^{t+1} + \mathbf{H}_{1}^{T}\mathbf{H}_{1}^{-1}\mathbf{H}_{1})^{-1}$$

$$= \mathbf{P}_1' - \mathbf{P}_1'\mathbf{H}_1^{\mathsf{T}}(\mathbf{H}_1\mathbf{P}_1'\mathbf{H}_1^{\mathsf{T}} + \mathbf{R}_1)^{-1} \mathbf{H}_1\mathbf{P}_1'$$

$$P_1' = X_1 H_1 P_1'$$
 (3.66)

The generalization to an arbitrary stage follows inductively and is accomplished by using the same arguments as above. Thus it follows that the recursive, maximum a posteriori estimate is described by the system

$$k_{k} = \Phi_{k,k-1} \mathbf{x}_{k-1} + K_{k} [\mathbf{x}_{k} - \mathbf{H}_{k} \Phi_{k,k-1} \mathbf{x}_{k-1}],$$
 (3.67)

where

$$= \mathbf{P}_{k}^{\prime} \mathbf{H}_{k}^{T} (\mathbf{H}_{k} \mathbf{P}_{k}^{\prime} \mathbf{H}_{k}^{T} + \mathbf{R}_{k})^{-1}$$
(3.68)

and

$$P_{k}' = \Phi_{k,k-1}P_{k-1}\Phi_{k,k-1}^{T} + Q_{k-1}$$
(3.69)

$$P_{k} = P_{k}^{\prime} - K_{k}H_{k}P_{k}^{\prime} . \qquad (3.70)$$

Once again, the Kalman filter equations have been derived.

K,

# 4. SUMMARY OF RESULTS

The filtering problem for time-discrete, nonlinear, stochastic systems has been considered within the general probabilistic framework provided by the application of Bayes's rule. This Bayesian approach provides insight into the general character of the behavior of the *a*psteriori density function for nonlinear systems. The development has been designed to provide an understanding of the difficulties arising in nonlinear filtering theory and provides a structure within which the nonlinear problem can be attacked. Further, the Bayesian approach leads naturally to the discussion of the maximum *a posteriori* and maximum likelihood estimation procedures. The former procedure is shown to be identical with deterministic least-squares when the plant and measurement noise sequences are assumed to be additive and gaussian. The recursive Bayesian and maximum a *posteriori* approaches are seen to yield the same filtering equations as the unbiased, minumum variance estimates when the plant and measurement systems are linear with gaussian noise.

The principal results of this presentation are stated in the following paragraphs. The general problem is stated in Section 1.2.









A AND AND AND







(3.65)

Section 2.1: The conditional mean  $\mathbb{E}[\mathbf{g}|\mathbf{Z}]$  is shown in Theorem 2.1 to provide the unbiased, minimum variance estimate of a random variable  $\mathbf{y}$  from measurement data  $\mathbf{Z}$ . Thus, knowledge of the *a posteriori* density  $p(\mathbf{g}|\mathbf{Z})$  would enable this estimate to be determined, as well as virtually any other type of estimate that might be desired.

It is shown in Theorem 2.2 and in Theorem 2.3 that a significant number of estimation criteria will lead to the same estimate as the minimum variance criterion. This is important because it indicates that the choice of the minimum variance oriterion because of its analytical tractability has broader application than might be otherwise appreciated.

Section 2.2: Recursion relations which describe the manner in which the *a posteriori* density function changes as new data become available are presented as Theorem 2.4. Some of the principal difficulties that are encountered in applying these results to nonlinear systems are discussed.

Section 2.3: The general results of Section 2.3 are applied to linear systems with gaussian noise. Theorem 2.5 shows that this system has a gaussian a posteriori density and that the Kalman filter equations describe the conditional mean and covariance of  $p(\underline{x}_k | \underline{z}_k)$ . It is pointed out that it is more convenient, at least in this case, to work with the characteristic function rather than the density itself to derive the desired results.

Section 3.1: The maximum a posteriori estimation procedure is presented essentially as a nonrecursive version of the Bayesian approach described in Section 2. This formulation is seen to be identical with the deterministic least-squares problem discussed in Section 1.1 when the plant and measurement noise sequences are assumed explicitly to be additive and gaussian.

Section 3.2: Some aspects of the well-known maximum likelihood procedure are discussed in this section. It is seem that this approach differs from the maximum a posteriori through the neglect of the a priori distribution assigned to the parameters to be estimated. The two are identical if  $p(\underline{X}_k)$  is uniform. Then the maximum a posteriori procedure which calls for the determination of the  $\underline{X}_k$  that maximizes  $p(\underline{X}_k|\underline{Z}_k)$  is identical with the maximum likelihood procedure in which  $p(\underline{Z}_k|\underline{X}_k)$  is maximized through the choice of  $\underline{X}_k$ .

The Gramer-Rao inequality is presented in Theorem 3.1. This inequality provides a lower bound for the error covariance matrix for any unbiased estimate of an unknown parameter. This result is generalized in Theorem 3.2 to the case where the parameter is a random variable with prescribed distribution. The lower bound is the Fisher information matrix. For a linear system with gaussian measurement noise and no plant noise, the Gramer-Rao inequality is shown to provide a lower bound that is identical with a bound derived in Section 3.3.3 of Chapte: 1. Also, the Fisher information matrix is shown to be identical to the observability matrix, introduced in Section 3.1.

Section 3.3: The determination of the states  $\underline{x}_k$  that maximize  $p(\underline{x}_k|Z_k)$  is shown to again yield the Kalman filter equations. The procedure that is used to accomplish the minimization makes use of the desire to obtain a recursive solution and promises to provide straightforward application to nonlinear systems.

The preceding discussion provides a brief description of the contents of Sections 2 and 3. The reader is directed to the references for more complete treatments of many aspects that have only been touched superficially.









A State Diversity Date of



REFERENCES			
2. Aoki, M.	Optimization of Stochastic Systems: Topics in Discrete-Time Stochustic and Adaptive Systems Optimization. Aomdomic Press, New York, 1966.		
3. Borenson, H.W.	A Nonlinear Perturbation Theory for Estimation and Control of Time-Discrete Stochastic Systems. PhD Dissortation, Department of Engineering, University of California, Los Angeles (also UCLA Engineering Report 68-2), 1968.		
4. van Trees, H.L.	Detection, Estimation, and Modulation Theory. Part I. John Wiley, New York, 1968.		
8. Ho, Y.C.	The Method of Least-Squares and Optimal Filtering Theory, Rand Corporation Memo RM-3329-PR, 1962.		
6. Ito, K.	On Stochastic Processes. Tata Institute for Fundamental Research, Bombay, India, 1961.		
7. Stratonovich, R.L.	Topics in the Theory of Randon Noise. (Translated by R.A.Silverman). Gordon and Breach, New York, 1963.		
8. Papoulis, A.	Probability, Random Variables, and Stochastic Processes. McGraw-Hill, New York, 1965.		
9. Sherman, S.	Non-Nean-Square Error Criterion. Institute of Radio Engineers, Transactions on Information Theory, 1T-4, 1958, pp.125-123.		
10. Sorenson, H.W.	Nonlinear Filtering by Approximation of the a posteriori Density. International Journal of Control, Vol.8, No.1, pp.33-51.		
11. Stratonovich, R.L.	Conditional Markov Processes. Theory of Probability and its Applications, Vol.5, No.2, 1960, pp.156-178.		
12. Kushner, H.J.	On the Dynamical Equations of Conditional Probability Density Functions with Applications to Optimal Stochastic Control Theory. Journal of Mathematical Analysis and Applications, 1984, pp.332-344.		
13. Buoy, R.S.	Nonlinear Filtering Theory. Institute of Electrical and Electronic Engineers.		

13. B Transactions on Automatic Control, Vol. AC-10, No. 2, April 1965, p. 198.

Conditional Probability Density Functions and Optimal Nonlinear Estimation. PhD 14. Fisher J.R. Dissertation, University of California, Los Angeles, 1366.

On the Estimation of State Variables and Parameters for Noisy Dynamic Systems. 15. Cox, H. Institute of Electrical and Electronic Engineers. Transactions on Automatic Control, Vol. AC-9, No. 1, January 1984, pp. 5-12.

16. Cox, H, Estimation of State Variables via Dynamic Programming. Joint Automatic Control Conference, Stanford University, 1964, pp. 376-381.

17. Detphmendy, D.M. Sequential Estimation of States and Parameters in Noisy Nonlinear Dynamical Systems. Srichar, R. Joint Automatic Control Conference, Troy, New York, 1965, pp. 56-63.

18. Mortensen, R.E. Maximum-Likelihood Recursive Nonlinear Filtering. Journal of Optical Theory and its Applications, Vol.2, No.6, 1968, pp.386-394.

19. Gura, I.A. Menrikson, L.J.

20. Cramer, H.

A Unified Approach to Nonlinear Esilation. Asrospace Corporation, Report TR-0200 (4306-02)-2, Los Angeles, 1989.

Kathematical Methods of Statistics. Princeton University Press, New Jersey, 1961."









# CHAPTER 7 - NONLINEAR FILTERING AND COMPARISON

# WITH KALMAN FILTERING

anti ta șe din a din

ſ

# by

# Lawrence Schwartz

Hughes Aircraft Company, P.O. Box 90919, Los Angeles, California 90009, USA

143

134

•i Smil

HALL THINK

Sec. Star

ł

# NOTATION

¢	expectation operator
•	conditional expectation
L	backward diffusion operator
ĩ	forward diffusion operator
R	Euclidian N-space
ê	equal by definition
≈	approximately equal
*	transpose
,	derivative
<b>p(:)</b>	probability density function
p(1]1)	conditional probability density function
9	partial differential operator
d	differential operator
, 0'ij	the element in the i <sup>th</sup> row and j <sup>th</sup> column of the matrix in brackets
Onve	Rvotago

· · ·

`\* \*\**T*\*

'.÷

÷

A THEFT

In O When they O WA

100 100

611

łż.

LIV BIRLA

Destant And

THE PARA DIG

#### CHAPTER 7 - NONLINEAR FILTERING AND COMPARISON WITH KALMAN FILTERING

Lawrence Schwartz

#### 1. INTRODUCTION

The field of nonlinear filtering is quite broad, and a general discussion of the various approaches is not an appropriate subject for a single chapter in a book dedicated to Kalman Filtering. The specific form of menlinear filtering treated herein is that which is most closely related to Kalman filtering, indeed, that which is the most natural generalization thereof: minimal-variance filtering. To further limit the scope, the derivation is limited to continuous-time minimal-variance filtering.

Curiously enough, the work that led to the successful derivation of nonlinear continuous-time minimal-variance filters does not stem from the paper by Kalman and Bucy<sup>1</sup> which contains the derivation of the linear continuoustime minimal-variance filter. Rather, the imposus derives from Stratonovich<sup>2</sup>, in an even earlier paper, who noted that all the information related to the optimal estimate of the state of a system is contained in the conditional probability distribution function of the state system the measurements. Thus, the more general filter is not an outgrowth of the more restricted one but the result of a parallel line of investigation.

While only one approach to nonlinear filtering is discussed in dotail, several nonlinear filters are computationally compared with a Kalman filter. All of the nonlinear filters mechanized reduce to the Kalman filter when the dynamical equation of the system is linear, the measurements are linearly related to the state, and the noise processes are Gaussian. An outline of the steps in the derivation is essential to set the stage for the following discussion. The first step in the analysis of any physical situation is the specification of a mathematical model; the choice should be made carefully, since the whole analysis depends upon the characteristics of the sole. For minimal variance, which is a probabilistic ordiption of optimality, the manipulations leading to the filter equations are made particularly simple by assuming that the random processes are white noises. The formal simplicity is gained at the expense of a certain amount of complication in the physical interpretation of the mathematical results.

Given the probabilistic criterion and the white-noise assumption, a natural mathematical model is the stochastic differential equation. Of course, the problem must be such that the stochastic differential equation satisfies existence and uniqueness conditions, which are different from those pertaining to nonstochastic differential equations. The essential difference stems from the fact that for white-noise models there is no bound on the forcing function and global conditions must be satisfied. From the stochastic differential equations for the system and the measurement, it is possible to derive a stochastic partial differential equation for the conditional density function. From the stochastic partial differential equation to derive a stochastic differential equation for the supected value of any scalar function of the state of the system; a finite set of such equations form the vector equation for the filter.

The exact equation for the filter requires the instantaneous evaluation of the conditional expectation of several functions of the state of the system. To simplify the problem, the original model can be replaced by an approximating stochastic differential equation, from which an approximate filter can be derived more simply. It seems remonship to require that the approximate model equations also satisfy existence and uniqueness conditions.

Guaranteeing existence and uniqueness for the approximate system does not quite do the same for the filter, but slightly stronger conditions on the equations suffice. At this point, one step remains in the validation: that of relating the stochastic differential equation for the filter to an ordinary differential equation for the actual mechanization. It is shown by Schwartz and Stear<sup>3</sup> that nonvalid filter equations similar to those previously derived by Schwartz and Bass<sup>6</sup> and by Fisher<sup>8</sup> for white-noise processes can be made computationally identical to valid filter equations.

; *R* 

#### 8. MATHEMATICAL NODEL

#### 8.1 White Noise

الأسطيحية المراقب المتراكر المرجور جرائب والمرادية

The usual mathematical formulation for a dynamical problem is a differential equation, nowadays most generally written in state-vector form:



(1).

where  $\dot{x}$  and f are n-vectors, u is an m-vector, and t is a scular. The meaning of (i) is well known for most input functions u, but the ensuing analysis deals with white-noise input functions, and (i) must be reinterpreted. The present section explains the problems associated with white-noise inputs, and outlined the development of the necessary calculus of stochastic processes. White noise in often described as a random process with a power spectral density which is a constant or, equivalently, an autocorrelation function which is a Dirac  $\delta$ -function. It is further noted that such a process has no physical meaning, since it would require infinite signal power. The foregoing definition is valid for stationary white noise, though the autocorrelationfunction definition can be extended to the nonstationary case by allowing a time-varying coefficient for the  $\delta$ -function.

The nonrealizability of white-noise processes is no reason to discard them; they occupy a place with respect to the family of stochastic processes analogous to the place of the Dirac  $\delta$ -function with respect to functions. Just as the  $\delta$ -function can be regarded as a limit of a sequence of unit-area pulses of decreasing width, a whitenoise process can be considered as the limit of a sequence of processes which are step functions. Moreover, both are useful only when their integrals are considered; indeed, both can be made mathematically rigorous only in terms of their integrals. To be somewhat imprecise, the  $\delta$ -function may be considered as the derivative of a unit step; with similar imprecision white-roise is the derivative of Brownian motion.

The practical reason for being concorned with white-noise processes is that, when differential equations are forced by white-noise, the solutions are Markov processes, i.e., the future is independent of the past. In other words, the solutions to differential equations forced by white-noise exhibit the stockastic analogue of the property of solutions to differential equations forced by ordinary functiona: given the state of the solution at some time and the forcing function from that time on, the subsequent evolution of the solution is stochastically independent of the previous history.

The mathematical problem associated with white-noise is nomewhat similar to that associated with the  $\delta$ -function: the meaning of the integral. The  $\delta$ -function is not really a function in the ordinary sense of the word, and no theory of integration can result in a value other than zero for

if the 8-function is assumed to be an ordinary function of t. However, by not asoribing values to  $\delta(t)$  and considering only its integral, it is possible to construct a meaningful theory. Similarly, no ordinary theory of integration can make sense of

∫w(t) dt ,

where w is a white-noise process. Here again, if no instantaneous value is given to w(t), a useful theory of stochastic integration is possible; that theory is outlined in the following section,

#### 1, 5 Stochastic Integrals

The exposition in this section is merely an outline of the mathemetical derivation of the stochastic integral, and it does not include any discussion of stochastic integrals of discontinuous random processes. A complete discussion of stochastic integration can be found in Skorokhod<sup>4</sup>. The underlying idex is the description of a white-noise process as the derivative of a Brownian motion; the major difficulty lies in the fact that a Brownian motion fails to be differentiable somewhere in every interval of nonzero length, with probability one. Thus, if b is a Brownian motion; db/dt has no meaning and

# fs(t)(db/dt) dt

is generally not defined, even for continuous functions g. But, if db is an increment of b, it has a welldefined stochastic description, and the Stieltjes integral

∫s(t) db(t)

is a possibly meaningful alternate form. However, b is not a function of bounded variation and even the Stieltjes integral is not defined. The stochastic integral is a succussionally meaningful momentization of the Stieltjes integral, in which the approximating sume are required to converge in probability to the integral, rather than to converge in the ordinary sense.

# 8.3 stochastic Differential Equations

and the contraction of the state of the

The discussion contained in the sequel is limited to the following special case of Equation (1):

$$\frac{dx}{dt} = f(t,x(t)) + g(t,x(t))u(t) , \qquad (2)$$

where x and f are n-vectors, t is a scalar, u is an m-vector unit white-noise process with independent elements, and g is an n x m matrix. A vector white-noise process has elements derived from Wiener processes;















a Wiener process is a unit Brownian motion, i.e., its increments have zero mean and variance equal to the interval of time over which the increment is defined. While Equation (2) is less general than Equation (1), it is sufficient for must practical applications. Let  $g^i$  denote the i<sup>th</sup> column of g and  $w_i$  the i<sup>th</sup> row of the vector Wiener process from which u is derived. Then, by formal multiplication of Equation (2) by dt and integration of the resulting expression. Equation (2) can be rewritten as

$$x(t) - x't_0) = \int_{t_0}^{t} f(s, x(s)) ds + \sum_{l=1}^{m} \int_{t_0}^{t} g^l(s, x(s)) dw_l(s) .$$
(3)

In Equation (3), the first integral is an ordinary integral, and the remaining m integrals are stochastic integrals. For simplicity, if the stochastic integral equation is satisfied by a process x with probability one, then Equation (3) is written in the form

$$dx = f(t, x) dt + g(t, x) dw(t)$$

The simplified form Equation (4) is referred to as a stochastic differential equation and is understood to be a shorthand notation for Equation (3).

The following formula is necessary for the derivation of stochastic differential equations for functions of solutions of other stochastic differential equations. The scalar version of the formula is proved by Skorokhod in Reference 6 (p.24ff); the vector version follows quite simply. Let x satisfy Equation (4) for  $t_0 \le t \le t_{f'}$ ; if a scalar function  $\phi(t,x)$  is defined with continuous second cross partial derivatives with respect to the .  $x_1$  for  $t_0 \le t \le t_f$  and for all x, then the process  $y(t) = \phi(t,x(t))$  entisfies the relation

$$d\mathbf{y} = \left(\frac{\partial\phi}{\partial t} + \frac{\partial\phi}{\partial x}\mathbf{f} + \frac{1}{2}\mathbf{g}^*\frac{\partial^2\phi}{\partial x^2}\mathbf{g}\right) dt + \frac{\partial\phi}{\partial x}\mathbf{g} dw \tag{8}$$

where  $\partial()/\partial x$  denotes the gradient (row) vector,  $\partial^{\frac{1}{2}}(y'\partial x^{\frac{1}{2}}$  denotes the Hessian (matrix of cross partials), and the "star" denotes matrix transpose.

## 2.4 Relation to the Physical Problem

í

1 States

 There are two interfaces between the physical situation and the mathematical model in the filtering problem: the reduction of the dynamics to a stoch which differential equation and the interpretation and mechanization of the stochustic differential equation for the filter as a computational algorithm. The mecond interface is considered first. The interpretation of the stochastic integral in the context of the actual estimation environment is not at all a trivial matter. The filtering algorithm will be a finite-difference approximation to Equation (2), with a represented by a sampled measurement, not a white-noise. The problem of the interpretation of Equation (2) and the approach to use for the integration is discussed at length by dray and Caughey<sup>7</sup>; they specify two approaches and propose a list of four pragmatic rules for choosing between the two approaches based on the interpretation of Equation (2). Another treatment of the difference between the two approaches is given by Wong and Zakai<sup>8</sup>.

The real difference between the approaches is in the choice of whether to use the ordinary calculus or the stochastic calculus. In the nomenclature of Gray and Caughey<sup>1</sup> the former choice is the physical approach and the latter choice is the mathematical approach. In contrasting the two approaches, the authors are quick to state that neither approach is inherently correct; the choice should be made according to their pragmatic rules:

(i) If g(t, x) is not actually a function of x, both approaches provide identical results.

(ii) If the problem is a strictly mathematical one, the mathematical approach must be used.

(iii) If Equation (8) is either an approximation to or a limit of the disorate problem

$$x(t_{k+1}) = x(t_k) / (t_{k+1} - t_k) = f(t_k, x(t_k)) + g(t_k, x(t_k)) u(t_k)$$

then the mathematical approach must be used,

(iv) If Equation (2) is either an approximation to a white-noise problem or the limit of a problem with short correlation time, then the physical approach must be used.

The computational effect of the difference between the two approaches is stated by Wong and Zakai<sup>a</sup> as follows: Let  $\{w^n\}$  be a sequence of piecewise linear approximations to the Wiener process in Equation (10) such that  $w^n \to w$ ; then if  $\{x^n\}$  denotes the sequence of corresponding solutions,  $x^n \to x$ , where x is the solution to

$$d\mathbf{z}_{i} = f_{i}(t, \mathbf{x}(t)) dt + \frac{1}{2} \sum_{j, \mathbf{k}} \mathbf{g}_{\mathbf{k}j}(t, \mathbf{x}(t)) \frac{\partial \mathbf{g}_{ij}}{\partial \mathbf{u}_{\mathbf{k}}} (t, \mathbf{x}(t)) dt + \sum_{j} \mathbf{g}_{ij}(t, \mathbf{x}(t)) d\mathbf{x}_{j} . \qquad (6)$$

They state zone reservations about the correctness of Equation (6) in the vector case, but the same form is implied by the results of Gray a.d Coughny'.





100 314 14 787A D 1

and a state of the state of the

-----




10

There is also a problem in relating the statistics of the real data to the statistics of the white-noise used in the model; this problem exists at both interfaces and is really the only one at the first. For simplicity, consider the following special case: Let u(t) denote a sequence of pulses of width  $\Delta t$  and of random height given by a Gaussian distribution of zero mean and variance  $\sigma^2$ . The autocorrelation function for u is a triangular spike of width  $\Delta t$  and height  $\sigma^2$ ; the area under the spike is then  $\sigma^2 \Delta t$ . It then seems reasonable that the equivalent white-noise is given by a covariance of weight  $\sigma^2 \Delta t$ . The general result is as follows: If an n-dimensional white-noise is given by a covariance of the form  $S(t)\delta(t-\tau)$ , an n-dimensional pulse-sequence approximating the process should be chosen from a population given by a covariance of  $S(t_1)/\Delta t$  for  $t_1 \leq t \leq t_1 + \Delta t$ . The case of continuous u is not quite so direct, though an equivalent formulation can be obtained by using the unorelated of a correlation time  $\tau$ , which is a time interval such that u(t) can be considered uncorrelated with  $u(t + \tau)$ .

Since the mathematical model is constructed under the masumption that the Wiener process has independent elements, one final step is required to model a noise with correlated elements. Let  $S(t)\delta(t-r)$  be the desired covariance, which implies S(t) is positive semi-definite for all t. Then there exists a matrix (which may be taken as symmetric)  $S^{1/2}$  such that  $S^{1/2}(S^{1/2})^* = S$ . If  $dv \stackrel{A}{=} S^{1/2} dw$ , the white-noise derived from vhas the proper dovariance. For notational simplicity it may be assumed that  $S^{1/2}(t)$  is incorporated into g(t, x), and the formalise of Equations (2), (3), and (4) is still valid.

#### 3. DERIVATION OF APPROXIMATE FILTER EQUATIONS

#### 3.1 Conditional Donaity Function

Let the dynamic equation of the system be given by Equation (2), and let the measurement be given by

$$y(t) = h(t, x(t)) + r(t) v(t)$$
,

where h is an l-vector,  $l \leq n$ , v is an l-dimensional unit white-noise and r is a nonsingular symmetric  $l \neq l$  matrix relating the unit white-noise to the modeled white-noise (equivalent to the matrix  $B^{1/2}$  just described). Since the mathematical model cannot handle white-noise directly, it is assumed that the measurement is derived from a process z given by

$$dz = h(t, x(t)) dt + r(t) db(t), \qquad (8)$$

where b is an l-dimensional Wiener process. The mathematical model of the system consists of the two vector equations (4) and (8). The problem is to find the minimal-variance estimate of x(t), given the process x(s) for  $t_0 \le s \le t$ ; that is, to find the estimate  $\hat{x}(t)$  such that the matrix given by

$$\epsilon (x - \overline{x}) (x - \overline{x})^{+} - \epsilon (x - \hat{x}) (x - \hat{x})^{+}$$

is positive somi-definite, where X is any estimate of x, the processes are evaluated at t, and the symbol  $\epsilon$  denotes expectation.

It is a simple exercise to show that the minimal-variance estimate of a random variable, given a related quantity, is simply the conditional expectation, so that

 $\hat{\mathbf{x}}(\mathbf{t}) = \boldsymbol{\epsilon}(\mathbf{x}(\mathbf{t})|\mathbf{x}(\mathbf{s}), \mathbf{t}_{\mathbf{n}} \leq \mathbf{s} \leq \mathbf{t}) \tag{9}$ 

and the problem is to find an equation for the conditional expectation.

The first step is to derive formally an expression for the stochastic differential of p(x|x), the conditional probability density function of x(t) given x(s),  $t_n \leq s \leq t$ . It can be shown that

$$\mathbf{p}(\mathbf{x}|\mathbf{x}) = \frac{\mathbf{a}(\mathbf{s}^{\Phi}|\mathbf{x})\mathbf{p}(\mathbf{x})}{\mathbf{a}(\mathbf{s}^{\Phi})} . \tag{10}$$

where p(x) is the probability density function for  $x(t_n)$  and

$$\hat{\Phi} = \frac{1}{2} \int_{t_0}^{t} h^{k} r^{-1} h \, ds + \int_{t_0}^{t} h^{k} r^{-2} \, dz \, . \tag{11}$$

Following Bucy<sup>9</sup>, let  $p(x|x) \oint Q/P$ . Also, let  $\partial \oint c(e^{\Phi}|x)$ . Then Q is explicitly a function of  $\theta$  and p(x), and  $\theta$  is explicitly a function of a random process  $\Phi$  given by

$$d\Phi = \frac{1}{2}h^{+}r^{-2}h dt + h^{+}r^{-1} db . \qquad (12)$$





(7)



(17)

(18)

xox laixo

In addition p(x(t)) is a function of t, while x(t) and z are assumed fixed. Then, using Equations (5) and (12),

$$\mathbf{q} = \frac{\partial \mathbf{q}}{\partial t} dt + \frac{\partial \mathbf{q}}{\partial \Phi} \left( \frac{1}{2} \mathbf{h}^{+} \mathbf{r}^{-2} \mathbf{h} dt + \mathbf{h}^{+} \mathbf{r}^{-4} db \right) + \frac{1}{2} \frac{\partial^{2} \mathbf{q}}{\partial \Phi^{2}} \mathbf{h}^{+} \mathbf{r}^{-2} \mathbf{h} dt , \qquad (13)$$

Now

$$\frac{\partial Q}{\partial t} = \frac{\partial \theta}{\partial t} p(x(t)) + \theta \frac{\partial p(x(t))}{\partial t} = \theta \tilde{L} p(x(t)) = \tilde{L} Q , \qquad (14)$$

where  $\mathcal{L}$  is the forward diffusion operator

$$-\sum_{l=1}^{n} \frac{\Im(x_{l};)}{\Im x_{l}} + \frac{1}{2} \sum_{l,j=1}^{n} \frac{\Im^{*}([x_{1}^{*}]_{1j};)}{\Im x_{l} \Im x_{j}} .$$
(15)

 $\frac{\partial q}{\partial \phi} = \frac{\partial \phi_{\pi}}{\partial z} = q , \qquad (16)$ 

Substituting Equations (14) and (16) into Equation (13) provides

 $dQ = \vec{L}Q dt + Q(h^*r^{-1}h dt + h^*r^{-1} db)$ 

Using Equations (18) and (15), and the definition of Q/P ,

$$d\mathbf{P} = d\left(\int_{\mathbf{R}_{n}} \mathbf{Q} \, d\mathbf{x}\right) = \int_{\mathbf{R}_{n}} (d\mathbf{Q}) \, d\mathbf{x}$$
$$= P\left\{\int_{\mathbf{U}}^{\mathbf{U}} \left[\int_{\mathbf{R}_{n}} \left(\frac{\mathbf{Q}}{\mathbf{p}}\right) \, d\mathbf{x}\right] \, d\mathbf{t} + \left[\int_{\mathbf{R}_{n}}^{\mathbf{Q}} \mathbf{p}^{\mathbf{h}^{\mathbf{r}}\mathbf{r}^{-\mathbf{g}}} \, d\mathbf{x}\right] \, d\mathbf{z}\right\}$$
$$= P \hat{\mathbf{h}}^{\mathbf{T}} \mathbf{r}^{-\mathbf{g}} \, d\mathbf{z} = P \hat{\mathbf{h}}^{\mathbf{T}} \mathbf{r}^{-\mathbf{g}} \, d\mathbf{t} + P \hat{\mathbf{h}}^{\mathbf{T}} \mathbf{r}^{-\mathbf{i}} \, d\mathbf{b} \quad .$$

Using Equation (5), with  $x \triangleq P$ ,  $\phi \triangleq P^{-1}$ ,  $f \triangleq P \widehat{h^*}r^{-1}$ , and  $w \triangleq b$ .

 $d(P^{-1}) = -P^{-1} \hat{h^*} r^{-2} (h - \hat{h}) dt - P^{-1} \hat{h^*} r^{-1} db .$ 

Finally, using Equation (11) with  $x \stackrel{\Delta}{=} (P^{-1}, Q)^*$ ,  $\phi \stackrel{\Delta}{=} QP^{-1}$ ,  $f \stackrel{\Delta}{=} (-P^{-1} \widehat{h^*} r^{-2} (h - \widehat{h})$ ,  $\widehat{\Box} Q + Q h^* r^{-2} )^*$ ,  $g \stackrel{\Delta}{=} (-P^{-1} \widehat{h^*} r^{-1}, Q h^* r^{-1})^*$ , and  $w \stackrel{\Delta}{=} b$ ,

$$i\left(\frac{\dot{q}}{\dot{p}}\right) = \left(\widehat{L}\left(\frac{\dot{q}}{\dot{p}}\right) + h^{\phi}r^{-i}h\frac{\dot{q}}{\dot{p}}\right)dt + h^{\phi}r^{-i}\left(\frac{\dot{q}}{\dot{p}}\right)db - \hat{h^{\phi}}r^{-2}(h-\hat{h})\left(\frac{\dot{q}}{\dot{p}}\right)dt - \frac{\dot{h^{\phi}}r^{-1}\left(\frac{\dot{q}}{\dot{p}}\right)dt - \hat{h^{\phi}}r^{-2}h\left(\frac{\dot{q}}{\dot{p}}\right)dt - \hat{h^{\phi}}r^{-2}\left(\frac{\dot{q}}{\dot{p}}\right)dt + (h-\hat{h})^{\phi}r^{-2}\left(\frac{\dot{q}}{\dot{p}}\right)(dz - \hat{h} dt)$$

Let  $\phi$  be any evaluar function, twice continuously differentiable in x :

 $\hat{\phi} = \int_{\mathbf{R}_{\mathbf{R}}} \phi(\mathbf{z}) p(\mathbf{x} | \mathbf{z}) d\mathbf{x}$ 

 $d\phi = \int_{R_n} \phi(z) (dp(z|z)) dz$ .

(20)

(19)

Substituting Equation (19) into (20) provides

$$\mathrm{d}\hat{\phi} = \int_{\mathbf{R}_{\mathrm{B}}} \phi(\mathbf{x}) \left( \tilde{U} p(\mathbf{x} \mid \mathbf{x}) + (\mathbf{h} - \hat{\mathbf{h}})^{\frac{n}{2}} \mathbf{r}^{-1} p(\mathbf{x} \mid \mathbf{x}) \left( \mathrm{d}\mathbf{x} - \hat{\mathbf{h}} \right) \mathrm{d}\mathbf{x} \right)$$

Let  $\mathcal{L}$  denote the formal adjoint of  $\mathcal{L}$  ,

$$\sum_{i=1}^{n} f_i \frac{\partial}{\partial x_i} + \frac{1}{2} \sum_{i,j=1}^{n} \left[ gg^{ij} \right]_{ij} \frac{\partial^2}{\partial x_i \partial x_j} . \tag{22}$$

Then Equation (21) becomes

$$\hat{\rho} = \widehat{L\phi} \, dt + (\widehat{\phi} \, \widehat{h} - \widehat{\phi} \, \widehat{h})^* r^{-2} (dz - \widehat{h} \, dt) \,. \tag{23}$$

3.2 Approximate Filter Equations

The use of Equation (23) as a differential equation of  $\hat{x}$  results in

$$d\hat{x}_{i} = \hat{f}_{i} dt + (\hat{x}_{i}\hat{h} - \hat{x}_{i}\hat{h})^{*}r^{-2}(dx - \hat{h} dt)$$
 (24)

which is not very practical because  $\hat{f}_i$ ,  $\hat{h}$ , and  $\hat{x_i}\hat{h}$  are needed continuously. As the first step in the approximation, let f and h be approximated by a second-degree expansion about  $x = \hat{x}$ ; also, for notational simplicity, suppress the explicit any trance of t as an argument of f, g, and h, since their dependence on time is incidental to the following r mipulations, Thus, adopting the summation convention

$$f_{1}(x) \approx f_{1}(x) + f_{1}(x)(x_{j} - \hat{x}_{j}) + \frac{1}{2} f_{1}(x)(x_{j} - \hat{x}_{j})(x_{k} - \hat{x}_{k}) , \qquad (25)$$

where

$$f_{i}^{(j)} \stackrel{\Delta}{\to} \frac{\partial r_{i}}{\partial r_{j}}$$
 and  $f_{ijk}^{(s)} \stackrel{\Delta}{\to} \frac{\partial^{2} f_{i}}{\partial x_{k}}$ .

A similar expression holds for h . From Equation (25),

$$\hat{t}_{1}(\mathbf{x}) \approx t_{1}(\hat{\mathbf{x}}) + \frac{1}{2} t_{1}^{2} t_{1}^{2} (\hat{\mathbf{x}}) (x_{j} - \hat{x}_{j}) (x_{k} - \hat{x}_{k})$$
, (26)

where  $(x_j - \hat{x}_j)(x_k - \hat{x}_k)$  is the conditional covariance of x and is denoted P<sub>11</sub>.

Similarly

 $\widehat{x_i t_i}(x) \approx t_{jk}^{(1)}(\hat{x}) P_{ki} + \hat{x}_i \hat{t}_j(x)$ (27)

Using Equations (26) and (27) for f and h in Equation (24) provides

$$d\hat{x}_{i} \approx f_{i}(\hat{x}) dt + \frac{1}{2} f_{ijk}^{(2)}(\hat{x}) P_{jk} dt + P_{ij} h_{k}^{(j)}(\hat{x}) r_{k}^{*} \int dx_{i} - (h_{i}(\hat{x}) + \frac{1}{2} h_{km}^{(3)}(\hat{x}) P_{mn}) dt ] .$$
(28)

The next step is to find a differential equation for P. In general, even with a second-degree expansion for the nonlinearities, an infinite sequence of differential equations is required, because all the moments are needed to describe the conditional density. However, by assuming an appropriate form for p(x|u), the sequence stops at P. The first assumption, used by Bucy<sup>5</sup>, is that third and fourth central conditional moments be neglected. In Schwarts and Bass<sup>4</sup> it is shown that the assumption is reasonable for a distribution with most of the probability mass sufficiently close to the mean. If it is assumed that p(x|s) is Gaussian, the sequence also stops at P, and there is no restriction on the size of the moments.

Since  $P_{ij}$  can be written  $(\widehat{x_i}x_j - \widehat{x_i}\widehat{x_j})$ ,  $dP_{ij}$  is derived in two parts. Let  $\phi = x_ix_j$  and use Equation (23) to find  $d\widehat{x_i x_j}$ . Since

 $\mathcal{L}\mathbf{x}_{i}\mathbf{x}_{j} = \mathbf{f}_{i}\mathbf{x}_{j} + \mathbf{f}_{j}\mathbf{x}_{i} + \mathbf{g}_{ik}\mathbf{g}_{ik} ,$ 

it follows that

$$d\widehat{x_1x_j} = \widehat{f_1x_j} dt + \widehat{f_jx_i} dt + \widehat{s_{ik}s_{jk}} dt + (\widehat{x_1x_j}h_k - \widehat{x_1x_j}\hat{h}_k)r_k^{\dagger}(dx_i - \hat{h}_i dt)$$



(20)



(21)

(30)

Next, let  $\phi = \hat{x}_{i}\hat{x}_{j}$  and use Equations (5) and (24) to obtain

$$\begin{split} i \hat{x}_{i} \hat{x}_{j} &= \frac{\partial (\hat{x}_{i} \hat{x}_{j})}{\partial \hat{x}_{k}} \left[ \hat{t}_{k} dt + (\hat{x}_{k} \hat{h}_{i} - \hat{x}_{k} \hat{h}_{i}) r_{im}^{2} (dx_{m} - \hat{h}_{m} dt) \right] + \\ &+ \frac{1}{2} \frac{\partial^{2} (\hat{x}_{i} \hat{x}_{j})}{\partial \hat{x}_{k} \partial \hat{x}_{i}} (\hat{h}_{m} \hat{x}_{k} - \hat{h}_{m} \hat{x}_{k}) r_{mn}^{2} (\hat{h}_{m} \hat{x}_{i} - \hat{h}_{m} \hat{x}_{i}) dt \end{split}$$

· Combining Equations (29) and (30) provides

$$dP_{1j} = (\hat{t}_1 \hat{x}_j - \hat{t}_1 \hat{x}_j + \hat{t}_j \hat{x}_1 - \hat{t}_j \hat{x}_1 + \hat{s}_{1k} \hat{s}_{jk}) dt - (\hat{x}_1 \hat{h}_k - \hat{x}_1 \hat{h}_k) \hat{x}_k^{2} (\hat{x}_j \hat{h}_l - \hat{x}_j \hat{h}_j) dt + + [\hat{x}_1 \hat{x}_j \hat{h}_k - \hat{x}_1 \hat{x}_j \hat{h}_k - \hat{x}_j \hat{h}_k \hat{x}_1 - \hat{x}_1 \hat{h}_k \hat{x}_j + 2 \hat{x}_1 \hat{x}_j \hat{h}_k] \hat{r}_k^{2} (dz_l - \hat{h}_l dt) ,$$
(31)

which is an exact equation. The derivation of the approximate form of Equation (31) is a todious algebraic exercise, and is covered in Bass, Norum and Schwartz<sup>10</sup>. For completeness, several intermediate results follow: For either assumption about the conditional density.

$$\widehat{x_{i}x_{j}(x_{k}-\hat{x}_{k})} = \widehat{x_{i}(x_{k}-\hat{x}_{k})}\hat{x}_{j} = \widehat{x_{j}(x_{k}-\hat{x}_{k})}\hat{x}_{i} = 0.$$

For the first assumption,

$$\widehat{\mathbf{x}_{1}\mathbf{x}_{j}}\widehat{\mathbf{h}}_{k} = \widehat{\mathbf{x}_{1}}\widehat{\mathbf{x}}_{j}\widehat{\mathbf{h}}_{k} - \widehat{\mathbf{x}}_{j}\widehat{\mathbf{h}}_{k}\widehat{\mathbf{x}}_{j} + 2\widehat{\mathbf{x}}_{1}\widehat{\mathbf{x}}_{j}\widehat{\mathbf{h}}_{k} = -\frac{1}{2}P_{1j}(\mathbf{h}_{k}^{(j)}(\mathbf{x})P_{1k}).$$
(32)

For the other assumption, the right-hand side of Equation (32) is

₫°ii

$$\frac{1}{2} \left( P_{ij} P_{ij} + P_{ij} P_{jj} \right) h_{k}^{(2)} (\hat{x}) , \qquad (33)$$

For simplicity, let the term in parentheses in Equation (33) be denoted by  $T_{ijkl}$ . Then, if g is also expanded in a first degree approximation, the following equations are obtained:

$$\approx P_{ik} f_{jk}^{(1)}(\hat{x}) dt + f_{lk}^{(1)}(\hat{x}) P_{kj} dt - P_{ik} h_{k}^{(1)}(\hat{x}) r_{lk}^{2} h_{kn}^{(1)} P_{nj} dt + + g_{ik} g_{jk}^{(1)}(\hat{x}) dt + g_{ikn}^{(1)} f_{kn}^{(1)}(\hat{x}) P_{nn} dt - - \frac{1}{2} P_{kl} h_{kk}^{(2)}(\hat{x}) r_{nn}^{-2} (ds_{n} - h_{n}(\hat{x}) - \frac{1}{2} h_{npq}^{(2)}(\hat{x}) P_{pq}) P_{ij}$$
(34)

and

$$dP_{ij} \approx P_{ik} f_{jk}^{(1)}(\hat{x}) dt + f_{ik}^{(1)}(\hat{x}) P_{kj} dt - P_{ik} h_{k}^{(1)}(\hat{x}) r_{lm}^{-1} h_{mn}^{(1)}(\hat{x}) P_{nj} dt + + g_{ik}(\hat{x}) g_{jk}(\hat{x}) dt + g_{kk}^{(1)} g_{jkn}^{(1)}(\hat{x}) P_{mn} dt + + \frac{1}{4} r_{ijkl} h_{mk}^{(4)}(\hat{x}) r_{mn}^{-4} (dg_{n} - h_{n}(\hat{x}) - \frac{1}{4} h_{mpq}^{(4)}(\hat{x}) P_{pq}) ,$$
(35)

where Equation (34) uses (32) and Equation (35) uses (33), and  $g_{ijk}^{(j)}$  denotes  $\partial g_{ij}/\partial x_k$ .

#### 4. SINULATIONS

#### 4.1 Description

The numerical investigation was conducted for eight different approaches to filtering, for two sets of dynamics, for two measurement schemes each. The two systems were chosen such that one satisfied existence and uniqueness conditions for stochastic differential equations while the other did not. No difficulties were anticipated in connection with filtering for the second system in the actual computer environment, and none were encountered.

#### 4.2 Filter Equations

Because of the large number of cases involved in the study, only first-order systems are considered. The first system is



States - States

Ĩ

(D1)	$\frac{dx}{dt} = -\frac{x}{1+x^2} + v$	(70)
(M11)	y = tan <sup>-1</sup> x + w	
(M1 2)	y = x + #	

where y and w are white-noises. With either measurement achame, (Mil) or (M12), the overall Hyntres Hatisfirs existence and uniqueness. The second system, which does not, is given by

> $\frac{\mathrm{d}x}{\mathrm{d}t} = -x^3 + v$ (D3) (37) = x + x<sup>1</sup> + w (M21) (M22)

The filtering schemes are outlined below. For the outline, f and h are the system nonlinear tipe an in Equations (2) and (7). The white-noise processes, v and w, are assumed stationary with covariances  $\sigma_{\psi}^{\sigma}$  and  $\sigma_{\psi}^{\sigma}$  respectively so that  $g = \sigma_{\psi}$  and  $r = \sigma_{\psi}$ . The subscript "n" denotes nominal,  $\hat{x}$  is the entireate of x, p is the approximate covariance, and a prime denotes differentiation. Note that for linear measurements, (M12) and (M22), certain terms vanish and there are only three different either. and (N22), certain terms vanish and there are only three different filters.

1. Linear

The linear filtering algorithm can be applied to a set of equations linearized about an  $a \ priori$  (Usuitual motion). The equations, derived by Kalman and Bucy<sup>23</sup> are

dx dt	E	$f'_{n}\hat{x} + \sigma_{y}^{-s}ph'_{n}(y - h'_{n}\hat{x})$	(38)
dp dt	•	$2pt_{\rm H}' = \sigma_{\rm H}^{-2} p^2 h_{\rm H}'^2 + \sigma_{\rm V}^2 +$	

2. Quasi-Moment Minimal-Variance

This is the filter derived by Schwartz and Bass", and independently by Fisher<sup>5</sup>.

$$\frac{d\hat{x}}{dt} = t(\hat{x}) + \frac{1}{2}t''(\hat{x})p + \sigma_{w}^{-2}ph'(\hat{x})[y - h(\hat{x}) - \frac{1}{2}ph''(\hat{x})] .$$

$$\frac{dp}{dt} = 2pt'(\hat{x}) - \sigma_{w}^{-2}p^{2}h'^{-2}(\hat{x}) + \sigma_{w}^{-2}p^{2}h''(\hat{x})[y - h(x) - \frac{1}{2}ph''(\hat{x})] + \sigma_{w}^{2} .$$

8. Truncated Minimal-Variance

This is the filter derived by Eass, Norum, and Schwarts10.

$$\frac{d\hat{x}}{dt} = f(\hat{x}) + \frac{1}{2} f''(\hat{x})p + \sigma_{\pi}^{-2}ph'(\hat{x})[y - h(\hat{x}) - \frac{1}{2}ph''(\hat{x})]$$

$$\frac{dp}{dt} = apf'(\hat{x}) - \sigma_{\pi}^{-2}p^{2}h'^{2}(\hat{x}) - \frac{1}{2}\sigma_{\pi}^{-2}p^{2}h''(\hat{x})[y - h(x) - \frac{1}{2}ph''(\hat{x})] + \sigma_{\pi}^{2},$$
(40)

#### 4. Modified Minimal-Variance

This filter is a compromise between Equations (39) and (40, which is based on difference in the driving terms in the p equation. By dropping the driving term, the filter is simpler, yet the response fails between the responses for the two preceding filters.

$$\frac{d\hat{x}}{dt} = f(\hat{x}) + \frac{1}{2} f''(\hat{x}) p + \sigma_{w}^{-2} ph'(\hat{x}) [y - h(\hat{x}) - \frac{1}{2} ph''(\hat{x})]$$

$$\frac{dp}{dt} = 2pf'(\hat{x}) - \sigma_{w}^{-2} p^{2} h'^{2}(\hat{x}) + \sigma_{v}^{4}.$$

(41)

(39)

31 here the d 

#### 8. Maximum-Principle Least Squares

This filter is derived by Detehmendy and Sridhar<sup>11</sup> for minimizing an integral-square-estimation-error criterion, using deterministic techniques sy the use of Pontriagin's maximum principle the minimization is "reduced" to a two-point boundary-value problem which is solved by an invariant imbedding technique using an approximation to one boundary condition.

$$\frac{dx}{dt} = f(\hat{x}) + \sigma_{w}^{-2} p h'(\hat{x}) [y - (\hat{x})]$$

$$\frac{dp}{dt} = 2p f'(\hat{x}) - \sigma_{w}^{-2} p^{2} h'^{2} (\hat{x}) + \sigma_{w}^{-2} p^{4} h''(\hat{x}) [y - h(\hat{x})] + \sigma_{v}^{2} . \qquad (42)$$

Actually, Equation (42) is a special case of the derivation in Detchmendy and Sridhar<sup>11</sup>; they used arbitrary weighting functions in the criterion integrand, while Equation (42) corresponds to the particular set of weighting functions that result in the Kalman filter for linear dynamics.

#### 6. Dynamic Programming Least Squares

This filter is derived by Cox<sup>12</sup> for a criterion which is similar to that used for the previous filter. The minimization is effected by dynamic programming using a quadratic approximation to the cost function.

$$\frac{d\hat{x}}{dt} = f(\hat{x}) + \sigma_{w}^{-2} ph'(\hat{x}) [y - h(\hat{x})]$$

$$\frac{dp}{dt} = 2pf'(\hat{x}) - \sigma_{w}^{-2} p^{2} h'^{2}(\hat{x}) + \sigma_{w}^{2}$$
(43)

A look at Equation (43) shows that this filter is essentially equivalent to using linear filtering about the computed mean, a technique that had been used heuristically previously.

#### 7. Discrete-Measurement Minimul-Variance

This filter is derived by Jazwinsky<sup>13</sup> for a minimal-variance oritorion under the assumption that the measurements arrive at isolated instants. The form presented here is the limiting form for continuous measurements.

$$\frac{d\hat{x}}{dt} = f(\hat{x}) + \frac{1}{2}f''(\hat{x})p + \sigma_{w}^{-2}ph'(\hat{x})[y-h(\hat{x})]$$

$$\frac{dp}{dt} = 2pf'(\hat{x}) - \sigma_{w}^{-2}p^{2}h'^{2}(\hat{x}) - \frac{1}{4}\sigma_{w}^{-2}h''(\hat{x})[y-h(\hat{x})] + \sigma_{r}^{2}.$$
(44)

Since the evolution of the system itself is considered continuous, the portion of the equations related to updating the estimate in the absence of measurements agrees with Equations (39), (40), and (41). The loss of the term in b''(3) is due to the difference between the physical and mathematical approaches discussed in dection 2. In the derivation of Equation (44), Jazwinsky uses the approximation to the conditional density that is used to derive Equation (40).

#### 8. Modified Discrete-Measurement Minimal-Variance

This filter is related to Equation (44) in the same way that Equation (39) is related to Equation (40), i.e., the conditional density is assumed to be Gaussian.

$$\frac{dx}{dt} = f(\hat{x}) + \frac{1}{2}f''(\hat{x})p + \sigma_{W}^{-2}ph'(\hat{x})[y-h(\hat{x})]$$

$$\frac{dp}{dt} = 2pf'(\hat{x}) - \sigma_{W}^{-2}p^{2}h'^{2}(\hat{x}) + \sigma_{W}^{-2}p^{2}h''(\hat{x})[y-h(\hat{x})] + \sigma_{\Psi}^{2}.$$
(45)

The application of Equations (38)-(45) to Equations (36) and (37) is straightforward and is not carried out here.

. Cara da la cara da la ca

The computer program used for the simulation study includes a simple rectangular-rule integration with constant step-size. The use of constant step-size allows each filter to be compared on the basis of the sume pseudo-random sequence. The random number generator is a combination of a standard uniform random sequence routine plus an

approximate transformation to a Gaussian random sequence. The program has two modes of operation: in one mode, only a single estimation is made with a pre-specified initial condition for the state; in the other mode, several runs are made for random initial conditions, and the statistics of the estimation errors are computed. The output from the program is a computer-propared plot showing the timb-history of the filter response or the error statistics. The plots show every 20th point with linear interpolation between.

#### 4.3 Results

As roted earlier, the simulation study was not intended to be a complete investigation of the computational characteristics of nonlinear filters. Two standard cases were used, which were computationally docide when used with the step-size chosen. For both dynamical equations (D1) of Equation (36) and (D2) of Equation (37), the standard run consists of 5000 points 0 001 second apart for a total problem time of five seconds, with  $\sigma_v^2 = 1$  and  $\sigma_w^2 = 10$ . Only the initial condition differs. Figures 1 and 2 show the state x and the measurement y for the two standard cases. The figures show the nonlinear measurement, but the noise is so large that there is not too much apparent difference between the linear and nonlinear measurements.

In the response to the standard inputs, there is a large difference between the response of the linear systems and the nonlinear systems, while the various nonlinear systems are remarkably similar. Figures 3 and 4 show the estimation error for the two standard cases. As hight be expected, linear measurements help the linear system. What might not be expected is that the relative error performance of the nonlinear filters within the shaded region is different for the two cases. To show that the results in Figures 3 and 4 are not peculiar to the particular initial condition and pseudo-random sequences, three representative filters were chosen for a set of ten runs. The mean and mean-square errors are shown in Figures 5 and 6 for initial conditions with variance one for the first case and variance 1/4 for the second case. It can be seen that the error statistics are still quite close, practically ind/stinguishable from a mean-square error point of view. Moreover, the relative error performance of the chosen filters is qualitatively unchanged.

Walle on the toric of statistical runs, it is interesting to note the comparison between the output of the p equation for an initial condition of one standard deviation and the mean-square error for ten runs for one case, as shown in Figure 7 for filter 2, dynamics D1, and measurement M11. Moreover, it was found that the change in the mean-square error for a larger number of cases, up to 50, while noticeable, was relatively small.

The effects of changing the statistics of the random sequences are similar for all the filters, and are intuitively reasonable. Changing  $\sigma_{\psi}$  has little effect on the initial response, though the higher  $\sigma_{\psi}$  the worse the ultimate following. Contrariwing increasing  $\sigma_{\psi}$  slows the response noticeably without affecting the ultimate tracking. The error responses for filter 2 with D; and M11 are shown in Figures 8 and 9.

The runs demonstrating the correctness of the noise model show the effects of improperly matching the filter parameters to the noise statistics. The effect of the mismatch on  $x - \hat{x}$  and p is shown in Figures 10 and 11. For all the curves shown, the uctual pseudo-random inputs were taken from identical populations: only the filter parameters varied. Evidently a statistical mismatch can scriously affect the performance of the filter. It also appears that the value of p generated by the filter is a good index of the performance, even when the statistica are poorly matched.

The final aspect studied on the computer was the effect of sampling the data. Two approaches to sampling were considered: sample-and-hold and pulse sampling. For the sample-and-hold runs, the effective noise variance was obtained using a  $\Delta t$  equal to the sampling period, rather than the computation interval. For the pulse-sampling runs,  $\sigma_{\pi}^{-1}$  was made zero for those intervals during which no measurements were made. Both types of sampling were applied to filter 2 with dynamics D1 and measurements M11. The response plots are shown in Figure 12. As should be expected, sampling is detrimental to the initial rosponse of the filter, though ultimately the estimate settles in to the proper value. From the one case considered, sample-and-hold appears somewhat better than impulse sampling, which may be due to the smoothing effect of the zero-order hold.

There is one problem in mechanizing the filters, which has not yet been discussed, that requires much more investigation: the effects of the filter parameters and the step-size on computational stability. For some combinations of parameters  $\sigma_{\rm v}$  and  $\sigma_{\rm w}$  and initial covariance  $y(t_0)$ , the itep-size must be made small to stabilize the computation during the initial portion of the run. The reason for the instrbility is the large value of the derivatives and the consequent large truncation errors introduced in the integration. A particularly insidious form of computational instability arcse in the course of the simulation stability is possible to be in a conditionally stable region such that one pseudo-random sequence with a given statistical description results in a stable response, while another sequence from the same population does not.

Apparently, nonlinear filtering is superior to linear filtering, although no one nonlinear filter offers any clear-cut advantage over any other from a performance point of view for the problem considered. It also appears that the filters derived from a continuous model are quite amenable to sampled-data use.

#### ACKNOWLEUGMENT

The research described in this paper was supported in part by Air Force Grant AFOSR-699-67.











le na Carlana a Carlana Investigation

30)

11

in **C**ity in C

#### REFERENCES

MARCE CALIES CALES AND A CALESCALE AND A

1.43

and trainage matthe

			N THE MAN
1.	Kalman, R.E. Buoy, R.S.	New Results in Linear Filtering and Prediction Theory. Transactions, American Society of Mechanical Engineers, Scrive D, Journal of Basic Engineering, Vol.83, March 1961, pp.95-108.	
2.	Stratonovich, R.L.	Conditional Markov Processes. Theory of Probability and its Applications, Vol.5, No.2, 1960, pp. 156-178.	
3.	Schwartz, L. Stear, E.B.	Computational Yalidity of Approximate Nonlinear Minimal-Variance Filters. Informa- tion Sciences, to appear April 1960.	
4.	Schwertz, L. Bass, R.W.	Extensions to Optimal Multichannel Nonlinear Filtering. Hughes Aircraft Company, Culver City, California, Report SSD 80220R, Fobruary 21, 1988.	
5.	Fisher, J.R.	Conditional Probability Density Functions and Optimal Nonlinear Estimation. PhD in Engineering, University of California, Los Angeles, 1966.	
6.	Skorokhod, A.V.	Studies in the Theory of Random Processes. Addison-Wesley, Reading, Mans., 1965.	
7.	Gray, A.H. Caughey, T.K.	A Controversy in Problems Involving Random Parametric Excitation. Journal of Mathematics and Physics, Vol.44, September 1986, pp.288-298.	
8.	Wong, E. Zakai, M.	On the Relation Between Ordinary and Stochestic Differential Equations. Electronics Research Laboratory, University of California, Berkeley, California, Remort 64-26, August 11, 1964.	
<b>9</b> .	Buoy, P. S.	Nonlinear Filtering Theory. Institute of Electrical and Electronic Engineers, Transactions on Automatic Control, Vol.AC-10. April 1965, p.197.	
10.	Bass, R.W. et sl.	Optimal Multichannel Nonlinear Filtering. Hughes Aircraft Company, Culver Uity, California, Report SSD 500C4R, August 25, 1965.	
11.	Detchmently, D.M. Swidhar, R.	Sequential Estimation of States and Parameters in Nolsy Nonlinear Dynamical Systems. Joint Automatic Control Conference, Preprints of Papers (AIAA, AIChE, ABME, IEEE, ISA) June 22–25, 1965, Rensenlaer Polytechnic Institute, Troy, New York, pp.56–63.	
12.	Cox, H.	Estimation of State Variables via Dynamic Programming. Joint Automatic Control Conference, Preprints of Papers (AIAA, AIChE, ASME, IEEE, ISA) June 24-26, 1964, Stanford University, Stanford, California.	
13.	Jazwinsky, A.H.	Nonlinear Filtering with Discrete Observations. AIAA Paper 06-38, presented at the AIAA 3rd Aerospace Sciences Meeting, New York, NY, January 24-26, 1986.	





•1•1•

والمحاجبة والمحاجبة المحاجبة

ا المان بي المان المان من المان ا المان المولية المان ال

(ent) California California

State of the state











5.00

1.55552

2000000000











ς.





Fig. 10 Effect of statistical mismatch on estimation error

うない。 それないしょう へんれい しんかい ひんしょう しんしん しょうしょう

5



ROULD

ST. Office the

Ditti siente Ditti

Fig.11 Effect of statistical mismatch on computed covariance

というない

Sec. 6





## CHAPTER 8 - LINEAR SMOOTHING TECHNIQUES

## (POST-FLIGHT DATA ANALYSIS)

# by

## Herbert E. Rauch

### Lockheed Missiles and Space Company Palo Alto, California, USA

.

**、** 

PRECEDING PAGE BLANK

# 163

<u>, i</u>...

•

# 

#### CHAPTER 8 - LINEAR SMOOTHING TECHNIQUES (Post-Flight Data Analysis)

Herbert E. Rauch

#### 1. INTRODUCTION

2

#### 1.1 Buckground

Bince the time of Gauss<sup>1</sup> people have been interested in the problem of least-squares estimation. Gauss not only formulated the estimation and devised a technique for solving it (a sequential algorithm called Gaussian elimination, which is used on modern computers), but he applied this to determine the orbit of the asteroid Geree. The limitation on his solution was that it only applied to the came where the system to be estimated is deterministic and the errors on the individual measurements are uncorrelated. When a system is deterministic, its evolution can be completely specified from the initial conditions and the differential equations of the system. Celestial bodies obey Newton's laws of motion and disturbing forces due to other celestial bodies are known to a reasonable degree of accuracy, so for practical purposes their motion can be considered deterministic. Terrestrial bodies, on the other hand, are subject to disturbing forces whose effects cannot be described exactly beforehand although their statistical properties might be known. When a system is described by differential equations subject to disturbances which are statistically figs-varying, the system can be usiled a stochastic process.

The pioneer work of Wieners discussed and solved the problem of linear least-squares estimation for stochustic processes. However, his results were mainly for stationary, non-time-varying systems, and there was some difficulty in obtaining numerical answers for large order systems. Many papers have appeared since then giving different solutions to the problem under more general conditions. The most widely used solution for filtering and prediction is that of Kalman<sup>3,4</sup> and Buoy<sup>5,6</sup>. The primary advantage of their solutions is that they apply to time-varying systems and the equations which specify the optimum filter are in the form of recursive difference sequations which have obvious computational advantages. However, Kalman and Bucy did not consider the important problem of smoothing. Filtering allows one to estimate present values of the variables of interest using present data, while smoothing allows one to estimate past values. A number of papers have presented different derivations of recursive solutions for the smoothing solution from different points of view for both discrete and continuous systems. Bryson and Frazior' gave one of the first recursive solutions for continuous systems using the calculus of variations and the method of maximum likelihood. Gimilar solutions with some alternative forms were given by Cor<sup>8</sup> using Dynamic Programming, Rauch, Tung and Striebel<sup>9</sup> using discrete time manipulations of Bayes's rule and the method of Maximum Likelihood, Lee<sup>10</sup> using the calculus of variations, and Meditoh<sup>11</sup> using projection arguments. Two more recent results were by Fraser's who expressed the solution as the combination of two filtered estimates from a forward and backward Kalman-Bucy filter, and Kallath and Frost<sup>13,14</sup> who made use of an innovations approach to linear smoothing. In their papers Kailath and Frost also give an interesting summary of work in filtering and smoothing with a list of over forty references,

Obviously, all these solutions should ultimately give the same answer to the same problem, but the different derivations give insight into the reasons for different recursive forms of the smoothing solution. For deterministic systems with linear equations, the solution simplifies considerably because the smoothed estimate of past velues is obtained directly from the filtered estimate of the latest value by integrating the differential equations of motion backwards. For linear smoothing of stochastic systems a similar procedure can be used in which differential equations are integrated backwards, starting with the latest filtered estimate. However, there are additional inputs to the differential equations involving earlier filtered estimates or measurements, depending on the form of the solution. The purpose of this presentation is to cultime the procedures and solutions which can be used in the linear emothing problem and to indicate how the result can be applied to a problem in post-flight data analyses which involves tracking a satellite disturbed by stochastic drag forces<sup>18</sup>.

#### 1.1 Notation

In general, lower case letters, such as u, v, x, and x, will denote column vectors while upper case letters, such as A, B, F, G, and  $\Phi$ , will denote matrices. The letter I represents the identity matrix. The superscript prime, as in A' or u', will denote the transpose of the matrix or the transpose of a column vector (which becomes a row vector). The product of a column vector and a row vector, such as xx' is a matrix. The symbol E represents expected value so that E[A] represents the expected value of the quantity A. The superscript -1 as in  $A^{-1}$  means the inverse of the matrix A. In all cases it will be assumed that the inverse of the matrix exists, although quite often, the inverse can be replaced by a pseudo-inverse or generalized inverse without changing the results. **PRECEDING PAGE BLANK** 

*ۿ؆؞؞؆ۮۿ؆ڰۿ*ٵ؞؉؆؞؞؇؞؞ڹ؞؇؞؇؞؇؞

#### 1.3 'Linear Estimation

For the discrete version of the linear estimation problem the system to be estimated can be described by the following set of matrix difference equations:

$$x_{k+1} = \Phi_{k+1} x_k + u_k$$
 for  $k = 1, 2, ..., N$ . (1.1)

The linear measurements obtained from the system are given by another set of matrix equations,

$$\mathbf{x}_{k} = \mathbf{H}_{k}\mathbf{x}_{k} + \mathbf{v}_{k}$$
 for  $k = 1, 2, \dots, N$ . (1.2)

The vector x represents the state of the system which is to be estimated while the vector x represents the known measurements. The vectors u and v are not known exactly, but they are zero mean independent random variables with known covariance. The variable u represents random changes in the state (dynamic noise) while the variable v represents random changes in the measurements (measurement noise). The subscript k represents the value of the quantities at the time of the  $k^{th}$  measurements (measurement noise) u is identically zero for all time, the system is said to be deterministic. The matrices  $\frac{\Phi}{2}$  (transition matrix) and H (output matrix) represent known quantities which can change from one measurement to the past.

Most physical systems will involve nonlinear equations, but it is assumed the above set of linear equations can be obtained by linearising about some nominal values for the state and the measurements. It may be that the physical system is governed by a set of linear (or linearized) differential equations rather than difference equations, although the measurements will take place at discrete times. In that case the original system differential equations must be integrated to obtain the required difference equation relating the change in state from one measurement to the next. Conversely, under certain conditions, in the limiting case as the time between measurements goes to zero, the discrete system will approach a continuous system.

The optimum estimate will be the linear estimate which minimizes the mean square error. Calculating the estimate requires knowing the mean and covariance of all the random variables of interest, but no higher moments. If all the random variables of an another moments, if all the random variables have a normal probability distribution, the estimate will be the conditional mean of the state given the measurements. Sometimes the estimate is also called the Maximum Likelihood estimate because it maximizes the conditional probability distribution.

Let  $\hat{x}_{j/k}$  denote the optimum estimate of the state  $x_j$  given all the measurements up to  $z_k$ . If j is greater than or equal to k, it is called filtering and prediction. If j is less than k, it is called smoothing. The optimum filtering solution as derived by Kalman, can be written, where B is the matrix gain on the Kalman filter,

$$\frac{\mathbf{\hat{s}}_{k+1/k} = \Phi_{k+1/k} \mathbf{\hat{s}}_{k/k-1} + \mathbf{\hat{s}}_{k}}{\mathbf{\hat{s}}_{k} = \mathbf{\hat{s}}_{k} - \mathbf{\hat{s}}_{k} \mathbf{\hat{s}}_{k/k-1}}$$
(1.3)

The initial conditions for the filtering solution are based on the  $a \ priori$  information. The quantity  $E_k$ , which contains the new information, is the difference between the actual measurement  $E_k$  and the best estimate of the measurement. This quantity will play an important part in the derivation of the moothing solution.

The above filtering solution is recursive in the sense that the desired estimate is calculated sequentially from previous estimates. The smoothing solution can also be written in a recursive form and, in fact, four different recursive versions of the smoothing solution will be presented here. In theory, these four versions will give identical results but, due to such things as computer programming or running time or the effect of roundoff errors, one way may be superior to another for a particular problem.

The first version, called fixed point smoothing, is useful when it is necessary to calculate the smoothed estimate at only one point. For instance, the smoothed estimate of only the initial conditions of the state might be desired. The form of the solution for fixed point emothing is similar to that for filtering as shown below, where the matrix gain B is a general weighting coefficient:

$$\hat{\mathbf{x}}_{\mathbf{k}/\mathbf{M}} = \hat{\mathbf{x}}_{\mathbf{k}/\mathbf{M}=1} + \mathbf{B}\hat{\mathbf{x}}_{\mathbf{M}} , \qquad (1,4)$$

المعاقبة والمعاوية والمحافظ

The second version, smoothing using filtered estimates, has two parts. First, the Kalman filtering solution is used to process the measurements sequentially, in the order in which they were received, to obtain the filtered estimate (forward aweep). Second, the smoothing solution is used to process the filtered estimate a second time, in reverse order, from the last to the first, to obtain the smoothed estimated (backward aweep). The initial condition for the backward sweep is the filtered estimate of the state at the final time, which is also the smoothed estimate of the state at that time. The form of the solution for smoothing with filtering is shown below, where the matrices A and B are general weighting coefficients:

$$\hat{\mathbf{x}}_{\mathbf{k}/\mathbf{N}} = \mathbf{A} \hat{\mathbf{x}}_{\mathbf{k}+1/\mathbf{N}} + \mathbf{B} \hat{\mathbf{x}}_{\mathbf{k}/\mathbf{k}} \, .$$

There is an alternative method in which the backward sweep involves measurements instead of the filtered estimates. This is the third version, which is called smoothing using measurements. The recursive solution can be written schematically as shown below, where the vector  $w_{\rm h}$  is also calculated recursively:







Carles Contractor

(2.2)

$$\hat{\mathbf{x}}_{\mathbf{k}/\mathbf{N}} = A\hat{\mathbf{x}}_{\mathbf{k}+1/\mathbf{N}} + B\mathbf{w}_{\mathbf{k}}$$
  
 $\mathbf{w}_{\mathbf{k}+1} = C\mathbf{w}_{\mathbf{k}} + D(\mathbf{x}_{\mathbf{k}} - H_{\mathbf{k}}\hat{\mathbf{x}}_{\mathbf{k}/\mathbf{N}})$ 

The fourth version, called fixed lag smoothing, can be used when it is desirable to smooth for a fixed number of measurements. The recursive solution involves both filtured estimates and measurements:

$$\mathbf{x}_{k+1/N+1} = \mathbf{A}\mathbf{x}_{k/N} + \mathbf{B}\mathbf{E}_{N+1} + \mathbf{C}\mathbf{x}_{k/k}$$

#### 1.4 Outline of Work

The remainder of this presentation examines the smoothing solution for the disorete system (Section 2) and the continuous system (Nection 3), and applies it to a problem in satellite tracking (Section 4). It is shown that the optimum estimate for the disorete system must satisfy the sampled Wiener-Nopf equation and an innovations approach is used to derive a solution. The innovations approach presented here does not have the generality and the elegant mathematics of that of Kailath and Frost<sup>13,14</sup>, but it uses some of their ideas. Four different versions of the smoothing solution and the associated covariance are presented.

The relations between the discrite system and the continuous system are examined and the continuous smoothing solutions are presented as the limiting once of the discrete ones. Finally, a problem is outlined involving a stochastic model for satellite drag, and the smoothing solution is applied to actual data.

#### 2. DISCRETE TIME SMOOTHING

#### 1.1 Statement of the Problem

The system to be estimated is described by a set of matrix difference equations,

$$\pi_{k+1} = \Phi_{k+1} \pi_k + u_k$$
 for  $k = 1, 2, ... N$ , (2.1)

The measurements obtained from the system are defined by the equations

where

and

ug = an n×1 column vector of statem (input) uncorrelated dynamic noise

Vy = an rx1 column vector of measurement (output) uncorrelated noise

 $\mathbf{z}_{k} = \mathbf{H}_{k}\mathbf{x}_{k} + \mathbf{v}_{k}$  for k = 1, 2, ... N,

5. = an rx1 olumn vector of known measurements

 $\Phi_{k+1}$  = the n×n transition matrix with known coefficients

H<sub>k</sub> = an r×n matrix with known coefficients.

The dimensions of the measurement  $z_k$  can vary from point to point as long as the dimensions of the associated vectors and matrices also vary accordingly. The vectors  $u_k$  and  $v_k$  represent vector-valued independent random variables with zero mean and a known covariance.

E(uku})	3	۹	if	X	=	j	and mero otherwise	
E(V <sub>k</sub> vj)	=	R <sub>k</sub>	it	k	ਕ	ł	and zero otherwise	
$E(u_k v_j^{\prime})$	×	ο.						(2.3)

In some treatments of this problem, it is assumed the vectors  $u_{\mu}$  and  $v_{\mu}$  are correlated if k = j. This slight modification will be discussed later. The matrices Q and R could be singular and the implications of this are also discussed later, although for the time being it will be assumed both matrices are non-singular.

In any real optimation problem there is an important distinction between a priori information and autual measurements. However, in order to simplify the notation and submequent explanations, it will be convenient to treat the a priori information formally as the initial measurement  $z_n$ , as shown below:

$$\mathbf{x}_0 = \mathbf{x}_0 + \mathbf{v}_0 \quad \text{where} \quad \mathbf{x}_0 = \mathbf{E}[\mathbf{x}_0]$$

$$\left[\mathbf{v}_0 \mathbf{v}_0'\right] = \mathbf{R}_0 = \mathbf{E}[(\mathbf{x}_0 \cdots \mathbf{x}_0)(\mathbf{x}_0 - \mathbf{x}_0)'] \quad (2.4)$$

Thus, the a priori information about the initial value of the state  $x_0$  is that it is a random variable with mean  $x_0$  and covariance  $R_0$ .

The state  $x_k$  is the quantity to be estimated. The best estimate of the state  $x_k$ , given the measurements  $s_j$  for  $j \in 0, 1, ..., N$ , is denoted by  $\Re_{k/N}$ , and it is a linear combination of the measurements with coefficients given by the nxr weighting matrix  $A_{k/j-1}$ 









				-	. )
۱. ۱			· · ·		
					-
	يەر يەرىيە ئىرى ھ	a na Bain In an Bail	میں است استاد کا	ىسىھر. 1 - 1	

151 TO 00127

$$\hat{x}_{k/N} = \sum_{j=0}^{N} A_{k/j} x_{j}$$
 (2.5)

The estimation procedure is called filtering if k equals N, prediction if k is greater than N, and smoothing if k is less than N. The error in the estimate, denoted  $X_{k/N}$ , is the difference between the state and the best estimate of the state:

$$\mathbf{\tilde{x}}_{k/N} = \mathbf{x}_{k} - \hat{\mathbf{x}}_{k/N} = \mathbf{x}_{k} - \sum_{j=0}^{N} \mathbf{A}_{k/j} \mathbf{x}_{j}.$$
 (2.6)

The partial derivative of the error with respect to the coefficients  $A_{k/j}$  is equal to the measurement  $z_{i,j}$ 

$$\partial X_{k/N} = - x_j . \qquad (2.7)$$

The best estimate of the state will be defined as the linear estimate which minimizes the mean square error, which is the trace of the covariance of the error  $P_{k/N}$ :

$$P_{k/N} = \mathbb{E}(\bar{x}_{k/N} \bar{x}_{k/N}), \qquad (2.8)$$

Bince the linear coefficients  $A_{k/j}$  are chosen to minimize the mean square error, the condition for this to be a minimum is that the partial derivative with respect to the coefficients is zero,

$$\partial TR(P_{k/N})/\partial A_{k/j} = 0$$
 for  $j = 0, 1, ..., N$ , (2.9)

Substituting the equation for the error into the covariance gives an expression for the mean square error which is quadratic in the coefficients. Carrying out the differentiation gives an expression which is linear in the coefficients - the sampled Wiener-Hopf equation - which is sufficient to determine the coefficients.

$$\partial TR(P_{k/N})/\partial A_{k/j} = 0 = -2E(3_{k/N}E_j)$$
 for  $j = 0, 1, ..., N$ . (2.10)

This expression can be interpreted by saying the error in the best estimate,  $\bar{x}_{k/N}$ , is uncorrelated with any of the measurements which made up that estimate. Since the best estimate is a linear combination of the measurements, this means the error in the best estimate is uncorrelated with the estimate itself.

$$\mathbb{E}(\hat{X}_{k/N} | \hat{X}_{j}) = 0$$
 for  $j = 0, 1, ... N$   
 $\mathbb{E}(\hat{X}_{k/N} | \hat{X}_{j/N}) = 0$ . (2.11)

#### 2.2 Innovations Approach

C 1554253 - DC 652 65 - DO 654 - DO 654 - DO

The innovation is the new information which comes from each measurement. The innovations approach to linear estimation is based on the assumption that the actual measurements and the innovation process (all the innovations) are equal in the sense that they contain the same information as far as linear operations are concerned. The new information is the difference between the actual measurement  $\mathbf{x}_k$  and the best estimate of the measurements. The best estimate of the measurement is the output matrix  $\mathbf{H}_k$  times the best estimate of the state  $\hat{\mathbf{x}}_{k/k-1}$  while the new information, denoted by  $\mathbf{B}_k$ , is the error in this estimate,

$$D_{k} = E_{k} - H_{k} X_{k} / k_{-1} = H_{k} X_{k} / k_{-1} + V_{k}$$
  
since  $X_{k} = H_{k} X_{k} + V_{k}$ . (2.12)

The best estimate of the measurement has the same properties as the best estimate of the state discussed previously, while the new information has the same properties as the error. The new information of the k<sup>th</sup> measurement is uncorrelated with any previous measurements and the new information is uncorrelated with the estimate. These properties of the new information can be used to derive a sequential estimation procedure.

Assume the current best estimate  $\hat{x}_{k/N-1}$  can be calculated as the sum of the previous best estimate  $\hat{x}_{k/N-1}$  and a coefficient times the new information  $\tilde{x}_{N}$ :

$$\hat{\mathbf{x}}_{k/N_{-1}} = \hat{\mathbf{x}}_{k/N_{-1}} + \mathbf{x}_{k/N} \hat{\mathbf{x}}_{N}$$
, (2.13)

If the coefficient  $B_{n/N}$  can be chosen so the error in the current best estimate is uncorrelated with any of the measurements, it will indeed be the best estimate because it satisfies the sampled Wiener-Hopf equation. The error in the current best estimate is composed of two terms, the error in the previous best estimate, and the error in the new information:

$$\mathbf{I}_{\mathbf{k}/\mathbf{N}} = \mathbf{I}_{\mathbf{k}} - \mathbf{I}_{\mathbf{k}/\mathbf{N}} = \mathbf{I}_{\mathbf{k}/\mathbf{N}-1} - \mathbf{I}_{\mathbf{k}/\mathbf{N}} \mathbf{I}_{\mathbf{k}}, \qquad (2.14)$$

Because they are errors in best estimates, these two terms are uncorrelated with the previous measurements,

Diff to the the total

(2.15)

. 109

St. Agreets i i

Č,

For the current measurement  $z_N$  the condition that the error be uncorrelated determines the coefficient  $B_{k/N}$ .

$$0 = E(\bar{x}_{k/N} \bar{z}_{N}') = E(\bar{x}_{k/N-1} \bar{z}_{N}') - B_{k/N} E(\bar{x}_{N} \bar{z}_{N}'), \qquad (2.10)$$

Therefore, the sequential estimation procedure in (2.13) gives the best estimate as long as the coefficient  $B_{k/N}$ satisfies Equation (2.16). For practical computation a more convenient form for the coefficient is obtained by using the following relations:

$$E(\mathbf{\tilde{x}}_{k/N-1}\mathbf{z}_{N}^{\prime}) = E(\mathbf{\tilde{x}}_{k/N-1}\mathbf{\tilde{x}}_{N}^{\prime}) = E(\mathbf{x}_{k}\mathbf{\tilde{x}}_{N}^{\prime})$$

$$E(\mathbf{\tilde{x}}_{N}\mathbf{z}_{N}^{\prime}) = E(\mathbf{\tilde{x}}_{N}\mathbf{\tilde{x}}_{N}^{\prime}) . \qquad (2.17)$$

These relations allow the coefficient  $B_{k/N}$  to be written in the form

$$\hat{\mathbf{x}}_{\mathbf{k}/\mathbf{N}} = \hat{\mathbf{x}}_{\mathbf{k}/\mathbf{N}-1} + \mathbf{B}_{\mathbf{k}/\mathbf{N}}\mathbf{z}_{\mathbf{N}}$$
$$\mathbf{B}_{\mathbf{k}/\mathbf{N}} = \mathbf{E}(\mathbf{x}_{\mathbf{k}}\hat{\mathbf{z}}_{\mathbf{N}}')\mathbf{E}(\tilde{\mathbf{z}}_{\mathbf{N}}\hat{\mathbf{z}}_{\mathbf{N}}')^{-1} . \qquad (2.18)$$

The noise  $v_{k}$  is uncorrelated with the error in the best estimate so the covariance of the new information  $Z_{k}$  can be calculated directly using (2.12),

$$\begin{split} \mathbf{E}(\mathbf{B}_{k}\mathbf{B}_{k}^{\prime}) &= \mathbf{H}_{k}\mathbf{E}(\mathbf{X}_{k/k-1}\mathbf{R}_{k/k-1}^{\prime})\mathbf{H}_{k}^{\prime} \\ &+ \mathbf{E}(\mathbf{v}_{k}\mathbf{v}_{k}^{\prime}) &= \mathbf{H}_{k}\mathbf{P}_{k/k-1}\mathbf{H}_{k}^{\prime} + \mathbf{R}_{k} , \end{split}$$
(2.19)

Multiplying the sequential error (2.14) by the state  $x_k$  gives a sequential procedure for calculating the covariance of the error,

$$E(\mathbf{x}_{k/N}\mathbf{x}_{k}') = E(\mathbf{x}_{k/N-1}\mathbf{x}_{k}') - \mathbf{B}_{k/N}E(\mathbf{x}_{N}\mathbf{x}_{k}')$$

$$P_{k/N} = P_{k/N-1} - \mathbf{B}_{k/N}E(\mathbf{x}_{N}\mathbf{x}_{k}')$$

$$= P_{k/N-1} - \mathbf{B}_{k/N}E(\mathbf{x}_{N}\mathbf{x}_{N}')\mathbf{B}_{k/N}'$$

$$E(\mathbf{x}_{k/1}\mathbf{x}_{k}) = P_{k/4} , \qquad (2.20)$$

since

and in the sources

To summarize, the acquential form of the optimum estimate and the associated covariance equation are given by (2.18) and (2.20). The form of the solution is quite general and does not depend on the system equations. The weighting coefficient  $B_{k/N}$  involves two covariances. One of these is given in (2.19) with k = N, but the other covariance is as yet unknown. When N = k - 1 the above equations apply to filtering and prediction. When N > k the equations are for fixed point smoothing because they show the change in the smoothed estimate at the  $k^{th}$  point.

#### 2.3 Filtering and Fixed Point Smoothing

Up to now it has not been necessary to examine in detail the system equations for the state or the measurement. However, to calculate the coefficient  $B_{e/N}$ , for the sequential estimation procedure, it will be necessary to find a way of calculating the corresponding povariances. Consider first the single-step estimation problem in (2.18), where k is replaced by k + 1 and N is replaced by k. This is the familiar Kalman filter equation, where  $B_{k+1/k}$  is the gain matrix for the filter,

$$\hat{x}_{k+1/k} = \hat{x}_{k+1/k-1} + \hat{B}_{k+1/k}\hat{a}_{k}$$
  
 $\hat{B}_{k+1/k} = \hat{x}(x_{k+1}\hat{x}_{k}) \hat{x}(\hat{x}_{k}\hat{x}_{k}^{\prime})^{-1}$ . (3.21)

For one-step estimation the unknown covariance and the gain matrix can be determined directly. ....

$$\mathbf{g}(\mathbf{x}_{k+1}\mathbf{x}_{k}') = \mathbf{g}(\mathbf{x}_{k+1}\mathbf{x}_{k} + \mathbf{u}_{k})(\mathbf{x}_{k}\mathbf{x}_{k/k-1} + \mathbf{v}_{k})' ] \\
 _ = \mathbf{g}(\mathbf{x}_{k+1}\mathbf{x}_{k} + \mathbf{x}_{k/k-1}\mathbf{x}_{k}') \\
 \mathbf{g}_{k+1/k} = \mathbf{g}_{k+1}\mathbf{x}_{k/k-1}\mathbf{x}_{k}' (\mathbf{x}_{k}\mathbf{x}_{k/k-1}\mathbf{x}_{k}' + \mathbf{x}_{k})^{-1} . \quad (2.22).$$

....

. . . . .

It should be pointed out here that, for the special case where the dynamic noise un and the measurement noise  $v_k$  are correlated, the unknown covariance has an additional term:

.....

$$\mathbb{E}(\mathbf{x}_{k+1}\mathbf{x}_{k}') = \Phi_{k+1}\mathbf{F}_{k/k-1}\mathbf{H}_{k}' + \mathbb{E}(\mathbf{u}_{k}\mathbf{v}_{k}').$$

To complete the molution to the one-step estimation problem, it is necessary to calculate a sequential relation for the covariance of the error PR+1/K which is used in the gain matrix. The best prediction must be the bost estimate times the transition matrix,

$$\mathfrak{L}_{\mathbf{k}+1/\mathbf{k}-1} = \mathfrak{P}_{\mathbf{k}+1} \mathfrak{L}_{\mathbf{k}/\mathbf{k}-1}$$

STATE PROVINCE MADE

The error in the best estimate is

$$\begin{aligned} \mathbf{x}_{k+1}/\mathbf{k} &= \mathbf{x}_{k+1} - \mathbf{x}_{k+1}/\mathbf{k}_{-1} \\ &= (\Phi_{k+1}\mathbf{x}_{k} + \mathbf{u}_{k}) - [\Phi_{k+1}\hat{\mathbf{x}}_{k}/\mathbf{k}_{-1} + \mathbf{B}_{k+1}/\mathbf{k}(\mathbf{H}_{k}\bar{\mathbf{x}}_{k}/\mathbf{k}_{-1} + \mathbf{v}_{k})] \\ &= \psi_{k+1}\bar{\mathbf{x}}_{k}/\mathbf{k}_{-1} + \mathbf{u}_{k} - \mathbf{B}_{k+1}/\mathbf{k}\mathbf{v}_{k} , \\ \psi_{k+1} &= \Phi_{k+1} - \mathbf{B}_{k+1}/\mathbf{k}\mathbf{H}_{k} , \end{aligned}$$
(2.23)

where

The covariance of the error is obtained by multiplying the error by the state and taking the expected value:

$$P_{k+1/k} = \mathbb{E} \langle \bar{X}_{k+1/k} X_{k+1}^{l} \rangle$$
  
=  $\mathbb{E} \bar{X}_{k+1/k} \langle \Phi_{k+1} Y_{k} + u_{k} \rangle^{l}$   
=  $\psi_{k+1} P_{k/k-1} \Phi_{k+1} + Q_{k}$ . (2.24)

The covariance for the current estimate can be written in a similar way,

 $P_{k/k} = \Phi_{k+1}^{-1} \psi_{k+1} P_{k/k-1} = (I - B_{k/k} H_k) P_{k/k-1},$  $B_{k/k} = \Phi_{k+1}^{-1} B_{k+1/k}$ 

where

$$P_{k+1/k} = \Phi_{k+1} P_{k/k} \Phi'_{k+1} + \Phi_{k} . \qquad (2,25)$$

These are Kalman filter equations for the onn-step estimation problem. To extend this to the smoothing problem, where N is greater than k, it is necessary to see how the errors in the estimate evolve. The matrix  $\psi_{k+1}$  in (2.23) shows how the estimation errors evolve from one point to the next point. Applying this to the points from k to N one can write

$$\bar{\mathbf{x}}_{N/N-1} = \psi_N \psi_{N-1} \dots \psi_{k+1} \bar{\mathbf{x}}_{k/k-1} + \cdots + \mathbf{terps} \text{ involving } u_k \text{ and } v_k \text{ for } k \leq 1 \leq N-1.$$
(3.26)

Since  $x_k$  is uncorrelated with the  $u_j$  and  $v_j$ , the unknown covariance can be written as

$$\mathbb{E}(\mathbf{x}_{k} \mathbf{B}_{k}^{\prime}) = \mathbb{E}[\mathbf{x}_{k} (\mathbf{H}_{N} \mathbf{X}_{N/N-1} + \mathbf{v}_{N})^{\prime}]$$

$$= \mathbb{E}(\mathbf{\overline{x}}_{k/k-1} \mathbf{\overline{x}}_{N/N-1}^{\prime}) \mathbf{H}_{N}^{\prime}$$

$$= \mathbb{P}_{k/k-1} \psi_{N/k}^{\prime} \mathbf{H}_{N}^{\prime} \cdot$$

$$\psi_{R/k} = \psi_{N} \psi_{N-1} \cdots \psi_{k+1}$$

$$\psi_{j+1} = \hat{\mathbf{y}}_{j+1} \sim \mathbf{B}_{j+1/j} \mathbf{H}_{j} \cdot$$
(2.27)

where

and .

#### 2.4 Other Smoothing Solutions

The fixed point smoothing solution given by (2.13), (2.19) and (2.27) will now be used to derive other recursive versions of the smoothing solution. At each point where the new information  $\mathbf{z}_j$  is received, it is multiplied by the weighting matrix  $\mathbf{B}_{k/j}$  in (2.18). The smoothed estimates at points k and k + 1 can be written as the weighted sum of all the new information.

$$\hat{x}_{k/N} = \hat{x}_{k+1/N} = \hat{x}_{k+1/k} - \sum_{j=k+1}^{N} \hat{u}_{k+1/j} \hat{z}_{j}$$
(2.28)

From (2.27) the previously unknown covariances and the weighting coefficient can be related to a new matrix  $C_k$  ,

$$\begin{split} \mathbf{E}(\mathbf{x}_{k} \mathbf{\hat{z}}_{N}^{\prime}) &= P_{k/k-1} \psi_{k+1}^{\prime} \psi_{N/k+1}^{\prime} \mathbf{H}_{N}^{\prime} \\ &= C_{k} P_{k+1/k} \psi_{N/k+1}^{\prime} \mathbf{H}_{N}^{\prime} \\ &= C_{k} \mathbf{E}(\mathbf{x}_{k+1} \mathbf{\hat{z}}_{N}^{\prime}) , \end{split}$$

 $C_{k} = P_{k/k-1} \psi_{k+1}' P_{k+1/k}$ 

(2.29)



1 hearth -

where

(2.30)

For the weighting coefficients in (2.28),

 $B_{k/j} = C_k B_{k+1/j} .$ 

since

$$\mathbf{B}_{\mathbf{k}/j} = \mathbf{E}(\mathbf{x}_{\mathbf{k}} \mathbf{Z}_{j}') \mathbf{E}(\mathbf{Z}_{j} \mathbf{Z}_{j}')^{-1} ,$$

#100

Subtracting  $C_k$  times the second equation in (2,28) from the first equation, and making use of (2,30) to cancel out the terms involving the new information, gives a relation between the smoothed estimate at points k and k + 1,

$$\bar{\mathbf{x}}_{k/N} - \bar{\mathbf{x}}_{k/k} = C_k (\bar{\mathbf{x}}_{k+1/N} - \bar{\mathbf{x}}_{k+1/k}) , \qquad (2.31)$$

This second version is called smoothing with filtered estimates. The smoothed estimate at k is calculated from the smoothed estimate at k + 1 and the filtered estimate. In the same way, the smoothed covariance (2.20) can be written as the sum of terms

$$P_{\mathbf{n}/N} = P_{\mathbf{n}/k} - \sum_{j=k+1}^{n} B_{\mathbf{n}/j} E(\widetilde{z}_j \widetilde{z}_j') B_{\mathbf{n}/j} \quad \text{for } \mathbf{n} = k \text{ and } k = 1.$$
 (2.32)

Premultiplying the equation with m = 1 + 1 in (2.31), by  $C_k$  and post-multiplying by  $C'_k$ , subtracting from the equation with m = k, and making use of (2.30) gives a relation between the covariance of the smoothed estimates at points k and k + 1.

$$P_{k/N} = P_{k/k} = C_k (P_{k+1/N} - P_{k+1/k}) C_k^{\dagger}, \qquad (2.33)$$

An alternative definition of the matrix  $C_k$  which is useful is obtained from (2.25),

$$C_{k} = P_{k/k} \Phi_{k+1}' P_{k+1/k} + P_{k+1/k} + P_{k/k} \Phi_{k+1}' = P_{k/k} \Phi_{k+1}' + P_{k+1/k} + P_{k+1/k}$$

since

. A slightly different form of smoothing using the filtered estimates is obtained by grouping terms and making use of some identities, in (2.24). Multiplying (2.31) by  $\Phi_{k+1}$  and subtracting  $\hat{x}_{k/N}$  gives

$$\begin{split} \Phi_{\mathbf{k}+1} \hat{\mathbf{x}}_{\mathbf{k}/N} &- \hat{\mathbf{x}}_{\mathbf{k}+1/N} &= (\Phi_{\mathbf{k}+1} \mathbf{C}_{\mathbf{k}} - \mathbf{I}) (\hat{\mathbf{x}}_{\mathbf{k}+1/N} - \hat{\mathbf{x}}_{\mathbf{k}+1/\mathbf{k}}) \\ &= - \mathbf{Q}_{\mathbf{k}} \mathbf{P}_{\mathbf{k}+1/\mathbf{k}}^{-1} (\hat{\mathbf{x}}_{\mathbf{k}+1/N} - \hat{\mathbf{x}}_{\mathbf{k}+1/\mathbf{k}}) , \\ \Phi_{\mathbf{k}+1} \mathbf{C}_{\mathbf{k}} &= \Phi_{\mathbf{k}+1} \mathbf{P}_{\mathbf{k}/\mathbf{k}-1} \psi_{\mathbf{k}+1}^{-1} \mathbf{P}_{\mathbf{k}+1/\mathbf{k}}^{-1} \\ &= (\mathbf{P}_{\mathbf{k}+1/\mathbf{k}} - \mathbf{Q}_{\mathbf{k}}) \mathbf{P}_{\mathbf{k}+1/\mathbf{k}}^{-1} , \qquad (2.35) \end{split}$$

since

with

The smoothing solution with measurements, the third version, is written in a similar form, but it requires some detailed algebra, making use of complicated identities, to show the equivalence. The measurement smoothing involves a new vector  $\omega_k$  which is calculated sequentially and, in the continuous version of the problem, can be interpreted as a fagrange multiplier. The smoothing solution involving measurements is

$$\Phi_{k+1}\hat{x}_{k/N} - \hat{x}_{k+1/N} = - Q_{k}\omega_{k}$$

$$\omega_{k-1} = \Phi_{k+1}'\omega_{k} + H_{k}'R_{k}^{-1}(z_{k} - H_{k}\hat{x}_{k/N}),$$

$$\omega_{k} = 0. \qquad (2.36)$$

Comparing (2.35) and (2.36) gives a direct interpretation of the vector  $\omega_{\rm g}$ , but no hint how to derive the recursion formula for  $\omega_{\rm g}$ ,

$$\omega_{k} = P_{k+1/k}^{-1} (\hat{x}_{k+1/k} - \hat{x}_{k+1/k}) , \qquad (2.37)$$

In order to derive the recursion formula it will be necessary to make use of the two relations

----

----

والمراجع والمحافظ والم

- - 1

$$P_{k}^{i} \lambda_{k} = P_{k}^{i} \lambda_{k-1} + H_{k} R_{k}^{i} H_{k}$$

$$P_{k}^{i} \lambda_{k}^{i} \lambda_{k} = P_{k-1}^{i} \hat{Y}_{k/k-1} + H_{k} R_{k}^{i} \mathbf{z}_{k} , \qquad (2.38)$$

These relations arise naturally when using the maximum likelihood derivation, but they seem artificial here. The first relation can be proved by multiplying  $P_k^{-1}_k$  by  $P_{k/k}$  in (2.25) to get the identity. The second relation can be proved by showing that

$$B_{k/k} = P_{k/k} H_k^{\prime} R_k^{-1}$$
 (2.39)

Starting with (2.37) and making use of the two relations as well as (2.31) gives the desired recursive relation for  $\omega_{\rm g}$ .



where

$$\begin{aligned} \hat{P}_{k+1}^{\prime} \omega_{k} &= \hat{\Phi}_{k+1}^{\prime} P_{k+1}^{-1} / k \left( \hat{x}_{k+1/N} - \hat{x}_{k+1/k} \right) \\ &= P_{k}^{-1} / k C_{k} \left( \hat{x}_{k+1/N} - \hat{x}_{k+1/k} \right) \\ &= P_{k}^{-1} / k \left( \hat{x}_{k/N} - \hat{x}_{k/k} \right) \\ &= (P_{k}^{-1} / k - 1 + H_{k}^{\prime} R_{k}^{-1} H_{k}) \hat{x}_{k/N} - P_{k}^{-1} \hat{x}_{k/k-1} - H_{k}^{\prime} R_{k}^{-1} x_{k} \\ &= (P_{k}^{-1} / k - 1 + H_{k}^{\prime} R_{k}^{-1} H_{k}) \hat{x}_{k/N} - P_{k}^{-1} \hat{x}_{k/k-1} - H_{k}^{\prime} R_{k}^{-1} x_{k} \\ &= P_{k}^{-1} / k - 1 \left( \hat{x}_{k/N} - \hat{x}_{k/k-1} \right) - H_{k}^{\prime} R_{k}^{-1} \left( x_{k} - H_{k} \hat{x}_{k/N} \right) \\ &= \omega_{k-1} - H_{k}^{\prime} R_{k}^{-1} \left( x_{k} - H_{k} \hat{x}_{k/N} \right) . \end{aligned}$$

$$(2.40)$$

The fourth and last version is called fixed lag smoothing because the smoothed estimate is always a fixed number of points, s, behind the latest measurement. It can be written as shown below where the vector  $\omega_{k-1}$  can be defined either by (2.37) or by the recursive relation in (2.36):

$$\hat{x}_{k/N} = \Phi_{k} \hat{x}_{k-1/N-1} + B_{k/N} \hat{x}_{N} + Q_{k-1} \omega_{k-1},$$

$$k = N - s, \qquad (2.41)$$

This relation is obtained by replacing k by k = 1 and N by N - 1 in (2.36) and substituting the resulting expression for  $\hat{X}_{k/N-1}$  into the fixed point solution (2.18). A recursive relation for the fixed lag covariance is obtained by replacing k by k = 1 and N by N - 1 in (2.33) and substituting the resulting expression for  $P_{k/N-1}$  into the fixed point covariance (2.20).

$$P_{k/N} = P_{k/k-1} - B_{k/H} \mathbb{E} \{ \mathbb{E}_{N} \mathbb{E}_{k}^{\prime} \} B_{k/N}^{\prime} + C_{k}^{-1} (P_{k-1/N-1} - P_{k-1/k-1}) (C_{k}^{-1})^{\prime}, \qquad (2.42)$$

Thus far, it has been assumed the matrices  $Q_k$  and  $R_k$  were non-singular. If the matrix  $Q_k$  is singular it will not change the form of the solutions, although if  $Q_k$  is zero the smoothing solutions simplify to

$$\hat{\mathbf{x}}_{\mathbf{k}/\mathbf{N}} = \boldsymbol{\mathcal{P}}_{\mathbf{k}+1}^{-1} \hat{\mathbf{x}}_{\mathbf{k}+1/\mathbf{N}}$$

If the matrix  $R_k$  is singular the smoothing solution involving measurements cannot be used because it contains  $R_k^{-1}$ . For the filtering solution, it is necessary that

$$H_k P_{k/k-1} H'_k + R_k$$

be non-singular, so there may be a problem if both  $Q_{\mu}$  and  $R_{\mu}$  are singular.

#### 3. CONTINUOUS TIME SMOOTHING

3.1 Statement of the Problem

The system to be estimated is described by a set of matrix differential equations

The continuous measurements obtained from the system are defined by the equation

where

= an n×1 column vector of state variables

a an nx1 column vector of system (input) uncorrelated dynamic noise

= an rx1 column vector of measurement (output) uncorrelated noise

s = an rx1 column vector of continuous known measurements

F = the n x n dynamics matrix with known coefficients .

H = the rxn output matrix with known coefficients.

The vectors u and v represent vector-valued continuous independent random variables (white noise) with zero mean and known covariance.

 $E(u(t)u(s)') = Q\delta(t-s)$   $E(v(t)v(s)') = R\delta(t-s)$  E(u(t)v(s)') = 0.

.







(3.1)

(3.2)

(3.3)





14

The property of the delta function  $\delta(t)$  is that the function is zero for  $t \neq 0$  but the integral of g(t) times the function is g(0) if the integral includes the origin. The *a priori* information is that the initial value of the state,  $x_0$ , is a random variable with mean  $\tilde{x}_0$  and covariance  $\tilde{P}_0$ ,

$$\mathbf{E}(\mathbf{x}_0) = \overline{\mathbf{x}}_0 \quad \text{and} \quad \mathbf{E}[(\mathbf{x}_0 - \overline{\mathbf{x}}_0)(\mathbf{x}_0 - \overline{\mathbf{x}}_0)'] = \overline{\mathbf{P}}_0, \quad (3.4)$$

The smoothed estimate of the state at time t, given all the measurements up to time T, is denoted by  $\hat{X}_{t/T}$ . The filtered estimate is denoted by  $\hat{X}_{t/t}$  or simply  $\hat{x}$ . The covariances of the errors in the smoothed and filtered estimates are  $P_{t/T}$  and  $P_{t/t} = P$  respectively. The linear estimate is a linear combination of the a priori information and an integral of the measurements

$$\hat{x}_{t/T} = A_{t/0}\bar{x}_0 + \int_{k=0}^{k=T} A_{t/k} z \, ds \, . \tag{3.5}$$

The best estimate is the linear estimate which minimizes the integral of the mean square error,

Minimize 
$$\int_{t=0}^{t=T} \mathbb{E}[(x - \hat{x}_{t/T})(x - \hat{x}_{t/T})'] dt$$
. (3.6)

#### 3.3 Continuous System as Limiting Form of Discrete System

The discrete system can be considered as the sampled version of the continuous system while the continuous system can be considered as the limiting form of the discrete. These results will be used to derive the continuous best estimate as the limiting case of the discrete best estimate. First, consider the continuous system. The linear differential equation can be integrated from time t to time s to get

$$\mathbf{x}(\mathbf{s}) = \Phi(\mathbf{s}, \mathbf{t})\mathbf{x}(\mathbf{t}) + \mathbf{u}^{*}(\mathbf{s}, \mathbf{t}) ,$$
  
$$\mathbf{u}^{*}(\mathbf{s}, \mathbf{t}) = \int_{\sigma=\mathbf{t}}^{\sigma=\mathbf{s}} \Phi(\mathbf{s}, \sigma) \mathbf{u}(\sigma) \, d\sigma , \qquad (3.7)$$

and where  $\Phi(\mathbf{s}, \mathbf{t})$  is the transition matrix which satisfies the differential equations

Let  $s = t + \epsilon$ , where  $\epsilon$  is small, and there is an obvious analogy between the continuous system and the discrete system,  $x_{k+1} = \Phi_{k+1}x_k + u_k$ .

$$\begin{aligned} \mathbf{x}_{k+1} &= \mathbf{x}(t+\epsilon) \\ \Phi_{k+1} &= \Phi(t+\epsilon,t) \\ \mathbf{u}_{k} &= \mathbf{u}^{*}(t+\epsilon,t) \end{aligned} \tag{3.8}$$

Since the covariance of the continuous dynamic noise  $u(\sigma)$  is a delta function, the covariance of the random input  $u_k$  is calculated as follows:

$$\mathbf{\tilde{x}}(\mathbf{u}_{\mathbf{k}}\mathbf{u}_{\mathbf{k}}') = \mathbf{Q}_{\mathbf{k}} = \int_{\mathbf{t}}^{\mathbf{t}+\epsilon} \Phi(\mathbf{t}+\epsilon,\sigma)\mathbf{Q}(\sigma)\Phi(\mathbf{t}+\epsilon,\sigma)' \, d\sigma \simeq \mathbf{Q}(\mathbf{t})\epsilon \,, \qquad (3.9)$$

Thus, starting with a continuous system, one can take a series of samples and convert it to a discrete system. The linear equation for the measurements can also be integrated from time t to time  $t + \epsilon$  to get an "average" value for the measurement. Assume for the moment that H(t) and x(t) are constant over that interval.

\*\*\*\*

$$\mathbf{i}(t) = (1/\epsilon) \int_{t}^{t} \mathbf{x}(\sigma) \, d\sigma = \mathbf{H}(t)\mathbf{x}(t) + \mathbf{y}^{\mathbf{x}}(t+\epsilon, t)$$
$$\mathbf{y}^{\mathbf{x}}(t+\epsilon, t) = (1/\epsilon) \int_{t}^{t+\epsilon} \mathbf{y}(\sigma) \, d\sigma \, . \tag{3.10}$$

Actually, the integral of the change in H(t) and x(t) will be of order  $\epsilon$ , but that can be neglected for small  $\epsilon$ . The analogy between the continuous measurements and the discrete measurements is

 $z_{k} = \overline{z}(t)$   $v_{k} = v^{*}(t + \varepsilon, t) . \qquad (3.11)$ 

where

Ĩ

The covariance of the noise is

$$\mathbb{E}(\mathbf{v}_{\mathbf{k}}\mathbf{v}_{\mathbf{k}}') = \mathbf{R}_{\mathbf{k}} = (1/\epsilon^2) \int_{\mathbf{t}}^{\mathbf{t}+\epsilon} \mathbf{R}(\sigma) \, d\sigma \simeq \mathbf{R}(\mathbf{t})/\epsilon \,, \qquad (3.12)$$

Next, consider the discrete system. The limiting form of the system difference equation is obtained by making a Taylor series expansion of the transition matrix for small < and neglecting terms of order < denoted by 0((\*) ,

$$\begin{aligned} \mathbf{x}(\mathbf{t}+\boldsymbol{\epsilon}) &= [\mathbf{I}+\mathbf{F}\boldsymbol{\epsilon}+\mathbf{O}(\boldsymbol{\epsilon}^{-})] \mathbf{x}(\mathbf{t}) + \mathbf{u}(\mathbf{t}) \\ \lim_{\boldsymbol{\epsilon}=0} \left[\mathbf{x}(\mathbf{t}+\boldsymbol{\epsilon}) - \mathbf{x}(\mathbf{t})\right]/\boldsymbol{\epsilon} &= d\mathbf{x}(\mathbf{t})/d\mathbf{t} . \end{aligned} \tag{3.13}$$

To summarize, the limiting form of the discrete system is obtained for small  $\alpha$  ,

 $\lim_{k \to 0} [x_{k+1} - x_k] / \epsilon = dx/dt$ Q. - QE  $R_{\rm h} \rightarrow R/\epsilon$ H, → H X. - X (3, 14)

#### 3.3 Filtering and Smoothing Solutions

The montinuous smoothing solutions are presented here as the limiting cases of the discrete solutions derived earlier. Applying the limiting process to the filtering solution in (2.21) and (2.22) and the covariance in (2.24) gives the continuous filtering solution.

$$dx/dt = F\hat{x} + B\hat{z}_t$$

$$dP/dt = FP + PF' - BHP + Q$$
with  $\hat{z}_t = z - N\hat{x}$  and  $B = PH'R^{-1}$ .
(3.15)

where  $\hat{\mathbf{x}} = \hat{\mathbf{x}}_{t/t}$  and  $\mathbf{P} = \mathbf{P}_{t/t}$  and the initial conditions are

 $\hat{\mathbf{x}}_0 = \bar{\mathbf{x}}_0$  and  $\mathbf{P}_0 = \bar{\mathbf{P}}_0$ .

The number of terms in the continuous case is fewer than that in the discrete case because

$$\left[\Pi_{\mathbf{k}}\mathbf{P}_{\mathbf{k}/\mathbf{k}-\mathbf{i}}\mathbf{H}_{\mathbf{k}}+\mathbf{R}_{\mathbf{k}}\right]^{-1} \rightarrow \left[\mathbf{HPH}^{\prime}+(\mathbf{R}/\boldsymbol{\epsilon})\right]^{-1} \rightarrow \mathbf{R}^{-1}\boldsymbol{\epsilon} . \tag{3.16}$$

Sometimes in a discrete problem it might be convenient to integrate the discrete difference equations for the covariance on a digital computer by changing them to the corresponding differential equations. As the limiting process shows, one requirement for this approximation to be valid is that

$$R_k \gg H_k P_{k/k-1} H_k'$$

Applying the limiting process to the fixed point emoothing solution in (2.18), (2.19) and (2.27) and the corresponding covariance in (2.20) gives the continuous fixed point smoothing solution.

and

$$dS_{t/T}/dT = D(T,t)S_T$$

$$d^{D}_{t/T}/dT = -D(T,t)RD(T,t)',$$

$$D(T,t) = P_{t/t}^{-1}\psi(T,t)H'R^{-1} \quad \text{and}$$

$$\psi(T,t) = \text{the matrix which satisfies the differential equation}$$

 $d\Psi(T,t)/dT = (F-BH)\Psi(T,t)$ 

Applying the limiting process to the smoothing solution involving the filtered estimate in (2.31) and the corresponding covariances in (2.33) gives the smoothing solution for the continuous case,

where

W.

(3.17)

$$d\hat{x}_{t/T}/dt = F\hat{x}_{t/T} + QP^{*1}(\hat{x}_{t/T} - \hat{x}_{t/t})$$

$$dP_{t/T}/dt = (P+QP^{-1})P_{t/T} + P_{t/T}(P+QP^{-1})' - Q.$$
(3.18)

These differential equations must be integrated backwards from T to 0 and the beginning conditions at  $t \approx T$  are the terminal conditions on the filter equations  $\hat{x}_{T/T}$  and  $P_{T/T}$ . The limiting form of the matrix  $C_k$  has been written as

$$C_{\mathbf{k}} \rightarrow \mathbf{I} - \boldsymbol{\epsilon} \left( \mathbf{F} + \mathbf{Q} \mathbf{P}^{-1} \right) , \qquad (3.19)$$

Applying the limiting process to the filtering solution involving the measurements in (2.38) gives the smoothing solution for the continuous case

$$d\omega/dt = -F'\omega - H'R^{-1}(z - H\hat{x}, ...) . \qquad (3.20)$$

Once again the differential equations are integrated backwards from T to 0 and the beginning conditions are the terminal conditions on the filter eviations  $\hat{\pi}_{T/T}$  and  $\omega(T) = 0$ .

Finally, applying the limiting process to the fixed lag solution of a seconds gives

4. 4.4. ...

$$d\hat{x}_{t/T}/dT = F\hat{x}_{t/T} + D(T,t)\hat{z}_{T} + Q\omega$$

$$dP_{t/T}/dT = (F+QF^{-1})F_{t/T} + F_{t/T}(F+QF^{-1})' - Q - D(T,t)FD(T,t)',$$

$$t = T - B,$$
(3.21)

where

#### 4. APPLICATION - STOCHASTIC DRAG MODEL

#### 4.1 Satellite Tracking with Stochastic Drag

This section presents an example of smoothing on satellite tracking where a stochastic model is used to improve scouracy. The practical problems which arise in smoothing are essentially the sate an those in filtering. A model for the system equations must be developed as well as a model for the measurements. Values must be chosen for the mean and covariance of the dynamic noise and the measurement noise and for the a priori estimate and its covariance. The primary difference between practical problems in filtering and those in mmoothing is in the effect of including dynamic noise on the model. If there is no dynamic noise in the model the system is deterministic and the smoothing solution is obtained directly from the final value of the filtered solution. When there is dynamic noise, the smoothing solution cannot be obtained so easily and may require significant additional work. Correlated noise on the measurements can have a similar effect to dynamic noise. To handle correlated part of the state equations are sugmented by including additional state variables for the correlated part of the noise.

The problem of trajectory estimation involves reconstructing the position history of a space vehicle from a series of measurements such as range, angles, or Doppler velocity. When there is only gravity scoeleration, due to engine thrust or air drag, the motion is deterministic and can be completely specified by six parameters. These six might be the position and velocity at mome initial time, or a set of mix orbit elements at the time of the ascending node. From the mix parameters, the current and future motion can be determined by numerical integration of the acceleration due to the earth's gravitational field or from a closed-form approximation to the integrated motion.

In general, when drag is included, the assumption is usually mode that its effect can also be represented as a deterministic phenomenon (for instance, as a polynomial with undetermined coefficients). This treatment may be adequate as long as drag is not a significant factor in the satellite motion. However, for low situade satellites, stochastic fluctuations in atmospheric density can cause noticeable changes in the motion.

The particular stochastic model presented here is based on the assumption that the statistical properties of the atmosphere can be adequately represented by a first-order stationary Gauss-Markov process. Therefore, the satellite motion can be determined by eight parameters that are a set of six mean orbit elements, a seventh parameter that is related to the constant average value of the drag, and an eighth parameter that is related to the instantaneous deviation of the stochastic drag from its mean value. A similar model could be developed when there is acceleration due to engine thrust. The stochastic fluctuations in thrust might be due to changes in the direction of thrust or the engine performance.

#### 4.3 Simplified Model for Analyzia

The main effect of the drag socelaration is to change the in-track motion of the satellite. The satellite arrives at some position in space a little sooner or later, depending on whether the drag has been a little larger or smaller than expected. The simplified discrete model of the motion will involve four state variables — the in-track angular position  $y_1$ , the in-track angular velocity  $y_2$ , the mean drag acceleration  $y_3$ , and the stochastic drag acceleration  $y_4$ .

S. C. S. S. S. S.

The continuous stochastic acceleration b(t) is assumed to be a zero-mean first-order Markov process with exponential correlation and a time constant 1/c . The differential equation describing the stochastic acceleration has stationary zero-mean white noise u(t) as the excitation,

$$db(t)/dt = -cb(t) + u(t)$$

$$\mathbb{E}[u(t)u(t+s)] = q\delta(s) ,$$

The discrete version of the stochastic acceleration is given by a difference equation,

$$b(t+T) = b(t) \exp(-\alpha T) + u^{*}$$

$$u^{*} = u^{*}(t+T,t) = \int_{s=t}^{s=t+T} \exp[-\alpha(T-s)] u(s) ds$$

$$E[u^{*}u^{*}] = \int_{s=0}^{s=T} \exp[-2\alpha(T-s)] q ds$$

$$= q[1 - \exp(-2\alpha T)]/20 , \qquad (4.2)$$

The dimersts stochastic modeleration  $y_{ij}$  changes every T seconds when it takes on the value of the continuous variable b(t). The system of differential equations for the angular motion are

$$dy_{1}/dt = y_{2} + y_{4}$$

$$dy_{2}/dt = y_{3} + y_{4}$$

$$dy_{3}/dt = 0$$

$$y_{4} = b(kT) \text{ for } kT < t < (k+1)T. \qquad (4.3)$$

Every T seconds noisy measurements of position are made. The difference equations for the discrete system 874

$$\begin{aligned} \mathbf{x}_{k+1} &= \Phi_{k+1} \mathbf{x}_{k} + \mathbf{u}_{k} \\ \mathbf{x}_{k} &= H_{k} \mathbf{x}_{k} + \mathbf{v}_{k} \\ \mathbf{E}(\mathbf{v}_{k}) &= 0 , \quad \mathbf{E}(\mathbf{v}_{k} \mathbf{v}_{k}') = \mathbf{R}_{k} \\ \mathbf{x}_{k}' &= (\mathbf{y}_{1}, \mathbf{y}_{2}, \mathbf{y}_{3}, \mathbf{y}_{4}) \\ \mathbf{u}_{k}' &= (0, 0, 0, \mathbf{u}^{\pm}) \\ \mathbf{u}_{k}' &= (0, 0, 0, \mathbf{u}^{\pm}) \\ \mathbf{v}_{k+1} &= \begin{bmatrix} \mathbf{1} & \mathbf{T} & \mathbf{T}^{2}/2 & \mathbf{T}^{2}/2 \\ \mathbf{0} & \mathbf{1} & \mathbf{T} & \mathbf{T} \\ \mathbf{0} & \mathbf{0} & \mathbf{1} & \mathbf{0} \\ \mathbf{0} & \mathbf{0} & \mathbf{e} \mathbf{x} \mathbf{p} (-\mathbf{0}\mathbf{T}) \end{bmatrix} \end{aligned}$$

 $H'_{L} = (1,0,0,0)$ .

The covariance of the *a priori* information is diagonal:

 $\mathbf{II}(\mathbf{x}_{0}\mathbf{x}_{0}') = \begin{cases} \mathbf{p} & \mathbf{0} & \mathbf{0} & \mathbf{0} \\ \mathbf{0} & \mathbf{p} & \mathbf{0} & \mathbf{0} \\ \mathbf{0} & \mathbf{0} & \mathbf{p} & \mathbf{0} \end{cases}$ (4.5) 0 0

For the numerical examples the sample time T will be held fixed at one revolution (T = 1), the correlation time 1/o will be two revolutions (c = 1/2) and the accuracy of the measurements will be set equal to unity. Three cases will be examined. The novariances of the stochastic drag q and the a priori information p for the three cases are

Case	q	p
1	10-2	1
2	10-"	1
3	10-2	100

(4.4)

(4.1)

STATE AN





In each case 25 measurements are taken, starting with  $\mathbf{z}_1$ . The diagonal elements of the covariance of the estimates of the state for case 1 are presented in Table I for both filtering and smoothing. Notice how smoothing decreases the error substantially. In Figure 1 the variances of the filtered and smoothed estimates for the angular position  $\mathbf{y}_1$  are plotted for both case 1 and case 2. In both cases the variance approaches a steady-state value where the information lost by dynamic noise balances the information guined from additional measurements. Reducing the variance of the stochastic drag reduces the steady-state variance of the estimates. In Figure 2, the variances for the angular position are plotted for both case 1 and case 3. Notice how the 'uitial conditions (the a priori information) have a significant effect on the first few estimates, although that rapidly dies out.

#### 4.3 Numerical Results with Real Data

The stochnatic drag model has been twated by using data taken during the first three days of satellite 1000  $Omioron^{16}$ . The satellite was hunched on November 12, 1060, with a period of 96.4 min, and was up for only 47 days. The apogee height, perigee height, and inclination were 614 statute miles, 113 statute miles, and 81.9°, respectively. Over the first three days the average constant drag was 24 x 10<sup>-4</sup> fractional decay of the semi-major axis per revolution, which corresponds to a period decay of about 0.2 sec/rev. This satellite was chosen because fluctuations in atmospheric density over the first two days caused a drag change of over 50% from the first day to the second, and because there were a large number of highly accurate Baker-Nunn optical sightings of the satellite. The total data consist of 13 parses over radar stations and six sets of optical sightings, spaced so that there is a gap of about 9 hours between the first and second day, when there were no measurements. (There was also a seventh set of optical sightings, which was discarded because it was completely inconsistent, by scores of miles, with all the rest of the data.)

Results were obtained using both optical and radar data but, in the result4 presented here, the radar data were used to determine the orbit, whereas the optical data were used as a check. This approach was particularly appropriate because the optical data were highly accurate (perhaps to less than one hundred feet) and also because the radar data were concentrated mainly in the northern hemisphere, whereas the optical data were mainly in the southern hemisphere. With a given set of data, the quantities that can be varied are the initial conditions (the initial estimate and its covariance), the r.m.s. drag of the stochastic model  $\langle \mathcal{O}_0 \rangle$ . It was desired to aliminate the effect of the initial conditions, so they were chosen with a rather large covariance and with an estimate which matched the data from the first few radar stations. For all the results presented here the initial conditions were held fixed, while only the r.m.s. drag and correlation of the model were observed here the initial conditions were held fixed, while only the r.m.s.

Since the most important effect of drag is to change the time equation, it is the in-track residuals that should be examined. Because the radar range measurements are so much more accurate than the angle measurements. the instruck residuals in position are determined by comparing the instruck position at the time when the range reaches a manimum (the point of closest approach). The weighted rams, in-track residuals for radar data alone are plotted in Figure 3 as a function of r.m.a. drag, with the correlation held fixed at two revolutions (about 3 hr). For very low r.m.s. drag, the stochastic model reduces to the deterministic model and the weighted r.m.s. residuals approach the value obtained using the deterministic model (6800 ft). Notice that the r.m.s. residuals for the stochastic model reach a minimum at about one-tenth of the value obtained using the deterministic model. For very high r.m.s. drag, the forward estimation procedure tended to be marginally stable. The high value of stochastic drag was used to overcorrect the estimate, first one way then the other. The r.m.s. in-track residuals for radar data alone are plotted again in Figure 4 as a function of the correlation, with the r.m.a. drag held fixed at  $4 \times 10^{-7}$ . The results show that the r.m.s. residuals were not very sensitive to the correlation. Once again, as the correlation gets very large, the stochastic model will approach the deterministic model. The estimate of the stochustic drag is plotted in Figure 5 as a function of revolutions for two values of the correlution with  $\sigma$  held fixed at  $4 \times 10^{-7}$ . Notice how the larger value of the correlation tends to mask the rapid fluctuations in the drag.

A point-by-point comparison of the in-track residuals at each rader pass is presented in Table II for the deterministic model with r.m.s. drag zero and the stochastic  $m_{\rm c}$  to l with the r.m.s. drag  $4 \times 10^{-7}$  and the correlation 10 rev. Comparing the two sets of results shows that the stochastic treatment of drag has reduced the residuals from the deterministic treatment using constant drag by an order of magnitude, from thousands of feet to hundreds of feet. In order to check the rader results, the optical data were included with zero weight, i.e., the residuals of the optical sightings were calculated, but no differential corrections were made. The residuals at three of the optical stations were less than 200 ft and at the fourth station, following the gap of 9 hr with no measurements, the residuals were 1500 ft. The prediction of the optical measurements after 3 rev and 6 rev following the end of the radar data gave residuals of 28,000 and 70,000 ft, respectively, indicating that there still might be some stochastic fluctuations on the third day. Also a capsule was separated from the satellite on the 31 orbit, and this may have influenced the drag characteristics.

### REFERENCES

1. Gauss, K.F.	Theory of the Motion of Heavenly Bodies Moving About the Sun in Conic Sections. Translated by C.H.Davis, Dover Publications, New York, 1963,	
2. Wiener, N.	The Extrapolation, Interpolation, and Smoothing of Stationary Time Series with Engineering Applications. Wiley, New York, 1949.	
3. Kalman, R.E.	A New Approach to Linear Filtering and Prediction Problems. Transactions, American Society of Machanical Engineers, Journal of Basic Engineering, Vol.82, March 1980, pp.34-38.	
4. Kalman, R.E.	New Methods in Wiener Filtering Theory. Proceedings, First Symposium on Engineering Applications of Random Function Theory and Probability. J.L.Bogdanoff and F.Koziu (eds), Wiley, New York, 1903.	
5. Bucy, R.S.	Optimum Finite-Time Filters for a Special Non-Stationary Class of Inputs. Johns Hopkins University, Applied Physics Emboratory, Baltimore, Md., Internal Nemo BHD-600, 1989.	
6. Kalman, R.E., Buoy, R.S.	New Results in Linear Filtering and Prediction Theory. Transactions, American Society of Mechanical Engineers, Journal of Basic Engineering, Series D, Vol.83, December 1961, pp.95-107.	
7. Bryson, A.E., Frazier, N.	Smoothing for Linear and Nunlinear Dynamic Systems. Aeronautical Systems Division, Wright-Patterson AFB, Ohio, Tech. Report ASD-TDR-63-119, February 1963.	
S. Cox, H.	On the Estimation of State Variables and Parameters for Noisy Dynamic Systems. Institute of Electrical and Electronic Engineers, Transactions on Automatic Control, Vol.AC-9, January 1964, pp.5-12.	
9. Rauch, H.E., et al.	Maximum Likélihood Betimates of Linear Dynamic Systems. AIAA Journal, Vol.3, August 1985, pp.1443-1450.	
10. Lee, R.C.K.	Optimal Estimation, Identification and Control. MIT Press, Cambridge, Mass., 1964.	
11. Meditch, J.M.	Orthogonal Projection and Discrete Optimal Linear Smoothing. Journal, Society for Industrial and Applied Mathematics, Vol.5, February 1967, pp.74-89.	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1
12. Framer, D.C.	A New Technique for the Optimal Smoothing of Data. BcD Dissertation, Department of Aeronautical Engineering, Massachusetts Institute of Technology, Cambridge, Mass., January 1967.	
13. Kailath, T.	An Innovations Approach to Least Squares Fatimation - Part I: Linear Estimation in Additive White Noise. Institute of Electrical and Electronic Engineers, Trans- motions on Automatic Control, Vol.AC-13, December 1968, pp.646-655.	
14. Kailath, T. Frost, P.	An Innovations Approach to Least Squares Estimation - Part II: Linear Smoothing in Additive White Noise. Institute of Electrical and Electronic Engineers, Transactions on Automatic Control, Vol.AC-13, December 1968, pp.655-660.	
15. Rauch, H.E.	Optimum Estimation of Satellite Trajectories Including Random Fluctuations in Drag. AIAA Journal, Vol.3, April 1965, pp.717-722.	

MANA

and a state of the state

State Laurent

TABLE I

Disgonal Elements of the Covariance

<b>Observation</b>	Filtered Estimate (P <sub>k/k</sub> )					
Point (k)	cov(y <sub>i</sub> )	ocv (y <sub>2</sub> )	cov(y;)	COY (Y, )		
0	1.00	1,00	1.00	0,0100		
1	0.69	1,31	0.92	0.0100		
2	0.80	1.31	0.54	0.0100		
3	0, 82	0.06	0.26	0.0100		
4	0.79	0, 68	0.13	0,0100		
5	0.75 .	0, 40	0.07	0,0100		
10	0.58	0.15	0,000	0,00995		
15	0.50	0,10	0.004	0,6000		
20	0.48	0.093	0,0028	0,0099		
26	0.47	0.089	0, 6020	0.0099		
Observation	 	Snoothed Est	imate (P <sub>k/N</sub> )			
Point (k)	cov(y1)	00Y (y2)	00V(y)	cov(y <sub>4</sub> )		
28	0.47	0.069	0.0020	0,0099		
-24	0.26	0,058	0.0020	0.0096		
23	0,18	0.036	0.0020	0.0091		
22	0.15	0.023	0.0020	0.0085		
21	0.18	0.017	0.0020	0.0080		
20	• 0.15	0.015	0.0020	0.0078		
15	0.135	0.014	0.0020	0.0078		
10	0.105	0.014	0/0020	0.0078		
5	0.14	0.015	0.0020	0.0078		
1	0.26	0.033	0.0020	0.0089		

#### TABLE II

0.082

0,0020

0,0094

0,45

	Data®	Revolution <sup>b</sup>	Deterministic <sup>o</sup>	Stochastic
1	R	0.4	44.6	2.7
2	0	4.0		0.1
8	R	5.1	5.7.	· 0.9
4	R	6.1	0.3	0.3
5	R	7.1	-1.6	0.5
6	R	8.1	-8.0	0.4
7	R	11.4	-4.4	0.2
	R	18.4	2.1	0.4
9	0	19.0		1.5
10	R	19.1	2.6	0.7
11	0	20.1		0.2
12	R	20.1	1.8	0.4
13	0	21.1		0.1
14	R	21.1	0,6	-0.1
15	R	22.1	2.6	2.1
16	R	28.1	0,6	0.3
17	R	28.4	4.0	0.1
18	0	31.1		28.0
10	•	94 1	1	70.0

### Comparisons of In-Track Mesiduals, 10<sup>3</sup> ft

<sup>B</sup>Optical = 0 and radar = R. <sup>b</sup>Since time of first ascending node.

12 M.A.

.

ويتحدد والمراجعة والمحاصية والمراجع والمراجع والمراجع والمحاج وا

0

<sup>c</sup>Radar data with constant drag model. <sup>d</sup>Radar data with zero weight on optical (for  $\sigma = 0.4 \times 10^{-6}$  $\theta_{c} = 10$  rev).

والمراجع والمحاجبة والمتحدية والمتحدية والمحاجب والمحاجب والمحاجب والمحاجب والمحاج والمحاج والمحاج والمحاج والم

179

KAR DE LE MAR 11.11 1998 **- 199** 





Fig.1 Variance history for two levels of stochastic disturbances (q)



Fig. 2 Variance history for two sets of initial conditions (p)



Fig.3 Root mean square in-track residuals versus root mean square stochastic drag









180

<u>N</u>



المراجع والمحمد والمراجع والمحمد والمح

υ.

155

•

## CHAPTER 9 - NONLINEAR SMOOTHING TECHNIQUES

TAX

The Manual Di

19474-1-1 44444

# hy John B.Peller

#### 1260 Arbolita Drive Les Habra, California 90631, USA

. . . .

# **PRECEDING PAGE BLANK**

#### CHAPTER 9 - NONLINEAR SMOOTHING TECHNIQUES

John B.Peller

#### 1. INTRODUCTION

すいことでは、このでは、このでは、このでは、

The so-called "nonlinear smoothing" problem is the problem of estimating the state of some noisy process at some time t, given noisy measurements related to the process over an entire measurement interval which includes the time t. The monlinear smoothing problem differs from the linear smoothing problem in that the differential equations describing the process and the observations can be nonlinear as well as linear. A typical nonlinear monothing problem is the post-flight estimation of the flight path of a missile based on tracking system measurements made during the entire duration of the flight. If the estimate of the missile's position and velocity at the midpoint of the flight is desired, the estimate can be based on all of the measurements made, including these made after the midpoint as well as these made before. This is the amothing problem. If either the differential equations describing the flight of the missile or the measurements made to determine the actual flight path (or both) are nonlinear, the problem of estimating the missile's position and velocity is called a nonlinear smoothing problem. In this chapter, methods of solving the onlinear smoothing problem will be presented.

Smoothing problems (or any estimation problems for that matter) may be classified according to the oriterion by which the quality of the estimate is to be judged. In this chapter, as in the rest of this publication, only probabalistic oriteria will be considered. For the smoothing problem, this is a marked departure from practice; normally, smoothing is performed by statistical techniques such as fitting a polynomial to a set of observations in a least squares fashion without considering the known properties of the noire in the measurements or in the plant.

In order to describe the classifications, some definitions are necessary. Let  $\underline{X}(t)$  be a vector function of time representing the state of the process or plant. Let  $\underline{y}(\underline{X}, t)$  be another vector function of both the state  $\underline{X}$  and time t; it will be called the observation process. Suppose the process runs from time t = 0 to the time t = T, and that the observation  $\underline{y}(\underline{X}, t)$  is recorded for all t in the interval from 0 to T. The smoothing problem is to estimate  $\underline{X}(t)$  for some t in the interval by operating on all the observations in the interval, Denote the estimate by  $\underline{X}(t)$ .

If  $\hat{g}(t)$  is selected so that the probability  $\hat{g}(t) = \chi(t)$  is maximized, the estimate  $\hat{g}(t)$  is called the maximum likelihood estimate of  $\chi(t)$ . If  $\hat{g}(t)$  is selected so that the covariance of the estimate  $\mathbb{E}[\{\chi(t) - \chi(t)\}[\chi(t) - \chi(t)]^T\}$  (where  $\mathbb{E}(.)$  is the expectation operator) is minimized, the estimate is called the minimum variance estimate. Minimum variance estimates are particularly useful in missile accuracy analysis. For a linear plant with additive gaussian white noise and linear observations with additive gaussian white noise, the state of the plant  $\chi(t)$  is a random process with a gaussian distribution, and any estimate of  $\chi(t)$  derived by a linear operation on the observations will also be a gaussian random variable. For this case, both the maximum likelihood estimate and the minimum variance estimate are equal to the max of a particular gaussian distribution derived by such a linear operation. Consequently, for any kind of linear estimation theory, it is unnecessary to differentiate between the maximum likelihood estimates.

For a nonlinear plant and nonlinear obsorvations, the probability density function of the estimate conditioned on all the observations is not generally gaussian. In this case, the maximum likelihood estimate (which can be shown to be equal to that value of  $\chi(t)$  for which the subject density function has its maximum value) will differ from the minimum variance estimate (which can be shown to be equal to the mean value or first moment of the subject density function). Thus, the procedure used to solve the nonlinear smoothing problem will differ depending upon which type of estimate is desired. Primary emphasis in this chapter will be placed on the problem of finding the minimum variance estimate, because this estimate is more useful in fields such as missile accuracy analysis and other probabilistic applications of the smoothing results. However, a particular approach for use in obtaining maximum likelihood estimates will also be discussed.

Minimum variance smoothing theory will be covered in Sactions 2 and 3 of this chapter. In Section 2, "exact" differential equations for the "smoothed expectation" of an arbitary function of the state are derived. In Section 3, these differential equations are approximated by other differential equations which are easily solved on a computer to obtain the minimum variance estimate. Section 4 presents a brief discussion of maximum likelihood nonlinear smoothing.

#### 3. EXACT DIFFERENTIAL EQUATION FOR THE SMOOTHED EXPECTATION

In Section 2.1, the problem is formulated, and initial steps are taken towards deriving the dusined exact differential equations. In particular, a differential form is derived for the probability density function of . **PRECEDING PAGE BLANK** 





the state conditioned on all the observations. In subsequent subsections, adjoint theory and various expansions are employed to convert this to an exact differential equation for the expectation of an arbitrary function of the state, conditioned on all the observations.

#### 3.3 Problem Statement and the Smoothing Density Function

Let the system by represented by the n-dimensional state vector:  $\underline{x}(t)$ , and let the observations  $\underline{y}(\underline{x},t)$  be represented by the m-dimensional vector  $\underline{x}(t)$ . Let these vectors satisfy the following equations:

$$i(t) = f(x, t) + j(t)$$
 (2.1)

and

$$(t) = g(x, t) - \eta(t)$$
, (2.2)

where  $\dot{f}(t)$  and  $\eta(t)$  are formally gaussian white noise processes. Equations (2.1) and (2.2) are considered to be formulisan of the following equations:

2

$$lx(t) = f(x, t) dt + d\zeta(t) + r$$
, (2.3)

$$dz(t) = g(x, t) dt + d\eta(t) + r$$
, (2.4)

where r denotes any terms for which the expectations satisfy

$$E(r) = O(dt)$$
, (2.5)

$$E(r^2) = O(dt^2)$$
, (2.6)

where  $O(.)/dt \rightarrow 0$  as dt  $\rightarrow 0$ . The function  $f(\underline{x}, t)$  is assumed to be piecewise continuous in t and to possess piecewise continuous first derivatives in  $\underline{x}$ . No explicit assumptions concerning  $\underline{g}(\underline{x}, t)$  are made herein. However, assumptions will be mode concerning the process  $\underline{z}(t)$  and the availability of a filtoring density function, the realization of which may constrain  $\underline{g}(\underline{x}, t)$  to have certain properties, in particular, to be piecewise continuous in both  $\underline{x}$  and t. Both functions,  $\underline{f}(\underline{x}, t)$  such  $\underline{g}(\underline{x}, t)$ , may be nonlinear.

The independent variable t is assumed to be constrained to some set, closed at least on the right, denoted by  $(t_0, T)$ . It is possible for  $t_1$  to be minus infinity, but T is finite. Let  $Z_{t_1, t_2}$  denote the set of random variables g(t) for  $t_1 \leq t \leq t_2$ :

$$Z_{t_{1},t_{2}} = \left\{ \underline{z}(t) \ni t_{1} \leq t \leq t_{2} \right\}.$$

$$(2.7)$$

For notational ease,  $Z_{t_0,\tau}$  will simply be denoted by  $Z_{\tau}$ . Assume that the conditional distribution of  $\underline{I}(t)$ , given  $Z_{\tau}$ , exists and is absolutely continuous. This implies that the conditional density of  $\underline{I}(t)$ , given  $Z_{\tau}$ , exists. This density is denoted by  $p_{\underline{I}(t)}[\mu]Z_{\overline{I}}]$ , where  $\mu$  is an n-dimensional argument. Because of the extensive use of this particular density, it is denoted by

$$\mathbf{s}(\boldsymbol{\mu}, \mathbf{t}) = \mathbf{p}_{\mathbf{x}(\mathbf{t})} [\boldsymbol{\mu} \mathbf{Z}_{\mathbf{x}}]$$
(2.8)

and is referred to as the smoothing density function. The numerous occurrences of the density function  $y_{g(t)}[\mu|Z_t]$  make it desirable to give it a special symbol and name. It will therefore be referred to as the filtering density function and denoted as

$$q(\mu,t) = p_{\pi(t)}(\mu Z_t)$$
 (2.9)

As the first step in obtaining a differential form for the amouthing density function  $s(\mu, t)$ , the smoothing density function is written in the form

$$\mathbf{P}_{\mathbf{g}(\mathbf{t})}(\underline{\mu}|\mathbf{Z}_{\mathbf{T}}) = \int_{-\infty}^{\infty} \mathbf{P}_{\mathbf{g}(\mathbf{t})}(\underline{\mu}|\underline{\mathbf{x}}(\mathbf{t}+\Delta \mathbf{t}) = \underline{\nu}, \mathbf{Z}_{\mathbf{T}}) \mathbf{P}_{\mathbf{g}(\mathbf{t}+\Delta \mathbf{t})}(\underline{\nu}|\mathbf{Z}_{\mathbf{T}}) d\nu, \qquad (2.10)$$

where the integration is over all n-components of  $\underline{\nu}$  (this convention will be used throughout).

At this point, the one and only assumption concerning the observation process is made. The observation process is assumed to be separable<sup>1</sup>. Separability implies that bounds taken over dense sets of points in  $(t_0, T)$  are the same as those obtained over the continuous parameter space, and that limits which are valid when approached sequentially are still valid when approached continuously and vice-verse. There is no effective loss of generality to assume that some process is separable, because if some process  $\underline{\chi}(t)$  is not separable, there exists a process  $\underline{\chi}(t)$  which is separable and which equals  $\underline{\chi}(t)$  with probability one:

 $P(z(t) = \bar{z}(t)) = 1$ , (2.11)

With the separability hypothesis, the process  $\underline{z}(t)$  is replaced with a discrete process  $\underline{z}(t)$  which is equal to  $\underline{z}(t)$  for t belonging to the set of rationals and equal to any constant for t belonging to the set of irrationals. Thus, only when t is a rational number does  $\underline{z}(t)$  convey may information concerning  $\underline{x}(t)$ 

**9** . .

Because the rationals are countable,  $\tilde{\underline{z}}(t)$  can be replaced with  $\underline{z}(i)$  and, momentarily replacing t with i , Equation (2.10) becomes

$$p_{g(i)}[\underline{\mu}|Z_{T}] = \int_{-\infty}^{\infty} p_{g(i)}[\underline{\mu}|\underline{x}(i+1)] = \underline{\nu} , Z_{T}]p_{g(i+1)}[\underline{\nu}|Z_{T}] d\nu . \qquad (2.12)$$

The first term in the integral can be written as

$$\mathbf{p}_{\underline{g}(i)}[\underline{\mu}|\underline{x}(i+1) = \underline{\nu}, Z_{T}] = \mathbf{p}_{\underline{x}(i)}[\underline{\mu}|\underline{x}(i+1) = \underline{\nu}, Z_{i}, Z_{i+1,T}]. \qquad (2, 13)$$

Because of the gaussian white noise assumption, Equations (2.3) and (2.4) represent Markov processes. Using the Markov property that the past and future are independent whon conditioned on the present,

$$p_{\underline{x}(1)}[\underline{\mu}|\underline{x}(1+1) = \underline{\nu}, Z_1, Z_{1+1, \tau}] = p_{\underline{x}(1)}[\underline{\mu}|\underline{x}(1+1) = \underline{\nu}, Z_1].$$
(2.14)

Substituting Equations (2.13) and (2.14) in (2.12),

$$p_{\underline{i}(1)}[\underline{\mu}|Z_{\underline{\tau}}] = \int_{-\infty}^{\infty} p_{\underline{i}(1)}[\underline{\mu}|\underline{x}(1+1)] = \underline{\nu}, Z_{\underline{i}}]p_{\underline{i}(1+1)}[\underline{\nu}|Z_{\underline{\tau}}] d\nu. \qquad (2.15)$$

By Bayes' rule for conditional densities,

$$P_{\underline{I}(\underline{i})}[\underline{\mu}]_{\underline{I}(\underline{i}+\underline{i})} = \underline{\nu}, Z_{\underline{i}}] = \frac{P_{\underline{I}(\underline{i})}[\underline{\nu}]_{\underline{I}(\underline{i}+\underline{i})} = \underline{\mu}, Z_{\underline{i}}]P_{\underline{I}(\underline{i})}[\underline{\mu}]Z_{\underline{i}}]}{P_{\underline{I}(\underline{i}+\underline{i})}[\underline{\nu}]Z_{\underline{i}}]}, \quad (2.16)$$

The denominator of Equation (2.16) can be evaluated by

$$p_{\underline{\mathbf{z}}(1+1)}[\underline{\nu}] \mathbf{z}_{1} = \int_{-\infty}^{\infty} p_{\underline{\mathbf{z}}(1+1)}[\underline{\nu}] \underline{\mathbf{x}}(1) = \underline{\mu}, \ \mathbf{z}_{1}] p_{\underline{\mathbf{z}}(1)}[\underline{\mu}] \mathbf{z}_{1}] d\mu.$$
(2.17)

By the Markov property,

$$p_{\underline{x}(i+1)}[\underline{\nu}|\underline{x}(i) = \underline{\mu}, \underline{z}_{\underline{i}}] = p_{\underline{x}(i+1)}[\underline{\nu}|\underline{x}(i) = \underline{\mu}], \qquad (2,18)$$

Using Equation (2.18), Equations (2.16) and (2.17) can be rewritten, respectively:

$$P_{\underline{x}(i)}[\underline{\mu}|\underline{x}(i+1) = \underline{\nu}, Z_{\underline{i}}] = \frac{P_{\underline{x}(1+1)}[\underline{\nu}|\underline{x}(i) = \underline{\mu}]P_{\underline{x}(i)}[\underline{\mu}|Z_{\underline{i}}]}{P_{\underline{x}(1+1)}[\underline{\nu}|Z_{\underline{i}}]}$$
(2.19)

and

SALAN - KOAL

*.* , ,

1

$$p_{\frac{1}{2}(\frac{1}{2}+1)}[\underline{\nu}|Z_{1}] = \int_{-\infty}^{\infty} p_{\frac{1}{2}(\frac{1}{2}+1)}[\underline{\nu}|X(1) = \underline{\mu}]p_{\frac{1}{2}(\frac{1}{2})}[\underline{\mu}|Z_{1}] d\mu . \qquad (2,20)$$

Thus, Equation (2.15) becomes

$$P_{\underline{I}(1)}[\underline{\mu}|Z_{\underline{T}}] = \frac{\int_{-\infty}^{\infty} P_{\underline{I}(1+1)}[\underline{\nu}|\underline{X}(1) = \underline{\mu}] P_{\underline{I}(1)}[\underline{\mu}|Z_{\underline{I}}] P_{\underline{I}(1+1)}[\underline{\nu}|Z_{\underline{T}}] d\nu}{P_{\underline{I}(1+1)}[\underline{\nu}|Z_{\underline{I}}]}, \qquad (2.21)$$

By the separability hypothesis, t and t + dt can be substituted for i and i + 1, respectively. Doing this and using Equations (2.8) and (2.9) in Equation (2.21),

$$\mathbf{s}(\underline{\mu},t) = \mathbf{q}(\underline{\mu},t) \int_{-\infty}^{\infty} \frac{\mathbf{p}_{\underline{\mathbf{g}}(t+dt)}[\underline{\nu}|\underline{\mathbf{x}}(t) = \underline{\mu}]}{\mathbf{p}_{\underline{\mathbf{g}}(t+dt)}[\underline{\nu}|\underline{\mathbf{x}}_{t}]} = (\underline{\nu}, t+dt) d\nu \qquad (2.22)$$

or

$$\mathbf{s}(\underline{\mu}, \mathbf{t}) = \mathbf{q}(\underline{\mu}, \mathbf{t}) \frac{\int_{-\infty}^{\infty} \mathbf{P}_{\underline{\mathbf{g}}(\mathbf{t}+\mathbf{dt})}[\underline{\nu}]\underline{\mathbf{x}}(\mathbf{t}) = \underline{\mu}]\mathbf{s}(\underline{\nu}, \mathbf{t}+\mathbf{dt}) d\nu}{\int_{-\infty}^{\infty} \mathbf{P}_{\underline{\mathbf{g}}(\mathbf{t}+\mathbf{dt})}[\underline{\nu}]\underline{\mathbf{x}}(\mathbf{t}) = \underline{\mu}]\mathbf{q}(\underline{\mu}, \mathbf{t}) d\mu}$$
(2.23)

Note that Equation (2.23) implies that the amounting density function is not explicitly dependent on the observation process in any manner. Of course, the filter density function  $q(\mu, t)$  would generally be expressed as a function of the observation vector and its statistics, so that the smoothing density function is implicitly a function of the observation process through the filter density function.

1 +144. 9.4
#### 2.2 Differential Equation for the Smoothed Expectation

In this section, preliminary steps are taken to obtain a differential equation for the "smoothed expectation" of an arbitrary function of the state. First, some definitions will be given. The smoothed expectation of an arbitrary function of the state,  $h(\underline{x})$ , is defined as:

$$\mathbf{E}_{\mathbf{a}}\mathbf{h}\left[\underline{\mathbf{x}}(\mathbf{t})\right] = \int_{-\infty}^{\infty} \mathbf{h}(\underline{\mu}) \mathbf{s}(\underline{\mu}, \mathbf{t}) \, \mathrm{d}\mu \,. \tag{2.24}$$

For later use, the conditional filtered expectation of an arbitrary function h(x) is defined as

$$\mathbf{E}_{\mathbf{f}}\mathbf{h}\left[\mathbf{x}(\mathbf{t})\right] = \int_{-\infty}^{\infty} \mathbf{h}(\underline{\mu})\mathbf{q}(\underline{\mu},\mathbf{t}) \, d\boldsymbol{\mu} \,. \tag{2.25}$$

Defining

$$dE_{gh}[\underline{x}(t)] = E_{gh}[\underline{x}(t)] - E_{gh}[\underline{x}(t-dt)]$$
(2.26)

and formally dividing by dt ,

$$\frac{dE_{gh}h[\underline{x}(t)]}{dt} = \frac{1}{dt} \left[ E_{gh}h[\underline{x}(t)] = E_{gh}[\underline{x}(t-dt)] \right]. \qquad (2.27)$$

The objective of this section is to obtain an expression of the term on the right side of Equation (2.27).

By definition (Equation (2.24)),

$$\mathbf{E}_{\mathbf{g}}\mathbf{h}\left[\underline{\mathbf{x}}(\mathbf{t}-\mathbf{d}\mathbf{t})\right] = \int_{-\infty}^{\infty} \mathbf{h}\left(\underline{\mu}\right)\mathbf{p}_{\mathbf{g}\left(\mathbf{t}-\mathbf{d}\mathbf{t}\right)}\left[\underline{\mu}^{2}\mathbf{Z}_{\mathbf{T}}\right] d\mu \qquad (2,28)$$

Substituting Equation (3.22) in Equation (2.28),

- -

---

$$\mathbf{E}_{\mathbf{g}}\mathbf{h}\left[\underline{\mathbf{x}}(t-dt)\right] = \int_{-\infty}^{\infty} d\mu h(\underline{\mu}) \int_{-\infty}^{\infty} \mathbf{p}_{\underline{\mathbf{x}}(t-dt)}\left[\underline{\mu}|\mathbf{Z}_{t-dt}\right] \frac{\mathbf{p}_{\underline{\mathbf{x}}(t)}\left[\underline{\nu}|\underline{\mathbf{x}}(t-dt) = \underline{\mu}\right]}{\mathbf{p}_{\underline{\mathbf{x}}(t)}\left[\underline{\nu}|\overline{\mathbf{x}}_{t-dt}\right]} \mathbf{p}_{\underline{\mathbf{x}}(t)}\left[\underline{\nu}|\mathbf{Z}_{\underline{\mathbf{x}}}\right] d\nu .$$
(2.29)

The smoothing density function in Equation (2.29) is evaluated at time t, as is the corresponding density function in  $E_{\rm m}h\left[\frac{\chi}{\chi}(t)\right]$ . This fact is used to advantage by deriving from Equations (2.27) and (2.29) an equation of the form

$$\frac{d\mathbf{E}_{gh}[\underline{z}(t)]}{dt} = \mathbf{E}_{g}\{\mathbf{K}_{h}[\underline{z}(t)]\}, \qquad (2,30)$$

where  $K_1$  is an operator yet to be determined. The simplest and quickest method of achieving this is first to note that the  $E_g$  operator corresponds to a scalar product and then to use the theory of adjoints. Denote the scalar product as

$$E_{a}h[x(t-dt)] = (h(\mu), e(\mu, t-dt)),$$
 (2.31)

If Equation (2,22) is written in the form

$$\mathbf{s}(\underline{\mu}, \mathbf{t} - d\mathbf{t}) = \mathbf{K}_{\mathbf{g}} \mathbf{s}(\underline{\mu}, \mathbf{t}) , \qquad (2.32)$$

where K, is the operator indicated by Equation (2.22), then Equation (2.31) can be written as

$$\mathbf{E}_{\mathbf{b}}[\mathbf{x}(t-\mathbf{d}t)] = \langle \mathbf{h}(\underline{\mu}), \mathbf{x}_{\mathbf{j}} \mathbf{n}(\underline{\mu}, t) \rangle . \tag{2.33}$$

Denoting the adjoint of  $K_{2}$  by  $K_{2}^{4}$  , Equation (2.33) becomes

$$\mathbf{E}_{\mathbf{a}} \mathbf{h} [\mathbf{x} (\mathbf{t} - \mathbf{d} \mathbf{t})] = \langle \mathbf{K}_{\mathbf{h}}^{\mathbf{h}} (\boldsymbol{\mu} \rangle, \ \mathbf{s} (\boldsymbol{\mu}, \mathbf{t}) \rangle, \qquad (2.34)$$

where K<sup>\*</sup><sub>3</sub>h(<sub>44</sub>) is given by

$$\mathbf{K}_{2}^{\bullet}\mathbf{h}(\underline{\mu}) = \int_{-\infty}^{\infty} \mathbf{h}(\underline{\nu}) \frac{\mathbf{p}_{\underline{\mathbf{g}}(\underline{\mathbf{t}})} \left[ \frac{[d] \mathbf{g}(\underline{\mathbf{t}} - d\underline{\mathbf{t}}) = \underline{\nu} \right] \mathbf{p}_{\underline{\mathbf{g}}(\underline{\mathbf{t}} - d\underline{\mathbf{t}})} \left[ \underline{\nu} | \mathbf{Z}_{\underline{\mathbf{t}} - d\underline{\mathbf{t}}} \right]}{\mathbf{p}_{\underline{\mathbf{g}}(\underline{\mathbf{t}})} \left[ \underline{\mu} | \mathbf{Z}_{\underline{\mathbf{t}} - d\underline{\mathbf{t}}} \right]} d\nu .$$
(2.35)

Note that the argument of  $K_{2}^{*}h$  is the argument after applying  $K_{2}^{*}$  to h.





The Constant Alask



Equation (2.27) can now be rewritten as

$$\frac{d\mathbf{E}_{g}h[\underline{x}(t)]}{dt} = \frac{1}{dt} \left\{ \langle h(\mu), \ h(\mu, t) \rangle - \langle K_{2}^{*}h(\mu), \ h(\mu, t) \rangle \right\}$$
(2.36)

and, by the linearity of scalar products,

$$\frac{dE_{gh}[\underline{x}(t)]}{dt} = \frac{1}{dt} \langle h(\underline{\mu}) - K_{h}^{*}h(\underline{\mu}), s(\underline{\mu}, t) \rangle , \qquad (2.37)$$

Equation (2.37) is the basic differential equation for the results to follow. It is necessary to evaluate the operator  $K_2^{\pm}$  first, and some necessary preliminary steps are presented in Section 2.3.

# 2.3 Expansion of a Particular Density Function

Before proceeding with the evaluation of the operator  $K_2^{\pm}$ , it is desirable to find a different expression for the density function  $p_{\frac{1}{2}(t)}[\underline{u}|\underline{x}(t-dt) = \underline{\nu}_1$ . This is accomplished in this section.

Because the integration in Equation (2.35) is over  $\nu$ , this equation can be rewritten as:

$$K_{2}^{*}h(\underline{\mu}) = \frac{1}{p_{\underline{f}(t)}[\underline{\mu}]Z_{t-dt}]} \int_{-\infty}^{\infty} h(\underline{\nu})p_{\underline{f}(t)}[\underline{\mu}]x(t-dt) = \underline{\nu}]p_{\underline{f}(t-dt)}[\underline{\nu}]Z_{t-dt}] d\nu. \qquad (2.38)$$

The term  $p_{g(t)}[\mu]x(t-dt) = \mu]$  is evaluated by consideration of the basic plant equation (Eqn. (2.3)) and the properties of the independent increment process. From Equation (2.3),

$$d\underline{\zeta}(t-dt) = \underline{x}(t) - \underline{x}(t-dt) - \underline{f}(\underline{x}, t-dt) dt + r. \qquad (2.39)$$

The Jacobian of this transformation is unity, so that, by first applying the Markov property and then Equation (2,39),

$$p_{\mathbf{x}(t)}[\underline{\mu}|\underline{x}(t-dt) = \underline{\nu}] = p_{\underline{x}(t)}[\underline{\mu}|\underline{x}(t-dt) = \underline{\nu}, \mathbf{z}_{t-dt}]$$
$$= P_{\underline{d}\underline{c}(t-dt)}[\underline{\mu} - \underline{x}(t-dt) - \underline{f}(\underline{x}, t-dt) dt + r]. \qquad (2.40)$$

Let

$$\mathbb{E}\left[d\zeta(t)\right] = 0 \tag{2.41}$$

$$E[d\zeta(t)d\zeta'(t)] = Q(t)dt + O(dt)$$
. (2.42)

Then the characteristic function becomes

$$\mathbf{g}(\mathbf{e}^{\underline{i}\underline{\omega}'d\underline{f}(\underline{t})}) = \exp\left\{-\frac{dt}{2}\underline{\omega}'\mathbf{Q}(\underline{t})\underline{\omega}\right\} + \mathbf{0}(dt) . \qquad (2.43)$$

With the aid of the inverse Fourier transform,

$$P_{d\underline{k}(t)}[\underline{\mu}] = \frac{1}{(2\pi)^n} \int_{-\infty}^{\infty} \exp\left\{-i\underline{\omega}'\underline{\mu} - \frac{dt}{2} \underline{\omega}'\underline{Q}(t)\underline{\omega}\right\} d\omega + O(dt) . \qquad (2.44)$$

Combining Equations (2.40) and (2.44),

$$p_{\underline{\mathbf{g}}(\underline{\mathbf{t}})}[\underline{\boldsymbol{\mu}}]\underline{\mathbf{x}}(\underline{\mathbf{t}}-d\underline{\mathbf{t}}) = \underline{\boldsymbol{\nu}}] = \frac{1}{(2\pi)^{\underline{\mathbf{n}}}} \int_{-\underline{\mathbf{m}}}^{\underline{\mathbf{m}}} \exp\left\{-\underline{\mathbf{i}}\underline{\boldsymbol{\omega}}'[\underline{\boldsymbol{\mu}}-\underline{\boldsymbol{\nu}}-\underline{\mathbf{f}}(\underline{\boldsymbol{\nu}},\,\underline{\mathbf{t}}-d\underline{\mathbf{t}})d\underline{\mathbf{t}}+\underline{\mathbf{r}}] - \frac{d\underline{\mathbf{t}}}{2}\underline{\boldsymbol{\omega}}'\underline{\mathbf{q}}(\underline{\mathbf{t}})\underline{\boldsymbol{\omega}}\right\} d\underline{\boldsymbol{\omega}} + O(d\underline{\mathbf{t}}) . \quad (2.45)$$

It is desirable to expand Equation (2.45) in a power series at this point:

$$P_{\underline{\mathbf{x}}(\underline{\mathbf{t}})}[\underline{\boldsymbol{\mu}}]\underline{\mathbf{x}}(\underline{\mathbf{t}}-d\underline{\mathbf{t}}) = \underline{\boldsymbol{\nu}}] = \frac{1}{(2\pi)^{\frac{1}{2}}} \int_{-\underline{\mathbf{u}}}^{\underline{\mathbf{v}}} \exp\left[-i\underline{\omega}'(\underline{\boldsymbol{\mu}}-\underline{\boldsymbol{\nu}})\right] \{1+i\underline{\omega}'[\underline{\mathbf{f}}(\underline{\boldsymbol{\nu}}, \underline{\mathbf{t}}-d\underline{\mathbf{t}}) d\underline{\mathbf{t}}]\} \times \left\{1-\frac{d\underline{\mathbf{t}}}{2}\underline{\omega}'\mathbf{Q}(\underline{\mathbf{t}})\underline{\omega}\right\} d\omega + \mathbf{0}(d\underline{\mathbf{t}}), \quad (2.46)$$

The following identities involving the Dirac delts function are easily verified:

Britten Barten Bient

 $\mathbb{R}^{n}$ 

$$\delta(\mu - \nu) = \frac{1}{(2\pi)^{n}} \int_{-\infty}^{\infty} \exp\{-i\omega'(\mu - \nu)\} d\omega, \qquad (2.47)$$

$$\frac{\partial \delta(\underline{\mu}-\underline{\nu})}{\partial \mu_{j}} = \frac{1}{(2m)^{n}} \int_{-\omega}^{\infty} -i\omega_{j} \exp\{-i\omega'(\underline{\mu}-\underline{\nu})\} d\omega, \qquad (2.48)$$

$$\frac{\partial^2 \delta(\underline{\mu}-\underline{\nu})}{\partial \mu_j \partial \mu_k} = \frac{1}{(2\pi)^n} \int_{-\infty}^{\infty} -\omega_j \omega_k \exp\left(-\frac{1}{\omega'}(\underline{\mu}-\underline{\nu})\right) d\omega , \qquad (2.49)$$

By the properties of the Dirac delts function,

$$\int_{-\infty}^{\infty} f(x, y) \,\delta(x - x) \,dx = f(z, y) \,. \tag{2.50}$$

Secause of the ways Equations (2.48) and (2.49) were obtained,

$$\int_{-\infty}^{\infty} \frac{\partial \delta(\underline{\mu}-\underline{\nu})}{\partial \mu_{j}} f(\underline{\nu}) d\nu = \frac{\partial}{\partial \mu_{j}} \int_{-\infty}^{\infty} \delta(\underline{\mu}-\underline{\nu}) f(\underline{\nu}) d\nu \qquad (2,51)$$

and

$$\int_{-\infty}^{\infty} \frac{\partial \mu_{j} \partial \mu_{k}}{\partial t_{j} \partial \mu_{k}} f(\nu) \, d\nu = \frac{\partial \mu_{j} \partial \mu_{k}}{\partial t_{k}} \int_{-\infty}^{\infty} \delta(\mu - \nu) f(\nu) \, d\nu \, . \qquad (2.52)$$

With the aid of Equations (2, 47) = (2, 49), Equation (2, 46) becomes

$$P_{\underline{x}(t)}[\mu|\underline{x}(t-dt) = \nu] = \delta(\mu-\nu) - \sum_{j=1}^{n} [f_{j}(\nu, t-dt)dt] \frac{\partial \delta(\mu-\nu)}{\partial \mu_{j}} + \frac{dt}{2} \sum_{j=1}^{n} \sum_{k=1}^{n} Q_{jk}(t) \frac{\partial^{2} \delta(\mu-\nu)}{\partial \mu_{j} \partial \mu_{k}} + O(dt) , \qquad (2.53)$$

which is the desired result of this section.

#### 3.4 Completion of Smoothed Expectation Equation

The derivation of the exact differential equation for the smoothed expectation of h(x) can now be completed. By using Equation (2,17), Equation (2,38) becomes

$$K_{z}^{h}(\underline{\mu}) = \frac{\int_{-\infty}^{\infty} p_{\underline{x}(t)}[\underline{\mu}|\underline{x}(t-dt) = \underline{\nu}]q(\underline{\nu}, t-dt)h(\underline{\nu}) d\nu}{\int_{-\infty}^{\infty} p_{\underline{x}(t)}[\underline{\mu}|\underline{x}(t-dt) = \underline{\nu}]q(\underline{\nu}, t-dt) d\nu}, \qquad (2.54)$$

5 . Se . Se . Se

For convenience, the numerator and denominator of Equation (2.54) will be denoted by Q and P , respectively. With the aid of Equation (2.53), Q can be written as

$$\mathbf{Q} = \int_{-\infty}^{\infty} d\nu h(\underline{\nu}) q(\underline{\nu}, t-dt) \left[ \delta(\underline{\mu}-\underline{\nu}) - \sum_{j=1}^{R} \left[ f_{j}(\underline{\nu}, t-dt) dt \right] \times \frac{\partial \delta(\underline{\mu}-\underline{\nu})}{\partial \mu_{j}} + \frac{dt}{2} \sum_{j=1}^{R} \sum_{k=1}^{R} Q_{jk}(t) \frac{\partial^{2} \delta(\underline{\mu}-\underline{\nu})}{\partial \mu_{j} \partial \mu_{k}} \right] + \eta(dt) . \quad (2.65)$$

Now straightforward application of Equations (2.50) - (2.52) yields

$$\mathbf{Q} = \mathbf{h}(\underline{\mu})\mathbf{q}(\underline{\mu}, \mathbf{t} - d\mathbf{t}) - d\mathbf{t} \left\{ \sum_{j=1}^{n} \frac{\partial}{\partial \mu_{j}} \left[ \mathbf{h}(\underline{\mu})\mathbf{q}(\underline{\mu}, \mathbf{t} - d\mathbf{t}) \left[ \mathbf{f}_{j}(\underline{\mu}, \mathbf{t} - d\mathbf{t}) \right] - \frac{1}{2} \sum_{j=1}^{n} \sum_{k=1}^{n} \frac{\partial^{2}}{\partial \mu_{j} \partial \mu_{k}} \left[ \mathbf{h}(\underline{\mu})\mathbf{q}(\underline{\mu}, \mathbf{t} - d\mathbf{t})\mathbf{q}_{jk}(\mathbf{t}) \right] \right\} + \mathbf{0}(d\mathbf{t}) . \quad (2.56)$$







100 9 14 Mar 17 10

10. 6 34.

**P** is simply evaluated by letting  $h(\underline{\mu}) = 1$  in Equation (2.56),

$$\mathbf{P} = \mathbf{q}(\underline{\mu}, \mathbf{t} - d\mathbf{t}) - d\mathbf{t} \left\{ \sum_{j=1}^{n} \frac{\partial}{\partial \mu_{j}} \left[ \mathbf{q}(\underline{\mu}, \mathbf{t} - d\mathbf{t}) \mathbf{f}_{j}(\underline{\mu}, \mathbf{t} - d\mathbf{t}) \right] - \frac{1}{2} \sum_{j=1}^{n} \sum_{k=1}^{n} \frac{\partial^{2}}{\partial \mu_{j} \partial \mu_{k}} \left[ \mathbf{Q}_{jk}(\mathbf{t}) \mathbf{q}(\underline{\mu}, \mathbf{t} - d\mathbf{t}) \right] \right\} + \mathbf{O}(d\mathbf{t}) . \quad (2.57)$$

Let  $\{\}_q$  denote the collection of terms in brackets  $\{\}$  in Equation (2.56). Let  $\{\}_p$  have a similar meaning for P, as given by Equation (2.57). Then

$$\frac{1}{P} = \frac{1}{q(\mu, t-dt)} \frac{1}{1 - \frac{dt()_{P}}{q(\mu, t-dt)} + 0(dt)}$$
(2.58)

Assuming  $\{\}_p/q(\mu, t-dt)$  is bounded, 1/P can be written as

$$\frac{1}{P} = \frac{1}{q(\underline{\mu}, t-dt)} \left( 1 + \frac{dt()_{P}}{q(\underline{\mu}, t-dt)} + O(dt) \right) . \qquad (2.59)$$

Substituting  $\{\}_q$  and  $\{\}_p$  as given in Equations (2.56) and (2.57), respectively, and simplifying, Equation (2.54) becomes

$$\frac{\mathbf{q}}{\mathbf{p}} = \mathbf{h}(\underline{\mu}) + d\mathbf{t} \left\{ -\sum_{j=1}^{n} \left[ \underline{f}(\underline{\mu}, \mathbf{t} - d\mathbf{t}) \right]_{j} \frac{\partial h(\underline{\mu})}{\partial \mu_{j}} + \frac{1}{2} \sum_{j=1}^{n} \sum_{k=1}^{n} \varphi_{jk}(\mathbf{t}) \frac{\partial^{2} h(\underline{\mu})}{\partial \mu_{j} \partial \mu_{k}} + \frac{1}{2} \sum_{j=1}^{n} \sum_{k=1}^{n} \varphi_{jk}(\mathbf{t}) \frac{\partial^{2} h(\underline{\mu})}{\partial \mu_{j}} \frac{\partial q(\underline{\mu}, \mathbf{t} - d\mathbf{t})}{\partial \mu_{k}} \right\} + O(d\mathbf{t}) .$$

$$(2.60)$$

In performing the simplifications leading to Equation (2.60), no use was made of the fact that Q is not a function of  $\mu$ . This fact will become important later. Since

$$\frac{1}{q(\mu, t - dt)} = \frac{1}{q(\mu, t) - \frac{\partial q(\mu, t)}{\partial t} dt + O(dt)}$$
$$= \frac{1}{q(\mu, t)} \left[ 1 + \frac{\partial q(\mu, t)}{\partial t} dt + O(dt) \right]$$
(2.61)

and

$$f_{j}(\mu, t-dt) = f_{j}(\mu, t) - \frac{\partial f_{j}(\mu, t)}{\partial t} dt + O(dt),$$
 (2.62)

it is apparent that Equation (2.60) is still valid whon all arguments t - dt are replaced by t. That is, the difference between the two expressions is of O(dt).

By definition.

 $\mathbf{K}_{\mathbf{p}}^{\mathbf{q}}\mathbf{h}(\underline{\mu}) = \frac{\mathbf{q}}{\mathbf{p}}.$  (2.63)

Comparing Equations (2.30) and (2.37), it is obvious that

$$K_{1}f = \frac{1}{dt} (I - K_{g}^{*})f . \qquad (2.64)$$

Equations (2.63) and (2.64) yield

$$\mathbf{K}_{i}\mathbf{h}(\underline{\mu}) = \sum_{j=1}^{n} \left[\underline{f}(\underline{\mu}, t)\right]_{j} \frac{\partial h(\underline{\mu})}{\partial \mu_{j}} - \frac{1}{2} \sum_{j=1}^{n} \sum_{k=1}^{n} \mathbf{Q}_{jk}(t) \frac{\partial^{2}h(\underline{\mu})}{\partial \mu_{j}\partial \mu_{k}} - \frac{1}{\mathbf{Q}(\underline{\mu}, t)} \sum_{j=1}^{n} \sum_{k=1}^{n} \mathbf{Q}_{jk}(t) \frac{\partial h(\underline{\mu})}{\partial \mu_{j}} \frac{\partial \mathbf{Q}(\underline{\mu}, t)}{\partial \mu_{k}} + \mathbf{O}(dt) .$$
(2.65)

A ST CONTRACTOR OF SALASS

Equation (2.30),

197

のないないです。

$$\frac{d\mathbf{E}_{\mathbf{g}}h[\underline{\mathbf{x}}(t)]}{dt} = \mathbf{E}_{\mathbf{g}}\{\mathbf{K}_{j}h[\underline{\mathbf{x}}(t)]\}, \qquad (2.66)$$

combined with Equation (2.65) is the desired differential equation for the smoothed expectation of an arbitrary function of x.

If the substitution

$$h(\mu) = \delta(\mu - \nu) \tag{2.67}$$

is made in both sides of Equation (2.66), the result is

$$\frac{is(\mu,t)}{dt} = -s(\mu,t) \sum_{j=1}^{n} \frac{\partial [\underline{f}(\mu,t)]_{j}}{\partial \mu_{j}} - \sum_{j=1}^{n} [\underline{f}(\mu,t)]_{j} + \frac{\partial s(\mu,t)}{\partial \mu_{j}} + \frac{1}{2} s(\mu,t) \sum_{j=1}^{n} \sum_{k=1}^{n} \frac{\partial^{2} q_{jk}(t)}{\partial \mu_{j} \partial \mu_{k}} - \frac{1}{2} \sum_{j=1}^{n} \sum_{k=1}^{n} q_{jk}(t) \times \frac{\partial^{2} s(\mu,t)}{\partial \mu_{j} \partial \mu_{k}} + \frac{1}{q(\mu,t)} \sum_{j=1}^{n} \sum_{k=1}^{n} q_{jk}(t) \frac{\partial s(\mu,t)}{\partial \mu_{j} \partial \mu_{k}} - \frac{1}{2} \sum_{j=1}^{n} \sum_{k=1}^{n} q_{jk}(t) \frac{\partial s(\mu,t)}{\partial \mu_{j} \partial \mu_{k}} - \frac{1}{2} \sum_{j=1}^{n} \sum_{k=1}^{n} q_{jk}(t) \frac{\partial s(\mu,t)}{\partial \mu_{j} \partial \mu_{k}} - \frac{1}{2} \sum_{j=1}^{n} \sum_{k=1}^{n} q_{jk}(t) \frac{\partial s(\mu,t)}{\partial \mu_{j} \partial \mu_{k}} + \frac{s(\mu,t)}{q(\mu,t)} \sum_{j=1}^{n} \sum_{k=1}^{n} \frac{\partial q_{jk}(t)}{\partial \mu_{j}} \frac{\partial q(\mu,t)}{\partial \mu_{k}} + \frac{s(\mu,t)}{q(\mu,t)} \sum_{j=1}^{n} \sum_{k=1}^{n} \frac{\partial q_{jk}(t)}{\partial \mu_{j}} \frac{\partial q(\mu,t)}{\partial \mu_{k}} + \frac{s(\mu,t)}{q(\mu,t)} \sum_{j=1}^{n} \sum_{k=1}^{n} \frac{\partial q(\mu,t)}{Q(\mu,t)} + \frac{s(\mu,t)}{Q(\mu,t)} \sum_{j=1}^{n} \sum_{k=1}^{n} \frac{\partial q(\mu,t)}{Q(\mu,t)$$

which is the differential equation for the smoothing density function.

It is possible to derive a differential equation for the smoothed expectation of an arbitrary function of the state in terms of filtered expectations. Striebel outlined a derivation of such an equation in Reference 2, based on the limit of a discrete formulation. Peller<sup>3</sup> used a similar method based on the formal teolulques with differentials used herein. The equation is presented here without derivation (see Peller<sup>3</sup>):

$$\frac{dE_{g}h[\underline{x}(t)]}{dt} = E_{f}\left\{L_{g}\left[\frac{a(\underline{x},t)}{q(\underline{x},t)}h(\underline{x})\right]\right\} - E_{f}\left\{h(\underline{x})L_{g}\frac{a(\underline{x},t)}{q(\underline{x},t)}\right\} + O(dt) , \qquad (2.69)$$

where L. is the backward diffusion operator:

$$L_{q}h(\underline{\nu}) = \sum_{j=1}^{n} r_{j}(\underline{\nu} + t - dt) \frac{\partial}{\partial \nu_{j}} h(\underline{\nu}) + \frac{1}{2} \sum_{j=1}^{n} \sum_{k=1}^{n} q_{jk}(t) \frac{\partial^{2}h(\underline{\nu})}{\partial \nu_{j}\partial \nu_{k}}.$$
 (2.70)

If the substitution

$$h(\mu) = \delta(\mu - \nu)$$
 (2.71)

is made in Equation (2.69), the result is the differential equation for the smoothing density function as already given by Equation (2.68), as is to be expected.

Equations (2.65), (2.65), and (2.68) are the principal results; they represent the "exact" differential equations satisfied by the smoothing expectation and smoothing density function. The boundary condition for Equation (2.68) is simply

$$e(\mu,T) = q(\mu,T)$$
, (2.72)

and the boundary condition for Equations (2.65) and (2.66) is given by the obvious relation

$$\mathbf{E}_{a}(h[\underline{x}(T)]) = \mathbf{E}_{p}(h[\underline{x}(T)]) , \qquad (2.73)$$

As was mentioned earlier, no use was made of the fact that Q is not a function of  $\mu$ . In fact, had use been made of this, it is obvious that one of the terms in Equation (2.65) would have been equal to zero. Because the results remain valid, these "exact" equations can easily be extended to cover the product noise case. First, for conciseness, Equation (2.65) is rewritten to show explicitly the possible dependence of Q on the system state:



and and and and the start

$$\mathbf{k}_{\mathbf{i}} \mathbf{h}(\underline{\mu}) = \sum_{\mathbf{j}=1}^{\mathbf{n}} \left[ \underline{\mathbf{f}}(\underline{\mu}, \mathbf{t}) \right]_{\mathbf{j}} \frac{\partial \mathbf{h}(\underline{\mu})}{\partial \mu_{\mathbf{j}}} - \frac{1}{2} \sum_{\mathbf{j}=1}^{\mathbf{n}} \sum_{\mathbf{k}=1}^{\mathbf{n}} \mathbf{q}_{\mathbf{j}\mathbf{k}}(\underline{\mu}, \mathbf{t}) \frac{\partial^{2} \mathbf{h}(\underline{\mu})}{\partial \mu_{\mathbf{j}} \partial \mu_{\mathbf{k}}} - \frac{1}{2} \sum_{\mathbf{j}=1}^{\mathbf{n}} \sum_{\mathbf{k}=1}^{\mathbf{n}} \mathbf{q}_{\mathbf{j}\mathbf{k}}(\underline{\mu}, \mathbf{t}) \frac{\partial^{2} \mathbf{h}(\underline{\mu})}{\partial \mu_{\mathbf{j}}} - \frac{1}{2} \sum_{\mathbf{j}=1}^{\mathbf{n}} \sum_{\mathbf{k}=1}^{\mathbf{n}} \mathbf{q}_{\mathbf{j}\mathbf{k}}(\underline{\mu}, \mathbf{t}) \frac{\partial^{2} \mathbf{h}(\underline{\mu})}{\partial \mu_{\mathbf{k}}} - \frac{1}{2} \sum_{\mathbf{j}=1}^{\mathbf{n}} \sum_{\mathbf{k}=1}^{\mathbf{n}} \mathbf{q}_{\mathbf{j}\mathbf{k}}(\underline{\mu}, \mathbf{t}) \frac{\partial^{2} \mathbf{h}(\underline{\mu})}{\partial \mu_{\mathbf{j}}} \frac{\partial \mathbf{q}(\underline{\mu}, \mathbf{t})}{\partial \mu_{\mathbf{k}}}$$

$$(2.74)$$

Now suppose

 $d\underline{\zeta}(t) = M(\underline{x}, t) d\underline{\omega}(t) , \qquad (2.75)$ 

where  $o_X(t)$  is an n-vector increment in an independent increment process and  $M(\underline{x}, b)$  is an  $n \times n$  matrix. Further suppose that

$$\mathbf{E}[d\omega(t)d\omega'(t)] = \Omega(t)dt + O(dt), \qquad (2.76)$$

then, by letting

$$Q(x,t) = M(x,t)\Omega(t)M'(x,t)$$
, (2.77)

Equations (2.66) and (2.74) apply directly, and the product noise case is accommodated.

#### 2.5 Commonts on the Exact Differential Equations

Equations (2.65), (2.66), and (2.68) together with the boundary conditions given by Equations (2.72) and (2.73) are in theory sufficient to molve any smoothing problem. In fact, Equations (2.68) and (2.72) are sufficient in themselves for this task mine once the smoothing density function is found, the smoothed expectation of any function of the system state can be found by integrating that function multiplied by the smoothing density function over all infinity in parameter space.

Although these equations are exact in the mean square sense, they are impossible to solve in practice and they must be replaced by solvable differential equations whose solutions are "approximately" equal to the solutions of these exact equations. The reasons that these equations are impossible to solve in practice are as follows. Equation (2.68) for the smoothing density function is a second-order partial differential equation in all components of the state vector whose domain is all infinity for this problem. The computer requirements that result from this type of problem far exceed what is considered practical by today's standards in computer technology. In addition, it is necessary to have the filtering density function,  $q(\mu, t)$ , available in analytical form or to have its first and second derivatives with respect to all state vector components available in any form. This is virtually never true in practice.

Now that the difficulties in solving for the smoothing density function are known, a brief note is in order on the difficulties in directly solving the equations for the smoothed expectation. First, note that the filtering density function and its first derivative with respect to all state vector components are required, something we have already agen to be almost never true. Second, note that the smoothing operator on the right side of Equation (2.66) operates on each term in Equation (2.65). This requires the evaluation of the 4moothed expectation of new terms. If these are computed by again applying Equations (2.65) and (2.60), still more new terms are introduced. In fact, if the process is continued, a countably infinite number of terms must be evaluated.

Thus, it is clearly impossible to solve directly for the smoothed expectation, or smoothing density function, and approximations which are more easily solved must be wought. In the next section, a particular set of approximations are developed which permit the determination of the minimum variance estimate. In a following section, a method of obtaining a maximum likelihood estimate will be described. This method is based on an entirely different procedure and will be presented without derivation.

#### 3. MINIMUM VARIANCE SMOOTHING APPROXIMATIONS

For linear systems, the smoothing density function is gaussian and is therefore completely described by the smoothed mean and covariance. The case of a nonlinear system is complicated by the fact that the smoothing density function is not necessarily gaussian, and, consequently, the smoothed mean and covariance may not completely describe it. In fact, as mentioned previously, a countably infinite number of parameters are necessary to completely describe smoothed parameters in the general case.

Because of this, approximations involving a finite number of parameters are sought. In the case of nonlinear filtering theory, two general approaches are most evident. One approach is to approximate the conditioned expectation of a function by a Taylor series truncated after the quadratic terms. This is the approach used by Budy<sup>4</sup>, Bass, Norum, and Schwartz<sup>5</sup>, and Schwartz<sup>6</sup>. It is equivalent to assuming certain higher order moments to be negligible. The second approach is to use quasi-moments. This approach, used by Fisher<sup>7</sup>, allows the conditional density function to be expressed in terms of the conditional mean, conditional covariance, and quasi-moment functions. Both approaches can be used for the nonlinear smoothing problem; the truncated Taylor series approach will be presented in this section. The quasi-moment technique has greater generality, but results











••••

Seren Den ein terme Diel

in very complex expressions when applied to the smoothing problem. Section 4.2 very briefly summarizes the idea of quasi-moments. (See Peller<sup>3</sup> for a more extensive discussion of the quasi-moment technique in nonlinear smoothing.)

#### 3.1 Nonlinear Minimum Variance Smoothing

For minimum variance smoothing, the conditional mean of the state is the desired estimate; thus, in Equations (2.64) and (2.65).

$$h_i[\underline{X}(t)] = X_i(t)$$
,  $i = 1, ..., n$ , (3.1)

so that

$$\frac{\tilde{z}}{\tilde{z}}(t) = \begin{bmatrix} E_{\mathbf{a}} h_{1} \\ \vdots \\ E_{\mathbf{a}} h_{\mathbf{a}} \end{bmatrix},$$
(3.2)

where the double overline in Heates the minimum variance smoothed estimate. From this point on, it is assumed that  $Q_{jk}$  is independent of the state. Using this assumption and substituting Equations (3.1), (2.1) and (2.64) into Equation (2.65),

$$\frac{d\bar{x}_{i}(t)}{dt} = \bar{f}_{i}(\underline{x},t) - \sum_{k=1}^{n} \bar{q}_{ik}(t) \overline{\frac{1}{q(\underline{x},t)} \frac{\partial q(\underline{x},t)}{\partial x_{k}}}.$$
(3.3)

Equation (3.3) is exact in the mean-square sense (and to within O(dt)). Approximations are now desired for each term on the right of Equation (3.3).

To approximate  $f_i(\underline{x}, t)$ , expand  $f_i(\underline{x}, t)$  in a Taylor series about  $\underline{\overline{x}}$  and drop terms of greater than second order:

$$\mathbf{f}_{i}(\underline{x}) \stackrel{i}{=} \mathbf{f}_{i}(\underline{\bar{x}}) + \sum_{j=1}^{n} \frac{\partial f_{i}(\overline{\bar{x}})}{\partial x_{j}} (x_{j} - \overline{\bar{x}}_{j}) + \frac{1}{2} \sum_{j=1}^{n} \sum_{k=1}^{n} \frac{\partial^{2} f_{i}(\overline{\bar{x}})}{\partial x_{j} \partial x_{k}} (x_{j} - \overline{\bar{x}}_{j}) (x_{k} - \overline{\bar{x}}_{k}) , \qquad (3.4)$$

where

$$\frac{\partial f_{i}(\vec{\underline{x}})}{\partial x_{j}} = \frac{\partial f_{i}(\underline{x})}{\partial x_{j}} |_{\vec{\underline{x}} = \vec{\underline{x}}}$$
(3.5)

$$\frac{\partial \mathbf{x}^{\mathbf{i}} \partial \mathbf{x}^{\mathbf{k}}}{\partial \mathbf{x}^{\mathbf{i}} \mathbf{x}^{\mathbf{i}}} = \frac{\partial \mathbf{x}^{\mathbf{i}} \partial \mathbf{x}^{\mathbf{k}}}{\partial \mathbf{x}^{\mathbf{i}} \mathbf{x}^{\mathbf{i}}} = \frac{\mathbf{x}^{\mathbf{i}}}{\mathbf{x}^{\mathbf{i}}}.$$
(3.6)

Applying the smoothing operator to both sides of Equation (3.4),

$$\vec{r}_{i}(\vec{x}) \doteq \vec{r}_{i}(\vec{x}) + \frac{1}{2} \sum_{j=1}^{n} \sum_{k=1}^{n} \frac{\partial^{2} \vec{r}_{i}(\vec{x})}{\partial x_{j} \partial x_{k}} (x_{j} - \vec{x}_{j}) (x_{k} - \vec{x}_{k}) . \qquad (3.7)$$

Noting the jkth element of the smoothing covariance matrix S is given by

$$B_{jk} = (x_j - \bar{x}_j)(x_k - \bar{x}_k) , \qquad (3.8)$$

Equation (3.7) becomes

$$\overline{f_{i}(\underline{s})} \stackrel{\pm}{=} f_{i}(\underline{s}) + \frac{1}{2} \sum_{k=1}^{n} \sum_{k=1}^{n} s_{jk} \frac{\partial s_{i}(\underline{s})}{\partial z_{j} \partial z_{k}} , \qquad (3.9)$$

An approximation for  $[1/q(\underline{x},t)][\partial_q(\underline{x},t)/\partial x_{\underline{x}}]$  is now sought. If  $q(\underline{x},t)$  were actually known in the form of some analytic expression, such as Fisher's quasi-moment expansion<sup>7</sup>, the desired approximation could easily be attained by techniques like those used to evaluate  $f_1(\underline{x})$ . 'In practice,  $q(\underline{x},t)$  is unlikely to be available in analytic form. For our purposes, it will be assumed that the filtered mean and covariance have been obtained (or approximated) by some technique such as that of Schwartz<sup>4</sup>. To obtain an approximation for  $[1/q(\underline{x},t)][\partial_q(\underline{x},t)/\partial x_{\underline{x}}]$ , it will be assumed that  $q(\underline{x},t)$  is nearly gaussian, so that it is adequately approximated by a gaussian density function having the 'computed filtered mean and covariance. Denoting the filtered mean and covariance by  $\underline{x}$  and P, respectively.

$$\mathbf{q}(\underline{x}, \underline{t}) \stackrel{\perp}{=} \frac{1}{(2\pi)^{11/2} (|\underline{p}|)^{1/2}} \exp\{-\frac{1}{2} [\underline{x} - \underline{x}]^{1} p^{-1} [\underline{x} - \underline{x}]\}, \qquad (3.10)$$

where the time dependence has been suppressed. By straight-forward evaluation (recalling that P<sup>-1</sup> is aymmetric),

$$\frac{\partial q(\underline{x},t)}{\partial x_{k}} \stackrel{i}{=} q(\underline{x},t) \sum_{l=1}^{n} (P^{-1})_{kl} (\overline{x}_{l} - x_{l}) . \qquad (3.11)$$

Consequently,

5727772A

$$\frac{1}{q(\underline{x},t)} \frac{\partial q(\underline{x},t)}{\partial x_{\underline{k}}} \stackrel{:}{=} \sum_{l=1}^{n} \left[ P^{-1}(t) \right]_{\underline{k}\, l} (\overline{x}_{l} - x_{l}) , \qquad (3.12)$$

Applying the smoothing operator to Equation (3.12).

$$\frac{1}{q(\underline{x},t)} \frac{\partial q(\underline{x},t)}{\partial x_k} \stackrel{i}{=} \sum_{l=1}^n \left[ P^{-1}(t) \right]_{kl} (\bar{x}_l - \bar{x}_l) , \qquad (3.13)$$

with the aid of Equations (3.3), (3.9), and (3.13), the differential equation for  $\vec{X}_1$  becomes:

$$\frac{d\bar{x}_{i}(t)}{dt} = f_{i}(\bar{x},t) + \frac{1}{2} \sum_{j=1}^{n} \sum_{k=1}^{n} B_{jk} \frac{\partial^{2} f_{i}(\bar{x})}{\partial x_{j} \partial x_{k}} + \sum_{k=1}^{n} \sum_{l=1}^{n} Q_{ik}(t) [P^{-1}(t)]_{kl}(\bar{x}_{l} - \bar{x}_{l}) .$$
(3.14)

A differential equation for the covariance is now desired, since it appears in Equation (3.14). By the linearity of the smoothing operator,

$$B_{jk} = (x_j - \bar{x}_j)(x_k - \bar{x}_k) = \bar{x}_j \bar{x}_k - \bar{x}_j \bar{x}_k$$
. (3.15)

Since Equation (3.15) implies

$$\dot{\mathbf{S}}_{\mathbf{j}\mathbf{k}} = \frac{d}{dt} \overline{\mathbf{x}_{\mathbf{j}} \mathbf{x}_{\mathbf{k}}} - \frac{d}{dt} \left( \overline{\mathbf{x}}_{\mathbf{j}} \overline{\mathbf{x}}_{\mathbf{k}} \right), \qquad (3.16)$$

it suffices to evaluate  $d(\overline{x_j x_k})/dt$  and  $d(\overline{x_j} \overline{x}_k)/dt$  . By letting

 $h(\mathbf{x}) = \mathbf{x}_j \mathbf{x}_k \tag{3.17}$ 

in Equations (2.64) and (2.65).

$$\frac{dx_{j}x_{k}}{dt} = f_{j}x_{k} + f_{k}x_{j} - Q_{jk} - \sum_{l=1}^{n} (Q_{j}x_{k} + Q_{k}x_{j}) \sum_{m=1}^{n} (P^{-1})_{lm}(\bar{x}_{m} - \bar{x}_{m}) .$$
(3.18)

By the linearity of the smoothing operator,

$$\frac{dx_{j}x_{k}}{dt} = \overline{t_{j}x_{k}} + \overline{t_{k}x_{j}} - \bar{q}_{jk} + \sum_{l=1}^{n} \sum_{m=1}^{n} \bar{q}_{jl} (P^{-1})_{lm} (\overline{x_{k}x_{m}} - \overline{\bar{x}}_{k}\overline{x}_{m}) + \\ + \sum_{l=1}^{n} \sum_{m=1}^{n} \bar{q}_{kl} (P^{-1})_{lm} (\overline{x_{j}x_{m}} - \overline{\bar{x}}_{j}\overline{x}_{m}) .$$
(3.19)

For mechanisation of Equation (3.19), it is sufficient to obtain a mechanizable suproximation to  $f_j x_k$  for all j and k, assuming that Equation (3.19) is mechanized for all j and k. It has screed been assumed that the third central moment is negligible. Since

$$\mathbf{E}_{\mathbf{s}}[(\mathbf{x}_{j} - \bar{\mathbf{x}}_{j})(\mathbf{x}_{k} - \bar{\mathbf{x}}_{k})(\mathbf{x}_{l} - \bar{\mathbf{x}}_{l})] = \mathbf{E}_{\mathbf{s}}[(\mathbf{x}_{j} - \bar{\mathbf{x}}_{j})(\mathbf{x}_{k} - \bar{\mathbf{x}}_{k})\mathbf{x}_{l}] - \mathbf{E}_{\mathbf{s}}[(\mathbf{x}_{j} - \bar{\mathbf{x}}_{j})(\mathbf{x}_{k} - \bar{\mathbf{x}}_{k})\bar{\mathbf{x}}_{l}], \qquad (3.20)$$

this assumption implies that

$$\mathbf{x}_{\mathbf{0}}[(\mathbf{x}_{j} - \mathbf{\bar{x}}_{j})(\mathbf{x}_{k} - \mathbf{\bar{x}}_{k})\mathbf{x}_{l}] \stackrel{\text{d}}{=} \mathbf{s}_{jk}\mathbf{\bar{x}}_{l}, \qquad (3, 21)$$

Multiplying Equation (3.4) by  $x_k$  ,

$$f_{j}(\underline{x}) x_{\underline{k}} \stackrel{i}{=} f_{j}(\underline{\ddot{x}}) x_{\underline{k}} + \sum_{l=1}^{n} \frac{\partial f_{j}(\underline{\ddot{x}})}{\partial x_{l}} (x_{l} - \overline{\ddot{x}}_{l}) x_{\underline{k}} + \frac{1}{2} \sum_{l=1}^{n} \sum_{\underline{n=1}}^{n} \frac{\partial^{2} f_{j}(\underline{\ddot{x}})}{\partial x_{l} \partial x_{\underline{n}}} (x_{l} - \overline{\ddot{x}}_{l}) (x_{\underline{n}} - \overline{\ddot{x}}_{\underline{n}}) x_{\underline{k}} , \qquad (3.22)$$



and subtracting  $\tilde{f}_j \tilde{x}_k$  from each side (using the expansion Equation (3.7) for  $\tilde{f}_j$  on the right side of Equation (3.22)):

$$f_{\mathbf{j}}(\underline{\mathbf{x}})\mathbf{x}_{\mathbf{k}} - \overline{\mathbf{x}}_{\mathbf{j}}\overline{\mathbf{x}}_{\mathbf{k}} \stackrel{i}{=} f_{\mathbf{j}}(\overline{\underline{\mathbf{x}}})\mathbf{x}_{\mathbf{k}} + \sum_{i=1}^{n} \frac{\partial f_{i}(\overline{\underline{\mathbf{x}}})}{\partial \mathbf{x}_{i}} (\mathbf{x}_{i} - \overline{\mathbf{x}}_{i})\mathbf{x}_{\mathbf{k}} + \frac{1}{2} \sum_{i=1}^{n} \sum_{m=1}^{n} \frac{\partial^{2} f_{i}(\overline{\underline{\mathbf{x}}})}{\partial \mathbf{x}_{i} \partial \mathbf{x}_{m}} (\mathbf{x}_{i} - \overline{\mathbf{x}}_{i})(\mathbf{x}_{m} - \overline{\mathbf{x}}_{m})\mathbf{x}_{\mathbf{k}} - \overline{\mathbf{x}}_{\mathbf{k}}\mathbf{x}_{i}(\overline{\underline{\mathbf{x}}}) - \frac{1}{2} \overline{\mathbf{x}}_{\mathbf{k}} \sum_{i=1}^{n} \sum_{m=1}^{n} \frac{\partial^{2} f_{i}(\overline{\underline{\mathbf{x}}})}{\partial \mathbf{x}_{i} \partial \mathbf{x}_{m}} (\mathbf{x}_{i} - \overline{\mathbf{x}}_{i})(\mathbf{x}_{m} - \overline{\mathbf{x}}_{m}) .$$

$$(3.23)$$

Applying the amoothing operator to both sides of Equation (3.23) and using Equation (3.21),

$$\overline{f_j(\underline{x})} x_{\underline{k}} \doteq \overline{f_j} \overline{x}_{\underline{k}} + \sum_{l=1}^{n} \frac{\partial f_l(\underline{x})}{\partial x_l} g_{l\underline{x}}. \qquad (3.24)$$

With the mid of Equations (3.7) and (3.24), Equation (3.19) becomes

$$\frac{d\bar{x}_{j}\bar{x}_{k}}{dt} \stackrel{i}{=} f_{j}(\bar{\bar{x}})\bar{\bar{x}}_{k} + f_{k}(\bar{\bar{x}})\bar{\bar{x}}_{j} + \frac{1}{2}\bar{\bar{x}}_{k}\sum_{l=1}^{n}\sum_{m=1}^{n} B_{lm}\frac{\partial^{2}f_{l}(\bar{\bar{x}})}{\partial x_{l}\partial x_{m}} + \\ + \frac{1}{2}\bar{\bar{x}}_{j}\sum_{l=1}^{n}\sum_{m=1}^{n} S_{lm}\frac{\partial^{2}f_{k}(\bar{\bar{x}})}{\partial x_{l}\partial x_{m}} + \sum_{l=1}^{n} B_{lm}\frac{\partial f_{l}(\bar{\bar{x}})}{\partial x_{l}} + \\ + \sum_{l=1}^{n} B_{lj}\frac{\partial f_{k}(\bar{\bar{x}})}{\partial x_{l}} - Q_{jk} + \sum_{l=1}^{n}\sum_{m=1}^{n} Q_{jl}(P^{-1})_{lm}(\bar{x}_{k}\bar{x}_{m}} - \bar{\bar{x}}_{k}\bar{x}_{m}) + \\ + \sum_{l=1}^{n}\sum_{m=1}^{n} Q_{kl}(P^{-1})_{lm}(\bar{x}_{j}\bar{x}_{m}} - \bar{\bar{x}}_{j}\bar{x}_{m}) .$$
(3.25)

It remains to evaluate  $d(\bar{x}_j \bar{x}_k) / dt$ . It is not really necessary to evaluate this term since, in evaluating  $S_{jk}$ , it is possible to generate  $\bar{x}_j \bar{x}_k$  by Equation (3.25) and to generate  $\bar{x}_j \bar{x}_k$  as the product of solutions to Equation (3.14). However, it is interesting to study the basic differential equation satisfied by  $S_{jk}$ , so an expression for the remaining term of Equation (3.16) will be derived.

For this case, both the ordinary calculus and the stochastic calculus imply (Peller<sup>3</sup>):

$$\frac{d\vec{x}_j \vec{x}_k}{dt} = \vec{x}_j \frac{d\vec{x}_k}{dt} + \vec{x}_k \frac{d\vec{x}_j}{dt}.$$
(3.26)

Using Equation (3.14) in Equation (3.26),

$$\frac{d\bar{x}_{j}\bar{y}_{k}}{dt} = f_{k}(\bar{x})\bar{x}_{j} + f_{j}(\bar{x})\bar{x}_{k} + \frac{1}{2}\bar{x}_{j}\sum_{l=1}^{n}\sum_{\mu=1}^{n}S_{lm}\frac{\partial^{2}f_{k}(\bar{x})}{\partial x_{l}\partial x_{m}} + \\
+ \frac{1}{2}\bar{x}_{k}\sum_{l=1}^{n}\sum_{\mu=1}^{n}S_{lm}\frac{\partial^{2}f_{j}(\bar{x})}{\partial x_{l}\partial x_{m}} + \sum_{l=1}^{n}\sum_{\mu=1}^{n}Q_{kl}[P^{-1}]_{lm}(\bar{x}_{\mu} - \bar{x}_{\mu})\bar{x}_{j} + \\
+ \sum_{l=1}^{n}\sum_{\mu=1}^{n}Q_{jl}[P^{-1}]_{lm}(\bar{x}_{\mu} - \bar{x}_{\mu})\bar{x}_{k}.$$
(3.27)

Using Equations (3.15) and (3.27) and recalling that  $S_{jk} = S_{kj}$  .

$$\frac{ds_{jk}(t)}{dt} = \sum_{l=1}^{n} \left[ s_{lk} \frac{\partial t_{k}(\vec{x})}{\partial x_{l}} + s_{lj} \frac{\partial t_{k}(\vec{x})}{\partial x_{l}} \right] + \sum_{l=1}^{n} \sum_{m=1}^{n} \left\{ Q_{jl}(p^{-1})_{lm} s_{mk} + Q_{kl}(p^{-1})_{lm} s_{mj} \right\} - Q_{jk} . \quad (3.28)$$

Equations (3.14) and (3.28) are the principal results of this section; they are the desired equations for approximate conlinear minimum variance smoothing.

anite a sugar

#### 3.2 Linear Minimum Variance Smoothing

It is interesting to determine the results of applying Equations (3.14) and (3.28) to the case of a linear system. For a linear system, Equation (2.1) becomes

$$\frac{1}{2}(t) = F(t) \underline{x}(t) + \hat{\zeta}(t)$$
, (3.29)

where F(t) is an  $n \times n$  matrix. Thus

 $f_{i}(\bar{\tilde{x}},t) = \sum_{j=1}^{n} F_{ij}\bar{\tilde{x}}_{j}$ (3.30)

and

$$\frac{\partial^2 f_i(\vec{k})}{\partial x_i \partial x_k} = 0, \qquad (3.31)$$

Substituting Equations (3.29) and (3.30) into Equation (3.14) yields

$$\frac{d\bar{x}_{i}}{dt} = \sum_{j=1}^{n} F_{ij}\bar{x}_{j} + \sum_{k=1}^{n} \sum_{l=1}^{n} Q_{ik}(t) \left[P^{-1}(t)\right]_{kl} (\bar{x}_{l} - \bar{x}_{l}) , \qquad (3.32)$$

or, equivalently,

$$\frac{d\bar{x}_{i}}{dt} = F_{\underline{x}}^{\underline{x}}(t) + Q(t)T^{-1}(t)[\bar{\underline{x}}(t) - \underline{x}(t)]. \qquad (3,33)$$

Proceeding similarly for Sik .

$$\sum_{l=1}^{n} \frac{\partial f_{j}(\bar{x})}{\partial x_{l}} \mathbf{g}_{lk} = \sum_{l=1}^{n} f_{jl}(t) \mathbf{g}_{lk}, \qquad (3.34)$$

which, when substituted into Equation (3.28), becomes

$$\frac{dS_{jk}(t)}{dt} = \sum_{l=1}^{n} (F_{jl}S_{lk} + F_{kl}S_{lj}) - Q_{jk} + \sum_{l=1}^{n} \sum_{m=1}^{n} Q_{jl} (P^{-1})_{lm} S_{mk} + \sum_{l=1}^{n} \sum_{m=1}^{n} Q_{kl} (P^{-1})_{lm} S_{mj}$$
(3.35)

or, equivalently,

$$\frac{dS(t)}{dt} = [P(t) + Q(t)P^{-1}(t)]S(t) + S(t)[P(t) + Q(t)P^{-1}(t)] - Q(t) . \qquad (3.36)$$

Equations (5.33) and (3.36) are identical to linear smoothing results obtained by Rauch, Tung, and Striebel<sup>4</sup> and repeated disewhere in this publication. Because the smoothing density function is gaussian, the minimum variance and maximum likelihood estimates are identical and are provided by the solution to Equation (3.33). Boundary conditions for Equations (3.33) and (3.36) are

and

$$S(T) = P(T)$$
, (3.38)

where T is the final time in the amoothing interval.

The fact that Equations (3.14) and (3.28) reduce to forms identical with previously derived results for linear systems is expected. Each approximation introduced in the derivations of the equations in this section is not an approximation at all when applied to linear systems. The filter density,  $q(\mu, t)$ , is gaussian in this case, and the third central moments are zero. The expansions about  $\frac{1}{2}$  are also exact in this case.

#### 3.3 An Iterative Approach

The approximate nonlinear minimum variance amouther has the advantage of relative simplicity, but is based on certain questionable assumptions. Assuming the third central moment to be negligible is equivalent to assuming there is no skewness to the density of g(t) conditioned on all the data. However, all sorts of simple nonlinearities produce donsities with significant skewness when driven by forcing functions containing gaussian noise. It is, of course, quite possible to use higher order approximations. Whether the improved performance would justify the increased complexity is questionable. Including a moment of order m as the highest moment in the approximation always requires the assumption that the moment of order m + 1 is negligible. Because oven



order moments (at least for near gaussian cases) are not at all negligible, it is clear that m + i should not be an even number. Consequently, to include the effects of skowness, both third and fourth order moments should be included. It appears that such a procedure may soon destroy the simplicity of the approach.

If one is truly concerned about the effect of assuming the third central moment to be negligible, the following iterative approach is suggested. In Equations (3.14) and (3.28), the term  $\partial f_j(\bar{X})/\partial x_i$  was used as a shorthand for  $\partial f_j(x)/\partial x_i$  evaluated at  $\chi = \bar{X}$ , and similarly for  $\partial^2 f_j(x)/(\partial x_i \partial x_i)$ . If these terms are evaluated for  $\chi = \Omega^0$ , where  $\hat{\chi}^0$  is the oth estimate of  $\chi$ , then the following iterative equations are suggested to use in place of Equations (3.14) and (3.28):

$$\hat{\mathbf{x}}_{1}^{q+1} = f_{1}(\hat{\mathbf{x}}^{q}, t) + \frac{1}{2} \sum_{j=1}^{n} \sum_{k=1}^{n} \mathbf{s}_{jk} \frac{\partial^{2} f_{1}(\hat{\mathbf{x}}^{q})}{\partial x_{j} \partial x_{k}} + \sum_{k=1}^{n} \sum_{l=1}^{n} \mathbf{Q}_{lk} [l^{p-1}]_{kl}(\hat{\mathbf{x}}_{l}^{q} - \tilde{\mathbf{x}}_{l}) , \qquad (3.39)$$

$$\dot{\mathbf{S}}_{jk}^{\sigma+1} = \sum_{l=1}^{n} \left[ \mathbf{S}_{lk}^{\sigma} \frac{\partial f_{j}(\hat{\mathbf{x}}^{\sigma})}{\partial x_{l}} + \mathbf{S}_{lj}^{\sigma} \frac{\partial f_{k}(\hat{\mathbf{x}}^{\sigma})}{\partial x_{l}} \right] + \sum_{l=1}^{n} \sum_{m=1}^{n} \left\{ \mathbf{Q}_{jl} \left[ \mathbf{P}^{-1} \right]_{lm} \mathbf{S}_{mk}^{\sigma} + \mathbf{Q}_{kl} \left[ \mathbf{P}^{-1} \right]_{lm} \mathbf{S}_{mj}^{\sigma} \right\} \right\}$$
(3.40)

This type of iteration is similar to quasi-linearization, and whereas no specific studies have been performed, it is anticipated that  $\frac{1}{3^{\circ}} \rightarrow \frac{1}{3}$  and  $3^{\circ} \rightarrow 3$  very rapidly, probably to high accuracy with only a couple of iterations. It should be noted that Equations (3.14) and (3.28) should provide botter estimates for nonlinear systems than linear theory, and the use of the iterative tochnique should permit quick convergence to very accurate estimates - even more accurate then Equations (3.14) and (3.28) used in non-iterative fashion.

# 4. MAXIMUM LIKELIHOOD SMOOTHING

For nonlinear systems, the smoothing density function will not, in general, be gaussian, thus the mean value (minimum variance estimate) will not, in general, be equal to the value for which the smoothing donsity function has its maximum value. In some cases, one is more interested in the maximum likelihood estimate. There are two approaches to this problem. One approach is to evaluate the maximum likelihood estimate directly, and a technique for this is presented in Section 4.1. The second approach is first to find the smoothing density function and then to find that value of the state for which the smoothing density function has its maximum value (for each t in the smoothing interval). A possible technique of doing this and the resultant difficulties will be briefly discussed in Section 4.2.

#### 4.1 Direct Approach

10

This direct approach is due to Bryson and Franier". Let the system be given by the nonlinear vector equation,

$$\dot{\mathbf{x}}(t) = f(\underline{\mathbf{x}}, \underline{\mathbf{0}}, t)$$
, (4.1)

where  $\underline{x}(t)$  is the n-dimensional state vector and  $\underline{g}(t)$  is a control vector independent of the state. Assume the a priori estimates are available:

$$[\underline{x}(t_n)] = \overline{x}_n \tag{4.2}$$

$$\mathbb{E}\left\{\left[\underline{x}(t_{n})-\overline{x}_{n}\right]\left[\underline{x}(t_{n})-\overline{x}_{n}\right]^{\prime}\right\}=\overline{P}_{n}$$
(4.3)

$$\mathbf{E}[g(t)] = \tilde{g}(t) \tag{4.4}$$

$$\mathbb{E}\{[\mathbf{a}(t) - \mathbf{\bar{o}}(t)][\mathbf{c}(\tau) - \mathbf{\bar{o}}(\tau)]^{\prime}\} = \overline{\mathbf{Q}}(t)\delta(t-\tau), \qquad (4,5)$$

Let the relationship between the system state  $\underline{x}(t)$  and the measurement process be given by

$$h[x(t), x(t), t] = 0$$
 (4.6)

for the case of no measurement noise (h is a p-dimension vector). Let

$$\mathbb{E}\left\{\left[\underline{x}(t) - \overline{x}(t)\right]\left[\underline{x}(\tau) - \overline{x}(\tau)\right]'\right\} = \overline{x}(t)\delta(t - \tau)$$
(4.7)

be an a priori estimate of errors in g(t). The Dirac delta function appearing in these equations indicates "white noise" in the plant, control, and measurements.

Bryson and Frazier find the maximum likelihood estimate of  $\underline{q}(t)$  and the initial condition,  $\underline{x}(t_0)$ . They also state that the maximum likelihood estimate  $\underline{x}(t)$  of the state  $\underline{x}(t)$  is obtainable from the maximum likelihood estimate  $\underline{q}(t)$  of the coutrol  $\underline{q}(t)$  by the equation

 $\hat{g} = f(\hat{g}, \hat{g}, t)$ , (4.8)



Ra Constantination For linear systems, this is clearly true. However, for nonlinear systems, it is not known to be true in general, and, in fact, the maximum of one probability density does not necessarily map into the maximum of another probability density function simply because of some functional relationship between the two variables. Therefore, the technique will be presented here with the understanding that the result of the computations will be the maximum likelihood estimate of the control, and that g(t) is simply that estimate of the state which is consistent with the maximum likelihood estimate of the control, i.e., that which satisfies Equation (4.8).

The maximum likelihood estimate of  $\chi(t_n)$  and  $\zeta(t)$  is found by minimizing the functional,

$$J = \frac{1}{2} \tilde{\chi}'(t_0) \tilde{P}_0^{-1} \tilde{\chi}(t_0) + \frac{1}{2} \int_{t_0}^{T} (\tilde{\chi}' \tilde{R}^{-1} \tilde{\chi} + \tilde{\Omega}' \tilde{Q}^{-1} \tilde{\chi}) dt , \qquad (4.9)$$

where

$$(\tilde{}) = (\hat{}) - (\tilde{})$$
 (4.10)

with the constraints

$$h(\hat{\boldsymbol{x}}, \hat{\boldsymbol{x}}, t) = 0 \tag{4.11}$$

and

$$\hat{X} = f(\hat{X}, q, t)$$
, (4.12)

By adjoining the constraints of Equations (4,11) and (4,12) and applying the calculus of variations, Bryson and Frazier show that the smoothed values of the control and initial condition satisfy

$$\underline{\hat{g}}(t) = \underline{\tilde{g}}(t) + \underline{\tilde{q}}(t) \left( \frac{\partial g}{\partial u} \right) \underline{\hat{\lambda}}(t) , \qquad (4.13)$$

$$\dot{\Delta}(t) = -\left(\frac{\partial A}{\partial \mathbf{x}}\right)'\dot{\Delta}(t) - \left(\frac{\partial \mathbf{h}}{\partial \mathbf{x}}\right)' \left[\left(\frac{\partial \mathbf{h}}{\partial \mathbf{x}}\right)'\right]^{-1} \mathbf{\ddot{x}}^{-1}(\mathbf{\hat{x}} - \mathbf{x}) , \qquad (4.14)$$

where

Į

$$\hat{I}(t) = \underline{\tilde{u}}(t) + \overline{\tilde{u}}\left(\frac{\partial I}{\partial \alpha}\right)' \underline{\lambda}(t) , \qquad (4.15)$$

$$\mathbf{\underline{h}}(\hat{\mathbf{a}}, \hat{\mathbf{a}}, t) = \mathbf{\underline{0}}, \qquad (4.16)$$

and the boundary conditions,

$$\hat{\mathbf{g}}(\mathbf{t}_{o}) = \hat{\mathbf{g}}(\mathbf{t}_{o}) + \hat{\mathbf{P}}_{o}\hat{\mathbf{\Delta}}(\mathbf{t}_{o}) \tag{4.17}$$

$$\underline{\lambda}(\mathbf{T}) = \underline{\mathbf{0}} , \qquad (4.18)$$

By this approach, the smoothing problem (for maximum likelihood estimates) becomes a two-point boundary value problem. The numerical solution of two-point boundary value problems on digital computers is often very difficult, although good results are often attained by quasi-linearization techniques. Assuming such techniques succeed in solving the numerical version of this two-point boundary value problem, then this approach has the advantage of not requiring a solution to the filtering problem before the smoothing problem can be worked.

#### 4.3 Quasi-Moment Approach

As mentioned previously, another approach to the maximum likelihood estimution problem is to determine the smoothing density function and then to find that value which maximizes the value of that function; this must be done for each value of t in the smoothing interval. In this section, the technique of expressing a density function in terms of quasi-moments is briefly described, and the reader is then referred to Peller<sup>3</sup> for further information on this approach. The material contained in this section is almost entirely derived from Fisher's work<sup>7</sup>, particularly as presented in Reference 10. In fact, this mention should simply be considered as a celective abridgment of Fisher's work.

Let  $p(\mu,t)$  be an n-dimensional density function, and let  $p_g(\mu,t)$  be an n-dimensional density function having the same mean and dovariance as  $p(\mu,t)$ . Let

$$\rho(\underline{\mu}, t) = \frac{p(\underline{\mu}, t)}{p_{\underline{\mu}}(\underline{\mu}, t)} .$$
(4.19)

Then, the coefficients of the expansion of  $\rho(\mu, t)$  in a series of multi-dimensional Hermite polynomials are the so-called quasi-moment functions. Since the multi-dimensional Hermite polynomials are a complete set of signa-functions over n-dimensional Euclidean space, any probability density function can be written as a series of these polynomials, provided  $\rho(\mu, t)$  is square-integrable, i.e., if

$$\int_{-\infty}^{\infty} \rho^2(\mu, t) \, d\mu < \omega \,. \tag{4.20}$$

It is also possible to approximate  $\rho(\mu,t)$ , if it satisfies Equation (4.20), to any specified accuracy in the integrable square error sense by a finite number of terms in the series.

Denoting the mean and covariance by  $\underline{m}(t)$  and C(t), respectively, and the characteristic functions of  $p(\underline{\mu}, t)$  and  $p_{\underline{g}}(\underline{\mu}, t)$  by  $\phi(\underline{\alpha}, t)$  and  $\phi_{\underline{g}}(\underline{\alpha}, t)$ , respectively. Finher shows that

$$\mathfrak{p}(\underline{\mu}, \mathfrak{t}) = \mathfrak{p}_{\mathfrak{g}}(\underline{\mu}, \mathfrak{t}) + \sum_{N=0}^{\infty} \frac{(-1)^{N}}{N!} \underbrace{\frac{j, k, \dots, l=1}{N}}_{N} \underbrace{K_{j, k, \dots, l}}_{N} (\mathfrak{t}) \frac{\partial^{N} \mathfrak{p}_{\mathfrak{g}}(\underline{\mu}, \mathfrak{t})}{\partial \mu_{j}, \partial \mu_{k}, \dots, \partial \mu_{l}}, \quad (4.21)$$

where

ΥĽ

$$K_{j,k,\ldots,l}(t) = \frac{1}{iN} \frac{\partial^{N}[\phi(\alpha,t)/\phi(\alpha,t)]}{\partial \alpha_{j} \partial \alpha_{k} \dots \partial \alpha_{l}} \frac{\partial^{N}[\phi(\alpha,t)]}{\partial \alpha_{j} \partial \alpha_{k} \dots \partial \alpha_{l}}$$
(4.22)

are the quasi-moment functions of order N . It can easily be shown that all quasi-moments of first and mecond order are zero.

Because the multi-dimensional Hermite polynomials corresponding to the  $n \times n$  matrix  $C^{-1}(t)$  are related to a generating function as follows,

$$H_{\underline{j},\underline{k},\ldots,l}(\underline{\mu},t) = (-1)^{N} \exp\left[\frac{1}{\mu} C(t)\underline{\mu}\right] \frac{\partial^{N} \exp\left[-\frac{1}{\mu} C^{-1}(t)\underline{\mu}\right]}{\partial \mu_{j}, \partial \mu_{k},\ldots,\partial \mu_{l}}, \qquad (4.23)$$

No I formation of

it is possible to rewrite Equation (4.22) in terms of the multi-dimensional Hermite polynomials:

$$p(\underline{\mu},t) = p_{g}(\underline{\mu},t) \left\{ 1 + \sum_{N=3}^{\infty} \frac{1}{N!} \underbrace{\sum_{j,k,\dots,l=1}^{n}}_{N,k,\dots,l=1} K_{\underline{j},\underline{k},\dots,l}(t) \underbrace{H_{\underline{j},\underline{k},\dots,l}}_{N} [\underline{\mu} - \underline{m}(t), t] \right\}.$$
(4.24)

Equation (4.24) shows that any probability density function can be expressed directly in terms of the mean, covariance, and quasi-moment functions. Equation (4.24) can be the basis of an approximation for the smoothing density function,  $a(\mu,t)$ . Peller<sup>3</sup> gives the basic development for this, although it is so tedious that parts of it are only outlined in terms of the required procedures. While there are certain advantages to the quasi-moment expansion of Equation (4.24), its use for the smoothing problem is not recommended, and it will receive no further discussion here.

#### 5. CONCLUSIONS

This chapter has developed the exact (in the mean-square sense) differential equations for the smoothing density function and the smoothed expectation of an arbitrary function of the state. These results were good for nonlinear systems and observations corrupted by additive gaussian white noise and were extended to the product noise case. In Section 3, approximate differential equations were presented for minimum variance smoothing, although it was assumed that the plant noise was not a function of the state for this case. One technique of maximum likelihood smoothing was presented and another suggested.

The equations developed in this chapter, particularly the minimum variance equations of Section 3, should be useful whenever probabalistic smoothing of data is desired, such as in estimating missile guidance system accuracy from noisy tracking data. Whereas statistical smoothing of data sometimes requires many passes over the data or many iterations to fit some polynomial to the data in a least squares fit, the technique presented here requires just two passes: one pass forward to get the filtered solution, and a second pass backwards to get the monothed solution. The forward pass should probably use nonlinear filter theory such as that presented by Schwartz' to obtain maximum value from the approach.

There are certain unresolved questions regarding the results presented in this chapter. One, all results have been derived formally: little regard to mathematical rigor has been given. The questions of existence and uniqueness of solutions to the various differential equations have not been treated. In any area dealing "ith

stochastic processes, this could be a serious omission since relatively little is known shout the conditions necessary to guarantee existence and uniqueness of solutions. Another question is the sensitivity of the results to uncertainties in the knowledge of the plant and of the various covariance matrices involved. These types of uncertainties are known to be potentially very serious in filter theory; some of these problems are treated elsewhere in this publication.

### ACKNOWLED GMENT

Nuch of the work contained in this paper was supported by various grants by the United States Air Force and a North American Rockwell fellowship.

#### REFERENCES

1.	Doob, J.L.	Stochastic Processes. Wiley, New York, 1953.
2,	Striebel, C.T.	Partial Differential Equations for the Conditional Distribution of a Markov Process Given Noisy Observations. Journal of Mathematical Analysis and its Applications, Vol.11, No.1-3, July 1965, pp. 151–160.
3.	Peller, John B.	Nonlinear Smoothing Theory and Applications. PhD Thesis, University of California, Lós Angeles, 1968.
4.	Bucy, R.S.	Nonlinear Filtering Theory. Institute of Electrical and Electronic Engineers. Transactions on Automatic Control, Vol. 10, No. 2, April 1965, p. 198.
5,	Bass, R.W. et al.	Optimal Multi-Channel Nonlinear Filtering. Hughes Aircraft Company, Culver City, California, Report SSD 50064R, August 1968.
8.	Schwartz, L.	Approximate Continuous Nonlinear Minimal-Variance Filtering. Hughes Aircraft Company, Culver City, California, Report SSD 60472R, December 1966.
7.	Fisher, James R.	Conditional Probability Density Functions and Optimal Nonlinear Estimation. PhD Thesis, University of California, Los Angeles, 1966.
8.	Rauch, H.E. et al,	On the Maximum Likelihood Estimates for Linear Dynamic Systems. Lockheed Aircraft Corporation, Sunnyvale, California, Technical Report 6-90-83-82, June 1963.
9.	Bryson, A.E. – Frazier, M.	Smoothing for Linear and Nonlinear Dynamic Systems. Published in the Proceedings of the "Optimum System Synthesis Conference", Aeronautical Systems Division, USAF, September 11-13, 1962. (Also Wright-Patterson Air Force Base, Ohio, ASD Technical Documentary Report, ASD-TOR-83-119, February 1963, pp. 354-364.)
0.	Fisher, J.R.	Optimal Nonlinear Filtering. pp.197-300 in "Advances in Control Systems", Vol.5 (edited by C.T.Leondes). Academic Press, New York, 1967.

201

ter paulater

ALCO MONTH

114. A. 17 A.

# CHAPTER 10 - GENERAL QUESTIONS ON KALMAN FILTERING

1.

State State

UNIT

71

# IN NAVIGATION SYSTEMS

#### рХ

.

Larry D. Brock\* and George T. Schmidt\*

 Assistant Design Project Engineer Hamilton Standard Farmington, Connecticut, USA

<sup>†</sup> Principal Engineer Instrumentation Laboratory Massachusetts Institute of Technology Cambridge, Massachusetts, USA

# **PRECEDING PAGE BLANK**

#### CHAPTER 10 - GENERAL QUESTIONS ON KALMAN FILTERING IN NAVIGATION BYSTEMS

#### Larry D. Brock and George T. Schmidt

#### 1. INTRODUCTION

i. I

Navigation systems are among the most popular areas for the application of Kalman filtering. Must of the major navigation system manufacturers have developed or proposed systems with Kalman filtering, and it is being used in several vehicles that are in operational use. Two examples are the Apollo space vehicle and the C-SA transport aircraft. Kalman filtering has now become an expected part of almost every proposed new navigation systems are not hard to find. There are at least three major complementary fautors that have come together at the proper time. These factors are an increased need, the mathematical tools, and the necessary equipment.

The need for effective on-board data processing is great for both space and terrestrial vehicles. The advent of manned travel to the moon, with the requirement for self-contained navigation, required an efficient method for the real-time processing of all navigation measurements. In terrestrial vehicles (both siroraft and marine) the increased need is caused primarily by the introduction of inertial navigation systems that have very good high-frequency remponse and give a continuous indication of position, velocity, and stitude. Ins. are also self-contained and operate without electromagnetic inputs or outputs. However, they do have errors that grow with time. Thus, for most applications, there is a desire to bound the position error growth with external aids. Accuracy requirements dictate that this external data he processed as efficiently as possible.

A mathematical technique was developed at just the right time to fulfill these needs. The Kalman filter theory, published<sup>1,2</sup> in the early 1960's, was recognized as an ideal solution to the mavigation data processing problems for both space<sup>3, 4</sup> and terrestrial vehicles<sup>5-10</sup>. These data processing problems can be fitted very nicely into the necessary Kalman filter assumptions, which means that the estimates of the desired navigation outputs are in fact very nearly optimum. This gives the design engineer confidence that the system is the best that can reasonably be expected, without the concern that some other technique might be found that would significantly improve the system. Also, the recursive form of the filter is convenient. A new optimum estimate can be made very shortly after each new measurement is obtained. Nor is it necessary to store a great mount of data or invert a large matrix, as might be necessary in more conventional, least-squares fitting techniques.

However, as excellent a solution as Kalman filtering is to the data processing problem, it would have been of only academic interest if it were not for the fantastic rate of development in electronic circuits, which makes possible very powerful digital flight computers. These three factors have come together at the proper time to create considerable interest in Kalman filtering in the navigation industry. This interest partially motivated this publication.

In this chapter, some of the initial steps necessary for the application of Kalman filtering will be discussed in general. The application of filtering to terrestrial navigation will then be illustrated by simple examples. Two methods are suggested for simplifying the problem in order that it can be more easily handled on a practical computer. Details are then given of the alignment and calibration of the inertial system in a spacecraft on top of a swaying launch vehicle. Practical implementation problems, as well as actual hardware difficulties, are discussed in detail. It is hoped that from these simple examples the reader will be in a better position to appreciate the applications that follow in the text.

#### 1. INITIAL STEPS IN THE APPLICATION OF KALMAN FILTERING

The Kalman filter itself is defined in very precise mathematical terms, but its application to actual physical systems is rarely a precise science. Considerable engineering experience is needed to properly identify the system to which the filter is to be applied, to adequately model that system, and then to develop a practical program that mechanizes the filter in the on-board computer. The optimization of the filter must include many factors which are difficult or impossible to describe mathematically, such as the trade-off between performance and computer size. The statistical parameters are rarely based on the actual statistics of the physical system, because these statistics are either too complicated or are not well known. The parameters are more likely to be chosen by less formal methods which attempt to maximize the performance in spite of the imperfectly known real world. For example, the covariance matrix in the operational Kalman filter will contain an estimate of the r.m.s. position accuracy obtained by the system, even if the filter parameters must be "adjusted" to obtain this match. The noises associated with parameters that are modeled might be arbitrarily increased to make up for parameters that were too small or not well enough known to be included in the model. **PRECEDING PAGE BLANK** 

205

1. 50

One of the first steps in applying Kalman filtering is to identify the system on which the filter is to be based. The choice of system is not as obvious as it might seem. The most direct approach is to estimate the desired parameters of the vehicle directly. A functional diagram of this approach is shown in Figure 1. The primary system on which the filter is based is the system of equations that describe the motions of the vehicle itself. This direct method is used in space navigation systems. This approach could also be used in terrestrial navigation systems where the Kalman filter uses all measurements, including inertial measurements, to produce estimates of the position and velocity of the vehicle directly. However, for practical reasons which will be discussed later, an indirect method is used where the filter estimates the error in an inortial navigation system. The corrected inertial output is then used to indicate the position and velocity of the vehicle. In this case, the system on which the filter is based is the one which describes the errors in the inertial system. A functional diagram is shown in Figure 2.

The system on which the filter is to be based is specified by a set of state variables that are defined formally by a set of mathematical relations. In practice, for actual physical systems, there is never enough information to satisfy these mathematical relations perfectly. Further, it is never practical to include more than a few of the major variables in the on-board computer. One of the basic design problems is the choice of state variables for the navigation filter and the associated trade-off of performance versus computer requirements.

As a first stop in the choice of state variables, theoretical error analyses and test data can be used to construct a set of state variables which describe the total system much more accurately than the description expected from the on-board filter. For a terrestrial margation system, this more complete description could require as many as fifty to sixty state variables with many more which could be identified but are not worth considering. The performance of filters based on various limited sets of state variables can then be evaluated by using computer simulations based on the more complete model. One is chosen which is shown by the simulations to best satisfy the system requirements. For terrestrial systems, practical filters have consisted of between zeven and twenty state variables. An effective operational test program is the only way to definitely confirm the design of the filter.

- The mathematical relations which give the formal definition of the state variables are:
- (a) The desired cutput must be some function of the state variables.

- (b) The measurements must be a function of the state variables and uncorrelated errors.
- (c) The state variables at one time must be a function of the state variables at a previous time, the controls to the system between the two times, and the uncorrelated (white) noise inputs between the two times.

A DISTRICT

1712-33.00

10 10 10 10 11

Dates a dirity

The first relation can usually be satisfied very easily. The desired output of a navigation system is the position, velocity, and possibly attitude of the vehicle with respect to some navigation coordinate frame. The variables which represent position and velocity can usually be included directly in the state space. The second relation involving the measurements can be more difficult to formulate. The measurements will generally conwist of two parts: a desired part which is some function of the states of the vehicle, such as position, velocity, or acceleration; and an undesired error. An error which can be considered uncorrelated between measurements does not require any additional state variables. However, if there are significant components of the error which are correlated, the model for the error must be added to the state variables.

The most effective overall design of the system can usually be obtained by minimizing the amount of preprocessing of the input data before it is used in the filter. One advantage of Kalman filtering is that measurements can be processed in raw form; this can greatly simplify the sensor sub-systems. For example, if an inertial system is to be aided by LORAN, it is not necessary for the LORAN equipment to solve for a position fix, which has then to be inserted in the filter to update the inertial system. The filter can be constructed to accept time difference measurements directly, thus simplifying the LORAN equipment. Another advantage of processing the raw measurements is that the model for measurement error can be much simpler. If the measurements are processed before being used by the filter, the modifications and cross-correlations of the error statistics caused by the preprocessing must be accounted for by the filter. Again using LORAN as an example, the errors in time differences can probably be adquately modeled as simple errors with no cross-correlations. The errors in latitude-longitude occrdinate would be complicated functions of LORAN station-vehicle geometry and they would be cruss-correlated.

The final step in the forming of the state vector is the addition of any variables necessary to describe the dynamic behavior of the state variables. The total set of state variables is the minimum number of dependent variables in the differential equations that describe the system. The inputs to the system can only be white noises and controls which are deterministic quantities that are known to the filter. Many of the basic sources of error in navigation systems are very highly autocorrelated as, for example, gyro drift rate. To satisfy the requirement for white noise inputs, a shaping filter must be devised which produces the desired correlated error. The input to the shaping filter is white noise, and the dependent variables in the shaping filter are added to the state space. If there are controls into the system which are not known quantities swilable to the filter, they too would have to be modeled as random noise.

Several general comments will now be made about the formulation of the state variables for space and terrestrial navigation aystems. The Kalman filter for a space navigation system is, in many ways, more straightforward than for a terrestrial system. The desired outputs are the position and velocity vectors of the vehicle; this requires six state variables. The measurements usually consist of optical angular measurements between celestial

#### 206

bodies and are functions of vehicle position. The optical measurements are usually relatively far spart, so that most of the error is uncorrelated with the previous error. In many cases the correlated part of the error will be insignificant, and no additional state variable will be required. In other situations, one additional variable might be required. The inertial measurements made in a spacecraft are obtained only during thrusting maneurers. During this time there would not normally be any other source of data on the change in velocity. Thus, the inertial measurement is taken as a deterministic control quantity and is not filtered. The dynamic equations for a space vehicle are also clear cut. If the position and velocity of a space vehicle are known at one time, the position and velocity for any future time can be predicted with great accuracy. The uncertainties due to solar and atmospheric pressures, gravitational field uncertainties, and computing inaccurations are very small unloss the vehicle is in a low orbit. Thus, the dynamic behavior can be described with the six basic variables with possibly a small amount of white noise input to account for all the disturbances. Therefore, for an on-board space pavigator with negligible correlated measurement errors, the entire filter can be based very nicely on six state variables.

The filter for a terrestrial system is not nearly as clear out. The desired outputs are again position and velocity. The measurements can come from inertial aensors, Doppler radar, and a great variety of other aids. The errors in any of these sensors are very highly correlated in time. The inertial instruments will add at least six variables - three attitude errors and three gyre drift rates - and possibly more for errors weld as scale factor and beresight errors. For radio aids, the sample period may be long enough for correlated errors to be neglected. The dynamic equations are the source of considerable difficulty if the Kalman filter is to be used in the direct approach as shown in Figure 1. This direct formulation requires that a statistical dynamic model for the vehicle be included in the state space. While the future state of a spacecraft can be predicted accurately for weeks, the future position of an aircraft can be predicted accurately for only a few seconds. The model used to describe these random motions is difficult to obtain, contains very high frequency components, and can be highly nonlinear. A direct filter for a terrestrial system is, therefore, difficult to formulate and puts great domand on the computer. Thus, although this direct filter approach is more intellectually astisfying, at the present time it is more of academic interest than of practical importance.

Even though this direct approach may not be practical until more powerful computers are available, the formulation of the navigation problem using this method can be very instructive and aid in the basic understanding of both the filtering process and the navigation process. This filter will give the optimum performance of an insertial navigation system even if no external data are available. This means the Kalman filter automatically solves some of the classical problems in inertial navigation. Optimum self-contained "damping" of the 84-minute and 24-hour error oscillations will be obtained, and the system will give the optimum indication of North by continuously "gyrocompassing". These statements are not as important as they might seen. It turns out that, for the levels of random motion that must be assumed for most vehicles, the optimum solution is only very slightly better than the deterministic solution of the navigation problem that has been used all along. The filter can only improve performance by separating the accelerations due to system errors from the actual accelerations of the vehicle. The system errors do have distinctive dynamic characteristics which make detection possible; however, their magnitudes are so much smaller than the accelerations of the vehicle that they are very difficult to separate. If any improvement could be expected at all, it would be for a system in a very stable vehicle, such as a submarine.

This direct filter will also give optimum performance when other sources of navigation information are added to the inertial data, and even when there are only external aids with no inertial data. For example, this filter could be used to combine Doppler, LORAN, and magnetic heading information without using an inertial system. It therefore has the advantage of continuing to operate after a failure in the inertial system.

If the navigation system does include an inertial system and some other source of navigation data, there is a completely different way of formulating the navigation problem which avoids most of the practical problems of the previous method. Instead of estimating the state of the vehicle directly, the filter is used to estimate the error state of an inertial navigation system. The inertial system follows the high-frequency motions of the vehicle very accurately, but has low-frequency errors which grow with time. The dynamic system on which the filter is based is the set of error equations for the inertial system, which are relatively well known, well behaved, low frequency, and essentially linear. The sample period can range from several seconds up to a minute without greatly influencing the sflectiveness of the filter. For these reasons, this method is used for virtually every practical terrestrial Kalman filter mechanization.

The forsulation of a Kalman filter for terrestrial navigation can best be illustrated by simple examples. Therefore, both methods are applied to a simplified navigation problem: the specific filter equations are defined, and simulation results are given. First, at the risk of being repetitious with previous chapters, the basic Kalman filter equations are stated. The equations are stated in the form most applicable to the navigation problem and are a mixture of the discrete form and the continuous form. The discrete form is used when a new measurement is introduced, and the continuous form is used between measurements. The equations are also a mixture of linear and nonlinear equations, the full nonlinear equations being used for the simulation of the system and of the measurement process. Linearization of these equations about the estimated state is used for the covariance matrix equations.

#### 3. SUMMARY OF BASIC FILTER EQUATIONS

It is assumed that the state of the entire system, including sensors, vehicle, and environment, can be described by differential equations of the form

$$(t) = \frac{1}{2} \left[ \frac{1}{2} (t), \frac{1}{2} (t), \frac{1}{2} (t), t \right],$$
 (3.1)

where g are known control inputs and g are white noises. For the assumptions to be made here, it can be shown that the control dous not affect the form of the optimum filter. Thus, the control variables will not be shown explicitly in the following discussion. It is assumed that measurements are made at discrete times according to the relation

$$\underline{\mathbf{n}}(\mathbf{t}_{\mathbf{n}}) = \underline{\mathbf{h}}[\underline{\mathbf{x}}(\mathbf{t}_{\mathbf{n}}), \underline{\mathbf{u}}(\mathbf{t}_{\mathbf{n}})], \qquad (3.2)$$

where  $y(t_m)$  are errors in the measurements that are uncorrelated between measurements. (Time-correlated errors in the measurements would have to be included in the state vector g.) Assuming that the optimal estimates are close enough to the true values for higher-order terms to be neglected, the optimum measurement process is given by Kalman's optimum linear filter. The basic equations are:

At a mensurement time

$$\hat{\underline{\mathbf{x}}} = \hat{\underline{\mathbf{x}}}' + \mathbf{E}' \mathbf{H}^{\mathsf{T}} (\mathbf{H}\mathbf{E}' \mathbf{H}^{\mathsf{T}} + \mathbf{U})^{-1} (\underline{\mathbf{u}} - \underline{\mathbf{h}} (\hat{\underline{\mathbf{x}}}', \mathbf{t}))$$

$$\mathbf{E} = \mathbf{E}' - \mathbf{E}' \mathbf{H}^{\mathsf{T}} (\mathbf{H}\mathbf{E}' \mathbf{H}^{\mathsf{T}} + \mathbf{U})^{-1} \mathbf{H}\mathbf{E}'$$

$$\hat{\underline{\mathbf{x}}} = \underline{\mathbf{x}} (\hat{\underline{\mathbf{x}}}, \mathbf{t})$$

Between measurements

where the prime (') indicates conditions that exist just before the measurement. The covariance matrix of estimation errors E is defined by

where "()" represents the expected or mean value,  $\delta \mathbf{x}$  is the error in the estimate, and

 $\dot{\mathbf{z}} = \mathbf{z}\mathbf{z} + \mathbf{z}\mathbf{z}^{T} + \mathbf{N}$ 

$$\frac{\left|\frac{dG}{2s}\right|^2}{s^2 r} = H$$

$$\frac{\left|\frac{dG}{2s}\right|^2}{s^2 r} = H$$

$$\frac{\left|\frac{1}{2}}{s^2 r} = \frac{1}{s^2 r}$$

$$\frac{\left|\frac{1}{2}}{s^2 r} = \frac{1}{s^2 r}$$

$$\frac{dG}{dG} = 0$$

The matrices R and Q are defined by

The optimum filter for a linearized system in which the higher-order terms can be neglected is shown in diagram form in Figure 3, where, at the sample times, the estimate of the state  $\underline{x}$  is changed by an impulse through the weighting factor,  $K = \Sigma' H^{2} (H\Sigma' H^{2} + U)^{-1}$ .

#### 4. ILLUSTRATIVE EXAMPLES

Both the direct and indirect forms of the Kalman filter are illustrated by applying them to similar simplified mavigation problems. For this example the vehicle is constrained to move only along a meridian on the surface of a spherical non-rotating earth. The vehicle is disturbed by random accelerations with gaussian distribution and an apponential subcorrelation function. Instrial measurements are obtained from an accelerometer that is mounted on a platform that rotates about an axis perpendicular to the meridian plane. The orientation of the ylatform is controlled by a single-degree-of-freedom gyro. The platform can be commanded to rotate at a given rate by a signal sent to the gyro, and it will also rotate due to gyro drift. The gyro drift rate is assumed to be a random-walk function beginning at some initial bias value. The scoelerometer measurements are used by the direct filter to produce statistical estimates of the position and velocity of the vehicle. For the indirect formulation, the accelerometer output is connected directly to the gyro through the proper gain to give a deterministic Schuler-tuned insertial navigator. The measurement in this case is the difference between inertial velocity and velocity from a Doppler radar. The Doppler radar is assumed to have a scale factor error that is also described by a random-walk from an initial bias value. A schematic drawing of the nevisation problem with the direct filter is shown in Figure 4. The diagram for the indirect formulation is basically the same except that Doppler data is also an input into the computer.

(3.3)

بر المنافع المنافعة الم



-----

The steps for applying a statistical filter are: (1) choose the state variables that describe the complete aystem; (2) determine the dynamic equations that describe these state variables; (3) derive the equations that relate the measurements to the state variables; (4) develop the matrices of partial derivatives that are used in the covariance matrix equations; and (5) if it can be assumed that the linear approximations of the errors are sufficient and that all signals and initial conditions have gaussian distributions, apply Equations (3.3). These steps are first applied to the direct form of the filter.

The state variables that describe this simplified navigation problem are vehicle acceleration a , velocity v , latitude L , platform misalignment from vertical  $\alpha$ , and give drift rate d . A diagram of the system is shown in Figure 5. The numbers at the output of the gains K mean discrete changes are made at the corresponding points in the system at the sample times. The equations describing the dynamics of these variables are

 $\dot{a} = -\sigma a + n_{a}$   $\dot{v} = a$   $\dot{L} = v/R$   $\dot{\alpha} = v/R + d + a$   $\dot{d} = n_{d}$ (4.1)

where  $\sigma$  is the inverse correlation time of the vehicle acceleration, R is the radius of the earth, and  $n_{a}$  and  $n_{d}$  are the white noise inputs into the models of vehicle acceleration and gyro drift rate, respectively. The  $\dot{\alpha}$  equation derives from the fact that the platform is rotated by  $\hat{\psi}/R$  in order to keep it aligned with the vertical while the true vertical rotates by  $\psi/R$ . The platform is also torqued by  $\hat{d}$  to compensate for the gyro drift rate d. The known control  $\sigma$  is therefore  $-\hat{\psi}/R - \hat{d}$ . The equations for the estimated state variables are the same as Equations (4.1) with the actual state variables replaced by the estimated variables and the noise terms deleted. Note that because of  $\sigma$ ,  $\hat{\alpha}$  and  $\hat{\alpha}$  are zero.

The measurement of the accelerometer is given by

$$\mathbf{m} = \mathbf{a}\cos\alpha + [\mathbf{g} \sim (\mathbf{v}^2/\mathbf{R})] \mathbf{s} in\alpha + \mathbf{n}_{\mathbf{n}}, \qquad (4.2)$$

where g is the acceleration of gravity and a \_ is uncorrelated noise in the measurement.

The F and H matrices are obtained by taking the partial derivatives of Equations (4.1) and (4.2) with respect to the state variables. The F matrix is given by

	<b>~</b> ~	0	0	0	0
	1	0	0	0	0
<b>F</b> =	0	8-1	0	0	0
	0	K" 1	0	0	1
	lo	0	0	0	٥

and the H matrix by

where A and D are defined by

$$H = \begin{bmatrix} 1 & 0 & 0 \\ g & - (\hat{v}^{/2}/R) \end{bmatrix} 0 \end{bmatrix}.$$

In the equation for H, the angle  $\alpha$  is small such that  $\cos \alpha$  can be assumed 1 and terms involving  $\sin \alpha$  can be neglected as second order. The noise matrices are given by

and

 $\mathbf{U} = \langle \mathbf{n}_{\mathbf{n}}^{\mathbf{f}} \rangle$ 

All terms are now defined so the optimum filter for this problem is completely defined by Equations (3.3).

An example of the response of this meridian system to typical inputs is given in Figures 6-9. The three sources of disturbances in the system are the random-walk gyro drift rate, the disturbance acceleration, and

209

(4.4)

the states

the random measurement error. The random-walk gyre drift increases the r.m.s. drift by 0.05 meru/hr<sup>1/2</sup>. (A meru is approximately 0.015 deg/hr.) The disturbance acceleration has an r.m.s. value of 0.003 ft/sec<sup>2</sup> and a correlation time of 1 min. The random measurement error has an r.m.s. value of 0.001 ft/sec<sup>2</sup>. The sampling period for the measurement is 1 sec. Also shown for the same inputs are the responses of a Schuler-tuned undamped system and a system with lead-lag damping. The accelerations of the vehicle in steady-state motion between maneuvers used in this example are optimistically small oven for a very stable vehicle such as a submarine. As the noise increases, the performance of the optimum system approaches very rapidly the performance of a deterministic Schuler-tuned system. However, this example does show that improvement is at least theoretically possible.

A Kalman filter is again applied to essentially the same navigation problem except that the indirect approach is used. The inertial system is mechanized in a closed Schuler loop and the filter uses Doppler data to estimate inertial errors. The state variables that describe the error equations to which the filter is to be applied are: Doppler scale factor error  $\Delta SF$ , inertial velocity error  $\Delta v_i$ , inertial latitude error  $\Delta I$ , platform misulignment  $\alpha$ , and gyro drift d. The errors are defined as the indicated values minus the true values. A diagram of the system is shown in Figure 10. Written in matrix form, the differential equations for the state variables are

∆sr		0	0	0	0	0	∆sf		n <sub>a</sub>	
Δi,		0	0	0	8	0	Δv <sub>1</sub>		0	
لكلاً.	Ŧ	0	R <sup>-1</sup>	0	0	0	ΔL	+	0	. (4, 5)
ά		0	-B <sup>-1</sup>	0	0	1	α		0	· · ·
ġ		0	0	0	0	0	d	ĺ	n <sub>d</sub>	
	,	-				_			L "-	

The measurement equation is

$$\Delta m = \left[-v_{1} \quad 1 \quad 0 \quad 0\right] \quad \Delta L \\ \Delta u_{i} \\ \Delta L \\ \alpha \\ d \\ d \end{bmatrix} + \epsilon_{u} \quad (4.6)$$

AND DEPARTMENT

The P and H matrices are given in Equations (4.5) and (4.6), respectively. The matrices U and N are defined by

where

# 

The initial estimates for the state variables are assumed to be zero. It is also assumed that there is no initial cross-correlation between state variables. The initial covariance matrix (E) is then a disgonal matrix made up of the expected mean-aquared value of each of the variables themselves, since the initial estimates are mero. With numbers specified for all the constants involved, all the information is now available to define the filter using Equations (3.3).

The results for the simulated operation of this system are given for the following typical values:

initial position error	=	0
initial velocity error	a	0
initial tilt	3	0.2 min
initial gyro drift	2	1.0 meru
initial scale factor error	=	0.3%
random-walk gyro drift (N <sub>d</sub> )	2	(0.1 meru) //hr
random scale factor (N_)	Э	(0.05%) <sup>\$</sup> /hr
unconvelated valuation with	-	6 6 knot

It is assumed that the vehicle accelerates to 500 knots, holds this velocity for one hour, accelerates to 1200 knots, and then stops after 4 hours. Figures 11-13 show the position, velocity, and tilt errors in the inertial system with and without the filter. Figures 14-15 show the estimates of gyro drift and scale factor error that are made by the filter. Figures 10-20 show the expected uncertainty in each of the state variables. These uncertainties are given by the square root of the diagonal elements of the covariance matrix. It can be seen that there is a drop in most of the uncertainties at one hour when the velocity changed. The change in velocity helps the filter distinguish between inertial errors and Depler errors, because the Doppler errors depend on velocity while the inertial errors do not.

By comparing Equations (4.3) and (4.5), the dynamic equations for the indirect filter appear very similar to these for the direct filter. In the first element of the state vector, Doppler scale factor replaces vehicle acceleration. There is one additional term which gives the feedback of misslignment error into velocity error that is due to the closing of the Schuler loop. Although the filters appear similar, they have very different characteristics. In the direct filter, the system is dominated by white noise driving the acceleration of the vehicle. The contribution of the inertial errors to the measurement is so small by comparison as to be very nearly unobservable. Also, the sample rate for the filter must be very high to accurately follow the accelerations of the vehicle. On the other hand, the measurement in the indirect filter is made up entirely of system errors - either inortial or Doppler. Thus the filter can very effectively identify and remove navigation errors and is almost independent of vehicle motions. Also, the sample rate only has to be fast enough to keep up with the slowly changing inortial and Doppler errors. Finally, since the system is corrected after each measurement, the predicted measurement difference is zero and there is no need to extrapolate the error state variables.

#### 5. APPLICATIONS, CONSEQUENCES, AND PROBLEMS OF FILTERING

Bosides the navigation problem, Kalman filtering also greatly aids the initial alignment and calibration of an inortial system. A filter of the same form an that used for navigation can be used, during pre-flight, to process external information in order to estimate the misalignments and miscalibrations of the system. The external information may be just the fact that the aircraft is not moving so that any bias velocity is velocity error. The state variables that are used in the pre-flight normally include platform misalignments, gyro drifts, and any other terms which might be subject to day-to-day variations, such as gyro and accelerometer scale factors. In addition, the calibration program might be run with the platform in several different attitudes to help the filter separate error sources.

The calibration problem and the navigation problem are so similar that in many cases the same computer program can be used. It is necessary to include in the navigation filter most of the major inertial system error sources, such as misalignments and gyro drifts. These parameters are the same ones that must be estimated during the preflight calibration program. Thus, with only quantitative changes in some of the statistical values describing the system, the navigation filter will also handle the calibration problem. For example, a filter which is designed to use Doppler radar measurements can be used for calibration by making the Doppler measurement zero and by greatly reducing the assumed error in the measurement.

In most cases, filtering techniques will both improve the accuracy of the calibration process and shorten the time required. A filter can easily handle the problem of motion during the alignment, such as wind buffeting, and can also be acchanized to account for the changing characteristics of components during warm-up. The same type of filter can be used to transfer alignment from a master system to another inertial system, such as in aircraft carrier operations. Alignment and calibration are considered further in Section 7.

Filtering techniques are also useful for post-flight shalyaes of navigation systems during a test program. In this application, all necessary system data is recorded during flight, along with all available reference dats. The reference data is used in the same way as it is used in flight; but reference data is usually much more accurate and the error model for the navigation system can be much more complete. In post-flight analyses even better use of the information can be made by extending the filter to include optimum smoothing techniques, which means that reference data are used from both sides of the point in question.

One of the most important consequences of the use of statistical filtering for navigation is that it changes -the basic figure of merit for the component parts of the navigation system. The dynamic and statistical nature of the error becomes very important, in addition to its absolute accuracy. What is ultimately important is the total accuracy of the overall system after the measurements have been processed by the filter. Thus, what is important is the ability of the filter to detect and compensate for the errors. For example, if a component has poor day-to-day stability but the error is nearly constant once warmed-up, the total system accuracy might be better than with a component with better absolute accuracy but with less stability during the flight. The parameters that give a measure of this lack of stability during flight are the noise inputs to the error equations. These are given by the N and U matrices in the filter equations (see Equations (3.3)). Thus, one of the important parameters of a component is the size of this noise term, which is a part of the error model for that component. For example, if gyro drift can truly be described by a random-walk model as used in the example in Section 4, then an important figure of merit for a gyro is the increase in mean-squared gyro drift per unit time. The units for this parameter would be (meru)<sup>2</sup>/hour. When a filter is used, this number might be more important than the drift rate itself.

An even more important consideration is the dynamic characteristics of the errors. If a filter is used to combine the information from two systems which have errors with the same dynamic characteristics, then the filter

.

can do little more than produce a statistical average which might reduce the error around 20 to 30%. On the other hand, if the errors have unique dynamic characteristics, then the filter will be able to distinguish the errors and obtain a possible 70 to 80% improvement. As was mentioned in the examples, this result is giveningcous when an inertial system and Doppler radar are combined. Most inertial errors are related to some inertial direction while Doppler errors are related to aircraft axes. Thus, when a turn is made, the filter is, to some degree, able to distinguish and calibrate the errors.

A vory valuable by-product of the use of statistical filtering is that an estimate is available at all times for the accuracy of the navigation system. This feature has at least three possible uses. First, the information could be displayed directly. The navigation system would then give the vehicle operator not only the indicated position and velocity, but the estimated accuracy of these numbers. In some situations this information could be invaluable. Secondly, a record could be kept of the acturl and estimated error in the system, for example, at terminal points. If there were, on an avorage, too much discrepancy between how well the system was doing and how well it thought it was doing, this would indicate that the system was not operating properly. The problem could be due either to some component which was erceeding specifications, or to improper modeling of the system. A third use of this statistical data would be the automatic editing of the input data. The filter alrendy hus an estimate of the expected error in the measurements. Thus, if a measurement is in error by more than three or four standard deviations, it can be automatically rejected and indication can be given to the operator that something might be wrong.

Two of the more important problems in the use of statistical filtering are the requirements for a large flight computer and the need for adequate statistical models of the component parts of the system. The computational problem is due to the necessity of computing, in real time, the optimum statistical weighting factors. The weighting factors involve the integration of a matrix differential equation for the covariance matrix (see Equations (3.3)). This matrix has dimensions  $n \times n$ , where n is the number of variables being estimated by the filter. The amount of computation required is roughly proportional to n<sup>3</sup>. Thus, the more complete the model, the worse the computational problem. For a particular navigation system there will be a trade-off between accuracy and computer capability. For any system there will be a point beyond which a larger computer gives very little improvement. Considerable knowledge of the system is needed to actually determine the accuracy/ computer-size trade-off. This fact leads into the other major problem. Some possible solutions to the computation problem are given in the next section.

The other major problem is the need for statistical models for all the instruments involved in the navigation system. Computer simulations of typical navigation problems using supposedly realistic statistical models produce outstanding performance when Kalman filtering is used. But the results in the real world are not necessarily this good bocause of unsuspected errors with which the filter is not capable of coping. A detailed statistical error model for an inertial system has not really been essential in the past, but now that the model is actually a part of the system, its determination is much more important. To be confident of having a complete model, it is necessary to test the system in an operational environment as near as possible to the one in which the system is actually going to be used. This flight test program is necessary to assure that the statistical filter can handle any peculiarity in the system. It is very helpful in this flight test program to record all information that is necessary to "re-fly" the flight on a ground based computer. With this recording it is possible to change the filter and determine what the results would have been without having to re-fly the miroraft. It is thus possible to optimize the filter with great savings in flight test expenses.

#### 6. METHODS FOR REDUCING COMPUTATION

212

The first method for reducing the computational load is to eliminate elements of the covariance matrix that have a negligible contribution to the operation of the system. The linear filter given by Equations (3.3) gives the "mathematically optimum" filter for the model that was assumed to represent the physical system; i.e. terms are included even if they would reduce the mean-squared uncertainty only in the minth or tenth significant figure. The assumed statistical model, however, represents the actual system to no more than two or three significant figures. The problem is to locate those terms that can be eliminated without significantly degrading the performance.

A method for producing an effective sub-optimum filter, which the authors have used with some success, is to use physical insight into the particular problem to identify the primary flow of information in the optimum filter. The optimum filter is then separated into smaller filters that preserve this primary flow of information while neglecting terms which have negligible effect. The details of the design of the sub-optimum filter are obtained by simulating both the optimum and sub-optimum filters for identical, typical problems. The design of the sub-optimum filter is then adjusted so that its performance matches that of the optimum filter as nearly as possible. This simulation process also indicates the efficiency of the sub-optimum filter. This approach has been relatively successful, particularly for simplifying the computation of the covariance matrix.

The second method for simplifying the optimum filter is to use pre-computed gains. If the filter equations are such that the gains at each measurement are only functions of time and of the a priori assumptions of the statistics of the noises and the initial state, then, by specifying the measurement schedule or rate, the gains may be pre-computed and stored in the vehicle's computer. For a small number of measurements this is a practical solution to the problem involved in implementing an optimum filter. For a large number of measurements, the pre-computed gains are usually smoothly varying with respect to time and may be approximated by suitable curves (straight lines, exponentials, etc.) that give almost identical filter responses as the true gains. Some of









the minor gains may even be noglected completely. This technique will be applied to the pro-launch calibration and alignment of an inertial platform in a spacecraft on top of a swaying launch vehicle and is, in fact, similar to the technique used in the Apollo Guidance Navigation and Control System.

#### 7. PRE-LAUNCH CALIBRATION AND ALIGNMENT

#### 7.1 Physical Basis for Calibration and Alignment

The inertial system to be calibrated and aligned includes gyroscopes and accelerometers. The known gravity acceleration magnitude is used to calibrate the accelerometers; the known inertial rate of rotation of the gravity vector at a point on the surface of the earth is used to calibrate the gyros. In this procedure, the system is approximately aligned to the local vertical coordinates (vertical, south, and enst), then the gyros are used to instrument an inertial coordinate system. The two approximately herizontal accelerometer outputs (south and east) are used by the optimum filter to generate estimates of the system minalignment with respect to the local vertical coordinates in the inertial coordinates instrument of the rotation of the gravity vector with the known rotation rate in the inertial coordinates only a third-order effect on the measured acceleration. Gyro failures can be closely associated with changes in drift due to acceleration of gravity along the input rats, so the pre-launch calibration of a gyro in a vertical position is highly desirable.

Since the estimates of the alignment and drift variables will depend on the measurement by the accelerometers of the rotation of the gravity vector in the inertial coordinates instrumented by the gyros, the major disturbances are the accelerometer quantization and the wind-induced sway of the launch vehicle. The model of the system for the optimum filter must include variables due to the sway. The complete filter must be simulated on a digital computer; it will be linear, so that by specifying the measurement achequile the optimum gains may be pre-computed. The gains will be approximated by functions that will be stored in the flight computer. The method for using this simplified filter in other platform positions, an illustration of a system text program, and practical hardware problems will be presented.

#### 7.2 Models

In this section the models of the launch vehicle and the inertial system will be presented. The launch vehicle bending dynamics in the north-south and east-west directions are approximated by identical second-order systems. The wind causing the vehicle sway is assumed to be exponentially correlated with a correlation time of  $1/\lambda$  sec. For convenience we assume the wind in each horizontal direction is uncorrelated with the other direction, as is indicated in Figure 21. The correlation function of the white noise required to produce the predicted value of the missile mean-squared sway can be found to be<sup>12</sup>:

$$\mathbf{n}\mathbf{w}\delta(\mathbf{t}-\tau) = \langle \mathbf{n}(\mathbf{t})\mathbf{n}(\tau) \rangle = \frac{(\mathbf{p}^2) 4\lambda \zeta \omega_n^2 (\omega_n^2 + 2\lambda \zeta \omega_n + \lambda^2)}{\lambda + 2\zeta \omega_n} \delta(\mathbf{t}-\tau) , \qquad (7.1)$$

where  $(p^2)$  is the predicted mean-squared missile sway, n(t) is the white noise generating the exponentially correlated wind, and  $\omega_n$  and  $\zeta$  are the natural frequency and damping ratio of the second-order approximation to the bending dynamics. The state vector for the sway variables in the south direction is

$$\begin{bmatrix} \dot{\mathbf{p}}_{\mathbf{a}} \\ \dot{\mathbf{v}}_{\mathbf{a}} \end{bmatrix} = \begin{bmatrix} 0 & 1 & 0 \\ 0 & 0 & 1 \\ -\lambda\omega_{\mathbf{n}}^{2} & -\omega_{\mathbf{n}}^{2} - 2\zeta\lambda\omega_{\mathbf{n}} & -2\zeta\omega_{\mathbf{n}} - \lambda \end{bmatrix} \begin{bmatrix} \mathbf{p}_{\mathbf{a}} \\ \mathbf{v}_{\mathbf{a}} \end{bmatrix} + \begin{bmatrix} 0 \\ 0 \\ n(\mathbf{t}) \end{bmatrix}$$
(7.2)

where  $p_s$ ,  $v_s$ ,  $s_s$  are the horizontal displacement, velocity and acceleration in the north-south direction. The model for the sway variables in the east direction  $(p_s, v_s, s_s)$  has the same form. For the computer simulations to follow, they have the values:  $\langle p^2 \rangle = 100 \text{ cm}^2$  east and south,  $\lambda = 0.1 \text{ sec}^{-1}$ ,  $\omega_p = 2.09 \text{ rad/sec}$ , and  $\zeta = 0.1$ .

The platform orientation with respect to the local vertical coordinate system is described by three small angles  $\alpha$ ,  $\beta$ , and  $\gamma$ . If the platform axes (x, y, z) were rotated by  $-\alpha$ ,  $-\beta$ , and  $-\gamma$ , the axes would coincide with the local vertical coordinates as shown in Figure 22. The state vector equation for this sub-state is given by

ſå	- ]		o	0	ŋ,	1	0	α		$t_x = \Omega_y$	
4	3		0	0	n,	0	1	ß		$\mathbf{t_y} + \boldsymbol{\Omega_h}$	
1	7	=	- Ω <sub>1</sub>	- î,	0	0	0	γ	+	$\mathbf{t_g} + \mathbf{d_g}$	, (7.3)
ď	x		0.114	0	0	0	0	dx		0	
Ĺġ	,		٥	٥.	U	0	٥	d,		0	

í

which has been assumed to be, for angle magnitudes of interest, a valid representation of the platform dynamics,  $\Omega_{\rm B}$  and  $\Omega_{\rm V}$  are the horizontal and vertical components of earth rate at the test site;  ${\rm d}_{\rm X}$ ,  ${\rm d}_{\rm Y}$ ,  ${\rm d}_{\rm Z}$  are the constant drifts for the vertical, south, and east gyros; and  ${\rm t}_{\rm X}$ ,  ${\rm t}_{\rm Y}$ ,  ${\rm t}_{\rm Z}$  are the torquing rates (if any) applied to the gyros. It is assumed that the torquing rates, the components of earth rate, and the east gyro drift are known perfectly, so that the vector on the right represents known control g(t) and is independent of the state

Using the small angle approximations, the south and cost accelerometer pulse rate outputs due to platform orientation in the gravity field may be written as

$$\begin{bmatrix} p_{0} \\ p_{0} \\ p_{0} \end{bmatrix} = z \begin{bmatrix} -\gamma \\ \beta \end{bmatrix}, \qquad (7, 4)$$

shere  $po_{g}$  and  $po_{g}$  represent the total pulse counts at some instant of time, and g is the local gravity  $(ca/rec^{2})$ . A 1 cm/sec pulse accolerometer quantization has been assumed.

Accelerometer pulse rates cannot be instantaneously measured, but the total pulses, which make up the output due to sway velocity and orientation in the gravity field, can be counted. The filter must distinguish between the pulse count due to the high frequency sway and the pulse count due to the slow rotation of the gravity vector in the inertial coordinates instrumented by the gyros. Inhorent in these measurements are quantization errors. The measurement vector is

$$\mathbf{a} = \begin{bmatrix} \mathbf{m}_{\mathbf{a}} \\ \mathbf{m}_{\mathbf{a}} \end{bmatrix} = \begin{bmatrix} \mathbf{V}_{\mathbf{a}} \\ \mathbf{v}_{\mathbf{a}} \end{bmatrix} + \begin{bmatrix} \mathbf{p}_{\mathbf{0}} \\ \mathbf{p}_{\mathbf{0}} \\ \mathbf{p}_{\mathbf{0}} \end{bmatrix} + \begin{bmatrix} \mathbf{n}_{\mathbf{a}} \\ \mathbf{n}_{\mathbf{a}} \end{bmatrix}.$$
(7.5)

The term "n\_" represents the quantization error at every sampling of the accelerometer pulse count registers. Although the quantization error is uniformly distributed, the measurement noise is assumed to be gaussian with zero mean. The measurements are made once a second with

$$(n_{i}^{4}) = ra. \qquad (7.6)$$

The state vector is 13-dimensional; the state vector differential equation is

$$dg/dt = F_{\Xi} + c + n(t)$$
. (7.7)

(See Figure 23 for 1, F, g, and g.)

With the derivation of the model for the system, the complete optimum linear filter is defined. The accelerometer pulse count registers will be sampled at constant rates. The estimated state vector is extrapolated between measurements according to

$$\frac{\mathbf{x}'}{\mathbf{x}} = \Phi \mathbf{x} + \mathbf{g}, \qquad (7.8)$$

and the covariance matrix according to

$$\mathbf{S}' = \mathbf{\Phi} \mathbf{E} \mathbf{\Phi}^{\mathsf{T}} + \mathbf{S}, \qquad (7,9)$$

where  $\Phi$  and 8 are pre-computed constant matrices for the time step of one second between measurements. They satisfy the following differential equations:

$$dS_{\rm d}t = FS + SF^{\rm T} + N \quad S(0) = 0 \quad (7.10)$$

and

 $d\Phi/dt = F\Phi$ ,  $\Phi(0) = I$  (7.11)

which may be integrated on a digital computer for a time step between measurements of 1 second. At the time of a measurement,  $\hat{j}$  and E are changed according to Equations (3.3) and 8, U, and N are given in Figure 23.

#### 7.3 Computer Simulations

A complete nonlinear simulation of the inertial platform in a swaying launch vehicle was made on a digital computer in order to simulate real accelerometer outputs. The initial misalignments were 1 degree on all axes; drifts were 10 meru for the vertical and south gyros and zero for the east gyro. (A meru is approximately 0.015 deg/hr.) The r.m.s. sway in each horizontal direction was 10 cm. The initial conditions for the covariance matrix were 1 degree of the alignment angles and 100 meru<sup>2</sup> for the gyro drifts. The sway variable variances were 100 cm<sup>2</sup>, 440 cm<sup>2</sup>/sec<sup>2</sup>, and 1900 cm<sup>2</sup>/sec<sup>4</sup> for position, velocity, and acceleration. All initial cross-correlation terms were assumed zero. The initial estimate of the state was a zero vector.

The response of the filter was excellent for these conditions. A number of runs were first made in which the matrix U was varied so as to cause good agreement between the r.m.s. error as determined by the filter and the actual error. The errors between the estimates and the actual values of szimuth angle. vertical gyro drift and south gyro drift are shown in Figures 24-26. The errors reach small values for the three cases in 15, 40, and 10 minutes respectively. Note that, since the measurements are taken every second, about 2400 samples are

:0) 



Arren Me hange story Bacht

required to estimate the drift of the vertical gyro. Clearly, any significant errors in the model for this system will result in drawtic errors in the vertical drift estimation. The errors in the estimates of the two leveling angles ( $\alpha$  and  $\beta$ ) are negligible after the first few measurements. The estimates of the sway variables are not particularly good, since we are sampling once per second and the sway period is 3 seconds. These simulations were run under assumed perfect knowledge of east gyro drift because of the classical result that east gyro drift cannot be identified from azimuth error. The fact that east gyro drift must be known presents no problem; as seen from Figure 26, the south gyro can be calibrated in about 10 minutes and the error shows little sensitivity to d<sub>g</sub>. This gyro can then be placed east by rotating the platform and a complete alignment and calibration made. (The question of other platform positions is discussed later.)

#### 7.4 Design of the Simplified System

ł

Carlo Carlos

The gains for the optimum filter may be pre-computed for all trials, since the measurement times will be the same and the *a priori* assumption for the statistics of the initial state vector and noises will not change. For the problem at hask, the implementation of the gains into the inertial system involved, first, the design of a simplified optimum filter. The gains for each state variable estimate depend on both accelerometer measurements and, in general, one gain is much smaller than the other and can be neglected. In this problem all cross-coupling measurement gains are neglected, e.g., vertical drift estimation depends primarily on the south accelerometer measurement gain for vertical drift estimation is not implemented. Typically the predominant gains vary as in Figure 27. The pog gain represents the number used to multiply the difference in the measured and predicted output of the south accelerometer and to update pog. The  $\alpha$  gain sultiplies the same difference. These gains can be approximated by exponentials and straight-line segments where, at distinct intervals, the time constants and slopes are changed to continually fit the approximate gains to the true gains. The gains for position, velocity, and acceleration quickly reach steady-state values and may be approximated by three constants.

The response of a simplified filter is shown in Figures 24-26. The process of design enters, since it required a number of runs using different slopes and time constants for the gains to get a good match with the response of the complete filter. In fact, in the end, the same exponential gains were used for both  $po_g$  and  $po_g$ , and the same exponential gains were used for both  $po_g$  and three sway variable gains, two initial conditions for exponentials, and sets of the following five numbers which are changed at ten discrete times: two time constants for exponentials ( $po_g$  and  $\beta$ ) and three slopes for straight-line segments ( $\alpha$ ,  $d_g$ ,  $d_g$ ). Table I summarizes these comments.

#### TABLE I

#### Approximate Gains

Variable	Accelerometer used	Form of approximate gain	
po <sub>s</sub>	south	exponential segments	
po.	east	same value as pog gain	
v,	south	constant	
v,	east	same as v <sub>a</sub>	
<b>A</b> _	south	constant	
P_	south	constant	
<b>1</b>	east	same value as a <sub>g</sub>	
P.	enst	same value as p	
α	south.	straight-line segments	
β	esst .	exponential segments	
γ	south	-β sain	
. d, *	south	straight-line segments	
dy	east	straight-line segments	

#### 7.5 Implementation

A slight variation of this simplified filter was implemented in the Apollo Guidance Navigation and Control System. Part of the program was concerned with initialization for platform positions other than the one considered here. The optimum filter, once implemented, does not change for other platform positions; the measurements that the filter gets are made to simulate the standard platform configuration. For example, if the platform area were vertical, north, and east, and if the sign of the north accelerometer output were changed to simulate a south accelerometer, the filter output for the variables in the south direction need only be interpreted as negative of their true values.















.

.....

her an an tem an and the second and be being





For some applications, it may be desirable to torque the mouth gyro at negative horizontal earth rate to keep the east axis level. The form of the filter and the filter gains do not change because perfect torquing is assumed; just add negative horizontal earth rate to the extrapolation of the angle  $\beta^3$ . It has also been found convenient to use a simple first-order extrapolation of the alignment angles  $(\alpha = \alpha + \dot{\alpha} dt)$ , etc.). The sway variables are extrapolated according to a sway transition matrix whose elements can be changed to compensate for variations in parameters between different launch vehicles.

Once the optimum filter has been implemented according to the simple muthod outlined, it can be readily adapted to various problems of alignment and calibration. For example, consider the following system test procedure in which the platform exem are identified as x, y, z:

- (1) run a 10-min test with x up, y south, z east to determine y gyro bias drift;
- (2) at 10 min read out y drift and use the angle estimates to align the platform to the local vertical coordinates; continue to hold that orientation by torquing the gyros at negative earth rate for 90 sec while counting pulses from the x accelerometer;
- (3) orient the platform to x down, y east, z south and run a 10-min test to determine z bias drift;
- (4) use the angle estimates at 10 min to align the platform and then torque for 90 sec at negative earth rate while counting x accelerometer pulses;
- (5) torque the south gyro with the negative of the horizontal component of earth rate for 45 min while determining vertical drift (x gyro). The y gyro bias, as determined in step 1, is d<sub>g</sub> for this step.

This procedure takes about 68 minutes, after which enough information is available to determine y gyro bias drift, a gyro bias, the sum of z gyro bias and acceleration-mensitive drift, and x accelerometer bias and scale factor. One can readily imagine how an automated system test procedure can be set up to completely calibrate the system in the swaying spacecraft. The last step in the program would be an alignment run to ready the system for launch. Through all of this, the basic simplified optimum filter does not change.

#### 7.6 Hardward Problems

ľ

2

The primary source of azimuth error is the uncertainty in calibration of the east gyro drift; the primary source of vortical drift estimation error is due to variations in the east gyro drift during a test. If the east gyro has a large drift due to acceleration along its input axis, then it is desirable to keep the input axis almost horizontal to minimize the variation in east gyro drift by torquing the mouth gyro a negative horimontal earth rate. Unfortunately, if the bias of the south gyro changes when it is torqued, then the mouth gyro calibration will yield two answers corresponding to the torqued and untorqued cases, respectively. (Torquing was done in the system test program.) Another problem could occur if the accelerometer biase change during the calibration. Figure 28 shows the effect of an exponentially changing east accelerometer bias on mouth drift estimation.

Beertow 4

24 D #5318248

Another possible problem area appears when using pulsed integrating accelerometers. If either horizontal accelerometer has a large dead-zone for near zero inputs, large transients in the filter output will repear. The form of the transients will vary depending on the time during the test that the accelerometer goes through the dead-zone. Vertical drift estimation is particularly sensitive to a dead-zone in the south accelerometer. In some cases a transient on the order of 800 meru has been observed; the filter never reached the correct value of drift at the end of 45 minutes because this vertical drift gain is small at the end of the test. As a practical solution to the problem, the platform is deliberately offset (between steps 4 and 5 in the system test program) from the vertical before beginning a vertical drift test so that the accelerometer never goes through null.

These problems are indicative of the strange results that can occur when the model for the system is incorrect. Philosophically, we have designed a total system test and as such it should indicate, in some manner, out-ofspecification conditions which would then require lower level testing. The model must therefore include all in-specification conditions and the designer's experience must be used to recognize out-of-specification situations.

#### 7.7 Laboratory Test Results

A significant portion of the filter design was done in the laboratory, since digital simulation can never equal actual hardware testing in pointing out the important problem areas. As part of the filter implementation verification, the Apollo inertial platform was suspended by a cable from the laboratory ceiling. The platform was pushed to simulate spaceoraft motion on the launch pad. Comparisons with independent measurement techniques verified the program accuracy to within the limits of the gyro performance.

#### 8. CONCLUSIONS

Navigation systems are one of the most important applications of Kalman filtering. Conversely, Kalman filtering is one of the most important new developments in navigation systems. It is likely that most future navigation systems will involve Kalman filtering in some form. However, considerable engineering experience is required



to develop an effective filter, because the physical systems are never precisely defined, the necessary statistical models are not well known, and the computer is never large enough. It is hoped the discussions in this chapter can be of some help in obtaining the understanding necessary to develop efficient navigation filters.

In summary, we might suggest eight steps for the successful implementation of a filter, such as was illustrated here:

- 1. Develop models.
- 2. Digitally simulate model and filter equations.
- 3. Domign simplified filter:
  - (a) restrict model,
  - (b) decouple variables,
  - (c) approximate gains.
- 4. Digital simulation of system and simplified filter.
- 5. Implementation into guidance computer,
- 6. Checkout.

- 7. Test experience.
- 8. Recycle to step 1.

The requirement to simulate the complete system with the filter equations has been inserted to demonstrate the filter's ability to estimate the quantities of interest. It has been all-too-common practice to simply run the filter equations and look only at the propagation of the covariance matrix. Finally, as practical experience indicates modeling errors, the sources must be found and practical solutions implemented.

#### REFERENCES.

1.	Kalman, R.E.	A New Approach to Linear Filtering and Prediction Problems. Journal of Basic Engineering, Transactions of the American Society of Mechanical Engineers, Vol.82, March 1960, pp.35-45.	
2.	Kalman, R.E. Bucy, R.S.	New Results in Linear Filtering and Prediction Theory. Journal of Basic Engineer- ing, Transactions of the American Society of Mechanical Engineers, Vol.83, March 1961, pp.95-108.	
3.	McLean, J.D. et al.	Optimal Filtering and Linear Prediction Applied to a Space Navigation System for the Circumlunar Mission. NASA TN D-1208, March 1962.	
4.	Battin, R.H.	A Statistical Optimizing Navigation Procedure for Space Flight. American Rocket Bociety Journal, Vol.32, November 1982, pp. 1681-1698.	
5.	Åström, K.J.	Some Problems of Optimul Control in Inertial Guidance. IBM Research Laboratory, Sun Jose, California, Research Paper RJ-229, November 1962.	
6.	Åström, K.J.	Some Problems of Indicating the Vertical in a Mowing Vehicle. Fom 2 Rapport A 2308-233, Försvarets - Forskningsanstalt, Avdelning 2, Stockholm 80, Sweden, October 1964.	
7,	Brown, R.G. Priest, D.T.	Optimization of a Hybrid Inertial Solar-Tracker Navigation System. Institute of Electrical and Electronic Engineers International Convention Record, Vol. 12, Part 7, 1964, pp. 121-135.	
8.	Fagen, S.L.	A Unified Approach to the Error Analysis of Augmented Dynamically Exact Inertial Navigation Systems. Institute of Electrical and Electronic Engineers Transactions on Aerospace and Navigation Electronics, Vol.ANE-11, No.4, December 1964, pp.234-248.	
9.	Bona, B.E. Smay, R.J.	Optimum Reset of Ship's Inertial Navigation System. Institute of Electrical and Electronic Engineers Transactions on Aerospace and Electronics Systems, Vol.AES-2, No.4, July 1968, pp.409-414.	
10.	Bona, B.S. Hutchinson, C.E.	Optimum Reset of an Inertial Navigator from Satellite Observations. Proceedings of the National Electronics Conference, Chicago, Illinois, October 1985.	h trange an en from





11. Schmidt, G.T.	The Application of Statistical Estimation Techniques to Inertial Component Calibration. AGARD-NATO Symposium on Inertial Nuvigation Components, Braunschweig Germany, May 1968.
12. James, H.M. et ml.	Theory of Servomechanisms. Dover Publications, New York, 1965, p.369.
13. Narkey, W. Hovorka, J.	The Mechanics of Inertial Position and Heading Indication. John Wiley, New York, 1961.
14. Schmidt, G.T. Brock, L.D.	Statistical Estimation in Inertial Navigation Systems. Journal of Spacecraft and Rockets, Vol.5, No.2, February 1968, pp.146-154.
15 Dusek H.	Theory of Error Propagation in Astro-Institut Guidance Systems for Low-Thrust East

Theory of Error Propagation in Astro-Inertial Guidance Systems for Low-Thrust Earth Orbital Missions. Proceedings of the American Rocket Society, Preprint No. 2682-62, ARS, Los Augeles, California, November 1962.



















There is a set of the local set





Ì



it Dinistanter Diff. 1

511

એ 🕒 અને પાસ્ટ નવી

the states

ж<u>ж</u>





Same in the second second second

in the set

Fig. 5 Disgram of simplified navigation system



Fig.6 Error in estimated latitude

12226



Pig.7 Error in estimated velocity



Fig.8 Error in estimated vertical

. C. .

ومناجبة والمعادية معتبي

221

6.00

1. A. A. A. A.



国家がある

C-2010-000

ľ,

mound

**.**.,

Kind Tributhant

inter the state





Fig. 10 Navigation system with indirect filter



ŝ

2000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 -

1078-021X X28

1505000



Fig. 13 Error in estimated vertical



Fig. 14 Actual and estimated gyro drift



Fig. 15 Actual and estimated scale factor

1.50.3 . 24











and burled representation headers that the transmi weather weather stated



Fig.18 R.M.S. vertical uncertainty


Fig. 19 R. M. S. gyro drift uncertainty



Fig. 20 R. M. S. Scale factor uncertainty

STAL PASIDIAN DA 1914 10 4400 K

226

1



になった。この

U VIZENE





Fig. 22 Emell angle definition























\*1\*0 🔘 (\*27

229

.

•

<sup>م</sup>لک مرکز میکند. این میکند میکند میکند این میکند میکند. این میکند میکند میکند میکند میکند میکند میکند میکند م



2 Vetermentel

**KONALISM** 

Aucenter A

Factor And

In State and Sicht

TIME (SECONDS)



# CHAFTER 11 - APPLICATION OF KALMAN FILTERING

1

1

ł

# THEORY TO AUGUENTED INERTIAL

# NAVIGATION SYSTEMS

k

J.B. Hudet Le

# Litton Systems, Inc. Guidance and Control Systems Division 8500 Canoca Avenue, V-ydiand Hills, California, UBA

Reproduced from O besi available copy.

A she have a

1

-

• • 4

Ì

5

Č.

· Car

231

ALL PROPERTY.

ent Second

a Section

#### CHAPTER 11 - APPLICATION OF KALWAN PILTERING Theory to Auguented Intertial Navigation Systems

#### J.R.Huddle

# 1. INTRODUCTION

In the past two decades different navigation denors have been combined with an suertial platform to obtain various multi-sensor systems for performing the navigation function. The main objuctive in integrating sensors into a system has been to increase the navigational accuracy obtainable when information from independent sensors is properly mixed. Another objective has been to increase the probability of success of a given mission that only a schewed through the existence of independent failure modes in a system. These modes are realized by different combinations of sensors, each caushle of providing useful information during a mission. Reduction in total system dost and whight Cau also result in such designs being obtained implicitly through a relaxation in important complexies in design complexity with attendar reductions in sensor size, weight and cost, increased relia. ~ billy and simpler maintenance.

Secure the development of Kalmas filtering theory, it was realized in the anvigation field that integrated apoten operation offered the opportunities already outlined for obtaining improved pavigation system characteristic. Nowever, the analysis and optimization of the error behavior in these systems is a complex task, since the media describing these errors are stochastic, time-variant and have high dimension. The designs for the control of error in multi-sensor systems were obtained mainly by using the classical frequency domain and root leasn techniques moordinated with trial and error design experimentation on analog or digital computers. The advect of the Kalman theory and the availability of more sophisticated digital computing inclusion destructs for investigating complex system error behavior has revolutionized the design of the error control mechanism for maky multi-sensor mavisation system or polications. The imputes for this change results from the existence of the least sensor of complex system can be examined, as provided through the use of high speed, large capacity digital computers. The significance of these points for practical design problems will become more evident later.

This chapter begins by describing the navigation function and the requirements which are caually imposed on mavigation systems. The major classifications within which most savigation sensors can be grouped are defined. To illustrate how these different navigation sensors have been employed in conventional designs, the theory of operation of three fundamental augmented inertial navigation systems is presented. With this discussion, a perspective is obtained for considering a Kalman modal.ization for these systems is a later section.

Since the computations' allocation for error control is an actual system is limited, constraints must be imposed on the complexity and configuration of any Kalman design proposed. Some of the practical problems affecting the configuration of the error controller in actual systems are discussed, and the basic operations which are implemented in multi-mensor navigation system applications are defined. The question of the complexity of the falmes design is also considered, and a procedure which has been used in error controller withes is described. Take design approach is then used to obtain a Kalman error control mechanization which is autished for applying each of the three summented inertial navigation system previously examined to a particular system mission.

The chapter concludes by illustrating the system performance for this mission, an obvined with the derived Enlane mochanization, and discussing the unique features of the three mavigation systems.

# 8. NAVEGATION AND T.E BENSORS EMPLOYED IN NAVIGATION SYSTEMS

. . .

#### 1.1 General

1.

 $\geq$ 

The function of a navigation system is to determine the position of a vehicle such that it can be direct : to other positions defined in the selected frame of reference. Terrestrial navigation is usually performed with respect to an orthogonal coordinate system affixed to the earth which has one axis coincident with the terrestrial pole and two axes in the equatorial plane. Vehicle position in this frame of reference is denoted in terms of the two argular mensures of latitude and longitude and the linear measure of altitude. As seen in Figure 1, longitude is measured ensured or westward about the polar axes from the zero longitudinal perician which passes through Greenwich, England. Latitude  $\hat{\varphi}$  is measured north or mouth from the equatorial plane about an axis orthogonal to the meridical plane is which the vehicle lies. Altitude h is measured positively or negatively with respect to see level along the local vertical at the latitude and longitude of the vehicle.

PRECEDING PAGE BLANK

and the states of the states of the

Reproduced from best available copy.

ĩ

١

1

£..

.

4

.

الإيدينان المست

зч L

Is addition to vehicle position, the savigation system is often required to provide velocity of the vehicle with reasest to the earth. The usegonvets of the velocity vector are usually desired along the orthogonal area of a coordinate system located a. the vehicle position. This reference is called the local geographic coordinate aystem and has one axis coincident with the local vertical and two Azes in the l.cal level or horizon plune, one along the local acrth line and the other directed eastward.

Another type of data often required from a mavigation system in vehicle attitude with respect to the local marines place and morth line. Is sophisticate, evidence systems, attitude information is required to accomplish automatic aircraft guidanne and control for wespon delivery, cruise to pre-selected destinations, and landing and take-off. Atsitude information is available shen the mavigation system incorporates a stabilized reference such as ca inertial plauform.

To satisfy some or all of the mavigation system requirements discussed, different types of sensors are employed. Most of these savigation seasors can be placed in one of three date;ories breed on the types of data that they provide. These three classes encompass savigation schoors which provide inertial, spert, or position information in some form. Nome of these mensors (4.g., inertial, LORAN) are capable of achieving some or all of the mavigation system requirements autonomously. Others (e.g., Doppler radar, range measuring equiments) require information from other mavigation symmetry, primarily an inertial platform to ad-emplish the navigation function. Each of the three classifications is now described to introduce unsuing material on the principles of operation of some basic augmented inertial pavigation systems.

# 1.1 Inertial Second

The basic element of an inertial sensor is the platform . The platform consists of a mechanical structure on which the inervial instruments are mounted. Up to three accelerometers are incorporated in the platform for measuring orthogonal specific force magnitudes. Gynamoopen are incorporated for the actitude measurement or stabilization of the nocelerometer triad to a specified or entation.

With knowledge of the local mass-sttraction vector, an inertial sensor can be used to determine the vector asseleration of the carrying-vehicle with respect to the inertial space. When this information is provided to a somputer, subsequent integrations can be performed to yield vehicle velocity and position. In addition to this basis emphility to provide sufficient information for autonomous mavigation, inertial sensors are used as at\*itude references providing the angular oriuntation of the vehicle with respect to the local horizon plane and morth line. This stitude data is used in the operation of other mavigation memors, as will be seen.

# . 1.3 apod Sentors

These seasors can provide speed of the carrying-vehicle along different wirections. The interpretation of this information as a vector, however, requires the use of an attitude reforence system. Two types of speed component ucasuring instruments exist. The first type determines speed with respect to the menuius in which the vehicle travels; as achieved by air-speed indicators or ships' .ous. The second type employs the Doppler frequency shift phenomenoe which is exemplified in Doppler somer or radar. These Doppler Wistruments provide speed with respect to a reflecting surface, usually the worth's surface. The attitude data provided to stabilize this information ons yield valoaity components along the it as a cographic coordinate axes. Bubaequent integration of the velocity component data in a computer can yield the present latitude, longitude and altitude of the vehicle,

#### 2.4 Position or Angular Separa

A number of Linus'fications exist for sensors which provide information useful 'n determining position. One type includes tracking systems, such as astro-trackers and ontical sights, which provide the angular priestation of a point of known inertial or geographic location with respect to a set of coordinate area of known orientation im the vehicle. This type of sensor requires an attitude reference and computer to obtain useful navigation data,

Another bas's classification for position sensors includes those which are besed on timing mechanisms. Included are systems which measure:

(a) Time betweep the transmission and receipt of a pulse of energy arom objects of known location. This is dens with ranging radar, laser range-finders and distance measuring equipments.

- (b) Difference between time of arrival of pulses from different transmitters with known location. This is Come with existing LORAN, and omega and proposed satellite pavigation systems.
- (c) Integrated Doppler shift data to provide the change in range to a reflecting or transmitting farget.

To obtain precision in these systems, bigh frequencies are employed. Multiple frequencies can be used to reacive asbiguities. In systems, these sensors are usually employed with other types of mavigation sensors and a digital computer. With an incrtial sensor, for example, dead-reckoning of vehicle position can be performed between the times at which position information is available, with the advantage that navigation data is available continuously.

Angular information can be provided with radio equipment having directional reduction characteristics. Szangles of such equipment are found in instrument landing systems at airports, which provide glide-slope and

felenter and a state of the second second



1

t

bearing ungle indications to landing sizersft. Another example is provide, by willtary TACAN and divilian canirange stations, which provide indications of vehicle bearing with respect to the respect ve timusmitters. The next domain vehicle beading indicator available is the magnetic company, which provides beading information relative to the magnetic poles of the earth.

## 3. CONVENTIONAL AUGMENTED INEBTIAL NAVIGATION SYNTEMS

#### 2.1 Some Al

This section reviews the manner in which different sensors have been employed in more conventional augmented inertial mavigation systems. The discussion will permit a better understanding of the advantages gained through the use of system error controluers based on kalmap theory, which are discussed later. To provide this introduction, the theory of operation for three augmented inertial navigation systems is discussed. The three mavigation systems that have been switched are commonly referred to as Doppler-inertial, LORM-inertial and matro-urertial.

An inertial sensor has been selected for each of these systems since, in amplications of any composity, the maximation system will include an inertial sensor. This occurs by sume the inertial sensor provides the most dynamic form of information possible through a direct measurement of acceleration and attitude often necessary for the operation of other new, gation sensors. The three navigation systems selected illustrate distinct methods of system operation through the character of the sensor  $rig_{abs}$  time inertial sensor. In each of the system, a sensor is a sensor to be assured to perform the required processing of measurement data to obtain the asvigation variable outputs. Since the inertial sensor plays a basic role in each system, is theory of its autonomous operation will be discussed first.

# S.S. Astananaus Inortial Navigation System Operation

Before an inertial platform can provide outputs useful for asvigation, the accelerometer area must be made emincident with the orientation assumed in the navigation "moputer. In most terrestrial navigation applications a Behmler-tuned mistform is employed. In such an inertial systes, two orthogonal accelerometers are initially sligned and them maintrimed in the local-level plane. A third orthogonal accelerometer, if used, will then be coincident with the local gravity vector. The maintenance of any fixed platform crientation with respect to the earth requires rotation of the platform with respect to inertial space to compensate for the rotation of the earth with respect to inertial space and any relative angular rate of the geographic position of the earthy while. The azimuth of the platform is defined as the angular rotation in the level plane of one of the level accelerometers from the local north line. We will assume that one of the level accelerometers is slowys pointed morthward, resulting in what is known as the morth-slaved platform sechamingtion.

Once the platform is aligned, the accelerometer orientation assumed in the mavigation computer allows the actual specific force measurements to be processed to yield useful mavigation information. For terrestrial mavigation, vehicle velocity and position with respect to the coordinate frame affixed to the earth is required. To derive this information from the accelerometer inputs, acceleration with respect to inertial space in first obtained by subtracting the effect of more local gravity from the specific force measurements. The derivative of vehicls velocity with respect to the earth, taken with respect to the accelerometer coordinate system, is ther obtained by isoving computed Coriolic acceleration components from these incrtici accelerations. The Coriolis acceleration components are dependent on the velocity and rotation rate of the accelerometer triad with respect to the earth and the rotation rate of the earth with respect to incrtisi space. The resulting three derivatives are

4	(v <sub>3e</sub> )	=	$s_{2g} + (\wp + 2h)_{He} V_{Se} - \wp_{Se} V_{He}^{\dagger} - c_{be}$	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1
c đi	[V <sub>hc</sub> ]	3	$\mathbf{s}_{\mathbf{R}\mathbf{p}} + \left( (\omega)_{\mathbf{E}\mathbf{d}} \mathbf{v}_{\mathbf{Z}\mathbf{d}} - (\omega + \mathbf{Z})_{\mathbf{Z}\mathbf{d}} \mathbf{v}_{\mathbf{E}\mathbf{d}} \right) ,$	(3, 1)
- 61	(v <sub>ge</sub> )		$\mathbf{s}_{\mathbf{z}\mathbf{p}} + \left[ (\rho + 2) \right]_{\mathbf{z}\mathbf{e}} \mathbf{v}_{\mathbf{d}\mathbf{e}} - (\mu + 2) \mathbf{u}_{\mathbf{e}} \mathbf{v}_{\mathbf{z}\mathbf{n}} \right] ,$	

where -

(# <sub>E,E,I</sub> )p	are specific force magnitudes measured by the ease, north and vertical excelerometers, respectively;							
( <sup>7</sup> 8,8,5) <sub>0</sub>	are computed components of vehicle velocity with respect to the earth along the local must, morth and vertical geographic axes, respectively;							
('8.8,8)e	are the instantaneous computed components of angular rate of the local geographic sizes with respect to the earth about the east, north and vertical axes, due to vehicle velocity with							

respect to the earth about the east, north and vertical axes, due to vehicle velocity eith respect to the earth;

ݩݯݵݷݞݷݵݥݷݵݷݾݵݯݘݵݷݥݵݥݷݵݷݯݻݷݸݻݷݵݾݻݘݵݵݚݵݑݕݵݷݵݷݵݷݵݷݵݷݵݷݵݷݵ ݽݵݵݿݵݵݷݸݵݛݵݷݵݷݵݷݵݷݵݷݵݵݥݵݵݷݵݵݷݵݵݷݵݵݷݵݵݷݵݵݷݵݵݷݵ

 $F_{\mathbf{W},\mathbf{S}}_{\mathbf{w}}$  are the compute ' components of the angular rate of the earth with respect to invertial space about the local morth and vertical axes (note  $H_{\mathbf{w}} \ge 0$ );

. ..

• ! .

# is the computed computeds of the local plugh-bob gravity vector along the local vertical.

The bracketed terms on the right side of Rematings (3.1) are the Coriolis acceleration components. Integration of three derivatives, which are with respect to the local geographic coordinate system. yields the computed values of vehicle velocity with respect to the earth expressed is local geographic duordinates.

The computed components of relative angular rate of the local geographic coordinates when added ... computed local earth-rate components are employed to rotate the gyros in inertial space to existain the local-level, merth-slaved orientation of the platform. The relative rates are obtained from the computed velocity components as

$$\begin{array}{c} \rho_{\mathbf{E}\mathbf{0}} = -\mathbf{V}_{\mathbf{H}\mathbf{0}}/\mathbf{R}_{\mathbf{H}\mathbf{0}} ,\\ \rho_{\mathbf{H}\mathbf{0}} = \mathbf{V}_{\mathbf{E}\mathbf{0}}/\mathbf{N}_{\mathbf{0}} ,\\ \rho_{\mathbf{E}\mathbf{0}} = \mathbf{v}_{\mathbf{E}\mathbf{0}}/\mathbf{n}_{\mathbf{0}} ,\\ \end{array}$$

$$(3.2)$$

where  $R_{me}$  is the computed meridional radius of the earth and  $N_{e}$  is the computed normal radius of the earth. The computed local earth rate components are obtained by projecting the earth rate about the pole onto the local month and vertical area:

 $\Omega_{\rm He} = \Omega \cos \Phi_{\rm C}$   $\Omega_{\rm He} = \Omega \sin \Phi_{\rm c}$ (3.3)

The emputed angular rate of change of the platform orientation with respect to inertial space is then

$$\begin{aligned} & \overset{\omega}{\mathbf{n}}_{\mathbf{n}} = & \overset{\rho}{\mathbf{n}}_{\mathbf{n}} \\ & \overset{\omega}{\mathbf{n}}_{\mathbf{n}} = & [\rho + i \overrightarrow{\mathbf{1}}_{\mathbf{n}}] \\ & \overset{\omega}{\mathbf{n}}_{\mathbf{n}} = & [\rho + i \overrightarrow{\mathbf{1}}_{\mathbf{n}}] \\ \end{aligned}$$

The actual angular rate of rotation of the platform with respect to inertial space differs from Equations (3.4) by a stathestic function due to errors in the platform gyros:

$$\begin{array}{c} \mathbf{a}_{\mathbf{2}\mathbf{p}} & \mathbf{a}_{\mathbf{1}\mathbf{0}} + \mathbf{b}_{\mathbf{2}} \\ \mathbf{a}_{\mathbf{2}\mathbf{p}} & \mathbf{z} & \mathbf{c}_{\mathbf{1}\mathbf{0}} + \mathbf{b}_{\mathbf{1}} \\ \mathbf{a}_{\mathbf{2}\mathbf{p}} & \mathbf{z} & \mathbf{c}_{\mathbf{1}\mathbf{0}} + \mathbf{b}_{\mathbf{2}} \end{array}$$

$$(3.5)$$

there.

Constant is an analysis of the ACC second

in the

**M**E. H. E) -

· . . . . .

als in the computed angular rate of change of the platform with respect to inertial space expressed about the local geographic axes;

 $(\omega_{0,N_1,U})_p$  is the sciual angular rate of change of the platform with respect to inertial space expressed about the local geographic axes;

(bg, H, g) are stochastic functions representing generalized gyro drift rate with respect to inertial space anyreased about the lucal generalizes.

The final step in determining the navigation variables is the computation of vehicle position in the earth describinate frame. Latitude and longitude are computed as integrals of the local relative angular rate vector sompoments about the local east sals and terrestrial pole, respectively. Altitude is obtained as the integral of vehicle velocity with respect to the earth along the local vertical axis:



ына с Анер, с на (**9.0)** 

where  $\Phi_{\rm d}$  is the computed vehicle latitude,  $\lambda_{\rm d}$  is the computed vehicle longitude and  $h_{\rm d}$  is the computed vehicle altitude.

A block diagram illustrating schematically the sechesization of a Schuler-tuned, north-slaved, free-inertial mavigation system defined by Equations (3,1)-(3,6) is shown in Figure 2.

<u>`````</u>



The advantages of caploying an inertial sensor derive from the availability of vehicle attitude information from the platform ginbals and dynamically-exact changes in vehicle vehicle vehicle, of the arcelerometry assurements. These types of information are usually necessary for the operation of other axionics subsystems in the vehicle. The mein drawback is employing an inertial system for autonomous navisation is that the errors is platform attitude and employed vehicle position and velocity have excillatory behavior. Stochastic solar assurements, therefore associated divergence in these errors as the operating time increases. To retain the valuable features of the inertial system operation maying the divergent and callatory error characteristic, other sensors have been input on avoid in mavigation systems to obtain bounde on the inertial system errors. These such as an input sensors measure accessed of position, the rate of crease of position or magular erientation. Since the error is this measured variable is generally bounded, at least a bound is obtained for the relevant inertial system error in the integrated rivigation system.

An appreciation of the complexity of the augmented inertial asvigation system design problem is gained "brough an examination of the equations describing the errors in the isertial system outputs. A rescription which is adequate for most applications is obtained with a set of nine time-variant, linear differential equations with stochastic amputs. These equations can be obtained by differencing the sets of Equations (3.1), (J.5) and (3.6) from these mame equations expressed in terms of only true values of "he variables involved, by then introducing the effects of errors in the accelerome.vrs and gyr(s, and them by ignoring second-order term) in the error variables.

The variables used to represent error in the inertial mavigation system are defined as follows, where the true value of the variables is question are double subscripted:

 $\begin{aligned} \delta \lambda_{\mathbf{a}} & \stackrel{\Delta}{\hookrightarrow} & [\lambda_{\mathbf{a}} - \lambda] \\ \delta \Phi_{\mathbf{a}} & \stackrel{\Delta}{\oplus} & [\Phi_{\mathbf{a}} - \Phi] \\ \delta \mathbf{b}_{\mathbf{a}} & \stackrel{\Delta}{\oplus} & [h_{\mathbf{a}} - \mathbf{b}] \end{aligned}$ 

are the errors in the computed vehicle longitude, intitude and mititude, respectively;

11.90

in the base of the second second second

....

1997 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 -1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 -1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 -

Ň

are the errors in the computed vehicle east, north and vertical velocity, respectively. See is the error in computed apparent gravity magnitude.

From these definition the following errors in the computer rate of refacion of the local seographic axes with respect to the wards and computed earth rate about the morth and vertical airs are derived to first order:

$$\delta \rho_{Be} \stackrel{\Delta}{=} \left[ \rho_{Be} - \rho_{B} \right] = \frac{-\delta V_{Be}}{R_{B}}$$

$$\delta \rho_{Be} \stackrel{\Delta}{=} \left[ \rho_{Be} - \rho_{B} \right] = \frac{\delta V_{EE}}{N}$$

$$\delta \rho_{Be} \stackrel{\Delta}{=} \left[ \rho_{2e} - \rho_{B} \right] = \left[ \delta \rho_{be} \tan \theta + \rho_{B} \sec^{2} \phi \delta \phi_{e} \right]$$

$$\delta \Omega_{Be} \stackrel{\Delta}{=} \left[ \Omega_{be} - \Omega_{b} \right] = -\Omega_{b} \delta \phi_{e}$$

$$\delta \Omega_{be} \stackrel{\Delta}{=} \left[ \Omega_{be} - \Omega_{b} \right] = \Omega_{b} \delta \phi_{e} .$$

The first set of differential equations given below describes the angular deviation of the andelerometer coordinate atom from the local geographic coordinate axes. This platform attitude error is obtained by integrating the difference in the instantaneous rate of rotation of the actual platform and local geographic moordinate axes with respect to incrtial space. Since it is nonvenient to perform this integration in the "eval geographic frame, these errors are defined by the three differential equations:

a strate of any and a second a

$$\frac{d}{dt} \left[ \phi_{\mathbf{g}} \right] = \left( \omega_{\mathbf{g}_{\mathbf{G}}} - \omega_{\mathbf{g}} \right) + b_{\mathbf{g}} + \left( \phi_{\mathbf{g}} \omega_{\mathbf{g}} - \phi_{\mathbf{g}} \omega_{\mathbf{g}} \right)$$

$$\frac{d}{dt} \left[ \phi_{\mathbf{g}} \right] = \left( \omega_{\mathbf{g}_{\mathbf{G}}} - \omega_{\mathbf{g}} \right) + b_{\mathbf{g}} + \left( \phi_{\mathbf{g}} \omega_{\mathbf{g}} - \phi_{\mathbf{g}} \omega_{\mathbf{g}} \right)$$

$$\frac{d}{dt} \left[ \phi_{\mathbf{g}} \right] = \left( \omega_{\mathbf{g}_{\mathbf{G}}} - \omega_{\mathbf{g}} \right) + b_{\mathbf{g}} + \left( \phi_{\mathbf{g}} \omega_{\mathbf{g}} - \phi_{\mathbf{g}} \omega_{\mathbf{g}} \right)$$

$$(3.7)$$

Reproduced from best available copy. 1.89

Any we want to be and the second

S 14.54

4-----

The second set of differential equations describes the propagation of error in the computed values of east, north and vertical velocity:

$$\frac{d}{dt} \left[ \delta V_{g_{c}} \right] = \left[ \Phi_{g} B_{N} - \Phi_{N} B_{g} \right] + \nabla_{g_{c}} - \left[ \delta \rho + 2 \delta \Omega \right]_{N_{0}} V_{g_{c}} - \left[ \rho + 2 \Omega \right]_{N} \delta V_{g_{0}} + \\
+ \left[ \delta \rho + 2 \delta \Omega \right]_{g_{0}} V_{N} + \left[ \rho + 2 \Omega \right]_{g} \delta V_{N_{0}} , \\
\frac{d}{dt} \left[ \delta V_{N_{0}} \right] = \left[ \Phi_{g} B_{g_{c}} - \Phi_{g} B_{g_{c}} \right] + \nabla_{N} - \left[ \delta \rho + 2 \delta \Omega \right]_{g_{0}} V_{g_{c}} - \left[ \rho + 2 \Omega \right]_{g} \delta V_{g_{0}} + \\
+ \delta \rho_{g_{0}} V_{g_{c}} + \rho_{g} \delta V_{g_{0}} ,$$

$$\frac{d}{dt} \left[ \delta V_{g_{0}} \right] = \left[ \Phi_{N} B_{g_{c}} - \Phi_{g} B_{N} \right] + \nabla_{g_{c}} - \delta \rho_{g_{0}} V_{N_{0}} - \rho_{g} \delta V_{N_{0}} - \delta g_{g_{0}} + \\
+ \left[ \delta \rho + 2 \delta \Omega \right]_{N_{0}} V_{g_{c}} + \left[ \rho + 2 \delta \Omega \right]_{N_{0}} V_{g_{0}} ,$$
(3.8)

where  $\nabla_{\mathbf{x},\mathbf{N},\mathbf{x}}$  are stochastic functions representing generalized measurement errors in the east, north and vertical accelerometers.

The final set of differential equations describes the propagation of error in the computed values of vehicle longitude, latitude and altitude:

$$\frac{d}{dt} \left[\delta \lambda_{0}\right] = \left[\delta \rho_{H_{0}} + \rho_{H} \tan \Phi \, \delta \Phi_{0}\right] \text{ sec } \Phi$$

$$\frac{d}{dt} \left[\delta \Phi_{0}\right] = -\delta \rho_{H_{0}} \qquad (3.9)$$

$$\frac{d}{dt} \left[\delta h_{0}\right] = \delta v_{H_{0}}$$

A block disgram illustrating the linear error model of the level channels of a Schuler-tuned, north-sizved, free-inertial mavigation system defined by Equations (3.7)-(3.9) is shown in Figure 3.

Due to the stochastic, time-variant, multi-variable form of these error equations, the application of classical frequency domain and root-locus technique is not effective for error controller design unless assumptions can be made which simplify the equations. In most applications, much assumptions result in constraints on the manner in which the system is operated. For example, platform alignment mechanimations have been designed with classical approaches, assumptions institute invariants and minimal coupling between error variables. Such assumptions are only valid if the vehicle undergoes no acceleration, which donstraints system operation significantly. Since Kalman theory accommodates the time-variant claracter of mavigation system behavior, only operational conditions which require a non-linear description of error behavior need be avoided when this approach is employed. Most design efforts using Kalman concepts for error control deal with the minimization of mechanization complexity that can be obtained for the given mystem performance requirements. Digital computers are usually required to deal with the synthesis problem effectively.

The introduction of Kalman theory has helped to change the viewpoint of analysts in augmented inertial navigation system design from one in which various other sensors simply aid the inertial sensor in performing the navigation function to one where each sensor is regarded as a source of information available to the navigation computer to be used in doriving values of the reduired navigation variables. With this new viewpoint, the system outputs at any instant of time, regardless of vehicle flight dynamics or intermittent availability of data from any sonsor, can be optimal in the sense that all information has been fully utilized. This complete absorption of sensor data is realized through the use of correlation information contained in the system error covariance matrix. The use of this matrix as a basis for measurement data mixing in Kalman filtering constitutes the unique and novel aspect of this design approach.

When Kalman theory is applied to terrestrial navigation systems, the covariance matrix is propagated in realtime in the system computer, due to the unpredictable nature of vehicle maneuvers and navigation sensor data availability. As a consequence, the instantaneous statistical state of the system, reflected by the covariance matrix, is always available to enable optimal utilization of new sensor measurements. In following sections we will describe the manner in which Kalman filter concepts are employed in error controller design for sugmented inertial navigation systems. For the present we describe the manner in which different navigation sensors have been used to control inertial navigation system errors with more conventional system mechanizations.

#### 3.3 Doppler-Inertial Navigation System Operation

A Poly in the second

In a Doppler-inertial navigation system, the sensor which augments the inertial sensor is the Doppler radar. The Doppler radar measures the frequency shift between a transmitted and reflected radio wave. This Doppler shift is proportional to the speed of the vehicle with respect to the reflecting surface along the direction in which the beam of energy is transmitted. Doppler radars have been developed which can provide three independent components of vehicle speed. These components are, however, expressed in a coordinate system related to the











Doppler antenna, whose orientation is not known with respect to the earth. Consequently, a Doppler radar cannot perform the navigation function autonomously, but requires that the Doppler antenna be stabilized mechanically or computationally using attitude information provided by an inertial platform or some other orientation reference. Once the stabilization of the Doppler speed measurements is achieved, which evolocity with respect to earth along the local east, north, and vertical axes is known, allowing changes in vehicle position with respect to the earth to be obtained from integration in the system computer.

The errors in the determination of vehicle speed in antenna coordinates from the Doppler frequency shift are due to various equipment calibration errors (s.g., antenna tomperature shifts, boam mimalignments, etc.) and the statistical character of the beam reflecting surface over which the vehicle travels. Since the beam has finite width, for a given vehicle speed a continuous spectrum of Doppler frequency shift information is extracted from the reflected beam. The centroid of this return is usually tracked and a scale assigned based on nominal operating conditions. If the character of the reflecting terrain changes markedly, the essigned scale factor will be in error. Most of this type of difficulty in using Doppler radar occurs over water where the surface smoothness distorts the return spectrum. Such problems have been solved to some extent by beam lobing. Over very smooth surfaces, however, the Doppler radar output cannot be used, as insufficient return energy is received. Autonomous navigation with Doppler radar is also degraded in over-water operation when surface ourrents exist. Without an independent reference, navigation with respect to the moving surface rather than the earth taken place. The important characteristic of the Dopplor errors is that they can be expressed as bounded fixed fractions of the vehicle speed components. Consequently, the errors in the Doppler speed components, expressed along the platform coordinate axes after data stabilization, are bounded. Proper blending of these data with the velocity components obtained from the inertial system will result in a system which has error components along these axes which are less than the bounds.

In a conventional coupling of the Doppler and inertial data<sup>11</sup>, the velocity components obtained from each sensor, or other suitable independent functions of these variables, are differenced from each other. For sufficiently small errors, these differences can be expressed as a linear combination of various errors associated with the Doppler and inortial sensors. These signals can then be fed back to correct or companyate the different sources of error in the navigation system. The eventual desired result is the reduction of error in all the navigation system outputs. A block disgram illustrating a conventional ground speed and drift angle damped northslaved inertial navigation system mechanization is shown in Figure 4. It can be meen in this figure that the differences between predicted and Doppler measured ground speed and drift angle are fed back through gains to correct the computed north and east velocities and after the computed platform spatial rate components.

#### 3.4 LOBAN-Institul Navigation System Operation

In a LORAN-inertial, navigation system the data used to sugment the inertial system in performing the navigation function is derived from a LORAN time-difference roceiver. LORAN employed as an autonomous navigation system provides position information only. The position is derived from measurements of the difference between the local time of redeipt of pulses transmitted from stations at known position on the surface of the earth. At least three stations are required if a complete position fix is to be obtained from LORAN. One of the stations, called the master, transmits the first pulse. This pulse when received by the two other stations, which are called slaves, is then retransmitted. Upon receiving the master pulse, the LORAN radio receiver in the vehiclo starts a precise clock which measures the times until the pulses are received from the slave stations. The constant difference of the distances from two fixed points defines a hyperboloid in space with the station positions as the foci. Consequently, the time-difference measurement obtained from a master-slave station pair will locate the vehicle on such a surface. Assuming the vehicle is near the surface of the earth, a line-of-position is defined. When a second time-difference measurement creates a line-of-position which intersects with the other, the position of the vehicle is defined. The reason that time-difference is measured in LORAN, as opposed to the time it takes a pulse to travel from a station to the vehicle, is that there is then no requirement for precise synchronization of the vehicle and station clocks with an absolute time standard. If precise synchronization were achieved, the derivation of range to two stations, using an absolute time standard and a time-of-receipt receiver on the vehicle, would suffice to obtain a complete position fix on the earth's surface.

Errors exist in the determination of true time-difference at the vehicle for a variety of reasons, including synchronization error at the slave stations in pulse retransmission, ambient receiver and atmospheric noise and local radio interference mear the vehicle. Another important source of error is that associated with the lack of procise knowledge of the speed of propagation of electromagnetic energy over the earth's surface. This is caused by stochastic variations in the propagation path definition. The result is an actual time-difference grid which is distorted relative to that predicted using geodetic data only.

The important characteristic of these errors in the use of LORAN in miding the inertial system in navigation is that they are bounded. Hence the error in position obtained from the time-difference measurements in areas of defined LORAN coverage will serve as an upper bound on the position error for the navigation system. Proper mixing of position information from the LORAN and inertial sensors can result in less error in all system outputs than could be achieved with either sensor operating autonomously.

In a conventional coupling of information from the LORAN and inortial sensors, time-difference predictions derived from the inertial system, or predicted geographic position information derived from the LORAN measurements, are subtracted from the appropriate variable from the other mensor. For sufficiently small errors, this difference can be expressed as a linear combination of errors associated with each of the sensors. Consequently, as in the Doppler-inertial system, the same basis exists for the design of a linear feedback controller







in count in the bas birts in the first

-

and thereares an and the second and the

to reduce the errors in the outputs of the system. This basis has been exploited in the past, where timeinvariant error control mechanism<sup>®</sup> have been designed. These mechanizations are usually far from optimal owing to the commentioned limited applicability of the time-invariance assumption for the inertial error model, as well as to the varying quality of time-difference data as the vehicle moves through a LORAN coverage area. The use of Kalman filtering concepts in the design of such a navigation system offers great benefits, as the timevariant behavior of the error can be accounted for automatically on a real-time basis.

#### 3.5 Astro-Inertial Navigation System Operation

The final navigation system to be discussed is one in which a star-tracker is employed to measure, at the vehicle, the angular orientation of astronomical bodies with known celestial position. Like the Doppler, an astro-tracker cannot be employed autonomously to determine vehicle position. An attitude reference with known orientation with respect to an earth-fixed coordinate system must be provided to allow an interprotation of the aighting angles. Normally, the optical axis of the tracker is oriented by slewing it through bearing and elevation angles, respectively, from an initial coincidence with the local north line. An astro-tracker usually incorporates a detection and mervo-motioniant system which drives the telescope elevation and bearing gimbals until a detected light source resides at the center of the field-of-view. When this is achieved, measurements of the bearing angle from the local north line in the local-level plane, and the elevation angle out of the local-level plane, can be read directly from telescope gimbals.

In an automated tracker system, the optical axis is first directed to a point in the sky where, based on the computed vehicle position and known celestial position of a star, the star would be expected to be found. Because of errors in the computed vehicle position and the attitude reference used, the star will deviate from the center of the field-of-view. The errors in the initial pointing angle can be determined by centering the star and measuring the changes in the telescope's angular orientation. If the angular errors in vehicle position and platform attitude are sufficiently small, the pointing angle errors can be expressed as linear combinations of these errors. For example, elevation and bearing angle errors could be linear functions of the errors in computed vehicle latitude and longitude, platform asimuth error, and the platform tilt errors about the local morth and east axes. These causes of pointing angle error can be expressed about axes of an orthogonal coordinate system which has one axis coincident with the optical axis of the tracker. Since the initial telescope crientation error can be expressed about axes orthogonal to the line-of-sight axis, the angular error component along the telescope axis is unobservable. The errors result from mechanical misalignment of the telescope, electrical bracker system itself are bounded. These errors result from mechanical misalignment of the telescope, electrical examples in the mulling control loops, and environmental offects, such as the index of refraction of the atmosphere.

In a conventional astro-inertial system, where the attitude reference is of high quality, it is usually assumed that no error exists in the platform level. Corrections are then made only to the computed vehicle position and the azimuth orientation of the platform. Usually in a star-tracker system, sequences of stara are shot to determine vehicle position. After each shot, residual error exists about the most recent line-ofsight, unless stars with orthogonal lines-of-sight are available. Conventional mechanizations usually do not depitalize on the relative orientation of the lines-of-sight that exist in a star shot sequence due to the bookkeeping involved. By automatically scoounting for the optimal use of all previous measurement data, Kalman theory offers a much more powerful approach for obtaining error separation for this navigation system.

## 4. PRACTICAL CONSIDERATIONS IN USING KALMAN THEORY IN NAVIGATION SYSTEM DESIGN

#### 4.1 General

The successful application of any method to achieve real-time error control in multi-sensor navigation systems requires an awareness of the constraints imposed by the actual hardware available for the system mechanization. In particular, the sensitivity of system performance, based on theoretical assumptions, to such phenomena as finite measurement rates and computation error resulting from computation equipment and/or computation algorithm characteristics to the determined to establish the efficacy of any error control concept. In the following paragraphs, the locations of the characteristics of navigation computers and sensors on the implementation of an error controlier based on Kalman linear correlation theory are outlined.

#### 4.2 Digital Computation

In the past, conventional augmented inertial navigation systems have been mechanized using analog computing elements. Such an approach is adequate for very simple mechanizations where stringent requirements for computational accuracy do not exist. More recently, the majority of augmented inertial navigation systems have employed various types of digital computing elements (e.g., general purpose and digital differential analyzer) to accomplish the required computations. Some of the advantages encountered in the use of these digital computers are expability for increased computations precision and flexibility in implementing logical functions. The ourrent trend toward increased processing rate and greater storage at reduced dost is an additional attraction. The disadvantages incurred in digital computation stem primarily from the finite rate at which logical and arithmetic operations can be performed. These finite rates generate requirements for the use of computational algorithms to accomplish mathematical operations. Computational algorithms can, however, achieve only imperfect realizations of the desired mathematical operations. The most familiar, and probably most important, of these operations in practice is that of integration. Computally, an error control mechanization for virtually any









A AND A AND





St. martin with

他们的自己的法

augmented inortial navigation system based on Kalman linear correlation theory will require that a considerable amount of precise arithmetic computation and logical switching he implemented. Consequently, the use of general purpose digital computers is required in these applications.

#### 4.3 Advantage of Kalman Mechanizations in Augmented Inertial Navigations Systems

The essential difference between Kalman and conventional navigation system error control methods results from the use of a covariance matrix, with the Kalman technique as a basis for computing the gain to be used for system correction when a measurement is made. The information contained in this matrix provides the means by which superior system performance can be achieved in comparison to that obtained with conventional approaches. Conventional error control generally involves the use of matrices of pre-stored gains, which are determined using a priori ansumptions. The use of the different gain sets is dictated as the real-time operating conditions of the system are ascertained.

As already noted, an error model for a navigation system is represented by sets of time-variant equations. Since there is infinite variety of possible flight and sensor data profiles for a navigating vehicle, the actual realization of these equations can only be made in real-time. Therefore, the nature of navigation system error propagation requires that a considerable number of gain matrices be stored in the digital computer to realize a conventional mechanization which accounts to any extent for the many sequences of system operating conditions possible. Usually this is not done in conventional designs, with the result that restrictions on operational capability of the vehicle are imposed to accommodate the mechanization inndequacies. Such restrictions on the operational capability of a system are extremely undesirable and can be avoided by using more sophisticated Kalman error control techniques.

The Kalman error dontrol mechanization avoids constraints on vehicle operation through real-time propagation and update of the error covariance matrix. The actual realization of arbitrary functions of time (e.g., acceleration and velocity components) present in the differential equations describing system error are used in the error covariance matrix propagation, resulting in a correct representation of the system error state. Also, the arbitrary availability and variation in quality of sensor measurement data during system operation. Real-time availability of sensor data is arbitrary, as the sensors may be intermittently operable. For example, accurate tracking of LORAN time-differences can be interrupted in the presence of high strength interference. Sensor data quality also varies with time as the vehicle moves into regions affecting sensor operation. With LORAN, the signal to noise ratio associated with information sent from a station usually drops as the distance between the vehicle and the transmitter increases. The quality of Doppler data changes with the terrain over which thme vehicle travels, the antenna station, and the vehicle epsed and altitude, all of which vary with time. Astrotracking capability depends on the effective brightness of available stars at the position of the tracker.

## 4.4 Sensor Measurement Processing

Ľ.

Different formulations of the Kalman filtering equations have been obtained. The initial derivation<sup>1</sup> assumed that the observation or measurement process was continuous and that each measurement was contaminated by an additive white noise. In practice neither of these assumptions can actually be realized.

In the first case, continuous measurements cannot be assumed when the equations are implemented on a digital computer, as the computer can achieve only a finite rate of processing and will, additionally, be required to perform functions other than that of error control. Consequently, in applications, only a finite rate of adsorption of mensor data can be realized.

The requirement for additive white noise contamination of the measurements in the initial formulation also presents difficulties. In all cases the assumption that the measurement-contaminating noise has short correlation time with respect to the system time constants cannot be made. A sensor in the process of extracting information in the presence of short-term noise employs some filtering. The filtering mascolated with this detection process will have the effects of correlating and reducing the magnitude of the contaminating noise process and introducing dynamic error in the information being measured. Equipment blages are further examples of long-term errors present in measurements.

The practical aspects resulting from digital implementation and sensor measurement smoothing have usually led to the use of the augmented state formulation of the equations with disorete observations in applications<sup>2</sup>. In this approach, coupling between the individual sensor subsystems is realized only at time instants when the computer processes sensor data to make system corrections. Between these update instants, the covariance matrix describing modeled system errors is propagated, reflecting the non-interaction of errors in the individual sensor subsystems. At the update time, measurements are made and compared with predicted values thus obtaining observed errors which are weighted with the Kalman gain to provide corrections to the modeled system errors.

Since the original derivation of the Kalman filtering equations, an alternate formulation by Bryson and Johansen<sup>2</sup> has been obtained which does not require the assumption that measurements are contaminated by white noise. With this formulation, continuous absorption of sensor data contaminated with correlated noise is theoreticely possible. The dimension of this filter is less than that of the augmented state form by the number of measurements contaminated by first-order correlated noise. The amount of computation required, however, is greater than that of the original formulation for filters of the same dimension. The Bryson-Johansen formulation was extended to the discrete case by Bryson and Herrikson<sup>3</sup>. This formulation is equivalent to the sugmented state formulation with the discrete measurements made at a finite rate. However, in the limit, the Bryson-Henrikson filter converges to the Bryson-Johanson filter, while the augmented state formulation is not defined. With continuous absorption of sensor data, possible for the case where only correlated noise exists in the sensor measurements, the augmented state formulation with discrete observations will be comparatively leas optimal. Results obtained by Bryson, Johanson und Henrikson imply that, as the sensor sumpling rate is increased, purformance achieved with the augmented state formulation will tend to the theoretical optimum obtained by Bryson and Johansen, until computational instability is reached. This result has been demonstrated by Rischel<sup>4</sup>.

If the frequencies associated with the difference of a sensor measurement and the predicted value of that measurement based on system computed navigation variables are long with respect to the measurement sampling frequency, then there will be little benefit in faster measurement sampling. In cruise navigation system applications, the sensor sampling intervals selected are soldon much shorter than the correlation time associated with the highest frequency noise. Shorter sampling periods have little utility except, perhaps, during the time when the inertial platform is being aligned. During this period, error deviations in system position and velocity will occur relatively more rapidly due to the large platform attitude errors. Therefore, a higher rate of system correction would be desirable for obtaining tighter hounds on these errors.

The sugmenting sensors in sugmented inertial navigation systems have time constants associated with their individual information processing procedures. If these time constants are long, significant dynamic error can result in the sensor output measurements. The dynamic inaccuracies are usually due to vehicle maneuvers. In most systems, acceleration information from the inertial platform can be used to compensate the dynamic error associated with the sugmenting sensor. This has been demonstrated in conventional LURAN-inertial systems<sup>9</sup>, where the rate of change of time-difference has been supplied to the LURAN receiver from the inertial systems<sup>9</sup>. Such a mechanization relieves the receiver from the task of tracking dynamic changes in time-difference, at least to the limits of the inertial errors, thereby committing it primarily to the task of separating transmitted information from noise. When this is done, long receiver time constants on the order of many seconds can be retained.

In some cases, the additional design complexity implied by such an approach may not be desired. What is done in these situations is to select time constants which achieve a balance between stochastic and dynamic error in the sensor processing design. In Doppler-inertial navigation systems, the time constant of the frequency shift tracker in the Doppler radar can be short, on the order of 0.1 second. The purpose of employing short-time constants in the detection process is to minimize the dynamic error in the variable being measured. When shorttime constants are used in the sensor filtering networks, a system correction utilizing the Kalman filtering equations at comparable rates may not be achievable due to allocation of computer time to error control. When this occurs, methods must be devised to minimize the information loss resulting from the slow sampling rate.

The most popular technique for minimizing this information lows<sup>10</sup> requires a high rate of comparison of the sensor measurements and the predicted values of these measurements. These differences, taken at the desired sampling rate, can be inserted into a pre-filter for smoothing between system correction instants. The pre-filtering will reduce the random sensor noise in this error difference, thereby improving the relative information content at the system update time. Note that this scheme does not introduce dynamic error in the sensor measurement, as only the error difference information, not the sequence of whole-value measurements, is being filtered. Consequently, the advantage gained in minimization of dynamic error in the measurements through the use of short sensor filtering time constants is rotained, while the information content of the observed error used for system correction is improved.

A first-order representation of this pre-filtering approach is indicated schematically in Figure 5. Here predicted values,  $\hat{m}$ , are differenced with the raw measurements, m. These differences are weighted with a filtering gain, k, and used to update a summing register. At the system update time, the smoothed difference, y, is weighted with the Kalman gain to obtain the system correction. The summing register is usually zeroed whenever its contents are used.

The principal characteristic to investigate before employing such a mechanization, is the degree of distortion imposed on information passing through the pre-filter. Unless the pre-filtering is modeled in the Kalman formulation, the actual error characteristics differ from the modeled error characteristics. In augmented instial navigation system applications, serious distortion will usually only occur during platform alignment modes. For these cases, an analysis of the distortion of error signals due to platform attitude error due to passage through the pre-filter should be made. Workable levals of distortion can be obtained with an appropriate selection of sensor and pre-filter time constants and the system update interval. 46-111-194

In this section we have considered some of the practical problems which arise in actually implementing a given error controller in the navigation momputer. In augmented inertial navigation systems, digital computers are required for the complex arithmetic and logical operations associated with a Kaiman error controller. Realtime propagation of the system error covariance matrix is required to accommodate the real-time dynamic maneuvers of the vehicle and the possible intermittency and variation in quality of sensor data. Finally, the reasons for finite discrete sampling of sensor data for system correction were discussed and a method for minimizing information loss was noted. In the next section, an approach is presented which has been used to alleviate some of the computational burden imposed on the digital computer. This approach deals with the synthesis of the error controller valuer than the technique: described for its implementation.

#### 5. SYNTHESIS FOR LINEAR ERROR CONTROLIERS USING KALMAN THEORY

#### 5.1 General

. J

「「「「「ない」」である

The control of error in any system requires a definition of the sources of error and their dynamic behavior. If the error model for the system is described by linear differential equations with random initial conditions and white-noise driving functions, it is a simple matter to directly apply Kalman's filtering theory for the control of these errors. If the model is non-linear, oither a linearization must be made or a different control algorithm must first be employed until the errors are reduced sufficiently for linear behavior to be realized.

The linear differential equations describing navigation system error are forced by stochastic functions. These stochastic driving functions have various degrees of auto-correlation, ranging from biases to essentially white noise terms. To devise dynamic models for such noise errors, correlation functions are derived from expirical data obtained about the noise sources. Linear dynamic systems excited by white noise are then synthesized from an analysis of these describing correlation functions to represent these error sources in the total system error model. Thus for multi-sensor navigation systems, information which is available about errors usually results in the definition of an extensive linear error model. The application of Kalman theory employing such error models results in significant computer computation time and storage requirements. To reduce these computational requirements, various methods have been tried in order to find simpler error control achemes which still obtain the benef cial correlation property. One of the most successful of these methods<sup>8</sup> involves the study of the complex error model itself to determine what simplifications can be made to reduce the computational burden. In this approach, one employs to advantage the fact that certain error components and dynamic interactions tend to dominate the error benevitor in the system outputs. This approach of simplifying the error model to reduce the computational requirements is important, because it enables the analyst to use his experience directly in determining the error controller design.

#### 5.3 Simplified Linear Error Controller Synthesis and Evaluation

Since the error model plays the central role in the Kalman linear error controller, its simplification yields a natural method of reducing the controller complexity. Generally, the idea is to retain the more dominant state variables and their dynamic interaction to define a simpler error model. To be specific, three steps are involved in this design simplification. The first step is to reduce the dimension of the refined error model by deleting those components of the error state vector which are feit to contribute least to overall error in the usable system outputs. The second step is to eliminate dynamic inter-couplings between these retained state variables which contribute least to the definition of their behavior. These two steps amount to obtaining simplified differential equations describing the more dominant state variables of the original error model. Once these steps are accomplished, refinement of the proposed filter based on the simplified error model is made. These frual step of refinement involves the adjustment of white noise observation or disturbance noise parameters. These parameters are selected so as to compensate to some extent for the simplifications made in the preceding steps.

To determine whether an error model simplification or parameter adjustment is appropriate, the effect on overall system performance must be ascertained. Since the systems considered in navigation applications are highly complex, little progress in making such determinations can be made in a simple way analytically. Consequently, digital computer programs which provide either simulation results or solutions to complex arrays of analytical equations must be employed in studying such systems. For general system studies, Monte Carlo simulation programs have been applied with success. Usually, however, in applying such programs a large number of runs must be made to obtain precise results. Precision is required in this synthesis problem, as the exact statistical effect on system performance due to a design change must be known for design decisions to be made.

For the case in which an adequate representation of the system can be obtained from a linear error model, the following approach<sup>4</sup> has proved useful in determining linear controller designs which are simpler than the optimal. This technique will be discussed for the design of a Kalman error controller based on a simpler error model of the system than the reference error model. The Kalman error controller impacts on the system error propagation at update times, which occur when a reliable comparison of data derived from any two independent sensors can be made and the system computer has sufficient time to perform all computations required for updating. Between the update instants, all system errors propagate freely from the initial values resulting after the most recent system update. Between the system correction times we assume that the actual system error behavior is described by the linear vector differential equation

$$\frac{d}{dt} [x(t)] = A(t)x(t) + \epsilon(t) , \qquad (5.1)$$

apei.e

x(t) is an n-dimensional vector describing all system errors which have initial values described by the covariance matrix:

 $\Sigma_{\alpha} \stackrel{\Delta}{\to} E[x(0) \times x(0)] :$ 

A(t) is an n x n matrix defining the linear dynamics of the errors in the refined or reference error model;



244 •

(t) is a zero-mean, white noise process, where

#### $Q(t)\delta(t - \tau) \stackrel{\Delta}{=} E[\epsilon(t) \times \epsilon(\tau)]$

With this notation, a well-known equation<sup>7</sup> describing the propagation of the covariance matrix corresponding to the system error x(t) is

$$\frac{d}{dt} \left[ \Sigma(t) \right] = \Lambda(t) \Sigma(t) + \Sigma(t) \Lambda^{T}(t) + Q(t) , \qquad (5.2)$$

Whenever an update is to be made, it is assumed that the comparison of data from different sensors results in a linear combination of the errors x(t), plus an additive white noise component. Thus the error which is actually observed is defined, in terms of the variables denoted below,

$$f(t) \stackrel{\text{\tiny def}}{=} H(t) x(t) + \eta(t) , \qquad (5.3)$$

where

y(t) is the m-dimensional error difference which is actually observed;

- H(t) is an m x n matrix describing the linear inter-relationship of the n-error components of x(t) entering into the observation error;
- $\eta(t)$  is a zero-mean, white noise process, where

$$R(t)\delta(t - \tau) \stackrel{f}{=} E[e(t) > < \epsilon(\tau)]$$

When an update is executed, an  $n \times m$  gain matrix, K(t), is used to weight the vector, y(t), to form a correction vector, c(t). This is then introduced into the system to result in the following change in the current system error:

$$x^{+}(t) = x^{-}(t) - o(t)$$
, (5,4)

where

 $x^{-,+}(t)$  is the current system error before and after the update is made, respectively;

 $c(t) \stackrel{\triangle}{=} K(t)y(t)$  is an n-dimensional correction vector;

K(t) is the n x m gain metrix used to weight the observed error difference, y(t).

The transition of the covariance matrix describing system error upon update is given by the well-known equation

$$\Sigma^{+}(t) = [I - K(t)H(t)]\Sigma^{-}(t)[I - K(t)H(t)]^{T} + K(t)R(t)K(t)^{T}, \qquad (5.8)$$

where  $\Sigma^{*,*}(t)$  is the covariance matrix before and after update, respectively.

For that case in which the optimal linear controller with respect to the error model is implemented, the optimal gain is given by the well-known equation<sup>1</sup>

$$K^{*}(t) = \Sigma^{*}(t)H(t)^{T}[H(t)\Sigma^{*}(t)H(t)^{T} + R(t)]^{-1}, \qquad (5.6)$$

The use of the optimal gain  $K^{+}(t)$ , allows Equation (5.5) to be written in the simplified form

 $\Sigma^{*}(t) = [I - K^{*}(t)H(t)]\Sigma^{-}(t) , \qquad (5.7)$ 

Our interest here is not the use of this optimal mechanization, but rather one which is less complex, where the reduction in complexity results from a direct examination of the system error in Equation (5.1). With this approach, the analyst first examines the error vector x(t), and deletes those components which he feels contribute in a relatively minor way to total error in the system epplication. Once this is done, a new error vector  $\bar{x}(t)$ , of reduced dimension  $\bar{n}$ ,  $\bar{n} \leq \bar{n} < n$ , is obtained. The analysis then examines the subset of elements of the error dynamics matrix A(t), which describe the interaction of the error components in  $\bar{X}(t)$ . Elements of this subfeviated portion of the matrix A(t) quot felt to describe significant interaction of the components of  $\bar{X}(t)$  are then eliminated. A new dynamic matrix  $\bar{A}(t)$ , of dimension  $\bar{n} \times E$ , can then be formed of the retained elements of A(t). The next step is to determine whether the disturbance white noise process,  $\epsilon(t)$ , might be modified to aid in describing the actual system error propagation. Usually an increase in the magnitudes of the singler error model, components in obtaining  $\bar{X}(t)$  from x(t). The final result is then a magnitude with the elimination of error components in obtaining  $\bar{X}(t)$  form x(t). The final result is then a simpler error model described by

$$\frac{d}{dt} \begin{bmatrix} \mathbf{x} \end{bmatrix} = \overline{\lambda}(t) \mathbf{x}(t) + \overline{\epsilon}(t) , \qquad (5.8)$$

where

X(t) is the E-dimensional error vector  $(m \leq \overline{n} \leq n)$ ;

 $\bar{\mathbf{x}}(t)$  is the  $\bar{\mathbf{x}} \times \bar{\mathbf{x}}$  dimensional dynamics matrix for the vector  $\bar{\mathbf{x}}(t)$ ;

**T(t)** is an fi-dimensional zero-mean, white noise process, where







# $Q(t)\delta(t-\tau) \stackrel{\Delta}{=} \mathbb{E}[\mathbb{Z}(t) < \mathbb{Z}(\tau)]$ .

Clearly, the equation describing the propagation of the covariance matrix corresponding to Equation (5.8) is

$$\frac{d}{dt} \left[ \widehat{\Sigma} \right] = \overline{\lambda}(t) \widehat{\Sigma}(t) + \widehat{\Sigma}(t) \overline{\lambda}(t)^{T} + \widehat{Q}(t) . \qquad (5.9)$$

The description of the observation process reculting from the use of the error model in Equation (5.8) is necessarily changed. The simplified observation matrix  $\vec{H}(t)$  is obtained by including only those elements in H(t) involving the error components rotained in  $\bar{X}(t)$ . Again to compensate the reduction in the uncertainty due to elimination of error vector components, the white noise process,  $\gamma(t)$ , might be modified in obtaining a model of observed error. As was done in the case of the disturbance noise process, the magnitudes of the observation noise components are usually assumed larger to introduce, artificially, uncertainty into the description of the simpler model. The final result is a representation of the error actually observed in terms of the variables denoted below:

$$y(t) \cong H(t)x(t) + f(t)$$
, (5.10)

y(t) is the actual m-dimensional observation vector;

where

 $\mathbf{R}(\mathbf{t})$  is an (a  $\times$  B) matrix specifying the relationship desired between components of the error vector  $\mathbf{X}(\mathbf{t})$ ;

7(t) is a mero-mean, m-dimensional white noise process describing additive noise in the observed errors (m  $\leq B$ ), where

If Kalman's algorithm for the computation of the gain matrix is now employed:

-

$$f(t)^{\pm} = \Sigma(t)f(t)^{T}(f(t)\Sigma(t)^{\pm}f(t)^{\mp} + f(t)^{\pm i},$$
 (5.11)

and used to weight the actual observed error, the correction obtained for the simplified error model at update is

\_

$$\mathbf{X}^{+}(t) = \mathbf{X}^{-}(t) - \overline{c}(t)$$
, (5.12)

where  $\mathbf{x}^{*,*}(t)$  is the modeled system error before and after update, respectively, and  $\overline{o}(t) \stackrel{\Delta}{\rightarrow} \overline{\mathbf{x}}^{*}(t)y(t)$  is the correction vector. The transition  $\mathcal{A}^{*}$  the covariance matrix corresponding to the simplified error model is expressed from Equation (5.12) as

$$\vec{\Sigma}(t)^* = [I - \vec{R}^*(t)\vec{H}(t)]\vec{\Sigma}(t)^* , \qquad (5,13)$$

Now that the technical details have been presented, a review is in order. First, by assumption, the analyst has been given a complex linear model, defining error propagation in Equation (5.1), and the observed errors in Equation (5.3), for some system. The objective is to design an error controller which, while having the statistical correlation attributes of Kalaan theory, has a lesser computational burden than that implied by the optimal mechanization. The optimal mechanization requires the use of Equations (5.2), (5.4), (5.6) and (5.7). The object introduced into Equation (5.1) and their impact on Equation (5.3). The results are the simplifications can be introduced into Equation (5.1) and their impact on Equation (5.3). The results are the simplified expressions (5.8) and (5.10). From these equations a less complex Kalman mechanization is implied, which would be implemented with Equations (5.9) and (5.11)-(5.13).

Clearly, the simpler error controller realises poorer performance than that achieved with a mechanization based on the complete error model. However, the performance loss incurred by using the simpler mechanization might be insignificant in terms of the performance required of the system. The question of practical importance which remains is that of determing the performance loss. The method by which this question is answered is also the manner by which the analyst will test the validity of the various simplifications and alterations of the original error model that are made.

The method by which the answers to these questions are obtained is quite simple. First the ultimate performance obtainable is determined by propagating the covariance matrix  $\Sigma(t)$ , using Equation (5.2), with optimal updating through the use of Equations (5.6) and (5.7). The solution for  $\overline{\Sigma}(t)$ , denoted  $\overline{\Sigma}_{\rm H}(t)$ , and, in particular, for the diagonal elements or error variances then serves as the reference against which performance obtained with any other mechanization can be compared. The second step involves the propagation of the two covariance matrices  $\overline{\Sigma}(t)$  and  $\overline{\Sigma}(t)$ .  $\overline{\Sigma}(t)$  is again propagated using Equation (5.2), while  $\overline{\Sigma}(t)$  is propagated by Equation (5.3). At the update instants, the gain  $\overline{K}^*(t)$ , using Equation (5.1), is computed and used in Equation (5.13) to update  $\overline{\Sigma}(t)$ . To determine the effect of using  $\overline{K}^*(t)$ , which is sub-optimal compared to  $\overline{K}^*(t)$  obtained by Equation (5.6) on the actual system errors, an (n × m) matrix K(t) is defined by augmenting  $\overline{K}(t)$  suitably with zeros. Then  $\overline{K}(t)$  is employed in Equation (5.5), which describes the effect on the actual system error covariance matrix. Comparison of the solution for  $\overline{\Sigma}(t)$ , denoted as  $\overline{\Sigma}_{\rm S}(t)$  and determined in the second step, with the reference solution  $\overline{\Sigma}_{\rm R}(t)$ , denoted as  $\overline{\Sigma}_{\rm S}(t)$  and determined in the second step, with the reference solution  $\overline{\Sigma}_{\rm R}(t)$ , obtained in the first step, then indicates the degree of performance degradation that results through the use of the proposed computationally-lees burdename mechanization.

5. Sec. C

AND ANTIMA 114











NACCOVAL AND MONTONIA ST

There is a significant increase in cost of the computer analysis with un increase in the dimension of the reference model in the synthesis approach described. Therefore, the analyst must often make certain approximations even in the reference error model. To obtain a suitable roferente model of manageable dimension, multiple sources of error which enter the system at the same point can be merged and represented as single noise error. To maintain the integrity of the design results, the statistical characteristics of this lumped noise source can be selected to more than account for the multiple sources of error represented. For example, in the design of augmented inertial navigation systems, a finely-detailed structure can be developed to model the error in the accelerometer measurements or the stability of the gyro orientation. Upon examining this detailed error structure and considering the environment in which the instrument must operate, the analyst can represent these errors as single sources of first-order correlated noise. With these assumptions used to obtain a simple stochastic gyro drift rate or accelerometer bias shift, the synthesis tochnique described could then be used to determine precisely the relationships between proposed design innovations and the system performanco.

Once a navigation system error controller is derived with this technique, some question might exist as to the validity of the assumptions made in using simple descriptions of the inertial instruments. This question can now be addressed in a meaningful mannor by employing a Monte Carlo program to simulate the system using refined error models and the newly-derived error controller. Accepting the imprecision of results obtainable with a few trials using the Monte Carlo program, gross deviations from expected system performance can be adequately identified at a reasonable cost.

In summarizing this section, a straightforward application of Kalman theory to augmented inertial navigation system design using all information about the sensor error characteristics can result in a design which is impractical in terms of the requirements imposed on the system digital computer. A method of obtaining an error control mechanization which has the desirable correlation property of a Kalman filter was described in detail. This approach is direct in the sense that it deals with the character of the modeled error characteristics of the sensors. A digital computation procedure was outlined whereby this synthesis approach could be utilized in designing error controllers for complex systems. This procedure has been used in the avionics systems industry, particularly in obtaining the sugmented inertial navigation system error control mechanization described in Section 7.

### 8. IMPLEMENTATION OF KALMAN ERROR CONTROLLERS IN AUGMENTED INERTIAL NAVIGATION SYSTEMS

#### 6.1 General

1

In a previous section, practical considerations influencing the implementation of a Kalman error controller in an augmented inertial navigation system were discussed. We now consider the detailed operations which are implemented to mechanize the derived error controller in the system digital computer. When the synthesis of the Kalman controller is dompleted, the design error model for the system is described by the following linear differential equation, which is represented schematically in the block diagram of Figure 6:

$$\frac{d}{dt} \left[ \mathbf{X}(t) \right] = \overline{A}(t) \mathbf{X}(t) + \overline{\mathbf{x}}(t) . \tag{6.1}$$

In augmented inertial navigation systems, measurements are made at discrete instants of time and compared with estimates of their values computed from the system navigation variables. For sufficiently small errors, the difference of the measurement and the predicted value can be represented as a linear combination of error states. The difference is called the observation error and is written

$$y(t_{i}) = \Pi(t_{i}) \Xi(t_{i}) + \eta(t_{i})$$
. (6.2)

When a measurement is made the observation error is multiplied by a Kalman-derived gain matrix to form a correction vector to update the system:

$$\sigma(t_1) = R(t_1)y(t_1)$$
, (6.3)

The updated value of the modeled error state vector of the system is expressed by

$$\mathbf{X}(\mathbf{t}_{i})^{*} = \mathbf{X}(\mathbf{t}_{i})^{*} + \overline{\mathbf{c}}(\mathbf{t}_{i}) . \tag{6.4}$$

When corrections are used to alter the values of the navigation variables within the system, servo-mechanism control of the modeled errors is realized. The implications of such a mechanization as opposed to the formal Kalman filter implementation, are noted later.

A description of the error behavior of the system for the operations defined above can be represented schematically by the block diagram shown in Figure 7. This diagram is analogous to that for a servo-mechanism whit, atterpts to nell the system outputs by feeding linear combinations of these outputs back through the gap,  $eescil_{i} \in Eeroman Shoory contributes to error controller design primarily through the algorithm used to$  $obta, where <math>e_{i} \in Aerois$ . The gain matrix obtained with Kalman theory takes cognizance of the statistical state of  $e_{i} = 10 - 2e^{-2} = ... Stars is corrected by the covariance matrix and the quality and type of observation available.$ 

As the second or state vector changes with time between the system correction instants as described by Equation 6.0.1, and covariances of these error states also change. These changes in the statistical state of

















the system error can be determined in real-time through the propagation of the system error covariance matrix. Alternate methods have been derived for performing the computations required for propagating this information. These methods include direct integration of the differential equation for the covariances and discrete propagation through use of the transition matrix. Other achemes<sup>4</sup> deal with root mean square values rather than the variances of the errors to reduce the precision requirements in the digital computation. In any case, the particular application under consideration and the equipment available for system implementation will determine which of these computational techniques is the most advantageous to employ. When a measurement is made with a sensor, the correction of the system requires the computation of the gain matrix. When a system correction is made, obanges occur in the statistical inter-relationships between the error state variables. At the update time the covariance matrix elements are altered to reflect the fact that the system correction was made.

A review of the above paragraphs concludes that the following basic operations must be implemented in the system digital computer to accomplish the mechanization of a Kalman error controller:

- 1. Propagation of covariance matrix variables between instants of time at which measurements are made.
- 2. Selection and sampling of the measurement data to be processed at the system update time.
- Computation, based on the values of system variables at the update time, of predicted values for the selected measurements and formation of the differences of these predictions with the actual measurement data to obtain observed errors.
- Computation of the Kalman gain matrix, based on covariance matrix values at the update time and the type and quality of the measurement data.
- 5. Weighting of the observed error with the Kalman gain matrix to obtain the correction term used to update the system variables whose errors are represented in the design error model.
- 6. Updating of the covariance variables at the update time to reflect the execution of the system correction.

The logical relationship of these basic operations is summarized schematically in Figure 8.

#### **5.9 Error Controller Configurations**

Two major approaches have been proposed for the configuration of the error control mechanism. In the first and most popular approach, the mystem correction vector determined at a measurement time is used to update values of the navigation variables and to biam noise sources within the system. This is the basic feedback scheme already discussed. In the second approach, only the system outputs are corrected using the values of propagated estimates of the system errors.

# 6.3 Feedback Correction

analan kissee wallon-kanger

In the feedback correction scheme, the system is organized schematically as shown in Figure 9. Here corrections are made to those errors in each subsystem which are included in the design error model. The feedback of the corrections derived by weighting the observed error with the Kalman gain matrix has two advantages. Since the Kalman controller is based on a linear error model of the system, any process which keeps this assumption valid is advantageous. Since proper feedback correction will tend to decrease error within the system, it is advantageous to maintain the linearity assumption made for the behavior of the errors. Classically, feedback has been used to neutralize automatically the effect of variations within a system. This insensitivity property inherent to feedback systems leads to the second advantage in employing the servo-mechanism approach. The more error states which are included in the design error model, the better should be the performance obtained by the system. However, the basic insensitivity realized by feedback to the system variations implies that the effect of many sources of error will automatically be reduced even when they are not included in the design error model.

#### 6.4 Output Correction

In the output correction scheme, the system is organized schematically as shown in Figure 10. Here no direct corrections are made to assumed sources of error in the subsystems. The linear error model equations for the system are used to propagate the error state vector estimate. A correction vector is then derived as a linear transformation of the propagated error vector and used to update only the system outputs.

The chief disadvantage of such an approach is that error is introduced through imperfect modeling. This occurs through two effects. Since error in the system is not corrected toward the null, as with the feedback method, the assumption of linearity can become invalid. Also, the insensitivity characteristic of the feedback scheme is lost. Consequently, a higher-dimensional linear error model will probably have to be employed as a basis of the Kalman filter to obtain performance comparable with that obtained using the feedback scheme. Another disadvantage of this approach is that an estimate of the modeled system error states must be propagated. This is avoided in the feedback mechanization. An advantage of this approach is that improper updates resulting from undetected spurious subsystem operation will affect only the outputs. In the feedback method an improper update can drive a subsystem to an improper operating region, resulting in a loss of overa'l system performance. However, such behavior can be avoided in the feedback method by developing effective checks on subsystem operation and the measured error difference.

247

1711 - M**O**L

A MANAGAN

# 7. A KALMAN ERROR CONTROLLER DESIGN FOR MULTI-SENSOR AUGMENTED INERTIAL NAVIGATION SYSTEMS

#### 7,1 General

The Doppler radar, astro-tracker, LOKAN receiver and inertial platform are representatives of different classes of navigation sensors having complementary characteristics which can be employed to advantage in an integrated navigation system. The autonomous and conventional integrated operation of these sensors has been discussed. The Doppler-inertial, astro-inertial and LORAN-inertial-navigation modes are re-examined below, using the more sophisticated Kalman linear correlation theory for more optimal mixing of the sensor data. The mechanization obtained is suitable for implementing any combination of the foregoing modes, including the fully-integrated astro-Doppler-inertial-LORAN navigation mode.

As previously noted, a straightforward application of Kalman theory may not be desirable or necessary in obtaining a navigation system error control mechanization: not desirable, owing to the magnitude of the computer requirements implied for an implementation based on a dotailed error model of each sensor; not necessary, as the rission to which the system is to be applied may not warrant a highly-complex mechanization. A thorough examination of the navigation system requirements generated by the intended mission must be made if the resulting design is to be cost-effective.

The mission considered in synthesizing the navigation system of this section is termed fast-reaction, meaning that accurate values of vehicle position, velocity and attitude are desired within a short period of time after the system is activated. In this mission no operational restrictions on the vehicle should result from any characteristic of the navigation system operation. For this mission the system error control mechanisation described below was derived using synthesis procedure described in Section 5. The performance of the navigation system, implemented with the derived Kalman mechanization, is illustrated for the fast-reaction mission in the principal integrated modes of operation.

#### 7.8 Derivation of the Kalman Mechanization

The straightforward application of Kalman theory to a detailed error model of an astro-Doppler-inertial-LORAN system could easily involve computations dealing with fifty or more error state variables having quite complex dynamic interactions. If a Kalman mechanization were based on such a model, corrections would be obtained for the accounted sources of equipment and environmental error, as well as the system's navigation outputs. In addition, through a detailed description of the dynamic interaction of variables, error propagation over the long-term would be accurately predicted.

An examination of the application of the navigation system to meet the requirements of the fast-remotion mission results in the primary requirement that the inertial platform be rapidly aligned regardless of the operational environment. Rapid platform alignment is necessary as the mission requires dynamically accurate attitude and velocity information, which are natural outputs only of the inertial system. Accurate values of these variables will not be available until platform alignment is achieved. If platform slignment and control of position and velocity errors are the principal objectives of the Kalman error control mechanization, it remains only to determine which of the system errors and interactions has especial importance in meeting this objective. Since position, velocity and attitude are the required system outputs, it is clear that errors in these variables must be controlled. Clearly then, a minimal mechanization would incorporate at least the errors in these variables as components of the state vector of the single error model.

Platform alignment is essentially a process of eliminating high initial values of attitude error and any error in computed velocity or position occurring as a result of attitude error. The existence in the system error model of biases or noise sources associated with equipment or induced by the environment will then only be relevant if such sources seriously affect alignment. The optimal control of errors resulting from stochastic inputs to a system is usually a problem which occurs more often in the steady-state. In navigation systems, such error control is of importance in applications involving long-term cruise missions requiring extremely high navigation accuracy. For the alignment application, many such sources are not likely to be very important and consequently, can be eliminated in the simple error model. It will be recalled that the loss in uncertainty, resulting from an elimination of these variables from the error model, can be partially accounted for by artificially increasing the variances of appropriate white noise sources in the simple model. However, exceptions to this philosophy exist. These exceptions can occur when the error in question is a long-term noise which constitutes a significant fraction of the difference between a measurement value and the prediction of that value during platform alignment. When this situation exists, the modeling error, resulting from an exclusion of the noise source in question, can cause a significant loss in system performance which cannot be offset by white noise parameter adjustments. One case of this sort occurs during gyro compassing of the inertial platform with Doppler speed data containing large biases. Another example is afforded by the case in which there exists significant distortion of the actual electronic LORAN time difference grid from the theoretical grid based on geodetic data. This distortion is caused by variations in the propagation speed of the electromagnetic energy over the earth's surface resulting from unaccounted variations in the nature of the paths of propagation.

In addition to the elimination of most noise sources in the alignment application, dynamic interactions between the retained error state variables can be reduced to a minimum by eliminating those defining error propagation characteristics which are long-term with respect to the fast-reaction mission. Such couplings are associated primarily, with the inertial sensor defining the characteristic 24-hour oscillations and modulating Poucault frequencies which are present in a detailed error model.

















and a second second

#### 7.3 Kalman Error Controller Description

After examining a detailed error model of the navigation system and the requirements of the fast-reaction mission along the lines outlined, the following minimal error control mechanization can be derived. This mechanization will incur only a small loss in system performance during platform alignment, if relatively accurate sensors are employed in the navigation mystem. The mechanization assumes the north-slave, Schuler-tuned platform implementation, but can accommodate any local-level platform mechanization with only slight modification. It is a minimal design in the sense that only the errors in the required system outputs are represented in the simplified error state vector, consequently reducing the model dimension to a minimum. The seven elements of this state vector are



where

大学では必要

55222 442 A

- $\delta \theta_{\pi,N}$  are the angular errors in system computed vehicle position, represented as rotations about the local geographic east and north axes, respectively;
- $\delta v_{n,M}$  are the errors in system computed vehicle velocity along the local morth and east axes, respectively;
- \$\phi\_N, \$\vec{\phi}\$ are the angular errors in platform attitude, represented as rotations about the local east, north and vertical axes, respectively. The east and north components are usually referred to as platform tilts while the third component is called platform wimuch misalignment.

The matrix defining the dynamic interaction of these moven error state variables is

	6	0	0	8*1	0	0	0
	0	0	R-1	0	٥	0	0
	0	0	0	0	-8 <sub>82</sub>	0	8 <sub>1p</sub>
٨́(t)	0	0	D	0	8 <sub>Np</sub>	-8 <sub>5p</sub>	0
	0	0	0	0	0	0	0
	0	0	0	R*1	دن <b>د</b> ۲۵	0	0
	0	0	-8-1	0	-cunc	0	0

This matrix has been reduced to include only those terms essential to alignment followed by cruise navigation for a short time period. The result is that only ten of the elements are non-zero. Of these, four are identical constants  $\mathbb{R}^{1}$ ,  $\mathbb{R}$  being a nominal radius for the earth and four are derived from the specific force measurements,  $\mathbb{S}_{\text{MD}}$ ,  $\mathbb{S}_{\text{ND}}$  and  $\mathbb{S}_{\text{ZD}}$ , along the local east, north and vertical axes, respectively. The final two elements are spatial rate terms,  $\omega_{\text{MD}}$ ,  $\omega_{\text{ND}}$ , being the domputed rates at which the platform must rotate about the local east and north axes, respectively, to remain locally-level, Clearly, this representation of inertial error behavior is much simpler than that described in Section 3.

The sero-mean, white disturbance noise vector, corresponding to Equation (7,1), can include up to seven components.

 $\vec{\xi} = \vec{\xi} \delta v_{\mu}$   $\vec{\xi} = \vec{\xi} \delta v_{\mu}$ 

to account to a certain extent for the deletion of noise state variables from the detailed system error model. The variances of the elements  $\hat{\mathcal{L}}_{\mathcal{O}_{\mathbf{N}},\mathbf{N},\mathbf{Z}}$  are of principal importance in accounting for errors which have an

(1.3)

249

(7.1)

1465 A. Y.

effect on the platform as does a gyro drift rate about the corresponding axis. The variances of the elements  $\overline{\xi}_{\delta V_{N,B}}$  are introduced to account for errors which have a stochastic effect on the velocity computation as does accelerometer noise, quantization error, misalignment, etc. Finally, the variances of the elements  $\overline{\xi}_{\delta \theta_{N,B}}$  reflect additional uncertainty in the position computation.

The description of the error model given above corresponds to propagation between instants of time at which system updates are made. To employ Kalman's algorithm at an update time, a model of observation error must be constructed. In an astro-Doppler-inertial-LORAN navigation system, the measurements that can be made with the sensors which augment the inertial sensor are Doppler ground speed, Doppler drift angle, LORAN time-difference(s), tracker elevation angle, and tracker bearing angle.

When one of these measurements is made, it is differenced with a predicted value of the measurement based on the computed values of the system navigation variables. If the associated errors are sufficiently small, the measurement difference can be represented as a linear combination of the errors involved. The coefficients of the error terms in such an expression are simply gradient components evaluated at the local operating point. Since noise terms associated with the tracker, Doppler and LORAN sensors are not incorporated in the simple error model, the observation matrices involve only gradient components relative to the errors in the navigation variables. These matrices are now defined for the different measurements.

The vehicle ground speed observation matrix is expressed as

$$\mathbf{R}_{\Delta \mathbf{V}_{\perp}} = \begin{bmatrix} \mathbf{0} & \mathbf{0} & \cos \theta_{\mathbf{0}} & \sin \theta_{\mathbf{0}} & \mathbf{0} & \mathbf{0} \end{bmatrix}, \qquad (7.4)$$

where  $\theta_{\rm d}$  is vehicle track angle and  $\Delta V_{\rm g} \stackrel{\Delta}{=} [V_{\rm gc} - V_{\rm gb}]$ , denotes the difference between computed,  $V_{\rm gd}$ , and Doppler-measured,  $V_{\rm gd}$ , ground speed.

The vehicle drift angle observation matrix is expressed as

$$\widetilde{H}_{\Delta\delta} = \begin{bmatrix} 0 & 0 & \frac{\sin \theta_{0}}{V_{g0}} & \frac{-\cos \theta_{0}}{V_{g0}} & 1 & 0 & 0 \end{bmatrix}, \qquad (7.5)$$

where  $\Delta \delta \stackrel{\circ}{\cong} [\delta_n - \delta_p]$  denotes the difference between computed,  $\delta_n$ , and Doppler-measured,  $\delta_n$ , drift angle.

The time-difference observation matrix is expressed as

$$\widetilde{H}_{\Delta_{\mathbf{f}}} = \frac{n}{0} \left[ \left( \sin \beta_{\mathbf{H}} - \sin \beta_{\mathbf{g}} \right)_{\mathbf{g}} \left( \cos \beta_{\mathbf{H}} - \cos \beta_{\mathbf{g}} \right)_{\mathbf{g}} = 0 \quad 0 \quad 0 \quad 0 \quad 0 \quad 1 \quad (7.6)$$

where  $\beta_{\rm H,S}$  are the computed bearing angles to the LORAN master and slave subtions relative to the local north line, respectively, C is the velocity of light, and  $\Delta t \cong [t_0 - t_1]$  denotes the difference between computed,  $t_0$ , and LORAN-measured,  $t_1$ , time-difference.

The elevation angle observation matrix is expressed as

$$\mathbf{H}_{\Delta \mathbf{n}} = [\sin \mathbf{B}_{\mathbf{n}} - \cos \mathbf{B}_{\mathbf{n}} = 0 \quad 0 \quad -\sin \mathbf{B}_{\mathbf{n}} \quad \cos \mathbf{B}_{\mathbf{n}}], \quad (7.7)$$

where  $B_0$  is the computed star bearing angle with respect to the local north axis, and  $\Delta E \stackrel{\Delta}{=} [B_0 - E_T]$  denotes the difference of computed,  $E_0$ , and tracker-measured,  $E_T$ , star elevation angle.

The bearing angle observation matrix is expressed as

 $\mathbf{R}_{\Delta \mathbf{R}} = \left[ \left( -\cos \mathbf{B} \tan \mathbf{E} + \tan \phi \right)_{\mathbf{C}} \left( -\sin \mathbf{B} \tan \mathbf{E} \right)_{\mathbf{C}} 0 \quad 0 \quad -1 \quad \left( \cos \mathbf{B} \tan \mathbf{E} \right)_{\mathbf{C}} \left( \sin \mathbf{B} \tan \mathbf{E} \right)_{\mathbf{C}} \right], \quad (7.8)$ 

where  $\phi_a$  is the computed latitude. E the computed star elevation with respect to the local-level plane, and  $\Delta B = [B_a - B_T]$  denotes the difference between computed,  $B_a$ , and tracker-measured,  $B_T$ , bearing.

The final elements of the error model to be defined are the zero-mean, additive white noise processes associated with each of the measurements cited above. As has been stated previously, uncertainty can be artificially associated with the observation process in the simple error model by increasing the variances of these noise components from values associated with the detailed error model. When the sensors used to align the inertial platform are collatively accurate, the components deleted in constructing a simple error model on which to base the Kalman mechanisation can be accounted for adequately by this method. For the above system, the vector denoting additive white observation noise would include:







The variances of the elements  $\eta_{\Delta V_R}$ ,  $\gamma_5$  are used to account for noise error in the Dopplor ground speed and drift angle measurements which vary with vehicle speed, altitude, stitude and torrain. The variance of the element  $\eta_{\Delta t}$  represents LORAN-measured time-difference hoise error, which varies with distance from the transmitting stations, local interference and some anomalies associated with the paths of propagation. Finally, the variance of the elements  $\eta_{\Delta t}$ , cas account for noise error associated with tracker measurements of star. elevation and bearing, respectively, which vary with star brightness and atmospheric conditions.

#### 7.4 Description of a Fast-Reaction Mission

We next illustrate the relative performance achievable in the different navigation modes when the Kalman feedback error controller described is employed. This is done for a nominal fast-reaction mission for which this controller was designed. The performance of the system for the major modes of navigation is discussed in detail later.

The vehicle flight dynamics result in the profile indicated schematically in Figure 11. In more detail, it was assumed that the aircraft spent two minutes on the ground after the inertial system was activated. Alignment of the platform was initiated when the inertial system was activated. The initial r.m.s. attitude errors of the platform was maximud to be 300 arc-seconds in till and 3600 arc-seconds in azimuth. The vehicle was assumed to remain stationary having no velocity during the two-minute ground alignment period. This results in the null-velocity mechanization commonly employed for ground alignment of inertial platforms. The assumption in reality is in error because of platform motion caused by wind buffeting and disturbances, resulting from the loading of fuel, cargo, armament etc., of the airframe. After the two-minute ground alignment period at coordinates of latitude  $\Phi = 12 \deg$ , and longitude,  $\lambda = 100 \deg$ , the vehicle was accelerated on a heading  $\theta = 180 \deg$  southward and vertically at 25 and 16 ft/sec<sup>2</sup>, respectively. When a ground speed of 1000 ft/sec was achieved, a turn of 160 deg at a rate of 3 deg/sec was initiated such that, at 1 min 40 act into the mission, the siroraft was directed northward. The aircraft proceeded at a club rate of 80 ft/sec until the oruise altitude of 10,000 ft was achieved. After one half hour, the aircraft one hour of operating time.

#### 7.5 System Performance for the Nominal Fast-Reaction Mission

The graphs shown in Figures 12-23 illustrate system performance obtained in the free-inertial, Dopper-inertial, LOKAN-inertial, and astro-inertial navigation modes for the fast-reaction mission. The performance measures displayed are the error in computed system position and . slocity and the szimuth misslighment at the inertial platform. The performance was obtained through the use of a digital computer Monte-Carlo simulation program. The curves reflect sample root mean aquare (r.m.s.) values of the indicated errors obtained from 30 trial runs. This sample size does not give precise results but indicates the appropriate order of magnitude of trends. Extensive models incorporating random bias, correlated noise and white noise were employed to represent the error characteristics of each navigation sensor. The bias errors present in each sensor are constant over any one simulation run. They vary from run to run by virtue of being selected randomly from a zero-mean, uniform distribution of prescribed standard deviation. These distributions were derived using digital random number generators. The essentially white noise sources in each sensor are simulated as sequences of samples selected randomly from sero-mean, uniform distributions of prescribed standard deviations derived in the same manner as for the bias errors. The correlated noise sources in each sensor are represented as exponential noise. Such noise is simulated by passing simulated white noise through a first-order digital smoother whose analog time constant is equal to the desired correlation time of the noise. The standard deviation of the exciting white noise sequence is selected to obtain the prescribed standard deviation desired for the correlated sequence.

In the following paragraphs, performance obtained in each of the navigation modes, as shown in the figures, is discussed. The mechanization used for error control in all the navigation modes was the Kalman feedback design, based on the extremely simple system error model described above. In each case the quality and quantity of data available from the defined and any unique feature of the particular mode of operation is discussed.

#### 7.6 Free-Inertial Navigation Mode

i.

The system performance obtainable in the free-inertial mode of mavigation depends oritically on the amount of time allotted to the inertial platform alignment on the ground. If the aircraft takes off with large platform attitude errors, relatively large errors in computed system position and velocity will result. This obtains as the acceleration information measured in platform coordinates is processed in the computer on the assumption that the accelerometer axes are doincident with the local geographic axes. In the augmented inertial navigation modes, completion of ground alignment is not as critical, as the desired coincidence of the actual accelerometer axes and the geographic coordinates can be obtained through the use of data from the augmenting sensor.

During stationary ground alignment, platform attitude error is removed in two stages, referred to respectively as leveling and gyro compassing. When the level accelerometers deviate from the local horizon plane, their sensitive axes detect components of the local gravity vector. These measured components result in errors in computed system position and velocity through subsequent integrations in the computer which can then be used to remove the platform tilts. Note, however, that, if the platform is assumed stationary, exact determination of the attitude error cannot be achieved when platform motion is caused by local stochastic disturbances.

If there exists a misalignment in the platform azimuth once the initial level error is approximately removed, the platform still tends to deviate from the level plane. This obtains as the platform is rotated to maintain

251

b within a

The support of the state

the local-level orientation as the earth rotates. Deviations of the platform axes from the assumed coordinates result in erroneous rotation of the platform coordinate system. The resulting tilt caused by the platform azimuth misalignment will again generate errors in computed system position and velocity. These errors then provide the basis to remove approximately the azimuth error. It should be poted that effective determination of the error in computed system position or velocity due to azimuth misalignment can occur only after platform level has been obtained. During ground alignment, platform leveling is achieved much more rapidly than the gyrocompassing of the platform azimuth. Further, as the latitude at which alignment occurs is increased, the time required for the azimuth gyrocompassing also increases. Noth these effects obtain as the coefficient in the linear error model of the inortial system which couples platform azimuth misalignment to platform tilt is small with respect to unity and decreases in magnitude as the latitude is increased. The speed at which overall alignment is achieved is also dependent on the system correction rate and the r.m.s. value and correlation characteristics of the stochastic motions of the platform. For the examples of this section, a system correction was made every second, and the null reference in velocity was assumed to be characterized by a white noise with an r.m.s. value of 1/8 ft/sec.

From this discussion we sue that the platform will be aligned only after the azimuth error is minimized. In Figures 12-14, system performance is illustrated for the free-inertial modes for ground alignment times of 2, 4 and 10 minutos. At 10 minutes, ground alignment has essentially been completed and subsequent errors shown in computed system position and velocity are due primarily to stochastic errors in the inertial equipment. From these particular results, the inertial system would be referred to as a 0.75 mile per hour inertial system. This inertial sensor error characteristic applies for all the performance curves shown in this section. At 2 minutes, a relatively larger azimuth missignment exists. Consequently, when the aircraft executes the flight dynamics described above, the acceleration measurement error due to azimuth missignment dominates the propagation of error in computed system position and velocity.

# 7.7 Doppler Inertial Navigation Mode

The system performance obtainable in the Doppler-inertial mode of navigation does not depend critically on the ground alignment time, as once the system is airborne, speed measurements made with the Doppler radar can be used to continue the platform alignment. During the period of airborne alignment, system performance will obviously be degraded with respect to the performance which results when a longer platform ground alignment time is assumed. Since the inertial and Doppler velocity vectors occurs instantaneously on take-off due to platform attitude error. The detection of attitude error results therefore from the differing longer term error characteristics of the two sensors. The characteristic errors in the inertial system have frequencies associated with the Schuler, 24-hour and Foucault oscillations. Bince the presence of these oscillations only in the platform attitude error affects the stabilized Doppler data, a basis for error separation at the velocity level exists. It turns out that air-borne platform alignment Doppler data is equivalent in form to ground alignment assuming a null velocity reference.

The rapidity at which the velocity error characteristics can be separated depends on the quality and quantity of the Doppler data. For two of the performance curves shown in Figures 15-17, the Doppler was assumed to provide horizontal speed components with an r.m.s. bias error of 0.1% and an r.m.s. white noise error of MK. Experiments made for this mission, as indicated in Figures 15-17 show that doubling of the r.m.s. bias values results in significant loss of system performance if the minimal Kalman mechanization is employed. Such sensitivities can be determined precisely through the use of the correlation analysis technique described in Section 5. In the cases considered in Figures 15-17, different white observation noise covariance matrices, R, were employed. These matrices differed over the time period from 2 to 30 minutes in which most of the platform asimuth misalignment is schieved with the Doppler data. When the R matrix is increased by a factor of three, the tuning factor, errors in computed velocity and the platform maint he significantly reduced. This occurs because the Doppler bias errors, unmodeled in the Kalman controller, can be accounted for to some extent, artificially by increasing the white observation noise variances.

Using the Doppler data, corrections to the modeled errors in computed system position, velocity and the platform attitude were made with the Kalman error controller every 5 seconds. These updates were initisted at the vehicle take-off. An important characteristic of system error using the Kalman mechanization is the correction of position realized with the velocity data. In conventional design, since the correlation information relating measured velocity error and position error is not available, no basis for updating the error in computed system position stints. Consequently, only a bounding effect on velocity error is realized in contrast to effective control of all the modeled errors with the Kalman design.

#### 7.8 LORAN-Inertial Navigation Mode

Gystem performance obtained in the LORAN-inertial navigation mode is essentially limited only by the quality and quantity of the geodetic position data derived from the LORAN time-differences. Unlike the Doppler updating process, extremely rapid alignment of the inertial platform can be realized with the receiver data for the fastreaction mission. This obtains as the platform does not play the data stabilization role with LORAN data that it does with Doppler data. When the microaft takes off, transient acceleration errors and, subsequently, velocity and position errors, exist in the inertial system as a result of the platform attitude error. These components of inertial system position error are readily detectable using time-difference data obtained in a region of defined LORAN coverage. Correction of platform attitude and computed velocity is achieved with the correlation data relating measured position error to error in computed system velocity and the platform missignment angles. The real-time availability of this information accounts sutomatically for any vehicle flight profile or sequence



.of measurements of LORAN time-differences. In conventional designs, a mochanization which accounts for the infinite variety of flight and aid data profiles possible would require an extensive amount of complex bookkeeping in the airborne computer, while being inherently sub-optimul.

The principal complication involved in employing LORAN is presented by any warpage error which exists between the actual time-difference grid and a predicted time-difference grid based on accurate computation using geodetic data. The warpage is due to unaccounted variations along the transmission paths between the LORAN stations and the aircraft, which affect the speed of propagation of the electromagnetic energy. Modeling of these errors is highly complex, as the changing propagation paths continually introduce new conductivity and reflecting surface variations. In the fast-reaction mission, the effect of grid distortion can be minimized by accepting an initial system position at the airfield which is based on actual time-difference data. This approach results in corrections to the system which are degraded only by the changes in distortion which occur between the airfield and point of initin) use of LORAN measurements. Consequently, if the distortion correlation distance is large with respect to distance traveled before the LORAN is used, system performance loss due to grid warpage modeling error will be small. Experiments have indicated that, if scourate stochastic models of the warpage phenomena are employed, only small corrections to the grid distortion result. Consequently, a LORAN-inertial system provides geodetic position information whose accuracy is limited by the LORAN time-difference grid distortion. Accurate maying the one still occur, however, when the aircraft and destination positions are specified in terms of measured time-difference coordinates.

For the performance curves shown in Figures 18-20, the LORAN time-difference data were assumed to be contaminated by first-order, 50 second correlated noises with r.m.s. values of 0.14 micro-seconds and white noises with r.m.s. values of 0.005 micro-seconds. The LORAN time-difference data was used to make a system correction every 10 seconds, these updates being initiated 60 seconds from the take-off time. In addition, constant bias or warpage error in the time of propagation from each transmitter to the sirdraft are assumed in two of the cases shown. The r.m.s. Value were 1 and 5 micro-seconds. In these cases, the system position was not initialized using the LORAN time-difference data with the result that highly degrading updates to platform azimuth alignment and domputed 6 stem velocity are made when the LORAN time-d fferences, error is to the transmission path biases appears in the computed system position. Except for the scanges in geodetic position errors due to the changing effects of the time-difference biases as the vehicle movie in the LORAN coverage area, the subsequent observation errors are caused primarily by the other system error surres. Consequently, with further updating, computed vehicle velocity and platform attitude error are brought under control as indicated in the figures.

The geometry involving an assumed LORAN chain of three transmitters and the aircraft profile is indicated in Figure 11. The master station is at  $13^{\circ}$  42' latitude and  $101^{\circ}$  6' longitude, the two slaves are at  $18^{\circ}$  18' and  $8^{\circ}$  45' latitude and  $90^{\circ}$  30' and  $108^{\circ}$  40' longitude, respectively. As the aircraft travels through the coverage area, the capability of deducing position from the two time-differences varies due to changes in the gradients relating time-difference error to geodetic position error. In this example, position can be determined to a r.m.s. precision of 1712 ft/meso at the airfield and 3901 ft/meso at the mission termination point for independent r.m.s. errors of a micro-mecond in both time-differences.

#### 7.9 Astro-Inertial Navigation Mode

Hystem performance in the astro-inertial navigation mode is limited by the capability of the tracker to lock onto a light source in the amblent noise environment and the accuracy with which the tracking axis can be related to the local geographic axes. The tracking precision is dependent on electronic component noise of the tracker and the brightness of the star relative to its immediate background. The accuracy in determining the anguing rientation of the tracker axis with respect to the local geographic coordinates is dependent on the actuacy of the mechanical alignment of the tracker gimbal axes to the reference inertial platform axes and to the accuracy of the alignment of these platform coordinates to the geographic frame.

In the fast-reaction mission, the dominant system error at the first star shot is the platform azimuth misalignment. This error is very observable at this time, as the error in computed system position and error associated with the tracker are relatively small. Consequently, an accurate alignment of the platform azimuth is achieved with the first system correction. Subsequent system corrections serve primarily to bound the three platform misslignment angles and the errors in computed system position.

Significant errors can be made in conventional astro-inertial system mechanizations, as these mechanizations usually assume that the platform reference coordinate system is locally-level. This assumption results in erroneous updates to computed position and platform azimuth to offset any existing tilt. An arc-second of position error oreated by an ard-second of platform tilt is approximately 100 feet. In a Kalman error controller design, the contribution of tilt to observed star position error can be modeled and accounted for properly in system updating. The sensitivity between attitude error and system position error implies, however, that tilt inducing errors require adequate modeling in the Kalman design. The main advantage of a Kalman mechanization in astro-inertial mavigation systems is the automatic accommodation of the infinite variety of star-to-vehicle position relationships possible. Data resulting from each star shot in an arbitrary sequence can be optimally employed, as consideration is automatically made for the residual error coincident with the optical axis and the destermined.

253

(O)ĭ

12 (1) 3/4 ( ).Ye

14-11-2-24

For the performance curves shown in Figures, di=23, the star elevation and bearing angle measurements are contaminated by a first-order correlated noise with r.m.s. value of 5 arc-seconds and a correlation time of 4 hours. In oddition, a white noise of r.m.s. value 2 arc-seconds was assumed. Two stars were considered in making the instem corrections. These stars had elevation angles of 30 and 60 degrees and bearing angles of zero and 120 4 wees, respectively. The shots alternated between the two stars, being made every 60 seconds and initiation 60 seconds after take-off. In Figures 21-23, selected curves from the preceding sets of figures have been displayed to permit convenient comparisons of system performance.

#### 8. CONCEDSION

This chapter has considered the application of Kalman filtering theory to error control in multi-mensor navigation mystems by detailing a particular error control design. The reader has been introduced to the busic -spea of existing navigation mensors and how their complementary characteristics have been exploited in the past in sugmented inortial navigation sensors. The practical aspects of implementing the much more sophisticated Kalman error controllers were discussed as well as practical synthesis techniques which have been employed to realize complicationally feasible designs. The design example which was considered illustrated system performance in the principal navigation modes to portray the particular characteristics of the navigation mensor sugmenting the inertial system.

The state-of-the-art in the application of highly-nomplex data processing techniques, such as Kalman filtering theory, is rapidly changing. The use of these sophisticated procedures is limited by the capability of digital computation equipment available for the synthesis of appropriate designs and the mechanization of these designs in actual systems. Limitations in some applications exist owing to the lack of information adequately identifying and describing all consequential system error characteristics.

As the cost of computation decreases and more information defining system error behavior becomes available, more marked improvements in navigation system accuracy can be achieved with existing memors. A relaxation of individual sensor accuracy requirements pormitted by more complex software design will, in turn, permit simpler sensor design. Mensor simplification coupled with natural design improvements can result in less costly, lighter, more reliable and more easily maintained sensors. These more sophisticated error control techniques imply a shift from complexity in hardware to complexity in software, resulting in navigation system dosigns with higher accuracy and reliability, simpler maintenance, and reduced size, weight and cost.



MARIA PORTA

### REFERENCES

- Kalman, R.E. New Methods and Results in Linear Prediction and Estimation Theory. Research Institute for Advanced Study, Baltimore, Maryland, Tech. Report 61-1, 1961.
   Bryson, A.E. Jr Linear Filtering for Time-Varying Systems Using Measurements Containing Colored Johansen, D.E. Institute of Electrical and Electronic Engineers, Transactions on Automatic Control, Vol. AC-10, Jan 1965, pp. 4-10.
- Bryson, A.E. Jr Estimation Using Sampled Data Containing Sequentially Correlated Noise. AIAA Henrikson, L.J. Guidance, Control and Flight Dynamics Conference, Aug 1907.
- Rishel, R.W. Convergence of Sampled Discrete Optimum Filters. Published in the Proceedings of the Joint Automatic Control Conference, June 1968, pp. 863-870.
  - Study and Analysis of Selected Long-Distance Navigation Techniques, Vols 1 and 11. Institute of Salence and Technology, University of Michigan, Dec 1962.
- 6. Bellantoni, J.F. A Square Root Formulation of the Kalman-Schwidt Filter. AIAA Journal, Vol.5. Dodge, D.W. No.7, July 1967, pp. 1309-1314.
- 7. Meditch, J.S.

5. O'Day, J.

at al.

Sub-Optimal Linear Filtering for Continuous Dynamic Processes. Aerospace Corporation, Los Angeles, California, Technical Report TOR-469(0107-35)-2, July 1964.

Systems Analysis and Mechanization Studies, DNL Navigation System Definition

Appendix J. US Air Force Contract AF33(615)67C-1184, Wright-Patterson Air Force

Final Engineering Report LNWN G/D Navigation System Integration. US Air Force

- 8. Huddle, J.R. et al.
- 9. Banbrook, H.W. et al.

10. Klementis, K.A.

11, Dworetzky, L.H.

Edwards, A.

Friedland, B.

et 11.

12, Richman, J.

Contract AF(657)-14389, Itema 2 and 3, Wright-Fatterson Air Force Base, Dayton, Chio, Jan 1966.

Bacs, Dayton, Chio, Oct 1967.

Synergistic Navigation Systems Study, Phases I and II. US Office of Naval Remearch Contract N00014-66-C0192, Oct 1966 and Nov 1967.

Principles of Doppler-Inertial Guidance. American Rocket Society Journal, Dec 1959, pp. 967-972.

Design of Mixer-Filter for Aircraft Navigation Systems. Published in the Proceedings of NAECON Conference, Dayton, Ohio, 1967, pp. 429-438.

 Priest, D.T.
 Optimisation of a Celestial-Inertial Navigation System Using Single Body Tracking, PhD Dissertation, Iowa State University, 1963.

14. Juhler, J.R. Berry, L.A. LORAN D Phase Corrections Over Inhomogeneous, Irregular Terrain. ESSA Technical Report IER 59-ITSA 56, Nov 1967.



ALL AND AND A DAY



18 Whether Didlet



(**O**))

1. 46 10. 1. 1

ALMESTER -

State Consultations

Fig.1 Earth-fixed and local geographic reference coordinate systems



.





 $\sqrt{2}$ 



Fig.1 Linear error model of level channels of Schuler-taned, month-slaved, free-inertial system

41.









Fig.6 Linear system error model serving as basis for Kalman error controller



Fig.7. Schematic representation of system error propagation with discrete update







And reasonal And

And Ottomation Outline





Fig.8 Sobematic diagram of basic operations required to implement Kalman feedback error controller



Fig.9 Schematic information flow in system employing feedback error control

الم جود م

A \* 4 \* 4 \*

. •

. . . . .

261









Louis Constant Constant





· "# ·







 $\bigotimes_{\lambda=99^{\circ} 30^{\circ}}^{\bullet=18^{\circ} 18^{\circ}}$ 





Land State

A Row Mark

1

Fig.11 Schematic illustration of aircraft flight profile and LORAN transmitter locations




1999991 (1999356) (1999369) (199956)

×.









And a market with



「「「「「「」」」

. . . . . . . . . . .

conducts which have been as



Milou-



## Fig.18 Position error CEP versus time. Loran-inertial mode



Fig.19 Velouity error CEV versus time. LORAN-inertial mode





Q

E C. N. S. S. S. S.

Å,

ð

3 1555-555 - 9227-555 - 1329-952 - 14645-55





. ری ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰

# CHAPTER 12 - APPLICATION OF KALMAN FILTERING TO BARO/INERTIAL HEIGHT SYSTEMS

17.13D

i.

X

loľ

A STATE OF STATE

Little - Wayne

## by

P.N. Barbam and P. Manville

## Avionic's Department, Royal Aircraft Establishment, Farnborough, Hampshire, England.



#### CHAPTER 19 - APPLICATION OF KALMAN FILTERING TO BARO/INERTIAL NEIGHT SYSTEMS

#### P.M.Barham and P.Manville

## 1. INTRODUCTION

The aviator has had no alternative, until recently, to the use of barometric pressure as the source of his height and height rate information. Despite the ingenuity of the designer, which has resulted in the development of some extremely sensitive and accurate instruments, the fact remains that the use of pressure necessitates ansumptions about the structure of the atmosphere in order to derive height or height rate. This has led to the concept of a standard atmosphere (e.g., ICAN). Errors can occur due to variations in the lapse rate, the height of the tropopause and in sea-level temperature and pressure. Even though a datum is established before take-off, this means that climbing or diving through a non-standard atmosphere will result in height rate errors of up to 8%, with corresponding errors in height. The two "actuals" shown in Figure 5 were recorded on the same day (ist March 1967), one over Cornwall and the second 600 nautical miles further north, over the Shetland Tales. In each case the pressure error is seen to correspond to a mean slope (of 0.055 and 0.015 respectively) with a random noise superimposed. With variations in the sign and magnitude of the mean slope, these are typical records. Effects such as lag and hysteresis make it very dangerous to rely on height rate accuracy at times when height rate itself is changing rapidly (such as during an overshoot when landing).

It was natural therefore that, with the advent of practical inertial platforms, the avionics system designer should have looked with interest to see what help he could get from the vertical inertial channel. Unlike the horisontal channels, which are comfortably bounded by the Schuler loop, the vertical channel is unstable, so that any accelerometer errors will cause the height error to increase as the square of time. Moreover the effect of gravity on the vertical accelerometer has to be cancelled. Since this correction is height-dependent, an error in assumed height will result in an acceleration error in the sense which will increase the original error.

However, it is easily seen that barometric and inertial height systems are largely complementary to each other. The former provides good height rate information in nearly level flight (subject only to the slope of the isobars in the vertical plane, which is typically 1 foot per mile standard deviation), but is poor in climbs or dives or in the presence of significant vertical acceleration. On the other hand the inertial system needs to be bounded by an external reference (such as barometric data) for sustained periods, but provides direct information about vertical acceleration and a good short-term reference for use during climbing or diving.

The use of such baro/inertial mixed systems is considered in more detail below.

#### 2. TRADITIONAL BARO/INERTIAL MIXING

The operation of a Kalman filter in a baro/inertial height system cannot be fully appreciated without first considering the more traditional mixing systems and their limitations.

#### 8.1 Fixed Gain Systems

The basic mixing loop is shown in Figure 1. There is no drift compensation for the inertial channel and, assuming that the system is trianed before flight, the dominant error at high stitude would probably be poor "g" compensation due to the use of erroneous height. In a closed-loop climb inertial height  $(h_1)$  will be forced to follow changes in pressure height  $(h_2)$  with resultant height rate errors of up to 8%. On the other hand, during an open-loop climb the system would soon become unstable because of the uncompensated acceleration errors. The transient which would normally cour on re-closing the loop could be reduced by a "fast re-set" phase, in which the justification  $K_1$  is temporarily increased.

Addition of drift compensation, as shown in Figure 2, would improve the open-loop climb as the inertial channel would now start in a well trimmed state. However, even with a fast re-set the system cannot settle without a lengthy transient on re-closing the loop as the act of forcing  $h_i$  to equal  $h_p$  will necessitate additional drift compensation to accommodate the new error in "g" correction.

Adding a pressure height correction loop, as shown in Figure 3, is a big improvement as it removes the transient on re-closing the inertial loop and is making an attempt to operate in "true" height. It operates in two different states based on the following assumptions:

(a) In level flight the barometric data gives perfect height rate (ignoring slope of isobars) and can be used to trim accurately the inertial channel. PRECEDING PAGE BLANK



(b) In a short period of climbing, the inertial channel may be considered perfect and used to determine the error in change of height as indicated by the barometric unit.

in practice some difficult compromises are necessary. Perfectly level flight can never be achieved and, to keep the inertial channel bounded, it is necessary to remain in the closed-loop state up to moderate height rates, in which case the pressure height rates can have significant errors. If, on the other hand, the switching takes place at lower height rates, the inertial channel could be left open-loop for excessive periods during slow climbs, with resultant instability.

#### 2,2 Variable Gain Systems

The above considerations laad one to examine the possibility of using a variable gain system, as shown in Figure 4. rather than direct switching between the third and fourth integrators. Both  $K_1$  and  $K_2$  can be made height-rato-dependent. In level flight  $K_3 = 1$  and  $K_4 = 0$ , so that the system becomes equivalent to the system in Figure 2. With increasing height rate  $K_4$  will increase and  $K_5$  will decrease. The error signal is now distributed between the inertial and barometric systems according to the values of  $K_5$  and  $K_4$ . This is similar to the sulti-dimensional correction to the state vector after a measurement with a Kahman filter, and the values of  $K_5$  and  $K_4$  are being used in a similar way to the appropriate terms in the gain matrix. Despite the increase of complexity the system in Figure 4 is still very imperfect compared with the Kahman filter described in Section 3.

With a Kalman filter the error between forecast and measurement results, in general, in corrections being fed back to all the elements in the state vector in accordance with the current values in the variance matrix. This is roughly analogous to  $K_1$  and  $K_2$  in Fibure 4 also being made variable. Of greater importance is the fact that, with the Kalman filter, the terms in the variance matrix reflect the previous history of the system and are not simple functions of height rate. For example, if a prolonged period of level flight had resulted in a very well trimmed inertial system, and hence correspondingly small values of variances associated with it, almost the whole disciplancy arising at the start of a climb would be attributed to an abnormal atmosphere, regardless of height rate. Finall, the filter described below has a built-in capability to learn the structure of the atmosphere and make shortterm use of this knowledge.

#### 3. THE KALMAN FILTER HEIGHT SYSTEM

#### 3.1 The State Vector

Since the purpose of the filter is to control the vertical channel of an inertial navigator it is obvious that the state vector should include terms representing the best estimates of the errors of inertial acceleration ( $\epsilon_{i}$ ), velocity ( $\epsilon_{i}$ ) and height ( $\epsilon_{i}$ ). Measurement will consist of a comparison of inertial and pressure height, so a term representing error in pressure height ( $\epsilon_{p}$ ) must be included. With the subscripts i, p and t representing inertial, pressure and true values respectively, these are defined as follows:

$$\mathbf{c} \mathbf{a}_{1} = \mathbf{a}_{1} - \mathbf{a}_{1} \qquad (3.1)$$

$$\mathbf{c} \mathbf{v}_{1} = \mathbf{v}_{2} - \mathbf{v}_{1} \qquad (3.2)$$

$$\mathbf{c} \mathbf{b}_{1} = \mathbf{b}_{1} - \mathbf{b}_{1} \qquad (3.3)$$

$$\mathbf{c} \mathbf{b}_{1} = \mathbf{b}_{1} - \mathbf{b}_{1} \qquad (3.4)$$

To enable the filter to learn something of the departures from standard of the atmosphere, a further element  $(\delta)$  is included in the state vector. This is defined by the relationship

 $\dot{\mathbf{h}}_{\mathbf{p}} = \dot{\mathbf{h}}_{\mathbf{t}}(1+\delta) \tag{3.5}$ 

and is, in effect, the slope of the plot of pressure height error against true height The complete state vector thus becomes

 $X = \begin{bmatrix} \mathbf{c} & \mathbf{h}_{i} \\ \mathbf{c} & \mathbf{v}_{i} \\ \mathbf{c} & \mathbf{a}_{j} \\ \mathbf{\delta} \\ \mathbf{c} & \mathbf{h}_{p} \end{bmatrix}.$ (3.6)

#### 3.3 The Transition Matrix

Examination of the chosen state vector shows that the first three terms are concerned with the open-'cop inertial mystem, and the last two terms with the atmospheric model. These two sets only interact as a result of the measurement process which, of course, builds up a strong correlation between inertial and pressure height estimates. However, in developing the transition matrix, which controls the updating process by the equations of motion, the two cets can be considered separately.















113m 🗰 Hilfer Hora 💼 1010 1

1144

Now the classical equations of motion for a uniformly accelerating mystem without any feedback can be expressed

$$\begin{bmatrix} \varepsilon & h_{i} \\ \varepsilon & v_{i} \\ \varepsilon & s_{i} \end{bmatrix}_{k+1} = \begin{bmatrix} 1 & t & \frac{1}{2}t^{2} \\ 0 & 1 & t \\ 0 & 0 & 1 \end{bmatrix} \begin{bmatrix} \varepsilon & h_{i} \\ \varepsilon & v_{i} \\ \varepsilon & s_{i} \end{bmatrix}_{k}$$
(3.7)

where t is the time interval between state k and state (k+1), and must be chosen so that the acceleration error may be considered constant during one step. However, in the vertical channel of an inertial navigator there exists an additional feedback. A height error in calculating the value of "g" will give an acceleration error of  $3 \times 10^{-6}$ ft sec<sup>-2</sup> for every foot of height error. If  $\epsilon_{0}$  is the acceleration error from all causes other than erroneous "g" compensation (i.e. untrimmed accelerometer drift) which may be considered constant between two successive steps, then the total error is

$$\epsilon_{i} = \epsilon_{i} + (3 \times 10^{-6}) \epsilon_{i}$$

Considering now the change in  $\in a_i$  during one cycle, we have

 $\Delta(\boldsymbol{\epsilon} \mathbf{n}_{i}) = 3 \times 10^{-6} \Delta(\boldsymbol{\epsilon} \mathbf{h}_{i})$ 

=  $3 \times 10^{-6} (t \in v_1 + \frac{1}{2} t^2 \in x_1)$ 

=  $(3 \times 10^{-6} t) \le v_i + (1.5 \times 10^{-6} t^2) \le u_i$ .

Hence

...

$$(\boldsymbol{\varepsilon} \ \boldsymbol{\mathbf{a}}_{i})_{k+1} = (\boldsymbol{\varepsilon} \ \boldsymbol{\mathbf{a}}_{i})_{k} + \Delta(\boldsymbol{\varepsilon} \ \boldsymbol{\mathbf{a}}_{i})$$

$$= (3 \times 10^{-6} t) (\epsilon v_{i})_{k} + [1 + (1.5 \times 10^{-6} t^{2})] (\epsilon a_{i})_{k} . \qquad (3.8)$$

The transition matrix for the inertial channel thus becomes

$$\begin{bmatrix} 1 & t & \frac{1}{2}t^2 \\ 0 & 1 & t \\ 0 & 3 \times 10^{-4}t & \begin{bmatrix} 1 + (1.5 \times 10^{-4}t^2) \end{bmatrix} \end{bmatrix}$$
(3.9)

Of the remaining terms in the state vector S should ideally have both a time and a distance correlated exponential decay, since it is obvious that any data on atmospheric structure would become of little use after a prolonged period of time, or at a distant location. However, in view of the difficulty of modelling a correlation with distance it was decided to use only an exponential time correlation with an arbitrarily chosen 10,000 second time constant. This is represented as

$$\delta_{k+1} = (1 - 10^{-4} t) \delta_{k} . \qquad (3.10)$$

From the definition of  $\delta$  in Equation (3.5) and of  $\epsilon$  h<sub>p</sub> in Equation (3.4), we have that

 $\Delta(\boldsymbol{\epsilon} \ \mathbf{h}_{p}) = (\mathbf{h}_{t} - \mathbf{h}_{p})t$  $= -\mathbf{h}_{t}t\delta$  $= -\Delta\mathbf{h}\delta .$ 

where

$$\Delta \mathbf{h} = \int_{0}^{t} \mathbf{h}_{c} \, \mathrm{d}\mathbf{t} \,, \tag{3.11}$$

h, being the computed best estimate of height rate.

Thus

$$(\mathbf{\epsilon} \mathbf{h}_{\mathbf{p}})_{\mathbf{k}+1} = (\mathbf{\epsilon} \mathbf{h}_{\mathbf{p}})_{\mathbf{k}} - \Delta \mathbf{h} \delta_{\mathbf{k}}$$
(3.12)

Combining (3.9), (3.10) and (3.12), the complete transition matrix for the state vector at (3.6) becomes

		[1	t	<b>i</b> t*	0	0		
		0	1	t	0	0		
Φ	=	0	3 × 10 <sup>-4</sup> t	$[1 + (1.5 \times 10^{-6} t^2)]$	0	0	•	(3, 13)
		0	0	0	$(1 + 10^{-4}t)$	0		
		lo	0	0	∆h	1		



#### 3.3 The Measurement Matrix

The measurement matrix required for baro/inertial mixing is

 $M = \begin{bmatrix} 1 & 0 & 0 & -1 \end{bmatrix},$ 

This selects from the state vector the terms

$$i h_i - \epsilon h_p = (h_t - h_i) - (h_t - h_p)$$
  
=  $h_p - h_i$ .

## 3.4 Plant Noise

#### 3.4.1 Acceleration

Plant noise on acceleration can arise from a number of causes, of which the following are of most significance:

- (a) random changes in bias level, which typically occur every 10 minutes on average with changes with a standard deviation of about 10°<sup>5</sup> g;
- (b)  $g^2$  distortion (with a force feedback accelerometer) with a typical level of 2 × 10<sup>-5</sup> g/g<sup>2</sup> for a typical inertial quality accelerometer.

Considering each of these in turn, the random changes increase the variance by  $10^{-10}$  g<sup>2</sup> every 10 minutes, or by  $17 \times 10^{-10}$  ft<sup>2</sup> sec<sup>-4</sup> in one 10 second cycle.

The second cause is more difficult to assess. However, an examination of the acceleration power spectral densities for a typical strike aircraft enables us to derive some approximate figures. Assuming an anti-vibration mounting with a 10 c/s resonant frequency, and typical damping, the power spectral densities are first weighted by the square of the amplitude transmission curve for the mounting and then integrated over frequency to provide mean square acceleration power levels for typical flight conditions. These can be converted into equivalent bies levels by application of the assumed  $g^2$  distortion ocefficient of  $2 \times 10^{-8} g/g^2$ . It is found that, for the aircraft investigated, the highest level occurs while pulling high "g". This is followed in descending order by take-off and landing and then by high IAS dives at a factor of 20 down on the high "g" case.

We are not interested in the absolute level of vibration, or the resultant bias, but in the variance of the shange of bias per computing cycle (10 seconds). Consider a simple example of a flight which has, on average, one high "g" manceuvre every 10 minutes (with a bias level of  $10^{-4}$  ft  $so^{-3}$ ), the rest of the flight being straight and level (negligible bias). This will result in two changes of bias level for each manceuvre, or one change in every 30 computing cycles. This gives a variance of the change of acceleration bias per step of  $3.3 \times 10^{-10}$  ft<sup>2</sup> sec<sup>-4</sup>, or about 20% of that due to random changes. Similar consideration of a high IAS dive occurring on average once every 5 minutes results in a negligible variance of  $0.02 \times 10^{-10}$  ft<sup>2</sup> sec<sup>-4</sup>, which indicates that only the high "g" manceuvres need be considered. Allowing a small increase in variance to account for other conditions, the plant noise consists of the addition each cycle of  $21 \times 10^{-10}$  ft<sup>2</sup> sec<sup>-4</sup> to the variance of acceleration error.

#### 3.4.2 Departure from Standard Atmosphere (8)

It was assumed that, overall, the standard deviation of  $\delta$  was 0.03, although there is some evidence that a higher value should be chosen. In prolonged level flight a plant noise contribution is required to maintain this level, despite the decay term introduced into the transition matrix. For a 10 second cycle this term is 0.999, which results in the variance p of  $\delta$  varying as

## $p_{k+1} = (0,999)^2 p_k + n$ ,

where n is the plant noise contribution. If  $p_{k+1} = p_k = (0.03)^2$  in the steady state, it is required that  $n = 1.8 \times 10^{-4}$  during level flight.

During climbs or dives it is obvious that the probability of a change of slope of the atmosphoric error is increased. Analysis of typical measured atmospheres (see Figure 5) indicates an additional contribution to the variance of  $\delta$  of about 4.8 x 10<sup>-4</sup> [Å] for a 10 second cycle, where  $|\mathring{h}|$  is the modulus of the height rate in it sec<sup>-1</sup>.

The total plant noise to be added to the variance of  $\delta$  each cycle is thus  $(1.8 + 4.8|\hat{h}|) \times 10^{-6}$ .

#### 3.4.3 Pressure Height

On the assumption that all isobaric surfaces are horizontal planes, any uncertainty in pressure height must arise from a lack of knowledge of the true value of  $\delta$  during a climb or dive. This is allowed for by the plant noise on  $\delta$  described above. However, in practice the isobaric surfaces in any region will be slightly sloped (with a standard deviation of about 1 foot per nautical mile). The system has no means of determining such a slope, apart from making height fixes (which are discussed below). To implement such a system would require two extra terms in the state vector (northerly and easterly slopes) which would couple into the pressure height correction term through terms in the transition matrix involving displacements determined by the horizontal channels. This



(3.14)











## 



Anti Orifate Birth Ortental

would be directly equivalent to the coupling of  $\delta$  into the height error by the  $\Delta h$  term. In practice such a refinement is hardly practical in the absence of a continuous external measurement, such as vertical Doppler, and would not be worth the extra complication. It must be appreciated, however, that without such terms the predicted height and height rate errors and their variances are relative to the isobaric surfaces and not to a horizontal datum. To allow for changes in slope, however, a small amount of plant noise is added directly to the pressure height. This amounts to the addition of  $4 \times 10^{-6}$  ft<sup>2</sup> to the variance of pressure height at each computing cycle.

#### 3.5 Measurement Noise

The measurement noise is the inability of the system to determine the correct static pressure of the undisturbed stmosphere outside the aircraft. This is essentially peculiar to a particular system and would need to be carefully studied in bach practical case. However, for the purposes of this study, some general observations were made for use in mathematical modelling. Sources of error considered included

- (a) random instrument noise,
- (b) hysteresis effects,
- (c) instrumental lags,
- (d) effects of wind gusts.

The first three errors were studied for a typical modern force re-balance air data system. The random instrument noise was found to have a standard deviation of about 0.11 millibar. Although relatively large under some conditions, hysteresis and lag effects were found to be very reproducible and, as such, could be allowed for by modelling in the digital computer of a practical system. In this case only the statistics of the error in modelling need be considered.

The actual pressure changes occurring due to gusts are, in general, small. Peak values of typically 1 millibar change in 1 kilometre can be recorded during thunderstorms and about an order less in ordinary "gusty" conditions. Hence the "all time, all weather" standard deviation must be very small. It should be noted, however, that a poor design of pitot-static will result in air velocity changes during gusts being recorded as pseudo-pressure-changes.

For the purposes of mathematical modelling the overall standard deviation of measurement noise was assumed to be 0.12 millibar for |h| < 25 ft sec<sup>-1</sup> and 0.32 millibar for |h| > 25 ft sec<sup>-1</sup>. These were converted to height errors by the usual relationship of  $(0.61^2 \times 10^{-7} + 27.8)$  feet per millibar. The two conditions were introduced on the grounds that lag effects are more significant during climba and dives. In practice, of course, the measurement noise used could result from a study of the actual system.

#### 3.6 Height Fixes

たとうと言語などなどのであった。

For a height fix using, for example, a radio altimeter, the required measurement matrix is

 $\mathbf{M} = \begin{bmatrix} \mathbf{1} & \mathbf{0} & \mathbf{0} & \mathbf{0} \end{bmatrix}.$ 

This selects  $\in h_1$  from the state vector, which is compared with  $(h_r + h_g - h_1)$ , where  $h_r$  is the radio altimeter height and  $h_g$  is the height of the ground above datum. Since in the steady state the normal baro/inertial measurements result in a correlation between inertial and pressure height which approaches unity, the same correction will usually be made to both.

#### 4. THE MATHEMATICAL MODEL

Orthodox modelling was used, making use of gaussian weighted random numbers of appropriate standard deviations to gamerate discrete errors for the "real world" model. One point of interest to this particular case is the model used for the atmosphere. Examination of a number of recorded atmosphere error plots, two of which are shown in Figure 5, had suggested a fairly well defined mean slope in each case, with random noise superimposed. The latter could be approximated by a series of straight line segments connecting points with a mean separation in height of about 5000 ft and with a standard deviation departure from the mean slope of 100 ft.

A series of random numbers, with rectangular distribution in the range 0 to 1, were drawn until one of them fell within the range 0 to 0.1. This was taken to represent a change point, each number in the series corresponding to a height change of 500 ft. Thus, if the seventh number drawn was the first to fall within the range 0 to 0.1, this represented a change point at 3500 ft. A gaussian weighted random number with standard deviation 100 ft was then drawn to represent the departure of this change point from the line of mean slope. A series of change points with mean separation 5000 ft were thus generated and were joined to provide a model of the atmosphere. The full lines in Figure 5 show typical models generated in this manner (with mean slopes of 0.06 and 0.15 respectively). It is mean they compare well with the actual records.

Due to the much lower level and similar frequency of  $g^2$  distortion changes compared with random accelerometer changes, it was not thought worthwhile to model both effects separately. Instead the frequency of random changes was slightly increased to compensate. Thus, on each cycle, a random number in the range 0 to 1 was drawn. If this fell within the range 0 to 0.02 a gaussian weighted random change of standard deviation 0.00032 ft sec<sup>2</sup> was added to the "real world" acceleration error.

275

ALL AN PORT of A

i o markin o k

#### 5. REBULTS

Bowe typical results obtained with the mathematical model are shown in Figures 6-10. No results have been shown for short duration high rate clubbs or dives, as the errors arising from these are negligible, due to the high degree of reliance placed on the inertial channel. It should also be appreciated that reversals of height rate also produce reversals of the "forcing function", that is the erroneous height rate derived from pressure, so that the Kalman filter is easily able to separate inertial and barometric errors. This is an important feature which, unlike the systems described in Section 2, enables the inertial channel to be trimmed accurately during typical terrain-following profiles. Theoretically the filter can distinguish between pressure and inertial errors whenever the height rate changes at all.

For this reason long climbs at low rates have been illustrated as representing the worst case for the filter. Figure 6 shows the height rate error and standard deviation for a climb to 60,000 ft at 150 ft sec<sup>-1</sup> through an atmosphere with mean  $\delta = 0.05$  after on initial settling period of level flight. A further five minutes of level flight is followed by a dive at the same rate, levelling out at ground level. The standard deviation just exceeds 0.2 ft sec<sup>-1</sup> at the end of the demont, but rapidly drops to a negligible amount on regaining level flight. The source for this run is within the standard deviation ourvo and shows no systematic trends. The corresponding height error with its standard deviation is shown in Figure 7 and reveals a similar pattern, the standard deviation regeners as it represents an especially difficult one for traditional height systems.

As the worst case for the Kalman filter height unit is a very prolonged climb at a constant rate, a similar run is illustrated in Figures 8 and 9, with a height rate of only 50 ft sec<sup>-1</sup>, so that the ascent to 60,000 ft and the subsequent descent each lasted for 20 minutes. This results in a considerable increase in the variances of the inertial system so that the effect of the mean value of  $\delta$  (0.0b) shows as a definite systematic error build-up on both velocity (Fig.8) and height (Fig.9). It demonstrates a very important feature of the Kalman filter. During the climb the erroneous pressure height changes were able to affect the inertial system as the reliance that could be placed on the latter decreased. However, the Kalman filter builds up in the variance matrix a complete set of correlations in the off-diagonal terms. Thus, on regaining level flight, although no measurement of absolute beight is made, the filter is not only able to correct the inertial velocity error, but is also able to correct the height error which had arisen as a result of this velocity error.

Figure 10 shows the Kalman filter estimate of  $\delta$ , with the true value for comparison, during the run corresponding to Figures 6 and 7.

#### 6. CONCLUSIONS

The Kalman filter can provide a usoful tool for the optimal mixing of baro/inertial height information. For any practical system it is essential that the form of the plant and measurement noise should be carefully studied and modelled as accurately as possible. It is not claimed that the system described above is optimised, but it represents a useful starting point for the study of specific aircraft systems.









4 4 1 1,10







11.19 **1** • •





Basic baro/inertial mixing system Fig.1



Fig. 2 System with drift compensation



1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 -1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 -1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 -

語がある







Fig.4 Variable gain system











والمراجع وال

Fig.6 Velocity error (150 ft sec<sup>-1</sup>)

المراجع h Cu

Ľ.

Soil Adented

Marshall I

William Start Bart





# CHAPTER 13 - APPLICATION OF KALMAN FILTERING TO THE C-5 GUIDANCE AND CONTROL SYSTEM

語語の説

2

10 A

Ţ

R Marsh

A PROVIDENCE

itin 🗩 (ei linge

يد و فلا ا

## by

Stanley F.Schmidt\*, John D.Weinberg<sup>†</sup> and John S.Lukesh<sup>†</sup>

 Analytical Mechanica Associates, Inc. Palo Alto, California, USA

<sup>†</sup> Northrop Corporation, Electronics Division Hawthorne, California, USA

# **PRECEDING PAGE BLANK**

Ň

1.000

<u>،</u> ۱

#### NOTATION

ane 🖬

## Coordinate frames

177	orthogonal, earth-fixed reference frame
2 <b>P</b>	orthogonal, wander azimuth, local vertical, ideal reference fr
CP	orthogonal, wander azimuth, local vertical, computational fram
7NF	orthogonal platform reference frame
AIP	non-orthogonal, accelerometer input axis frame
01 <b>7</b>	non-orthogonal, gyro infut axis frame
opi	orthogonal, i <sup>th</sup> gyro input-output-spin axis frame

Operational and structural symbols

-	(superior arrow) three-dimensional vector
•	(superior dot) time differentiation
d dt	time differentiation
T	(supersoript) watrix transpose
-1 <del>i</del>	(superscript) matrix inverse
⊽	vector gradient
3	approximately equal
C-1	inverse Laplace transformation
•	Laplace variable
<b>E</b> []	ensemble expectation operation

## Navigational symbols

7	apecific force acting on ball
7,	7 in IP
7 <sub>p</sub>	Î in PRF
τ <sub>λ</sub>	f in AIF
Zei	ž in Gri
Ä	estimate of $\vec{I}_1$ in OF .
ī,	accelerometer outputs in AIF
ĩ	inertial acceleration of ball
õ	gravitational force (por unit mass) acting on ball
	acrodynamic force (per unit mass) acting on aircraft
Ŧ	thrust force (per unit mass) acting on mircraft
7 -	Doppler velocity vector
ſ	plumb-bob gravity
Ā,	Coriolis and centripetal acceleration estimate in CP













ar Dina tra Orient



······

		280	
ជ	earth rate in CP		
$\vec{v}_L$	aircraft level velocity with respect to earth in CF		
13	sircusft rate in CF		
λ	airoraft latitude (computed)		
ե	aircraft longitude (computed)		
0 <sub>pl</sub>	platform heading (computed)		
0.sa	ball asimuth readout		in the second
0 <sub>H</sub>	platform heading (indicated)		
V <sub>us</sub>	Aircraft groundspeed (domputed)		
0 <sub>01</sub>	aircraft ground track angle (computed)		1994 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 -
C	earth's curvature matrix (computed)		
W	angular rate matrix (computed)		
h	aircraft altitude (computed)		
V <sub>L</sub> -	aircraft vertical velocity (computed)		
₩e	commanded (uncompensated) gyro torquing rates in CP		
	commanded (compensated) gyro torquing rates in CF		
ũ,	gyro drift rates in GIF		
Ξ <sub>p</sub>	total inertial angular rate of platform in PRF		
T <sub>PISP</sub> .	IF to FRF transformation matrix		
TptA	PRF to AIF transformation matrix		J
Taisp	GIF to PRF transformation matrix		}
T <sub>Ptai</sub>	PRF to UF1 transformation matrix		
T	EFF to OF transformation matrix		
ā, ·	Rocelerometer bias errors in AIF		1.1.62
ä,	Rodelerometer noise errors in AIF		
a	accelerometer bias error calibrations in AIP		
(s <sub>ij</sub> ) .	movelerometer scale factor and input axis orthogonality error calibration matrix		A PROVIDENCE
≓ € <sub>b0</sub>	gyro fixed drift rate calibration in GIF		
(m <sub>13</sub> )	gyro scale factor and input axis orthogonality error calibration matrix		
[u <sub>ij</sub> ]	gyro mass unbalance error calibration matrix		
<b>82</b> 10	diagonal gyro scale factor error matrix		21. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1
57 A	diagonal accelerometer scale factor error matrix		
a <sub>i</sub>	simulator error generating function for ith gyro		
Kalman filter	aymbol a		
Y	observation vector		

augmented navigation state vector X

sa i

	• •	
286		· · · · · · · · · · · · · · · · · · ·
Ĩ,	augmented navigation error state vector	0
£	estimated sugmented navigation error state vector	le af list gain an an an Air
x	error in f	
¢	transition matrix for 1	
P	nevigation system error covariance matrix	
R	ngvigation system forcing error noise matrix	a trad a tradition of the second s
K,	Kalman gain: vector, 😑 b	
K∦ <sup>≥</sup>	epsilon gain vector	
Δ <b>y</b>	observation residual	a the second
M	observation matrix (discrete observation)	
¢	"epsilon" gain scalar	
C	observation noise scalar	: د به از مع ا ( مز مع از مع ا ا مع مع از مع
ts	start of Kalman cycle	والمسالية والمسالية والمسالية المسالية المسالية والمسالية والمسالية والمسالية والمسالية والمسالية والمسالية وا
t	end of Kulman cycle	
Δt	Kalman cycle time	
ti	time within a Kalman cycle	
n	dimension of t	
u <sub>so</sub>	Nystem control vector	
Î(Î)	estimate control algorithm	
ê(î)	system control algorithm	A share
Other symbols		
Z	navigation system state vector	
U	navigation system forcing function vector	
p	vector defined by Equation (2.1), as indicated in text	
H	navigation system description function	
v	vector defined by Equation (2.2), as indicated in text	TIME PANTAN MAR
Q	vector of random time-uncorrelated errors in the navigation observation	
8	function defined by Equation (2.2)	
P	sugmented navigation system description function	
u	vector of augmented navigation system white noise forcing functions	
Q	functional, time-dependent relationship between navigation observations and augmented navigation aystem state	
ī	aircraft position	
Z	simulator true position and velocity vector	
	vector of principal augmented navigation system errors	in the Ren tief berte Ratht
•1	vector of forcing sugmented navigation system arrors	
α	watrix of linearized • to • coefficients	

۰,

.

to **P**res

ドンシンシンシン オイト・シント しんしん



#### CHAPTER 13 - APPLICATION OF KALMAN FILTERING TO THE C-5 GUIDANCE AND CONTROL SYSTEM

#### Stanley F. Schmidt, John D. Weinberg and John S. Lukesh

#### 1. INTRODUCTION

Ņ

.

<u>ن</u>

The Inertial-Doppler Navigation System developed by Northrop for the USAF/LockFeed C-5 heavy logistics transport<sup>1</sup> is designed to provide an accuracy sufficient for dropping supplies and/or personnel at desired locations, in any kind of weather, anywhere on earth. The requirements of high accuracy and all-weather capability are satisfied by combining information from self-contained and ground-based navigation aids. The measurements from the navigation aids are processed by the on-board navigational computer which uses algorithms derived from Kalman filter theory<sup>2</sup>.

Baction 2 discusses the conceptual design of the overall navigation system, describes the system components, defines the system's functional capabilities and accuracy requirements, and discusses the development approach to the Kalman filter.

Section 3 formulates the mathematical bases for the damign and for the error evaluation programs. There follows a discussion of (a) the major constraints that were imposed on the design by the hardware implementation, (b) the alternate design solutions that were considered and evaluated, and (c) the final solutions adopted. At the completion of this phase, design definition was sufficiently complete for the detailed real-time programming of the computer to begin.

The "Design Evaluation and Refinement" phase (Section 4) required a detailed simulation program to validate the design and to develop refinements in problem areas. Because extensive system laboratory tests and some system flight tests had already been conducted at the time of writing, it has been possible to include in this section come comparative results between simulated and real system behavior. These comparisons desonstrate the performance capability of the design, and also show how the simulator was used to locate and correct trouble areas.

Section 5 is a briaf summary of the overall C-5 Kalman navigation filter development effort described in the earlier sections.

Finally, Section 6 takes a brief glance at future trends in the area of Kalman filtering application to realtime, digital computer implementation of augmented inertial and other types of navigational systems.

#### 2. SYSTEM DESCRIPTION, PROBLEM DEFINITION AND APPROACH

Development of the Kalman filter for the C-5 navigation system started with the following areas already more or less completely defined:

(i) The functional design of the overall navigation system.

- (ii) The functional capabilities and accuracy goals necessary to satisfy the various G-5 mission requirements.
- (iii) The detailed specifications on the performance characteristics of the various subsystems.

The central objective was to develop the algorithms to be used in the on-board computer which would enable the achievement of these functional and accuracy capabilities. Pursuit of this objective involved such attention to mathematical error models - both functional and statistical - of all of the subsystems. The constraints of on-board computer memory, word length, and speed were also of fundamental importance.

. Pp. 28.14 288. blick



**O**t

1 MAY SHO

1.11

#### 3.1 Fundamental Concepts

As will be seen, the C-5 navigation system in its primary modes of operation uses an inertial measurement unit to maintain a continuous knowledge of the position and velocity of the aircraft. It will also be seen that the Kalman filter is used to update the inertial navigation system when data from other navigation aids is available.

The following is a brief description of the fundamentals of inertial navigation systems and the practical application of Kalman filter theory to them.

## 2.1.1 Inertial Navigation Systems\*

Figure 1 is a highly simplified block diagram illustrating the fundamental operation of an inertial navigation aystem. The principal components, the inertial measurement unit (IMU) and the computer are shown.

The IMU contains a stable platform on which three gyros and three linear accelerometers are orthogonally mounted. As implied by its name, the stable platform maintains an orientation, communded by the computer, which is independent of the orientation of the case. This is accomplished, as indicated in Figure 1, by sensing error signals from the gyro output axis pick-offs and, through feedback, by providing the appropriate platform drive nulling torques. This "inner loop" feedback control system must have sufficient gain and dynamic response to keep the pick-off errors small, regardless of IMU case attitude motions.

If the inner loop functions properly, then the platform maintains the orientation dictated to the gyros by the computer. Each gyro, however, is not a perfect device, of course, and tends to exhibit a small "drift rate". That is, even when the gyro output axis pick-off is nulled, the gyro case (fixed to the platform) has a rotation rate about the input axis direction which differs slightly from the commanded rate.

These drift rates, and the unwanted output axis<sup>†</sup> torques which attend them, arise because of internal rotational unbalances in the gyro about its output axis, gyro output axis turquer scale factor errors, and from several other, more subtle, mochanisms. If the computer could somehow estimate the unwanted output axis torques, then their effect on platform orientation could be removed by applying appropriate angular rate compensating signals to the gyro output axis torquers.

As indicated in Figure 1, the accelerometer outputs are the measured specific force which is sent to the digital computer in the form of pulse trains. The specific force vector,  $\vec{T}$ , is given as

#### $\vec{I} = -\vec{i} + \vec{0}$

where  $\vec{a}$  is the inortial acceleration vector and  $\vec{G}$  is the gravitational attraction per unit mass. If the position is known,  $\vec{G}$  can be calculated.

Hence, it can be seen that the navigation equations perform the functions of accepting specific force and performing the other computations necessary so that position and velocity are continuously available.

To initialize operation, the system must be aligned so that the coordinate frame of the stable platform and computer are coincident. The initial position and velocity must also be known.

The inertial navigation equations used in the C-5 are referred to as a wander azimuth geodetic vertical mechanization. This means that: (a) the coordinate frame has two axes in the locally-level tangent plane, and (b) no attempt is made to maintain the level axes in a preferred azimuth direction (such as north). The platform is initially aligned by driving it with the torquers until the outputs of the two level accelerometers are more. Once it has thus been leveled, its azimuth relative to north is determined by monitoring the level gyro command rates, which are necessary to keep it level despite the rotation of the earth. The arc tangent of the ratio of these two rates then provides an estimate of the platform heading relative to north. This operation is called self-gyrocompassing. During subsequent operation, position and velocity data are computed and applied through the gyro torquers to maintain the platform locally-level despite aircraft motion across the surface of the serth.

#### 2.1.2 Kalman Filter Theory

Because of the various error sources in gyros, accelerometers, the initialization process, and navigation equation approximations, the position and velocity indicated by the inertial navigation system is in error. The error also grows with time after initialization.

Data which are functionally related to the true aircraft position and velocity can also be obtained from many other sources. As examples, barometric pressure provides an indication of altitude, Doppler radar provides

\* Reference 3 is highly recommended for those desiring further information.

<sup>1</sup> The output suis of a single-degree-of-freedom gyroscope is that axis (normal to the spin axis) about which the spinning rotor is free to rotate with respect to the gyro case. The input axis (normal to the spin and output axes) is therefore the direction about which the gyro is sensitive to rotational case motion with respect to inertial space.

والمراجبة والمراجبة والمتعار والمعار والمتعار والمعار والمعار والمعار والمعار والمعار والمعاد والمعاد والمعار والمعالية

a measure of ground speed, and so forth. Conceptually, it is desired to use this external data to update the inertial havigation system; unfortunately these data, too, are imporfact, because of measurement errors and other effects (e.g., the pressure-altitude relationship is non-constant).

Kalman filter theory provides a means by which algorithms can be derived for digital computer processing of such measurement data observations". The resultant algorithms are feasible for use with on-board computers in that there is no requirement for storage of large quantities of data; that is, each observation is processed sequentially as it occurs in time, and after it is processed the raw data can be overwritten with new data. As originally presented, the Kalman algorithms were restricted to:

- (a) Linear dynamic systems.
- (b) Observations linearly related to the state vector.
- (c) Observations occurring at equally spaced time intervals.

The practical<sup>†</sup> application of the theory requires removal of these three restrictions. Removal of the third restriction is trivial and is readily accomplished by stating the original algorithm given by Kalman in the two parts:

- (1) at an observation;
- (2) between successive observations.

Theoretical proofs which remove the first two restrictions for nonlinear systems are not available. Practically, however, one should recognize that almost all real systems obey approximately linear relationships in the vicinity of a given solution. As a result, a practical filter for nonlinear systems can be derived by appropriate use of variational equations about a solution which is "close to" the desired solution. The desired solution in the case at hand would be the true value of the navigation state vector (aircraft position, velocity, etc.) which the observations are used to estimate. If this solution were known there would be no need for the filter. Hence, since it is not, the most obvious second choice is linearization shout the best estimate of the navigation state vector, which is indeed available. If the solution converges at more observations are taken, the linearization approximation becomes more and more valid.

The algorithms are presented here in one of the many forms which are applicable to a general nonlinear problem\*\*. Following this, the overall implementation of the filter in the C-5 navigation system is discussed.

Given a system which obeys the differential equations

$$\mathbf{Z} = \mathbf{H}(\mathbf{Z}, \mathbf{U}, \mathbf{p}, \mathbf{t}) ,$$

where

Ţ

Z = normally defined state vector;

- U = vector of forcing functions; U may be composed of known functions of time and/or random functions of time;
- p = vector of constants whose values are required to give a unique set of Equations (2.1).

The vector observations of this system obey

$$Y(t) = S(2, V, t) + q(t)$$
, (2.2)

where

V = a vector of deterministic quantities (e.g., constants) necessary to uniquely define the equations:

q(t) = random, time-uncorrelated errors of observation<sup>††</sup>.

An augmented state vector X can now be constructed, viz.:

 $\mathbf{X} = \begin{pmatrix} \mathbf{U} \\ \mathbf{y} \\ \mathbf{v} \end{pmatrix}$ 

such that \*\*\*

## $\dot{X} = F(X, t) + u(t) .$

The word observation is chosen here to indicate a measurement used in the sense of Reference 2. This distinction is
made because not all measurements are processed in this sense, as will be seen later.

The first known study results of a practical application of the filter are given in Reference 4.

\*\* Several forms are presented in Reference 5.

Any time-correlated measurement errors can be defined as components of V. This also requires definition of differential equations for these components with white noise type driving functions.

\*\*\* These techniques are presented in more detail in Reference 5, although the notation there is somewhat different.



(2.1)

(2.3)

COLUMN TWO IS

:::::::

where u(t) = white noise (random forcing functions).

Almo,

11.62.5

ţ,

١.

#### Y(t) = Q(X,t) + q(t) .

#### 2.1.3 Modified Kalman Filter Equations

The modified Kalman filter for such a system and its observations is summarized by Equations (2.5) - (2.11), for which the following definitions and assumptions apply:

## (a) $\nabla_{\mathbf{x}}(x) |_{\mathbf{x} = \hat{\mathbf{x}}}$ denotes the gradient of the bracketed function evaluated at $\mathbf{x} = \hat{\mathbf{x}}$ .

- (b) t<sub>m</sub> denotes the observation time, t<sub>o</sub> denotes the time of the last change in the estimated state (which coincides with the time of the last observation).
- (c) The subscript a denotes "after the observation is included" and the subscript b denotes "before the observation is included". Hence, from Equation (2.7), at an observation time.  $P_b = P(t_m)$ , and  $P_m$  in Equation (2.9) becomes  $P(t_o)$  in Equation (2.7) when updating to the next observation time. Further comments (regarding multiple observations occurring at the wame time) are included in the subsequent discussion.

Between observations (time update)

$$\hat{\mathbf{x}}(t) = \hat{\mathbf{x}}(t_0) + \int_{t_0}^{t} \mathbf{F}(\hat{\mathbf{x}}, \tau) d\tau$$
 (2.5)

$$\frac{d\Phi}{dt} = \nabla_{\mathbf{X}}[\mathbf{F}(\mathbf{X},t)] \Big|_{\mathbf{X}(t) = \hat{\mathbf{X}}(t)} \Phi(t;t_0) \Phi(t_0;t_0) = \mathbf{I}$$
(2.6)

At observation time, t<sub>a</sub> (observation update)

$$P(t_{n}) = \Phi(t_{n}; t_{n}) P(t_{n}) \Phi^{T}(t_{n}; t_{n}) + R(t_{n}; t_{n})$$
(2.7)

$$\hat{\mathbf{x}}(\mathbf{t}_{1}) = (\mathbf{K}_{1} + \mathbf{K}_{2}) [\mathbf{Y}(\mathbf{t}_{1}) - \hat{\mathbf{Y}}(\mathbf{t}_{2})]$$
(2.8)

$$P_{a}(t_{H}) = P_{b} - K_{i}MP_{b} + K_{i}(MP_{b}M^{T} + C)K_{i}^{T}$$

$$(2.9)$$

$$\hat{t}(t_{m} + \tau) = \Phi(t_{m} + \tau; t_{m}) \hat{x}(t_{m})$$
(2.10)

$$\hat{X}_{a}(t_{a} + \tau) = \hat{X}_{b}(t_{a} + \tau) + \hat{x}(t_{a} + \tau)$$
, (2.11)

Equation (2.5) simply represents an integration of the differential equations,

Ŷ

using  $\hat{X}(t_p)$  as initial conditions.  $\hat{X}(t)$  is the best estimate of X(t), because the random function u(t) of Equation (2.3) is a mero mean white noise variable.

Equation (2.6) defines the differential equations for the transition matrix, which relates small deviations in  $\hat{X}(t)$ , that is,

$$\Phi(t;t_0) = \frac{\partial \dot{X}(t)}{\partial \dot{X}(t_0)}$$

By defining

then, approximately,

$$\widetilde{\mathbf{x}}(\mathbf{t}) = \Phi(\mathbf{t}; \mathbf{t}_0) \widetilde{\mathbf{x}}(\mathbf{t}_0) + \int_{\mathbf{t}_0}^{\mathbf{t}} \Phi(\mathbf{t}; \tau) \mathbf{u}(\tau) d\tau .$$

Hence, defining the covariance matrix  $P(t_o)$  by

$$P(t_a) = E[\tilde{x}(t_a) \tilde{x}^{T}(t_a)],$$

where E[] means the expected value of the bracketed quantity, and using the white noise definition of the random variables, u, it follows that:

 $P(t_{n}) = E[\bar{i}(t_{n})\bar{i}^{T}(t_{n})] = Equation (2.7) .$ 



(2.4)

 $R(t_{a};t_{o})$  in Equation (2.7) is the random uncertainty in the state caused by the random variable u.

Equation (2.8) defines the change in the state estimate resulting from the observation, and Equation (2.9) updates the covariance matrix P to include the effects of the observation.

In these equations:

$$\begin{split} Y(t_m) &= \text{ actual observation} \\ \widehat{Y}(t_m) &= G(\widehat{X}, t_m) = \text{ computed (estimated) observation} \\ Y(t_m) &= \widehat{Y}(t_m) &= \text{ observation residual} \\ & K_1 &= P_b M^T / (M P_b M^T + C) \\ & K_g &= M^T \epsilon / (M P_b M^T + C) \\ & K_g &= M^T \epsilon / (M P_b M^T + C) \\ & M &= \nabla_X \left[ G(X, t_m) \right] \Big|_{X=\widehat{X}} \\ & \epsilon &= \text{ control gain of epsilon technique} \\ & C &= E[q(t_m) q^T(t_m)] \; . \end{split}$$

The equations are for a single observation, that is, each observation type is sequentially processed.  $K_1$  is the conventional gain for the Kalman linear filter.  $K_2$  represents one type of fix [see References 7 and 8] for preventing divergence of the filter resulting from unmodeled and computational errors. In this chapter, the use of  $K_2$  is called the "spainon" technique.

K, causes overweighting of the most recent observation in Equation (2.8). This, in turn, causes an additive growth of the covariance matrix in Equation (2.9).

Because of the computation time required to carry out the calculations,  $\hat{X}(t_m)$  is only available after a small delay. Equation (2.10) therefore prodicts  $\hat{X}(t_m)$  ahead to time  $t_m + \tau$ , where  $\tau$  is larger than the computation time. When real-time reaches  $t_m + \tau$ , Equation (2.11) is used to modify the total system state estimate.

## 2.1.4 C-5 Navigation System Considerations

From the preceding discussion on fundamentals of inertial navigation systems, a method of implementing the Kalman filter in a navigation system can be recognized. This is to use the basic inertial navigation system in the role of solving Equation (2.5). To facilitate this, the components of the state vector X(t) were chosen to include quantities (such as position and velocity), estimates of which are carried in the digital computer registers. Also included are the attitude of the inertial platform of the IMU, gyro drift rates, accelerometer biases, measurement errors, and so on. This augmented state vector then contains all the quantities notations.

The u(t) of Equation (2.3) can be interpreted as random forcing functions in the inertial system or in the measurement devices which prevent perfect time updating. Physical evidence has shown that gyros and many other real devices have errors which vary with time and/or other environmental factors. Hence, this random forcing function concept provides a means of including such effects in the filter algorithms.

Figure 2 is a block diagram depicting the conceptual real-time implementation of a Kalman filter, in accordance with the above standpoint, in an on-board navigation computer. In particular, the sequence of operations involved in a single cycle of the on-board filter is shown. Although the figure is largely self-explanatory, it should be noted that careful attention to sequencing these operations, so as to minimize or eliminate possible harmful control lage, is inherent in the diagram.

Details of the Kalman filter which is mechanized in the C-5 navigation system are discussed in later sections.

#### **3.2** Navigation System Description

The C-5 navigation system consists of the Inertial-Doppler Navigation Equipment (IDNE) and a number of suxiliary navigation sids, manufactured by several sub-contractors. Northrop Electronics Division is the system integrator as well as the supplier of major equipment items, including the IMU and the on-board digital computers. Figure 3 is a simplified block diagram of the system'. As shown in the block diagram, a high degree of redundancy was incorporated in the design to enhance overall reliability. The two on-board digital computers are the heart of the basic system. The navigator's control and indicator panels enable him to decide what mode to select, what navigation information to use or reject, and so forth.

Footnote: The measurement dimension is assumed here to be  $1 \times 1$ . This slices use of scalar division, rather than the matrix inversion required in the more general case of an  $m \times 1$  measurement.

\* Reference 1 gives some additional details.











the second s

#### The Digital Computer

The computers shown in Figure 3 are Northrop Electronics Division general-purpose, parallel operation digital computers (NDC-1051A). The word length is 23 bits. The memory sizes are 12,288 and 8,192 words, respectively, for the primary and auxiliary computers. Other characteristics are as follows:

Memory type:	Random access core, non-volatile DRO			
Instructions:	51, with two instructions per word and half-word arithmetic capabilities			
Add time:	Half-word 6 microseconds Full-word 8 microseconds			
Multiply time:	Half-word 28 microseconds Full-word 74 microseconds			
Other features:	Indexing Indirect addressing Roll table (push-down stack) Interrupt capability.			

#### The INU

The IMU (Inertial Measurement Unit), shown in Figure 3, is the Northrop floated ball, gimballess platform. Three Northrop single-degree-of-freedom floated, rate-integrating gyroscopes are orthogonally mounted inside the floated ball. Each gyro error signal is detected and amplified. The amplified signals are applied to ballmounted torquers which cause the ball to rotate. This high-gain, three-axis, feedback attitude control system rotationally isolates the floated ball from the case, maintaining a stable platform to a high degree of accuracy. Each gyro bas an input torquer whose command signals are controlled by the digital computer. Hence, the platform attitude relative to a given reference frame can be changed by quantities computed by the digital computer.

Three Kearfott linear pendulous accelerometers are also orthogonally mounted inside the floated inner ball. The accelerometer-sensed signals are sent to the digital computer in the form of a train of pulses. The number of pulses in a given unit of time divided by the time interval is a measure of the average specific force exerted on the IMU case over the interval. The frequency of the pulses for the C-5 system is high enough for this average specific force to be considered as a continuous quantity, for all practical purposes.

The INU also contains an electrostatic, three-axis, spherical angle measuring system. This "ball readout" provides a measure of the orientation of the platform case (outer ball) relative to the stable member (inner ball).

#### Doppler Radar

The Doppler radar, manufactured by the GPL Division of General Precision Systems. Inc., supplies the computer with measurements of ground speed and drift angle\*. By utilizing information from the "ball readout" mentioned previously, it is possible to obtain the ground speed relative to platform axes.

#### LORAN

The LORAN receiver is manufactured by Collins Radio Company. When available, the LORAN data is converted by the digital computer to position information and displayed to the navigator. Subject to his judgment and control, the LORAN data can also be used to update the best estimate of the state vector which includes position, velocity, and other quantities.

Ter Starte

#### TACAN

The TACAN receivers, manufactured by Hoffman Electronics, provide slant range and bearing information to ground stations at known locations. The navigator can use this data, when it is available, in a similar manner to that described for the LORAN.

#### Multi-mode Radar (MMR)

The multi-mode radar, manufactured by Norden Division of United Aircraft, provides many modes of operation for the C-5. Two modes are used to obtain navigation information. In the terminal phase of flight, the navigator can, through a cathode ray tube display, sight on landmark patterns whose position relative to desired supply and/or personnel drop location is known. Slant range and bearing data is provided by the MMR during the terminal phase. The navigator can also use the MMR for sighting on landmarks whose absolute position (latitude and longitude) are known. In this mode the measurements of range, bearing, and landmark location go directly to the Kalman filter.

\* The drift angle is the angle butween the ground speed vector and the aircraft longitudinal axis.

## Attitude-Heading Reference Unit (AURU)

The two redundant attitude and heading reference units, shown in Figure 3, are manufactured by Lear Siegler. Data from these units is used to provide a back-up mode of operation in case of an IMU malfunction.

#### Central Air Data Computer (CADC)

「「あっていていた」

9

The two air data computers are manufactured by Elliot Brothers. One of the functions of the CADC is to provide barometric pressure altitude data to the navigation system. (See Reference 1 for other functions.)

#### Inertial-Doppler Navigation Equipment

Figure 4 is a functional block diagram of the C-5 Inertia)-Doppler Navigation Equipment, which performs the navigation function in the primary mode of operation. This major subsystem of the navigation system is supplied by Northrop Electronics Division. It contains elements organized and located as shown in Figure 4. The Doppier antenna and HUU are in the nose visor of the aircraft. The electronics rack contains the A-D/D-A (analog-todigital and digital-to-analog) converters, the digital computers, and the Doppler signal data converter. The control and indicator panels are in the navigator's area.

Before installation in the aircraft, a tape reader, which is part of the auxiliary ground equipment, is used to load the real-time computer program and associated corstants into the memory of the primary and/or auxiliary computer. The mavigator also has access to some parts of the computer memory through a keyboard input.

#### 2.3 Navigation System Functional Capabilities and Accuracy Goals

#### 2.3.1 System Operating Modes

As can be seen from the discussion relative to Figure 3, the C-5 navigation system has numerous equipment redundancies to enhance overall system reliability. In addition, the navigator can select any one of four basic navigation modes, depending upon equipment availability. A brief description of these primary modes is now given.

Free Inertial This mode uses the IMU, digital computer, and barometric pleasure measurements of altitude.

Dappler-Inertial This mode adds the Doppler measurements to these used in the free-inertial mode.

Doppler Dead-Reckoning In case of IMU failure, if the Doppler and AHRU equipments are available, this -mode can be selected.

True Airspeed Dead-Reckoning If both the IMU and the Doppler radar fail, AHRU and airspeed data from the CADC can be used to continue dead reckoning.

In any one of these basic modes, position data from LORAN, TACAN, multi-mode radar, or visual information can also be used to update the best estimate of the state vector.

In addition to these navigation modes, there are also the platform alignment (ground or air) modes. Here, as in the navigation modes, the Kalman filter is used for weighting the various available data to continually optimize navigation and platform alignment.

Finally, a special terminal navigation mode uses multi-mode radar data to attain the high relative-position accuracy required for determining the computed air release point (CARP) in cargo or personnel drops.

#### 2,3,2 IDNE Performance Goals

The design performance goals for the C-5 navigation system are those defined for system initialization and navigation in the Doppler-inertial and free-inertial modes, as follows.

(a) Ground alignment (-75° to +75° latitude):

- (1) Warm-up from -65°F and alignment after warm-up, both in 25 minutes.
- (2) Initial position uncertainty: 1.0 nautical mile (30).
- (b) Airborne alignment (-75° to +75° latitude):
  - (1) Alignment after warm-up in 20 minutes.
  - (2) Initial position uncertainty: 5.0 nautical miles (30).

(c) Navigation (free-inertial and Doppler-inertial):

- (1) Ground alignment first 5 hours: 0.75 nautical mile per hour CEP, - accord 5 hours: 1.25 nautical miles per hour CEP.
- (2) Airborns alignment first 5 hours: 1.0 mautical mile per hour CEP, - second 5 hours: 2.0 mautical miles per hour CEP.

295











W. A. Analderica A. W.

These design goals are explained as follows. The system should be able to be aligned within 25 minutes from cold start (-65°F) at latitudes  $-75^{\circ}$  to  $+75^{\circ}$ , even when the initial location is not known to better than a 1 nautical mile (30) tolerance.

Following ground alignment, the ensemble position error growth rate should to no larger than 0.75 nautical mile per hour CEP\* for the first five hours and 1.25 nautical miles per hour CEP for the second five hours, for either the Doppler-inertial or the free-inertial modé. Following air alignments, the corresponding goals are 1.0 and 2.0 nautical miles per hour, even with a 5 nautical mile (30) initial position uncertainty.

The free-inertial mode performance goals are, of course, the more stringent. That is, Doppler data provides added measurements which, if propenly used, must improve the system performance over that obtained with the free-inertial mode.

Obviously, the operational environment for ground alignment is far more severe than a laboratory environment. The C-5 is a heavy logistics transport and the INU and other equipment are located in the nose visor. The nose visor is open while the aircraft is loading cargo. The ground alignment process must be carried out during visor motions created by winds and cargo-loading transients. Definitions of approximate, but realistic, statistical models for such factors were required during the Kalman filter development effort.

#### 2.3.3 Scheduling Requirements

296

An unusually stringent schedule was maintuined on the entire C-5 program by Lockheed, Northrop, and the other equiprent suppliers. For that reason, when considering various system refinements in which a number of alternatives presented themselves, decision deadlines had to be mot, sometimes requiring a less thorough than desired evaluation of each alternative.

#### 2.4 Problem Summary and Approach

The C-5 Kalman filter development problem can be summarized as follows.

- Given: (a) Performance specifications and physical characteristics of the C-5 subsystem hardware. (b) Required overall C-5 system functional capabilities and performance goals.
- Define: The algorithms for the Kalman filter portion of the real-time program, and all interfaces with other sub-programs of the computer.

A logical first question was: "does a solution exist?". At the start of the problem, relevant information from many sources (such as practical experience and data from conventional inertial navigation systems) was available. Theoretical considerations, preliminary analysis, and practical experience all contributed to give a high degree of confidence that a solution must exist. In fact, as will be seen later, probably many alternative solutions exist. Hence, as the design gradually evolved, engineering judgment played a large role in the selection of the specific solution from the available alternatives.

In solving a problem of this type, it is logical to first ask four primary questions:

- (1) What is known about the problem?
- (2) What additional information must be obtained?
- (3) How is this additional information to le obtained?
- (4) Can the generation of this information be organized so that it is available when needed?

The answers to question (1) are summarized in the paragraph above, and the other questions are discussed below. Note that answers to question (3) are the solutions to the problems resed implicitly by question (2). The answer to question (4) is really the approach to solving the overall problem. That is, the overall problem is separated into many small problems, the solutions of which are organized to provide a logical time sequence of overall system development. Typical problems and solution approaches involved in the C-5 Kalman filter design development are illustrated in the following paragraphs.

#### Example 1

1.

Preliminary statistical information was needed on system performance in its principle modes of operation (e.g., Doppler-inertial navigation following a ground alignment). Adequate statistical error analysis techniques for linear systems were already available. Further, past applications had shown that such techniques were applicable to nonlinear problems, if the idea of a nominal trajectory representing the mean value of the ensemble were used. Use of linear variational equations for small deviations from the nominal trajectory therefore provided the desired linear equations for propagating and relating subsystem and system errors for C-5.

\* The CRP (circular error probability) usage for ensemble position error growth is interpreted as meaning that, for an arbitrary system selection and test evaluation, the probability is 0.5 that the position error is less than the specified number.







- Karalan Karalan Karalan Karalan Karalan Karalan Karalan Karalan

#### To use this theory several steps were required:

- (1) Derivation of an approximate but realistic mathematical model of the inertial system.
- (2) Derivation of variational equations for the inertial system error model.
- (3) Development of subsystem error models.
- (4) Development of general-purpose digital computer programs for calculating ensemble error analysis data.

These error analysis programs, though preliminary in some respects, must be sufficiently floxible to enable investigation of performance sensitivity to measurement cycling rates, non-included error sources, and the like, Early answers to questions in such areas were mandatory to allow reasonably accurate, first-cut assessment of the demands which the final Kalman filter program would make on the on-board computer.

#### Example 2

いいいうし

ni I

Information was needed relating the number of state variables of the Kalman filter to real-time computer speed and memory requirements.

Since, given the algorithms, an experienced real-time programmer can estimate time and memory factors, this problem was solved by out-and-try procedures. Initially, "ball park" answers were found. These act as guidelines until new information requires refinements. As described in Section 3, such data, together with the results from the ensemble error analysis programs, indicated a need for development of a special data smoothing algorithm for the raw data, before Kalman filter processing.

#### Example 3

Information was needed about such nonlinear phenomena as the build-up of computational error in the on-board Kalman solutions. No known theory could provide answers to such questions. Simulation of the system on a general-purpose digital computer, however, promised a very practical solution. Honce, development of a simulator and study of such effects under simulated conditions was the solution chosen. As the design evolved prior to laboratory and flight tests, the simulator was expected to become the primary tool for validation and/or checking of algorithm refinements. If the simulator design were made sufficiently versatile, it would also help in understanding the causes of, and determining fixes for, anomalistic behavior of the real hardware in later stages of the system development.

These are a few examples of the considerations which influenced the approach to the overall problem solution. From the three examples given, it should be reasonably obvious that the problems could not all be solved simultaneously. An orderly, time-sequenced approach had to be formulated with overlaps in obvious areas, and with contingencies for\_redirection based on new information. Such an approach adapted itself naturally to the overall system development time schedule. The sequence of presentation in the next two sections follows, to a large degree, the overall approach adopted by Northrop management at the start of the development effort.

## 3. DESIGN DEVELOPMENT AND DESCRIPTION

The previous section has briefly discussed some of the problems which had to be solved in the development of a practical C-5 Kalman filter. The most important initial steps were the derivations of:

- (a) the system state vector,
- (b) dynamic equations for the state vector,
- (c) variational equations for propagating errors in the state vector,
- (d) statistical models for the error sources.

Furthermore, a general-purpose computer program was needed early for evaluating effects of various subsystem design parameters on system performance.

As data was gathered from the above tasks, the constraints imposed by the limitations of the roal-time computer were investigated. These were:

- (a) real-time availability,
- (b) computational precision.
- (c) memory availability.

This section follows the overall development in the sequence given, and summarizes the design algorithms of the real-time computer that resulted from this phase of the development.

#### 3.1 Mathematical Modeling

The general definition of an appropriate mathematical model for a real system is one of the most difficult tasks in the application of Kalman filter theory. The decision as to what degree of approximation is appropriate is primary, since unnecessary complexity will result in unacceptable real-time computer storage and memory requirements.







To illustrate, consider the problem of defining a G-5 system state vector. The primary objective of the navigation system is to provide continuous, accurate indications of the C-5 mircraft position, velocity, and heading<sup>2</sup>.

The differential equations for siroraft acceleration can be written in the form

<u>₽</u> = 8+8+4.

where

Z = miroraft Acceleration

W = gravitational force/unit mass.

A = aerodynamic force/unit mass,

Y = thrust force/unit mass.

The velocity  $\vec{x}$  and position  $\vec{z}$  would comprise mix vector elements each. To describe  $\vec{\lambda}$  and  $\vec{T}$ , a great many more state variables would be needed. This definition would permit treatment of the INU accelerometer outputs and many other sensed quantities (e.g., engine settings, control deflections, etc.) as observations. If, however, the differential equations were written in the form

 $\frac{H}{2} = \vec{q} + \vec{r}$ , (3.2)

where  $\vec{f}$  represents INU accelerometer outputs, then it would no longer be necessary to model the aircraft attitude motions, engine dynamics, etc. A large number of state variables would thereby be replaced by the outputs of a measuring instrument. In the use of Equation (3.2), however, the INU accelerometer outputs are not observations in the Kalman filter theory sense. Instead, they are forcing functions. Nevertheless, as will be seen, errors in these "measured" forcing functions can be accounted for with this approach.

For either of the two sets of Equations (3.1) or (3.2), Doppler data, barometric pressure, LORAN, TACAN, and the like, all of which are only functions of position 'and/or velocity, can be treated as observations. Suppose now that the IMU fails. Assuming that Doppler data is still evailable, the equation

 $\dot{\vec{z}} = \vec{z}(t) \tag{3.3}$ 

can now be written, where  $\vec{g}(t)$  is the Doppler velocity measurements. The Doppler data is no longer an observation in the filter theory sense; however, the LORAN and TACAN data, which are functions of the position state variables,  $\vec{z}$ , are still observations.

The definition implied by Equation (3.2) was the one selected for C-5. As montioned earlier, a wander azimuth, geodetic vertical frame had already been selected for the navigation equations. Hence, the differential Equations (3.2) were changed to define the vehicle acceleration and velocity in this rotating reference frame.

Care in state vector definitions also allowed the equations for the dead-reckoning modes to be sub-sets of the equations for the Doppler-inertial modes. This permitted a much simpler overall formulation for the multi-mode system.

#### 3.1.1 Free-Inertial Mode Model

As already discussed, it was planned to use the free-inertial navigator (in the primary m-wise) for updating the total estimate of state in the time intervals between measurement cycling of the Kalman filter. Therefore, the need for an integration of nonlinear differential equations as part of the Kalman filter was obviated. However, a derivation of an approximate model of the system was needed as a basis for an approximate transition matrix for the Kalman filter. As described earlier, the differential equations for the system state were

$$\dot{X} = F(X,t) + u$$
, (3.4)

Using these, the variational equations:

Å ≖ A(t)¥

were required, where

$$A(t) = \nabla_{x} [(F(x, t))]_{x(t) = \hat{x}(t)}$$
(3.6)

 $\hat{X}(t)$  is the estimate of system state. Several of the components of  $\hat{X}(t)$  are available from the inertial navigator.

The inertial navigator consists of the IMU and the navigation equations solved in the NDC-1051A computer (see Figure 1). A reasonable model for Equation (3.4) can be defined by writing approximate differential equations for the IMU computer system. To illustrate how this can be accomplished, consider the modeling of the way by which the specific force vector is obtained for input to the navigation equations (see Figure 5).

\* An inertial, non-rotating coordinate frame is assumed here for simplicity of description.

(3. 1)

(3.5)





· · · · · · · · · · · · · · · · · · ·	114	1.4	4 /	
_ F.A., F				

The true, vector specific force  $\vec{f}_1$  is transformed to the platform reference frame by the transformation  $T_{FIFF}(t)$ . Since the accelerometer-sensitive axes are not precisely erthogonal, the non-orthogonal transformation.  $T_{FFF}(t)$ , transforms the specific force to accelerometer axes. In addition, accelerometers have scale factors which are not precisely known; the diagonal matrix SPA represents these factors. Finally, the accelerometer outputs are also corrupted by bias offsets and noise, represented by the two vectors  $\vec{B}_b$  and  $\vec{B}_h$  respectively. The vector  $\vec{f}_0$  represents the simplified, continuous model for the outputs of the actual accelerometers.

Standard C-5 system procedures include provision for estimation of all the above accelerometer-related errors, and for a set of gyro-related errors discussed below as well, during periodic, laboratory calibration operations on each IMU. These numbers are then incorporated into the computer to appropriately compensate the accelerometer inputs to, and the gyro command outputs from, the computer during motual system operation. In this connection, the vector  $\vec{a}_{b0}$  in Figure 3 denotes the calibration for accelerometer null bias  $\vec{a}_{b}$ , and the 3 x 3 matrix  $[a_{ij}]$ denotes the calibration for the accelerometer scale factor and input axis orthogonality errors, that is

$$[\mathbf{a}_{11}] \approx [\mathbf{T}_{\mathbf{p}_{2A}} \mathbf{SPA}]^{-1}$$

The estimated specific force vector  $\vec{A}_{g}$  can therefore be written in terms of all these quantities. The error in  $\vec{A}_{g}$  due to each one of the contributing accelerometric error sources can, in turn, be approximated by first-order variations of these relationships.

The error in  $\vec{A}_{g}$  caused by the errors in the transformation  $T_{p_{1,2}p}(t)$  must also be derived. These latter errors are called tilts because, ideally,  $T_{p_{1,2}p} = I$ , if  $\vec{f}_{1}$  is defined in the true local tangent plane reference system.

Continuing with Figure 5, the vector  $\vec{A}_g$  feeds the navigation equations, which are shown schematically in Figure 6. Although the NDC-1051A calculations are discrete rather than continuous, their repetition rate is high (12 Hz). Hence the error differential equations for these calculations can be approximated by taking gradients of the nonlinear equations with respect to the various variables.

Three of the nine fundamental state variables can be identified by referring to Figure 6. These are the integrator outputs:

$$V_{i}, \overline{V_{L}} = \text{vertical and level velocities}$$
  
h = altitude.

Two others, latitude and longitude, are obtained from the transformation T. The final fundamental state variables are the three rotations necessary for defining the transformation  $T_{p_{1,2}p}(t)$  discussed above.

5

à

The closure of the loop (i.e., the gyro signal processing) is illustrated in Figure 7. The gyro torquing signals, represented by the vector  $\omega_0$ , are transformed by the calibration matrix  $[g_{ij}]$ . This matrix is a calibration for the gyro scale factor errors (diagonal matrix SPO) and the gyro input axis non-orthogonalities (matrix  $T_{0i2P}$ ). Note that  $T_{0i2P}$  transforms rates from the gyro input axis frame to the platform reference frame. Also,

Shown also are the  $[u_{ij}]$  matrix, which compensates for gyro drift rates caused by static mass unbulance, and the vector  $\vec{s}_{bc}$ , which is a calibration of the fixed gyro drift rate. In Figure 7,  $\vec{\omega}_{d}$  is the actual vector drift rate of the gyros, referred to the input axis. Reference 3 gives many details on the modeling of such factors.

The final output is the total inertial rate of the platform,  $\vec{a_p}$ . Again, first-order variations of the various equations are used to define errors in  $\vec{a_p}$  created by the various error sources.

The process just outlined led to the definition of approximate variational equations for the free-inertial navigator errors in the state-space form:

(9 x 1) (9 x 9) (9 x 41) (41 x 1)  

$$\dot{\bullet} = \alpha(t) \bullet + \beta(t) \bullet_r$$
.

There are nine error state variables and 41 error forcing functions. Some of the error forcing functions were assumed to be constants (e.g., accelerometer and gyro misslignments), while others were prescribed functions of the flight path (e.g., static mass unbalance drifts); however, known fluctuations in gyro drift rate and accelerometer noise needed a random characterization. This was accomplished by assuming that such errors were representable by white noise driving a first-order filter in accordance with Figure 8.

Before the final, augmented state vector formulation is defined, a brief consideration of the observation modeling is in order.

A THE REAL PROPERTY AND REAL PROPERTY.

#### 3.1.2 Observation Models

The process of appropriately describing the observations and their error models follows a pattern similar to that described for the free-inertial model.

In certain instances (e.g., a LORAN time-difference pair and inertial navigator latitude and longitude), a nonlinear equation relater the inertial and subsystem measurements; in others, a direct linear relation exists (e.g., barometric pressure altimeter and inertial altitude indications). In either case, however, linear relations (M matrices) between inertial and subsystem errors at a measurement time were developed by simple first-order variations.

Modeling of the time propagation of subsystem errors necessitated the use of time-correlated random variables. These were all approximated by white noise into a first-order filter, as already mentioned. This process led to the definition of 17 more error state variables (constant blases and/or correlated noise) of the type described, by the vector V of Equation (2, 2).

#### 3.1.3 Final Linear Dynamic Error Model

Augmontation of the fundamental state vector to include all the deterministic type of error sources led to a final dynamic error model of the form

$$\begin{array}{c} \mathbf{67 \times 1} & (\mathbf{67 \times 67}) & (\mathbf{67 \times 1}) & (\mathbf{67 \times 1}) \\ \dot{\tilde{\mathbf{x}}}(\mathbf{t}) &= \mathbf{A}(\mathbf{t}) \tilde{\mathbf{x}}(\mathbf{t}) + \mathbf{n}(\mathbf{t}) & . \end{array} \right\}$$

$$(3.7)$$

The vector  $\bar{\mathbf{x}}(t)$  represents the total augmented error state vector. The vector  $\mathbf{n}(t)$  represents the white noise inputs for some of the augmented state variables. Most of the components of  $\mathbf{n}(t)$  are identically zero. It should be noted that this model assumes a smooth time propagation of system errors. In reality, because of the quantimed, iterative nature of digital computer signal processing, the actual errors really exhibit staircase behavior. However, because of the fineness of the quantization (typically one part in  $2^{27}$ ), and the high iteration and sampling rates employed (e.g., 12 Hz), the continuous approximation is an adequate one and, of dourse, highly convenient for simplifying any necessary mathematical manipulations of the model.

#### 3.2 Ensemble Error Analysis Program

At this stage of the overall development effort it was possible to start the construction of a 67-variable ensemble error analysis program, hereafter called Program I for convenience of reference. The purpose of this program was to establish the system performance possible if the C-5 Kalman filter incorporated all 67 state variables. This error model, although still approximate, contained all the error sources considered as having some significance; it is therefore hereafter referred to as a "complete" error model.

Data from this program was intended to serve as a reference standard, against which the performance of subsequent, less sophisticated alternatives° could be measured. Construction of the overall program required;

- (1) A means of generating desired flight profiles.
- (2) Input means for controlling all the assumed initial error sources, the flight profile, and the sequence of observations.
- (3) The algorithm for the change in the error covariance matrix resulting from use of the observations.
- (4) The algorithm for updating the error covariance matrix between the observations.

Of these, items 1 and 2 are program details which will not be pursued here. For item 3, an optimal filter was assumed. Item 4, however, merits some discussion, since it incorporated some approximations which were submequently used in the actual, on-board Kalman filter program.

#### 3.2.1 Prediction Between Observations

The algorithm for prediction of the covariance matrix across the intervals between successive operations was developed as follows.

If  $\bar{X}(t)$  denotes the error in the state estimate at time t , then

$$\tilde{a}(t_{m}) = \Phi(t_{m}; t_{0}) \mathbf{x}(t_{0}) + \int_{t_{0}}^{t_{m}} \Phi(t_{m}; \tau) \mathbf{n}(\tau) d\tau \qquad (3.8)$$

describes the error propagation across the prediction time interval ( $\Delta t = t_{g} - t_{g}$ ) between successive observations! Defining the covariance matrix of the error by

• It is apparent that the on-board computer could not solve such a voluminous problem in real-time and, as will be seen, it is not practical to do so.

<sup>†</sup> Later this time interval will be seen to be the actual cycling time of the Kalman filter.











-----



(3.9)

(3.10)

and the second state of the second second

 $P(t_m) = E\{\tilde{X}(t_m)\tilde{X}^T(t_m)\},\$ 

it follows that

$$(t_n) = \Phi(t_n + \Delta t; t_n) P(t_n) \Phi^T(t_n + \Delta t; t_n) + R(t_n + \Delta t; t_n), \qquad (t_n + \Delta t; t_n)$$

where

$$\mathbf{R}(\mathbf{t}_{0} + \Delta \mathbf{t}; \mathbf{t}_{0}) = \int_{\mathbf{t}_{0}}^{\mathbf{t}_{0} + \Delta \mathbf{t}} \int_{\mathbf{t}_{0}}^{\mathbf{t}_{0} + \Delta \mathbf{t}} \Phi(\mathbf{t}_{0} + \Delta \mathbf{t}; \tau) \mathbb{E}\{\mathbf{n}(\tau) \mathbf{n}^{\mathsf{T}}(\boldsymbol{\xi})\} \Phi^{\mathsf{T}}(\mathbf{t} + \Delta \mathbf{t}; \boldsymbol{\xi}) d\tau d\boldsymbol{\xi}.$$
(3.11)

Equation (3.10) is the desired algorithm for updating the error covariance matrix between observations. However, suitable approximations for the transition matrix,  $\Phi$ , and for the system noise matrix, R, needed to be devised. As an illustration, the development of the transition matrix proceeded as follows.

The transition matrix obeys the differential equations:

 $\dot{\Phi}(t) = A(t)\Phi(t)$  $\Phi(0) = I$ 

When A is a constant matrix, a closed-form solution for  $\Phi$  can be obtained:

$$\Phi(\Delta t) = \Gamma^{-1}[(\mathbf{s}\mathbf{I} - \mathbf{A})^{-1}], \qquad (3.12)$$

where s is the Laplace variable, and I is an identity matrix of the same dimension as A. When the dimension of A is large, however, it is generally hopeleasly intractable to algebraically carry out the indicated matrix inversion. Other, approximate methods must therefore be used. One of the simplest of these involves the use of the power series expansion:

$$\bar{\Psi}(\Delta t) = \sum_{i=0}^{\infty} \left(\frac{1}{i!}\right) (A\Delta t)^{i} = I + (A\Delta t) + \frac{1}{2!} (A\Delta t)^{2} + \dots \qquad (3.13)$$

The series is bruncated at a point sufficient to provide the desired accuracy. However, if A is timedependent, which is the case for C-5 mirborne modes, the corresponding series is:

$$\Phi(\Delta t) = I + (A\Delta t) + \frac{1}{2l} (\dot{A} + A^2) (\Delta t)^2 + \dots \qquad (3.14)$$

The higher-order terms not shown contain higher-order derivatives of A. There are some difficulties associated with attempting to use a truncated form of this expansion. These stem from the impracticality of obtaining the first- and higher-order derivatives of A, if second- or higher-order accuracy is required. This impracticality arises because A naturally contains only the basic institut navigation signals, up to the soceleration level, as elements. Use of terms involving A or higher derivatives would therefore require the artificial, approximate generation of otherwise unavailable acceleration derivatives.

On the other hand, the accuracy involved in the use of only the first two terms in the expansion depends on the size of  $\Delta t$ . It was evident early in the program that, because of the multiple non-Kalman functions required of the C-5 computer, only a small percentage of its real-time use could be devoted to cycling the Kalman computations. In addition, even with the large anticipated reduction in the size of the state vector (from 67), these computations would still be lengthy and therefore time consuming. All of this suggested that the design, even at this stage, should anticipate the need for a large  $\Delta t$ , i.e., of the order of many seconds to a few minutes. To accompodate this need, the following approach was adopted.

One of the important properties of the transition matrix is its exponential character; that is, the transition matrix for state evolution between times  $t_1$  and  $t_3$  is the product of the transition matrices associated with state evolution from  $t_1$  to  $t_2$  and from  $t_2$  to  $t_3$ , respectively. A simple extension of this property gives:

 $\Phi(t_{n};t_{0}) = \Phi(t_{n};t_{n-1})\Phi(t_{n-1};t_{n-2}) \dots \Phi(t_{2};t_{1})\Phi(t_{1};t_{0}) , \qquad (3.15)$ 

where  $t_{\mathbf{R}} = t_{\mathbf{R}}$ .

Denoting  $t_i \sim t_{i-1} \simeq \Delta t_i$ , and setting  $\Delta t_i \simeq \Delta t/n$ , where  $\Delta t \simeq t_m - t_o$ , the  $\Delta t_i$  could then be visualized as the n equal time intervals between n + 1 successive updates of all navigation signals by either the navigation computer (C-5 filter mechanization) or the navigation sub-routine (ensemble error analysis program), and  $\Delta t$  could be visualized as the Kalman cycle time itself. Since  $\Delta t_i$  could now be assumed quite small (down to 1/12 second in the navigation computer, and arbitrarily small in the ensemble program), the truncated form,

$$-\Phi(\Delta t_i) = I + \{A_{1-1}\Delta t_i\} + \frac{1}{2}\{A_{1-1}\Delta t_i\}^2, \qquad (3.16)$$

was tentatively assumed sufficiently accurate for extrapolation over a  $\Delta t_i$  interval. Substitution of Equation (3.16) into (3.15) gives

$$\Phi(\Delta t) = \prod_{i=1}^{n} \left[ I + (A_{i-i}\Delta t_i) + \frac{1}{2}(A_{i-i}\Delta t_i)^2 \right]$$
(3.17)

or, for n > 1 (see footnote),

$$\Phi(\Delta t) = 1 + \Delta t^* \sum_{i=1}^{n} A_{i-1} + \frac{1}{t} (\Delta t^*)^2 \sum_{i=1}^{n} A_{i-1}^* + (\Delta t^*)^2 \sum_{i=2}^{n} A_{i-1} \sum_{k=0}^{i-2} A_k , \qquad (3.18)$$

where  $\Delta t^* = \Delta t_1 = \Delta t/n$ , and terms of order higher than  $(\Delta t^*)^*$  have been dropped.

Equation (3.18) still presented difficulties, primarily because mechanization of the last (double summation) term suggested considerable complexity. This term was therefore tentatively dropped, as were all the remaining second-order terms, except for certain principal ones associated with constant elements of the matrix  $A_{i-1}$ .

The residual expansion finally adopted was thus

$$\Phi(\Delta t) = I - A_{j}\Delta t + \frac{1}{4}A_{0}^{2}\Delta t^{2} + \sum_{i=1}^{n} \left[A_{NG_{i-1}}\Delta t^{*}\right], \qquad (3.19)$$

where  $A_0 + A_{NO_{i+1}} = A_{i+1}$ , and  $A_0$  and  $A_{NO_{i+1}}$  are respectively the constant and non-constant portions of the matrix  $A_{i+1}$ .  $A_0^2$  represents a matrix of melected, principal constant elements of  $A_{i+1}^2$ .

For preliminary evaluation, a small program was constructed to check the validity of Equation (3.19) if used recursively to approximate stationary linear dynamics (e.g., the C-5 ground alignment situation) over long periods of time. Although this program used only a limited-dimension model, the model incorporated the principal dynamical characteristics associated with the stationary solution (5.19) and the stationary shall percentage drifts relative to an exact solution (obtained by use of Equation (3.12) on the limited-dimension model over 10-hour periods, even with  $\Delta t$  as large as several minutes.

The airborne modes require additional non-stationary A matrix terms that depend on the aircraft flight profile. Evaluation of the adequady of Equation (3, 19) in airborne modes was therefore deferred until after the development of the system simulation program, described later in this chapter.

#### 3.2.2 Observations

Three types of information external to the inertial navigation subsystem would be available for comparison with its outputs in the ground alignment mode and the inertial and Doppler-inertial navigation modes:

- (a) baromotric altitude from the CADC;
- (b) groundspeed and drift angle from the Doppler;
- (c) the knowledge that the nose visor is essentially stationary\* during ground alignment.

Observation of the information described by (c) is discussed here to illustrate the general approach.

Since the aircraft is parked during ground alignment, either the zero velocity or the fixed position of the sircraft can be used as the information reference against which to compare, respectively, the inertial navigator velocity or position<sup>†</sup>. However, since the use of velocity-level measurements was already anticipated for Dopplarimertial navigation, velocity comparison was selected for ground alignment. This provided a greater uniformity between the ground alignment and Doppler-inertial mode mechanizations.

If either one of the institial navigator level velocity readings is denoted by v the corresponding Kalman observation Y, used in ground alignment, could therefore be

Y = v.

The signal v can be decomposed into the form:

 $\mathbf{v} = \mathbf{v}_{\mathbf{A}} + \delta \mathbf{v} , \qquad (3.20)$ 

American Jule

where  $v_A$  is the actual visor velocity component, due to wind and loading along the level direction associated with v, and  $\delta v$  is the error in the signal v arising primarily from initial tilts of the platform with respect to the vertical, and from unculibrated platform drift rates.

Footnote: The notation from here on has been purposely simplified for compactness; thus,  $\Phi(\Delta t)$  is used for  $\Phi(t_a; t_0)$ ,

and the product notation  $\prod_{i=1}^{n}$  implies the ordered product of matrix factors (i = 1 to n from right to left).

\* Except for the motions induced by wind and aircraft loading.

\* Either reference, if properly used, is entirely equivalent to the other.

302

Ń

. 1
From the standpoint of Equation (3.20), it would appear that  $\delta v$  is the signal containing information, and  $v_A$  is the noise from which it must be separated. However, this assumption about  $v_A$  is justified only if  $v_A$  consists mostly of noise which is of high frequency with respect to both the  $\delta v$  signal content and the contemplated observation iteration rate. Because of the frequency response characteristics of the massive C-5 aircraft to random wind and loading disturbances, this proved to be true for ground alignment. If it were not true, it might have been necessary to introduce error state variables to account for systematic low-frequency behavior of  $v_A$  to meet alignment performance goals. Such variables, which generalize the system error state observations.

Since &v was an early, natural choice as one of the essential all-mode state variables, and since the notation relating measurements to state is

$$\mathbf{Y} = \hat{\mathbf{Y}} = \mathbf{H}(\tilde{\mathbf{x}} - \hat{\mathbf{x}}) + \mathbf{q}$$

by association. M was therefore defined as:

# 1 2 3 ... δν ... n M = [0 0 0 ... 1 ... υ].

The choice of two scalar measurements instead of one (two-dimensional) vector measurement for ground alignment deserves comment. This choice d.c. not require matrix inversion. Instead, two successive passes are made through the observation update portion of the Kalman equations. Each pass only involves a division for the inversion of the quantity (MPM<sup>T</sup>+C) in Equation (2.9). In fact, all observations were defined as scalars to make this processing simplification uniform.

Although this approach is theoretically exact only if the noise associated with each separate observation is uncorrelated with that associated with every other observation, ground alignment proved to be the only C-5 case where this condition was not substantially met. The resulting performance of ground alignment, as will be seen, nevertheless met design goals, despite this theoretical infringement.

Having defined the observation matrices associated with all C-5 modes and information sources, it was then possible to complete the 67-variable ensemble error analysis program (Program I) by incorporating the following matrix equations:

 $\begin{array}{l} \mathbf{Q} = \mathbf{M}^{\mathbf{P}}\mathbf{M}^{\mathbf{T}} + \mathbf{C} \\ \mathbf{b} = \mathbf{P}\mathbf{M}^{\mathbf{T}}\mathbf{Q}^{-1} \\ \mathbf{P} \neq \mathbf{P} - \mathbf{b}\mathbf{M}^{\mathbf{P}} \end{array}$  (3.21)

These, together with Equation (3.10), completed the set of recursion equations necessary to obtain the time history of the covariance matrix P of navigator errors through a typical C-5 flight operation.

To simulate such flights, the program was equipped with a general, flexible flight profile sub-routing capable of providing great circle cruise, climb, descent and turn phases in any commanded sequence. The navigation signals thus generated were used to drive the dynamic elements of the transition matrix.

Program I was coded in FORTRAN IV single precision for use on the IBM 7090 or 360 computers. On completion all checkout, the immediate, extensive use of Program I at this stage established the ability of a 67-variable. All precision 7090 or 360 muchanization to meet or exceed all C-5 navigation system performance goals.

### 3.3 Design Review and Discussion

At this stage of the development the overall final filter design was beginning to take shape. Approximately half the development time had elspsed, however, and many of the difficult problems imposed by the real-time computer constraints had yet to be faced.

### 3.3.1 Overall Design

The overall design mechanization for the multi-mode system was reasonably well defined. Figure 9 is a functional block diagram of the overall mechanization, showing the role of the filter in the overall navigation information processing. Although this is essentially an analog diagram, definition of the overall sequencing of the digital calculations was also well advanced, as will be seen below.

In this connection, it was recognized that the actual Kalman filter would require an appreciable time to process a single set of observations and to produce revised error estimates for application as system corrections. While this was going on, data on current position and velocity would continue to be required for inputs to other, non-navigational functions (A.g., sircruft steering) and for display. Since such signals might be available at several different points (e.g., groundspeed from corrected Doppler groundspeed or from the horizontal navigation equations in Figure 9), one source had to be uniformly selected as "best" between successive Kalman corrections. Because of their relatively high update frequency and their relative smoothness, the (corrected) inertial navigation signals were chosen as the best outputs for this purpose.









# 3.3.2 Kalman Filter Design

Table I summarizes the C-5 Kalman filter equations. The computations shown, together with the formation of the various matrices involved, comprise one full Kalman processing cycle in the on-board computer, and are given in the order of actual processing.

As discussed earlier in this chapter, the Kalman equations are modified to include the "epsilon" weighting technique and to provide for system and estimator control. At one time it was contemplated not to mechanize control at all, but rather to perform only open-loop error estimation. Navigation signals would then be corrected with these estimates only for output to other equipments. This approach was discarded in favor of the closedloop control technique shown, because it was recognized that otherwise the linearization required would be done about the uncontrolled system state. A large error build-up in this uncontrolled state would cause a correspondingly larger departure from the desired linear properties.

The essentials of the control technique used, which are summarized symbolically in Table I by the notations f() and g(), are as follows:

(a) Impulsive Control Wherever possible, the contents of a computer register whose error is modeled as a state vector element are corrected by adding the estimate of error (with the proper sign) to the computer register. Simultaneously, the register containing the error estimate is set to zero. This is done to each such register pair one in every Kalman computational cycle.

(b) Finite Duration Control The above impulsive type of control is used uniformly for all registers controlled by the C-5 Kalman mechanization, with one important exception. This exception portains to those two state vector elements which model the two components of platform tilt; i.e., the two components of the small angular deviation between the actual and desired platform vorticals. To remove the actual tilts, the gyros are commanded to torque the platform vertical at a uniform rate over the next Kalman interval. This rate is exactly that required to take out the estimated tilt in the platform in one Kalman cycle. At the same time that these leveling rates are first applied, the platform drift rate estimates are set to these computed leveling rates. The current tilt estimates are left untouched.

1 JANA . V.

# 11:24

Note that a basic, built-in invariant implicit in these control techniques is the difference between the actual and the estimated system behavior. This invariance allows retention of the simple, open-loop covariance matrix recursion relations unchanged, if it is assumed that the additional errors created by the application of control are negligible. This assumption was made at this stage of development for later validation.

The off-optimal "epsilon" factor was introduced in the design algorithms as a precautionary weasure to combat unmodeled and numerical errors (see References 7 and 8).

### 3.3.3 Other Relevant Information

Concurrently with the effort described thus far, some additional data was being accumulated as a result of related work on another system. For example, an NDC-1051 computer (24-bit word) had been programmed for Kalman estimation in the fine ground align mode, and laboratory tests were conducted with an INU of different design from another program. The results were not overly encouraging.

One of the problems was the difficulty of maintaining a positive definite covariance matrix. The other appeared to be connected with the use of fixed-point arithmetic. (nly poor estimates of gyro drift rates were obtained. This seemed to be tied to the loss of numerical significance in the covariance matrix due to the fixed-point calculations. These results were one of the reasons for the introduction of the "epsilon" technique in the basic mechanization equations for the C-5 system.

### 3.4 Constraint-Imposed Problems and Solutions

One of the principal problems which had to be dealt with at this point was the identification of the smallest system error model which could be mechanized for the on-board Kalman filter, and still meet system performance goals. Use of the complete, 67-variable model of Program I, for example, would require far too much memory and real-time of the on-board computer. However, Program I results did indicate that the complete model, if mechanized, would substantially surpass system performance goals. This, therefore, provided a reasonable performance margin to work with, in terms of the loss of performance inevitably expected from on-board use of mailer models.

Actual quantitative evaluation of this performance loss is a relatively subtle problem which cannot be solved, for example, by comparison of 67-variable model Program I results with results from an analogous program incorporating the smaller model. Rather, a more general type of computer error analysis program is required. As will be seen, the stringent overall C-5 development schedule had a direct effect at this time on the selection of the type of program to be constructed to fill this role.

In addition, definition of the airborne computer tape sub-routines had been proceeding concurrently in all other non-Kalman areas. Since design in these more conventional areas naturally proceeded faster than it did in the more novel Kalman area, the programs for these sub-routines, together with their attendant word count and time allocations, began to crystallize zomewhat ahead of those for the Kalman sub-routine. This process led to the early establishment of word count and real-time allocations for the Kalman sub-routine. These allocations constituted the two primary constraints on the filter design.

These contraints, together with the accuracy goals and all-mode operation requirements, comprised the framework within which the Kalman design had to be completed. The problems created by these contraints, the methods of attack used, and the final design solutions adopted are now discussed.

## 3.4.1 Computer Stornge Considerations

1

'n.

The relative fractions of the total computer storage allocated to the various functions of a real system like this one is always a matter of compromise, requiring continuous trade-off evaluations. A Kalman filter having sufficient dimension to satisfy the C-5 requirement was new, although experience did exist on more limited implementations. However, these implementations were inadequate as a basis for estimating the C-5 Kalman storage requirement. Nevertheless, the storage available was defined quite.closely, because the 12,288-word total (primary computer) was fixed and the requirement for most of the other routines could be closely estimated on the basis of experience.

Depending on how much program compaction effort was to be applied to the other routines, the storage available for the Kalman routine ranged from 2000 to 3000 words. It was immediately evident that the model size must be reduced (from 67) as far as performance goals would allow.

In conjunction with this effort, it became evident that the minimal word length to be used in carrying out all arithmetic operations would have to be determined. If double, rather than single, precision operations were used, these would obviously double the non-instruction storage requirements. This consideration of word length also naturally raised a question as to the type of arithmetic to be used (fixed- or floating-point).

Finally, further consideration strongly suggested that the requirement for multi-mode filter operation might demand a sizeable portion of the overall Kalman word allocation for intermode switching.

#### 3.4.2 Error Model Dimension

It was recognized immediately that a natural, two-step iterative process, together with the techniques necessary to carry out these steps, would be required to determine the minimal error model for the Kalman filter mechanization. These steps were (1) a method for systematically discarding error variables from the complete (67-variable) model, and (2) a method for evaluation of the reduced models thus obtained.

This latter need - for a reduced model evaluation tool - led to the development of the second major ensemble error analysis program (hereafter referred to as Program II). Two types of formulation, of differing capabilities, were considered. The first and more general of these was based on an extended set of reductive covariance matrix equations (see Reference 9). This type of formulation allows complete freedom in discarding (a) arbitrarily selected error variables, and (b) arbitrarily selected elements of the residual transition and system noise matrices.

A second and simpler formulation was also available<sup>10</sup>. This one allowed calculation of the increase in error produced by discarding constant state vector elements only. Despite this theoretical limitation, it had the overriding advantage that it allowed rigorous handling of all bias errors (which were predominant) and permitted much faster program construction by only requiring the addition of a few mimple operations to Program I. For practical reasons, then, and in the light of the tight development schedule requirements, the second and simpler formulation was adopted.

An efficient method for discarding variables had to be devised, since the turn-around time for Program II results was one or two days. Hence, the obvious brute-force approach involving removal of one variable at a time could not be used. Instead it was necessary to try to identify at once some extensive sets of variables whose removal would, in a few steps, closely approximate the desired minumal error model. These discarded sets, which finally comprised no less than 51 of the original 67 variables, were arrived at as follows.

Between the time that Program I (the original 67-variable model) was formulated and Program II was completed and available, it had been determined for a variety of practical reasons, not to provide outputs from certain on-board navigation equipments as Kaluan filter inputs. The state variables originally included in the 67-variable model to account for the errors in these measurements were thus automatically discarded as a first stop. For example, although two radar altimeters would be aboard the C-5 aircraft, these would only furnish altitude above terrain, and up to a maximum of only 2500 feet. Since only the estimation of altitude above sea level was contemplated for the Kaluan filter, and since the C-5 aircraft would typically spend only a magligible percentage of mission time below 2500 feet over the ocean, use of the radar altimeter dats by the filter was excluded. The radar altimeter state variables in the original error model could therefore be discarded.

In addition, where several elements had been used in the complete model to describe the high, intermediate, and low-frequency components of the error in a subsystem navigation reference signal (e.g., Doppler groundspeed), these were compressed into one, where plausible, by eliminating all but the largest amplitude component.

Finally, the largest of the trial sets of discarded elements was composed of the velocity and/or accelerationsensitive errors associated with the IMU gyros and accelerometers. These errors, although important in a tactical aircraft design application, were expected to produce only minor performance degradations in such a huge transport mircraft as the C-5.

305















teren Bitter ftrit Breiten

7 er 1 g 🖸 🖉 🖓 🖓

Tables II and III summarize the 67-to-16 model size reduction finally attained by the techniques just described. Figure 10 illustrates the small ensemble performance degradation of the final 16-variable model, relative to the complete model, for a typical C-5 great circle flight, as obtained by Program II.

As protein of

# 3.4.3 Word Length/Arithmetic Type

In parallel with the effort to reduce error model size, an investigation was conducted to determine the most suitable word length/arithmetic type for the filter.

One of the problems identified almost at the start of this effort was the large dynamic range requirements of certain covariance matrix elements during the transient phases of filter operation. For example, it was estimated that platform tilts of, typically, one degree would have to be reduce: to about one second of arc in a short period at the beginning of ground alignment. This reduction was necessary to achieve sufficiently accurate platform drift rate calibration during the remaining ground alignment time. The ratio of starting tilts to residual tilts is about 3600 to 1. The tilt variances carried in the covariance matrix must, however, traverse the square of this range (i.e., about 10<sup>7</sup> to 1). A single word of the airborne computer is only 28 bits (27 bits plus sign) in length. Hence, if single precision, fixed-point arithmetic were used, the accuracy of settled tilt variance computations would only be about 3 binary bits. The effect of round-off and truncation errors on a word of only 3-bit significance during this critical alignment period was believed to be intolerable.

At this juncture, a wide variety of alternative solutions seemed available, comprising all remaining conceivable combinations of (i) word length, and (ii) fixed- or floating-point arithmetic. However, in the light of further sorutiny, these options were not all practical possibilities.

Before the final reduction to 16 variables, a podel of dimension 24 had been attained in the parallel error model size reduction investigation. With this size as uses, a preliminary sizing of the overall Kalman subroutine storage requirements had produced an estimate of 3300 words, predicated on the use of double precision, fixed-point arithmetic.

Although this figure was still in the ball park, it was evident at this point that 3300 words could be made available only if a considerable programming effort on the other routines was undertaken. (Because of experience gained on previous programs, most of the other routines already made efficient use of storage and further competion would have been difficult.) Therefore, the necessity for implementing the double precision fixed-point arithmetic approach (or any method that required more than a single 28-bit word) was questioned. Attention was then focused on the use of floating-point arithmetic. Fixed-point arithmetic requires careful scaling of all variables no that adequate significance is retained, while avoiding any danger of overflow. The scaling conideration is unusually complicated in the Kalman filter routine because of the extensive matrix operations. A significant advantage of the floating-point approach, therefore, is that the wooling tak is avoided entirely.

The hardware peculiarities of the machine itself now completed the rapidly narrowing restriction of choice. The NDC-1051A computer had a built-in (i.e., hardwired) capability for performing either full-word or half-word fined-point arithmetic. It was not practical to contemplate a split of the full 28-bit word into a mantless comprising most of these 28 bits and a small exponent to provide just enough dynamic range to satisfy covariance matrix requirements. (Although such a split could be used if accompanied by attendant sub-routines to provide erithmetic operations, such sub-routines were expected to be somewhat slow. This would have resulted in an instriction use of computer time.)

The design choice was thus narrowed to the use of floating-point arithmetic, utilizing a single 28-bit word shared equally by mantisse and exponent. In particular, the restriction to a short 14-bit (13 bits plus sign) mantisse created a need for evaluation of filter performance with respect to round-off and truncation errors. Since this evaluation was clearly beyond the existing or easily augmented capabilities of either Program I or Program II, a requirement for a new type of evaluation program was established.

Finally, it is pointed out here, for reference later in this chapter, that the adoption of floating-point erithmetic was the first and major contribution toward increasing the single Kalman cycle execution time. Some preliminary estimates, made at about the time this technique was adopted, indicated uninterrupted execution times of about 10 seconds. The preliminary real-time budget allocated about 20% of computer real-time during ground alignment and 5% in flight to the Kalman routine. Inis led to estimates of one to three minutes for on-board execution. As will be seen, the unavoidable emérgence of these slow Kalman cycles led to a challenging problem later in the development.

# 3.4.4 Multi-mode Requirements

ーンシンド

The proceeding paragraphs have centered on the effort to optimize design of the Kalman filter algorithms with respect to storage requirements. These efforts were characterized by use of the error model associated with the most complicated mode of C-5 pavigation system operation. This is the Doppler-Barometric-Inertial mode, augmented with position fixes from any or all of the on-board position-measuring sub-system equipments.

However, as has been pointed out, many different modes of system operation were contractually required. The evror model associated with each such mode differed with every other to varying degrees. Hence the problems of how to mechanize all such modes, and how to rwitch between them, had to be addressed. Multi-mode operation implies mode switching in somewhat arbitrary sequences, and optimal operation requires continuation or relatively complex transfer of the estimation process. For these reasons a unified, multi-mode Kalman filter dusign was formulated in solving the problem of minimizing storage requirements with this selected approach, the concept of the standard error model proved particularly useful, as follows.

This standard model was composed of the separate error models for the inertial navigator, the Doppler (groundspeed and drift ingle) inputs, and the CADC (barometric altitude) input. As such, it constituted the largest dimension (10) model required for dynamic error propagation in any navigation and alignment mode. Further, the state vector, trunkition matrix, and system noise matrix required to characterize any other mode could in every case be described either in terms of sub-sets of the standard slute vector and transition and system noise matrices, or as slightly modified sub-sets whose dimensions never exceeded 16. To a large extent, these convenient error model sub-sets axisted because of the care which had been previously taken, with regard to inter-mode uniformity, in the design of the signal flow in each of the mode mechanizations.

The use of this concept made it possible to view all non-standard modes as involving significant information in only certain well-dofined portions of the estimate and control vectors and of the covariance and system noise matrices, during both time and accasement update operations. The overall filter operations could then always be done on a fixed dimension (10) basis, independent of mode, without corruption of the information-carrying portion of these matrices by the portions carrying meaningless information. The continuity of state variables domain to all modes (e.g., position errors) when facilitated the design of inter-mode switching and initialization. Rigid conformity to these principles, in conjunction with incorporation of the 16-variable model and single-word floating-point arithmetic, ultimately led to an all-mode design which satisfied the 2200-word allocation.

# 3.4.5 Computer Real-Time Allocation

Throughout the early and intermediate design definition phases, there was a gradual but nearly inevitable drift toward a design requiring more and more execution time. This drift was recognized, and investigation of its implications on the final design was therefore initiated.

The first problem was the need for development of a transition matrix capable of extrapolation across relatively large time intervals. The second problem was the potential loss of reference information (s.g., Doppler groundspeed and drift angle measurements in Doppler-inertial operation). This latter problem, which would arise in the less frequent sampling imposed by a slower filter cycle, was initially ignored because it was assumed that simple measurement averaging over the slow Kalman computational cycle was a solution. This assumption led to the appearance of an unanticipated problem late in the development.

#### 3.4.6 Transition Matrix

The early effort in this area has already been discussed in some depth. The need for evaluation of the preliminary formulation developed at that time had already created the first need for a new tool with capabilities beyond those of ensemble Programs I and II. The system simulation program (SSP) developed to fill this and many other roles in the overall Kalman filter development is described later in this chapter.

### 3.4.7 Measurement Averaging

One of the more stringent design goals of the C-5 navigation system design was the short time allowed for the alignment and calibration of the inertial platform prior to inertial or sugmented inertial navigation. This led to the attendant opecification of the Kalman filter as the only approach believed capable of achieving the desired calibration levels in such short times. However, as with all statistical filter methods, success of the Kalman filter depends on the smouth of measurement data available in a specified time interval.

As an example, the velocity signal to be used as the basic ground alignment measurement would be corrupted by the velocity errors produced by the effects of wind and cargo loading on the parked aircraft during the alignment period. The time required to complete a single Kalman computational dyple was expected to be about one minute. However, if the standard Kalman formulation, which car accept only a single measurement in each cycle, were used, only about one noisy data sample could be processed each minute. On the other hand, it had already been determined, almost at program inception, that data sampling rates of at least one sample per second were recuired for adequate noise smoothing<sup>4</sup>.

To this end, it was proposed simply to pre-average the (once per second) velocity readings over the Kalman interval and use the result for the measurement once per minute. This seemed such an obviously satisfactory solution that it was accepted on the spot, and attention was directed to other areas. Much later in the program this decision was re-examined. A small ensemble error analysis program was constructed to quantitatively evaluate the performance of this averaging process with respect to platform drift rate recovery. The results were unsatisfactory: rather than converging, the solution diverged - to a point far beyond that required to meet minimal ayetam navigation performance goals.

Analysis of this phenomenon led to a simple conclusion. Not only was the signal noise being smoothed (as desired) by the averaging process, but so was the signal itself! This created serious driving error (e.g., platform drift rate) identification problems for the standard an formulation, which is based on use of signal samples rather than signal averages.

\* This information had been obtained from a small, special-purpose alignment evaluation program, incorporating a simplified, single-axis leveling loop model.

307

STATE AMAGE VILLE

Add and the state of the state

A modification of the Kalman measurement processing formulation, designed to accommodate averages rather than samples, was therefore evidently necessary. This was undertaken, and led to a simple modification of the stundard formulation. Finally, a simple ensemble error analysis program was constructed to evaluate the effect of this method. Figure 11° shows a sample of the successful results of compensated averaging, as well as the divergent characteristics obtained with uncompensated averaging.

The modification developed is based on the assumption that one may rofer measurements to a lixed epoch (time point) of the state vector<sup>†</sup>:

$$r(t) = M(t)\Phi(t;t_0)x(t_0) + q(t)$$
 (3.22)

If n measurements  $y_i(t_i)$  in the interval  $t_a \rightarrow t_a + \Delta t$  are averaged, it follows that

$$-\frac{1}{n}\Sigma y_{i}(t_{i}) = \frac{1}{n}\left[\Sigma M_{i}(t_{i})\Phi(t_{i};t_{o})\right]x(t_{o}) + \frac{1}{n}\Sigma q(t_{i}), \qquad (3,23)$$

Since  $q(t_i)$  is assumed to be a random variable, an averaged measurement can be defined as

$$\vec{y} = \vec{M}x(t_{o}) + \vec{q}, \qquad (3.24)$$

where

$$\overline{\mathbf{y}} = \frac{1}{n} \Sigma \mathbf{y}_{i}(\mathbf{t}_{i})$$

$$\overline{\mathbf{u}} = \frac{1}{n} \Sigma M_{i}(\mathbf{t}_{i}) \Phi(\mathbf{t}_{i}; \mathbf{t}_{o})$$

$$\overline{\mathbf{q}} = \frac{1}{n} \Sigma \mathbf{q}$$

A DOMAS A

246

#### n = number of measurements.

The technique just described was incorporated into the design. Figure 12 compares this formulation with the normal Kalman formulation.

#### 4. DESIGN EVALUATION AND REFINEMENT

As already mentioned, evaluation of the effect on system performance of many important, specific areas of the basic Kalman filter design lay beyond the capabilities of any existing closed-formula theory, as well as these of the ensemble error analysis Programs I and II. These areas included the single-word floating-point arithmetic, approximations to the elements of the transition and data average observation matrices, the linear approximation to system error behavior, and the like.

Such evaluation could, of course, be left until it could be accomplished with the real system hardware, as soon as this became available. However, not only would this kind of an approach be highly wasteful of precious development time, but also - however carefully designed and sonitored - feal system tests can rarely provide the kind of environmental control essential to understanding all-mode system performance mensitivities to individual effects such as those which were in question here. The considerations led to the development of a digital simulation of the C-5 aircraft navigation system. This section describes this simulation, and the way in which it was used to evaluate, validate, and refine the filter design.

# 4.1 Simulator Description

Simulation, as the name implies, is only an approximation to reality. Since the C-5 simulator was, in fact, a program designed for use in a general-purpose digital computer (IFM 7090 or 360), its approximations were embodied in the mathematical models used to simulate real system hardware operation.

Figure 13 is a functional block diagram illustrating the anjor elements which are required to simulate the marigation system. As illustrated, a flight profile simulator is necessary for generating the true specific force,  $\vec{t}_1(t)$ , acting on the IMU, and for supplying to the measurement simulation the true position and velocity, denoted by the vector Z(t). In particular, the flight profile simulator was made sufficiently versatile to simulate take-off. landing, cruise, and in-flight mneuvers of the C-5 aircraft.

The IMU simulator accepts the true specific force from the flight plan simulator, and the commanded platform torquing rates from the 1051A mirborne computer simulator. The Kalman filter, which is a part of the 1051A simulation, accepts information from the simulated inertial navigation computations and from the measurement simulator, and feeds back corrections to the navigation simulation.

The single-cycle time of only 10 seconds used for generating Figure 11 results was based on a very early (incorroct) assumption about the Kalman filter computational complexity. For corresponding results for cycle times more appropriate to C-5, see Reference 14.

<sup>†</sup> The notation here and in Figure 12 has been simplified for brevity by means of the symbols

**Ι-**Ϊ-Î, **J-**Υ-Ŷ-Δγ.

# 4.1.1 IMU Simulation

The IMU model used in the simulation program is illustrated in F m = 14. The vector  $\vec{I}_1$  is the true specific force obtained from the Flight Profile block of Figure 13. The matrix  $T_{p_1,p_1}(t)$  is the transformation from the coordinate system in which  $\vec{I}_1$  is computed to the platform reference frame. In particular, corruption of the specific force vector by accelerometer-rolated errors is simulated, as shown across the top of the figure (see also Figure 5), and results in the measured specific force vector,  $\vec{I}_0$ . The corruption of the figure (see also Figure 5), and results in the measured specific force vector,  $\vec{I}_0$ . The corruption of the figure (see also Figure 7). In addition to the gyro errors discussed earlier and shown in Figure 7, this more detailed model includes a breakdown of gyro drift rates,  $\vec{\omega}_0$ , into bias, programmed time-function, and mass unbalance drift rates. Constant of these for the  $i^{\text{th}}$  gyro (i = 1,2,3) is represented in this figure by the box  $G_1$ . Generation of the mass unbalance drift rates, in particular, requires resolution of the specific force vector into the matrix  $T_{p_10}$ , which produces the resolved specific force vector  $\vec{I}_0$ . The overall resultant rate vector  $\vec{\omega}_1$  is the actual inertial rotation rate of the platform in platform reference coordinates.

Updating of the matrix  $T_{\text{Plup}}$  is done in a separate portion of the program in accordance with a sot of matrix differential equations<sup>3</sup>. In addition to  $\vec{\omega}_{1}$ , these equations also require as inputs the components of the inortial rate of rotation of the true locally tangent, ideal wender azimuth angle frame in this frame itself. These components are, in turn, generated by the Flight Profile computations, using  $\vec{f}_{1}$  resolved into this ideal reference frame.

Some of the simplifying assumptions built into this simulation were:

- (a) The measured specific force and the gyro torquing rates are continuous.
- (b) The inherent, dynamic instrument lags of the gyro and the accelerometers are negligible.
- (c) The dynamic lags of the platform serve loops are negligible.

These simplifying assumptions, which were mandatory if the simulation were to run at efficient speeds - i.e., much faster than real time, were motivated by the fact that the principal natural frequencies of inertial system error propagation are extremely low (e.g., Schuler frequency 1 cycle per 84.4 minutes). With respect to these, the lags and sampling rates of the above neglocted effects lay in ranges which were, respectively, too short and too high, to effect output errors of interest.

# 4.1.2 NDC-1051A Navigation Equation Simulation

Figure 15 shows some of the details of the computations performed in the 1051A computer. As already discussed, laboratory calibrations are performed on each IMU to establish the elements of the calibration vectors and matrices shown in the block diagram. For example, the null bias calibration will cancel the effects of vector  $\vec{n}_b$  of Figure 14, except for errors in the calibration and variations in  $\vec{n}_b$  caused by environmental factors.

Under idealised conditions of perfect calibration, with reference to Figures 14 and 15.

$$[\mathbf{n}_{11}] = [(\mathbf{T}_{\mathbf{p},\mathbf{A}})(\mathbf{SFA})]^{-1}$$
(4.1)

and

$$[\mathbf{g}_{ij}] = [(Brq)T_{01p}]^{-1}, \qquad (4.2)$$

Also the  $[u_{ij}]$  matrix and the constant drift rate calibration vector produce a commanded drift rate. This will exactly cancel the fixed and static mass unbalance drifts of the gyros under the idealized conditions. This type of calibration "marries" the specific IMU and its 1051A computer so that the remaining errors are small. These same matrices are used in the simulation of the C-B microaft mavigation system. Inputs to the simulator control the assumed calibration errors for any specific case.

The wander asimuth equations are solved by numerical integration. No attempt is made to simulate any part of the actual 1051A arithmetic for the operations shown in Figure 15. Parallel studies in the development of the real-time program had shown that the errors introduced by this assumption are insignificant.

#### 4.1.3 Kalman Filter Simulation

Rather than attempting to design an all-mode simulator, different versions of the simulated Kalman filter sub-routine were programmed for each mode under study. This approach was used to reduce complexity and run time. Also, special versions were developed to isolate causes of certain problem areas. For example, versions with and without the 14-bit arithmetic mentioned below, and with and without data averaging, were constructed.

For final validation, the sequence of calculations, the 14-bit floating-point arithmetic, and all other factors were simulated to a high degree of realism. To this end, a special machine language sub-routine was constructed for masking the 7090 word to simulate the 14-bit floating-point arithmetic. This simple sub-routine is used in the appropriate computations to introduce the additional errors caused by the 14-bit floating-point arithmetic. Relative care was used in the order of calculation to preserve as much accuracy as possible. These constructed for used in the real-time program for the same purpose.

309

1 1 400

state and states and states

1. 4. 78

in e un freren CE and

All values of all the controllable quantities (for example, the constants of the R matrix, the value of the 's gain factor, the variance of the random error in observation, etc.) are controlled by input and/or overlay of block data. Observation sequences, Kalman cycle time, etc., are also input in a similar manner.

2011年1月1日日

# 4.1.4 Observation Simulation

The observation simulation models necessary for the various modes of study were made a part of the Kalman filter sub-routine. Models were simulated as realistically as possible to include all the relevant error sources. These errors, which could only be described by a random process, were included by using a selected sample or samples. This was justified on the basis that the primary purpose of the simulation was to detect and correct major problems. Once problem areas are corrected, the real system should behave in a reasonably linear manner as far as the error propagation is concerned.

# 4.2 Example Uses of the Simulation Program

It should be reasonably clear from the preceding discussion that the digital computer simulation program is not only quite versatile but can also be readily modified. By means of appropriate input data, values can be selected for quantities controlling

- (a) the flight plan,
- (b) the initial errors,
- (c) the constant and time-varying IMU errors,
- (d) the observation sequence and the Kalman cycle time,

# as well as many others.

To a novice, such versatility poses a problem of where to begin. This problem can be beat addressed by simply reviewing what the program was designed to test. The possible problem areas in the Kaluan filter could be expected to fall in the following categories:

- (a) Nonlinear phenomena.
- (b) Numerical inaccuracies and problems.
- (c) Effects of inherent computational dolays.
- (d) Unmodeled or improperly modeled errors.
- (e) Approximations in the equations.

It was therefore planned to set up computer runs designed to exercise the simulated filter in a worst-case sense with respect to each of these categories. Analysis of the resulting data would then undover problems, if present. Although selection of a worst case is by no means a simple problem, theory can be used to provide sensible guidelines. For example, the basic design is based on linear theory. Hence, one of the first questions is "how linear is this system?". To obtain the answer, the simulator was used to generate time historics of actual mavigation system errors for various sets of initial errors. These were compared to errors predicted by the transition matrix in accordance with

# $\hat{\mathbf{x}}(\mathbf{t}) = \Phi(\mathbf{t}; \mathbf{t}_a) \hat{\mathbf{x}}(\mathbf{t}_a) \ .$

These runs validated that, for all practical purposes (i.e., within reas nable values for  $\tilde{x}(t_0)$ ), the system errors propagate linearly. Similar checks were made on the measurement functions. These also indicated that the residuals were linearly related to estimate errors for all practical purposes. Hence the linearity assumptions had been proved sound.

In checking the remaining areas, a two-step process was generally adopted:

- (a) Demonstration that the filter behaved normally when the only errors in the system were those included in the filter.
- (b) Adding unmodeled errors to determine degradation in performance or the existence of otherwise hidden problems.
- 4.2.1 Fine Ground Align Mode Description

The purpose of the fine ground alignment mode is to:

- (a) Precisely level the platform. Platform deviations from level are called tilts.
- (b) Obtain accurate coincidence between platform and computer azimuths. The difference between the azimuths is called a heading error.
- (c) Precisely estimate drift rates of the level channel gyros (gyros 2 and 3).

The fine ground align mode starts after 10 minutes of warm-up, during which time coarse alignment of the platform is accomplished. At the start of fine ground alignment the platform is thus reasonably close to its desired attitude. Tilts are fractions of a degree, and coarse gyro-compassing has reduced the besking error to a few degrees.

Latter y

STORAGE STREET

Figure 16 is a simplified schematic of the operation of a single level channel during fine ground alignment. As illustrated, the accelerometer on the platform senses a tilt proportional to gravitational force times the sine of the tilt angle  $\phi$ . The velocity register output, v, will ramp as a result of the tilt angle. Because of platform drift rate, an approximate quadratic effect (in time) will also exist, as shown in Figure 17. The total output shown is the sum of the two effects.

The data averaging described earlier samples the velocity register at a high rate to accumulate an average reading. At 45 seconds, this average is input to the Kalman filter, which computes estimates of the state errors. However, because of the time required for their computation, these estimates are not applied until the end of the second cycle. At this time, the velocity register is reset to a small value by the estimated velocity error, and leveling rate and platform drift rate estimates are also applied.

buring the third cycle, the platform is almost perfectly leveled by the dommanded leveling rate<sup>†</sup>. In subsequent cycles, the residual quadratic sawtoothing of the volocity register is steadily reduced as the platform drift rate is more and more accurately estimated and compensated.

During the operation just described, the platform is held in a single, fixed-azimuth orientation which, to facilitate the discussion, is here taken us cardinul (i.e., the platform axes are aligned approximately north and east). In this orientation, the filter ufter a short time forces the platform into a steady-state, level condition in which, although the input axis north gyro drift rate has been accurately calibrated, the input axis east gyro has not. Instead, the platform (and therefore the input axis of this gyro) has a heading error just sufficient to couple enough of the Earth's rotation rate\*\* into this gyro to cancel its drift rate. Thus, after single-position gyrocompassing, as the above operation is called, both a significant heading error and an cest gyro drift rate remain, although the total platform drift rate about the east axis is compensated.

To remove these important remidual errors, two-position gyrodompansing is employed for C-5 fine ground alignment. That is, after a 8- or 7-minute period in the first position, the platform is slewed 90 degrees in azimuth, and the first-position process is repeated. Because of the reorientation of the level gyro input axes, however, the heading error is removed, and the input axis east (now north) gyro is also calibrated during this second operation.

## 4.2.2 Nominal Results of Fine Ground Align Mode Simulation

Į

μ

The Kulman fine ground alignment technique just described is most easily evaluated when no unmodeled errors are included in the simulation run. This is because unmodeled errors produce extra error effects at the modeled variable levels which unfortunately do not behave in accordance with the filter model. For example, a scale factor error in the input axis morth gyro in the first position produces an apparent fixed drift rate about the north axis (in the horizon plane), due to the torquing rate applied to this gyro to counter the Earth's rotation rate in this direction. This is duly compensated by the filter in the first position by means of a fixed re-bias rate. After repositioning, however, this scale factor effect is abaent, since no Earth rate compensation torquing is applied to this gyro when its input sits is west. However, the compensating re-bias rate remains to produce an effective drift rate, and therefore a balancing heading error, in the second position.

Since such effects would unnecessarily complicate the evaluation of performance with respect to other specific filter design areas, unmodeled errors were not included in the simulation runs used to evaluate the fine ground alignment filter design until the design had been essentially completed and validated.

In addition, in order to establish a fixed frame of reference for comparisons between runs, a single set of nominal values for modeled errors was used, as well as a nominal alignment configuration. Specifically, all runs simulated two-position fine ground alignment at 45 degrees north latitude, with the input axes of gyros 2 and 3 representively east and north in the first position, and north and west in the final position. The time spent in each position was about 7 minutes, which, together with the 1-minute siew, accounted for the overall maximum goal of 15 minutes<sup>††</sup>. As to modeled errors, initial tilts about axes 2 and 3 were assumed to be respectively -0.32 and +0.40 degree, gyros 2 and 3 were each assigned a +0.02 degree per hour drift rate, and the platform heading error was initiated at +2.0 degrees.

Figures 18, 19, and 20 summarize the results of a simulation run with this nominal configuration, after most of the major problem areas had been ironed out. In particular, Figure 18 is the simulated time history of the north velocity register for this run. The ramps due to the large initial platform tilts in each position are evident. The velocity error due to this tilt is largely removed at the beginning of the third Kalman cycle in each position, based on the average of the velocity data collected in the first Kalman cycle. Based on the same data, the initial tilt is also accurately estimated and largely removed by the application of an appropriate platform leveling rate during the third cycle. In subsequent cycles, the drift rate and its tilt and velocity error effects are gradually estimated and removed.

- This was the expected duration of the final Kaiman cycle time widway in the development. It was reduced to 22.5 seconds later in the program.
- This leveling is reflected in the velocity register readings, whose instantaneous slope is proportional to the instantaneous tilt of the platform.
- \*\* Specifically the northward component, in the horizon plane.
- \*\* Por investigation purposes, operation in the second position was often extended beyond 15 minutes overall, and alignment performance in inter-cardinal orientations of the platform were also examined.

Although velocity and platform tilt errors are thus quickly removed, calibration of the level gyro drift rates and removal of the heading error is a much slower process. This is illustrated in Figures 19 and 20, from which it is evident, in particular, that significant calibration of the number 2 level gyro drift rate is still proceeding at the end of the allotted 15 minutes of alignment time".

## 4.2.3 Effects of Large Unmodeled Errors

Early isboratory results for the ground might mode were obtained before the calibration constants of Figure 10 were available. Figure 21 shows a typical time history of the north velocity register obtained from the actual system at that time. Comparison with Figure 18 shows that the behavior of the roal system was far from nominal. It was recognized, however, that large unmodeled errors could easily produce such performance. Since the final system would be precalibrated, there was little need for concern if the effects seen in Figure 21 were indeed caused by these errors.

The simulator was used to confirm this hypothesis: Figure 22 shows the north velocity time history obtained from a simulated fine ground alignment, for which large uncellibrated scale factor errors on the gyros were assumed. The unmistakable similarity between the simulated and actual velocity responses in Figures 22 and 21 gave some confidence that the irregularities observed in the real system results were in fact caused by its lack of collibration, rather than any filter design deficiencies. In particular, it is evident that, in each case, the platform was not adequately leveled by the command rate applied during the third cycle. In the simulation, this was known to have been caused by the large (10%) gyro scale factor error, which produced a net leveling displacement of the platform during this cycle which was 10% larger than that commanded. The resulting throughout later phases of system test, to identify problems due to lack of, or to faulty, system calibration,

#### 4.2.4 Problems Caused by the 14-Bit Floating-Point Arithmetic

As mentioned earlier, several versions of the sub-routine which simulated the on-board filter computations were available. This made it possible to address the ovorall problem in a reasonably orderly manner. For example, it was first established that satisfactory results were obtained with the single precision floatingpoint 7090 arithmetic (28 bits and sign), before simulating the 14-bit arithmetic.

It was also suspected, on the basis of prior experience, that there would likely be problems in maintaining a positive definite covariance matrix. As described earlier, however, two techniques for combating potential numerical problems and unmodeled errors had been provided<sup>6</sup>. To recapitulate, these two techniques were the addition of the R matrix in the time update operation

$$P(t) = \Phi P(t_{\lambda}) \Phi^{T} + R$$
, (4.3)

and the "epsilon" technique in the measurement update operations:

$$\mathbf{P}_{\mathbf{x}} \simeq \mathbf{Y} - \mathbf{P}\mathbf{M}^{\mathsf{T}}\mathbf{M}\mathbf{P}/(\mathbf{M}\mathbf{P}\mathbf{M}^{\mathsf{T}} + \mathbf{C}) + \mathbf{M}^{\mathsf{T}}[\epsilon^{2}/(\mathbf{M}\mathbf{P}\mathbf{M}^{\mathsf{T}} + \mathbf{C})]\mathbf{M}.$$
(4.4)

Neither of these techniques, however, could provide a satisfactory solution to the numerical problem that occurred as a result of the use of the proposed 14-bit. floating-point arithmetic. This was the loss of positive definiteness (i.s., the occurrence of negative dirgonal terms) in the covariance matrix on the first or second filter cycles in the simulated fine ground alignment mode. The problem did not occur initially, when the standard, Kalman discrete-measurement formulation was simulated, but only later, the first time the new, averaged-measurement formulation was tried. This tended to cloud the fundamental source of difficulty. Previous experience had definitely indicated that the most likely problem area would be in the execution of Equation (4.4). The principal changes in this equation introduced by the averaged-measurement formulation were:

(a) the M vector had wore non-zero elements;

(b) the C scalar (variance of the random error in measurement) was significantly smaller.

Once these factors were recognized, the value of  $MPM^T$  was compared with the value of C. The problem was then immediately recognized. C was so small that early in the sequence (when P is large) it was effectively zero<sup>†</sup> with respect to addition to  $MPN^T$ . Also, the third term of Equation (4.4), which contains the "epsilon" gain, was effectively zero<sup>†</sup> in comparison to the second term.

Increasing either  $\epsilon$  or C to prevent the difficulty was not reasonable. The results of such increases would prevent good saturation later in the measurement sequence. Perhaps, in retrospect, an appropriate choice would have been to:

. (a) Test if C < MPN<sup>T</sup>.

(b) If true, assign (MPM<sup>T</sup> + C) = (1+k)MPM<sup>T</sup>, where the quantity k could be selected somewhere in the range 0.001 < k < 0.1.</p>

 However, the level of calibration achieved in this time was more than adequate. This was established by the excellent inertial navigation performance (not shown) achieved in a simulation of a flight following this alignment.

<sup>†</sup> i.e., iess than the last bit in the 14-bit register.





#### The actual choice used was to assign

# $(MPM^T + C) = (1,01)(MPM^T) + C$ ,

The 1.01 factor provented loss of positive definiteness early in the sequence. Later mensurements were weighted slightly less than optimally, but the performance loss was small in comparison to that produced by other factors (e.g., unmodeled error sources).

# 4.2.5 The Approximate Transition Matrix

angalan kukuku angalan sarata

ŀ

The transition matrix for the time-varying linear variational equation

 $\dot{x}(t) = A(t)x(t)$ 

was to be approximated by the formulation given in Equation (3, 19). Two potential problems were evident in this expansion:

(1) Were the second-order terms retained sufficient for adequate prediction across the Kalman interval?

(2) Could the non-constant portion.  $A_{\rm HC}$ , of the A matrix be treated as a constant across the interval?

In the ground align mode the matrix A is constant. Hence, one of the early uses of the simulator was to verify the adequacy of Equation (3.19) for use in this mode. This investigation, quickly completed, resulted in the addition of a few more selected second-order (constant) terms to the elements of the original formulation.

In the case of airborne mode filter operation, the question as to whethu, or not the non-constant elements of A could be treated as constants across each Kalman interval, required a much more extensive investigation with the simulator. This was because results in this area were (not surprisingly) found to depend strongly on (a) the hature and extent of the particular mansuvers assumed, (b) the Kalman cycle time, and (c) the order in which filter operations were carried out (see Table I).

At the time of writing, these investigations were essentially complete, and the constancy assumption had been proved valid, subject to certain conditions on the above factors which were realizable in the C-5 application. However, any detailed discussion of this area, which, it is to be noted, has general importance far beyond the C-5 application, is outside the scope of this chapter.

# 4.3 Simulated Ground Align/Barometric-Inertial Performance\*

Figure 23 shows the gyro calibration history for a simulated fine ground alignment, in which all errors believed to be significant (modeled and unmodeled) were included. In this case, unlike that of Figure 19 where only modeled system errors were included, the final, steady-state, level gyro compensation values attained represent lumped compensations for the combined effects of gyro bias drift rates and the effective drift rates produced by gyro scale factor and input axis alignment calibration matrix errors.

It is evident from Figure 23 that essentially no calibration of the number 1 (azimuth) gyro drift rate is attained in the overall 15-minute fine alignment mode period. This is because the sensitivity of the level velocity register averages, which are the Kelman filter inputs in this mode, to azimuth drift rates is considerably less than it is to level drift rates. This, however, is a limitation, not of the Kalman filter approach, but rather of the basic, two (level) position gyrocompassing technique; no other filter type could (statistically) produce a significant aximuth drift rate calibration within the 15-minute alignment time using the same positioning sequence and geometry.

Fine alignment performance in the presence of the unmodeled gyro errors can only be honestly evaluated, by simulation of follow-on barometric inertial flight. The goal (see Section 2) is a barometric-inertial navigation CEP growth rate during ten-hour flights of no larger than 0.75 nautical mile per hour during the first five hours and 1.25 nautical miles per hour in the second five hours.

Figure 24 gives the navigator performance in terms of the actual eas\* and west position errors that accumulated for this particular set of unmodeled gyro errors. Shown also are the "ensemble" results taken from the Kalman filter covariance matrix. This was a simulation of a barometric-inertial flight, starting at 45 degrees latitude, accelerating east to 600 ft/sec horizontal velocity, climbing to 33,000 feet altitude, and thereafter oruising at constant altitude and speed on a great circle path. It is seen in the figure that the actual east position error exceeds the standard deviation obtained from the Kalman filter covariance matrix. This was not unexpected because:

(a) The 16-by-15 Kalman filter covariance matrix does not include all the error sources.

(b) The sample errors chosen were larger than would, on average, be obtained from a statistical sample.

The overall trends of the two curves (actual and atandard deviation) have the same behavior. This gives confidence that the simplifications used in the filter are appropriate for long-time propagation.

<sup>6</sup> We vertical channel of a three-degrees-of-freedom free-inertial navigator is unstable. For the C-5 navigator, stability is maintained by using the generalized all-mode Kalman filter with an altitude observation based on the barometric altimeter. Therefore, this mode of operation is referred to as "barometric-inertial" instead of free-inertial.





Costen Constitution Constitution



Previous ensemble analysis had shown that straight flights produce slightly worse barometrid-inertial performance than flights which have turns. It was therefore decided to establish a flight that included a turn, as a further test for the filter in the barometric-inertial mode.

The path chomen was a level accoloration in the north direction to 600 ft/sec, simultaneous with a climb to 33,000 feet. This was followed by cruine for 26 minutes, then a 90-degree right turn taking 60 seconds, and finally a long terminal great circle cruise on the new leg at 600 ft/sec. Figure 2D gives the navigation results for these conditions. Alignment and error conditions were identical to these used in obtaining the results of Figure 24.

As can be seen in comparing Figures 24 and 25, the actual errors are slightly smaller for the latter. This is a (one-case) confirmation of the date obtained from the ensuable analysis programs.

An interesting property of the results of Figures 24 and 25 is that sustained oscillations at both earth rate frequency and Schuler frequency are contained in the errors. This is particularly evident in the north position error time histories. The approximate error model used in the Kalman filter reflucts these same oscillations and retains a fairly good phase relationship. This gives confidence that the entire Kalman filter error covariance matrix does remain a good model of the actual sets of errors. This is not too important for fine alignment followed by otherwise unaugmented barometric-inertial operation in the subsequent flight. If, however, additional data such as a LORAN position fix becomes available later in the flight, correct weighting of the observation to improve mayistion and to calibrate the IMU depends on the validity of the covariance matrix,

#### 4.4 Laboratory Test of Ground Align/Barometric-Instal Performance

Reliable confirmation of the design of the C-5 Kalman filter in the fine ground align mode of operation became possible for the first time only when the IMU precalibration program checkout was completed, and this program became available for use in determining the gyro and accelerometer error compensation matrix elements (see Section 4.1).

This allowed laboratory system fine align tests to be conducted for the first time with calibrated IMU's. Since filter design had been producated from the start on the use of much precalibrated IMU's and guite a few simulation results incorporating this assumption were already available, extensive, direct, and meaningful comparisons between actual and simulated performance could now be carried out.

Figures 26 and 27 show, respectively, the east velocity register and the drift rate compensation histories during typical laboratory system fine align tests with calibrated IMU's at this stage of the program,

Comparison of Figures 25, 21, and 18 reveals that IMU calibration had removed the mis-modeling difficulties, the effects of which are evident in Figure 21, so that velocity register behavior of the calibrated system was very similar to that shown in Figure 18 for a simulated, *perfectly* precalibrated system.

Comparison of Figures 27 and 23 further substantiates that the (14-variable) filter-incorporated system error model was adequate, as indicated by the highly similar simulated and laboratory system test results shown for the gyro drift rate compensation histories during ground align.

In addition, the laboratory test depicted in Figure 27 was extended (as shown) to establish azimuth drift rate (gyro 1) compensation behavior if more alignment time were available in a real C-5 operational situation\*. It was evident that, despite the weak sensitivity involved in extracting azimuth drift rate information from level velocity averages, complete calibration could nevertheless be accomplished in about half an hour<sup>†</sup>.

Finally, Figure 28 shows the inertial navigation performance of a typical laboratory system following nominal (i.e., 15-minute, two-position) fine ground align. The excellent performance (relative to the performance goal of 0.75 mautical mile per hour for the first five hours (see Section 2.3)) shown here was a strong overall confirmation, beyond that obtainable from operation in the fine align mode only, of filter design adequacy.

#### 4.5 Flight Test of Ground Align/Baroaetric-Inertial Performance

Although results (such as that shown in Figure 28) clearly demonstrated adequate performance of the C-5 filter design for ground align in the laboratory, and proved a high degree of confidence for its expected performance in this mode in the field, *demonstrated* field capability could, of course, only be obtained in actual flight test.

At the time of writing, performance results obtained from the first series of flight tests have confirmed not only the adequacy of filter design for fine ground alignment, but for barometric-inertial navigation as well.

In addition, these results have given a high degree of confidence that the design will also prove adequate in the remaining system modes of filter operation, which will be evaluated in the test program phases now being entered.

Calibration of the asimuth, as well as the level gyro drift rates, provides even better subsequent inertial mavigation.
† This had been predicted by earlier simulation results (not shown).











ti in the state of the state of

N.2-MAR CARLES LONGIN CARLON



# 5. SUMMARY

The C-5 Kalkan filter software development effort, which has been outlined in this chapter, resulted in a final design which was optimum in the practical sense that it

- (a) met the software performance specifications;
- (b) fitted into the real-time, on-board computer;
- (c) act the pertinent, overall software development schedules.

To the extent possible, modern sample-data estimation and control theory was not only used in developing the design, but in some areas actually extended (e.g., the "epsilon" and averaged-measurement techniques). However, as is typical of most practical problems, in a few areas, because theory was inadequate or development time limited, approximate solutions had to be constructed, based on engineering judgment and experience. Much of the development effort was devoted to the constructed, use of the three major error analysis and usstem simulation programs (Programs I, II, and III). These programs were used to validate or reject alternatives at almost every major filter design decision point throughout the program. Costly redesign and tape reprogramming offorts late in the system development were thereby avoided. Program III. in particular, also proved an invaluable tool for isolation and correction of both hardware and software "but," during system test.

# 6. FUTURE TRENDS

1

It is only in the past few years that digital computers have become available which are sufficiently fast, capacious, light and small to render feasible real-time on-board solutions of problems the size of that mechanized in the C-5 computer. The rapid advance in technology which created these computers continues: next-generation models, with vastly improved capabilities, will soon be available. In particular, the use in such systems of advanced, optimal, information processing techniques like Kalman filtering, which were hitherto ignored because of their attendant memory and real-time cost, have now become feasible. Further, the superior performance available in practice from such techniques, with respect to prior methods, ensures their future as the eventual standard techniques in such applications. This future will be all the more rapidly realized, however, if batter solutions to some of the significant current problems attending the solut mechanization of such techniques are reached. With respect to Kalman filter mechanizations, two principal current problems are:

(a) The Need for an Extremely Accurate System Description. A Kalman filter must incorporate both a model of the dynamic behavior of the system to which it is applied and the statistics of the noise inputs forcing the system. Mechanised performance depends on the fidelity of the mechanised model to the real system, and on the accuracy with which the assumed statistics characterize the actual random input processes. Such information cannot in general be easily or cheaply obtained in the course of a real system development, where time and funds are limited. Further work on appropriate modifications of the Kalman gain equations, which would provide a high degree of insensitivity to system and input mis-modeling or under-modeling, is necessary<sup>2</sup>.

(b) Round-off Error Sensitivity. The standard recursive Kalman formulation for determining optimal measurement weighting involves the use of essentially squared, covariance matrix information. Round-off errors associated with the fixed computer word length, which are introduced at each computational step, produce an unduly rapid rate of weighting error build-up, when compared with what could be obtained with a formulation based on unsquared information. Although several such formulations have recently been advanced, more work is needed to establish whether these do not cost more in attendant speed and memory requirements than they gain in accuracy relative to the standard formulation.



315











ir in the state of the state of

and a second state of the second state of the

X

ĩ

# REFERENCES

2.57

HK X

11 11

A JANTING

1. Stein, Kenneth J.	Self-Contained Avionics Broaden's Scope of C-5 Missions. Aviation Week and Space Technology, Vol.87. No.21, November 20 1967, pp.192-202.
2. Kalman, R.E.	A New Approach to Linear Filtering and Prediction Problems. American Bodiety of Mechanical Engineers Transactions, Journal of Basic Engineering, Vol.82, Series D, No.1, 1900, pp.35-45.
3. Broxmeyer, C.	Inertial Navigation Systems. McGraw-Hill, New York, 1984.
i. Smith, G.L. et al.	Application of Statistical Filter Theory to the Optimal Estimation of Position and Velocity on Board a Circumlunar Vehicle. NASA Technical Report R-135, Washington, DC, 1962.
5. Scimidt, 6. F.	State Space Techniques Applied to the Design of a Space Navigation System. Pro- ceedings of the 1962 Joint Automatic Control Conference, Session 11, Paper 3.
). Leondes, C. (Editor)	Advances in Control Systems. pp.1-68, 219-339 in Vol.3. Academic Press, New York, 1966.
7. Sohmidt, S.F.	Estimation of State with Acceptable Accuracy Constraints. Analytical Mechanics Associates, Inc., Palo Alto, California, Technical Report 67-16 (Contract NAS 5-11048), 1967.
). Schmidt, S.F.	Compensation for Modeling Errors in Orbit Determination Problems. Analytical Mechanics Associates, Inc., Palo Alto, California, Technical Report 67-168 (Contract NAS 5-11048), 1967.
). Fagin, S.L.	Recursive Linear Regression Theory, Optimal Filter Theory, and Error Analyses of Optimal Systems. Institute of Electrical and Electronic Engineers, International Convention Record, Part I, Session 30, 1964, pp.216-240.
). Lukesh, J.S.	Simulation of the Nortronics Kalman Filter-Augmented C-5 Inertial-Doppler Navigator. Proceedings of the National Aerospace Electronics Conference (NAECON), Dayton, Ohio, May 1968, pp.1.1-180.
l. Sekiguchi, J. Berg, R.	The C-5 Navigation Computing System. Proceedings of the National Aerospace Elec- tronics Conference (NAECON), Dayton, Ohio, 1968, pp.189-194.
7. Miller, B.J.	Information Processing in the C-5 Navigation Computer. Proceedings of the ION National Space Meeting on Simplified Manned Guidance, Navigation and Control, Cocce Beach, Plorida, 19-21 Pebruary 1968, pp.255-280.

# AUTHORS AND REFERENCE AUTHORS

Berg, Richard G.	Northrop Corporation, Electronics Division, Hawthorne, California.					
Bromasyer, Charles	Raytheon Co., Nassachusetts.					
Fagin, Samuel L.	Sperry Gyroscope Co., Division of the Sperry-Rand Corporation, Great Neck, NY.					
Gunckel, Thomas L. II	Autometics Division, North American Rockwell Corporation, Anaheim, California,					
Kalman, R.E.	Research Institute for Advanced Study, Baltimore, Maryland,					
Leondes, C.T.	Department of Engineering, University of California, Los Angeles, California.					
Lukesh, John 5.	Northrop Electronics Division, Hawthorne, California.					
McGee, L.A.	NASA Ames Research Center, Moffatt Field, California.					
Miller, Bruce J.	Northrop Electronics Division, Hawthorne, California.					
Samuelson, H.R.	Northrop Electronics Division, Hawthorne, California.					

· References 10-12 are not directly referred to in the text; however, they are included for general information.

Schwidt, Stanley F.	Analytical Mechanics Associates, Palo Alto, California.
Sekiguchi, Joh	Northrop Electronics Division, Hawthorns, California.
Smith, G.L.	NASA Ames Research Conter, Noffatt Field, California.
Soremson, H.W.	AC Electronics Division, General Motors Corporation, El Segundo, California.
Stein, Kenneth J.	Aviation Week and Space Technology, McGraw-Hill Co., New York, NY
Steybe, D.E.	Northrop Electronica Division, Hawthorne, California.
Weinberg, John D.	Northrop Electronics Division, Hawthorne, California.

Strate Light and

A TANK A TANK

NE PHAther and Ch

# TABLE I

Kalman Filter Equations

	0p	peration		Mat	rix formula	tion
Obs Res	erv idu	ation al	Δy	3	Y - Ŷ	
Eet Con	im tro	ator Dl	Ŷ	z	f(î)	
Obs	<b>e</b> r1	Ation	٩	-	MPM <sup>T</sup> + C	
Upd	ate		b	=	(P+€I)h <sup>T</sup> Q	- 1
			â	2	<b>x̂</b> + b(∆y)	
			P		P - PM <sup>T</sup> Q <sup>1 I</sup> M	P + N <sup>T</sup> ¢ <sup>2</sup> Q <sup>- 1</sup> N
Tra Cal	ns: cui	ltion Matrix Lation	<b>Φ(</b> Δt)	*	Equation (	3. 19)
Tim			Ŷ	<b>`</b> =	Φ(Δt) 🕯	
Upd	at (		P	2	<b>Φ(</b> Δt)₽ <b>Φ<sup>T</sup></b> (Δ	t) + R(Δ†)
Ely a Con	ite) itri	<b>n</b> 51	• <sup>U</sup> sc	2	E(Ŷ)	
عادمیں ہو		Quantity and	definitio	n		Dimensions
2 = system error estimate vector			n x 1			
P = estimate error covarianc			ovariance	mat	rix,	n x n
$\Phi$ = system error transition matrix			аха			
R = system noise covariance matrix			n×n			
Δy	×	observation residual			1 x 1	
Y	Ŧ	observation				1 × 1
Ŷ		observation estimate			1 x 1	
M	#	pessurement math	rix			1 × n
C	×	observation nois				1 x 1
۹	=	observation res	idual cova	ria	nce matrix	1 x 1
b	=	weighting vector	r			n × 1
I	×	identity matrix				· n × n
¢	21	"epsilon" facto	r			1 x 1
u <sub>so</sub>	z	system control	vector		•	n×1
£()	R	estimator contro estimate to aya	ol algoriti tem	; 194	applies	
<b>g()</b>	2	aystem control : corrections to	algorithm; be syplied	ca tu	sputes system	
Δt	=	period of Kalman	n cyclo			
۵	2	state vector di	mension			

A. 1. 1. A. 1. A.

ST.ST.ST.S.S.

**.** 

~ ~ ~ ~ ~ ~ ~ ~

A 1 7 1 1 4

17-16 0 the second provide the 1 10 7

i ette 🗖 🖅

# TABLE II

< ] •

TANKS OF BUSIES

ļ

# Overall C-5 Error Model

_	Node i			
Brror Snurce	Complete	<b>He</b> chanized		
Inertial Measurement Unit	50	13		
Doppler Groundspeed Drift Angle Vertical Volocity	3 3 3	1 1 0		
Altitude Reference Error Barometric Radar Altimeter	2 3	1 0		
Position Reference Error Horizontal Position	2	0		
Heading Reference Error AHRU	1 .	0		
Totals	67	16		

# TABLE .III

# . C-5 Inertial Navigator Error Nodel

	Node l				
Error source	Complete	Nechanized			
Velocity Error	3	3			
Position Error	3	3			
Platform Attitude Error	3	3			
Gyro Drift Rate Error Correlàted Bias Scale Factor Ramp Mass Unbalance Input Axis Alignment	333	3 0 0 0 0 0			
Accelerometer Error Correlated Bias Scale Pactor Input Axis Alignment	3 3 3 6	1 0 0 0			
Ball Readout	2	•			
Totals	50	13			

• A single state element is used to model TMU readout/Doppler drift angle azimuth error.

319



trestant Production State

in the second second



Fig.1 Simplified block diagram of an inertial navigation system



Fig. 2 Block diagram of Kalman filter operation sequence



Velocity

320

i.

hivis: (\* 15



たいと

Sources States

Fig. 3 Block diagram of C-5 navigation system equipments



Fig.4 C-5 inertial-Doppler navigation equipments







proce

Accelerometer signal

Pig. 5



Q





La Carta Carta

ιų Ka







where:

- a, m gain constant assumed for particular variable
- . = time constant assumed for particular variable
- s = Laplace variable

Fig.8 Random noise model



Antra Cherolander Child



he - Dell's erry Order











his Augerty - 4

tines and all states and the



ETANDARD KALMAN FORMULATION (RECURSIVE, SINGLE-MEASUREMENT PROCESSING)

N NEASURKHENTS (y,)

TIME SEQUENCING

AVERAGED-MEASUREMENT FORMULATION

(RECURSIVE, AVERAGED-MEASUREMENT PROCESSING)

ONE AVERAGED

TIME STOUENCING

\*EXCLUDES SYSTEM FORGING NOISE TERMS



in the second second

، ز به از ف از می

*ݥݛݡݔ*ݯݡݸݷݷݸݥݸݵݸݬݸݔݑݸݾݱݑݸݖݛݾݸݛݛ





. 🕒 . 🗸

. A Marine Carlo C

وستعرب فسيله بالعباب ويتبع وتبعي وسناه ويتبع



K)

770

Fig. 17 Typical single-channel'velocity bistory in fine ground align









330

<u>``</u>





Ę

<u>.</u>







Fig. 22 Simulated north velocity history of an uncalibrated system

331

110 M 10 10 10 10 10

32 . Hantan



Fig. 24 Simulated Parometric-inertial position error time histories (great circle flight).

Ň

MARKAN SARAFES

West Carl

le se entre entre



Fig. 25 Simulated barometric-inertial position error time histories (flight with two great circle legs)



Fig. 26 East velocity history for single-position fine ground alignment in laboratory system test





TELEVICE STORE





Fig. 28 Laboratory system free-inertial asvigation performance following fine ground alignment



# CHAPTER 14 - APPLICATION OF KALMAN FILTERING TECHNIQUES

たいでのないのである。

THE PART

F.S.S.S.S.S.S.S.

222155566

# TO THE APOLLO PROGRAM

# þy

Richard H.Battin and Gerald M.Levine

# Instrumentation Laboratory Massachusetts Institute of Technology Cambridge, Massachusetts 02139, UNA

.

ale and a state first China

gerre - manuari interesses interest internet internet within and in the start. Within a starter and the first a

NOTATION	۰
1.0.1.0.1.0.0	<u> </u>

.

	· · · · · ·
•	measured angle
A <sub>L</sub>	angle with respect to vertical in unknown landmark calculation
•	denominator of weighting vector
8 <sub>d</sub>	disturbing acceleration
•#	semi-major axis of horizon ellipse
Þ	measurement geometry vector
b <sub>0</sub> , b <sub>1</sub> , b <sub>2</sub>	partitions of geometry vector
Ъ <sub>Н</sub>	semi-minor axis of horizon ellipse
C(X)	special transcendental function
0	apead of light
đ	factor in horiaon delculation
E	COVARIANCO MATRIX
1	Sx C partition of covariance matrix
1	general derivative function
f(q)	special function in Encke's method
9	gravity gradient matrix
٠ <b>E</b>	special gravity vector
I	identity matrix
1	unit vector
1AB	unit vector B relative to A
J <sub>kil</sub> , J <sub>kil</sub> , J <sub>ezM</sub>	coefficients of gravity potential functions
k1 - kn - kn	derivative functions
¥	transformation matrix
2	Bero matriz
P <sup>r</sup> k	derivatives of Legendre polynomials
q	measured quantity
a	special variable in Encke's method
R	messured range
LAB	position vector B relative to A
r <sub>s</sub>	earth equatorial radius
L.	position vector of landmark
r <sub>M</sub>	mean lunar radius
rp	radius of primary body
Epv(c)	conic position vector
E <sub>v</sub>	position vector of vehicle
	•

والموافقة والمرجعة والمراجع المحاج والمحاج والمحاج والمحاج والمحاج والمحاج المحاج المحاج المحاج المحاج



Barrene Britten pinter Bart

r <sub>xe</sub>	projection of I <sub>CL</sub>
B(1)	special transcendental function
I	transformation matrix
t, t <sub>o</sub>	time
±0, ±1	horizon vectors
u	projection operator
Unit(g)	unit vector in direction of <u>q</u>
Y <sub>AB</sub>	velocity vector B relative to A
Yev(C)	conic velocity vector
1	error transition matrix
H	6×6 partition of H matrix
¥., · ¥. · · · ¥.	3x3 sub-matrices of g matrix
X X X	vector partitions of T matrix
× .	generalized anomaly in Kepler's equation
x <sup>H</sup>	component of position vector in horizon coordinates
X <sub>¥Ÿ</sub>	component of position vector in moon coordinates
3 <sub>H</sub>	component of position vector in horizon coordinates
JWA	component of position vector in moon coordinates
j.	special vector in measurement incorporation
Ho + Hi + He	three-dimensional partitions of g vector
M <sub>0</sub> , M <sub>1</sub> , , M <sub>8</sub>	components of g vector
α.	angular error in landmark tracking
ត	mean squared measurement error
a <sub>e</sub>	reciprocal of semi-major axis
ß	NR shaft angle
γ	factor in measurement incorporation
Δt	integration time step
Ł	position deviation vector from conic
ð <sub>e</sub>	deviation in quantity q
8 <b>/3 , 80</b> -	RR shaft and trunnion.angle biases
6	position error vector
4	parameter error vector
ข	velocity error vector
θ.	NR trunnica angle
<i>H</i> <sub>1</sub>	gravitational constant of body A

velocity deviation vector from conic

ALC: TYPE ACT

1995

and the

Sec. 14

11-12A

¥

 $\cdot _{i_{1}}$ 







a la contra la contra de la contr

# 338 angle measurement error variance σţ landmark error variance $\sigma_{\rm L}^2$ variance of angular error $\alpha$ $\sigma^{\sharp}_{\alpha}$ co-latitude ¢ special vector in Nyström's method ₫ polar angle component of landmark tracking error χ special vector in Nyström's method Ý weighting vector ω three-dimensional partitions of $\omega$ $\varphi_{0}, \varphi_{1}, \varphi_{2}$

# Subscripts

ĺ

P

2

Ŕ

2 COLORADIAN

C	command module
a	dimension of state vector to be estimated
	earth
N .	horison
L.	lunar module or landmark
×	100m
	pessured
	stage in integration time step
,	primery body
Q /	secondary body
▲ 2	sub
•	star
۷	vehicle
X , Y , B .	coordinate axes

# Superscripts

r	transpose of vector or matrix				5
•	altered variable	•		·	

# Abbreviations

M		gommand module
	•	command module computer
	166.	display and keyboard
		inertial measuring unit
1.14		lunar module
1.147		lunar module computer
		rendezvous radar
		somning telescope
1901 1		eastant
BAT		ware high frequency
AN.		

and the second se	. الجيسة بيجيش بيدراسي
and the state	
المتعقب محتر	
ليعشد شيتكر	
- 10 - 14 A - 10 - 10 - 10 - 10 - 10 - 10 - 10 -	Inter & Cambons
HALDAS A	
naunden 🗰 🗤 s	
an ganger an	
HALLBAR WY	
а а. доба <b>ция</b> на с	



WINAM MARC

and an and a start of the second s
# CHAPTER 14 - APPLICATION OF KALMAN FILTERING TECHNIQUES TO THE APOLLO PROGRAM

# Richard H.Battin and Gorald M.Levine

## 1. COASTING FLIGHT NAVIGATION

Navigating the Apollo spacecraft in consting flight involves two processos. First, frequent revigation measurements are made to improve the estimate of the spacecraft's position and velocity. Record, a prediction of the orbit is made periodically so that small corrections to the speed and direction of motion can be applied using the propulsion system if the spacecraft is not following the intended course.

Predicting the course of Apollo during prolonged periods of coasting flight is the same as the astronomer's problem of predicting the positions of the moon and planets. There are several considerations which influence the ability to make long-range predictions. First of all are the mathematical techniques used for solving the equations of motion. Unless rather elaborate computational techniques are employed, numerical errors will propagate and rapidly degrade the solution. Second, the accuracy of predicting position and velocity also is subject to our knowledge of the physical properties of the solar system. Finally, and most important of all, is the accuracy of initial conditions - the values of position and velocity at the time from which the prediction is made.

In order to insure accurate initial conditions, it is necessary periodically to correct the estimate of spacecraft position and velocity using data gathered from optical or radar measurements. Use of a space sextant allows the astronaut, for example, to measure the apparent elevation of a star move the earth's horizon or to measure the angle subtended by the directions to a star and to a landmark on the moon. At the time a measurement is made, the best estimate of the spacecraft's position and velocity is contained in the on-hoard digital computer. Then, since the directions of the astar and the locations of landmarks are known, it is possible to calculate the spected value of the angle to be measured.

When the expected value of this measurement is compared with the value actually measured, the difference can be used to correct the estimate of the spacecraft's position and velocity. A sequence of such measurements separated in time, together with an accurate mathematical description of the solar system, will eventually produce estimates with sufficient precision to permit corrective maneuvers to be made with confidence.

### 1,1 Navigation Instruments

To accomplish the lunar landing objectives of the Apollo mission, two vehicles are used - the command module (CM) and the lunar module (LM). Each vehicle is equipped with mensors for enquiring navigation measurement data, together with a digital computer for information processing and orbit prediction (see Reference 1). By appropriate utilization of these instruments, the Apollo astronauts can molve the three major coasting flight navigation problems of the lunar mission: (1) dislunar-middourse navigation of the CM to and from the moon; (2) navigation of the CM in orbit about the moon; and (3) navigating both the CM and LM during the rendezvous phase in lunar orbit.

Reference orientation is maintained in both vehicles by means of a device called an inertial measuring unit (IMU). This instrument is basically a small platform supported and pivoted so that the spacecraft is free to rotate about it. On this platform are mounted three gyromcopes that sense and prevent any rotation of the platform from occurring.

In the command module a rigid structure mounted to the spacecraft, called the pavigation base, provides a common mounting structure for a telescope, a sextant and the base of the IMU gimbal system. Precision angle transducers on  $c_{\rm tell}$  of the axes of the optical instruments and on each of the axes of the IMU gimbals permit the indicated angles to be processed in the command module computer (CMC) to generate the components of the optical targets in inertial system stable-member coordinates.

The sextant (SXT), shown schematically in Figure 1, is a 28-power, narrow-field-of-view instrument having two lines of sight. One of these lines of sight is fixed to the spacecraft and is simed at a landmark or the horizon by turning the vehicle in space using orientation commands to the attitude control system. The second line of sight can be pointed at a star through the use of a two-sxis hand controller. One sxis of this motion, the shaft axis, is parallel to the landmark line of sight. The shaft drive changes the plane in which the navigation angle is measured by rotating the head of the instrument as a whole. The trunnion drive asts the navigation angle by tilting the trunnion axis mirror. By superimposing the two images in the field of view and signaling this event to the computer by depressing a mark button, the navigator can measure the angle between the lines of sight to a star and either a landmark or the horizon.



The Scanning telescope (SCT), shown in Figure 2, is a unity-power, wide-field-of-view instrument having a single line of sight. Both shaft and transion control of this line of sight is possible. The shaft angle is made always to follow the sextant shaft angle, while the trunnion can be selected so that the astronaut may look either along the star or landmark line of sight of the sextant. The wide field of view aids considerably in star and landmark recognition.

A very high frequency (VHF) link between the two vehicles exists which is normally used for inter-vehicle voice communication. However, a signal path through this VHF link from the CM to the LM and back to the CM makes possible a measurement of the range between the vehicles. Data from this automatic VHF range-link is used in the CMC during rendezvous navigation to complement the manually acquired optical measurements.

Optical navigation sightings are not required in the lunar module. However, to mid in a successful rendezyous with the command module, a rendezvoue radar (RR), shown in Figure 3, is provided and mounted near the LM inertial measuring unit so that direction data can be related between the two. With this instrument the range and range rate of the CN with respect to the LM as well as the direction, in terms of shaft and trunnion angles of the redar antenna, are made available to the lunar module computer (LMC) for state vector updating.

The command and lunar module computers are designed to handle a relatively large and diverse set of on-bound data processing and control functions. Some of the special requirements for this computer include (a) real-time solution of several problems simultaneously on a priority basis, (b) efficient two-way communication with the navigator, (c) capability of ground control radio links, and (d) multiple signal interfaces of both a discrete and continuously variable type. The memory section has a cycle time  $\sim 1.2 \,\mu \sec$  and consists of a fixed (read only portion of 36,864 words, together with an erasable portion of 2044 words. Each word in memory is 16 bits long (15 data bits and an odd parity bit). Data words are stored as signed 14-bit words.

Nost of the computer programs relevant to guidance and navigation are written in a pseudocode notation for economy of storage. This notation is encoded and stored as a list of data words. An "interpreter" program translates this list into a sequence of sub-routine linkages. Thus, the small basic instruction set is effectively expanded into a comprehensive mathematical language, which includes matrix and vector operations, using numbers of 28 bits and sign.

1 440

The display and keyboard (DSKY), illustrated in Figure 4, serves as the communication medium between the computer and the navigutor. The principal part of the display is a set of three registers, each containing five decimal digits, so that a word of 15 bits can be displayed in one register by five octal digits. Three registers are used because of the frequent, need to display the three components of a vector. Data are entered in the computer by the astronaut through the keyboard. When the computer requires a response from the astronaut, certain lights are caused to fissh on and off in order to attract his attention.

# 1.1 Navigation Data Processing

The recursive formulation of the optimum linear estimator, as originally deviaed by R.E.Kalman, is ideally suited to the space navigation problem - ospecially when a vehicle-borne computer is utilized to process data obtained from on-board instrumentation. With the Kalman estimator, measurement data may be incorporated sequentially, as they are obtained, without recourse to the batch processing techniques required by other methods. Furthermore, within the framework of a single computational algorithm, estimates of quantities in addition to position and velocity, such as radar biases, may be included by the simple expedient of increasing the dimension of the state vector. Finally, matrix inversion, with all of its numerical pitfalls, may also be avoided by regarding all measurement data as single-dimensional or scalar information.

Each computer in the two vehicles maintains an estimate of the position and velocity vectors of both its own and the other spacecraft. These two state vectors are normally six-dimensional but at times are augmented by certain parameters which must also be estimated as part of the navigation process.

In cislumar-midcourse navigation the command module computer only sinvolved and the CM state vector is of six dimensions. However, when the command module is navigating in luner orbit, it is necessary also to estimate the position vector of the particular landmark which is being tracked. This is conveniently accomplished by utilizing a nine-dimensional state, the first all elements of which are the components of the CM position and velocity vectors in moon-mentered, non-rotating rectangular coordinates, while the last three elements are the components of the lunar landmark position vector.

Two separate rendezvous navigation programs, one in such computer, are used simultaneously during the rendezwous phase of the mission with each solving the navigation problem independently of the other. In the CMC the six-dimensional state vector of either the CM or the LM can be updated from the measurement data obtained with the CM sensors. Normally, it is the LM state vector which is altered, but the mode is at the option of the astronauts. The selection of the updating mode is based primarily on which vehicle's state (a more accurately known initially and which vehicle is solve in controlling the rendezvour maneuvers.

Since the rendezvous radar of the LM is not structurally mounted with the IMU, significant unknown biases in the knowledge of the direction of the radar antenna are possible. In order to achieve the required accuracy during rendezvous, it is necessary to include the RR angle biases as components of the state vector to be estimated in the LMC. Although the augmented state vector then has eight components, it is treated as ninedimensional for computational convenience with the ninth element zero.

# 2. EXTRAPOLATION OF THE STATE VECTOR AND ERROR TRANSITION MATRIX

The estimates of position and velocity are maintained in the spacecraft computers in non-rotating restangular coordinates and are referenced to either the earth or the moon. An earth-contored equatorial coordinate-system is used when the vehicle is outside of the lunar sphere of influence. Inside of this sphere the center of coordinates coincides with the center of the upon. The extrapolation of position and velocity is made by a direct numerical integration of the equations of motion.

The basic equation may be written in vector form as

$$\frac{\mathrm{d}^2}{\mathrm{d}t^2} \mathbf{x}_{\mathrm{PV}} + \frac{\mu_{\mathrm{P}}}{r_{\mathrm{PV}}^2} \mathbf{x}_{\mathrm{PV}} = \mathbf{A}_{\mathrm{d}} , \qquad (2.1)$$

where  $\underline{r}_{PV}$  is the vector position of the vehicle with respect to the primary body P, which is either the sarth or moon, and  $\mu_p$  is the gravitational constant of P. The vector  $\underline{n}_d$  is the vector acceleration which prevents the motion of the vehicle from being precisely a conic with P at the focus.

If  $\mathbf{s}_{d}$  is small compared with the central force field, direct use of Equation (2.1) is inefficient. As an alternative, the integration may be accomplished by employing the technique of differential accelerations suggested by Encke (see pages 189-190 of Reference 2).

### 3.1 Racke's Method

At time  $t_0$ , the position and velocity vectors  $x_{pv}(t_0)$  and  $y_{pv}(t_0)$  define an o-outsting conic orbit. The vector difference  $\underline{\delta}(t)$  butween the actual and conic orbits satisfies the following differential equation:

$$\frac{d^2}{dt^2} = \frac{\mu_p}{r_p^2 v(0)} \left[ \left( 1 - \frac{r_p^2 v(0)}{r_p^2 v} \right) r_{pv} - \frac{5}{2} \right] + \mathbf{a}_d , \qquad (2.2)$$

subject to the initial conditions

 $\underline{\hat{g}}(t_0) = \underline{Q}, \qquad \frac{d}{dt} \underline{\hat{g}}(t_0) = \underline{V}(t_0) = \underline{Q},$ 

where  $E_{PV(C)}$  is the osculating conic position vector. The numerical difficulties which would arise from the evaluation of the coefficient of  $E_{PV}$  in Equation (2.2) may be avoided. Since

$$\mu_{V}(t) = I_{\mu_{V}(d)}(t) + \underline{\delta}(t) , \qquad (2.3)$$

it follows that

 $1 = \frac{r_{p \ V}^3}{r_{p \ V}^3} = - f(q_0) = 1 - (1 + q_0)^{3/4},$ 

where

$$\sigma = \frac{\left(\underline{\delta} - 2\chi_{\mu\nu}\right) \cdot \underline{\delta}}{r_{\mu\nu}^2} . \tag{2.4}$$

The function f(q) may be conveniently evaluated from

Ľ

$$f(q) = q \frac{3+3q+q^2}{1+(1+q)^{3/2}}.$$
 (2.5)

Encke's method may now be summarized as follows:

(i) Position in the osculating orbit is calculated from

$$\underline{\underline{x}}_{PV(G)}(t) = \left[1 - \frac{x^2}{x_{PV}(t_0)} G(\alpha_0 x^2)\right] \underline{\underline{x}}_{PV}(t_0) + \left[(t - t_0) - \frac{x^2}{\sqrt{\mu_p}} G(\alpha_0 x^2)\right] \underline{\underline{y}}_{PV}(t_0) , \qquad (2.6)$$

where

$$\alpha_{0} = \frac{2}{r_{PV}(t_{0})} - \frac{v_{PV}(t_{0})^{2}}{\mu_{P}}$$
(2.7)

and x is determined as the root of Kepler's equation in the form

$$\sqrt{(\mu_{p})(t-t_{0})} = \frac{r_{py}(t_{0}) \cdot y_{py}(t_{0})}{\sqrt{\mu_{p}}} x^{2} C(\alpha_{0} x^{2}) + [1 - r_{py}(t_{0})\alpha_{0}] x^{2} B(\alpha_{0} x^{2}) + r_{py}(t_{0}) x .$$
(2.8)

Die als miles Daar A DATA AND AND AND



The special transcendental functions S and C are defined by

$$\begin{array}{c} \mathbf{S}(\mathbf{x}) &= \frac{1}{31} - \frac{\mathbf{x}}{51} + \frac{\mathbf{x}^2}{71} - \dots \\ \mathbf{C}(\mathbf{x}) &= \frac{1}{21} - \frac{\mathbf{x}}{41} + \frac{\mathbf{x}^2}{61} - \dots \end{array}$$
 (2.9)

(ii) Deviations from the osculating orbit are obtained by a numerical integration of

$$\frac{d^2}{dt^2} \frac{\delta(t)}{\delta(t)} = -\frac{\mu_p}{r_p^2 v(c)(t)} \left[ f(q_c) r_{pv}(t) + \delta(t) \right] + \underline{h}_d(t) . \qquad (2.10)$$

The first term on the right-hand side of the last equation must remain small, i.e. of the same order as  $\underline{\mathbf{n}}_d(t)$ , if the method is to be efficient. As the deviation vector  $\underline{\delta}$  grows in magnitude, this term will eventually increase in size. Therefore, in order to maintain the efficiency, a new excutating orbit should be defined by the true position and velocity. The process of selecting a new conic orbit from which to calculate deviations is called rectification. When rectification occurs, the initial conditions of the differential education for  $\underline{\delta}$  are again zero and the right-hand side is simply the perturbation acceleration  $\underline{\mathbf{n}}_d$  at the time of rectification.

(iii) The position vector  $\mathbf{r}_{py}(t)$  is computed from Equation (2.3) using Equation (2.6). The velocity vector  $\mathbf{y}_{py}(t)$  is then computed as

$$(\mu_{\psi}(t) \simeq \underline{Y}_{\mu_{\psi}(0)}(t) + \underline{V}(t),$$
 (2.11)

where

where

$$\underline{\mathbf{y}}_{\mu\nu(d)}(t) = \frac{\sqrt{\mu_{\mu}}}{\mathbf{r}_{\mu\nu(d_{d}})\mathbf{x}_{\mu\nu(d)}(t)} \left[ \alpha_{0} \mathbf{x}^{3} \mathbf{B}(\alpha_{0} \mathbf{x}^{4}) - \mathbf{x} \right] \underline{\mathbf{r}}_{\mu\nu}(t_{0}) + \left[ 1 - \frac{\mathbf{x}^{2}}{\mathbf{r}_{\mu\nu(d)}(t)} \mathbf{C}(\alpha_{0} \mathbf{x}^{2}) \right] \underline{\mathbf{y}}_{\mu\nu}(t_{0}) , \quad (2.12)$$

# 2.2 Disturbing Acceleration

The form of the disturbing acceleration  $\mathbf{g}_{ij}$  to be used depends on the phase of the mission. In earth or lunar orbit only the gravitational perturbations arising from the non-apherical shapt of the primary body P need be considered. During translumar and transcarth flight, the gravitational attraction of the sun and the secondary body Q (either earth or moon) are relevant forces. A summary of the vericus cases appears below.

are the derivatives of the Legendre polynomials;

$$\cos \phi = \mathbf{i}_{\mathbf{E}\mathbf{V}} \cdot \mathbf{i}_{\mathbf{E}}$$

is the cosine of the angle  $\phi$  between the unit vector  $\underline{i}_{gv}$  in the direction of  $\underline{i}_{gv}$  and the unit vector  $\underline{i}_{g}$  in the direction of the north pole;  $\underline{r}_{g}$  is the equatorial radius of the earth; and  $J_{gg}$ ,  $J_{gg}$ ,  $J_{gg}$  are the coefficients of the second, third and fourth harmonics of the earth's potential function. The subscript R denotes the center of the earth as the origin of coordinates.

(11) Translumar and Transvarth Flight\*\*

$$\mathbf{s}_{d} = -\frac{\mu_{q}}{r_{qv}^{2}} \left[ f(a_{q}) \mathbf{x}_{pq} + \mathbf{x}_{pv} \right] - \frac{\mu_{s}}{r_{av}^{2}} \left[ f(a_{s}) \mathbf{x}_{ps} + \mathbf{x}_{pv} \right] , \qquad (2.14)^{\dagger}$$

where the subscripts Q and S denots the secondary body and the sun, respectively. Thus, for example,  $r_{pg}$  is the position vector of the sun with respect to the primary body. The arguments  $q_{()}$  are calculated from

\* See page 198 of Reference 2.

\*\* See page 12 of Reference 2.

المراجع والمحاج والمحاج

<sup>†</sup> During the periods of transverth and translutar flight when the vehicle is near the primary body, the sisturbing modelerstion which is used is the sum of Equation (3.14) and either (3.13) or (3.17), whichever is appropriate.



2 🖬 🗄

$$q_{()} = \frac{(t_{PV} - 2t_{P()}) \cdot t_{PV}}{r_{P()}^2}$$
(2.15)

and the function f from Equation (2.5).

In the violaity of the lunar aphere of influence a change in origin of coordinates is made. Thus

$$\frac{\mathbf{r}_{\mathbf{P}\mathbf{V}}(t) - \mathbf{r}_{\mathbf{P}\mathbf{Q}}(t)}{\mathbf{Y}_{\mathbf{P}\mathbf{V}}(t) - \underline{Y}_{\mathbf{P}\mathbf{Q}}(t)} = \mathbf{r}_{\mathbf{Q}\mathbf{V}}(t) - \underline{Y}_{\mathbf{P}\mathbf{V}}(t) \right\}$$
(2.10)

(iii) Lunar Orbit\*

$$\mathbf{\hat{z}}_{d} = \frac{\mu_{M}}{r_{MV}^{2}} \left\{ \sum_{k=\bar{v}}^{4} J_{kH} \left( \frac{r_{H}}{r_{MV}} \right)^{k} \left[ P_{k+1}^{\prime} (\cos \phi) \mathbf{1}_{MV} - F_{k}^{\prime} (\cos \phi) \mathbf{1}_{Z} \right] + 3 \eta_{22H} \left( \frac{r_{H}}{r_{MV}} \right)^{2} \left[ 4 \frac{x_{HV} y_{HV}}{x_{MV}^{2} + y_{HV}^{2}} \left( \mathbf{1}_{HV} \times \mathbf{1}_{Z} \right) + \frac{x_{MV}^{2} - y_{HV}^{2}}{x_{MV}^{2} + y_{HV}^{2}} \left( (5 \cos^{2} \phi - 3) \mathbf{1}_{HV} - 2 \cos \phi \mathbf{1}_{Z} \right) \right] \right\}, \quad (2.17)$$

where  $r_{\rm H}$  is the mean lunar radius;  $x_{\rm MV}$  and  $y_{\rm MV}$  are the lunar equatorial plane components of  $r_{\rm MV}$ ; and  $J_{22\,\rm M}$  is that coefficient of the moon's potential function associated with the asymmetry of the moon about its polar axis.

# 2.3 Error Transition Matrix

The position and velocity vectors, as maintained in the computer, are only estimates of the true values. As part of the Kalman estimation technique, it is also necessary to record statistical data for the processing of navigation measurements.

If  $\underline{\sigma}(t)$  and  $\underline{\gamma}(t)$  are the errors in the estimates of the position and velocity vectors, respectively, then the six-dimensional covariance matrix  $\underline{e}_{ee}(t)$  is defined by\*\*

$$\mathbf{E}_{\boldsymbol{\ell}\boldsymbol{\ell}}(\mathbf{t}) = \begin{pmatrix} \overline{\underline{\boldsymbol{\xi}}(\mathbf{t})\underline{\boldsymbol{\xi}}(\mathbf{t})^{\mathsf{T}}} & \overline{\underline{\boldsymbol{\xi}}(\mathbf{t})\underline{\boldsymbol{\eta}}(\mathbf{t})^{\mathsf{T}}} \\ \overline{\underline{\boldsymbol{\eta}}(\mathbf{t})\underline{\boldsymbol{\xi}}(\mathbf{t})^{\mathsf{T}}} & \overline{\underline{\boldsymbol{\eta}}(\mathbf{t})\underline{\boldsymbol{\eta}}(\mathbf{t})^{\mathsf{T}}} \end{pmatrix}.$$
(2.18)

As noted in Section 1, for cortain applications it is necessary to expand the state vector and the covariance matrix to more than six dimensions, in order to include the estimation of landwark locations in the CMC during orbital navigation and the rendezvous radar tracking biases in the LMC for rendezvous pavigation. For this purpose a nine-dimensional covariance matrix is defined as follows:

$$\mathbf{E}(t) = \begin{pmatrix} \mathbf{E}_{\mathbf{6},\mathbf{6}}(t) & \overline{\mathbf{c}}(t)\overline{\mathbf{c}}^{\mathsf{T}} \\ \overline{\mathbf{2}}(t)\overline{\mathbf{c}}^{\mathsf{T}} \\ \overline{\mathbf{c}}\overline{\mathbf{c}}(t)^{\mathsf{T}} & \overline{\mathbf{c}}\overline{\mathbf{2}}(t)^{\mathsf{T}} \\ \overline{\mathbf{c}}\overline{\mathbf{c}}(t)^{\mathsf{T}} & \overline{\mathbf{c}}\overline{\mathbf{c}}^{\mathsf{T}} \end{pmatrix}, \qquad (2.19)$$

where the components of the three-dimensional vector  $\zeta$  are the estimation errors of the three additional variables to be estimated.

To take full advantage of the operations provided by the interpreter in the spacecraft computers, the covariance matrix is restricted to either six or nine dimensions. In the LNC rendezvous navigation procedure, only two additional quantities are estimated. However, a dummy variable is added to the state vector to make it nine-dimensional.

The appropriate relation to be used in extrapolating the  $E_{s,j}(t)$  matrix is obtained as follows. The basic equation of motion of the vehicle with respect to the primary body may be written in the form:

$$\frac{\mathrm{d}^2}{\mathrm{d}t^2} \mathbf{L}_{PV} + \frac{\mu_P}{r_{PV}^2} \mathbf{L}_{PV} + \frac{\mu_Q}{r_{QV}^2} \mathbf{L}_{QV} + \frac{\mu_Q}{r_{PQ}^2} \mathbf{L}_{PQ} = \mathbf{E} \,.$$

where the vector g encompasses the gravitational acceleration of the sum as well as the other disturbances arising from the asymmetrical shapes of the earth and moon. This equation con be linearized about the best estimate of the vehicle's state and, to a first order of approximation, the estimate error vectors will satisfy the resulting linear differential equation. We have

\* See page 3 of Reference 2.

\*\* The transpose of a vector or a matrix is denoted by a superscript T.



$$\frac{\mathrm{d}}{\mathrm{d} t} \begin{pmatrix} \boldsymbol{\mathfrak{g}}(\mathbf{t}) \\ \boldsymbol{\eta}(\mathbf{t}) \end{pmatrix} = \begin{pmatrix} \boldsymbol{\mathfrak{g}} & \mathbf{I} \\ \boldsymbol{\mathfrak{g}}(\mathbf{t}) & \boldsymbol{\mathfrak{g}} \end{pmatrix} \begin{pmatrix} \boldsymbol{\mathfrak{g}}(\mathbf{t}) \\ \boldsymbol{\eta}(\mathbf{t}) \end{pmatrix},$$

where I and O are the three-dimensional identity and zero matrices, respectively.

The matrix  $\underline{O}(t)$  is the three-dimensional gradient of the gravitational field with respect to the components of the position vector  $r_{\mu\nu}$ . If we neglect the gradient of the vector  $r_{\mu\nu}$ , then it is easy to show that

$$g(t) = \frac{\mu_{p}}{r_{pv}^{\delta}(t)} \left[ 3 \underline{r}_{pv}(t) r_{pv}(t)^{T} - r_{pv}^{2}(t) \underline{I} \right] + \frac{\mu_{q}}{r_{qv}^{\delta}(t)} \left[ 3 \underline{r}_{qv}(t) r_{qv}(t)^{T} - r_{qv}^{2}(t) \underline{I} \right] , \qquad (2.30)$$

which is a sufficiently good approximation for translunar and transearth flight. For orbital mavigation about the primary body only the first term expression for g(t) need be included.

Since

$$\underline{\mathbf{E}}_{\boldsymbol{\epsilon}\,\boldsymbol{\epsilon}}(t) = \begin{pmatrix} \underline{\boldsymbol{\epsilon}}(t) & (\underline{\boldsymbol{\epsilon}}(t)^{\mathsf{T}} & \underline{\boldsymbol{\eta}}(t)^{\mathsf{T}}) \\ \boldsymbol{\boldsymbol{\eta}}(t) \end{pmatrix}$$

it is easy to show that the six-dimensional covariance matrix satisfies the following matrix differential equation:

$$\frac{\mathrm{d}}{\mathrm{dt}} \mathbf{E}_{44} = \begin{pmatrix} \underline{0} & \mathbf{I} \\ \underline{0} & \underline{0} \end{pmatrix} \mathbf{E}_{44} + \mathbf{E}_{44} \begin{pmatrix} \underline{0} & \underline{0} \\ \mathbf{I} & \underline{0} \end{pmatrix}.$$

Secause of accumulated numerical inaccuracies, it is possible that the covariance matrix may fail to remain positive definite after a large number of computations, as it theoretically must. An innovation to avoid this problem, which has also the advantage of significantly reducing certain computational requirements, is to replace the covariance matrix by a matrix  $\underline{\mathbb{M}}(t)$  , called the error transition matrix. The  $\underline{\mathbb{M}}(t)$  matrix has the property:

### E(t) = . W(t) W(t)

and thus, in a sense, is the square root of the covariance matrix. If needed, the covariance matrix may be dotermined as the product of the matrix  $\underline{W}(t)$  and its transpose, thereby guaranteeing it to be at least positive semi-definite.

The chief computational advantage of the M matrix lies in the simplicity of the differential equation which it satisfies. From the differential equation for  $E_{dd}$ , the fact that the components of the vector  $\chi$  do not change with time, and Equation (2.31), it is obvious that

$$\frac{d}{dt} \frac{\Psi}{\Psi} = \begin{pmatrix} 0 & 1 & 0 \\ \underline{0}(t) & \underline{0} & \underline{0} \\ \underline{0} & \underline{0} & \underline{0} \end{pmatrix} \Psi. \qquad (2,22)$$

Now let the nine-dimensional matrix  $\underline{W}(t)$  be partitioned as

 $\underline{\underline{\mathbf{H}}} = \begin{pmatrix} \underline{\underline{\mathbf{H}}}_0 & \underline{\underline{\mathbf{H}}}_1 & \cdots & \underline{\underline{\mathbf{H}}}_0 \\ \underline{\underline{\mathbf{H}}}_0 & \underline{\underline{\mathbf{H}}}_{10} & \cdots & \underline{\underline{\mathbf{H}}}_{17} \\ \underline{\underline{\mathbf{H}}}_{10} & \underline{\underline{\mathbf{H}}}_{19} & \cdots & \underline{\underline{\mathbf{H}}}_{26} \end{pmatrix},$ 

Then, we have

$$\frac{d}{dt} \underline{w}_{i+s}(t) = \underline{u}(t)\underline{w}_{i}(t)$$

$$\frac{d}{dt} \underline{w}_{i+s}(t) = \underline{u}(t)\underline{w}_{i}(t)$$

$$\frac{d}{dt} \underline{w}_{i+s}(t) = \underline{0}$$
(2.24)

Thus, the extrapolation of the M matrix may be accomplished by successively integrating the vector differential equations

$$\frac{1}{t^2} \underline{w}_i(t) = \underline{Q}(t) \underline{w}_i(t) \quad i = 0, i, \dots, 8. \qquad (2.25)$$

Finally, then, if D is the dimension of the matrix  $\underline{W}(t)$ , the differential equations for the  $\underline{w}_i(t)$  vectors are simply

$$\frac{d^{2}}{dt^{2}} \underline{w}_{i} = \frac{\mu_{p}}{r_{p}^{2}(t)} \left\{ 3[\underline{i}_{p}(t) \cdot \underline{w}_{i}(t)] \underline{i}_{p}(t) - \underline{w}_{i}(t) \right\} + \frac{\mu_{q}}{r_{q}^{2}(t)} \left\{ 3[\underline{i}_{q}(t) \cdot \underline{w}_{i}(t)] \underline{i}_{q}(t) - \underline{w}_{i}(t) \right\}$$
(2.26)  
$$i = 0, 1, ..., D - 1$$

with the second term omitted for orbital navigation about the primary body P .

Set page 207 of Reference 2.





12.211

(2.23)





**的**和新聞

# 2.4 Numerical Integration

1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 -

The extrapolation of the state vector and the error transition matrix requires the solution of D + 1 secondorder vector differential equations, specifically Equations (2.10) and (2.28). Thuse are all special cases of the form

$$\frac{d^2}{dt^2} \underline{y} = \underline{f}(\underline{y}, t) , \qquad (3.27)$$

in which the right-hand side is a function of the independent variable and time only. Nyström's method, described in Reference 3, is particularly well-suited to this form and gives an integration method of fourth-order accuracy, while requiring only three computations of the derivatives per time step. (The usual fourth-order Runge-Kutta integration methods require four derivative computations per time step.) The mecond-order system is written as

 $\frac{d}{dt} \underbrace{\chi}_{z} = \underbrace{\pi}_{z}$   $\frac{d}{dt} \underbrace{z}_{z} = \underbrace{f}(\underbrace{y}, t)$ (2.28)

and the formulae are summarized as follows:

$$\begin{aligned} \mathbf{y}_{n+1} &= \mathbf{y}_n + \mathbf{\phi}(\mathbf{y}_n) \Delta t \\ \mathbf{g}_{n+1} &= \mathbf{g}_n + \mathbf{\psi}(\mathbf{y}_n) \Delta t \\ \mathbf{\phi}(\mathbf{y}_n) &= \mathbf{g}_n + \frac{1}{2} (\mathbf{g}_1 + 2\mathbf{g}_2) \Delta t \\ \mathbf{\psi}(\mathbf{y}_n) &= \frac{1}{2} (\mathbf{g}_1 + 4\mathbf{g}_2 + \mathbf{g}_3) \\ \mathbf{g}_1 &= \mathbf{f}(\mathbf{y}_n, \mathbf{t}_n) \\ \mathbf{g}_2 &= \mathbf{f}(\mathbf{y}_n, \mathbf{t}_n) \\ \mathbf{g}_3 &= \mathbf{f}(\mathbf{y}_n + \frac{1}{2} \mathbf{g}_n \Delta t + \frac{1}{2} \mathbf{g}_2 (\Delta t)^2, \mathbf{t}_n + \frac{1}{2} \Delta t) \\ \mathbf{g}_3 &= \mathbf{f}(\mathbf{y}_n + \mathbf{g}_n \Delta t + \frac{1}{2} \mathbf{g}_2 (\Delta t)^2, \mathbf{t}_n + \Delta t) \end{aligned}$$
(2.29)

For efficient use of computer storage as well as computing time, the computations should be performed in the following order:

- (i) Equation (2.10) is solved using the Nyström formulas (2.29). It is necessary to preserve the values of the vectors  $\underline{r}_{py}$  at times  $t_n$  and  $t_n + \Delta t/2$  for use in the solution of Equations (2.26).
- (ii) Equations (2.26) are solved one at a time using formulae (2.29), together with the values of  $r_{PV}$  which resulted from the first step.

It has been found experimentally that the maximum value that the integration time step  $\Delta t$  can have is either  $0.3(r_{p_V}^2/\mu_p)^{3/2}$  or 4000 seconds, whichever is the smaller.

### 3. INCORPORATION OF MEASUREMENT DATA

An important feature of the Apollo navigation method is that measurement data from a wide variety of sources may be incorporated within the same framework of computation. Associated with each measurement is a D-dimensional vector  $\underline{b}$  representing, to a first order of approximation, the variation in the measured quantity  $\underline{Q}$  which would result from variations in the components of the state vector. Thus, each measurement establishes a component of the spacecraft state vector along the direction of the  $\underline{b}$  vector in state space.

By sigebraically combining the  $\underline{M}$  matrix, the <u>b</u> vector and a mean-squared a priori estimation error  $\alpha^2$  in the measurement, there are produced a weighting vector  $\underline{\omega}$  and the step change to be made in the error transition matrix to reflect the changes in the uncertainties in the estimated quantities as a result of the measurement. The weighting vector  $\underline{\omega}$  has D components and is determined so that the observation data is utilized in a statistically optimum manner.

According to the Kalman estimation theory for a one-dimensional scalar measurement, the weighting vector is determined from

 $\underline{\omega} = \frac{1}{a} \underline{\mathbf{E}} \mathbf{b} , \qquad (3.1)$ 

where E is the value of the covariance matrix extrapolated to the time of the measurement and

 $\mathbf{a} = \mathbf{b}^{\mathrm{T}} \mathbf{g} \mathbf{b} + \mathbf{a}^{\mathrm{T}} . \tag{3.2}$ 

For convenience, define the vector g as

$$\mathbf{F} = \mathbf{X}_{\mathbf{I}}\mathbf{P}$$

so that Equations (3.1) and (3.2) may be written

$$\underline{\omega} = \frac{1}{2} \mathbb{E}_{\mathbf{Z}}$$
(3.4)

$$\mathbf{a} = \mathbf{z}^{\mathsf{T}} \mathbf{z} + \overline{\alpha^{\mathsf{T}}} \,. \tag{3.5}$$

Then, if  $\delta Q$  represents the difference between the quantity actually measured and its expected value based on the extrapolated value of the state vector, the change in the state vector is simply  $\omega \delta Q$ ,

As a result of the measurement, the statistics embodied in the covariance matrix  $\underline{E}$  or its square root  $\underline{B}$  must be altered. Again, the Kalman theory dictates that the new value for  $\underline{E}$ , denoted by  $\underline{E}^{\bullet}$ , is obtained from

$$\mathbf{E}^{\mathbf{A}} = \mathbf{E} \left( \mathbf{I} - \frac{1}{\mathbf{a}} \mathbf{b} \mathbf{b}^{\mathrm{T}} \mathbf{E} \right)$$

or, in terms of the W matrix and the z vector,

$$\underline{\underline{g}}^{\text{th}} = : \underline{\underline{\pi}} \left( \underline{\underline{1}} - \frac{1}{a} \underline{\underline{\pi}} \underline{\underline{n}}^{\text{T}} \right) \underline{\underline{\pi}}^{\text{T}} , \qquad (3, 6)$$

where I is the D-dimensional identity matrix.

What is desired, of course, is a formula for updating the  $\underline{N}$  matrix rather than the  $\underline{Z}$  matrix. The objective will be achieved if a square root can be found for the parenthesized factor in Equation (3.6). Indeed, the desired result is obtained by determining the value of the parameter  $\gamma$  such that

$$\left(\underline{\mathbf{I}} - \frac{1}{\mathbf{a}} \underline{\mathbf{a}} \underline{\mathbf{a}}^{\mathsf{T}}\right) = \left(\mathbf{I} - \frac{\gamma}{\mathbf{a}} \underline{\mathbf{a}} \underline{\mathbf{a}}^{\mathsf{T}}\right) \left(\underline{\mathbf{I}} - \frac{\gamma}{\mathbf{a}} \underline{\mathbf{a}} \underline{\mathbf{a}}^{\mathsf{T}}\right)^{\mathsf{T}}.$$

By straightforward computation it is seen that  $\gamma$  must be

$$\gamma = \frac{1}{1 + \sqrt{\alpha^2/a}}, \qquad (3.7)$$

so that the new value W\* is computed as

$$\overline{\mathbf{x}}^{\dagger} = \overline{\mathbf{x}} - \gamma \omega \underline{\mathbf{z}}^{\mathsf{T}} . \tag{3.8}$$

In order to take full advantage of the three-dimensional vector and matrix operations provided by the interpreter in the computer, the mine-dimensional <u>matrix</u> is stored sequentially as follows:

$$\mathbf{w}_{0}^{\mathsf{r}}$$
,  $\mathbf{w}_{1}^{\mathsf{r}}$ , ...,  $\mathbf{w}_{24}^{\mathsf{r}}$ ,

Then, by defining the three-dimensional sub-matrices



so that

(3.9)

(3.10)

(3, 3)

and partitioning the possibly nine-dimensional vectors b, w, and g as



the computations developed in this section are conveniently performed as



11)

$$\underline{\underline{z}}_{j} = \underbrace{\underline{\underline{z}}_{k=0}^{\underline{D}_{-1}}}_{\underline{\underline{x}}_{k=0}} \underbrace{\underline{\underline{w}}_{j+3k} \underline{\underline{w}}_{k}}_{\underline{\underline{x}}_{k}}, \quad \underline{\underline{a}} = \underbrace{\underline{\underline{x}}_{k=0}^{\underline{D}_{-1}}}_{\underline{\underline{x}}_{k} - \underline{\underline{x}}_{k}} \underbrace{\underline{z}}_{k} + \overline{\underline{\alpha}}_{k}^{\underline{\underline{x}}} \right\}$$

$$(3)$$

where the subscripts i and j range, respectively, from 0 to D-1 and from 0 to (D/3) - 1.

It is worth emphasizing that the particular formulation of the Kalman estimator, developed in this and the previous section, has achieved what is felt to be a minimum in practical computational efficiency as well as a most sparing use of erasable memory locations in the small vehicle-borne Apollo computers. The latter is possible primarily because of the introduction of the error transition matrix instead of the more conventional covariance matrix.

It should be remarked, however, that straightforward application of the  $\underline{\mathbb{Y}}$  matrix techniques would not have been possible had the dynamics of the state vector model included what is commonly called "process noise". Since we have elected to neglect process noise in favor of computational compactness, the gradual and inevitable decay of the elements of the  $\underline{\mathbb{Y}}$  matrix must be countered in practice by a periodic re-initialization of those elements. It should be clear that, without either including process noise or re-initializing the  $\underline{\mathbb{Y}}$  matrix, eventually all measurement data would be ignored simply because they would be given a zero wrighting factor.

# 4. COMMAND MODULE CIBLUNAR - MIDCOURSE NAVIGATION

During the translumar and transmarth phases of the Apollo mission, navigation data can be obtained with the sextant by measuring the angle between the lines of sight to a star and an earth or moon horizon or landmark. When the navigator depresses the mark button, indicating to the computer that the two targot optical images are properly superimposed in the SNT field of view, the time of the measurement and the measured angle, i.e. the SNT trunnion angle A, are automatically recorded in the CMC. The navigator must then inform the computer, through the DBKY, of the identity of the star and the particular feature of the earth or moon involved in the sighting. These data are used by the computer to determine: (i) the CM state vector estimate and its associated error transition matrix extrapolated to the measurement time, as described in Section 2; (ii) the estimated CM position vector  $\underline{r}_{\rm V}$  relative to that body used in the measurement (since this may be either the primary body P or the secondary body Q, the single subscript will serve without automatics); (iii) the unit vector  $\underline{r}_{\rm L}$  in the direction of the perticular star used (there are data for 37 stars in the CMC fixed memory which the navigator can identify by code number); and (iv) the position vector  $\underline{r}_{\rm L}$  of the landmark, assuming a star-landmark measurement.

### 4.1 Star-Landmark Mensurements

Because of the extremely high accuracy required for middourse navigation measurements, observation data must be corrected for aberration prior to processing. Aberration is the term used to describe the change in the apparent direction of an object due to the velocity of the observer normal to the line of sight to the object. It is only when this perpendicular velocity has a magnitude of tens of thousands of feet per second that the aberration correction is necessary. It should be remarked that the correction is not required for rendezvous or orbit mavigation.

The apparent direction of the star is computed from

$$i_{\alpha}^{+} = \operatorname{Unit}\left(i_{\alpha} + \frac{\chi_{PV} + \chi_{QE}}{c}\right), \qquad (4.1)$$

where the notation Unit  $(\underline{q})$  is understood to mean a unit vector in the direction of the vector  $\underline{q}$ . The vector  $\underline{x}_{eg}$  is the velocity of the earth relative to the sun, and c is the speed of light. The velocity of the sun relative to the star is already taken into account for the basic star directional data  $\underline{j}_{g}$ , and the relative velocity between the earth and mean is negligible for the case in which the mean is the primary body. The expected direction of the landmark is obtained from

$$\frac{iv_L}{2V_L} = \text{Unit}\left(\frac{1}{2V_L} + \frac{V_{PV}}{\alpha}\right). \tag{4.2}$$

where

 $\underline{i}_{VL} = Unit(\underline{r}_L - \underline{r}_V)$ 

is a unit vector defining the estimated direction of the landmark relative to the vehicle.

Using the corrected unit vectors corresponding to the directions of the two SXT lines of sight, the CMC computes the measured deviation  $\delta Q$  as

$$\delta \mathbf{Q} = \mathbf{A} - \cos^{-1} \left( \mathbf{j}_{\mathrm{H}}^{*} \cdot \mathbf{j}_{\mathrm{VL}}^{*} \right) \,. \tag{4.3}$$



Although for cialumar-mideourse navigation the CM state vector is six-dimensional, only the first three components of the measurement geometry vector  $\underline{b}$ , defined in Section 3, will be non-zero, since the measured quantity is independent of velocity except through the small aberration correction. To determine this non-zero partition  $\underline{b}_a$  of the  $\underline{b}$  vector, we compute the first-order differential of the expression

$$\mathbf{r}_{VL}^{*} \operatorname{coc} A = \mathbf{i}_{u}^{*} \cdot \mathbf{r}_{VL}^{*}$$

with the result that

 $\cos A \, \delta r_{VL}^{\pm} - r_{VL}^{\pm} \sin A \, \delta A = \underline{i} + \delta \underline{r}_{VL}^{\pm} ,$ 

Then, since

$$\delta r_{VL}^{*} = \frac{\Gamma_{VL}^{*} \cdot \delta \Gamma_{VL}^{*}}{r_{VL}^{*}} \quad \text{and} \quad \delta r_{VL}^{*} = -\delta r_{PV} ,$$

we have

$$\nabla_{\mathbf{L}} \sin \mathbf{A} \delta \mathbf{A} \simeq \left( \mathbf{i}_{\mathbf{H}}^{*} - \cos \mathbf{A} \mathbf{i}_{\mathbf{V}\mathbf{L}}^{*} \right) \cdot \delta_{\mathbf{T} \neq \mathbf{V}}$$

Hence

٥ř

$$\underline{b}_{0} = \frac{1}{r_{VL}^{+} \sin A} \left( \underline{i}_{B}^{+} - \cos A \underline{i}_{VL}^{+} \right)$$

$$\mathbf{k}_0 = \frac{1}{\mathbf{r}_{VL}^0} \text{ Unit } \left[ \mathbf{i}_{\mathbf{s}}^* - (\mathbf{i}_{\mathbf{s}}^* \cdot \mathbf{i}_{VL}^*) \mathbf{i}_{VL}^* \right] .$$

Finally, the measurement error variance is computed from

$$\overline{\alpha}^{2} = \sigma_{\lambda}^{2} + \frac{\sigma_{L}^{2}}{z_{VL}^{2}}, \qquad (4.b)$$

where  $\sigma_{k}^{2}$  and  $\sigma_{k}^{2}$  are the assumed error variances in the SXT trunnion angle and the landmark, respectively.

### 4.3 Star-Horizon Heasurements

The processing of star-horizon measurement data is the same as that required for star-landmark data, provided the landmark location vector  $\underline{r}_i$  is replaced by a vector from the spacecraft to the horizon. The determination of this horizon vector is made according to the following geometrical arguments.

Consider first an earth-horizon measurement. The star direction  $j_{es}$  and the estimated CM position vector  $\Sigma_v$  determine a plane. Assuming that the horizon is at a constant altitude above the earth's surface, the intersection of this measurement plane with the horizon is approximately an ellipse. The crientation of this horizon ellipse is defined in terms of three mutually orthogonal unit vectors:

 $\frac{\mathbf{i}_{2}}{\mathbf{i}_{0}} = \operatorname{Unit}\left(\mathbf{i}_{0} \times \mathbf{r}_{V}\right)$   $\mathbf{i}_{0} = \operatorname{Unit}\left(\mathbf{i}_{0} \times \mathbf{i}_{U}\right)$   $\mathbf{i}_{1} = \mathbf{i}_{2} \times \mathbf{i}_{0},$  (4.6)

where  $\underline{i}_{g}$  is a unit vector in the direction of the earth's polar axis. Referring to Figure 7, it is clear that  $\underline{i}_{0}$  and  $\underline{i}_{1}$  coincide with the semi-major and semi-minor axes of the horizon ellipse, respectively. The plane containing the horizon ellipse is inclined with respect to the earth's equatorial plane by an angle I, where

$$\min \mathbf{I} = \underline{i}_1 \cdot \underline{i}_2 \quad (4.7)$$

The shape of the horizon ellipse is determined by the lengths of its major and minor axes. Assuming the contour of the earth to be well-approximated by the so-called Fincher ellipsoid, the semi-major axis of the horizon ellipse  $a_{\rm H}$  is simply the sum of the semi-major axis of this ellipsoid and the constant horizon altitude. Likewise, the semi-minor axis  $b_{\rm H}$  is found by adding the horizon altitude to that value of the radius of the Fischer ellipsoid which corresponds to a latitude equal to the inclination angle I.

The problem of determining the vector  $\mathbf{r}_i$  is readily solved in the horizon coordinate system for which the x and y axes coincide with the directions  $\mathbf{j}_0$  and  $\mathbf{j}_1$ , respectively, as illustrated in Figure 8. The matrix

 $-\underline{\mathbf{H}} = \begin{pmatrix} \underline{\mathbf{i}}_{0}^{\mathrm{T}} \\ \underline{\mathbf{i}}_{1}^{\mathrm{T}} \\ \underline{\mathbf{j}}_{2}^{\mathrm{T}} \end{pmatrix} \qquad (4.8)$ 

will serve to transform vectors from the original coordinate system to the horizon system. Let  $r_{\rm H}$  and  $i_{\rm eH}$  represent the components of  $r_{\rm V}$  and  $i_{\rm e}$  in horizon coordinates, so that

(4.4)



tation Officiation O seite



Further, define  $\underline{t}_0$  and  $\underline{t}_1$  as vectors from the point  $(x_H, y_H)$  to the two points of tangency with the horizon ellipse.

The vectors  $f_n$  and  $f_1$  are obtained by solving simultaneously the equation of the horizon ellipse:

 $\frac{a_{1}^{2}}{x_{1}^{2}}+\frac{p_{1}^{2}}{y_{1}^{2}}=1$ 

 $\frac{\mathbf{x}\mathbf{x}_{H}}{\mathbf{a}_{1}^{2}} + \frac{\mathbf{y}\mathbf{y}_{H}}{\mathbf{b}_{1}^{2}} = 1 \ .$ 

and the equation of the line tangent to the ellipse and passing through the point  $(x_{\mu}, y_{\mu})$ :

We have

$$I_{I} = \frac{1}{d} \begin{pmatrix} x_{H} \pm \frac{u_{H}}{b_{H}} y_{H} v'(d-1) \\ y_{H} \neq \frac{b_{H}}{u_{H}} x_{H} v'(d-1) \\ 0 \end{pmatrix}, \quad I = 0, 1, \quad (4.10)$$

where

$$d = \frac{x_1^2}{x_1^2} + \frac{y_1^2}{y_1^2}.$$

# "The upper and lower signs in Equation (4.10) correspond, respectively, with i = 0 and i = 1.

The two points of tangency,  $\underline{\mathbf{r}}_0$  and  $\underline{\mathbf{r}}_1$ , correspond to the two horizon points,  $\underline{\mathbf{r}}_{LH}^+$  (near horizon) and  $\underline{\mathbf{r}}_{LH}^0$  (far horizon), but not necessarily respectively. By definition, the near horizon is associated with that vector  $\underline{\mathbf{r}}_1$  which makes the smaller angle with the star vector. The morizon vector is then

using for  $\underline{t}$  either  $\underline{t}_n$  or  $\underline{t}_1$ , whichever is appropriate.

ŝ,

In the CMD, moon-horizon measurements are processed under the assumption that the moon has a spherical shape. The determination of the horizon vector  $\mathbf{r}_L$  is made as for the earth-horizon case but with the moan radius of the moon used for both  $\mathbf{s}_N$  and  $\mathbf{b}_N$ .

### 5. COMMAND MODULE NAVIGATION DURING RENDEZVOUS

To accomplish rendezvous with the CM and LM in the Apollo mission, two types of navigation data are obtained with the CM-based sensors. As described in Section 1, these data are accuired both manually with the optics and automatically through the VHF range-link. They serve in complementary roles for rendezvous navigation. The optical data yields information in the directions normal to the CM-to-LM line of sight, whereas the VHF rangelink gives information along the line of sight. Taken together, they provide an excellent rendezvous navigation

It is recognized that the optimum method of applying the Kalman estimation technique to rendezvous navigation would be to solve the twelve-dimensional problem of simultaneous estimation of both the CN and LN states. However, the limitations of a relatively small eramable numbery products the storage of the required twelvedimensional error transition matrix. In the CMC rendezvous navigation procedure (and in the LMC as well), the estimated state vector of one vohicle only is altered by the navigation data. Consequently, the error transition matrix is of manageable siz-dimensional size and is extrapolated with whichever state vector is being estimated.

Because of this simplification, together with some other related factors, the error transition matrix does not maintain an accurate representation of the covariances of the relative state errors between the two vehicles. This situation necessitates a re-initialization of the matrix at various times during the rendezvous, essentially to restart the solution in both the CMC and LMC.







AND AND A

222222

Π1

### 5.1 Optical Measurements

Approximately once per minute during the navigation portions of the rendezvous phase, the Apollo navigator sights the LM using the reticule pattern in the SXT star line of sight. When the mark button is depressed, the time of the measurement, the SXT shaft and trunnion angles, and the three IMU gimbal angles, which describe the orientation of the navigation base with respect to the inertially stabilized platform, are all automatically recorded in the CMC.

From these five angles and the known orientation of the inertial platform, the measured direction,  $\underline{i}_{m}$ , of the LM as observed from the CM is obtained. It is convenient to consider the unit vector  $\underline{i}_{m}$  as having been found by the simultaneous measurement of the angles between the lines of sight to the LM and two stars. The data are processed by selecting two convenient unit vectors (flotitious star directions), converting the vector  $\underline{i}_{m}$  to an equivalent set of two artificial star-LM measurements, and using the measurement incorporation procedure of Section 3 twice — once for each artificial measurement. These two unit vectors are chosen such that they and the estimated line of sight vector form an orthogonal triad.

The processing of each of the two artificial measurements is similar to the cislumar-mideourse navigation procedure described in Section 4. Let  $\underline{r}_{po}$  and  $\underline{r}_{pL}$  be the CA and LM position vectors, respectively, extrapolated to the time of the measurement.

For convenience, the first fictitious star direction is chosen to be

$$\underline{\mathbf{i}}_{ai} = \operatorname{Unit}\left[(\underline{\mathbf{i}}_{oL} \times \underline{\mathbf{i}}_{a}) \times \underline{\mathbf{i}}_{oL}\right], \qquad (5.1)$$

where

$$L_{OL} = Unit(\underline{r}_{PL} - \underline{r}_{PO})$$

is the estimated CN-to-LM line of sight. The measured deviation is then given by

$$\delta q = \cos^{-1} \left( \frac{1}{2} + \frac{1}{2} \right) - \frac{\pi}{2} , \qquad (5.2)$$

since  $j_{u_1}$  and  $j_{0L}$  are perpendicular. This orthogonality also permits a simplification in the calculation of the geometry vector  $\underline{b}_0$ . Indeed, from Equation (4.4), we have

$$\underline{b}_0 = \pm \frac{1}{r_{ct.}} \underline{1}_{a1} . \tag{5.3}$$

The plum or minus high is selected, respectively, according as the CM or LM state vector is to be updated.

Following the alteration of the appropriate state vector, the second artificial measurement is incorporated by recomputing the vector  $\underline{i}_{OL}$  and selecting the fictitious star direction as

$$i_{n_1} = \text{Unit}(i_{n_1} \times i_{\text{UL}})$$
 (5.4)

The measurement error variance  $\overline{\alpha^2}$  for each incorporation is a constant which is the sum of the assumed error variances of the SXT and knowledge of the INU orientation.

### 5.2 WF Range Link Kensurements

Asynchronous with the manually-coquired optical data, and at approximately the same frequency. VHF range measurements are automatically taken. Again, the time of the measurement and the measured CM-to-LM range R are recorded in the CMC.

The measured deviation for this range measurement is simply

والمراجع والمحالية والمعارية والمحالية وال

$$\delta q = R - r_{ot}$$

To determine the geometry vector  $\underline{b}_0$  , we compute the first-order differential of

$$\mathbf{R}^{\mathbf{f}} = (\underline{\mathbf{r}}_{\mathbf{PL}} - \underline{\mathbf{r}}_{\mathbf{PC}}) \cdot (\underline{\mathbf{r}}_{\mathbf{PL}} - \underline{\mathbf{r}}_{\mathbf{PC}})$$

under the assumption that it is the CM siste vector which is to be estimated. There results

$$R\delta R = - (r_{PL} - r_{PO}) \cdot \delta r_{PO}$$

 $\delta R = \underline{i}_{GL} \cdot \delta \underline{r}_{PL}$ 

 $\delta \mathbf{R} = -\mathbf{1}_{\mathbf{RL}} \cdot \delta \mathbf{r}_{\mathbf{PC}} \, .$ 













(8. 6)

Albert A. Seehington and A. erken



Hence

(5,6)

(6.1)

where the upper or lower sign depends, respectively, on whether the CM or LM state vector is to be changed.

### 6. LUNAR MODULE NAVIGATION DURING RENDEZVOUS

While the CMC is actively acquiring and processing data during the rendezvous phase of Apollo, the same navigation problem is being simultaneously and independently solved in the LMC, using tracking information gathered from the rendezvous radar mounted on the LM vehicle. After RR tracking acquisition of the CM is established, the LMC records, at approximately one minute intervals, the measured range R and range rate R of the CM with respect to the LM, together with the shaft angle  $\beta$  and the trunnion angle  $\theta$  of the gimballed radar dish. In addition to these four measured quantities, the time of the measurement and the three IMU gimbal angles are also noted.

An described in Section 1, eight variables are estimated as part of the navigation procedure. The usual three components of the position and velocity vectors,  $\underline{r}_{p_V}$  and  $\underline{v}_{p_V}$ , of the selected spacecraft, LM or CM, with respect to the primary body P, constitute the first six components of the state vector. The estimates of the biases,  $\delta\beta$  and  $\delta\theta$ , in the RR shaft and trunnion angles are the seventh and eighth elements. A diamay variable is used for the ninth component to facilitate three-dimensional vector operations.

The measured quantities produce four sequential alterations of the nine-dimensional state vector through four separate applications of the measurement incorporation method described in Section 3. The incorporation is, of course, performed recursively, i.e. at each stage the new components of the state vector, resulting from the previous update, are used in computing the next update in the sequence. The measurement error variances  $\overline{\alpha^2}$ are based on a priori knowledge of the redar performance.

### 8.1 Range and Range-Sate Neasurements

The measured deviation in range and the associated measurement geometry vector  $\mathbf{h}_0$  are the same as for the VHF measurement of range in the CM and need not be further discussed. For the range-rate data, the measured deviation is easily seen to be

$$\delta \mathbf{q} = -\mathbf{\hat{\mathbf{x}}} - (\mathbf{\underline{y}}_{PO} - \mathbf{\underline{y}}_{PL}) + \mathbf{\underline{j}}_{LO} , \qquad (2.5)$$

The geometry vector <u>b</u> for the range-rate measurement provides the first case encountered thus far for which both the <u>b</u><sub>0</sub> and <u>b</u><sub>1</sub> partitions are different from zero. To determine the <u>b</u> vector, we compute the first-order differential of the relation

$$\mathbf{x} = \frac{(\mathbf{r}_{PC} - \mathbf{r}_{PL}) \cdot (\mathbf{y}_{PC} - \mathbf{y}_{PL})}{[(\mathbf{r}_{PC} - \mathbf{r}_{PL}) \cdot (\mathbf{r}_{PC} - \mathbf{r}_{PL})]^{1/2}}$$

with the assumption that it is the CM state vector which is to be estimated. We usve

or. alternative

Stars of

$$\dot{\mathbf{k}} = \frac{1}{r_{c}^{2}} \underline{r}_{Lc} \times \langle \underline{\mathbf{y}}_{Lc} \times \underline{\mathbf{r}}_{Lc} \rangle \cdot \delta \underline{\mathbf{y}}_{Fc} + \frac{1}{r_{c}} \underline{\mathbf{x}}_{Lc} \cdot \delta \underline{\mathbf{y}}_{Fc} .$$

 $\delta \mathbf{\hat{x}} = \frac{1}{r_{1,0}} \left( \underline{\mathbf{y}}_{LC} \cdot \delta \underline{\mathbf{x}}_{PC} + \underline{\mathbf{r}}_{LC} \cdot \delta \underline{\mathbf{y}}_{PC} \right) - \frac{1}{r_{1,0}^2} \left( \underline{\mathbf{r}}_{LC} \cdot \underline{\mathbf{y}}_{LC} \right) \underline{\mathbf{r}}_{LC} \cdot \delta \underline{\mathbf{r}}_{PC}$ 

To obtain the proper relationship if the LM state vector is to be estimated, we need only note that  $\delta_{\mathbf{r}_{PL}} = \delta_{\mathbf{r}_{PC}}$ . Thus, the non-mero partitions of the <u>b</u> vector are

with the choice in sign dependent on the particular state vector to be updated (plus for CM and minus for LM).

### 6.3 Hadar Antenna Angle Measurements

The rendenvous radar antenna dish is at the origin of the RR cartesian coordinate system, as seen in Figure 9. Shaft notion takes place about the positive  $\cdot y$  axis and the shaft angle  $\beta$  is measured from the positive xaxis. Trunnion motion occurs in a plane normal to the xx plane and containing the shaft axis y. The configuration is such that the trunnion axis would coincide with the x axis for zero shaft angle with the trunnion angle  $\theta$ , under those circumstances, measured from the x axis.

Let  $\underline{i}_{\underline{n}}$ ,  $\underline{i}_{\underline{n}}$ ,  $\underline{i}_{\underline{n}}$  be unit vectors along the RR coordinate axes. By means of the recorded IMU gimbal angles, together with the knowledge of the inertial orientation of the IMU, the components of  $\underline{i}_{\underline{n}}$ ,  $\underline{i}_{\underline{n}}$ ,  $\underline{i}_{\underline{n}}$ , are readily









obtained in basic reference coordinates. It is usy to verify with reference to the figure that the measured deviation for the shaft angle is

$$\delta \mathbf{Q} = \beta - \left[ t_{\mathbf{n}n^{-1}} \left( \frac{\mathbf{A}_{\mathbf{n}} \cdot \mathbf{A}_{\mathbf{L}0}}{\mathbf{A}_{\mathbf{n}} \cdot \mathbf{A}_{\mathbf{L}0}} \right) + \delta \beta \right]. \tag{6.3}$$

while that for the transion angle is

$$\Theta = \theta - \left[ \sin^{-1} \left( -\frac{1}{2v} + \frac{1}{41,0} \right) + \delta \theta \right], \qquad (6.4)$$

The position vector of the CM with respect to the LM is expressible in terms of components along the RR coordinate axes as

 $\underline{r}_{PO} - \underline{r}_{PL} = \underline{r}_{LO} \begin{pmatrix} \sin\beta\cos\theta \\ -\sin\theta \\ \cos\theta \end{pmatrix}$ 

1

To obtain the measurement geometry vector corresponding to the shaft angle, we again compute the differential

$$\delta_{\mathbf{L}_{\mathbf{P}\mathbf{C}}} = \mathbf{r}_{\mathbf{L}\mathbf{C}} \begin{pmatrix} \cos\beta\cos\theta \\ 0 \\ -\sin\beta\cos\theta \end{pmatrix} \delta\beta$$
$$= \mathbf{r}_{\mathbf{L}\mathbf{C}}\cos\theta \text{ Unit } (\underline{\mathbf{i}}_{\mathbf{y}} \times \underline{\mathbf{i}}_{\mathbf{L}\mathbf{C}}) \delta\beta$$

Hence

$$\delta \beta = \frac{1}{r_{LC} \cos \theta}$$
 Unit  $(\underline{i}_{y} \times \underline{i}_{LC}) \cdot \delta \underline{i}_{PC}$ ,

Similarly, for the trunnion angle, we have

$$\delta_{\mathbf{I}\mathbf{PC}} = \mathbf{r}_{\mathbf{LC}} \begin{pmatrix} -\sin\beta\sin\theta\\ -\cos\theta\\ -\cos\theta \end{pmatrix} \delta\theta$$
$$= \frac{\mathbf{r}_{\mathbf{LC}}}{\cos\theta} (\mathbf{j}_{\mathbf{Y}} \times \mathbf{j}_{\mathbf{LJ}}) \times \mathbf{j}_{\mathbf{LC}} \delta\theta ,$$

so that

$$\delta \theta = \frac{1}{r_{\rm LC} \cos \theta} \left( \frac{1}{2\gamma} \times \frac{1}{2 LC} \right) \times \frac{1}{2 LC} \cdot \delta r_{\rm PC} \, .$$

The quantity  $r_{LC} \cos \theta$ , which appears in the expressions for  $\delta \beta$  and  $\delta \theta$ , is simply the length of the projection of the  $\underline{r}_{LC}$  vector in the xx plane. By denoting this length by  $r_{\chi_R}$ , we may calculate its value from

$$r_{xx} = r_{L0} \sqrt{\left[1 - (\underline{i}_{y} \cdot \underline{i}_{L0})\right]} , \qquad (6.5)$$

$$\underline{b}_{0} = \pm \frac{1}{r_{RE}}, \text{ Unit } (\underline{i}_{W} \times \underline{i}_{LC})$$
(6.6)

and, for the trunnion angle vector,

$$\underline{b}_{0} = \pm \frac{1}{r_{xx}} \left( \underline{i}_{y} \times \underline{i}_{LO} \right) \times \underline{i}_{LO} , \qquad (3.7)$$

where the plus or minus sign, as before, indicates either the CM or LN state vector is to be updated.

Since the antenna angles are independent of velocity, we conclude that  $b_1 = 0$  for both. However, because we are also estimating the angle biases, the  $\underline{b}_2$  partitions of the <u>b</u> vectors are not zero. Indeed, for the shaft angle,

and, for the trunnion angle,

 $\begin{aligned} \mathbf{b}_{\mathbf{g}} &= \begin{pmatrix} 1 \\ 0 \\ 0 \end{pmatrix} \\ \vdots \\ \mathbf{b}_{\mathbf{g}} &= \begin{pmatrix} 0 \\ 1 \\ 0 \end{pmatrix}, \end{aligned}$ 



The estimation of the radar biases is included regardless of whether it is the CM or LM state vector which is being estimated. Since the radar biases do not depend on which vehicle's state vector comprises the first six components of the estimated state, there is no sign selection associated with these <u>b</u>, partitions.

# 7. COMMAND MODULE ORBIT NAVIGATION

When the Apollo spacecraft is in either earth or lunar orbit, navigation data can be obtained by optical measurements of the lines of sight to landmarks. These planotary surface features can be either of the "known" or "unknown" variety. A known landmark is an identifiable feature whose coordinates are known and tabulated. In contrast, an unknown landmark is any surface feature whose coordinates are known and tabulated. In contrast, an unknown landmark is any surface feature whose coordinates are known and tabulated. The brief period during which it is visible. The mechanics of the measuring process is quite similar to the CM-to-LM line of sight measurement procedure described in Soction 5.

# 7.1 Known Landmark Measurements

Ľ,

LEVELON LEVENS LEVENS

As indicated in Section 1, orbit navigation involves a nine-dimensional state vector, the last three components of which are the coordinates of the landmark. Since the tracking period for one landmark is very short (less than one minute in the case of the earth and only two or three minutes during lumar orbit), all navigation data for any particular landmark are acquired before the processing begins. At the conclusion of the tracking, the landmark partition of the state vector is initialized from the identification data entered by the navigator into the computer.

Sufficient CMC erasable storage is allocated for five measurements on a single landmark. The data from each of these sightings consists of the time of the measurement and the set of angles described in Section 5. From these agles the measurement unit vector  $\underline{i}_{-}$  slong the CM-to-landmark line of sight is computed.

Er. h of the measured unit vectors is converted to an equivalent set of two artificial star-landmark measuremen' in exactly the same manner as in the rendezvous navigation procedure discussed in Section 5. The only difference is that the geometry vector is nine-dimensional and is given by

$$\underline{\mathbf{h}} = \frac{1}{\mathbf{x}_{\rm VL}} \begin{pmatrix} \mathbf{i}_{\rm s} \\ \underline{\mathbf{0}} \\ -\underline{\mathbf{i}}_{\rm s} \end{pmatrix}, \qquad (7.1)$$

where  $r_{vi}$  is the current estimated distance between the vehicle and the landmark, and  $\frac{1}{20}$  denotes the direction to the artificial stars.

### 7.8 Unknown Landmark Neasurements

The nine-dimensional orbit navigation technique provides a means of mapping on the surface of a planut a point which is designated only by a number of sets of optical tracking data. This process may be used either to locate a desired LM landing site which may have unknown coordinates or to map features on the surface of the moon.

Assume that a landmark has been tracked and N sets of optical navigation data have been acquired. If the marigator cannot — or chooses not to — identify the landmark, it is then treated as an unknown landmark. In this process the data from the first navigation measurement are used to compute an initial estimate of the landmark location. The mine-dimension state vector is then formed, and the data from the remaining N - 1 sightings are incorporated exactly as if the optically-designated point had been an identified landmark.

The determination of the initial estimate of the landmark position vector  $\mathbf{r}_{L}$  is accomplished as follows. Let  $\mathbf{i}_{L}$  be the measured unit vector from the vehicle to the landmark calculated from the data of the first measurement. Then

$$\underline{\mathbf{r}}_{\mathbf{L}} = \underline{\mathbf{r}}_{\mathbf{P}\mathbf{V}} + \underline{\mathbf{r}}_{\mathbf{V}\mathbf{L}} \mathbf{\mathbf{j}}_{\mathbf{L}} , \qquad (7.2)$$

with ret obtained by applying the law of cosines to the triangle defined by Equation (7.2). We then have

$$r_{UL}^2 = (2r_{PV} \cos A_L)r_{VL} + (r_{PV}^2 - r_{P}^2)^2 = 0$$
, (7.3)

where  $r_p$  is the radius of the primary body, and  $A_1$  is the angle between the directions from the vehicle to the center of the primary body and the landmark and is computed from

$$\operatorname{cos} A_L \approx -\underline{i}_m \cdot \operatorname{Unit}(\underline{r}_{py})$$
 .

Molving Equation (7.3) and selecting the appropriate sign yields

$$\mathbf{r}_{\mathbf{VL}} \simeq \mathbf{r}_{\mathbf{P}\mathbf{V}} \left[ \cos A_{\mathbf{L}} - \left( \frac{\mathbf{r}_{\mathbf{P}}^2}{\mathbf{r}_{\mathbf{P}\mathbf{V}}^2} - \sin^2 A_{\mathbf{L}} \right)^{1/2} \right]. \tag{7}$$

353







A STAR PROVINCE MANY



. 4)

Thus, the initial estimated location of the unknown landmark is established. The remaining N-1 sets of data are then used as for a known landmark, according to Section 7.1.

### 7.3 Initialization of the Error Transition Matrix

Beveral different landmarks are tracked by the Apollo havigator is navigate in orbit about the earth or moon. As a consequence, special methods are required to reinitialize the error transition matrix each time a new landmark is acquired. This is necessary to reflect the fact that the initial landmark location errors are not correlated with the errors in the estimated CM position and velocity vectors. Therefore, before processing the measurement data associated with a new landmark, it is necessary to convert the nine-dimensional error transition matrix  $\underline{W}$  to a six-dimensional matrix having the same CM position und velocity error variances and covariances. A new nine-dimensional matrix is then formed by augmenting appropriate landmark uncertainty information.

As the first stage in the initialization process it is necessary to determine a square root of the sixdimensional partition  $\mathbf{E}_{s,s}$  of the nine-dimensional covariance matrix  $\mathbf{E}$ , defines in Equation (2.19). From Equations (2.21) and (3.8), it is clear that

$$\mathbf{\overline{H}}_{6.6} = \begin{pmatrix} \underline{\mathbf{W}}_{3}^{\mathsf{T}} & \underline{\mathbf{W}}_{1}^{\mathsf{T}} & \underline{\mathbf{M}}_{2}^{\mathsf{T}} \\ \underline{\mathbf{W}}_{3}^{\mathsf{T}} & \underline{\mathbf{W}}_{4}^{\mathsf{T}} & \underline{\mathbf{M}}_{6}^{\mathsf{T}} \end{pmatrix} \begin{pmatrix} \underline{\mathbf{W}}_{0} & \underline{\mathbf{W}}_{0} \\ \underline{\mathbf{W}}_{1} & \underline{\mathbf{M}}_{4} \\ \underline{\mathbf{W}}_{2} & \underline{\mathbf{W}}_{4} \end{pmatrix}.$$
(7.5)

The square root of  $\ {\bf g}_{ee}$  , i.e., a six-dimensional matrix  $\ {\bf y}_{ee}$  such that

Nee Hes : (7.6)

is not unique. However, a convenient method, which is used in the CMC; is to select a triangular form for  $\underline{W}_{dS}$ , so that the resulting set of algebraic equations for the elements of  $\underline{W}_{dS}$  is most easily solved.

The second part of the <u>H</u> matrix initialization depends on whether the landmark being tracked is known or unknown. For a known landmark, the new nine-dimensional <u>H</u> matrix is formed as:

$$\underline{\mathbf{n}} = \begin{pmatrix} \mathbf{u}_{0}^{T} & \underline{\mathbf{n}}_{1}^{T} & \underline{\mathbf{0}} \\ \mathbf{u}_{0}^{T} & \underline{\mathbf{n}}_{0}^{T} & \underline{\mathbf{0}} \\ \underline{\mathbf{0}} & \underline{\mathbf{0}} & \underline{\mathbf{n}}_{0}^{T} \end{pmatrix}, \qquad (7.7)$$

with the upper left-hand six by six partition being the square root matrix  $\underline{W}_{od}$  found in part one. The threedimensional sub-matrix  $\underline{W}_{o}$  is given a value consistent with the expected errors in the knowledge of the landmark location. In particular,  $\underline{W}_{o}$  is chosen so that

$$\underline{\mathbf{y}}_{\mathbf{1}}^{\mathsf{T}}\underline{\mathbf{x}}_{\mathbf{0}} = \underline{\mathbf{y}}_{\mathbf{1}}^{\mathsf{T}}, \qquad (7.8)$$

where  $\zeta$  is the vector error in the landmark position.

If, on the other hand, the measurements are made using an unknown landmark, the error in the initially computed estimate of the location of the landmark will be a function of the uncertainties in the CM position estimate, the tracking accuracy and the altitude of the landmark above the gravitational center (see Reference 4). In this case, part two for the initialization of the  $\underline{W}$  matrix is more complicated.

The basic relationship among the quantities of interest is Equation (7.2). Errors in  $r_{\rm FV}$  and  $\underline{i}_{\rm a}$ , denoted respectively by  $\underline{q}$  and  $\underline{\delta}_{\underline{i}_{\rm a}}$ , will produce an error in the landmark location  $r_{\rm L}$ . However, this error is clearly in the plane of the landmark, i.e., perpendicular to  $\underline{r}_{\rm L}$ . Also, an error in  $r_{\rm VL}$  will result in a landmark position error in the direction of  $\underline{i}_{\rm d}$  and of a magnitude  $\delta r_{\rm L}/(\underline{r}_{\rm L})/(\underline{i}_{\rm TL})$ , where  $\delta r_{\rm L}$  is the error in the landmark altitude. Thus, the total error  $\underline{i}_{\rm d}$  in the landmark location is given by

$$\underline{\underline{\mathbf{r}}} = \underline{\underline{\mathbf{u}}}(\underline{\mathbf{c}} + \mathbf{r}_{\mathsf{VL}}\delta\underline{\mathbf{j}}_{\mathsf{H}}) + \delta\mathbf{r}_{\mathsf{L}} \frac{\sqrt{|\underline{\mathbf{x}}_{\mathsf{L}}^{\mathsf{T}}\underline{\mathbf{L}}_{\mathsf{L}}|}}{\underline{\mathbf{j}}_{\mathsf{H}}} \underline{\mathbf{j}}_{\mathsf{H}}, \qquad (7.8)$$

 $\underline{\mathbf{u}} = \mathbf{I} - \frac{\mathbf{I}_{\mathbf{u}} \mathbf{\overline{u}}}{\mathbf{I}_{\mathbf{u}}^{T}} \tag{7.10}$ 

is the projection operator which assures that only the components of  $\underline{\epsilon}$  and  $\delta \underline{i}_{\underline{n}}$  in the plane normal to  $\underline{r}_{\underline{i}}$  are related to the landmark location error.

Now, consider a coordinate system in which the direction  $\frac{1}{100}$  is along one of the coordinate axes. Then, if T is the transformation matrix which relates the selected axis system and the original reference system, we have

where







$$\mathbf{J}_{\mathbf{R}} \rightarrow \mathbf{T} \begin{pmatrix} \mathbf{0} \\ \mathbf{0} \\ \mathbf{1} \end{pmatrix}.$$

The error in 1 may be expressed as:

121 - 12 C

$$\delta_{\underline{i}_{\underline{\alpha}}} = \underline{I} \begin{pmatrix} \alpha \cos X \\ \alpha \sin X \\ 0 \end{pmatrix},$$

where  $\alpha$  is the small random angle between the true and measured directions to the landmark. The polar angle X is defined in the plane normal to  $\underline{j}_{m}$  from the coordinate axis to the projection of the measured direction of  $\underline{j}_{m}$ .

Assume that  $\alpha$  and  $\lambda$  are statistically independent random variables with zero means. Further, assume that  $\lambda$  is uniformly distributed over the interval  $-\pi$  to  $\pi$ . Then, for the covariance matrix of the uncertainty in  $\underline{1}$ , we obtain

$$\overline{\underline{\delta}_{\underline{i}\underline{n}}}^{T} = \frac{1}{2}\sigma_{\alpha}^{2} \left( \begin{pmatrix} 1 & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 0 \end{pmatrix} \right)^{T}$$
$$= \frac{1}{2}\sigma_{\alpha}^{2} \left( \underline{j} - \underline{i}_{\underline{n}} \underline{j}_{\underline{n}}^{T} \right) ,$$

where  $\sigma_{\alpha}^{\sharp}$  is the variance of  $\alpha$ .

Finally, since g and the velocity error  $\eta$  are statistically independent of  $\delta_{i_{n}}$  and  $\delta_{r_{i_{1}}}$ , we may calculate the following covariance matrices from Equation (7.9):

$$\frac{\zeta e^{T}}{\zeta \Omega^{T}} = \underline{U} \underline{e} \underline{v}^{T}$$

$$\frac{\zeta u^{T}}{\zeta \Omega^{T}} = \underline{U} \underline{e} \underline{v}^{T}$$

$$\frac{\zeta u^{T}}{\zeta \Omega^{T}} = \underline{U} \underline{e} \underline{v}^{T} + \frac{1}{2} \sigma_{\alpha}^{2} v_{\nu L}^{2} \underline{U} (\mathbf{I} - \underline{i}_{\alpha} \underline{i}_{\alpha}^{T}) \underline{U}^{T} + \sigma_{L}^{2} \frac{\underline{\Gamma}_{L}^{T} \underline{L}_{L}}{(\underline{i}_{\alpha}^{T} \underline{i}_{\alpha})^{2}} \underline{i}_{\alpha} \underline{J}_{\alpha}^{T}$$

$$(7.11)$$

'As 's result, the W matrix is initialized as

$$\underline{\mathbf{H}} = \begin{pmatrix} \mathbf{H}_0^T & \mathbf{E}_1^T & \mathbf{Q} \\ \mathbf{H}_8^T & \mathbf{H}_4^T & \mathbf{Q} \\ \mathbf{U}\mathbf{H}_0^T & \mathbf{U}\mathbf{H}_1^T & \mathbf{H}_8^T \end{pmatrix}, \qquad (7.12)$$

where I is now Mutermined as the three-dimensional triangular square root of

$$\mathbf{E}_{0}^{T} \mathbf{W}_{0} = \frac{1}{2} \sigma_{n}^{2} \mathbf{r}_{0}^{2} \mathbf{U} \left( \mathbf{I} - \mathbf{j}_{n} \mathbf{j}_{n}^{T} \right) \mathbf{U}^{T} + \sigma_{1}^{2} \frac{\mathbf{E}_{0}^{T} \mathbf{E}_{0}}{\left( \mathbf{j}_{n}^{T} \mathbf{j}_{n} \right)^{2}} \mathbf{j}_{n} \mathbf{j}_{n}^{T} . \qquad (7.13)$$

After incorporating the data obtained from the tracking of a landmark, all 81 elements of the <u>N</u> matrix will again, in general, be non-zero, indicating correlations among all components of vehicle position and velocity and landmark position. This matrix is extrapolated, as described in Section 3, until a new landmark is acquired necessitating a new initialization.

# 8. FLIGHT EXPERIENCE OF THE APOLLO NAVIGATION SYSTEM

The first manned trip to the vicinity of the moon of Apollo 8 during December 1968 gave an excellent test of the Apollo system's on-board navigation capability (see Reference 5). Although ground tracking navigation was the primary mystem, the on-board navigation system had the task of confirming a safe trajectory and providing a back-up for return to earth in the remote chance that ground assistance became unavailable for on-board use.

Apollo 8 was to use sun illuminated visual horimons rather than landmarks for operational simplicity, even though, as confirmed from earth orbit by Astronaut Don Eisele in Apollo 7, the earth's horizon does not provide a distinct target for visual use. Moreover, the filter in the sextant beamsplitter, designed originally to enhance the contrast between water and land when looking down at the earth, filters out the blue in such a way as to make the horizon even more indistinct. Originally a blue sensitive photometer had been designed for horizon detection in the prototype sextant models, but was removed from the production systems, since a decision had been made that ground tracking would be the primary source of middourse navigation. Without the photometer, interest in the earth's horizon as a visual target resulted in demonstrations on simulators that, in some subjuctive way, the human with a little experience can choose an altitude sufficiently repostable, at least as good as ±3 kilometers. Accordingly, a few weeks before the Apollo 8 launch, the navigation command modulo pilot, Jim Lovell, spent a few



hours on the sextant earth horizon simulator at the Massuchusetts Institute of Technology in Cambridge for training and to childrate the horizon altitude he seemed to prefer. No was remarkably consistent in shoosing a location 72.8 kilometers above the sea level horizon. This value was recorded in the CMC as part of the pre-launch mission load.

The plan for the mission was to examine the sextant angle measurements make warly while still near the earth and, based on the spaceoraft state vector determined by ground trucking, infer in real-time the horizon altitude Lovell was using. After the first eleven sightings on the earth at distances of about 30 thousand manifold miles from earth, it was estimated that he was using an 18.2 kilometers altitude and the CMC was releved with this new value. (Later during the mission it was agreed that a truer estimate was nearer 23 kilometers, but the value was not changed since the difference then was too small to be of concern.) Following the horizon calibration, the first mideourse correction of minose 25 ft/sec was performed. The large size of this correction was due to trajectory perturbations, resulting from the maneuvers performed in getting the spaceoraft safely away from the third stage of the launch vehicle.

After the mideourse correction, the CMC state vector was made to agree with the value obtained from ground tracking. The important parameter, predicted perilume altitude, was 60.7 nautical miles - very close to the true value estimated later to be 68.8. The next 31 navigation measurements were made using the earth's horizon, modeled at 18.2 kilometers altitude. Being sufficiently far from both earth and mean during this time, it is not surprising that the initially good state vector was degraded. At the end of this period, the indicated perilume was 32 mantical miles below the moon's surface. With the most fine sufficiently far from the sightings, still using the earth's horizon, the predicted perilume increased to 92.9 mantical miles - make altitude of the sarth's horizon was unimportant for these sightings since the distance from earth was now approximately 150,000 nautical miles, so that the 10 arc-second accuracy of the saxtant was the predominant source of error.

The next group of 16 sightings was made using the moon horizon at a distance of about 80,000 hautical miles. As would be expected, the first few of these resulted in fairly large changes in the estimated atste vector, while the remaining had a very small affect. At the end of this group of measurements, the indicated perilume was 67.1 nautical miles. The final set of 15 translumar sightings was made about 35.000 nautical miles from the moon with little additional effect on the perilume estimation. The final estimate was 67.6 nautical miles or about 1.3 nautical miles lower than the value later reconstructed from ground tracking dats. At this time the on-board and ground tracking dats were practically identical and consideration was given to using the on-board state vector for lunar orbit insertion. Although the state vector update hardly changed the on-board value, it was performed since there was up overriding argument to deviate from the flight plan.

The transearth flight of Apollo 8 after 10 lunar orbits also provided a good measure of the on-buard navigation capability. The transearth injection measurer of the aervice propulsion system was targetted by ground data and executed in back of the moon by the on-board digital autoplict and guidance systems. This J522.5 ff/eec mansuver was followed by a single middourse correction of 4.8 ff/acc 14.7 hours later, resulting in entry conditions at 600,000 feet altitude above the earth which were 0.8 ff/acc faster and 0.1° shallower than planned.

Although the primary navigation during this period was again the ground tracking network, 138 on-board navigation measurements were performed by Lovell was amonitor and back-up. In order to determine what would have happened without ground assistance, the actual on-board measurements were incorporated in a simulation with the computer initialized to the actual on-board state vector as it existed when the spacecraft emerged from bohind the moon. The single transcarth middourse correction was added appropriately to this simulation, in accordance with that actually measured by the inertial guidance system. (In the actual flight a new ground determined state vector was loaded into the computer at the time of this menever.)

The last of the 138 measurements was completed 16 hours before entry. The incorporation of these measurements in the simulation left a hypothetical on-woard estimate of entry flight path angle at 400,000 feet of  $-6.38^{\circ}$ , as compared with the ground tracking estimate of  $-6.48^{\circ}$ . This 0.1° difference was well within the mafe tolerance of  $\pm 0.5^{\circ}$ . Another parameter of concern at the entry interface is the error in knowledge of altitude rats. The simulated on-board estimate of this quantity differed from that estimated by ground tracking by 230 ft/sec. However, the conservative allowable tolerance is  $\pm 200$  ft/sec.

It should be apphasized that, in the event ground data were not available, the plan was to continue the onboard measurements to optimize the final middourse correction and state vector for safe earth atmospheric entry. In the absence of actual flight data, a continuation of the simulation using the planned sighting program was made with standard deviation errors in the sextant of 10 arc-seconds and in the horizon of 3 kilometers. In addition, bias errors of 5 arc-seconds in the sextant and 4 kilometers in the horizon were included. The resulting estimation error in the entry  $s_{12}$  at the entry interface had a standard deviation of  $0.03^{\circ}$  and a bias of  $0.007^{\circ}$ . The corresponding altitude rate uncertainty had a standard deviation of 41.1 ft/sec and a bias of 26.5 ft/sec. The capability of the on-board navigation system to bring the spacecraft safely back from the moon seems clearly to have been demonstrated.



tradining Tangana in Tangana in Stati tan

368

# REFERENCES

1. Miller, J.E. (Editor) et al.	Space Navigation Guidonce and Control. Technivision Ltd, Maidenhead, England, 1968.
2. Battin, R.H.	Astronautical Guidance, McGraw-Hill, New York, 1904.
3. Henrici, P.	Discrete Variable Methods in Ordinary Differential Equations. John Wiley, Now York, 1962.
4. Bellantoni, J.F.	Unidentified Landmark Navigation. AIAA Journal, Vol.5, August 1967, pp.1478-1483.
5 Unio 0.0	Analla Cantonal Children and Navigation Sustan A December Papart Burnented at

Apollo Control, Guidance and Navigation System - A Progress Report. Presented the National Space Navigation Neeting of the Institute of Navigation. Houston, Texas, April 1969.

XOX

.6

# State Construction Construction

2007 STATE - 19 2010

1.2.2.2.2.2.2.1.2.2.

Ë

sented at







### Coasting flight navigation Fig. 5



Fig. 6 Weighting vector calculation





المدار والموار والم

# CHAPTER 15 - SOME APPLICATIONS OF KALMAN FILTERING

现入为相关的认为

4.5

# IN SPACE GUIDANCE

# J.F. Bellantoni

by

# National Aeronautics and Space Administration Electronics Research Conter Cambridge, Massachusetts

V

# PRECEDING PAGE BLANK

The material in this chapter represents the views of its author and is not to be construed as an official position of the US Government or of the Electronics Research Center.













# J.F. Bollantoni

# 1. INTRODUCTION

Among the first applications of linear optimal recursive filtering were those made by Battin<sup>1</sup> and by MoLeon, Smith, Schmidt and McGee<sup>2,22</sup> to the estimation of the position and velocity of a spaceoraft. The field of space mavigation still provides a fruitful area of application of this technique. Two of the applications described in this chapter deal directly with space navigation: the error analysis of a satellite navigation system, and unidentified landmark navigation. The third application deals with initial alignment of an inertial guidance system that will indicate vehicle position and velocity during boost into a free-fall trajectory.

### 2. ERROR ANALYSIS OF A SATELLITE NAVIGATION SYSTEM

# 2.1 The Kalman Filter as a Tool in Error Analysis

One of the useful properties of the linear recursive optimal filter (Kalman filter) is that it generates its own error analysis in the process of estimating the state. In fact, the covariance matrix P(t) of error in the estimate  $\hat{\mathbf{x}}(t)$  , upon which the error analysis rests, may be computed recursively over the time of interast without benefit of real data. This is of importance in the preliminary design stages of aircraft and spacecraft guidance systems, since a prediction of system performance may be made knowing only the statistics of the instrument errors and the times of the observations. The basic equations are that, for the propagation of the covariance matrix between measurement times.

$$\frac{d}{dt} P(t) = P(t)P(t) + P(t)P^{T}(t) + O(t)\Gamma(t)O^{T}(t), \qquad (2.1)$$

where

- $P(t) = E[x(t) \hat{x}(t)][x(t) \hat{x}(t)]^T$ 
  - = covariance of error in estimate  $\hat{x}(t)$
- F(t) = system matrix in state equation
- $x(t) = state vector, given by <math>\dot{x}(t) = F(t)x(t) + G(t)w(t)$
- $\hat{\mathbf{x}}(t) = \mathbf{x}_{\mathbf{x}}$  interestimate of  $\mathbf{x}(t)$
- $[\Gamma(t)\delta(t) = \pi[\pi(t) \pi(t)][\pi(t) \pi(t)]^{T}$ 
  - w(t) = white-noise driving state equation
  - $\Psi(t) = Ew(t) = 0$
  - $\delta(t) = Dirac delta function of t$
  - Q(t) = input matrix for w(t)
  - E() = expectation of ()
  - $()^{\overline{n}}$  = transpose of ()

so that, for the covariance matrix  $P_i^*$  after incorporation of data at time t, ,

# $\mathbf{P}_{i}^{*} \stackrel{\text{\tiny left}}{=} \mathbf{P}_{i}^{*} - \mathbf{P}_{i}^{*} \mathbf{H}_{i}^{T} \mathbf{H}_{i} \mathbf{P}_{i}^{*} \mathbf{H}_{i}^{T} + \mathbf{R}_{i} \mathbf{J}^{**} \mathbf{H}_{i} \mathbf{P}_{i}^{*} \ ,$

**PRECEDING PAGE BLANK** 



where

 $P_{\underline{i}}^{*} = P(t_{\underline{i}}^{*})$   $P_{\underline{i}}^{*} = P(t_{\underline{i}}^{*})$   $R_{\underline{i}} = R(t_{\underline{i}})$ 

 $H_i$  = observation matrix at time  $t_i$ 

 $t_i^-$ ,  $t_i^+$  = times just before and just after observation at  $t_i^-$ .

An alternate form of Equation (2.1), useful when the state transition matrix  $\phi(t_i, t_{i-1})$  is known or may be approximated, is  $P(t_i) = \phi(t_i, t_{i-1})\phi^{\dagger}(t_i, t_{i-1}) + Q(t_i, t_{i-1})$ (2.1)

$$\mathbf{t}_{i}^{*} = \phi(\mathbf{t}_{i}, \mathbf{t}_{i-i}) \mathbf{P}(\mathbf{t}_{i-i}^{*}) \phi^{*}(\mathbf{t}_{i}, \mathbf{t}_{i-i}) + \mathbf{Q}(\mathbf{t}_{i}, \mathbf{t}_{i-i}) , \qquad (2, 1)'$$

where

and the second second

1 110

$$\mathbf{R}(\mathbf{t}_{\mathbf{i}},\mathbf{t}_{\mathbf{i}-\mathbf{i}}) = \int_{\mathbf{t}_{\mathbf{i}-\mathbf{i}}}^{\mathbf{t}_{\mathbf{i}}} \phi(\mathbf{t}_{\mathbf{i}},\tau) \mathbf{U}(\tau) \Gamma(\tau) \mathbf{Q}^{\mathsf{T}}(\tau) \phi^{\mathsf{T}}(\mathbf{t}_{\mathbf{i}},\tau) \, \mathrm{d}\tau \quad .$$
(2.3)

If approximations, adequate for the purposes of error analysis, are available for  $\phi$  and Q, then Equation (2.1)' is superior to Equation (2.1) in that it avoids the integration implicit in Equation (2.1).

The assumptions underlying this type of error analysis impose limitations that may preclude its use in some applications. The major assumptions are:

(1) The state vector x(t) is adequately described by a linear differential equation driven by white-noise;

 $\hat{x}(t) = F(t)x(t) + G(t)w(t)$ , (2.4)

where F(t) and G(t) are known,

(3) The observations  $Z(t_i)$  are linearly rolated to the state  $x(t_i)$  by

$$Z(t_i) = H(t_i) X(t_i) + V(t_i),$$
 (2.5)

where  $H(t_i)$  (w  $H_i$ ) is known and  $v(t_i)$  is white-noise.

(2) The mean and variance of w(t) and v(t) are known.

(4) The minimum variance linear filter  $\hat{R}_i$  is used to improve the state estimate, as follows:

$$(t_{i}^{\dagger}) = \hat{\chi}(t_{i}^{\dagger}) + \hat{R}_{i}[Z(t_{i}) - H_{i}\hat{\chi}(t_{i})]$$
 (2.8)

The last assumption (4) is not essential. If the filter is linear, but not necessarily minimum variance, Equation (2.2) may be replaced by the more general expression

$$\mathbf{P}_{i}^{*} = (\mathbf{I} - \mathbf{K}_{i}\mathbf{H}_{i})\mathbf{P}_{i}^{*}(\mathbf{I} - \mathbf{K}_{i}\mathbf{H}_{i})^{T} + \mathbf{K}_{i}\mathbf{R}_{i}\mathbf{K}_{i}^{T}, \qquad (2.7)$$

where  $K_i$ , a general linear filter, has been used in Equation (2.6) instead of  $\hat{K}_i$ . A simple derivation of Equation (2.7) is given in Appendix A. It is easy to show that Equation (2.7) reduces to Equation (2.2) when  $\tilde{K}_j = \hat{K}_j$ , where

$$\hat{R}_{i} = P_{i}^{*} H_{i}^{2} [H_{i} P_{i}^{*} H_{i}^{2} + R_{i}]^{-1}, \qquad (2.8)$$

Equation (2.7) is of use in analyzing the error of a linear, sub-optimal filter. In particular, the suboptimal filter being analyzed may be a Kalman filter bused on (1) an approximate state transition matrix, or (2) a state containing only some of the variables of the problem, or (3) some approximation in the filter designed to save computation bine or space.

The choice of state vector, and the choice of models for the variables contained therein, determine the state transition matrix  $\phi$  and the noise covariance Q in Equation (2.1)'. For error analysis purposes it is often possible to make enough approximations in the error models to allow use of the integrated form Equation (2.1)' rather than Equation (2.1). The integrated form will be used exclusively in the remainder of this section.

The choice of measurement types determines the observation matrix H and the covariance R of observation noises in Equation (2.2). Along with the assumed observation times and the initial value  $P_0$  for P, the four matrices

φ, Q, H, N

completely determine the computation of the state covariance matrix P and hence characterize the error analysis.



THE STREET

**HO**R

### 3.3 The Satellite Navigation Problem

The position and velocity of a vehicle in earth orbit are usually calculated from a series of measurements made by ground radar stations and instruments, such as horizon trackers, star trackers and landmark telescopes, located on board the vehicle. In order to compare the merits of various combinations of instruments, it is mecessary to estimate the not accuracy with which any particular combination predicts position and velocity and, in some cases, whicle attitude. If it is assumed that a minimum variance filter is employed to process the data, then Equations (2.1) or (2.1)<sup>7</sup> and (2.2) give the second-order statistics of the position and velocity, and of any other variables included in the state vector. If the filter is linear, but not necessarily minimum variance, then Equation (2.2) is to be replaced by Equation (2.7) to achieve the same results.

The scope of the error analysis is essentially determined by the choice of (a) the state vector, (b) the measurement types, and (c) the realism desired in determining when the measurements coour. The last consideration should not be overlooked as a factor in the complexity of a digital computer program.

The most convenient place to start the analysis of a practical filter problem such as this has been found to be with (b), the measurement types, rather than with the state vector. The reason is that many of the variables in the state vector are noises or biases originating in the measuring instruments. In addition, a deft definition of the state variables, suggested by the measurement matrix, often reduces the complexity of the state transition matrix. Hence, the observation types will be discussed in Section 2.3 and the choice of state variables in Section 2.4.

The inertial coordinates X, Y, Z describing the vehicle position are shown in Figure 1, and the vehicle vertical coordinates  $\xi$ .  $\eta$ ,  $\zeta$  are shown in Figure 2.

# **3.3** Ubservation Types

The purpose of this section is to describe some of the observations that are commonly used in satellite orbit determination. Table I lists the H-matrices corresponding to these observations.

### Ground Based Radar

All All All and a start of the start of the

Figure 3 shows the geometry of a tracking station. The four observations are

2	P	*	LTURO		<b>`</b> (	2. (	D)	

$\mu = \frac{1}{dt} \rho = range rate$	(2.10)
--	--------

A.	ų	tan <sup>-1</sup>	(⊅.€/⊅.â)	asianuth	(2.11)

 $\overline{n} = \sin^{-1} \left( \overline{\rho} \cdot \hat{h} / \rho \right) = \text{elevation}, \quad (2.12)$ 

where the symbols used are defined in Figures 1, 2 and 3. Each of these observations is affected by the vehicle position and/or vehicity and can be expected to give partial orbit determination information. However, the uncertainties  $\Delta e$ ,  $\Delta n$ ,  $\Delta h$  of the station's own location are a major source of error, comparable to the noise level on range, and must be considered as variables in the state vector in any realistic error analysis. Table I, therefore, shows the H-matrices corresponding to station location errors as well as to vehicle position and velocity. Finally, the biases our range and range rate and in azimuth and elevation are often important and are slice shown in Table I. These biases are denoted by  $\Delta \rho$ ,  $\Delta \dot{\rho}$ ,  $\Delta \dot{\rho}$  and  $\Delta E$ .

### Inertial Measurement Urit

Most experiments carried on a satellite require knowledge of the vehicle's orientation in inertial space in order to produce weful data. Since it is often impractical to control the vehicle's attitude precisely, a gyromomorphically stabilized platform is often required. Experimental observations are then made (perhaps using the vehicle itself as an intermediate coordinate frame) relative to the gyromomorphically stabilized axes. The stabilized coordinate frame itself may be referenced to inertial space by a star tracker that makes observations of two or more stars. It will be assumed that the gyroscopically stabilized axes are maintained coincident either with the inertial axes XYZ of Figure 1 or with local vertical axes  $\xi \mathcal{H}_{\lambda}^{c}$  of Figure 2. Where necessary, the two cames will be distinguished.

Observations made relative to the synchooplically stabilized (IMU) area will contain errors due to the disorepancy between the true and the indicated crientations of the IMU. This discrepancy may be represented as small rotations  $\theta_1$ ,  $\theta_2$ ,  $\theta_3$  about the IMU gyroscopic axes, i.e. along the XYZ or  $\frac{2}{2}\eta^2 \xi$  axes. These gyro angles  $(\theta_1, \theta_2, \theta_3)$  will be needed in the state vector, not only because they are important errors in many of the observations to be described, but because they are a large part of the attitude error of the vehicle. In many missions, attitude has an importance equivalent to that of position and velocity. A suitable model for  $\theta_1$ ,  $\theta_2$  and  $\theta_3$  will be described in the section dealing with the state variables. It should be noted that only the attitude indication of the IMU is of interest, since its accelerometers, if any, give practically no output in the absence of thrust, the case under consideration.

367









and the state of the state

annosi Carandola Caralla

### Horison Tracker

A horizon tracker may be pictured, for the sake of error analysis, as four independent sighting telescopes, two pointing to the forward and backward horizons in the orbital plane, and two pointing to the left and right horizons in a plane normal to the orbit (see Figure 4). The tracker measures the four angles  $\Phi_1$ ,  $\Phi_2$ ,  $\Phi_3$ ,  $\Phi_4$  of the telescopes relative to the  $\frac{2}{2} \hat{\gamma} \hat{\zeta}$  coordinate frame at the vehicle. These angles are combined to give an average stadimetric angle

$$\bar{\Phi}_{1} = \frac{1}{2}(\Phi_{1} - \Phi_{2} + \Phi_{4} - \Phi_{3})$$
(2.13)

and two angles.  $\Phi_{g}$  and  $\Phi_{g}$ , that indicate the direction of the local vertical relative to the  $\hat{e}\hat{\gamma}\hat{\xi}$  coordinate frame:

$$\overline{\Phi}_{g} = \frac{1}{2} (\Phi_{1} + \Phi_{g}) \tag{2.14}$$

$$\Phi_{3} = \frac{1}{2} (\Phi_{3} + \Phi_{4})$$
 . (2.15)

One of the major errors in these observations is in the deviations  $h_1$ ,  $h_2$ ,  $h_3$ ,  $h_4$  of the sensible horizon heights from their assumed values  $H_1$ ,  $H_2$ ,  $H_3$ ,  $H_4$ . Corresponding to the errors  $\delta \Phi_1$ ,  $\delta \Phi_2$ ,  $\delta \Phi_3$  in the observations of Equations (2.13), (2.14) and (2.15), there are three combined horizon deviations,

$$\tilde{h}_{1} = \frac{1}{2}(h_{1} + h_{2} + h_{3} + h_{4}) = \sqrt{(R^{2} - R^{2}_{RR})}\delta \Phi_{1}$$
 (2.16)

$$\tilde{h}_{\mu} = \frac{1}{2}(h_{\mu} - h_{\mu}) = \sqrt{(R^2 - R^2_{\mu\mu})} \delta \tilde{\Phi}_{\mu}$$
 (2.17)

$$\tilde{h}_{a} = \frac{1}{2}(h_{a} - h_{a}) = \sqrt{(R^{2} - R^{2}_{a})} \delta \tilde{\Phi}_{a},$$
 (2.18)

where R is the distance from the menter of the earth to the vehicle and  $R_{\rm EH}$  is the distance from the center of the earth to the assumed horizon altitude. It is possible to take these combined horizon deviations as three independent state variables. A convenient model for them, consisting of time correlated zero-mean noises, will be discussed in the next section. In addition, it is often necessary to consider the bias errors in the  $\tilde{\Phi}_1$ ,  $\tilde{\Phi}_2$ ,  $\tilde{\Phi}_3$  observations produced by a constant error in the estimated horizon heights. Such an error will produce a bias in  $\tilde{\Phi}_1$ , but not in  $\tilde{\Phi}_2$  and  $\tilde{\Phi}_3$ . Hence it is often realistic to introduce a state variable  $\Delta \tilde{H}_1$  for horizon bias affecting the stadimetric angle  $\tilde{\Phi}_1$ .

A second major error in the combined observations is in the tracking instrument itself. Instrument errors appear in the angles  $\Phi_1 \ \ \Phi_2 \ \ \Phi_3 \ \ \Phi_4$ , and hence in  $\Phi_1 \ \ \Phi_2 \ \ \Phi_3 \ \ \Phi_4$ . To describe them one may postulate combined instrument errors  $\phi_1 \ \ \phi_2 \ \ \phi_3 \ \ \Phi_4$ , and hence in  $\Phi_1 \ \ \Phi_2 \ \ \Phi_3 \ \ \Phi_4$ . To describe them one may postulate combined instrument errors  $\phi_1 \ \ \phi_2 \ \ \phi_3 \ \ \Phi_4$ , and hence in  $\Phi_1 \ \ \Phi_2 \ \ \Phi_4$ . These errors may be introduced as time-correlated variables in the state vector, if a suitable model is found. A rather complete model would include biases  $\Delta \Phi_1 \ \ \Delta \Phi_4$  and  $\Delta \Phi_3$ , and purely white-noise components. In general, the time constant of the time correlated error is small enough to allow this error to be considered as white-noise.

A third error that appears in the horizon tracker observations is that due to the IMU error angles  $\theta_1$ ,  $\theta_2$ ,  $\theta_4$ . Here the two cases of IMU orientation must be distinguished. Moreover, errors in IMU crientation do not affect the stadimetric angle  $\Phi_1$ , but do add directly to the observations  $\Phi_2$  and  $\Phi_3$  of the vertical.

In summary, Table 1 lists the matrices relating the effective horizon tracker observations  $\tilde{\Phi}_1$ ,  $\tilde{\Phi}_2$ ,  $\tilde{\Phi}_3$  to the following variables in the state vector: vehicle position errors  $\mathbf{x}$ ,  $\mathbf{y}$ ,  $\mathbf{s}$ ; effective horizon deviations  $\tilde{\mathbf{h}}_1$ ,  $\tilde{\mathbf{h}}_2$ ,  $\tilde{\mathbf{h}}_3$ ; IMU error angles  $\theta_1$ ,  $\theta_2$ ,  $\theta_3$ ; effective tracker errors  $\tilde{\Phi}_1$ ,  $\tilde{\Phi}_2$ ,  $\tilde{\Phi}_3$ ; effective horizon tracker biases  $\Delta \Phi_1$ ,  $\Delta \Phi_2$ ,  $\Delta \Phi_3$ ; horizon deviation bias  $\Delta \mathbf{H}_1$ .

### Star Tracker

A star tracker provides highly accurate attitude information to the IMU axes and, thereby, to other instruments. The observations are train angle (TR) and elevation angle (EL), relative to the  $\frac{2}{3}$   $\frac{2}{3}$  reference frame, of the unit vector from the vehicle towards the star. Figure 5 shows these observations. Note that here TR and EL are measured relative to the vehicle vertical coordinates, even if the IMU axes are along XYZ. For simplicity, the same geometry is assumed for star and landmark observations.

The errors in the TR and EL observations are those in the instrument itself and in the IMU orientation, because star directions are known with high precision.

The instrument errors are divided, as usual, into two correlated errors  $\Omega_1$ ,  $\Omega_2$ , bias errors  $\Delta TR$ ,  $\Delta BL$ , and pure white-noise. The total instrument error is often of the order of arc-seconds.

The IMU orientation errors  $\theta_1$ ,  $\theta_2$ ,  $\theta_3$  usually dominate the total error if more than a few minutes have elapsed since the last star observations.











A DIA AND A DIA

intelle Priste start Dieter

· 309

# Landmark Sensor

5.63%

いたので

COLOR DA

Observation of a landmark of known location on the earth may be made with a telescope by a drew member.

The geometry of a landmark observation is assumed to be similar to that of a star observation (see Figure 5). But, since the landmark is not at infinity, as is a star, the observation is affected by, and hence yields information about, the vehicle's position as well as its attitude. The errors involved may be modelled as time correlated errors  $\pi_1$ ,  $\pi_2$ , bias errors  $\Delta TR$ ,  $\Delta EL$ , and pure white-noise, just as in the case of the star tracker.

### Rodar Altimeter

This instrument measures RA, the distance  $|\vec{R} - \vec{R}_{gg}|$  between the vehicle and the sub-satellite point  $\vec{R}_{gg}$  (see Figure 6). It involves only the position variables of the state, as shown in Table I.

The major errors in the observation are in knowledge of the altitude of the sub-satellite terrsin, in tidal heights, and in the instrument itself. The characterization of the terrain uncertainty as a time-correlat first-order Markov process is valid only as a first approximation, since this error is highly orbit dependent.

The radar altimeter information is restricted to the radial direction, as may be seen in Table I. Nence cross-track and down-track errors cannot be removed by this observation.

### 2.4 State Vector; the $\phi$ and Q Matrices

# 2.4.1 Partitioning for Biases

The variables introduced in Section 2.3 from examination of the observations will be included in the state. Table II light those state variables that are not binses; Table III lists the bias variables. The total number of variables is seen to be close to 50, if only one ground radar is used; if a computer program were constructed to calculate the basic covariance equations, Equations (2.1)' and (2.2) as they stand, approximately 2500 computer locations would be required for each of the  $P_{1+1}$ ,  $P_1$ ,  $\phi$  and Q matrices. Moreover, 0.25 × 10<sup>4</sup> operations (multiplies and adds) are required for a single matrix product such as  $\phi P$ . At 10 microseconds per operation, Equation (2.1)' slone would require about 5 seconds of computer time. When six or more ground stations are employed simultaneously, the number of state variables is about 85, each matrix requires about 7200 locations, and Equation (2.1)' requires about 25 seconds. It is, then, clearly impossible to simulate a one or two day mission, in which thousands of observations are made, without consuming large amounts of computer time and space.

It is possible to alleviate considerably the computer storage and time problem by (1) processing only one (scalar) observation at a time and (2) partitioning Equations  $(2, 1)^{t}$  and (2, 2) into sub-matrices for the bias variables.

(1) If the noise covariance R in Equation (2.2) is diagonal, it is possible to process one row of the H matrix at a time. (As pointed out in Reference 4, it is always possible to transform Equation (2.2) so that R is diagonal.) This avoids matrix inversion and saves on storage space for PH<sup>T</sup>. Equation (2.2) is then

$$\mathbf{P}^* = \mathbf{P}^* \leftarrow (\mathbf{P}^* \mathbf{H}_{\mathbf{k}}^{\mathsf{T}}) (\mathbf{P}^* \mathbf{H}_{\mathbf{k}}^{\mathsf{T}})^{\mathsf{T}} / (\mathbf{H}_{\mathbf{k}} \mathbf{P}^* \mathbf{H}_{\mathbf{k}}^{\mathsf{T}} + \mathbf{R}_{\mathbf{k}}) , \qquad (2.19)$$

where  $R_{\rm c}$  is the  $k^{\rm th}$  diagonal element of R and H<sub>c</sub> is the  $k^{\rm th}$  row of H . Equation (2.1)' is still

$$\mathbf{P}_{i}^{*} = \phi(\mathbf{t}_{i}, \mathbf{t}_{i-1}) \mathbf{P}_{i-1}^{*} \phi^{\mathrm{T}}(\mathbf{t}_{i}, \mathbf{t}_{i-1}) + \mathbf{Q}(\mathbf{t}_{i}, \mathbf{t}_{i-1}) .$$
(2.20)

(2) Pure bias states have identity elements in the state transition matrix and [0] in the Q matrix. Partitioning Equation (2.20) gives

$$\phi = \begin{bmatrix} \phi_{a} & 0 \\ 0 & 1 \end{bmatrix} \quad P = \begin{bmatrix} P_{a} & P_{aB} \\ P_{Ba} & P_{B} \end{bmatrix} \quad Q = \begin{bmatrix} Q_{a} & 0 \\ 0 & 0 \end{bmatrix}$$
(2.21)

Here B stands for bias, s for something else.

The equations then become

$$\begin{bmatrix} \mathbf{P}_{\mathbf{g}}(\mathbf{t}^{-}) & \mathbf{P}_{\mathbf{g},\mathbf{g}}(\mathbf{t}^{-}) \\ \mathbf{P}_{\mathbf{g},\mathbf{g}}^{T}(\mathbf{t}^{-}) & \mathbf{P}_{\mathbf{g},\mathbf{g}}(\mathbf{t}^{-}) \end{bmatrix} = \begin{bmatrix} \phi_{\mathbf{g}}\mathbf{P}_{\mathbf{g}}(\mathbf{t}_{0})\phi_{\mathbf{g}}^{T} + \mathbf{Q}_{\mathbf{g}}\phi_{\mathbf{g}}\mathbf{P}_{\mathbf{g},\mathbf{g}}(\mathbf{t}_{0}) \\ \mathbf{P}_{\mathbf{g},\mathbf{g}}^{T}(\mathbf{t}_{0})\phi_{\mathbf{g}}^{T} & \mathbf{P}_{\mathbf{g},\mathbf{g}}(\mathbf{t}_{0}) \end{bmatrix}$$
(2.22)

and

$$\begin{bmatrix} \mathbf{P}_{\mathbf{g}}(\mathbf{t}^{+}) & \mathbf{P}_{\mathbf{g}}(\mathbf{t}^{+}) \\ \mathbf{P}_{\mathbf{g}}^{T}(\mathbf{t}^{+}) & \mathbf{P}_{\mathbf{g}}(\mathbf{t}^{+}) \end{bmatrix} = \begin{bmatrix} \mathbf{P}_{\mathbf{g}}(\mathbf{t}^{-}) & \mathbf{P}_{\mathbf{g}}(\mathbf{t}^{-}) \\ \mathbf{P}_{\mathbf{g}}^{T}(\mathbf{t}^{-}) & \mathbf{P}_{\mathbf{g}}(\mathbf{t}^{-}) \end{bmatrix} = \frac{\mathbf{I}}{\mathbf{D}} \begin{bmatrix} \mathbf{\Delta}^{T} \mathbf{\alpha} & \mathbf{u}^{T} \mathbf{\beta} \\ \mathbf{\beta}^{T} \mathbf{\alpha} & \mathbf{\beta}^{T} \mathbf{\beta} \end{bmatrix},$$
(2.23)

\_\_\_\_\_





the state for the la

where

$$t = H_{a}(t)P_{a}(t^{*}) + H_{a}(t)P_{ab}^{T}(t^{*})$$
(2.24)

$$\beta = H_{g}(t)P_{gg}(t^{-}) + H_{gg}(t^{-})$$

 $D = H_{m}(t)P_{m}(t^{*})H_{m}^{T}(t) + B(t)P_{m}(t^{*})B^{T}(t) + 2H_{m}(t)P_{mm}(t^{*})H_{m}^{T}(t) + B_{k} , \qquad (2.26)$ 

The first of these two equations may be summarized as

C

P <sub>s</sub> (t")	Ξ	$\boldsymbol{\psi}_{\mathrm{H}}(\mathrm{t},\mathrm{t}_{\mathrm{O}})\boldsymbol{\mathrm{P}}_{\mathrm{H}}(\mathrm{t}_{\mathrm{O}}^{\mathrm{A}})\boldsymbol{\phi}_{\mathrm{H}}^{\mathrm{T}}(\mathrm{t},\mathrm{t}_{\mathrm{O}})$	$(\mathbf{b}, \mathbf{t}_0) + \mathbf{q}_{\mathbf{s}}(\mathbf{t}, \mathbf{t}_0)$	(2.27)
P <sub>ss</sub> (t <sup>-</sup> )	E	$\phi_{\rm B}({\rm t},{\rm t}_{\rm 0}){\rm P}_{\rm BB}({\rm t}_{\rm 0}^{\rm +})$	TIME PROPAGATION	(2.28)
P <sub>B</sub> (t <sup>*</sup> )	r	P <sub>B</sub> (t <sup>+</sup> <sub>0</sub> ) .		(2. 29)

The second may be summarized as

$$P_{B}(t^{*}) = P_{B}(t^{*}) - \alpha^{T} \alpha / D$$

$$P_{BB}(t^{*}) = P_{B}(t^{*}) - \alpha^{T} \beta / D$$

$$P_{B}(t^{*}) = P_{B}(t^{*}) - \beta^{T} \beta / D ,$$
(2.30)
(2.31)
(2.32)

where

= 
$$H_{g}(t)P_{g}(t^{-}) + H_{g}(t)P_{gg}^{T}(t^{-})$$
 (2.33)

$$\beta = H_{a}(t)P_{ab}(t') + H_{b}(t)P_{a}(t')$$
(2.34)

$$D = \alpha H_{a}^{T}(t) + \beta H_{a}^{T}(t) + R_{b} , \qquad (2.35)$$

The above time and data equations are forms that may be programmed. The former are used to extrapolate the error matrices  $P_{a}$ ,  $P_{aB}$ ,  $P_{a}$ ,  $P_{a$ 

# 2.4.2 Equations for $\phi_{\rm s}$ and $Q_{\rm s}$

The state transition matrix  $\phi_{\rm g}$  for the error vector of Table II is obtained from the deterministic description of the error process, in the integrated form

$$\mathbf{I}(\mathbf{t}) = \phi_{\mathbf{g}}(\mathbf{t}, \mathbf{t}_{0}) \mathbf{I}(\mathbf{t}_{0}) + \int_{\mathbf{t}_{0}}^{\mathbf{t}} \phi_{\mathbf{g}}(\mathbf{t}, \sigma) \mathbf{f}(\sigma) \, d\sigma \,, \qquad (2.36)$$

where X is the error,  $\phi_{\mu}$  its state transition matrix and  $B(\sigma)$  the driving noise.

ά

The Q matrix is obtained as the covariance of the integral on the right side of the Equation (2.36):

$$\mathbf{Q}_{\mathbf{g}} \simeq \int_{\mathbf{t}_{0}}^{\mathbf{t}} d\lambda \int_{\mathbf{0}_{0}}^{\mathbf{t}} d\mu \,\phi_{\mathbf{g}}(\mathbf{t},\lambda) \,\mathbf{m}[\mathbf{n}(\lambda) \,\mathbf{n}^{\mathrm{T}}(\mu)] \phi_{\mathbf{g}}^{\mathrm{T}}(\mathbf{t},\mu) , \qquad (2,37)$$

where E[] indicates expectation.

In practice, the  $\phi_{\rm B}$  and  ${\bf Q}_{\rm B}$  matrices are themselves partitioned into sub-matrices appropriate to the matural groupings of variables in the state. For the errors shown in Table II, the partitioning of Figures 7 and 8 are appropriate for  $\phi_{\rm B}$  and  ${\bf Q}_{\rm B}$ , respectively. Note that the dynamic biases  $\delta\mu$ ,  $\delta {\bf O}_{\rm N}(1)$ , ...,  $\partial {\bf C}_{\rm N}(4)$ ,  $\delta {\bf D}_{\rm O}/{\bf D}_{\rm B}$  have been retained in  $\phi_{\rm B}$  because they affect position and velocity via  $\phi_{\rm B}$ . The various non-zero sub-matrices of Figures 7 and 8 will now be discussed.

# Vehicle Position and Velocity

These are the error quantities of primary interest in most applications. A closed form for the two-body state transition matrix  $\phi_{\rm RV}$  is given by Kochi<sup>5,6</sup>. A realistic picture of error propagation, however, includes

i**s C**L

**ROMAN** 

(2.25)

A STATE AND A STATE AND A STATE



11 January of 1841

potential  $C_1$ ,  $i = 3, 3, 4, \ldots$ , and, for low altitude satellites, the uncertainties in air drag. A mothod for treating these three classes of uncertainties as biases (constant errors) is given in Appendix B. A more sophisticated treatment for air drag is warranted in many cases<sup>2-11</sup> and extensive models have been developed for the errors in the geopotential<sup>12-16</sup>. The equations of Appendix B, however, are adequate to predict the correct order of sagnitude of drag and potential errors for earth orbits from about 50 to 500 miles altitude. In particular, the comparatively large errors produced by the uncertainty in  $C_g$  can be accurately predicted by these equations.

the effects of uncertainties in the earth's gravitational parameter  $\mu$  , in the zonal harmonics of the geo-

The state transition sub-matrix  $\Phi_D$  may be calculated by numerical integration and filled into  $\Phi_g$  as shown in Figure 7. The portion of  $A_g$  corresponding to position and velocity is zero, as shown in Figure 8, if drag is treated as a bias in  $D_0$ . This portion of  $Q_g$  would be non-zero if a stochastic drag model were used.

# Gyro Errora

The model of gyro drift is taken as a fixed drift rate plus drift rate noise:

$$\theta(\mathbf{t}) = \theta_0 + \omega_0 \tau_{0,0} (\mathbf{1} - \mathbf{e}^{-|(\mathbf{t} - \mathbf{t}_0)/\tau_{0,0}|}) + \omega_{\mathbf{B}} (\mathbf{t} - \mathbf{t}_0) + \int_{\mathbf{t}_0}^{\mathbf{t}} \tau_{0,0} (\mathbf{1} - \mathbf{e}^{-|(\mathbf{t} - \sigma)/\tau_{0,0}|}) \eta_{\omega}(\sigma) \, d\sigma \,, \qquad (2.38)$$

where

- $\theta(t) = total angular error about input axis of gyro$
- ω<sub>o</sub> = initial gyro drift rate
- $\tau_{\rm sp}$  = gyro drift noise time constant
- 71. = uncorrelated drift noise
- $\omega_n$  = gyro drift rate bias.

The gyro drift rate process itself is taken to be white-noise through an RC filter:

$$\omega(\mathbf{t}) = \omega(\mathbf{t}_{0}) \mathbf{e}^{-\left[(\mathbf{u}-\mathbf{t}_{0})/\tau_{0}\mathbf{D}\right]} + \int_{\mathbf{t}_{0}}^{\mathbf{t}_{0}} \mathbf{e}^{-\left[(\mathbf{t}-\sigma)/\tau_{0}\mathbf{D}\right]} \eta_{\omega}(\sigma) \, \mathrm{d}\sigma \, . \tag{2.39}$$

The gyro drifts about each of the three orthogonal IMU axes are assumed to be mutually independent and each follows the model of Equations (2.38) and (2.39). A more refined model involving time-dependent drift rate would be needed for long periods of gyro operation without reset; such a case would occur if reentry or maneuver had to be made long after star tracker failure. The present model, however, is adequate for non-failure mode operation.

When the covariances of  $\theta(t)$  and  $\omega(t)$  are taken for the three gyros, the combined state transition matrix for gyro angles, drift rate noise and drift rate biases is

		li 🛛	0	0	τ <sub>αp</sub> (1 - X)	0	0	t t <sub>o</sub>	0	٦		
		0	1	0	0	$ au_{g_D}(1-X)$	0	0	$\mathbf{t} - \mathbf{t}_{o}$	0		
		0	0	1	i	0	$\tau_{qp}(1-X)$	0		t-t <sub>o</sub>		
		0	0	0	x.	0	0	0	0	0		
¢ <sub>ed</sub>	Ŧ	0	0	0	0	x	0	,   0	σ	0	•	(2.40
		0	o	0	0	0	x	0	0	0		
		0	0	Û,	0	0	0	i i	0	0		
		0	0	0	0	0	o	0	1	0		
		0	0	0	   0	0	0	10	Ø	1		

where  $X = e^{-|(t-t_0)/\tau_{0D}|}$ .

A. A.

The covariance matrix of the accumulated noise is

		• •	
ο κοί ο κο	0	0 0	
O O A O O B	0	0 0	
B O O C O N	0	0 0	
G <sub>αμ</sub> = 0 в 0 0 С G	0	<b>0</b> 0	. (2.41
оовоос	1   0	0 0	
	0	0 0	
0 0 0 0 0 0	0	<b>0</b> 0	
م ه ه م م	0	ه و	

where

 $A = \sigma_{0D}^{2} \tau_{0D} [2(t - t_{0}) + \tau_{0D}(1 - X^{2}) - 4\tau_{0D}(1 - X)]$   $B = \sigma_{0D}^{2} \tau_{0D} (1 - X)^{2}$  $O = \sigma_{0D}^{2} (1 - X^{2}) \cdot$ 

and

# $\sigma_{\rm do}^{\rm s}$ = variance of gyro drift noise

# Horison Deviations

The combined horizon deviations  $\tilde{h}_1$ ,  $\tilde{h}_2$ ,  $\tilde{h}_3$  are dependent on the actual horizon deviations  $h_1$ ,  $h_2$ ,  $h_3$ ,  $h_4$  at the points of tangency of the telescopic lines of sight and the sensed horizon. For a reasonable approximation, one may assume<sup>1</sup> that  $h_1$ ,  $h_2$ ,  $h_3$ ,  $h_4$  are correlated in time and location with an autocorrelation function

$$CF(\tau, \delta) = \sigma_{HD}^2 e^{-|\tau/\tau_H| - |\delta/\delta_H|}$$
, (2.42)

where  $\tau$  is the time between measurements of deviations h in height of horizon points separated by a distance  $\delta$ ,  $\tau_{\rm H}$  and  $\delta_{\rm H}$  are constants determined by observation. Here  $\sigma_{\rm HD}^2$  is the variance of h<sub>1</sub>, h<sub>2</sub>, h<sub>3</sub>, or h<sub>4</sub>. This approximation is applicable to the CO<sub>2</sub> 14-16 micron spectral band. Using this autocorrelation twotion, and taking into account the vehicle velocity but ignoring the correlations due to the adjacency of successive orbits, Wildox<sup>18</sup> has calculated the variance of the position error along the vertical axis  $\frac{2}{3}$  as

$$\frac{R^2}{R_{\rm g}^2} \frac{(1+2a+b)}{4} \sigma_{\rm HD}^2 , \qquad (2.43)$$

and the variance of the position errors along the horizontal axes  $\,\,\hat{\eta}\,$  and  $\,\,\hat{\zeta}\,$  as

$$\frac{R^{2}(1-b)}{(R^{2}-R_{g}^{2})^{2}}\sigma_{HD}^{2}, \qquad (2.44)$$

where

-(1/#H) 008-1 (Rg/R)#

L = radius of earth

$$\sigma_{\rm HD}^2$$
 = variance of h<sub>1</sub>, h<sub>2</sub>, h<sub>3</sub> or h<sub>3</sub>.

In terms of the errors  $\delta \Phi_1$  ,  $\delta \Phi_2$  ,  $\delta \Phi_3$  in the combined tracker observations, the vertical position error is





раналаранан калараларан калараларан 1995 - Аларан Каларанан каларан каларан 1995 - Фенераларан Каларанан каларан калара and the horizontal position errors are

R54 . (2.46

Combining Equations (2.45), (2.43) and (2.16) gives the variance of  $~{\rm \tilde{h}}_i$  :

$$\operatorname{var}(\tilde{h}_{1}) = \sigma_{HD1}^{2} = \frac{1+2a+b}{4}\sigma_{HD}^{2}$$
, (2.47)

and the variances of  $\tilde{h}_{2}$  ,  $\tilde{h}_{3}$  :

$$\operatorname{var}(\tilde{h}_2) = \sigma_{HD2}^2 = \frac{(1-b)}{2} \sigma_{HD}^2$$
 (2.48)

$$\operatorname{var}(\tilde{h}_{g}) = \sigma_{HU_{3}}^{2} = \frac{(1-b)}{2} \sigma_{HD}^{2}$$
 (2.49)

The three combined horizon deviations may thus be considered as the outputs of three first-order linear systems driven by three independent white-noise sources. The model for each is

$$H(t) = H(t_0)e^{-|(t-t_0)/\tau_H|} + \int_{t_0}^{t} e^{-|(t-\sigma)/\tau_H|} \eta(\sigma) d\sigma . \qquad (2.50)$$

The  $\phi$  and Q for the three deviations are .

$$\phi_{\rm ND} = \begin{bmatrix} e^{-|(t-t_0)/\tau_{\rm HD}|} & 0 & 0 \\ 0 & e^{-|(t-t_0)/\tau_{\rm HD}|} & 0 \\ 0 & 0 & e^{-|(t-t_0)/\tau_{\rm HD}|} \end{bmatrix}, \quad (2.51)$$

$$\Theta_{\rm ND} = \begin{bmatrix} \sigma_{\rm HD1}^2 (1-y^2) & 0 & 0 \\ 0 & \sigma_{\rm HD2}^2 (1-y^2) & 0 \\ 0 & 0 & \sigma_{\rm HD2}^2 (1-y^2) \end{bmatrix}, \quad (2.52)$$

where

5.24.2

$$= e^{-i(t-t_0)/\tau_{HD}}$$

and, as shown in Reference 18,

$$\tau_{\rm HD} = \tau_{\rm H} \delta_{\rm H} / (\delta_{\rm H} + 2\pi \tau_{\rm H} / T_{\rm p}) , \qquad (2.54)$$

where  $T_0$  is the orbital period of the satellite.

# Horison Tracker

The combined instrument errors  $\tilde{\phi}_i$ ,  $\tilde{\phi}_j$ ,  $\tilde{\phi}_j$ , are illustrated in Figure 4. The deterministic model for each is that of white-noise through an RC filter, the three noise components being assumed uncorrelated with each other. The model for each error is

ễ(t)	z	$\vec{\phi}(t_0) e^{-\left  (t-t_0)/\tau_{\rm HT} \right } + \int_{t_0}^t e^{-\left  t - t_0 \right } dt$	- (t-σ)/THT ηφ(σ) dσ		(2.55)
ф <sub>ит</sub>	=	[•- (t-t <sub>0</sub> )/* <sub>HT</sub>   0 0	0 e- (t-t <sub>g</sub> )/+ <sub>HT</sub>   0	0 0 e <sup>- (t-t</sup> o)/*HT	(2. 56)
9 <sub>HT</sub>	Ħ	σ <sub>HT1</sub> (1-e <sup>-2 (t-t<sub>0</sub>)/r<sub>HT</sub> ) 0 0</sup>	0 0 HT2(1-e <sup>-2 (t-t<sub>0</sub>)/<sub>"MT</sub> ) 0</sup>	$ \begin{bmatrix} 0 \\ 0 \\ \sigma_{\rm HT2}^2 (1 - e^{-2   (t - t_0) / \tau_{\rm HT}  } ) \end{bmatrix} , $	(2.57)





where

- <b>1</b> HT 1		variance of	$\tilde{\phi}_{i}$
0 <sup>2</sup> HT 2	Ŧ	variance of	<i>ф</i> ,
с. НТ 1	=	variance of	õ,

 $\tau_{\rm HT}$  = time constant for the horizon tracker errors.

If a four-telescope tracker is employed, as described in Section 2.3,

$$\sigma_{\rm HT1}^2 = \sigma_{\rm HT}^2/4 \tag{2.58}$$

$$\sigma_{\rm HT2}^2 = \sigma_{\rm HT3}^2 = \sigma_{\rm HT}^2/2 , \qquad (2.59)$$

where  $\sigma_{\rm HT}^2$  is the actual variance of a single telescope angle.

# Star Tracker

The time correlated errors in the star tracker train and elevation angle are shown in Figure 5. The error process for each is assumed to be white-noise through an RC filter, the two components being uncorrelated with each other:

$$\Omega(t) = \Omega(t_0) e^{\tau |(t-t_0)/\tau_{BT}|} + \int_{t_0}^{t} e^{\tau |(t-\sigma)/\tau_{BT}|} \eta_{BT}(\sigma) d\sigma . \qquad (2.60)$$

The state transition matrix and noise matrix are

$$\varphi_{\rm sr} = \begin{bmatrix} e^{-|(t-t_0)/\tau_{\rm sr}|} & 0\\ 0 & e^{-|(t-t_0)/\tau_{\rm sr}|} \end{bmatrix}$$
(2.61)

$$\mathbf{P}_{\mathbf{ST}} = \begin{bmatrix} \sigma_{\mathbf{ST}1}^{2} (1 - e^{-2 | (t - t_{0})/\tau_{0.1}}) & 0 \\ 0 & \sigma_{\mathbf{ST}2}^{2} (1 - e^{-2 | (t - t_{0})/\tau_{\mathbf{ST}}}) \end{bmatrix} & (2.82) \end{bmatrix}$$

where ,

 $\sigma_{\rm ST1}^{\rm g}$  = variance of star tracker train angle error

 $\sigma_{\rm STR}^2$  = variance of star tracker elevation angle error

 $\tau_{\rm BT}$  = correlation time of star tracker errors.

# Landmark Sensor

The time correlated noises in the landmark sensor train angle and elevation angle (Fig. 5), are state variables. The error process for each is assumed to be white noise through an RC filter, the two components being uncorrelated with each other:

$$\pi(t) \simeq \pi(t_0) e^{-|(t-t_0)/\tau_{LS}|} + \int_{t_0}^{t_0} e^{-|(t-\sigma)/\tau_{LS}|} \eta_{LS}(\sigma) d\sigma , \qquad (2.63)^{-1}$$

The state transition matrix and noise matrix are

$$\phi_{LS} = \begin{bmatrix} e^{-|(t-t_0)/\tau_{LS}|} & 0 \\ 0 & e^{-|(t-t_0)/\tau_{LS}|} \end{bmatrix}$$

$$Q_{LS} = \begin{bmatrix} \sigma_{LS1}^2 (1-e^{-2|(t-t_0)/\tau_{LS}|}) & 0 \\ 0 & \sigma_{LS2}^2 (1-e^{-2|(t-t_0)/\tau_{LS}|}) \end{bmatrix}$$
(2.65)

MAN AND

where

 $\sigma_{\text{Let}}^{\mathbf{F}} = \text{variance of landmark sensor train angle}$ 

 $\sigma_{\text{Las}}^{\text{B}}$  = variance of landmark sensor elevation angle

 $\tau_{\rm L, g}$  = correlation time of landmark sensor errors.

The geometry chosen for landmark sensor and star tracker is not ideal from a mechanization point of view, since it requires infinite train rates when the star or landmark is along the vertical. Nevertheless, it was chosen because it is simpler to compute and because it should give results representative of the reveral practical configurations porsible.

### Gravity, Oblateness and Drag Blases

The state transition matrix relating fixed uncertainties in the gravity, oblateness and drag parameters to position and velocity is given by  $\phi_0$ , described in Appendix B. This matrix is obtained most conveniently by pumerical integration of the differential equations. The state transition matrix relating the biases at time  $t_0$  to the same biases at time t is just the identity matrix I. The  $Q_0$  matrix for these biases is zero.

# 2.5 Interproting the Covariance Matrix

### 2.5.1 Yehicle Position Error Covariances

The vehicle position error covariances contained in P<sub>2</sub> are in inertial coordinates. They may be converted to local vertical coordinates. If T indicates transpose,

$$\operatorname{cov}\{\mathbf{X}_{\mathsf{LV}}\mathbf{X}_{\mathsf{LV}}^{\mathsf{T}}\} = \operatorname{cov}\{(\mathbf{S}^{\mathsf{T}}\mathbf{X}_{\mathsf{T}}) \cdot (\mathbf{S}^{\mathsf{T}}\mathbf{X}_{\mathsf{T}})^{\mathsf{T}}\} = \mathbf{S}^{\mathsf{T}}\operatorname{cov}\{\mathbf{X}_{\mathsf{T}}\mathbf{X}_{\mathsf{T}}^{\mathsf{T}}\}\mathbf{S}$$

 $= S^{T} P_{s(1-3)} S_{1}$  (2.66)

where  $X_{LV}$  is the position error vector in local vertical coordinates,  $X_{\chi}$  the same vector in inertial coordinates,  $P_{K(1-3)}$  is the upper left  $3 \times 3$  of  $P_{K}$ , and S is an orthogonal transformation from local vertical to inertial coordinates:

8,11	æ	$\cos \Psi \cos \Omega - \sin \Psi \sin \Omega \cos i$	8	oor B	(2.67)
8 <sub>12</sub>	æ	$-\sin\Psi\cos\Omega-\cos\Psi\sin\Omega\cosi$	¥	-sin $\beta$	(2.68)
8 <sub>13</sub>	×	sipΩ sin i	×	0	(2.69)
8 <sub>tl</sub>	7	cosΨ sinΩ ≻ sinΨ cosΩ cos i	3	sin $\beta$	(2.70)
8,,	z	$-\sin\Psi\sin\Omega + \cos\Psi\cos\Omega\cos i$	T	cos ß	(2.71)
s <sub>t1</sub>	×	-sini comΩ	×	0	(2.72)
8 <sub>31</sub>	z	sin¥ sin i	Ħ	0	(2.73)
8,,,	x	cos T sin i	=	0	(2.74)
n.,	z	cos i	Ŧ	'n, .	(2.75)

where  $\Psi$  and i, given in Figure 9(a) are used if the inclination is not zero, and  $\beta$ , given in Figure 9(b), is used if the inclination is zero.

# 2.5.2 Vehicle Velocity Error Covariances

The velocity error covariances contained in  $P_s$  may be transformed from inertial to local vertical coordinates. The computation is

$$\operatorname{ov}(\overline{V}_{LV}\overline{V}_{LV}^{T}) = \operatorname{ov}((S^{T}V_{T})(SV_{T})^{T}) = S^{T} \operatorname{oov}(\overline{V}_{T}\overline{V}_{T}^{T})S$$
$$= S^{T}P_{B(N-6)}S, \qquad (2.76)$$

يسترين والمسترين والمسترين والمرارية والمناج والمتعاد والمتعام والمتعارية والمتعارية والمتعارية والمتعارية والمتعادية

where  $\overline{V}_{LV}$  is the velocity error in local vertical coordinates,  $\overline{V}_{I}$  is the velocity error in institute coordinates and  $P_{B(1-4)}$  is  $P_{B(I,J)}$  for i = 4, 5, 6 and J = 4.5, 6. The matrix B is that given above for position covariances.









2000 X 100

# 2.5.3 Gyro Misalignment Covariances

The elements of  $P_{g(I,J)}$  for I,J = 7,8,9 are the covariances of the gyro misalignments along platform axes 1,2,3. If the platform axes coincide with the local vertical no transformation is needed. If the platform axes coincide with the inertial coordinate frame a transformation to local vertical axes is needed. The calculation is as follows:

$$dov(\vec{\partial}_{LV}\vec{\partial}_{LV}^{T}) = dov((\mathbf{s}^{T}\vec{\partial}_{I})(\mathbf{s}^{T}\vec{\partial}_{I})^{T})$$
$$= \mathbf{s}^{T}\mathbf{P}_{\mathbf{s}}(\mathbf{1},\mathbf{s},\mathbf{s})\mathbf{s} . \qquad (2.77)$$

where S is as above,  $P_{s(1,0,0)}$  is  $P_{s(1,1)}$  for I, J = 7, 8, 9 and  $\theta$  is the vector gyro misalignment.

# 2.5.4 Vehicle Attitude Covariances

The vehicle stitude error  $\mathcal{P}$  is the sum of gyrc misslignment  $\theta$  and the rotation error  $\epsilon$  due to position and velocity errors:

$$\vec{\gamma} = \vec{\theta} + \vec{\epsilon}$$
 (2.78)

$$\simeq \Theta + \left(\frac{\mathbf{\vec{R}} \times \delta \mathbf{\vec{R}}}{\mathbf{R}^2} + \frac{\mathbf{\vec{\xi}} \cdot \delta \mathbf{\vec{V}}}{\eta \cdot \mathbf{v}} \mathbf{\vec{\xi}}\right) , \qquad (2.79)$$

1.1.1

where  $\hat{\xi}\hat{\gamma}\hat{\zeta}$  are the local vertical unit vectors of Figure 2 and  $\delta K$ ,  $\delta V$  are position and velocity error. Factoring out SR and SV gives

$$\vec{\gamma} = \vec{\theta} + \left[ \left[ \vec{\mathbf{K}} \mathbf{x} \right] \frac{1}{\mathbf{R}^2} + \frac{\vec{\mathbf{K}} \times \vec{\mathbf{V}}}{\left( \vec{\mathbf{K}} \times \vec{\mathbf{V}} \right)^2} \vec{\mathbf{K}} \right] \cdot \left\{ \begin{array}{l} \delta \vec{\mathbf{K}} \\ \delta \vec{\mathbf{V}} \end{array} \right\}$$
$$= \vec{\theta} + \left[ \theta_A \right] \cdot \vec{\mathbf{x}} , \quad \text{where } \vec{\mathbf{x}} = \left\{ \begin{array}{l} \delta \vec{\mathbf{K}} \\ \delta \vec{\mathbf{V}} \end{array} \right\}_{(\text{INERTIAL})}$$
(2.80)

and  $[\theta_i]$  is known in terms of  $\mathbb{R}$  and  $\nabla$ , in inertial coordinates. If  $\vec{\theta}$  is in a local vertical system, the attivude covariance, in local vertical coordinates, is

$$\operatorname{cov}(\widetilde{\gamma}_{LV}\widetilde{\gamma}_{LV}) = \operatorname{cov}\{(\widetilde{\mathbf{s}}^{\mathsf{T}} \widetilde{\mathbf{d}}_{1} + \mathbf{s}^{\mathsf{T}} [\theta_{A}] \widetilde{\mathbf{x}}) (\widetilde{\mathbf{s}}^{\mathsf{T}} \widetilde{\mathbf{d}}_{1} + \mathbf{s}^{\mathsf{T}} [\theta_{A}] \widetilde{\mathbf{x}})^{\mathsf{T}}\}$$

$$= \operatorname{cov}\{\widetilde{\mathbf{s}}^{\mathsf{T}} (\widetilde{\mathbf{d}}_{1} + [\theta_{A}] \widetilde{\mathbf{x}}) (\widetilde{\mathbf{d}}_{1}^{\mathsf{T}} + \widetilde{\mathbf{x}}^{\mathsf{T}} [\theta_{A}]^{\mathsf{T}}) \widetilde{\mathbf{s}}\}$$

$$= \widetilde{\mathbf{s}}^{\mathsf{T}} \operatorname{cov}\{\widetilde{\theta}_{1} \widetilde{\theta}_{1} + (\theta_{A}] \widetilde{\mathbf{x}} \widetilde{\theta}_{1}^{\mathsf{T}} + \widetilde{\theta}_{1} \widetilde{\mathbf{x}}^{\mathsf{T}} [\theta_{A}]^{\mathsf{T}} + (\theta_{A}] \widetilde{\mathbf{x}} \widetilde{\mathbf{x}}^{\mathsf{T}} [\theta_{A}]^{\mathsf{T}}) \widetilde{\mathbf{s}}\}$$

$$= \widetilde{\mathbf{s}}^{\mathsf{T}} \operatorname{cov}\{\widetilde{\theta}_{1} \widetilde{\theta}_{1} + (\theta_{A}] \widetilde{\mathbf{x}} \widetilde{\theta}_{1}^{\mathsf{T}} + \widetilde{\theta}_{1} \widetilde{\mathbf{x}}^{\mathsf{T}} [\theta_{A}]^{\mathsf{T}} + (\theta_{A}] \widetilde{\mathbf{x}} \widetilde{\mathbf{x}}^{\mathsf{T}} [\theta_{A}]^{\mathsf{T}}) \widetilde{\mathbf{s}}$$

$$= \widetilde{\mathbf{s}}^{\mathsf{T}} \left[ \mathbf{P}_{\mathbf{s}}(\theta\theta) + [\theta_{A}] \mathbf{P}_{\mathbf{s}}(\mathfrak{x}\theta) + \mathbf{P}_{\mathbf{s}}(\theta\mathfrak{x}) [\theta_{A}]^{\mathsf{T}} \right] + \left[ \theta_{A} \mathbf{P}_{\mathbf{s}}(\mathfrak{x}) [\theta_{A}]^{\mathsf{T}} \right] \widetilde{\mathbf{s}} , \qquad (2.81)$$

where

$$\begin{split} P_{s}(\gamma\gamma) &= P_{s}(I,J) & \text{for} & I = 7-9 , J = 7-9 \\ P_{s}(\chi\gamma) &= P_{s}(I,J) & \text{for} & I = 1-6 , J = 7-9 \\ P_{s}(\gamma\chi) &= P_{s}(I,J) & \text{for} & I = 7-9 , J = 1-6 \\ P_{s}(\chi\chi) &= P_{s}(I,J) & \text{for} & I = 1-6 , J = 1-6 \\ \end{split}$$

If  $\theta$  is in an inertial frame, the attitude covariance in local vertical coordinates is

$$\operatorname{cov}\{\vec{\gamma}_{\mathsf{LV}}\vec{\gamma}_{\mathsf{LV}}^{\mathsf{T}}\} = \operatorname{cov}\{\{\vec{\theta}_{\mathsf{LV}} + \mathbf{S}^{\mathsf{T}}[\theta_{\mathsf{A}}]\,\mathbf{X}\}\{\vec{\theta}_{\mathsf{LV}} + \mathbf{S}^{\mathsf{T}}[\theta_{\mathsf{A}}]\,\mathbf{X}\}^{\mathsf{T}}\}$$

$$= \operatorname{cov}\{\overline{\theta}_{LV} \overline{t}_{LV}^{T}\} + \mathbf{S}^{T}[\overline{\theta}_{A}] \operatorname{cov}\{\overline{\mathbf{X}} \overline{\theta}_{LV}^{T}\} + \operatorname{cov}\{\overline{\theta}_{LV} \overline{\mathbf{X}}^{T}\} [\overline{\theta}_{A}]^{T} \mathbf{S} + \mathbf{S}^{T}[\overline{\theta}_{A}] \operatorname{cov}\{\overline{\mathbf{X}} \overline{\mathbf{X}}^{T}\} [\overline{\theta}_{A}]^{T} \mathbf{S}$$

$$= P_{\mathbf{s}(\gamma\gamma)} + \mathbf{s}^{\mathrm{T}}[\theta_{\mathrm{A}}] P_{\mathbf{s}(\chi\gamma)} + P_{\mathbf{s}(\gamma\chi)}[\theta_{\mathrm{A}}]^{\mathrm{T}} \mathbf{S} + \mathbf{s}^{\mathrm{T}}[\theta_{\mathrm{A}}] P_{\mathbf{s}(\chi\chi)}[\theta_{\mathrm{A}}]^{\mathrm{T}} \mathbf{S} .$$
(2.82)

The remainder of the covariance matrix is readily interpreted and needs no transformation.

2.6 Determining the Times of the Observations

The logic and search procedures that must be programmed to simulate realistically the times at which observations are made may constitute almost half of the computer program coding.
## 2.6.1 Ground Based Radar

Most radar observations taken with the antenna at a low elevation angle are unreliable and are discurded. As a result any one radar  $c_{m}$  take readings only when the vehicle is within a cone extending out from the radar station and with its axis along the local vertical of the station. Typically, the half-cone angle is between  $80^{\circ}$  and  $85^{\circ}$ . The maximum range of the radar puts a spherical cap on this cone of visibility. A search procedure must determine, for each radar in the tracking not, whether the vehicle falls within this care of visibility. The rotation of the earth complicates the picture in a minor way. A variable search step is found to be indispensible, in many cases, to avoid excessive computation time.

#### 2.6.2 Horizon Tracker

This instrument is limited in operation only by the power and data processing capacity of the vehicle.

#### 2.6.3 Star Tracker

The major considerations in determining the times of observation for a star tracker are (a) stellar magnitude, (b) obscutation of the star by the earth, and (c) adequate angular separation of stars.

#### 2.6.4 Landwark Sensor

The considerations here are very similar to those for the ground-based radur observations. Indeed, the same search procedure (coding) may be used for lendmarks as for radars, with suitably different parameters.

## 2.6.5 Radar Altimeter

It is usually desired to investigate the effect of operating the altimeter only over water, since the tidal uncertainty is such less than the terrain uncertainty and easier to determine. Storing a facsimile of the landsea configuration may be avoided, for error analysis purposes, by allowing altimetric readings for a fraction of the time corresponding to the percentage of water underlying the orbit.

#### 8.7 Results of Typical Computer Suns

The results of a typical computer program for on-board measurements along are recorded and discussed in Reference 18. The instruments employed are horizon tracker, star tracker, radar alimeter. The computer runs indicate, among other things, the oscillatory nature of cross-track error, the value of a radar altimeter in sustracting the monumentation of down-track errors, and the futility of increasing the horizon tracker data rate beyond the limit set by the horizon deviation time constant.

The results of a typical computer program that combines both on-board and ground-based data are shown in Figure 10. This computer run assumed three radar stations tracking a satellite in a 250 nautical miles circular orbit. Ground radar range readings were taken at 10, 22 and 24 hours. They have a bias of 35 ft, white noise of 10 ft, station location biases of  $\Delta e = \Delta_n = 75$  ft, and  $\Delta h = 25$  ft. The on-board horizon tracker was assumed to have a bias of 0.015 deg and a noise variance of 0.25 deg<sup>2</sup> with a 1 second time constant. The horizon deviations were assumed to have  $R_g \delta_H = 2500$  nautical miles, and a variance  $\sigma_{HD}^2 = 1.4$  (nautical miles)<sup>2</sup>. A star tracker with 0.16 deg bias, 1 second time constant and  $4 \times 10^{-6} (deg)^2$  variance was assumed, along with a radar altimeter of 0.01 nautical mile white-noise and 0.005 nautical mile bias. These results are not realistic beyond about one day of orbital time, because no drag or uncertainties in the gravity model were included.

When drag bias  $\delta D_0/D_0$  and gravitational bias  $\delta \mu/\mu$ , but not biases in  $C_n$ , are accounted for, an error analysis of ground radar tracking yields results such as shown in Figure 11. The radar location biases of 200 ft in each of  $\Delta e$ ,  $\Delta n$ ,  $\Delta h$  were estimated during the run and at 13.2 hours were, on the avarage, 141 ft. Radar range readings were taken at times indicated by a solid triangle in Figure 11. The program also attempted to estimate  $\delta \mu$ ,  $\delta C_u$ ,  $\delta C_u$ ,  $\delta C_d$ , and  $\delta C_d$ , but no significant changes in their covariances were observed because of the projection of the run.

## 3. BELF-CONTAINED ALIGNMENT OF AN INERTIAL QUIDANCE SYSTEM

#### 3.1 Description of Problem

As the accuracy and sensitivity of gyroscopes and accelerometers increase, so will the importance of selfcontained alignment of inertial navigators. This section describes the application of the Kalman filter to the problem of estimating the orientation of an inertial mensor coordinate frame using only the signals from the gyroscopes and accelerometers. The vehicle in which the instruments are located is undergoing random. But limited, translations and rotations, such as a launch vehicle before lift-off, or an aircraft being loaded preparatory to take-off. The procedure to be described applies both to sensors stabilized by means of ginbals and to sensors mounted directly on the vehicle body. For simplicity it is assumed that one gyro input axis and one accelerometer input axis is along each of three mutually orthogonal directions. These directions define the sensor coordinate frame.

At the time that the fine alignment procedure commences, the sensor coordinate frame will have a small vector misalignment & relative to some intended inertially-fixed alignment frame (see Figure 12). The vector & is

a random variable with known mean and covariance at  $t_0$ , determined by the errors in some coarse alignment scheme that has been completed before  $t_0$  and that need not concern us. The orientation of the intended alignment frame is accurately known in inertial space; changes in orientation of the someor frame relative to its actual alignment at  $t_0$  are also accurately known (in the case of body-mounted semeors, because of continuous calculation of the direction cosine matrix, and, in the case of a gimballed system, because of procisely known torques applied to the gyros). The purpose of the Kalman filter then is to estimate the initial misaligne-int vector  $\mathbf{v}$  from the accelerometer readings. The data are the vectors  $\tilde{\mathbf{z}}(t_1)$  ( $\mathbf{i} = 1, 2, \dots, N$ ):

$$\overline{\mathcal{R}}(t_i) = -\overline{\mathbf{r}}(t_i) - \overline{\mathbf{r}}(t_i) \times \overline{\mathbf{r}} + \overline{\mathbf{v}}(t_i), \qquad (3.1)$$

where  $g(t_i)$  is the local gravity vector at the alignment site at time  $t_i > t_0$  and  $\nabla(t_i)$  is the sum of vehicle acceleration and accelerometer noise. In order to simplify the filter equations,  $\nabla(t_i)$  is taken to be zero mean noise. It should be noted that  $g(t_i)$  is known in any given inertial coordinate system because the geodetic latitude, longitude and altitude of the alignment site are precisely known for all i,  $t_0 \leq t \leq t_n$ .

## 3.9 Recursive Filter for the Problem

If one denotes by  $\theta_1^-$  and  $\theta_1^+$  the filter's estimate of  $\theta_1^-$  just before and just after the incorporation of the data incorporation equations are

$$\hat{s}_{i}^{+} = \hat{s}_{i}^{-} + K_{i} [\bar{z}(t_{i}) - \hat{z}(t_{i})]$$
(3.2)

$$\mathbf{i}^* = \mathbf{F}_1^* - \mathbf{K}_1 \mathbf{H}_1 \mathbf{F}_1^* \,, \qquad (3.3)$$

where

R. C. Starte

$$= \mathbf{P}_{i}^{\mathbf{H}} \mathbf{I}_{i}^{\mathbf{T}} \mathbf{P}_{i}^{\mathbf{T}} \mathbf{H}_{i}^{\mathbf{T}} + \mathbf{R}_{i}^{\mathbf{T}} \mathbf{I}^{\mathbf{T}}$$

$$(3.4)$$

= Kalman filter matrix at time  $t_i$  ,

$$\hat{\mathbf{z}}(\mathbf{t}_{i}) = -\mathbf{\ddot{s}}(\mathbf{t}_{i}) - \mathbf{\ddot{s}}(\mathbf{t}_{i}) \times \hat{\mathbf{e}}_{i}$$

- = estimated observation at  $t_i$
- $H_{i} = [\underline{a}(t_{i}) \times]$ (3.6)
  - = antisymmetric matrix corresponding to
    oross product by \$(t\_1)

## $R_i \simeq \text{covariance of } \overline{V}(t_i)$ -

 $= \mathbf{E}(\nabla(t_i)\nabla(t_i)^T)$ 

The time propagation equations are particularly simple in inertial space:

K.

$$\hat{\theta}_{1}^{*} = \hat{\theta}_{1-1}^{*}; \quad \hat{\theta}_{0}^{*} = 0 \tag{3.8}$$

$$P_1^{-} = P_{1+1}^{+}; \quad P_0^{+} = \operatorname{cov}(\hat{s}_0^{+}).$$
 (3.9)

The observation matrix H , however, in time-varying if viewed in an inertial frame, because **E**(t) is rotating at the earth's rate:

$$\mathbf{R}(\mathbf{t}_{i}) = [\theta(\mathbf{t}_{i}, \mathbf{t}_{0})] \mathbf{\ddot{x}}(\mathbf{t}_{0}), \qquad (3.10)$$

where  $[\theta(t_i, t_0)]$  is a matrix representing the earth's rotation from  $t_0$  to  $t_i$ . To be specific, let us express all vectors and matrices in an inertial coordinate system having one axis  $(u_2)$  along the earth's spin axis and the other two pointing to points fixed on the cellstial equator (Fig. 13). In this coordinate frame, with  $u_1$ directed to the vernal equinox, one has

 $\left[\theta(t_1, t_0)\right]_{u_1 u_2 u_3} = \begin{bmatrix} \cos \theta & -\sin \theta & 0\\ \sin \theta & \cos \theta & 0\\ 0 & 0 & 1 \end{bmatrix}$ (3.11)



(3.5)

(3.7)

Reserved Divide a metric Divide by

(3.13)

(3.14)

$$\begin{bmatrix} \mathbf{g}(\mathbf{t}_{1}) \times ]_{\mathbf{u}_{1}\mathbf{u}_{2}\mathbf{u}_{3}} = \begin{bmatrix} \mathbf{0} & -\mathbf{z}_{3} & \mathbf{z}_{2} \\ \mathbf{z}_{3} & \mathbf{0} & -\mathbf{z}_{1} \\ -\mathbf{z}_{2} & \mathbf{z}_{1} & \mathbf{0} \end{bmatrix} = \mathbf{H}_{1} , \qquad (3.12)$$

$$= (\mathbf{t}_{i} - \mathbf{t}_{i}) |\vec{\Omega}_{\mathbf{g}}|$$

 $\overline{\Omega}_{\mathbf{g}}$  = magnitude of earth's spin

$$\mathbf{s}_i = -|\mathbf{\tilde{s}}| \sin \lambda$$

θ

$$\mathbf{g}_{g} = -|\mathbf{g}| \cos \lambda \cos \mathbf{u}^{T}$$
 (3.10)

$$\mathbf{s}_{3} = -|\mathbf{\tilde{g}}| \cos \lambda \sin \mathbf{L}^{*}$$
(3.16)

|2| = magnitude of local gravitational acceleration

## $\lambda$ = local geodetic latitude

L\* = local hour angle of Aries.

The selection of state variables in this alignment problem results in a time-varying observation matrix, but trivial time propagation equations for the state and its covariance matrix. In general, such a compromise is desirable in applying the Kalsan filter because

- (a) the covariance time propagation equation is a matrix equation, requiring integration or the calculation of a state transition matrix, .
- (b) the observation matrix H is calculated only when data is received, and need not be stored between data times, and
- (c) no numerical inaccuracy is introduced in propagating the state or its covariance between data times.

## 3.3 Batched Filter for the Problem

The recursive nature of the Kalman filter lends itself to an alignment procedure that may have to be terminated at any time. Such is the case when a quick reaction capability is needed, as in reacue launches or when the usual alignment procedure has failed a short time before launch. At the other extrome, the operation of a recursive filter over an extended period of time can result in numerical, and geometric difficulties<sup>19-21</sup>, adequately discussed elsewhere in this publication, that make its continued use of dubious value.

If adequate time is available to collect, say. N pieces of data and if it is destinear these data all at once after the last is received, batch processing formulas are more appropriate than recursive ones. In that case one may ascemble the data  $Z(t_i)$ , i = 1, 2, ..., N, into a single data vector

$$Z_{N,1} = \begin{bmatrix} Z(t_1) \\ Z(t_2) \\ \vdots \\ Z(t_N) \end{bmatrix}, \qquad (3.17)$$

calculate the estimated observations, using only the estimated state at  $t_n$  ,

 $\hat{Z}_{N,1} = \begin{bmatrix} -\bar{g}(t_1) - \bar{g}(t_1) \times \hat{e}_0^+ \\ -\bar{g}(t_2) - \bar{g}(t_2) \times \hat{e}_0^+ \\ \vdots \\ -\bar{g}(t_N) - \bar{g}(t_N) \times \hat{e}_0^+ \end{bmatrix}, \quad (N.18)$ 

and apply the Kalman Filter equations in the form

 $\begin{aligned} \hat{\theta}_{N}^{+} &= \hat{\theta}_{0}^{+} + P_{N}^{+} H_{N,1}^{T} R_{N,1}^{-1} (Z_{N,1} - \hat{Z}_{N,1}) \end{aligned}$ (3.19)  $P_{N}^{+} &= \left[ (P_{0}^{+})^{-1} + H_{N,1}^{T} R_{N,1}^{-1} H_{N,1}^{-1} \right]^{-1} , \end{aligned}$ (3.20)



where

A ROAD AND A

の見たいでは、「日本の」の目的では、「日本の」ので、「日本の」ので、「日本の」ので、「日本の」ので、「日本の」ので、「日本の」ので、「日本の」ので、「日本の」ので、「日本の」ので、「日本の」ので、「日本

$$H_{H_1}^{\dagger} = [H_1^{\dagger}, H_2^{\dagger}, \dots, H_N^{\dagger}]$$
 (3.21)

When the matrix multiplications are carried out it is found that no more than 3×3 matrices are involved for

$$H_{N,1}^{T}(\mathbf{Z}_{N,1} - \hat{\mathbf{Z}}_{N,1}) = \sum_{i=1}^{N} H_{i}^{T} R_{i}^{-1} [\mathbf{Z}(t_{1}) + \hat{\mathbf{z}}(t_{1}) + \hat{\mathbf{z}}(t_{1}) \times \hat{\mathbf{0}}_{0}^{+}]$$
(3.23)

and

くいで

380

nhóra

$$H_{N,1}^{T} R_{N,1}^{-1} H_{N,1} = \sum_{i=1}^{N} H_{i}^{T} R_{i}^{-1} H_{i} . \qquad (3.24)$$

These formulas reduce to a weighted least squares estimate, as expected<sup>22</sup>, when  $\theta_n^+ = 0$  and  $(\mathbb{P}_n^+)^{-1} = 0$ .

## 3.4 Physical Significance and Accuracy of Filtering

The physical significance of the clignment filtering is easily seen in Figure 13. The true gravity vector at the alignment site rotates in a predetermined done about the earth's spin axis. This gravity vector, as seen through the missing of inertial memory, is estimated to be  $\mathbf{E}_{est} = \mathbf{Z}(t) + \mathbf{Z}(t) : \mathbf{\hat{e}}$ , where  $\mathbf{\hat{e}}$  is the estimate of missing ment, while in actuality it is  $\mathbf{E}_{true} = \mathbf{Z}(t) + \mathbf{Z}(t) \times \mathbf{\hat{v}}$ , where  $\mathbf{\hat{s}}$  is the true missing ment. Both  $\mathbf{E}_{est}$  and  $\mathbf{E}_{true}$  rotate at the same rate as  $\mathbf{\hat{s}}$  shout the earth's spin axis and sweep out parts of cones from to to  $\mathbf{t}_N$ . The function of the filter is to bring these two partial cones into near coincidence by bringing to to  $t_{M}$ . The function of the filter is to bring these two partial cones into near coincidence by bringing of closer to Y. In the case of the repursive form, an adjustment is made at each data point, while in the batch form the estimate cone piece is adjusted to fit the data (and  $\phi_0^+$ ) in a single step. This picture of adjusting a partion of a cone to fit the observations explains why, for given  $t_1$ ,  $t_2$ , ...,  $t_N$  and given  $\Psi(t_1)$ , the alignment process is most accurate at the equator and inoperative at the poles: the conic surface generated by E in a given time is greatest at the equator and nil at the poles. Analytically, the observability matrix for the problem is just Equation (3.21), which may be written

> [[**X**(t<sub>1</sub>) × ]] H<sub>N,1</sub> = [[[(t<sub>0</sub>) ×]] i [[[(t<sub>0</sub>) ×]]

and this is seen to be singular if  $\mathbf{F}(t_1) = \mathbf{F}(t_2) = \dots = \mathbf{F}(t_N)$ , which occurs at the poles of the earth.

If a general measure of the accuracy of the alignment is desired, one way employ the ratio of the determinant of  $P_{n,1}^{*}$  to the determinant of  $P_{c}^{*}$ . Potter and Framer<sup>23</sup> have shown that this ratio is, in the present case,

$$\left|\mathbf{P}_{N,1}^{+}\right| \left| \mathbf{P}_{0}^{+} \right| = \left| \mathbf{R}_{N,1} \right| \left| \left| \mathbf{R}_{N,1} + \mathbf{H}_{N,1} \mathbf{P}_{0}^{+} \mathbf{H}_{N,1}^{T} \right| .$$
(3.26)

From this, it is seen that a large value of  $H_{N,1}P_0^+H_{N,1}^-$  will result in a large reduction of initial alignment error. This matrix, however, is composed of  $N^2$  3x3 sub-matrices, of which the (i,j)<sup>th</sup> sub-matrix is

$$H_{1} \mathbb{P}_{1}^{*} H_{1}^{T} = [\mathbf{Z}(t_{1}) \times ] \mathbb{E}(\hat{\mathbf{e}}_{0}^{*} \hat{\mathbf{e}}_{0}^{*T}) [\mathbf{Z}(t_{1}) \times ]^{T}$$
  
$$= \mathbb{E}[(\mathbf{Z}(t_{1}) \times \hat{\mathbf{e}}_{0}^{*}) (\mathbf{Z}(t_{1}) \times \hat{\mathbf{e}}_{0}^{*})^{T}], \qquad (3.27)$$

where f indicates expectation. If the earth were not rotating,  $f(t_i) = f(t_i)$  for  $0 \le i$ ,  $j \le N$ . In this case an initial misalignment  $\hat{e}_0^*$  parallel to  $\mathbf{g}(t_0)$  would make all the sub-matrices zero. Hence one may conclude, as expected, that the earth's rotation is necessary to remove initial misalignments about the vertical. Similarly, one can conclude that initial misalignments about axes normal to  $g(t_0)$  are reduced even without the earth's rotation.

### 4. UNIDENTIFIED LANDWARK NAVIGATION

## 4.1 Introduction

Unidentified landmark navigation<sup>9 u-27</sup> is an orbital navigation technique that relies heavily upon the recursive properties of the Kalman filter. The method produces improvement of vehicle position and velocity information

(3.25)





(3, 22)

R,

using only passive, on-board measurements. Since the essential information required is the direction from the which to an unidentified ground point, relative to inertial space, the typical instrumentation comprises a landmark telescope, a star tracker, and a computer. It is not necessary to store an extensive list of landmark locations and identification side. Moreover, the complete elimination of the setremat's participation is also a possibility, because it is necessary only to track patterns and not to identify them.

The methods available fall into two groups: (1) those such as Reference 24, in which the landmark location biases are introduced as state variables, resulting in a nine-dimensional state vector; and (2) there similar to Reference 20, in which the landmark biases do not appear explicitly in the state vector, which is then the usual six-vector of position and velocity deviations.

The first type, comprising what will be termed the one-sight and the two-sight methods will be discussed here because they clearly illustrate the role of the Kalman filter. A discussion of both types of methods is given in Reference 26.

Assuming that a computer on the orbiting vehicle is available and that a Kalman filter is employed to process navigational data from other than landmarks, the following additional computations must be performed in order to implement the unidentified landmark method under consideration: (1) a larger transition matrix of the state vector; (2) the estimated value of the landmark observation minus the true value; (3) the mensurement matrix, relating the landmurk observation to the state; (4) the best initial estimate of the landmark's position; and (5) the initial covariance of the state when a new landmark is located. These five required calculations will be discussed in turn.

## 4.2 State Transition Matrix

The nine state variables T,  $\nabla$ , T, representing deviations from nominal in position, velocity, and land-mark location, have a state transition matrix from  $t_a$  to t given by

 $\phi(\mathbf{t}, \mathbf{t}_{0}) = \begin{bmatrix} \phi_{\mathbf{r}\mathbf{r}} & \phi_{\mathbf{r}\mathbf{v}} & \mathbf{0} \\ \phi_{\mathbf{v}\mathbf{r}} & \phi_{\mathbf{v}\mathbf{v}} & \mathbf{0} \\ \mathbf{0} & \mathbf{0} & \phi_{l\,l} \end{bmatrix}, \quad (4.1)$ 

where the foul upper loft sub-matrices constitute the usual error propagation matrix for vehicle position and velocity, and the lower right sub-matrix gives the transition between landmark biases  $T_0$  at time  $t_0$  and their value T at time t. If T is in a planet-fixed coordinate system, then  $\phi_{ll}$  is the identity matrix I; if T is in inertial space, then

$$\phi_{11} = (\Omega)$$

where  $(\Omega)$  is the rotational transformation representing planetary rotation from time  $t_0$  to t. The most convenient coordinate system for  $\tilde{I}$  is one fixed in the planet, at least from the point of view of the state transition matrix computation.

#### 4.3 Entimated Observations

The observation is some function of vehicle position  $\vec{R}$ , velocity  $\vec{V}$ , and landmark location  $\vec{L}$  determined by the instrumentation. The computer calculates the observation  $Y_{\mathbf{R}}$  using estimated  $\vec{R}$ ,  $\vec{V}$ , and  $\vec{L}$  and subtracts it from the actual observation  $Y_{\mathbf{T}}$  supplied by the instruments. The difference  $Y_{\mathbf{T}} - Y_{\mathbf{R}}$ , weighted by the linear filter matrix, is used to correct the deviations  $\mathbf{F}$ ,  $\nabla$ ,  $\vec{I}$ . In practice, it is often computationally expedient to consider the "observation" to be some function of the actual instrument readouts.

In Table IV the first column gives the instrument readouts, the second column gives the computation of the, observation  $Y_T$  in terms of these readouts, and the third column gives the computation of the observation  $Y_g$  from estimated  $\overline{R}$ ,  $\overline{V}$ , and  $\overline{\Gamma}$ . It will be noticed that the landmark sensor angles T (train angle) and D (depression angle) may be processed together to yield the unit vector measurement  $\beta$  or processed separately to give two scalar measurements. Also, the calculation of  $(S)_{\overline{R}}$ , since it does not concern landmark navigation exclusively, is not given in detail.

#### 4.4 Monsurement Matrices

The third calculation required for the linear filter is the matrix M relating the state X to the observation deviation.  $\overline{y}$ . In the present case,  $\overline{x}$  includes deviations  $\overline{L}$  of the estimated landmark location from the nominal location. The measurement matrix may be obtained by differentiation of the estimated observation function, shown in column 3 of Table IV, with respect to  $\overline{x}$ ,  $\overline{V}$ ,  $\overline{L}$ . The results for the two choices of landmark observation  $Y_{m}$  are given in Table V.

## 4.5 Initial Estimate of Landmark Position

From the estimated vehicle position and the observed direction in inertial space from vehicle to landmark, the on-board computer must extract a first estimate of landmark location. Two ways of doing this will be described.

## One-sight method

If the moon or planet is assumed spherical of radius  $R_{g}$  near the landmark, then the observed line of sight  $\beta$  may be extrapolated from the vehicle position  $\bar{R}$  to where it intersects the sphere. This is the estimated position  $\bar{L}$  of the landmark. It is computed as

where

$$\rho = -\beta \cdot \mathbf{\bar{R}} - [\mathbf{R}_{-}^{2} - \mathbf{\bar{R}}_{-}^{2} + (\mathbf{\bar{R}} \cdot \mathbf{\beta})^{2}]^{1/2}$$

The major inacouracy of this method is in  $R_{\rm g}$ . Even if an ellipsoidal formula is used to compute  $R_{\rm g}$  for the earth, or a table stored, the terrain on which the landmark is situated may depart from these approximations by a significant distance.

## Two-sight method

If two sightings  $\beta_1$  and  $\beta_2$  are taken at times  $t_1$  and  $t_2$  and the estimated vehicle positions  $\overline{R}_1$  and  $\overline{R}_2$  recorded, then the on-board computer may estimate the location of the landmark with an error proportional to the average sighting distance times the instrumentation angular uncertainty. In this case, Reference 24 gives the formulas for  $\Gamma$  for a non-rotating planet. In vector form, they are

 $L_{1} = (L_{1}^{\prime} + L_{1}^{\prime\prime})/2$ ,

where.

$$\begin{split} \mathbf{\tilde{L}}_{1} &= \text{ two-alght estimate of } \tilde{\mathbf{L}} \text{ at } \mathbf{t}_{1} \\ \mathbf{\tilde{L}}_{1}^{\prime} &= \mathbf{\tilde{R}}_{1} + (\mathbf{\tilde{R}}_{p} - \mathbf{\tilde{K}}_{1}) \cdot (\beta_{1} - \mathbf{O}\beta_{2})\beta_{1}/(1 - \mathbf{C}^{2}) \\ \mathbf{\tilde{L}}_{1}^{\prime\prime} &= \mathbf{\tilde{R}}_{p} - (\mathbf{\tilde{R}}_{p} - \mathbf{\tilde{R}}_{1}) \cdot (\beta_{p} - \mathbf{O}\beta_{1})\beta_{p}/(1 - \mathbf{C}^{2}) \\ &= \mathbf{O} = \beta_{1} \cdot \beta_{p} \end{split}$$

and

If the planet is rotating with constant angular velocity  $\Omega$ , then, so shown in Appendix C, it is necessary only to replace  $K_g$  by  $(\Omega)^{-1}K_g$  and  $\beta_g$  by  $(\Omega)^{-1}\beta_g$ , where  $(\Omega)$  is the matrix representing planetary rotation from  $t_1$  to  $t_g$ .

Use of the two-sight method introduces a slight sequencing problem. It is obvious from the geometry that the two sights upon which the initial landmark estimate is based should be well-separated, preferably with 90° between  $\beta_1$  and  $\beta_2$ . But the linear filter cannot process readings until the initial landmark estimate is made. Hence, in this method, it is necessary to collect all the readings on a given landmark, select two wellseparated for the two-sight estimation, and then process the rest in order. If twelve T and D readings and their times are taken in the process, then about 36 storage locations are required. The sequence of operations is as follows:

(1) Store R.

(2) Collect and store  $\beta_i$  and  $t_i$ , i = 1, ..., N.

(3) Extrapolate  $\overline{R}_1$  to  $\overline{R}_N$  .

(4) Calculate  $E_1$  from Equation (4.3), using  $\bar{R}_N$  and  $\bar{\rho}_N$  in place of  $\bar{R}_2$ ,  $\bar{\rho}_2$ .

(5) Process the data  $\beta_i$ ,  $t_i$  for i = 2, ..., N-1 as though they occurred in real time. The result is  $\overline{R}_{N-1}$  and  $\overline{V}_{N-1}$ , the best estimates at  $t_{N-1}$ .

#### 4.6 Initial Covariance Matrix

The initialization of the covariance matrix of landmark and vehicle errors when a new landmark is placed can be a critical on-board computation, as the computer runs to be discussed later will show. The extent of the on-board computation depends on the method employed to estimate the landmark location when it is first sighted. The two methods just described lead to different covariance matrices. For the first, the one-sight method, the proper covariance is

 $\operatorname{oov}(\overline{x}) = \begin{bmatrix} P_{TT} & P_{TV} & P_{TI} \\ P_{TV}^{T} & P_{VV} & P_{VI} \\ P_{TI}^{T} & P_{VI}^{T} & P_{II} \end{bmatrix}, \qquad (4.4)$ 

(4.3)

A REAL PROPERTY OF A REAL PROPER



where

$$\mathbf{P}_{\mathbf{r}\,l} = \mathbf{P}_{\mathbf{r}\,\mathbf{r}}\mathbf{U}^{\mathsf{T}} + \frac{1}{2}\rho^{2}\sigma_{\theta}^{2}\mathbf{U}[\mathbf{I} - \beta\beta^{\mathsf{T}}]\mathbf{U}^{\mathsf{T}} + \frac{p\rho^{\mathsf{T}}\sigma_{\mathsf{T}}^{2}}{(\beta^{\mathsf{T}}\mathbf{L}/\mathbf{L})^{2}}$$

$$(4.5)$$

and

 $\mathbf{U} = \mathbf{I} - \beta \mathbf{E}^{\mathsf{T}} / \beta^{\mathsf{T}} \mathbf{E} , \quad \beta = \mathbf{E} - \mathbf{\bar{R}} , \quad \hat{\boldsymbol{\rho}} = \beta / \rho \qquad (4.6)$ 

 $\sigma_{\theta}^{\beta}$  = variance of instrumentation angular noise

 $\sigma_{\rm T}^{\rm f}$  = variance of terrain uncertainty.

For the second method, the two-sight method, the proper covariance is the same form as Equation (4, 4), but where

$$P_{pl} = (AP_{pr} + BP_{Vr})^{T}, P_{Vl} = (AP_{pv} + BP_{Vv})^{T}$$

$$P_{ll} = AP_{pr}A^{T} + AP_{rv}B^{T} + BP_{rv} + BP_{vv}B^{T} + D_{l}K_{l}D_{l}^{T} + D_{l}K_{r}K_{l}^{T}$$

$$(4.7)$$

and

	٨	*	$(\frac{1}{4})(1 + \phi_{rr})$	
	B	.=	(†) <i>φ</i> <sub>rv</sub>	
	D,	x	$(\frac{1}{2})\rho_1 \left[1 + \beta_1^* \beta_1^{*T}\right] U_1$	. (4.8)
•	D,	2	$(\dagger)\rho_{s}[x+\beta_{i}\beta_{i}^{T}]u_{s}$	. (4.0)
	K,	Ħ	$(\dagger)\sigma_{\mu_1}^{\dagger}[1-\beta_1\beta_1^{\dagger}]$	
	K,	z	$(\frac{1}{2})\sigma_{\sigma_{\pi}}^{3} [I - \hat{\rho}_{\pi}^{*} \hat{\rho}_{\pi}^{*T}]$	

In the preceding,  $\beta^*$  is written for  $(\Omega)^{-1}\beta$ , and the subscripts 1 and 2 refer to the times of the first and last sightingn. The quantities 1,  $\phi$ ,  $\rho$ ,  $\sigma_s^2$ , U, and  $(\Omega)$  have the same meanings as previously. Note that  $\sigma_s^2$ , the variance of termin roughness, does not appear in the second method, as to be expected. The derivations of these initial covariances are rather involved and will not be presented here.

## 4.7 Computer Simulation

Because the one-sight method appears to require less computation than other methods, a program was donstructed<sup>14</sup> to try it out. The program calculated the covariance of the nine-variable state by the linear filter equations. It simulated white-noise on the readings and the terrain error in estimating the landmarks. The error in the optimal estimate of the state was calculated and compared to the covariance matrix as an approximate deek.

The first significant result obtained is that the one-sight method does not work unless the covariance matrix is properly initialized for each new landmark. This result was obtained in deveral computer runs, a typical pair of which is shown in Figures 14 and 15. The conditions for these runs are listed in Table VI.

Figure 14 shows the reduction in the magnitude of the sctual position error and the square root of the trace of the position covariance when the covariance matrix was initialized for each landmark as:

 $e_{00} \psi(\vec{x}) = \begin{bmatrix} P_{PT} & P_{TV} & 0 \\ P_{TV}^{\Psi} & P_{VV} & 0 \\ f_{1} & 0 & D \end{bmatrix} = \Phi \times \Phi , \qquad (4.9)$ 

where  $P_{rr}$ ,  $P_{rv}$ ,  $P_{vv}$ , are the 3x3 sub-matrices of the vehicle position and velocity at the time of sighting the landmark, and D is a diagonal matrix of elements each equal to the trace of  $P_{rr}$ . Although this initialization procedure seems conservative, the results are disappointing; while the covariance matrix gradually approaches zero, the actual error, which it supposedly represents, goes to infinity.

In contrast, Figure 15 shows the identical computer run (same instrument noise, same terrain error, etc.), but with the covariance matrix initialized as given by Equations (4.4), (4.5) and (4.6). The actual arror here



is in much better agreement with the covariance matrix. From these and other similar pairs of runs, it is concluded that the covariance of the landmark error and cross correlation of landmark and vehicle errors must be calculated correctly at each new landmark for the one-sight method.

The computer runs just discussed are for a lunar satellite. Figure 16 shows the results of the one-sight method for an earth satellite. Soven landmarks are observed, all on the dayside. About six pairs of rendings (T and D), spaced 7.2 seconds apart, are taken on each landmark. This measurement schedule is judged to be within the capability of a human operator, cloud cover and other dutius allowing.

With the run shown in Figure 10 as a nominal case, the following parameters were porturbed: sensor noise, orbital altitude (e = 0), terrain error. The results are shown in Figures 17-19. These curves represent the square root of the trace of the covariance matrix immediately after, and one-half hour after, the last landmark observation.

## 5. CONCLUDING REMARKS

The facility with which the Kalman filter dispatched the complicated estimation problems described in Swetions 2, 3 and 4 should not obsoure two serious difficulties in its use, of which only a very brief discussion will be given here:

(1) The need to linearize the problem without loss of accuracy.

(2) The presence of computational difficulties.

## 5.1 Nonlinearity

The linearity problem enters, generally, through the approximation made in the state transition matrix or that made in the observation matrix. These two matrices represent linear approximations to what may be nonlinear relations.

In the case of orbital navigation, the measurement nonlinearity often produces a more serious error than does the state transition matrix approximation. This is usually true for radar range readings<sup>3</sup>. The problem of measurement nonlinearity may be overcome in an error analysis, such as described in Section 2, by using data simulated from a linear model, rather than real data or data generated by a nonlinear model. An error analysis thus constructed will not have nonlinearity errors, but will not be a reliable guide to the nonlinearity effects to be encountered in a program that processes real data.

In the case of inertial system alignment, Section 3, nonlinearity can enter only in the measurement matrix, since the state transition matrix is exactly the identity. The extent of the nonlinearity error depends on the size of the estimates  $\hat{s}$ .

Nonlinearity errors can be expected to occur in unidentified landmark navigation, both because of the measurement model and because of the state equation. The simulations described in Section 4, however, were based on a monlinear measurement process. For this measurement hat measurement nonlinearity does not seriously affect the landmark navigation methods of Section 4.

#### 5.2 Computational Difficulties

All three applications discussed in this chapter are susceptible to the numerical inaccuracies caused by a finite computer word length. These enter in many ways and have varied mechanisms and cures<sup>19-21</sup>, the discussion of which is taken up elsewhere in this publication. Only two areas of difficulty, which have appeared in the simulation programs of Sections 2 and 4 of this chapter, will be mentioned. They are the calculation of the covariance matrix

P\* = P\* - KHP\*

and the calculation of the matrix product

المراجع والمراجع والمراجع المراجع والمراجع والمراجع والمحاص والمحاص والمحاص والمحاص والمحاص والمحاص والمحاص وال

## PHT .

The first of these leads to an inaccurate result if  $P^*$  and KHP<sup>\*</sup> are almost equal. This is because the magnitude of the error (which is determined by the number of digits employed in the computer), divided by the magnitude of the answer (which may be extremely small if  $P^*$  and KHP<sup>\*</sup> are almost equal), yields s'large fractional error in the new  $P^+$  matrix. This situation arises, as it did in the simulations of Section 2, when a series of highly accurate range readings are taken by a ground radar on a satellite whose position uncertainty in approaching the station is very large. The initially large covariance matrix is reduced by a series of subtractions of the form  $\neg$ -KHP to one that is smaller by several orders of magnitude, at least in the direction of the line of sight from the station to the vehicle. If these orders of magnitude equal or exceed the number of digits in the computer word, all numerical significance is lost.

المراجع والمراجع والمحارب والمراجع والمحارب والمحارب والمحاج وا













|--|



Sin Binte werk Breite



le e la 🌒 📲 la cara 🔘 even

The second type of computation,  $PH^T$ , causes numerical inaccuracies in the same circumstances as the first. In fact it can be seen that the covariance calculation of Equation (2.2), for the case of  $R_1 = 0$ , leads to a new covariance matrix  $P^*$  such that  $P^*H^T = 0$ . But neither  $P^+$  nor  $H^T$  is zero; each zero in the product must be the result of a computation of the form

If the elements of  $P^+$  and of  $H^T$  are large, the zero result can come about only by the subtraction of equal, large numbers with the attendant loss of numerical significance. In the case of radar range readings,  $H^T$  is just  $\beta_0$ , the unit vector from ground station to vehicle; after many highly accurate range rendings the vehicle's position uncertainty, as represented by P, may still be large, but is distributed almost exclusively in a plane poreal to  $\beta_0$ . Hence the product  $P\beta_0$  is almost zero, even though the individual elements of P may be large. This situation is typical of many in which highly accurate, closely-spaced observations are made.

## ACKNOWLEDGMENTS

The work on which Soctions 2 and 4 of this chapter are based was largely carried out while the author was employed in the Sperry Rand Systems Group, Division of the Sperry Rand Corporation, Great Neck, NY. The efforts of E.Wu and R.Massey in constructing the computer program of Section 2 are gratefully schowledged.

REFERENCES

Gon	erul	
1.	Battin, R.L.	A Statistical Optimizing Navigntion Procedure for Space Flight. American Rocket Society Journal, Vol.32, No.11, Nov. 1962, pp. 1681-1686.
2.	McLean, J.D. et al.	Outimal Filtering and Linear Prediction Applied to a Space Navigation System for the Circumlunar Mission. NABA TN D-1200, March 1962.
3.	Denham, W.F. Pines, S.	Sequential Extination When Measurement Function Newlinearity is Comparable to Measurement Error. AIAA Journal, Vol.4, No.6, June 1988, p.1071ff.
Ref	erences for section 2	
4.	Andrews, A.	A Square Root Formulation for the Kalman Coveriance Equation. AIAA Journal, Vol.6, 1968, pp.1165-1186.
5.	Kochi, K.C.	Exact First-Order Navigation-Guidance Mechanization and Error Propagation Equations for Two-Body Reference Orbits. AIAA Journal, Vol. 2, 1984, pp. 385-388.
6.	Kochi, K.C.	Exact Two-Body Error Sensitivity Coefficients. AIAA Journal, Vol.2, 1984. p. 1503.
7.	Maxwell, J.A. Dorfman, M.	The Accuracy of Satellite Orbit Determination. AIAA Paper 63-155, June 1983.
8,	Noe, K.	Stochastic Models for Errors in Orbit Predictions for Artificial Earth Satellites. American Rocket Bociety Journal, Vol.32, No.11, Nov.1962, pp.1726-1728.
9.	Karrenberg, H.K. et al.	Yariations of Satellite Position with Uncertainties in the Mean Atmospheric Density. American Rockot Society Journal, Vol.32, No.4, April 1942, p.876ff.
10.	Rauch, H.E.	Optimum Estimation of Satellite Trajectories Including the Effect of Random Variations in Drag. AIAA Paper 64-306, July 1964.
11.	Wengert, R.E. Jones, J.F.	Orbit Determination Accuracies Using Systematic Error Adjustments. AIAA Paper 64-395, July 1964.
12.	Guttmann, P.T.	The Effects of Uncertainties in Some Zonal Harmonics of the Earth's Gravity Potential on Satellite Position Detection. 85D TDR 83-174 TOR-289 (4110)-2, AD415437.
13.	Kaula, W.N.	Determination of the Earth's Gravitational Field. Review of Geophysic Vol.1, No.4, Nov.1983, pp.507–551. (Also NASA RP-132.)



## an a wards and the state of the

and the second of the second second

F.

۰,

996		
14.	Khula, Ŵ.M.	The Comparison and Combination of Satellite with Other Determination of Geodetic Parameters. Goddard Space Flight Center Contributions to the COSPAR Musting, May 1962, pp.119-124.
15.	Kaula, W.M.	A Review of Geodetic Parameters. NABA TN D-1847, May 1963.
16.	Proskurin, V.P. Betrakov, V.V.	Perturbations in the Motion of Artificial Satellites Due to the Oblateness of the Marth. American Rocket Society Journal, Vol.31, No.1, Jan. 1961, pp. 117–123. (Translated by Research Information Bervice, New York, from Ryullsten' Instituta Teoretischeskoy Astronomol, Vol.7, No.7 (90), 1960, pp.537–548.)
17.	McArthur, W.G.	Horizon Sensor Navigation Errors Resulting from Statistical Variations in the CO <sub>2</sub> 14-16 Micron Radiation Band. Published in the Proceedings of the 9th Symposium on Ballistic Missile and Space Tochnology, San Diego, California, August 12-14, 1964.
18.	Wilcox, J.C.	Self-Contained Orbital Navigation Systems with Correlated Measurement Errors. Published in the Proceedings of the AIAA/ION Guidance and Control Conference, Minnespolis, Minnesota, 10-18 August, 1965. (The variances referred to are the disgonal elements of Equation (42) of this Reference.)
Xø f	erences for Section 3	,
19,	Schlee, F.H. et ml.	vergence in the Kalman Filter. AIAA Journal, Vol. 5, 1987, pp. 1114-1120.
20.	Mendelwohn, J. G'Dwyer, W.	Astrody, mice of Lunar Satellites, Part I: Orbit Determination. Vol.I - Summury and Conclusions, Vol.II - Derivations - Grumman Final Report on Contract NAS 8-5015, NASA Report RE-168, Oct.1963.
21.	Bellantoni, J.F. Dodge, K.W.	A Square-Root Formulation of the Kalman-Schmidt Filter. AIAA Journal, Vol.5, No.7, Aug. 1967. p. 1309ff.
22.	Smith G.L. et al.	Application of Statistical Filter Theory to the Optimal Estimation of Position and Velocity on Board a Circumlunar Vehicle. NAMA TR R-136, Appendix F, 1962.
23.	'Potter, J.E. Fraser, D.C.	A Formula for Updating the Determinant of the Covariance Matrix. AIAA Journal, Vol.5, No.7, July 1967, pp.1352-1354.
Xe (	erences for Section 4	
24.	DeNessa, E.J. Dittrich, M.S.	Orbit and Landwark Determination During Lunar Orbit. Numter's Themis. Department of Aeronautics and Astronautics, Nassachusetts Institute of Technology, May 17, 1983.
25.	Levine, G.M.	A Method of Orbital Navigation Using Optical Sightings to Unknown Landmurks. Published in the Proceedings of the AlAA/ION Guidance and Control Conference, New York, 1965.
26.	Bellantoni, J.F.	Unidentified Landmark Navigation. AIAA Journal, Vol.5, No.8, 1967, pp.1478-1483.
27.	Toda, N.F. Bohles, F.H.	Autonomous Orbital Navigation by Optical Tracking of Unknown Landmarks. Published in the Proceedings of the AIAA/JACC Guidance and Control Conference, New York, 1966.



A Courses

AN AREA D





4**9**5114

an an that the second secon

## APPENDIX A

## Derivation of Equation (2.7)

If the linear filter K is used to update the state estimate  $\hat{x}^-$ , what is the covariance  $P^+$  of the error in the resulting estimate,  $\hat{x}^+$ ? Let

 $\hat{\mathbf{x}}^+ = \hat{\mathbf{x}}^- + \mathbf{K}(\mathbf{z} - \mathbf{H}\hat{\mathbf{x}}^-) ,$ 

where

$$z = Hx + v$$

$$E(v) = 0$$

$$E(vv^{T}) = R = known$$

$$E(x - \hat{x}^{-})v^{T} = Ev(x - \hat{x}^{-})^{T} = 0$$

A direct calculation of P<sup>+</sup> from its definition gives

$$P^{+} = E(x - \hat{x}^{+})(x - \hat{x}^{+})^{T}$$

- = x[x 2" K(p H2")][x 2" K(x H2")]<sup>T</sup>
- = F[e" KHe" KV][e" KHe" KV]<sup>†</sup>
- =  $\mathbf{E}[(\mathbf{I} \mathbf{K}\mathbf{H}\mathbf{e})\mathbf{e}^{T} \mathbf{K}\mathbf{v}][(\mathbf{I} \mathbf{K}\mathbf{H})\mathbf{e}^{T} \mathbf{K}\mathbf{v}]^{T}$
- $= (\mathbf{I} \mathbf{KH}) \mathbf{E} [(\mathbf{e}^{*}) (\mathbf{e}^{*})^{T}] (\mathbf{I} \mathbf{KH})^{T} + \mathbf{KE} [(\mathbf{v}) (\mathbf{v})^{T} \mathbf{K}^{T}] \mathbf{KE} [(\mathbf{v}) (\mathbf{e}^{*})^{T}] (\mathbf{I} \mathbf{KH})^{T} (\mathbf{I} \mathbf{KH}) \mathbf{E} [(\mathbf{e}^{*}) (\mathbf{v})^{T}] \mathbf{K}^{T}$
- =  $(I KH)P^{-}(I KH)^{T} + KRK^{T}$ ,

where  $\bullet$  has been used for  $x - \hat{x}$  and P for  $E(ae^{T})$ .

## APPENDIX B

## State Transition Netrix for Air Drag, Gravity and Oblateness Coefficients

This Appendix derives the differential equations for vehicle position and velocity errors produced by constant errors in the drag parameter, oblateness coefficients and gravitational parameter of an earth satellite.

The vehicle position  $\overline{R}$  and velocity  $\overline{V}$  are given by

D,

 $\frac{d}{dt}\begin{bmatrix} \bar{R} \\ \bar{V} \end{bmatrix} = \begin{bmatrix} \bar{V} \\ \bar{P}_{\mu} + \bar{P}_{0B} + \bar{D} \end{bmatrix}, \qquad (B.1)$ 

where

 $\vec{F}_{\mu} = -\mu \vec{R}/R^{2}$   $\vec{F}_{08} = -\partial U_{08}/\partial \vec{R}$   $\vec{D} = \frac{1}{2} D_{0} \sqrt{V} e^{-K_{0}(R-R_{0})}$   $U_{08} = -\mu \sum_{n=1}^{\infty} C_{n} (R_{n}^{n}/R^{n+1}) P_{n} (n/R)$ 

and

<b>0</b> '	К <sub>0</sub> .	R <sub>0</sub>	=	drag constants	
R <sub>H</sub>			=	radius of earth corresponding to	¢,

drag parameter

		۲ ۲
	ا مرالم مرالم ارمار م	
iça.		







= coefficient of a<sup>th</sup> zonal harmonic in geopotential = n<sup>th</sup> Legendre polynomial P<sub>n</sub>(a) = gravitational parameter of earth.

It is easily shown that

$$\mathbf{F}_{OB} \approx (\mu/R^3) \sum_{n=1}^{\infty} C_n (R_{\mathbf{E}}/R)^n \left[ \overline{kR} \frac{d\mathbf{P}_n(\mathbf{s})}{d\mathbf{s}} \sim (n+1)\mathbf{P}_n(\mathbf{s})\overline{k} - \mathbf{s} \frac{d\mathbf{P}_n(\mathbf{s})}{d\mathbf{s}} \overline{k} \right], \qquad (B.2)$$

where

2/R

C<sub>B</sub>

μ

= unit vector normal to equator. Ŕ

The deviations  $\delta \bar{R}$  and  $\delta \bar{V}$  in position and velocity corresponding to biases  $\delta \mu$ ,  $\delta C_i$ ,  $\delta D_0/D_0$ , in gravitational parameter  $\mu$ , oblateness coefficients  $C_i$ , and drag parameter  $D_0$ , follow the differential equations

$$\frac{d}{dt} \begin{bmatrix} \delta \vec{R} \\ \delta \vec{V} \end{bmatrix} = \begin{bmatrix} 0 & I \\ M_{R} & N_{V} \end{bmatrix} \begin{bmatrix} \vec{R} \\ \vec{V} \end{bmatrix} + \begin{bmatrix} 0 & \\ \delta \vec{F}_{\mu} + \delta \vec{F}_{0B} + \delta \vec{D} \end{bmatrix}.$$
(B.3)

where

and

$$\begin{aligned} \partial \overline{F}_{\mu} / \partial \overline{K} &= -\mu \left[ I - 3\overline{K}\overline{K}/R^2 \right]/R^3 \\ \partial \overline{F}_{0R} / \partial \overline{K} &= (3\mu/2)R_E^2 \frac{C_2}{R^2} \left\{ \left[ (R^2 - x^2)I + \left(\frac{7x^2}{z^2} - b\right)\overline{K}\overline{R} \right] - 2x[6\overline{K}\overline{R} + \overline{R}\overline{K}] + 2R^2\overline{K}\overline{K} \right\} \\ \partial \overline{0} / \partial \overline{R} &= -K_0 D_0 \sqrt{V}\overline{R} e^{-K_0(R-R_0)}/2R_0 \\ \partial \overline{D} / \partial \overline{V} &= \frac{1}{4} (D_0 / V) \left[ \overline{V}\overline{V} + V^2 I \right] e^{-K_0(R-R_0)} \\ \delta \overline{F}_{\mu} &= -\overline{R} \delta \mu / R^3 \\ \delta \overline{F}_{0R} &= \frac{\mu}{R^2} \sum_{n=1}^{\infty} \left( \frac{R_R}{R} \right)^n \left[ -P_n (n+1) + e \frac{dP_n}{de} \overline{R} + R \frac{dP_n}{de} \overline{K} \right] \delta C_n \\ \delta \overline{D} &= (\frac{1}{4}) D_0 V \overline{V} e^{-K_0(R-R_0)} \delta D_0 / D_0 \end{aligned}$$

The state transition matrix,  $\Phi_0$ , is obtained by integrating from  $t_0$  to t the matrix differential equation corresponding to Equation (B.3):

 $\frac{d}{dt} \begin{bmatrix} \phi_{D} \end{bmatrix} = \begin{bmatrix} 0 & I \\ \mathbf{M}_{R} & \mathbf{M}_{V} \end{bmatrix} \begin{bmatrix} \phi_{D} \end{bmatrix} + \begin{bmatrix} A \end{bmatrix}$  $[\Phi_{\rm p}(0)] = [0]$ 

(8.4)

where

$$(A) = \begin{bmatrix} 0 & 0 & 0 & 0 & 0 \\ \frac{\delta \vec{\mathbf{F}}_{\mu}}{\delta \mu} & \frac{\delta \vec{\mathbf{F}}_{0B1}}{\delta C_{j}} & \frac{\delta \vec{\mathbf{F}}_{0B1}}{\delta C_{j}} & \frac{\delta \vec{\mathbf{F}}_{0Bk}}{\delta C_{k}} & \frac{\delta \vec{\mathbf{F}}_{0Bk}}{\delta C_{k}} & \frac{\delta \vec{\mathbf{F}}_{0B1}}{\delta C_{l}} & \frac{\delta \vec{\mathbf{D}}_{0}}{\delta D_{0}/D_{0}} \end{bmatrix} .$$

1991 - The Part of the Part of

The typical quantity  $\delta F_{OBD}$  in the above is

$$\frac{\int_{0}^{n} \frac{\partial P_{n}}{\partial C_{n}} = \frac{\mu}{R^{3}} \left( \frac{R_{R}}{R} \right)^{n} \left\{ - \left[ (n+1)P_{n}(s) + s \frac{dP_{n}(s)}{ds} \right] R + R \frac{dP_{n}(s)}{ds} R \right\},$$

where

$$\frac{P_{n+1}(s)}{ds} = \frac{2n+1}{n+1} \frac{sP_n(s)}{sP_n(s)} - \frac{n}{n+1} \frac{P_{n-1}(s)}{n-1}$$

$$\frac{dP_{n+1}(s)}{ds} = \frac{2n+1}{n+1} \left[ s \frac{dP_n(s)}{ds} + P_n(s) \right] - \frac{n}{n+1} \frac{dP_{n-1}(s)}{ds}$$

and

 $P_0 = 0, \quad \frac{dP_0}{da} = 0$  $P_1 = s, \quad \frac{dP_1}{da} = 1.$ 

Equation (E.4) will yield the transition matrix relating position and velocity errors to the normalized biases  $\delta\mu$ ,  $bc_i$ ,  $\delta c_i$ ,  $\delta$ 

## APPENDIX C

## Two-Sight Landmark Location for a Rotating Planet

The problem is to obtain an estimate  $E_i$  of landmark position at time  $t_i$  given observation  $\beta_i$  and vehicle position  $\vec{R}_i$  at  $t_i$ , and  $\beta_g$ ,  $\vec{R}_g$  at time  $t_g$ , ,

The landwark position at  $t_i$  may be estimated as

$$\vec{\mathbf{L}}_{i} = \vec{\mathbf{R}}_{i} + \rho_{i} \hat{\rho}_{i} , \qquad (\mathbf{C}, 1)$$

where  $\rho_{i}$  is unknown. The landmark location at time t, may be estimated as

$$\vec{\mathbf{L}}_{p} = \vec{\mathbf{R}}_{p} + \rho_{p} \dot{\rho}_{p} , \qquad (C.2)$$

where  $\rho_g$  is unknown. Since  $\tilde{L}_2 = (\Omega)\tilde{L}_1$ , where  $(\Omega)$  is the rotational transformation of the planet from  $t_1$  to  $t_2$ , the last equation becomes

$$\vec{\mathbf{L}}_{1} = (\Omega)^{-1} \vec{\mathbf{R}}_{2} + \rho_{2}(\Omega)^{-1} \hat{\rho}_{2} . \qquad (C.3)$$

Equations (C. 1) and (C. 3) are six scalar equations in five unknowns, and hence two solutions,  $\bar{L}_1'$  and  $\bar{L}_1''$ . may be obtained by straightforward algebra:

$$\vec{L}_{1}^{\prime} = \vec{R}_{1} + (\vec{R}_{2}^{*} - \vec{R}_{1}) \cdot (\beta_{1} \beta_{1} - C \beta_{2}^{*} \beta_{1}) / (1 - C^{2}) , \qquad (C.4)$$

 $\vec{\mathbf{L}}_{1}^{\prime\prime} = \vec{\mathbf{R}}_{g}^{*} - (\vec{\mathbf{R}}_{g}^{*} - \vec{\mathbf{R}}_{1}) \cdot (\hat{\boldsymbol{\rho}}_{\chi}^{*} \hat{\boldsymbol{\rho}}_{g}^{*} - c \hat{\boldsymbol{\rho}}_{1} \hat{\boldsymbol{\rho}}_{\chi}^{*}) / (1 - c^{c}) , \qquad (C.5).$ 

where

 $\overline{\mathbf{n}}_{g}^{\phi} = (\Omega)^{-1} \overline{\mathbf{n}}_{g}, \quad \hat{\boldsymbol{\rho}}_{g}^{\phi} = (\Omega)^{-1} \hat{\boldsymbol{\rho}}_{g}$   $\mathbf{C} = \hat{\boldsymbol{\rho}}_{1} \cdot \hat{\boldsymbol{\rho}}_{g}^{\phi}.$   $(\mathbf{U}, \mathbf{6})$ 

These solutions are not identical because of noise on the data  $\beta_1$  and  $\beta_2$ . Their average is a better estimate than either one alone. Thus,

$$\vec{L}_{1} = (\vec{L}_{1}' + \vec{J}_{2}')/2$$
 (C.7)

It is seen that the only effect of the planet's rotation is to introduce  $(\Omega)^{-1}$  in front of  $\bar{R}_2$  and  $\bar{\rho}_2$  .



Š

43.45.4 \* 4 \* 4 4 h

# TABLE I

Observation Matrices

Observation	State variables	Observation matrix
ρ	х,у, в	A <sub>0</sub>
	$\Delta \bullet$ , $\Delta n$ , $\Delta h$	$(-\hat{ ho}_{\mathbf{G}}\cdot\hat{\mathbf{e}}, -\hat{ ho}_{\mathbf{G}}\cdot\hat{\mathbf{n}}, -\hat{ ho}_{\mathbf{G}}\cdot\hat{\mathbf{h}})$
	Δρ	:
Þ	х,у, г	$[(\overline{\mathbf{v}} - \overline{\mathbf{v}}_{\mathbf{a}}) - (\overline{\mathbf{v}} - \overline{\mathbf{v}}_{\mathbf{a}}) \cdot \hat{\rho}_{\mathbf{a}} \hat{\rho}_{\mathbf{a}}] / \rho_{\mathbf{a}}$
	<b>. . .</b> ż	$\rho_{a}/\rho_{a}$
	$\Delta \bullet$ , $\Delta n$ , $\Delta h$	(ê·a, ñ·a, ĥ·a)
	<b>ά</b> Δ	1
٨		$(\hat{\mathbf{e}} \cos \mathbf{A} - \hat{\mathbf{b}} \sin \mathbf{A})/\rho_{\alpha} \cos \mathbf{E}$
	$\Delta \bullet$ , $\Delta n$ , $\Delta h$	$\left\{ \hat{\mathbf{o}}(\mathbf{K}, \rho_n^2 - \rho_n/\rho_{nn}^2) + \hat{\mathbf{n}}(\mathbf{K}, \rho_n, \rho_n + \rho_n/\rho_{nn}^2) + \bar{\rho}_n \mathbf{x} [\hat{\mathbf{n}} \rho_n, \mathbf{K}, + (\hat{\mathbf{n}} \rho_n + \hat{\mathbf{o}} \rho_n)/\mathbf{R}_n \rho_{nn}^2] \right\}$
-	$\Delta \mathbf{e}, \Delta \mathbf{n}, \Delta \mathbf{h}$	$(\hat{\mathbf{r}}_{0} / \mathbf{R}_{0} + \hat{\mathbf{r}}_{0} / \mathbf{R}_{0} = \hat{\mathbf{r}} + \hat{\mathbf{r}}_{0} \hat{\mathbf{A}} / \hat{\mathbf{r}}_{0})/\hat{\mathbf{r}}_{0}$
~		1
- -		
Ψi	Х, У, Б с	$\left(-RR_{\rm H}/R^{\circ}/(R^{\circ}-R_{\rm EH}^{\circ})\right)$
	n <sub>1</sub>	$(1/\sqrt{(R^2 - R_{\rm EN}^2)})$
	μ. 	_   1///===== \
	ΔΦ.	
7		
Φ2	X, Y, Z	- <del>-</del>
	$\theta_1, \theta_2, \theta_3$	(∩,0,-1) for IMU on ず行く axes
		- C for the on XYZ axes
	n <sub>2</sub>	$1/r (R^ R_{BM})$
	φ <sub>2</sub> ΔΦ	
Φ,	X, Y, 2	- ζ /R
	$\theta_1, \theta_2, \theta_3$	(0,1,0) for IMU on ξηζ axes
	$\sigma_1, \sigma_2, \sigma_3$	77 IOF INU ON ATZ axes
-	<sup>n</sup> ₃ ∧a	1//(R <sup>-</sup> ~R <sub>EN</sub> )
	μ <sup>ωψ</sup> ι <del> </del> <del> </del>	
	<sup>4</sup>	•



State Property and

14 10 19 19 19 19

ł

		TABLE I (continued)
Observation	State variables	Observation matrix
TRETAR	$\theta_1$ , $\theta_2$ , $\theta_3$	$\hat{f} = \rho_{sf}(\rho_{s\eta}, \hat{f}) + \rho_{sf}(\hat{\xi})/\delta'$ , for IMU on XYZ exes
	$\theta_1, \theta_2, \theta_3$	$(1, -\rho_{\mu\xi}\rho_{a\eta}/\delta', -\rho_{a\xi}\rho_{a\xi}/\delta')$ , IMU on $\hat{\xi}\hat{\eta}\hat{\xi}$ axes
	Ω <sub>i</sub>	1
	Δ TR <sub>BT AR</sub>	1
ELOTAR	$\theta_1$ , $\theta_2$ , $\theta_3$	$-\hat{\rho}_{s} \times \hat{\xi}/\sqrt{\delta^{2}}$ , IMU on XYZ axes
	$\theta_1, \theta_2, \theta_3$	(0, $-\rho_{sl}/\sqrt{\delta'}$ , $\rho_{s\eta}/\delta'$ ), IMU on $\hat{\xi}\hat{\eta}\hat{\xi}$ axes
	Ω <b>,</b>	2
	ΔEL <sub>STAR</sub>	1
TR <sub>lmk</sub>	x,y,z	Ê x A_/8"
	$\theta_1, \theta_2, \theta_3$	$\hat{\mathcal{E}} = \rho_{L\mathcal{E}}(\rho_{L\eta}\hat{\eta} + \rho_{L\zeta}\hat{\zeta})/\delta''$ , IMU on XYZ axem
	θ1, θ2, θ3	$(1, -\rho_{L_{\ell}}\rho_{L_{\eta}}/\delta'', -\rho_{L_{\ell}}\rho_{L_{\ell}}/\delta''), \text{ IMU on } \hat{\xi}\hat{\eta}\hat{\xi}^{\bullet} \text{ axes}$
	<i>π</i> <sub>1</sub>	1
	∆ TR <sub>LMK</sub>	1
el <sub>lux</sub>	x, y, z	Ê/18"
	0, 0, 0, 0,	$-\hat{\rho}_{\rm L} \times \hat{\mathcal{E}}/\sqrt{\delta''}$ , IMU on XYZ axes
	$\theta_1, \theta_2, \theta_3$	(0, $-\rho_{L\ell}/\sqrt{\delta}$ ", $\rho_{L\eta}/\sqrt{\delta}$ "), ZMU un $\hat{\zeta}\hat{\eta}\hat{\zeta}$ exes
	# <sub>1</sub>	1
RA	x, y, x	Ĩ./R
	A	•

NOTE: Auxiliary quantities for Table I are given in Figures 1-5 and as follows:

. +4

R =  R	R <sub>EH</sub> = distance from center of earth
$\vec{V}_{0} = \vec{\omega}_{g} \times \vec{R}_{0}$ $\vec{\omega}_{g} = \text{angular velocity of earth}$ $\rho_{0} =  \vec{P}_{0} $ $\vec{\omega}_{c} = \vec{\omega}_{g} \times \hat{\rho}_{0} + (\vec{V} - \vec{V}_{0}) \cdot \hat{\rho}_{0} \hat{\rho}_{0} / \rho_{0} - \frac{1}{2} (\vec{V} - \vec{V}_{0}) / \rho_{0}$	$\rho_{\mathbf{n}\xi} = \hat{\xi} \cdot \hat{\rho}_{\mathbf{n}}$ $\rho_{\mathbf{n}\eta} = \hat{\eta} \cdot \hat{\rho}_{\mathbf{n}}$ $\rho_{\mathbf{n}\xi} = \hat{\xi} \cdot \hat{\rho}_{\mathbf{n}}$ $\delta' = \rho_{\mathbf{n}\xi}^{*} + \rho_{\mathbf{n}\eta}^{*}$
$K_{1} = \tan \lambda_{0} / R_{0} \rho_{en}^{2}$ $R_{0} =  \tilde{R}_{0} $ $\rho_{e} = \hat{e} \cdot \rho_{q}$ $\dot{\rho}_{n} = \hat{n} \cdot \rho_{0}$ $\rho_{en}^{2} = \rho_{e}^{2} + \rho_{n}^{2}$ $\rho_{e} = \hat{h} \cdot \rho_{n}$	$\begin{aligned} \beta_{L} &= \beta_{L} /  \beta_{L}  \\ \rho_{L\xi} &= \hat{\beta}_{L} \cdot \hat{\xi} \\ \rho_{L\eta} &= \hat{\beta}_{L} \cdot \hat{\eta} \\ \rho_{L\zeta} &= \hat{\beta}_{L} \cdot \hat{\eta} \\ \hat{\xi}^{\prime \prime} &= \hat{\beta}_{L} \cdot \hat{\xi} \\ \hat{\xi}^{\prime \prime \prime} &= \rho_{L\eta}^{2} + \rho_{L\zeta}^{2} \end{aligned}$

TABLE II

## List of State Variables (Excluding Biases)

	vohicle X inertial position error
	vehicle Y inertial position error
	vehicle Z inertial position error
	vehicle inertial velocity error along X axis
	vehicle inertial velocity error along Y axis
·	vehicle inertial velocity error along Z axis
	gyro 1 angular error, due to drift
	gyro 2 angular error, due to drift
	gyro 3 angular error, due to drift
	gyro 1 random drift rate, time correlated
	gyro 2 random drift rate, time correlated
	gyro 3 random drift rate, time correlated
	gyro 1 drift rate bias
	gyto 2 drift rate bias
	gyro 3 drift rate bias
	horigon phenomenological deviation affecting altitude
	horizon phenomenological deviation affecting vertical
	horizon phenomenological deviation affecting vertical
	horizon mensor time correlated error in $\Phi_i$
	horizon sensor time correlated error in $\Phi_2$
	horizon sensor time correlated error in $\Phi_3$
	star tracker time correlated error in TR <sub>g</sub>
	star tracker time correlated error in EL
	landmark sensor time correlated error in TR <sub>L</sub> .
	landmark sensor time correlated error in EL
ı	Diam in earth's gravity, $\mu$
	biases in zonal harmonics of gravitational potential, $2 \le N(1)$ , $N(2)$ , $N(3)$ , $N(4)$

8C<sub>N(1)</sub> : 8C<sub>N(4)</sub>

δυ₀/D₀

÷

والمراجع المراجع والمراجع والمراجع والمحالية والمحالية والمحالية والمراجع المراجع والمحالية والمراجع والمحالية

π<sub>1</sub> π<sub>2</sub> δμ

362

x Y z x Y 2 θ<sub>1</sub>

θ<sub>2</sub> θ<sub>3</sub> ω<sub>1</sub> ω<sub>2</sub> ω<sub>3</sub> ω<sub>81</sub> ω<sub>82</sub> ω<sub>83</sub>

 $\begin{array}{c} \mathbf{\tilde{h}}_{1} \\ \mathbf{\tilde{h}}_{2} \\ \mathbf{\tilde{h}}_{3} \\ \mathbf{\tilde{\phi}}_{1} \\ \mathbf{\tilde{\phi}}_{2} \\ \mathbf{\tilde{\phi}}_{3} \\ \mathbf{\tilde{\phi}}_{3} \\ \mathbf{\tilde{\phi}}_{1} \\ \mathbf{\tilde{\phi}}_{2} \\ \mathbf{\tilde{\phi}}_{3} \\ \mathbf{\tilde{\phi}}_{1} \\ \mathbf{\tilde{\phi}}_{1} \\ \mathbf{\tilde{\phi}}_{2} \\ \mathbf{\tilde{\phi}}_{3} \\ \mathbf{\tilde{\phi}}_$ 

bias in drag parameter, D<sub>0</sub>

والمراجع والروار والروائي فكروان والمراجع والروائي والروائي

· · ·

a a star a star







tine führe breis Dieter

· 1 · 11 💼

1 acres of

THE THE PARTY OF

-----

## TABLE 111

## List of Sias State Variables

Δ٠	ground radar station location error, easterly
Δn	ground radar station location error, northerly
Δh	ground radar station location error, vertical
Δρ	ground radar range bias
٥¢	ground radar range rato bias
44	ground rader azimuth bias
Δĸ	ground radar elevation bias
$\Delta \tilde{\Phi}_i$	horizon tracker stadimetric angle bias
∆ <b>∛</b> ₽	horizon tracker vertical angle bias on $\xi$ axis
۵ <b>Φ</b> 3	horizon tracker vertical angle bias on $ \hat{\eta} $ axis
∆Ĥ <sub>1</sub>	horizon deviation bias, affecting stadimetric altitude
$\Delta TR_g$	star-IMU train angle bias
∆rl <sub>s</sub>	star-IMU elevation angle bigs
∆tr <sub>l</sub>	landmark-IMU train angle bian
$\Delta \mathbf{EL}_{\mathbf{L}}$	landmark-IMU elevation angle bias
ΔRA	radar altimeter bias

## TABLE IV

Observations for Unidentified Landmark Navigation

Instrument readout	True observation, Y <sub>T</sub>	Estimated observation, Y <sub>E</sub>
Star tracker $\alpha_i$ , $\delta_i$ (1 = 1, 2, 3)	$(Y)_{T} = (S)$ . where $(S) = (\hat{S}_{11}, \hat{S}_{12}, \hat{S}_{13}) (\hat{S}_{v1}, S_{v2}, S_{v3})^{-1}$ $S_{v1} = (\cos \delta_{1} \cos \alpha_{1}, \cos \delta_{1} \sin \alpha_{1}, \sin \delta_{1})^{T}$ , $i = 1, 2, 3$	(Y) <sub>E</sub> = (S) <sub>E</sub>
Landmark sensor T , D	Vector observation:	$\hat{Y}_{\mathbf{E}} = \frac{\mathbf{L} - \mathbf{R}}{ \mathbf{L} - \mathbf{R} }$
Landmark sensor 7 D	Scalar observations: Y <sub>T</sub> = T - Y <sub>T</sub> = D	$Y_{g} = \tan^{-1} \frac{(\overline{L} - \overline{R}) \cdot \hat{S}_{2}}{(\overline{L} - \overline{R}) \cdot \hat{S}_{1}}$ $Y_{g} = \sin^{-1} \frac{(\overline{L} - \overline{R}) \cdot \hat{S}_{1}}{ \overline{L} - \overline{R} }$

th and elevation of it star relative α, δ, unit column vector of ith star relative to inertial space ŝ matrix transformation from vehicle frame to inertial space =  $(\hat{B}_1, \hat{B}_2, \hat{B}_3)$ (#) × = train angle of landmark, measured about vehicle S1 axis T D

= depression angle of landmark, measured normal to vehicle  $S_2S_3$  plane

1, 1 1 4 1 A 4 1 4

(8). = (5) estimated

والمراجع والمرجع والمرجع والمواجع والمرجع والمرجع والمرجع والمرجع والمرجع والمرجع والمرجع والمرجع والمرجع والم

のためのない

Section 3

= transpose, when used as a superscript T

## TABLE V

Measurement Matrices for Unidentified Landmark Navigation

Observation	Mensurement matrix
Ϋ <sub>κ</sub> = β	$\mathbf{M} = \left[\frac{-1}{\rho} \left(\mathbf{I} - \hat{\rho}\hat{\rho}\right),  \overline{0},  \frac{1}{\rho} \left(\mathbf{I} - \hat{\rho}\hat{\rho}\right)\right]_{3\times9}$
Scalar observations:	
$\mathbf{Y}_{\mathbf{E}} = \tan^{-1} \frac{\vec{\rho} \cdot \hat{\mathbf{S}}_2}{\vec{\rho} \cdot \mathbf{S}_2}$	$\mathbf{M} = \begin{bmatrix} -(\rho_2  \hat{\mathbf{s}}_1 - \rho_1  \hat{\mathbf{s}}_2)^{T} / (\rho_2^2 + \rho_3^2)  ,  \mathbf{\tilde{0}}  , \\ \\ (\rho_2  \hat{\mathbf{s}}_3 - \rho_3  \hat{\mathbf{s}}_2)^{T} / (\rho_2^2 + \rho_3^2) \end{bmatrix}$
$Y_{\rm H} \doteq \sin^{-1}\frac{\hat{\rho}\cdot\hat{\theta}_1}{\hat{\rho}}$	$\mathbf{M} = \begin{bmatrix} -\langle \hat{\mathbf{S}}_1 - \hat{\rho} \rho_1 / \rho \rangle^T / \langle \rho_2^2 + \rho_3^2 \rangle^{1/2} ,  \overline{0} , \\ (\hat{\mathbf{S}}_1 - \hat{\rho} \rho_1 / \rho)^T / \langle \rho_2^2 + \rho_3^2 \rangle^{1/2} \end{bmatrix} = 1 \times 9$

NOTES:  $\vec{\rho} = \vec{L} - \vec{R}$ ,  $\hat{\rho} = \vec{\rho}/\rho$ ,  $\rho = |\vec{\rho}|$   $\hat{\mathbf{s}}_1 \hat{\mathbf{s}}_2 \hat{\mathbf{s}}_3 =$  unit vectors along vehicle axes 1,2,3  $\vec{L} =$  estimated landmark location vector  $\vec{R} =$  estimated vehicle position vector  $\rho_1, \rho_2, \rho_3 =$  components of  $\hat{\rho}$  along vehicle axes 1,2,3  $\vec{0} = (0, 0, 0)$ 

## TABLE VI

Conditions for Computer Runs of Figures 14 and 15

Orbit: 100-nautical-mile circular lunar orbit

Gravitational parameter	$1.7313 \times 10^{14} \text{ft}^3/\text{sec}^2$
Semi-major axis	1038.5 nautical miles
Eccentricity	0,0
Instination to lunar equator	0.0 deg
Start time, to	0.0 hr

Measurements: Unit vector from vehicle to landmark

0.258 hr
0.003 hr
0,999 hr
0.001 rsd, r.m.s., actual

Landmark estimation method: one-sight

Terrain altitude variation

0.250 nautical mile, r.m.s., actual 0.250 nautical mile, r.m.s., a priori





 $\hat{\mathbf{x}}, \hat{\mathbf{x}}, \hat{\mathbf{x}}, =$  unit vectors east, north, and vertical at the station  $\hat{\mathbf{x}} \cdot \mathbf{v}$  ector from center of earth to vehicle.  $\cdot (\mathbf{x}, \mathbf{y}, \mathbf{z})$   $\hat{\mathbf{v}} \cdot \mathbf{v}$  velocity of vehicle relative to center of earth  $\cdot (\dot{\mathbf{x}}, \dot{\mathbf{y}}, \dot{\mathbf{z}})$   $\hat{\mathbf{x}}_{\mathbf{G}}^{*}$  vector from center of earth to station,  $\cdot (\mathbf{x}_{\mathbf{G}}, \mathbf{y}_{\mathbf{G}}, \mathbf{z}_{\mathbf{G}})$   $\hat{\mathbf{r}}_{\mathbf{G}}^{*}$  vector from station to vehicle,  $\cdot \hat{\mathbf{x}} - \hat{\mathbf{x}}_{\mathbf{G}}$  $\hat{\mathbf{x}}_{\mathbf{G}}^{*}$  station geodetic latitude

Fig. 3 Ground radar geometry



VIEW IN ORBITAL PLANE

396

VIEW NORMAL TO ORBITAL PLANE

Fig. 4 Horizon and horizon tracker geometry

A E(VERTICAL) LANDMARK æ , STAR PR PL P VEHICLE EL NORMAL TO ORBIT  $TR = TAN^{-1}(-\rho,\zeta/\rho,\eta)$ TR EL = SIN<sup>-1</sup>  $(\rho, \xi)$ ŝ (TRACK) UNIT VECTORS DESCRIBED IN ^ 7 FIG.2 Ŷ  $\overline{P_L}$  = vector from vehicle to Landmark

397

N. CONTRACTO

 $\beta_s$  = UNIT VECTOR FROM VEHICLE TO STAR TRL, ELL = TRAIN ANGLE, ELEVATION ANGLE TO LANDMARK TR<sub>s</sub>, EL<sub>s</sub> = TRAIN ANGLE, ELEVATION ANGLE TO STAR.

Fig. 5 Landmark sensor, star tracker geometry



Fig.6 Geometry of radar altimeter measurement



「おおおい」という

の影響やすると見た



Fig.8 Partitioning of Qg





Fig. 9 Geometry of local vertical to inertial transformation



Fig. 10 Typical mavigation error plot for both ground and on-board observations

• • •

509

XOU:

Row:

11

Line Printing Di













**Mentilitien** 





10 1 13 2

In Alight The

IN CALERA Meren

Fig. 16 Actual and estimated position error for complete initialization of covariance matrix at each landwark, earth orbit





ic (

1 200



Fig. 19 Accuracy against terrain errors, earth orbit

# CHAPTER 16

TAL:

11 1 1 1 1 1

and Award

PROVINCIAL

W This may

die Da

# SECTION I - APPLICATION OF KALMAN FILTERING FOR THE ALIGNMENT OF CARRIER AIRCRAFT INERTIAL NAVIGATION SYSTEMS

## by

James T.Koubs

Litton Systems, Inc., Woodland Hills, California, USA



## CHAPTER 16

## SECTION I - APPLICATION OF KALMAN FULTERING FOR THE Alignment of carrier Aircraft Inertial Navigation Systems

James T. Kouba

## 1. INTRODUCTION

Same raise

in-unum-second hospital harden harden harden harden

Because missions generally flown by military sircraft require covertness in a hostile environment, the need for a self-contained passive navigation system has long been recognized. Being self-contained and passive means respectively that the system does not have to rely on any external equipment such as LORAN, Omega, etc., nor radiate to perform the navigation function. Since covertness is not necessarily required for all phases of a mission, the military navigation system generally comprises of a multitude of subsystems, including active and non-self-contained elements, so that maximum use is made of all available information. However the system invariably has and requires the inherent capability of performing the navigation function in a self-contained, passive manner.

This requirement is emphatic for aircraft operating from an aircraft carrier. The military functions performed by these aircraft require flight over water for extended periods of time without the benefit of external aids to navigation and often in a hostile environment. Additionally these aircraft are required to return to the aircraft carrier, which may be concealed by weather and electromagnetic silence. Without a self-contained, passive navigation system these mayait factios would be impossible.

The only navigation system satisfying these requirements is the inertial navigation system. It provides in a self-contained, passive mannor all the required druise navigation information. It is, however, recognized that certain functions to be performed will require contact in one form or another with the outside world, so that the inertial mavigator cannot be the only element of the mavigation system.

Since the performance of an inertial navigator is highly correlated with the initialization accuracy, it is imperative that a highly reliable and accurate initialization process be devised. The recognition of this problem and the need for its solution as applied to carrier aircraft has, in the past few years, led to intensive research and analysis to resolve the multitude of operational and performance problems associated with it. The following sections discuss and analyze the carrier aircraft initialization process and logically develop the requirements. Because of the classified nature of the problem, the material will be limited to a qualitative description of the processes involved. The actual quantitative performance requirements and the detrils of the solutions are omitted.

Section 2 is a brief review of the essential features of inertial navigation and includes a generalized mechanization diagram for four different local level mechanizations. An essential ingredient in the design of a Kalman filter is the error model of the system being controlled. To this end Section 3 presents a generalized error model for the four mechanizations described in Section 2. Section 4 defines the initialization process and Sections 5. 6. and 7 discuss the multitude of problems encountered in carrier deck, catapult, and in-flight alignment. Finally the chapter concludes by illustrating the various features of a Kalman filter for carrier deck alignment.

#### 2. THE INERTIAL NAVIGATION SYSTEM

The essential elements of the inertial navigation system are the instrument cluster (platform) (consisting of gyros and accelerometers) that is gimbaled to the airframe and a digital computer. The instrument cluster's geographic orientation is controlled by torquing signals supplied by the computer, which in turn accepts the outputs of the accelerometers and computes yelocity and position.

More specifically consider a right-handed orthogonal coordinate system (x, y, z) associated with, and rigidly attached to, the instrument cluster. The accelerometers and gyros are arranged so that

(1) the positive sensitive axes of the accelerometers coincide with the x, y, x axes,

(ii) torquing signals applied to the gyros will selectively produce angular rates of the platform about the x, y, x axes,

For mircraft inertial navigation systems a local-level mechanization is generally employed. In this mechanization the platform is controlled, ideally, via its gyros so that its z axis is always collinear with the local perpendicular of the reference good, independent of the motion of the mircraft. There ar four different types of local-level mechanizations that can be considered. One of these, called north-pointing, is such that the y axis of the platform ideally slways points north (x axis points east and x up). For the remaining three mechanizations, called wander azimuth, the y axis is not constrained to remain north but is permitted to "wander" at a



controlled rate relative to north. The computer in turn computes the wander angle by essentially (not directly) integrating the wander angle rate. Actually the wander angle, latitude, and longitude are all implicitly computed by solving a set of differential equations for a set of direction comines.

Comparatively the north-peinting mechanization is considerably simpler than the wander azimuth mechanizations. However it does have the disadvantage of not having a polar onpublility. This results from the fact that, as the navigation system approaches the pole, the azimuth gyro (z gyro) torquing rate required to maintain the platform north-pointing because excessively large. Such is not the case for the wander azimuth mechanization since the platform is not constrained geographically.

In the sequel it will be shown that the wandor azimuth mochanization is particularly advantageous for the solution of the alignment problem while the sircraft is on the flight deck of the sircraft carrier. This results principally because the wander azimuth platform can initially be at any angle relative to north and the computer's wander angle adjusted to agree with it.

A complete, detailed derivation of these four local-level mechanizations can be found elsewhere<sup>1</sup>. For future reference Figure 1 diagrammatically presents a generalized mechanization block diagram for local-level free-institunavigation systems. Although it theoretically represents all four local-level mechanizations, in actual practice many variations are used. This is especially true for the north-pointing mechanization wherein many simplifications arise and, instead of computing direction cosines, latitude and longitude are obtained directly. For simplicity the vertical channel mechanization has been omitted. Table I lists the definitions of the symbols used in Figure 1, and Table II lists the vertical platform angular rates for the various mechanizations. It should be noticed that for any of the wander azimuth mechanizations the explicit reduction of the direction cosines to latitude, longitude, and wander angle is required for display purposes only.

The next section presents an error model for these mechanizations,

#### 3. THE INERTIAL ERBOR MODEL

The object of the initialization process is to remove, insofar as possible, the deviation of the actual system from an ideal, error-free system. In general the estimation of the deviation is made by comparing (subtracting) a set of reference variables with (from) a set of communaurate variables computed by the actual system and inferring from these all deviations of interest. The inference process requires a model relating the quantities that are observed with all quantities of interest.

Adequate estimate and control of the initialization process must take into account both the deviation of reference variables and the system variables from the error-free system variables. Thus a necessary requirement for the design of an alignment technique is a model of the error propagations of both the inertial system and the reference system, i.e. the differential and algebraic equations governing the deviations of the actual systems from the ideal, error-free systems.

The reference system will vary, depending upon what is available and desirable to use at the time the initialization process is being invoked. For example, if the initialization process is carried out while the aircraft is on the ground, no external information is required except initial position, and the initialization process is carried out in a completely self-contained manner. This is the case because the a priori knowledge that the aircraft is stationary on the ground implies that the velocity is zero (to within the small oscillations caused by wind buffeting) so that the reference variables are chosen to be the initial manually inserted position and zero velocity. In contrast, if the initialization process is carried out on the deck of an aircraft carrier, the velocity and position of the aircraft will vary as a function of time because of the ship's motion. In this case a system external to the aircraft is required so that adequate information is available to describe the motion of the aircraft system.

Of singular interest here is the error model for the aircraft inertial system, since it is required in all cusos. Figure 2 displays a generalized error block diagram for the local-level free-inertial navigation system. It is a linear error model durived by essentially subtracting the error-free system equations from the actual system equation and retaining only first-order terms. For simplicity the vertical channel has been omitted. As can be seen, all four local-level mechanizations are included in this figure. And, as might be expected, various simplications can result by choosing one particular mechanization of interest; e.g., for the free azimuth mechanization,  $\rho_{\rm x}=0$ . Also it should be noted that  $\delta\rho_{\rm x}$  can be eliminated from the error model, since the errors  $\psi_{\rm x}$  and  $\delta \theta_{\rm x}$  always appear in the form  $\phi_{\rm x}=\delta\theta_{\rm x}=\Psi_{\rm x}$  as inputs into the various nodes.

At times it may be of interest to know the latitude, tongitude, and wander angle errors themselves. For nonpolar regions they are given by

$$\delta \lambda = \lambda_{n} = \lambda_{n} = (\delta \theta_{n} \sin \alpha + \delta \theta_{n} \cos \alpha) \sec \phi \qquad (3.1)$$

Secondary of

$$\delta \phi = \phi_{\mu} - \varepsilon_{\mu} - \varepsilon \delta \phi_{\mu} \sin \alpha - \delta \theta_{\mu} \cos \alpha \qquad (3.2)$$

$$\delta \alpha = \alpha_{\mu} - (\omega_{\mu} = \partial \theta_{\mu} - \delta \lambda \sin \psi, \qquad (3.3)$$

when the subscript C means computed and the subscript T means true (error free). For polar regions the relationships become nonlinear functions of  $\delta \partial_{\mathbf{x}}$ ,  $\delta \partial_{\mathbf{y}}$  and  $\delta \partial_{\mathbf{x}}$  because  $\delta \lambda$  and  $\delta \alpha$  do not remain small.

In the derivation of this error model, which is not presented here<sup>2</sup>, three coordinate systems are used:

(1) True, defined by the error-free system.

- (ii) Platform, defined by the notual instrument cluster.
- (iii) Computer, defined by the actual computed values of the direction comines.

To maintain a linear error model, the angular separation between these coordinate systems must remain small (less than 10 degrees).

The angular separation between the true and the computer coordinate systems is called the position error,  $\delta\theta_x$ ,  $\delta\theta_y$ , and  $\delta\theta_z$ . That is the position errors (angular) are the small angular rotations  $\delta\theta_x$ ,  $\delta\theta_y$ , and  $\delta\theta_z$  about the true x, y, z axes necessary to bring the true axes into coincidence with the computer axes.

The angular separation between the true and the platform coordinate system is called the platform tilt error,  $\phi_{\mathbf{x}}$ ,  $\phi_{\mathbf{y}}$ , and  $\phi_{\mathbf{x}}$ . That is the tilt errors are the small angular rotations  $\phi_{\mathbf{x}}$ ,  $\phi_{\mathbf{y}}$ , and  $\phi_{\mathbf{x}}$  about the true  $\mathbf{x}$ ,  $\mathbf{y}$ ,  $\mathbf{x}$  axes necessary to bring the true axes into coincidence with the platform axes.

The angular separation between the computer and the platform is defined by the angles  $\Psi_{\chi}$ ,  $\Psi_{\chi}$ ,  $\Psi_{\chi}$ , which are given by the components of the vector difference  $\Psi = \phi - \delta \theta$ . That is the computer-platform errors are the small angular rotations,  $\Psi_{\chi}$ ,  $\Psi_{\chi}$ ,  $\Psi_{\chi}$ , about the computer x, y, a axes necessary to bring the computer axes into coincidence with the platform axes.

The velocity error and the platform angular rate error are defined by the vector difference between the computed values and the true values, i.e.,

$$\frac{\delta \mathbf{V}}{\delta \mathbf{V}} = \mathbf{Y}_{\mathbf{U}} - \mathbf{Y}_{\mathbf{T}} \tag{3.4}$$

$$\frac{\delta \rho}{\rho} = \rho - \rho_{\rm T} \,, \tag{3.5}$$

Table III lists the definitions of the remaining variables appearing in Figure 2.

## 4. THE INITIALIZATION PROCESS

The initialization process for the institut navigation system requires both the initialization of the mechanization differential equations with the proper velocity, position, and wander angle and the physical alignment of the platform to the required geographically referenced orientation. Each type of mechanization considered here requires (ideally) that the platform be aligned so that its z axis coincides with the local perpendicular of the reference geoid. The azimuth orientation for north-pointing systems requires that the y axis of the platform point north: whereas for the wander azimuth systems a dichotomy exists in that the platform azimuth must agree with the computed wander azimuth angle or vice versa.

In principle the initialization process could occur before, during, or after the aircraft is catapulted from the carrier. In each of these cases the aircraft is in motion relative to the earth, so that some internal or external (to the aircraft system) reference sensor is required. Without any further definition or restriction on the availability of reference sensors, each case can have a multitude of different possible solutions depending upon the reference sensors chosen. To limit the ensuing discussions it is therefore necessary to limit the choice of reference sensors considered. For alignment while on the warrier deck the reference sensor chosen is the ship's inertial navigation system (SINS). Alignment during detapulting is preceded by position and velocity initialization and fine leveling, using SINS data, and employs ship heading or transponder ranging to perform the azimuth alignment while catapulting. For in-flight alignment the reference considered is the Doppler radar velocity sensor. A discussion of each of these techniques is given in the following sections.

Before examining these techniques it is necessary to discuss the quantitative meaning of alignment. Since in any case the initialization process, after a practical amount of time, will result in a residual deviation from the ideal (error-free) system, it is fair to ask when is a system considered to be aligned? To answer this question it must be realized that what is actually of interest is the accuracy of size subsequent navigation. Thus it is possible to quantitatively define alignment implicitly by specifying the accuracy of the subsequent navigation. However this is usually not done, and a redundant specification of both the alignment and navigation accuracy is made. Incompatibility does not present a problem, because the alignment accuracy is derived from the navigation accuracy; and there exists a high correlation between the two. Thus the system is considered to be aligned when it has not the quantitative specification dictated by the user.

## 5. CARRIER DECK ALIGNMENT

It is generally recognized that alignment while on the deck of the carrier will produce the most accurate ensuing navigation. However, because of the complexity of handling the multitude of aircraft on the flight deck, .





a free and the second and the second s



No 1946 ta Maria a Partika 🖤 Alimita



severe restrictions are placed on the acceptable solutions to the alignment procedure and technique. Thus, in the design of an alignment technique while the aircraft is on the carrier deck, full account must be taken of the operational, environmental, and control constraints imposed.

Before the strict imposition of the operational constraints two techniques had been proposed that, for historical reasons, are worth mentioning. The first technique, called trunsfer alignment<sup>3</sup> employs a portable master inertial system physically and electrically mated with the mircraft system. The close proximity of the two platforms and the angular precision of the mating permit accurate gimbal matching for minute alignment and ucceleration matching for level alignment. The second technique, called insertion<sup>4</sup>, employs a portable mircraft inertial navigutor (PINS) that is prealigned at a shiphoard console. By including a battery and sufficient computational capability in the portable unit, it can be removed from the alignment console, carried to, a waiting mircraft, and inserted, ready for operation.

As will be seen, the operational constraints imposed rule out those techniques as possible solutions and implicitly require that the alignment take place with the inertial navigator in the alreraft. The combination of the three aforementioned construints determines the form and size of the error controller employed in the alignment technique. A discussion of each of these constraints and their rumifications follows.

### 5.1 Operational Constraints

To avoid interference with any of the normal deck-handling procedures such as fueling, arming, and taxing, the alignment is to be initiated and completed without any manual intervention or restriction on the location and movement of aircraft. This succinctly summarizes the operational constraints. Two immediately obvious ramifications are that no power cable and no data cable can be used for the alignment technique. In turn, the lack of the power oable requires that the alignment be achieved during the apoch of engine turn-on to catapulting. Also, solutions involving special alignment stations on the carrier deck to afford sufficient time with engines on for the alignment to be completed are implicitly ruled out, i.e., no lotering for the express purpose of alignment is permitted since this would interfere with the normal deck-handling procedures.

Without an additional specification the time allotted for alignment is ambiguous. Therefore, after consideration of the normal movements of an aircraft subsequent to engine turn-on, an alignment time is specified (by the user) excluding any taxing periods.

The imposition of a specific alignment time results in an instrument performance problem<sup>5,4</sup>. Both the accelerometers and gyros will function within specification if they are operated within a relatively narrow range about a design temperature. This normal operating temperature is, for proment day instruments, considerably above expected ambient temperatures. Thus, when the platform is initially turned on (energizing the heaters), the instruments will cycle through a thermal transient until equilibrium is achieved. During this thermal transient, the instruments will exhibit transient drift and bias characteristics that are large enough and last long enough to degrade the alignment process.

Additionally other portions of the system will exhibit undesirable characteristics during this thermal transient phase. The cluster structure may exhibit excessive warpage due to thermal gradients and this may result in instrument non-orthogonalities and consequently degrade alignment. Also the multitude or electronic packages required to perform such functions as pulse shaping and scaling for gyro torquing, accelerometer proof mass restoration, and formation of the velocity output pulses, may exhibit abnormal behavior during the thermal transient.

The solution to these problems requires a careful mechanical and thermal design of the various instruments, structures and electronic packages involved so that the resulting undesirable thermal transient behavior, if not reduced to a negligible value, is repeatable and predictable as a function of suitably measured temperatures.

Since the aircraft system requires the SINS data to accomplish alignment and no data cable is permitted, it is necessary to resort to a telemetering system to obtain the required information. This leads to a problem relating to the existence of an electromagnetic radiator on the aircraft carrier. It is obvious that at times it would be deemed necessary to maintain the carrier electromagnetically silent. If these times of silence coincide with the times that alignment is to be performed, an incompatibility arises. The solution to this problem is one of more specifically defining what is meant by electromagnetic silence. Electromagnetic silence is desirable to avoid the possibility of the carrier revealing its presence to an enemy. Certainly, if the radiation pattern and power are so designed that, for all practical purposes, it is detectable only on the carrier deck, use of an electromagnetic radiator during a period of silence will not endanger the corrier. Thus appropriate achievable design constraints on the telemetering system will increase its utility to cover all times.

Additional ramifications do result because of the operational constraints imposed. However, they result because of the coupling of the operational, environmental and control constraints and will therefore be discussed as they arise.

#### 5.2 Environmental Constraints

Carrier deck alignment is to be performed while the carrier is pitching, rolling, yawing, heaving, surging, and swaying. In addition to these motions the carrier may be making heading changes to perform evasive manuavers or to head into the wind in anticipation of launching or receiving alloraft. Additionally the alroraft itself will be buffeted by winds over the flight deck, thus causing motion that is not sensed by SINS. The deaign of the alignment technique requires that each of these environmental conditions be taken into account.

410

5

The principal effect of these motions is related to the reference data used for the alignment process. The reference data supplied by the SINS to the aircraft system via the data link consist in part of position and velocity, as measured by the SINS itself. Since the aircraft may be situated anywhere on the flight deck, the SINS reference data. The data required to perform this compensation consist of the vector distance between the SINS and the aircraft plus the carrier's pitch, roll and heading, and their rates. (See Figure 3.) Since the SINS inherently provides these angles and angular rates it is reasonable to require that these be included in the data telemetered to the aircraft. On the other hand the distance between the SINS and the aircraft will vary from aircraft to aircraft and even change for a given aircraft if, during alignment, it taxies to a new deck position. Prohibition of the manual insertion of lever arm components, a restriction of the lever arm components.

#### 5.3 Control Constraints

1.2.2.2

5744 S

\*\*\*\*\*\*\*\*

15444444

The control constraints are concerned with the physical limitations of the instruments involved and the limitations imposed by the present state of development of the control theory to be used to perform the alignment.

The primary physical limitation of the instruments that affects the alignment problem is the maximum torquing rate that can be applied to the gyros. In reality a dual phenomenon is of concern. First there exists an absolute maximum torquing rate that, if exceeded, will essentially destroy the gyro torquing coils. Second, as this maximum is approached from the lower side, the accuracy with which the desired angular rate is achieved decreases, i.e., torquing nonlinearities become significant.

In the application of modern control theory to the carrier deck alignment problem the limitation that significantly affects the solution is the imposition of linearity. The reason for this will become apparent in the following discussion.

The gyro torquing limitation coupled with the hands-off operational requirement and the fact that the institument cluster's azimuth is initially completely unknown leads to a nonlinear estimation problem. To understand how this results it is necessary to review the technique of coarse alignment of the instrument cluster.

In order to apply linear control processes to align the platform, it is necessary to bring the platform close to the desired orientation, so that linearization of the angular deviations can be assumed. Theoretically linearity can be assumed for angular deviations up to 10 degrees. From a practical point of view it is desirable to achieve a considerably smaller deviation, since then the required angular rates to align have more reasonable magnitudes. For sircraft systems the platform can be approximately leveled by nulling out the pitch and roll simbal readouts. This process is carried out while the gyros are being brought up to speed by simply torquing the gimbals with their own synchro readout signals. This technique is sufficiently accurate to justify linearization of the control problem, since it results in a deviation of approximately 1 degree from level for sircraft on the ground and 2 to 3 degrees for sircraft on the deck of a carrier. The larger level coarse alignment errors on the carrier result from the pitching and rolling of the ship.

The azimuth of the platform is usually slewed to approximately north (and for wander azimuth mechanizations the computer wander angle is initialized at zero) by torquing the azimuth gimbal synchro with the difference between the output of the azimuth gimbal synchro and a synchro signal obtained from a magnetic compass indicating the magnetic morth heading of the aircraft. Additionally the magnetic heading can be compensated for magnetic variation to make the output a substant more azimuth alignment more accurate.

After the coarse alignment phase has been completed and the gyros are brought up to speed, the platform is placed under the control of the gyros. At this time the fine alignment phase can begin, or siternatively additional coarse alignment in level can be achieved by simply torquing the gyros with the output of the accelerometers. The additional coarse level alignment reduces the design problem for the fine alignment phase, since it reduces the dynamic range over which it must operate.

The application of this technique for shipboard coarse alignment unfortunately fails. This is because the magnetic heading device does not function with sufficient accuracy while in the vicinity of the massive steel deck and superstructure of the aircraft carrier. Thus, although the platform can be brought to a reasonably small coarse level alignment, no direct information is available to slew its azimuth to an approximate geograph's referenced direction. If manual insertion of data were permitted the problem could be easily solved by visually estimating (to within 10 dogrees) the angle between the ship's heading and the aircraft's heading (the spotting angle, see Figure 3). This angle, along with the ship's heading from SINS, would suffice for coarse azimuth alignment by either slaving the azimuth gimbal to a pre-assigned direction (such as north) or by initializing the computer wander azimuth angle.

It should be pointed out that if the mechanimation is such that the platform is required to have a specific asimuth direction (as, for instance. In the north-pointing mechanization) then the platform must be close to that desired azimuth before the gyros have assumed control. The resson for this is that, because of the restriction on maximum gyro torquing, an excessive amount of time would be consumed if it were required to torque the platform through a large angle.

The solution to this problem is achieved by choosing one of the wander azimuth mechanizations and employing, after the gyros have assumed control, a nonlinear estimation process to estimate the platform's wander angle and initialize the computer.

411

In North Chr.

ł

For the ground alignment case (zero reference velocity) the wander angle of the platform is estimated from the residual velocity error after leveling has achieved steady state<sup>7</sup>. A similar procedure can be used for carrier deck alignment, but it must be dynamical, since the aircraft will be accelerating because of the carrier's motion.

This section has examined carrier dock alignment, detailing the multitude of ramifications of the operational, environmental, and control constraints imposed. In summary, carrier deck alignment must take place to a specified accuracy and within a specified time under the following conditions:

- (i) No power cable,
- (ii) No data cable,
- (iii) No manual insertion of spotting angle.
- (iv) No manual insertion of lover arm components.
- (V) No loitering for the express purpose of alignment.
- (vi) No restriction on location.
- (vii) Taxiing permitted.

In Section 8 the filtering techniques employed in the solution of this problem are discussed.

## 6. CATAPULT ALIGNMENT

The most time-consuming phase of the initialization process is the szimuth alignment of the platform, principally because of the rate at which the azimuth information propagates into measurable observables. In an unaccelerated environment the szimuth missiignment propagates into the lovel channel at a rate proportional to either earth rate or spatial rate. This represents a relatively weak coupling; and if the measurable observables are contaminated with extraneous information, considerable time must be spont extracting the missignment. In an accelerated environment, the azimuth misslignment propagates into the level channel proportional to the level accelerated environment, the azimuth misslignment propagates into the level channel proportional to the level acceleration which, if large enough, provides a much stronger coupling and reduces the azimuth alignment time. This line of reasoning leads to the concept of performing the azimuth alignment portion of the initialization process while the micraft is being enough from the carrier.

PUT AND THE P

Several different approaches have been devised to perform catapult alignment. One approach, developed by NADC<sup>\*</sup> Johnsville, compares the velocity increment direction as computed by the aircraft system with the catapult direction as obtained from the SINS-computed ship's heading plus the angle between the ship's heading line and the catapult. A second approach employs two transponders mounted on the carrier and compares the aircraft computed and transpondermeasured relative position to estimate the azimuth misalignment. Various modifications and generalizations of these two techniques are obviously possible.

Before catapulting the siroraft both techniques require the same amount of preparation, i.e. alignment up to and including fine leveling. If possible it is desirable to adhere to all the previous operational constraints. However, at its present state of development, the velocity increment approach requires one manually inserted piece of data: the catapult from which the miroraft is to be launched. This results because the technique requires the angle between the ship's heading line and the catapult. Since there are four catapults, a simple manual switch aslection is necessary so that the computer can delect the correct pre-stored angle. Other than this exception the techniques are encumbered with the same tasks as carrier dock alignment, i.e. lever arm estimation, automatic coarse azimuth alignment, relative velocity compensation, permission of aircraft movement, and fine level alignment. It should be pointed out that, since the maintain alignment is delayed until catapulting occurs, the thermal transient problem is somewhat relieved in that the platform will have more time to reach thermal equilibrium.

The transponder technique, albeit requiring additional equipment both on the aircraft and on the aircraft carrier, does not require any manual intervention. In fact the problem of lever arm estimation is reduced to simply computing the aircraft's location from the transponder data received, thus relieving the alignment technique of the burden of implicit estimation of the lever arm.

With respect to the velocity increment approach, there are several significant effects that take place during catapulting, which must be considered in the design of the technique and in ussessing the accuracy of the process. The more significant of these will now be discussed.

While being catepulted the aircraft is forcibly constrained to the dock and to follow the catepult track. However, because the track will not precisely constrain the lateral motion of the aircraft, an error in the reference direction arises. This is because the premise of the technique is that the aircraft (actually its platform) moves along a precisely defined line relative to the ship. Inherently the technique has no method for compensating or measuring this motion. The problem is additionally complicated by the fact that the lateral motion of the wheel in the track can possibly cause an emplified lateral motion of the platform depending upon its location in the mircraft. Without a technique for compensating or measuring this motion it represents an irremovable error.

Superimposed on the lateral track motion is the motion of the mircraft's inertial system due to the ship's pitching, rolling and yawing (during the launching of mircraft the heading of the ship is wilfully maintained

" US Naval Air Development Center, Johnsville, Warminster, Pennsylvania,

constant). Since the inertial system measures earth-referenced velocities, it will be sensitive to the velocities induced by these angular motions coupled through the lever arm. This motion can easily be compensated for by using the pitch, roll, and yaw rates as provided by SINS through the telemetering system and the estimated lever  $\gamma$  a.

"inally it should be recognized that the accuracy of the ship's heading and the surveyed angle of the catapult (near prominently and directly into the accuracy of the azimuth alignment achievable. Senility of the ship's novading data is also of concern. Any lags induced because of the finite data rate of the telemetering system will induce errors proportional to the angular rate. Senility can be compensated for by increasing the data rate or introducing prediction techniques before using the data.

The transponder technique similarly has several effects to be considered. Since it is utilizing information at the position level to estimate the azimuth misalignment, it is not plagued by the irremovable catapult error nor the necessity for relative velocity correction. However, since the position as measured by the transponder is relative to the ship's body coordinates, it is necessary to resolve the measured position into the local-level coordinate system. This, of course, requires the pitch, roll, and heading of the ship.

Because of the relatively short base line available for the transponding system, small-distance errors can lead to relatively large azimuth errors. Thus it is important to obtain an accurate survey of the transponders relative to the SINS. Another significant possible source of error arises because the transponder's measured position is not precisely the position of the platform. However, this displacement error, to a large extent, can be compensated for by using the inertial system data.

The single must significant drawback of the transponder technique is the need to add additional equipment to the already overcorowded aircraft avionic system and to require an additional radiator to the aircraft carrier. Design constraints, similar to those imposed on the telemetering system, are required to maintain safety and security under electromagnetic silence.

#### 7. IN-FLIGHT ALIGNMENT

Appreciating the numerous difficulties associated with carrier deck and catapult alignment it is worthwhile to consider that alignment be delayed until after the Aircraft has left the carrier. This delay would eliminate the problem of interference with the deck handling procedures and significantly reduce the problem associated with the platform thermal transients. However it is encumbered with problems not heretofore considered. The discussion will be limited to performing in-flight alignment using a Doppler radar as the reference system. The details of the synergistic Doppler-Inertial system have been adequately discussed in Chapter 11, so that only problems peouliar to the carrier aircraft application will be considered here.

To establish the state of the navigation function and the inertial platform at the time in-flight alignment begins, it is necessary to discuss the sequence of events relative to the navigation process from engine turn-on until the start of alignment. It is apparent that this epoch should be kept as small as possible since relatively poor, if any, navigation would be available and would introduce position errors that are irremovable without additionally complicating the navigation function by requiring position updating.

At the time in-flight alignment begins two conditions must be met. First, since it is used as the reference system during alignment, the Doppler must be functioning properly. This is specifically pointed out because a commun, although not necessary, characteristic of a Doppler radar is that it will not track the aircraft velocity, until it has exceeded some specified minimum altitude. Thus, until the aircraft has attained this altitude and the Doppler begins to track the velocity, slignment cannot begin. Second, the platform must be coarsely aligned so that the Doppler data can be transformed, at least approximately into the navigation coordinate system and also to permit linearization of the alignment process.

One technique for achieving these two conditions is to require the aircraft, after leaving the carrier and climbing to a safe altitude, to fly straight and level. This flight condition is maintained long enough to achieve coarse platform alignment and set the platform under gyro control. Coarse alignment is performed by nulling the pitch and roll gimbals and slaving the azimuth gimbal to magnetic north. This procedure has several disadvantages. It increases the time before the beginning of alignment and thus the time during which relatively poor navigation is performed. Also it places a constraint on the maneuvers that the aircraft can perform after takeoff from the carrier. Jastly for the pilot to fly straight and level he must have some reference of the vertical. This could be the horizon, if visibility permits, or some ancillary device.

A second technique would be to perform a coarse alignment on the carrier deck and, after the gyros have assumed control of the platform, to maintain the coarse alignment by pendulous erection and magnetic heading slaving. This procedure avoids the necessity of restricting the aircraft's maneuvers and permits alignment to begin as soon as the Doppler begins tracking the velocity. Careful consideration must be given to the mechanization of this technique, since the pendulously ersoted platform will tilt exorbitantly during large horizontal accelerations (catapulting and turning) and the magnetic heading device, as mentioned previously, will have large errors while on the carrier deck.

Excessive platform tilting can be avoided by inhibiting erection when the level accelerometer outputs exceed some specified value. The specific value would be ascertained by an analysis of the error dynamics. The coarse
414.

azimuth alignment problem is handled as follows. Since an erroneous magnetic north will be indicated on the deck of the carrier, completion of coarse azimuth alignment must wait until the aircraft has left the ship. However, at this time large azimuth changes of the platform cannot take place within a rensonable amount of time because of the limit on gyru torquing. It is therefore necessary to employ a wander azimuth mechanization and perform coarse azimuth alignment by proper adjustment of the computer wander angle and not the platform. Thus, while on the carrier duck, the platform is at first coarsely aligned to the aircraft heading; and, after the syros wars resumed control, the computed wander angle is slaved to the difference between the platform heading and "wilcated magnetic heading (see Figure 4). Since large alterations in the computed wander angle can easily be made, no particular difficulty arives when the indicated magnetic heading changes significantly as the aircraft leaves the aircraft carrier.

The foregoing discussion establishes the state of the navigation function and the inertial platform at the time in-flight alignment begins. The problems associated with the over-water alignment process have been adequately discussed in Chapter 11 and will be briefly reiterated here:

- (i) Distortion of Doppler spectrum return over water.
- (ii) Loss of Doppler return over very smooth water surface.
- (iii) Doppler radar measures velocity relative to water current rather than earth.

(iv) Alignment process squarement in form to ground alignment and therefore requires a relatively long time.
 (v) During alignment, navigation pollumination is degraded relative to a pre-aligned system.

Bocause of the unpredictability of the sea conditions and the length of time required for in-flight alignment, there is a preferous to perform the alignment on the carrier deck.

#### 8. CARCIER DECK ALIGNMENT FILTER

As pointed out in Section 5, the application of a Kalman filter to the carrier deck alignment problem must be preceded by an estimation of the wander azimuth angle. Because of complete lack of knowledge of the azimuth orientation of the platform, the initial wander azimuth angle is uniformly distributed between  $-\pi$  and  $\pi$ . An extension of the error model presented in Section 2 to include the unknown wander angle results in a complex monlinear model involving sines and cosines of the wander angle. Thus a direct application of a Kalman filter is not possible.

Because the Kalman filter requires a linear error model, it is necessary to estimate the large-angle azimuth offset by some other meaks. Since the linear range of the filter  $(\sin \theta \simeq \theta)$  extends to  $10^{\circ}$ , a gross estimate can be obtained using a minimum variance estimator<sup>11</sup> comprising a three-state model (azimuth offset, two platform tilts) which relates commensurate aircraft and SINS velocities. The scheme is invoked after pendulous leveling has occurred and the gyros have taken control of the platform. For convenience the vertical platform axis is torqued so that the coarse azimuth offset remains constant. Once the estimate of azimuth offset is obtained, the alignment process is linear, and the standard form of Kalman filter is used to perform the fine alignment.

One of the sajor considerations in the application of a Kalman filter is the number of states to be modeled. As the number of states is increased the performance of the system will improve. However, a point of diminishing returns is moon reached, since the inclusion of relatively insignificant error processes improve the performance slightly but increase the computer requirements enormously. The choice of the number of states for the application under consideration is actually based on a dual use of the filter. The filter is required not only to perform the carrier deck slignment but also to provide the suboptimal error control for Doppler-inertial navigation. The size of the filter is therefore dictated by the mode requiring the most number of states.

Consideration of the significant error processes leads to a choice of 13 states. For the carrier deck alignment process they are

Position errors
Velocity errors
Platform tilts
Asimuth error
Level gyro drifts
Horiscotal lever arms
Reference velocity error

The model for the first seven errors is obtained directly from Figure 2. The level gyre drifts and reference velocity errors are modeled as first-order correlated noise processes. Only the horizontal lever arm components are modeled, since during alignment the aircraft is always on the carrier deck and the vertical lever arm (relative to the body axes) is known. While the aircraft is stationary on the flight deck the body-axis lever arm components are time-invariant, so that the model chosen to represent them is simply a constant. This type of model is the same as a first-order correlated moise process with infinite correlation time.

















Although level gyro bias is included in the carrier align filter, practically no gyro biasing will occur. Because of the high quality of the gyros required and the short time of alignment, the other errors of the system will dominate and not permit biasing to occur. However they are included for two reasons. First, they are required for Doppler-inertial navigation. Second, they can be used for periodic gyro biasing to remove the long-term (order of months) drifts. This periodic gyro biasing would be employed at scheduled maintenance time. Under these conditions sufficient time is available to reduce the errors of the system to such an extent that the biasing will occur.

Periods of taxing are handled as follows: while the aircraft is stationary on the carrier deck its wheel brakes are on. If the aircraft taxis the wheel brakes are released. Thus a simple switch indicating the state of the brakes provides the computer with a discrete. Using this discrete the computer will temporarily suspend alignment whenever the brakes are released and the system reverts to free-inertial navigation. During this time the lever are estimates are altered by integrating the difference between the aircraft computed velocity and the SINS telemetered velocity compensated for lever arm effects. This process will be in error because complete alignment bas not occurred. Therefore during the taxing the filter covariance matrix is propagated to reflect the fact that the errors in the lever arm estimates are changing. Once the brake is reapplied the system reverts to the alignment -process.

The observation process consists of comparing the aircraft computed velocity with the SINS-telemetered velocity compensated for lever arm effects. The errors associated with this process are the errors in aircraft-computed velocity and azimuth, the errors in the SINS velocity, thu errors in the lever arm estimates, and the errors in the SINS pitch, roll, heading and their rates. Since the filter states do not model the errors in the SINS pitch, roll, heading, and their rates, these errors are approximated as a white noise in the observation process.

The task of filter synthesis and performance estimation requires an extensive digital computer simulation. The basic computational modules of the simulation program consist of

(1) the filter,

(ii) the real-world system error model,

(iii) dynamics.

The filter module is a replice of the filter to be programmed on the aircraft's digital computer. As such it will be manipulated to obtain a compromised suboptimal performance. The basic objective is to obtain adequate control of the real world system error model and at the same time minimize the computational complexity. The real-world system error model module consists of the error models of all the various subsystems involved in the alignment process, i.e. the aircraft inertial system, the SINS, the telemetering system, etc. If Nonte Carlo techniques are employed it is not necessary that these error models be linear. The dynamics module simulates all the various motions of the ship and the aircraft. For the carrier deck alignment problem it is important to include the dynamics of the ship's course and its pitching, rolling, and yawing<sup>12</sup>. Additionally it is required to simulate the aircraft's motion on the carrier deck, while it is catapulting and after catapulting.

The simulation process for filter synthesis and performance estimation consists of controlling the real-world system error model with the gains generated by the filter under test for a particular set of dynamics. The literature contains many examples of such studies for problems similar to the one under consideration here  $1^{3-14}$ .

#### 9. CONCLUSION

This chapter has considered the application of Kalman filtering to the problem of alignment of carrier aircraft inertial navigation systems. Considerable detail has been presented to establish the constraints under which the filter is to be designed.

In view of its usage in the solution of the problem, the local-level wander azimuth mechanization and error model were presented. For carrier deck\_alignment the interplay between the operational, environmental, and control constraints were used to define the conditions under which the filter must perform.

Catapult and in-flight alignment were examined as alternative or supplementary techniques. Problems similar to those encountered in carrier deck alignment were found, and the same or alternative solutions were discussed.

Finally a brief discussion of the filter for carrier deck alignment was given.







3.

S. 17.7.3

1.00000 (100000)

3303773 | - 62462234 - 520623

1. Brockstein, A.J.

1968.

1966.

16-18, May, 1966.

Kouba, J.T.

2. Pinson, J.C.

4. Miller, B.

5. Powers, H.B. Jr

6. MoMurran, M.

7. Lindner, B.I.

Henderson, V.D.

#### REFERENCES

Derivation of Free Inertial, General Wander Azimuth Mechanization Equations.

Inertial Guidance for Cruise Vehicles. pp. 113-187 of "Guidance and Control of

Final Engineering Report, Transfer Align Demonstration. US Navy Bureau of Wespons

Contract NOw 62-0993-f, prepared by Litton Systems, Guidance and Control Division,

Faster Carrier Inertial Alignment Sought. Aviation Week, Mouraw-Hill, 21 November,

Analytic Thermal Compensation of Fast Reaction Inertial Navigation Systems.

Fast-Reaction Alignment of Inertial Navigators. Hpace/Aeronautics, July, 1987.

Sensitivity Study in a Estimation and a Updating for a Wander Azimuth System.

Litton Systems, Guidance and Control Division, Publication 5958, 15 February, 1987.

Presented at The National Aerospace Electronics Conference, Dayton, Ohio,

Aerospace Vehicles", C.T. Leondes (Ed.), McGraw-Hill, New York, 1963.

Litton Systems, Guidance and Control Systems Division, Publication 7973, September,

















tetem Omitae fink Diebte



revit Din en birth Dalla



Boty, R.L. Initial Conditions and Alignment. pp.176 and 209 "Inertial Guidance", G.R. Pitman, Nease, R.F. (Ed.), Wiley, New York, 1962.
 Kouba, J.T. At Sea SCRAMALIGN(1). Litton Hystems, Guidance and Control Division, Office Correspondence JTK-6812-174, 9 December, 1968.
 Brockstein, A.J. Minimum Variance Solution to the At-Sea Coarse-Alignment Problem, Litton Systems, Guidance and Control Division, Publication 8907, March 1969.
 Liebelt, P.B. An Introduction to Optimal Estimation. Addison-Wesley, Gambridge, Mass., 1967, p.164.

Reference No. 2113-357, June, 1963.

12. Brockstein, A.J. Design of a Model to Simulate Significant Aircraft Carrier (CVA-59) Motions, Litton Eystems, Guidance and Control Systems Division, Publication 7793, June, 1968.

 Hodson, A.E.
 Optimal Leveling and Gyrocompassing of Inertial Systems. Institute of Electrical and Electronic Engineers, Transactions on Aerospace and Navigational Electronics, March, 1963.

 

 14. Jurenka, F.D.
 Optimum Alignment of an Inertial Autonavigator. Institute of Electrical and Leondes, C.T.

 Electronic Engineers, Transactions on Aerospace and Electronic Eystems, Vol.AES-3, No.6, November, 1967.

Danik, B. Integrated Inertial Velocity-Aided and Position-Aided Aircraft Navigation.
 Presented at the 24th Annual Meeting of the Institute of Navigation, 21 June, 1938.

16. Sutherland, A.A. Jr

The Kalman Filter in Transfer Alignment of Inertial Guidance Systems. Journal of Spacecraft, Vol.5, No.10, October, 1968.



..

#### TABLE I

N

Secondo

A STATE OF A

Definition of Variables for Figure 1

and the second s	
I, Y, H	Navigation system coordinate axes
A <sub>x</sub> , A <sub>y</sub>	Accelerometer outputs
V <sub>z</sub> , V <sub>y</sub> , V <sub>z</sub>	Velocity components relative to Earth
$P_{\mathbf{z}}, P_{\mathbf{y}}, P_{\mathbf{z}}$	Platform angular rate components relative to Earth
ນະ ບໍ່ ບັ	Earth rate components
Ъ	Geodetic altitude
•	Mean equatorial radius of the Earth
1	Flattening of the Earth
¢	Geodetic 1stitude
λ	Longitude
α	Asimuth wander angle
° <sub>rj</sub>	Direction cosines defining latitude, longitude and azimuth wander angle. Specifically
	$C_{xx} = \cos \alpha \cos \lambda - \sin \alpha \sin \Phi \sin \lambda$
	$\mathbf{C}_{\mathbf{x}\mathbf{y}} = -\mathbf{sin} \ \alpha \ \cos \lambda - \cos \alpha \ \mathbf{sin} \ \Phi \ \mathbf{sin} \ \lambda$
	C <sub>xe</sub> = cos Φ sin λ.
	$O_{\chi g} = \sin \alpha \cos \Phi$
	C <sub>Yy</sub> = dos a cos Φ
	$C_{y_{H}} = \sin \Phi$

high S I DESCRIPTION

#### TABLE II

#### Wander Angle Rate and Vertical Platform Angular Rate for Local-Lavel Mechanizations

Hechan	isation	å	P <sub>B</sub>
North-poin (no polar	ting (a=0) capability)	0	$\frac{C_{TT}\rho_{T}+C_{TT}\rho_{T}}{1-C_{TE}^{2}}C_{TE}$
	Free-azimuth	->;c**	0
	Foucault	$= (\dot{\lambda} + \Omega) c_{\gamma_{\Xi}}$	$= \Omega c_{\chi_{III}}$
Wander Azimuth (polar Gapsbility)	Unipolar Upper sign Northern Hemisphere Lower sign Southern Hemisphere	*	$=\frac{O_{Yy}\rho_y+O_{Yy}\rho_z}{(C_{Yy}\pm 1)}$

#### TABLE III

Definition of Variables for Inertial Error Model

$R_{g} = 1/a [1 - h/a -$	$\begin{array}{l} \text{approximation to the}\\ f(1-3C_{TH}^2-C_{Ty}^2)] &= \text{radius of curvature}\\ & \text{ in the } \mathbf{X} \ , \ \mathbf{X} \ \ \text{plane} \end{array}$
$R_y = 1/u [1 - h/n -$	approximation to the $f(1 - 3C_{Ty}^2 - C_{TR}^2)] - radius of curvature in the y , z plane$
A' <u>s</u>	vertical acceleration minus gravity
r e	gravity
⊽ <sub>z</sub> , ⊽ <sub>y</sub>	sacelerometer errors
e <sub>x</sub> , e <sub>y</sub> , e <sub>x</sub>	CYTO STTOTE
<u> = e + u</u>	inertial angular rate

 $n_{L}$ 



ł.

1







#### Fig.2 Generalized error block diagram for local-level free-inertial navigation systems

R. 48.

P\_ 60,















the second





#### A. GEOMETRY

- . NO RESTRICTION ON POSITION OF AIRCRAFT ON FLIGHT DECK
- NO DATA OR POWER CABLE
- NO MANUAL INSERTION OF LEVER ARM
- NO MANUAL INSERTION OF SPOTTING ANGLE
- AIRCRAFT MAY BE MOVED AFTER ALIGNMENT . HAS BEEN INITIATED
- CAPABILITY OF ALIGNING IN FIVE MINUTES
- B. DESIRED OPERATIONAL CHARACTERISTICS

Fig.3 Carrier deck alignment geometry



- FLATFORM HEADING (AZIMUTH GIMBAL SYNCHRO READOUT)

ANTER A CONTRACTOR

- INDICATED MAGNETIC NORTH Ψm

 $= \Psi_0 - \Psi_{n0}$ 

Pig.4 Wander angle slaving to magnetic north



In Carte Martin

يكر والأ

14 1 a CO 1'

#### CHAPTER 16

97 XAMARAA

tein Officiality Ott

#### SECTION II - NAVIGATION AT SEA USING THE INVARIANTS

#### FORM OF KALMAN FILTERING

27

#### N.W. Senbrook\* and H. Halesandaris \*

#### \*Litton Systems Inc., Woodland Hills, California, UMA

<sup>†</sup>Matellite Positioning Corporation 18033 Ventura Eloci Encino, Californis, USA

### PRECEDING PAGE BLANK

NOTATION

Σ, σ	covariance matrices
Ŷ	estimates of state variables
x	state variable matrix
7	aysten dynamics matrix
Q	system noise matrix
Ŧ	internal system noise
H	observation matrix
	observation
۷	mensurement noise
<b>Q</b> .	covariance of system noise
t	time
8 <mark>k</mark>	Kronecker delta
Aij	'position-dependent covariance matrix elements
R <sub>ij</sub>	position/non-position covariance matrix elements
s <sub>ij</sub>	non-position-dependent covariance matrix elements '
σ <sub>ij</sub>	elements of standard covariance matrix
Ĺ	transposed state variable matrix
υ <sub>r</sub> , υ <sub>s</sub>	matrices used in covariance matrix update
$\sigma_{\rm H}, \sigma_{\theta}, \sigma_{\theta}^{\pm}$	standard deviation of measurement noise
a <sub>p</sub> , a <sub>s</sub>	transposed observation matrix
K <sub>r</sub>	Kalman gains
Superscript -	indicates before fix
Superscript +	indicates after fix
Bubscripts X, Y, X,	platform axes
φ <sub>t</sub>	true latitude
λ <sub>T</sub>	true longitude
$\phi_{\mathbf{s}}$	asimuth angle error
έφ	latitude error .
8 <b>λ</b>	longitude error
۷ <sub>px</sub>	platform x-velocity
۷ <sub>۵۶</sub>	platform y-velocity
n.	earth rotation vector
Ω <sub>N</sub> , Ω <sub>L</sub>	north component of earth rate
Ω <sub>¥</sub>	vertical component of earth rate

an gel Cherner and Plymer

The Arris

			hing Marines
		425	
0	platform rearis component of south rate		
··•π, · Ω	platform where component of anyth rate		
<b>.</b>	platform wanting leaving component of south wath $(x = 0)$		
"# 2	platform-tilt vector		
ф_	tilt about platform x-axis		
от. Ф_	tilt about platform y-axis		
· 7 8			and a start of the
6 6.	X-gyro drift rate		
-11 4 E	Y-gyro drift rate		
ייי, אר געייי, אר	error in x and y components of earth rate		
10177	drift rate of aximith wander angle		
<u>_</u>	niatform enstial rate ventor	•	
۰ ما _	waria anatial rata		
<b>TET</b>	e-mae specar and		
~1T	a-maa gywyan invy w_avig gygtial wata		
ייזי ז	y-and system into .		Sold State of Control
~	$v_{av}$ is oracle rate $(-V_{av}, v)$		
~1 0	a-max diwit into (-v <sub>py</sub> /n)		
<b>~</b> 3	presition and figure		فسيد والمشاه
P1 (P1)			
** **	Note very route along the august		
v <sub>1</sub> , oy <sub>€</sub>			
* <b>5'</b> • * H			
CK0	Rooterdevier noise constant		
15, U			
^ <u>z</u> , ^y	x- and y-axis acceleration components		
0 <sup>4</sup> <sup>2</sup> , 0 <sup>4</sup> y			
aug, aug	x- and y-axis cortoils errors		AND THE PARTY
	gyra-moise-model constant		
	Begitive rediprocal or gyro correlation time		
CKA	Doppler noise model donstant		
-	Regative reciprocal of Doppler noise model correlation time		مانتیک (چمان منت مت منتخط (همان منت مت
<b>ట</b> 	White Hoise		
ω <sub>ι</sub>	white-noise components on state variables		
0, 0 <b>-</b>	observations		
h	altitude		الله من المراجع المراجع المستحد ما ما مناطق المنابع من المراجع ا
87 <sub>01</sub>	Doppler lateral-velocity error		

(\*\* N/A (\*\* \* \*\*\*

heine Bertenten Bere

.

n de

1 N U

		i te de la constante de la cons La constante de la constante de La constante de la constante d
		La statistica da la constatistica da la constatistica da la constatistica da la constatistica da la constatisti La constatistica da la constatis
426		
8v <sub>pr</sub> '	Doppler range-velocity error	۵٬۵۰۵ مې د ۲۵ مې د ۲۵ ۲۰۰۵ مې د ۲۰۰۵ مې د ۲۵ مې د ۲۰۰۵ مې د ۲۰۰ ۲۰۰۰ مې د ۲۰۰۵ مې د ۲
v <sub>0</sub> , v <sub>T</sub>	ground speed	
87. 08	total Doppler-drift-angle error	for the formation of the second s a contract of the second se
ψ <sub>₽</sub>	platform heading (graft to X platform axim)	
87 <sub>2</sub> , 87 <sub>7</sub>	velocity difference of inertial-Doppler units	
ðv <sub>a</sub>	error in ground speed (including conversion)	
81	Doppler momile factor	
8V. <sub>1</sub>	fluctuation error	
8V <sub>DE</sub>	x-axis Doppler error	
٥٧ <sub>07</sub>	y-axis Doppler error	مربع المربع ا المربع المربع
٨ <sub>B</sub>	vertical acceleration	
μ <sub>i</sub> , σ <sub>i</sub>	weesterometer pon-orthogonalities (i = 1, 3)	
∀ <sub>x</sub> , ∀ <sub>y</sub>	accelerometer biss errors	
₩ <sup>AX</sup>	x-habelerometer scale factor	
#* <sub>A2</sub>	y-addelerometer scale factor	
V	reciprocal of accelerumeter correlation time	
TRK	z-gyro borquing signal from filter	
TYE	y-gyra torquing signal from filter	
<b>e</b> 1	reciprocal of gyro correlation time	
7	amplitude of gyro poise drift	
A	suplitude of accelerometer noise	
w <sub>rz</sub>	I-gyro samle-factor error	
#***	y-gyto scale-fautor error	
# <sub>13</sub>	E-gyro scale-factor error	
R <sub>a</sub> , y, m	gyro torquing rates	
kj – kø	grn mon-orthogonalities	
a <u>i</u>	acceleration-sensitive drift coefficients for gyros	Kan Chinya ang kana
° <sub>ih</sub>	acceleration-squared drift coefficients for gyros	
τ <sub>e</sub>	gyrn-drift correlation time	6 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -
7)	integration algorithm constant	
μ	integration algorithm constant	

Ÿ.

<u>к</u>,

ł.

#### CHAPTER 16

#### SECTION II - NAVIGATION AT SEA USING THE INVARIANTS FORM OF KALMAN FILTERING

#### H. W. Banbrook and H. Halamandaris

#### 1. INTRODUCTION

Mechanization concepts employing optimal control techniques have been applied to guidance and navigation problems over the past five years, the object being to produce a system which exhibits a minimum error variance under the constraint of time-optimality. The latter is attained by separating the signal from the noise in the aid sensor data and maximally utilizing this data.

Inertial measurement units (INU) exhibit error characteristics which grow with time. A mechanization utilizing aid sensors to damp and/or "bound" the inertial-unit-error characteristics is desired. By mixing a Doppler radar which possesses good long-term stability and an inertial measurement unit which has good shurt-term stability, the overall system accuracy is gr y improved over that attainable from either mensor independently.

The inertial system is the pumary sensor and the Doppler serves as the sid sensor. Observations made by the Doppler are contaminated by the sea current drifts, since the Doppler measurements are relative to the water mass. The application discussed in this section specifically addresses an airborne application; however, the mathematical mechanisation as configured is directly applicable to a shipboard inertial unit damped by an EM log or Doppler sonar. In either the airborne or shipboard onse, the largest error contributor in the Doppler observation is the sea current drift. The hardware biases characteristic of modern airborne and shipboard sonar Dopplers are relatively shall compared to the sea noises.

This section contains a brief summary of Kalman filtering theory which, through a transformation of variables, is extended to the invariants form of Kalman filtering. A north-seeking model is discussed herein for illustrative purposes.

Practical considerations play an important role in configuring a Kalman filter; for instance, to mechanize an optimal filter for an inertial navigation system requires an expanded memory mirborne (or shipborne) computer. Due to weight, volume, cost, power reliability and maintainability constraints, it is not, in general, desirable to implement a fully-optimized system. Therefore, suboptimal filter mechanizations are generally influenced by the above-mentioned constraints. In keeping with this thinking, the invariants technicus minimizes the number of differential equations to be programmed in the covariance matrix. Also scaling requirements, for the ouvariance matrix, are not as stringent because of the bounded characteristics of subsets of the covariance matrix. Further, the theoretical accuracy realizable from the invariants ford of the Kalman filter is as good as the classical Kalman filter. However, the word-length requirements can be relaxed somewhat for the invariants technique because it is less sensitive to truncation and round-off effects than the plassical Kalman filter.

To illustrate the potential accuracy improvement attainable from the implementation of the optimal filter. Section 5 shows the result of a simulation for which the filter was run as an (open-loop) estimator rather than a (closed-loop) controller. The degree of predictability can be seen from the illustration in Section 5. The conclusion is that the invariants form of Kalman filtering can indeed predict the states of the system.

#### 2. KALMAN FILTER THEORY FOR THE INVARIANTS TECHNIQUE

#### 2.1 General

- 18 CO 18

This subsection contains the summary of Kalman's equation and the application of the Kalman filter based on the Litton invariants technique. The general method is that employed by Kalman. Certain modifications have been made to accommodate constraints imposed by the suboptimal characteristics of the mechanized filter. Also certain refinements, such as exponential torquing of the gyros and Doppler, have been incorporated to simulate the effect of exponentially-correlated noise sources. The following discussion presents the general philosophy of the Kalman filter.

The inertial navigation element (IMU plus computer) provides the complete navigation function; however, by using redundant data, its performance can be improved. The primary aid sensor considered (the Doppler radar) provides velocity information in the form of ground speed and Doppler drift angle. The Doppler radar also requires a heading reference such that combination of these sensors can also perform the navigation function, although it is somewhat degraded. An alternative use of the Doppler velocity is to damp the inertial system. The sid sensors also provide intermittent position data.

427





Thus it is seen that the various other avionic subsystems provide a redundant measurement of position and velocity. Optimal usage of this redundant information requires a dutailed examination of the relative errors of the various sonsors in addition to the availability of the data.

There are various methods of utilizing the redundant information, their quality depending heavily on the secounting for the many types, particular magnitudes and spectra of error sources. Of prime concern are classical and Kalman mechanizations. Since all mechanizations are imperfect, the problem is to extract signal from noise. The classical mechanization approach is to let the systems do much of the filtering by utilizing feedback.

This chapter considers Kalman filtering for airborne alignment and navigation. It is well known that such a filter technique results in a minimum variance error for the assumed-system-error model.

The error propagation of the inertial unit and Doppler error are both strongly flight-path-dependent, hence time-varying. Since the flight path is not fixed for all possible missions, it is clear that there should be some onboard accounting for flight path in the filtering, as referred to later, to obtain best system performance. Where intermittent or occasional information is subplied, such as position data, the Kalman filter is especially suited because of its capability of making large corrections throughout the inertial system. Provided the system error model is sufficiently realistic, improved system performance is obtained by employing Kalman filtering techniques. The mechanization advantages of classical feedback are obtainable in Kelman filtering. With a digital computer, it is important to note that the selection of filtering methods is reduced to only two considerations;

#### (1) System performance

#### (ii) Computer requirements.

This implies a sophisticated model because there are many important error sources and types of error propagation. Any increase of sophistication of the error model typically implies increased programming requirements. Utilization of results of investigations of computation techniques closely related to the deeper aspects of the Kalman filter approach are evidently not only warranted, but should be given priority.

A frequently considered aspect of the use of any filtering technique is the dependence on correctness of the hypothesized error model. If a sophisticated error propagation model is used, then unrealism can stem only from inacouracy of statistical estimates of error source magnitudes, e.g. drift-rate characteristics of the particular gyro. If the mechanization of the Kalman filter is a feedback system, as is the proposed system, then the advantages of the relative insensitivity of that type of system to unexpectedly large errors are attained. In all simulations of Kalman filters at Litton, it should be mentioned that error sources have particular characteristics in each run, randomly chosen from a distribution with the assumed characteristics.

The principal difficulty in applying Kalman filtering to sirborne systems was the large computer capacity required to mechanize the myriad equations. Litton has managed, by transformation of variables, sequential computation, and approximations, to simplify Kalman's general mechanization equations.

The transformation of variables is one in which the majority of the elements of the system covariance matrix are invariant to the position aid sensor data. This transformation of variables is presented in more detail in a later section.

#### 2.2 Classical Kalman-Filter-Equation Summary and Mathematical Preliminaries

The mathematical development of the classical Kalman-filter equations is formulated in myriad recent publications, including Kalman's complete documentation<sup>1</sup>; hence, these equations will not be dorived here but merely summarized and described in general terms.

Table I summarizes the Kalman/Litton equations that would typically be mechanized in an aided inertial navigation system for alignment and navigation. The notation employed is that used by Kalman. Underlined capital letterm signify matrices, underlined lower-came letterm signify vector quantities (column matrices). A prime signifies the transpose of, the "hat" sign signifies best estimate of, the + sign signifies the pseudo-inverse of, and E means the expected value of. The classical Kalman filter is modified slightly in that the best estimates of state variables I are updated with information including the N<sup>th</sup> increment of time.

In setting up a Litton mechanization, the first step is to select a set of state variables. These will normally be any and every quantity which we wish to update and sufficient states to adequately describe the dynamics of the system.

The next step is to represent the system dynamics by a set of first-order linear differential equations of the form of Equation (2, 1). These exist as a set of auch equations by defining the state variables as error quantities. For example, the state variables are not defined as "position" and "velocity", but as "error in position" and "error in velocity". The atandard linearized error  $c_{\rm station}$  (neglecting second-and-higher-order terms in error quantities) will then provide equations of the form of Equation (2, 1). Inspection of these equations yields the  $\xi$  and  $\underline{G}$  matrices, where  $\xi$  is referenced as the system dynamics matrix.

The <u>Gw</u> terms represent internally generated noise. Gyro drift noise is an example of internal noise represented by elements of <u>Gw</u>.













	16.70	<b>.</b> 44	Ľ,
	· · ·		



at this of the strate of the

428

į,

-000000

いる言いたい

WARRAND X

TABLE (

Summary of Kalman Filtering Equations

Defining Equations	
Statu variables: $\underline{x} = \underline{Cx} + \underline{Ow}$	(2. 1)
Obsorved quantities: $z = Hz + y$	(2. 2)
Initialize with best available estimates of state variables $\underline{x}$ and cover matrix $\sum$ (error in estimates)	ariance
Between updating points:	
(a) Propagate state variable estimates by $\frac{dg}{dt} = Eg$	(2.3)
(b) Propagate covariance by $\frac{d\Sigma}{dt} = \underline{r}\underline{\Sigma} + \underline{\Sigma}\underline{r}' + \underline{Q}\underline{Q}'$ ,	(2.4)
where $\sum = \operatorname{cov}[\underline{g}] = \mathbb{E}[\underline{x}, \underline{x}']$ $\mathbf{Q} = \operatorname{cov}[\underline{y}] = \mathbb{E}[\underline{x}\underline{w}']$ $\mathbf{R} = \operatorname{cov}[\underline{y}] = \mathbb{E}[\underline{y}\underline{y}']$	
Note: To use Equations (3.4) through (3.7) we assume $\mathbb{E}[\underline{wv}'] = 0$ , i.e. $\underline{w}$ and $\underline{v}$ are uncorrelated.	
(a) Correct state variables by $\Delta \underline{x} = \underline{K}(\underline{x} - \underline{H}\underline{x})$	(2.5)
$\mathbf{K} = \Sigma \mathbf{R}_1 \left[ \mathbf{R} \overline{\Sigma} \mathbf{R}_1 + \mathbf{R} \right]_{\frac{1}{2}} = \frac{\mathbf{D}}{\overline{\Sigma} \mathbf{R}_1}  .$	(2.8),
where $D = \underline{H} \underline{\Sigma} \underline{H}' + \underline{R}$ is a scalar quantity if updates are based on only measurement at a time.	y one
(b) Correct covariance matrix by $\Delta \Sigma = -KH\Sigma = -DKK'$ .	(2.7)

The observed or measured quantities used to update the state variables must also be specified. These are position checkpoints and Doppler velocity measurements. These "observables" must be expressed as a linear function of the state variables, as in Equation (2.2). The linearized error equations expressing errors in observations as functions of errors in the state variables provide Equation (2.2) and the observation matrix  $\underline{H}$ .

The  $\underline{\mathbf{y}}$  vector represents measurement errors. In most practical applications, the measurement errors  $\underline{\mathbf{y}}$  are uncorrelated with (statistically independent of) the internal noise  $\underline{\mathbf{y}}$ . To simplify the remainder of the equations, ascume that  $\mathbf{E}[\underline{\mathbf{wy}}'] = 0$ .

Equations (2.3) and (2.4) are the mechanization equations for real time computation of the estimated values of the state variables and the expected variance in these estimates. Equations (2.3) are essentially the navigation equations which are already mechanized in the inertial navigation system. Equation (2.4) is new to the mechanization. The covariance  $\Sigma$  represents a running estimate of the accuracy or confidence in the system estimates of each state variable. In effect, the system CEP is continually computed, implying that the covariance matrix is a "arunning error analysis.

Equations (2.3) and (2.4) are differential equations which are solved by integrating from some known starting point. Initialization is a familiar problem for Equation (2.3) and corresponds to aligning the platform, and setting in initial velocity and positions into the computer. To initialize Equation (2.4), an estimate of platform alignment, gyru calibration and computer initialization accuracy must be made. Such estimates are based on error analyses and actual test data. These estimates are not critical to the process if they are not grossly underectimated. Studies show that it is better to overestimate the error for the self-correlated terms (variances) rather than to underestimate, whereas it is better to underestimate the cross-correlated terms (covariances). Upon initializing Equations (2.3) and (2.4) the computer continuously tracks the state variables and updates the covariance matrix by integrating Equations (3.3) and (3.4) in real time.

When aid information is obtained from separate measurements or observations, the present estimates of the state variables and covariance matrix are upuated according to Equations (2.5) and (2.7). Equation (2.5) states that a comparison of the measured value z with the present estimate <u>Hf</u> be made, and this difference signal multiplied by a feedback gain <u>K</u>. One of the simplifications instituted by Litton is to serially update after a measurement. Even if data from two or more measurements are available simultaneously, the data can still be processed serially as several single measurements and the system sequentially updated. This procedure reduces each of the <u>H</u> matrices to a single row matrix. This in turn makes the quantity  $\underline{H\Sigma H}' + \underline{R}$  a scalar quantity, and the inverse of a scalar D is simply its reciprocal  $D^{-1}$ .

Once the filter gain  $\underline{k}$  and the scalar D from Equation (2.6) are computed, the corrections to the elements of the covariar 'matrix become almost trivial, as seen in Equation (2.7). Following the update corrections, the systems compute ions revert back to Equations (2.3) and (2.4) in order to propagate the system estimates forward in real time until the next system updating occurs.

In addition to the above, the development of the correction invariant formulas requires a cursory knowledge of tensor notation, specifically, the concept of the summation convention. In tensors, to denote different quantities, superscript indices are employed. In the development which follows, subscripts are used throughout except on the Kronecker delta terms.

In writing homogeneous linear and quadratic functions summed over some index, the following types of formulas, respectively, result:

 $+ \mathbf{a}_{12}^{*}\mathbf{x}_{1}\mathbf{x}_{2} + \mathbf{a}_{13}^{*}\mathbf{x}_{1}\mathbf{x}_{3} + \dots + \mathbf{a}_{1n}^{*}\mathbf{x}_{1}\mathbf{x}_{n} + \\ + \mathbf{a}_{21}^{*}\mathbf{a}_{21}\mathbf{x}_{1} + \mathbf{a}_{22}^{*}\mathbf{x}_{23}\mathbf{x}_{3} + \dots + \mathbf{a}_{2n}^{*}\mathbf{x}_{23}\mathbf{x}_{n} +$ 

SOM PANHAN DAKE

 $\sum_{y=1}^{n} a_{y} x_{y} = a_{1} x_{1} + a_{2} x_{2} + a_{3} x_{3} + \dots$ 

 $\sum_{k=1}^{n} \sum_{j=1}^{n} a_{kj} x_k x_j = a_{11} x_1^2 + a_{2j} x_2^2 + \ldots + a_{nn} x_n^2 +$ 

and

S. Carlanda

Sect

.....

These two equations are systems of the first order and systems of the second order, respectively. A system of order zero has no indices and thus is invariant.

 $+ \dots + a_{i_1 i_1} x_{i_1} x_{i_1} + \dots$ 

In the course of the derivations that follow, the summation signs  $(\Sigma)$  would be used quite frequently. To present a more compact appearance for the formulas, the  $\Sigma$  sign will be dropped and is implied whenever an index is repeated in the expression, e.g.,

$$\sum_{j=1}^n \mathbf{a}_j \mathbf{x}_j \triangleq \mathbf{a}_j \mathbf{x}_j$$

and, since j is repeated, the summation over j is implied on the right-hand side of the expression above. The convention applies in similar manner to the double summation, since the indices j and k are both repeated. To facilitate the analysis, a dummy index is frequently employed. An index not repented is a free index and spans any dimension assigned.

The Kronecker delta  $(\delta_1^k)$  is a mixed tensor of the second order (rank two), because it employs both subscripts and superscripts and has the unique property that

 $\delta_j^k = \begin{cases} 0 & \text{if } j = k \\ 1 & \text{if } j = k \end{cases}$ 

2.3 General Implications of Correction Invariants

The error propagation of the inertial system can be shown to be computed by differential equations in the covariance matrix, which, by a special transformation of variables, has the form

A B B B

where  $\underline{R}, \underline{S}$  are invariant under Kalman corrections with respect to a chosen subset of observed state variables and  $\underline{A}$  is the inverse of the matrix of that chosen subset of variables which are, in part, directly observed by aid sensors.

The A submatrix, which comprises the small portion of non-invariants of the computation set, is functionally the inverse of a partition of the error matrix and, as such, has diagonal terms that vary proportionally to system accuracy which, in turn, lies between zero and a sharply defined upper band. In contrast, the direct error quantities are relatively unbounded. Useful information is best mapped in terms of accuracy measure in statistics for combination with other information. (Thus, defining accuracy  $A \approx 1/\sigma_p^2$ , where  $\sigma_p^2$  is the variance of the position data, it follows that the accuracy obtainable by many observations is the sum of  $A_1$ .)

Defining Kalman's covariance matrix as  $\sigma$ , with subscripts x and y denoting the position variables, the A submatrix is functionally

$$A_{ps} = \sigma_{ps}^{-1} = \begin{vmatrix} \sigma_{yy} & \sigma_{yy} \\ \sigma_{xy} & \sigma_{yz} \\ \sigma_{yy} \sigma_{xx} & \sigma_{yy} \end{vmatrix}$$

The R invariants, which appose in the Kalman weighting for inertial system trim-up, possess the functional behavior of ratios of related covariances. The ratios have relatively limited range compared to related correlations. Functionally, the R submatrix is

$$R_{rs} = \sum_{k=1}^{k} \sigma_{rk} A_{ks}$$

where r > k,  $s \leq k$ , and r and s refer to the members of the R submatrix.

The S invariants (comprising the larger part of modified matrix equations), which correspond to correlations of non-observables, are the most insensitive to the aid data profile. This offers a potential memory reduction by using approximations which do not materially compromise performance optimality. The S submatrix elements have the same units as the corresponding members of the classical covariance matrix ( $\sigma$ ) and functionally are

$$\mathbf{S}_{\mathbf{rs}} = \boldsymbol{\sigma}_{\mathbf{rs}} - \sum_{k=1}^{\mathbf{k}} \mathbf{R}_{\mathbf{rk}} \boldsymbol{\sigma}_{\mathbf{ks}} \, ,$$

where r, s > k.

The choice of correction invariants with respect to position observations leads to error propagation differential equations in which latitude error effects in level and vertical components of earth rate and Coriolis computation are largely eliminated from the R and S submatrix elements. This presents a memory saving and removes problems of computation of very small terms whose integrated magnitude over 10 hours may, nevertheless, be substantial.

The correction invariants tend to retain the simple cross-channel characteristics of the pure inertial system, otherwise disturbed by the aiding process. These characteristics are principally the result of azimuth error and, usually to a lesser degree, latitude error entering earth rate estimates. With computation in invariant form, it has been demonstrated that a reduction in differential equation programming is feasible, the obviated equations being replaced with a few closed form azimuthal correlation relations which enable azimuth correction based on accelerative effects (important at high latitudes). The computation techniques proposed for the invariants are presented in the subsequent paragraphs.

The A submatrix is restricted to contain only position variables; the R submatrix contains the covariance of one position variable with one non-position variable. The S submatrix contain the covariance of all nonposition variables with non-position variables.

#### 2.4 Algebruic Form of Correction Invariants

An observation involving a given subset of system error  $\epsilon_k$ ,  $k = 1, \ldots, K$  of form

$$\underline{z} = \sum_{k=1}^{k} e_{k} \mathbf{G}_{k} + \mathbf{N}$$

is utilized for Kalman corrections (analogous to Kalman's expression,  $\underline{z} = \underline{H}x + \underline{\nu}$ ) of each system variable, where it may be shown that the change upon correction implies that

$$\Delta \Sigma_{rs} = -\frac{U_{r}^{2}U_{s}}{\sigma_{N}^{2} + a_{k}u_{k}} = -U_{s}^{2}U_{r}^{2}/\sigma_{N}^{2}, \qquad (2.8)$$

where  $\sigma_N^2$  is the observation error variance, r and s refer to variable labels, the + and - superscripts refer to after and before system correction, and

 $A_{ps} \neq \sigma_{ps}^{-1}$ 

$$\mathbf{U}_{\mathbf{r}} = \sum_{k=1}^{\mathbf{g}} \sigma_{\mathbf{r}\mathbf{k}} \mathbf{G}_{\mathbf{k}} \, .$$

Furthermore, Ann is defined

431

(2.9)

It may be proven, from the form of the Kalman correction formulas presented above, that

$$R_{rs} = \sum_{k=1}^{K} \sigma_{rk} A_{ks}, \qquad s < K^{\dagger}$$
$$B_{rs} = \sigma_{rs} - \sum_{k=1}^{K} R_{rk} \sigma_{ks}, \qquad s, r > K$$

undergo changes under system correction  $\Delta R_{rs} = 0$ ,  $\Delta B_{rs} \approx 0$ ; that is,  $R_{rs}$ ,  $S_{rs}$  are correction invariants for the chosen invariant set (position). Note that the "avariants are not functions of the observation weights ( $G_k$ ) nor observation accuracy. In particular, it is seen that invariants for any position observation components are the same. It can be shown that

$$\Delta A_{\rm ps} = 0_{\rm p} 0_{\rm s} / \sigma_{\rm R}^2, \quad p, s \leq K. \tag{2.10}$$

The error propagation of A, B, S may be expressed directly by differential equations in terms of A, B, S alone. The Kalman system corrections have weight

$$K_{\mu} = \frac{U_{\mu}^{+}}{\sigma_{N}^{2}} = \frac{1}{\sigma_{N}^{2}} \frac{\Sigma}{s} R_{\mu s} U_{s}^{*}, \qquad (2.11)$$

where  $U_{a}^{*} = \Sigma_{g} G_{p} A_{ap}^{-1}$ . For the case where the correction invariants are defined with respect to position-type observations,

$$\sigma_{\rm sp} = \Lambda_{\rm sp}^{-1} = \begin{bmatrix} \Lambda_{\rm ss} & -\Lambda_{\rm 12} \\ -\Lambda_{\rm 13} & \Lambda_{\rm 11} \\ \Lambda_{\rm 11}\Lambda_{\rm ss} & -\Lambda_{\rm 12}^{-1} \end{bmatrix} \, . \, .$$

This implies that, upon receipt of a position fix, dnly the A matrix (3 elements) need be updated; the R and S submatrices (39 elements) are invariant, and, hence, do not change. An abbreviated proof of A. R, S properties follows:

(a) The correction formulas for A<sub>ps</sub> defined with respect to a selected observation type are obtained from the ordinary Kalman formula for that observation type after taking the first difference of the identity.

$$A_{pk}\sigma_{kr} = \sigma_{pk}^{-1}\sigma_{kr} = \delta_{p}^{r}, \quad p, r, k, \leq K.$$

$$\Delta A_{pk}\sigma_{kr}^{*} + A_{pk}^{*}\Delta \sigma_{kr} = 0$$

Thus

 $\Delta A_{pk}\sigma_{kr}^{\dagger} - A_{pk}^{-} U_{k}^{\dagger} U_{r}^{\dagger} / \sigma_{\theta}^{\dagger} = 0 , \qquad (2.12)$ 

where a summation convention is used on paired indices. Multiplying by  $(\sigma_{rs}^{-1})^*$  and noting that

$$A_{pr}U_{r} = \sigma_{pr}^{*1}U_{r} = \sigma_{pr}\sigma_{rs}G_{s} = G_{p}, \quad s \leq K,$$

obtain

$$\Delta A_{ps} - A_{pk}^{*} U_{k}^{*} U_{r}^{*} A_{rs}^{*} / \sigma_{\theta}^{2} = 0 , \qquad (2.13)$$

whence the correction formula

$$\Delta A_{pa} = \frac{a_p a_s}{\sigma_s^2}, \quad p, a < K. \qquad (2.14)$$

(b) The invariance of  $R_{pg}$  defined with respect to a selected observation type is shown, taking the first difference of the defining equation for R,

 $\Delta R_{re} = \Delta \sigma_{rk} A_{ke}^{*} + \sigma_{rk}^{*} \Delta A_{ke} \qquad k , s \leq K.$ (2.15)

Substituting the ordinary Kalman correction and the  $\Delta A_{\rm ps}$  expressions

$$\Delta \mathbf{R}_{\mathbf{rs}} = \frac{-\mathbf{U}_{\mathbf{r}}^{T}\mathbf{U}_{\mathbf{k}}^{*}}{\sigma_{\mathbf{d}}^{2}} \mathbf{A}_{\mathbf{ks}}^{*} + \sigma_{\mathbf{rk}}^{-} \frac{\mathbf{G}_{\mathbf{k}}^{+} \mathbf{s}}{\sigma_{\mathbf{d}}^{2}}.$$
 (2.16)

and using Equation (2.12) and the definition of  $U_{\mu}$  .

These restrictions are redundant; e.g., all 5 invariants are zero where position is involved, and  $R_{gg} = 1$ ,  $R_{yy} = 1$  and  $R_{gy} = 0$ .







(2. 11)

August 1

Destant b

And the start of the

$$\Delta \mathbf{R}_{\mathbf{F}\mathbf{S}} = \frac{-\mathbf{U}_{\mathbf{F}}^{*}\mathbf{O}_{\mathbf{g}}}{\sigma_{\mathbf{g}}^{*}} + \sigma_{\mathbf{F}\mathbf{k}}^{*} \frac{\mathbf{O}_{\mathbf{g}}\mathbf{O}_{\mathbf{g}}}{\sigma_{\mathbf{g}}^{*}} = \frac{\mathbf{O}_{\mathbf{g}}}{\sigma_{\mathbf{g}}^{*}} \left[ -\mathbf{U}_{\mathbf{F}}^{*} + \sigma_{\mathbf{F}\mathbf{k}}^{*}\mathbf{O}_{\mathbf{k}} \right] = 0 \; .$$

Hence R<sub>rs</sub> is invariant.

у / ч, () - ч на

22222

South States

(c) The invariance of  $S_{rs}$  defined with respect to a selected observation type is shown by taking the first difference of the defining equation, using the relation  $\Delta R_{rk} = 0$ ,

$$\Delta \mathbf{s}_{rs} = \Delta \sigma_{rs} - \mathbf{R}_{rk} \Delta \sigma_{sk}, \quad k \in \mathbf{K}, \quad (2.17)$$

and then substituting the ordinary Kalman expressions to obtain

$$\Delta \mathbf{S}_{ra} = \frac{-U_{a}^{*}U_{r}^{*}}{\sigma_{\theta}^{2}} + \mathbf{R}_{rk} \frac{U_{a}^{*}U_{k}^{*}}{\sigma_{\theta}^{2}}$$
$$= \frac{U_{a}^{*}}{\sigma_{\theta}^{4}} \left[ -U_{r}^{*} + \mathbf{R}_{rk}^{*}U_{k}^{*} \right]$$

using  $R_{rs}^{+} = R_{rs}^{+}$ .

Substituting the defining expression for R<sub>wk</sub> ,

$$\frac{d^{+}}{rk}U_{k}^{+} = \sigma_{rp}^{+}A_{pk}U_{k}^{+} = \sigma_{rp}^{+}G_{p} \cdot = U_{r}^{+}.$$

Hence  $\Delta S_{rs} = 0$ ; that is,  $S_{rs}$  is correction invariant relative to the associated type of observation.

#### 2.5 Correction Formulas for Disparate Observations

The correction invariant set for observations  $\theta$  (involving, for example, position errors) are not invariant under corrections with different kinds of observations (such as involving velocity or axisuth errors). The following is a derivation of correction formulas for disparate observations, which are shown to be closely similar to ordinary Kalean formulas but in the variables of the correction invariant set. The algebraic derivation differs for unassociated observations as a result. In Equation (2.8) the terms  $U_{\alpha} \rightarrow U_{\alpha}$  where<sup>†</sup>

$$U_{\rm p} = \sigma_{\rm rs}^2 G_{\rm s} , \quad {\rm s} \in \mathbb{K}$$

$$U_{\rm s} = \sigma_{\rm rs}^2 G_{\rm s} , \quad {\rm q} > \mathbb{K} .$$
(2.18)

That is, q is not in the set of variables for which  $A_{ps}$  is defined. Consequently,  $U_r A_{rs} \neq G_r$  and Equation (2.14) does not hold. Indeed, for the unassociated observations  $U_r A_{ts} = \sigma_{rq} G_q A_{rs} = G_q R_{qs} = R_{\sigma s}$ , where R is the correction invarant set of the computation process and denoting  $R_{\sigma s} = G_q R_{qs}$  for  $\Theta$  disparate type of observation. The Equation (2.10) is replaced with

$$\Delta A_{\mu\nu} = \frac{R_{\rho+\mu}R_{\rho+\mu}}{\sigma_1^2}, \qquad (2.19)$$

which is a formula computationally similar to the ordinary Kalman formula except for sign of the correction. The correction  $R_{ra}$  for unassociated observation type is obtained from the R-matrix defining Equation (3.13) and

$$\Delta \sigma_{\mathbf{rk}} = \frac{-\theta_{\mathbf{r}}^* \theta_{\mathbf{k}}}{\sigma_{\mathbf{r}}^*} \qquad (2.20)$$

Further manipulation yields the correction formula for R<sub>rs</sub> with unassociated observation:

$$\Delta \mathbf{R}_{\mathbf{a}} = \frac{-\mathbf{R}_{\mathbf{a}}^{*} \mathbf{s}_{\mathbf{f}}^{*} \mathbf{s}_{\mathbf{f}}^{*}}{\sigma_{\mathbf{a}}^{*}} \mathbf{r} > \mathbf{K}$$
(2.21)

The implied computation is directly analogous to the ordinary Kalman correction formulas.

The corrections of Srs for unassociated types is obtained in the same manner.

\* Retaining use of summation convention on paired indices.

**CALCER** 

Personal and

The correction formula for unassociated types of observations is

$$\Delta S_{rm} = - \frac{S_{rd+}^* S_{md+}^*}{\sigma_{d}^*} \quad \text{where} \quad r, n \ge k \,.$$

The implied computations are again directly analogous to the ordinary Kalman correction formulas.

#### 2.6 Computational Features of Correction Invariants

The Kalman filtering computations relating to error growth, when executed in correction invariant variables, have a number of markedly different characteristics from the direct form with impact on computer cost, including effects of the following:

- (i) Obviation of the bulk of the error-matrix update computations following inertial system corrections with aid data, this being the consequence of the invariance proporties. This feature makes a wired incrementalcomputation program of digital differential analyzer program attractive because of substantial reduction of memory requirements associated with instruction memory, the variable memory, however, being increased to a lesser degree.
- (11) Reduced word-length requirements and/or decreased computation error sensitivity.
- (a) The bulk of the variables in the correction invariants computations have readily determined variation ranges which change little for disparate flight and aid data profiles. In contrast, variances which have the range of the square of errors which themselves have wide range are unwidely computationally and can require monitoring to prevent overflow under varying operational conditions.
- (b) The A matrix, which comprises the small portion of non-invariants of the computation set, is functionally the inverse of a partition of the error matrix and as such has diagonal terms which vary proportionately to system accuracy, which lies between zero and a sharply defined upper band. In contrast, the direct error quantities are relatively unbounded. Useful information is best mapped in terms of accuracy measure in statistics for combination with other information. (Thus, defining accuracy A =  $1/\sigma^2$ , where  $\sigma^2$  is the variance, it follows that the accuracy obtainable by many observations is the sum of accuracy.)
- (a) The R invariants, which appear in the Kalman weighting for inertial system trim-up, have functional behavior of ratios of related variances. The ratios have relatively limited range compared to related correlations.
- .(d) The 5 invariant# (comprising the larger part of modified matrix equations), which correspond to correlations of non-observables, are the most insensitive to aid data profile, offering a potential of extensive memory reduction by approximations which do not materially compromise performance optimality.
- (iii) Memory saving by reduced computation program relating to cross-channel inertial variable correlations.

The correction invariants tend to retain the simple cross-channel characteristics of the pure inertial system, otherwise disturbed by the aiding process, which are principally the result of azimuth error (and latitude error entering earth rate estimates to a usually smaller degree). With computation in invariants form, it has been demonstrated that a 20-40% reduction in *differential equations programming* is feasible, the obviated equations being replaced with a few closed form azimuthal correlations which enable capability for azimuth correction based on accelerative effects (important at high latitudes).

#### 3. SYNTEM-ERROR MODELS

The basic error sources considered are characteristic of the inertial sensors (gyros and succlerometers) and missignments of these sensors. Random errors which are described by statistical autocorrelation functions were considered for the sensors. These error models are described in more detail in the subsequent sections. A northseeking coordinate frame is assumed with the axes orientation shown in Figure 1. (The inertial-system error-model block diagram is shown later in Figure 5.)

3.1 Imertial Sensor-Error Hodels

#### 3.1.1 General

and the second design the second s

The inertial sensor-error models are contained in the general IMU error model; however, a brief description is presented here to clarify the models. In the description of the various sensors, the noise function considered is exponentially correlated with an autocorrelation function of the form

 $R(\tau) = \beta^{\dagger} e^{-\nu} |\tau| ,$ 

where  $\beta^{2}$  is the noise-amplitude function and  $\nu$  is the reciprocal of the correlation time characteristics of the noise. Non-white noise drivin; functions of this nature are generated by driving a shaping filter with white noise having an autocorrelation function

and a state of the second state of the second state of the second state of the second second second second second

$$\mathbf{R}'(\tau) = 2\nu\beta^2\delta(\tau) \ .$$

















Bill and the Constitution of the

ario 🔘 (\* 27

The shaping filter is shown in Figure 2.

No thermal models are considered for the inertial instruments.

#### 3.1.2 Accelerometers

Errors which are characteristic of the accelerometers in this analysis are blas, random, scale factor and mon-orthogonalities. Bias is defined as day-to-day or long-term repeatability and is assumed constant over any mission. Mosle-factor error refers to calibration: errors and long-term stability of the accelerometer sensing mechanism (including electronics). This error is assumed constant for the flight duration. Random is a shortterm change due to shifts in the accelerometer bias and scale factor errors. Non-orthogonalities are composed of two basic errors - mechanical missighments in mounting the accelerometer and mechanical missighments internal to the accelerometer.

#### 3.1.3 Gyros

. 9. J.

C.45874

222.22 22

 $\Sigma_{i,j}$  ,  $\omega_{i,j}$ 

Й С

1.1.5

The gyro terminology employed is in general consistent with the standard gyro terminology prepared by the Gyro and Accelerometer Panel of the Electronic Parts Committee, Acrospace Industries Association. The error sources considered are acceleration-insensitive bias and random drifts, acceleration-sensitive drifts (mass unbalance and compliance or anisoelesticity), torquer scale-factor errors and mechanical misalignments.

The bias error is assumed constant over any sission, since it is defined as the day-to-day or turn-on repeatability. The random drift is described by an exponential autocorrelation function and does change over any given mission. Torquer scale-factor errors are due primarily to electronics and calibration errors and remain constant over the mission. Mechanical misalignments are composed of gyro-mounting errors and internal gyro mon-orthogonalities.

The differential equation describing the gyro-noise model to be used in the Kalman filter is:

where

# $\dot{z} = DD \times c + CKD \times \delta_{B}^{T},$ $\dot{\delta}_{B}^{T} = \begin{cases} 1 & \text{if } \tau = B \\ 0 & \text{if } \tau \neq B \end{cases}$ $DD = \frac{1}{\tau_{d}}$ $\tau_{g} = \text{ correlation time of gyro noise}$ $CKD^{2} = \frac{c^{2} \times DD}{2}$

e = gyro random-drift-rate amplitude .

#### 3.1.4 Doppler-Error Model

The Doppler-error model simulated is composed of several error types, which are shown in Figure 3. The errors considered are bore-sight, scale factor, high frequency electronics noise, low-frequency terrain (overwater) noise, conversion biases and flexure errors. The basic differential equation describing the Doppler noise model is

# $\delta \dot{V}_{D} = BB \times \delta V_{D} + CKA \times \delta T_{B}$ $\delta T_{B} = \begin{cases} 1 & 12 & \tau = B \\ 0 & 12 & \tau \neq B \end{cases}$ T = -1

 $\tau_{\rm B}$  = Doppler correlation time.

Also, shown in Figure 4 is the resolution of the Doppler velocity through platform heading and drift angle into platform axis components. The symbols  $\delta V_x$  and  $\delta V_y$  represent the differences between platform velocities  $(V_{p_2}, V_{p_3})$  and the respective Doppler velocities.

#### 3.3 INU Krror Model

The IMU error model is shown in Figure 5 with each axis decoupled pictorially but functionally coupled. The level channels are completely shown in the figure with only the vertical gyro. The vertical loop is not presented because a conventional third-order baro-inertial loop is assumed; hence, the vertical-velocity errors are small.

Functions describing each of the errors are presented and discussed below, and terminology is consistent with that employed in most guidance texts and that specified by the Gyro and Accelerometer Panel of the Electronic Parts Committee, Aerospace Industries Association.

Coriolis Errors

$$\begin{split} &\delta \mathcal{O}_{\chi} = -\delta (\rho_{\chi} + 2\Omega_{\chi}) V_{\mu \gamma} - (\rho_{\chi} + 2\Omega_{\chi}) V_{\gamma} , \\ &\delta \mathcal{O}_{\psi} = \delta (\rho_{\mu} + 2\Omega_{\mu}) V_{\mu \psi} + (\rho_{\mu} + 2\Omega_{\mu}) V_{\psi} , \end{split}$$

**Cross-Coupling Acceleration Errors** 

$$f_{1}(\vec{A})_{\chi} = -\phi_{\chi}A_{\chi} + \phi_{\chi}A_{\chi}$$
  
$$f_{1}(\vec{A})_{\chi} = +\phi_{\chi}A_{\chi} - \phi_{\chi}A_{\chi}$$

Accelerometer Non-orthogonalities

$$\begin{split} t_{g}(\bar{A})_{\chi} &= -\mu_{g}A_{\chi} + \sigma_{g}A_{\chi} \\ t_{g}(\bar{A})_{\chi} &= \mu_{1}A_{\chi} - \sigma_{1}A_{g} \\ & \beta_{\chi}^{g} e^{-\nu|\tau|} + \nabla_{\chi} \end{split}$$

<u>,</u><u></u><u>β</u><sup>\*</sup> e<sup>-ν|+</sup> | + ∇<sub>v</sub> ,

Accelerometer Noise and Bias

 $\mathbf{W}_{AX} = \mathbf{X}$  modelerometer scale-factor error  $\mathbf{SF}_{AY} = \mathbf{Y}$  modelerometer scale-factor error  $\mathbf{A}_{\mathbf{X}}$ ,  $\mathbf{A}_{\mathbf{Y}}$  are true modelerations.

TFN = Gyro torquing term from filter

 $\delta \rho_{g} = \tan \phi_{g} \frac{V_{\chi}}{R} + \frac{V_{\chi T}}{R} \sec^{2} \phi_{g} \delta \phi_{T}$ 

 $\delta \Pi_{y} = \Omega \cos \phi \, \delta \phi_{T} = \delta \Omega_{g} \, .$ 

$$\begin{split} \delta \rho_{\chi} &= - V_{\chi} / R \\ \delta \Omega_{\chi} &= + \Omega \sin \phi_{\chi} \delta \phi_{\chi} \end{split}$$

 $\delta \rho_{\rm Y} = V_{\rm X}/R$ 

Computational-Error Rates

Gyro	Noise	and	Bias	
------	-------	-----	------	--

$$\begin{split} \gamma^{a}_{\mathbf{X}} & \mathbf{e}^{-\mathcal{C}[\tau]} + \mathbf{c}_{\mathbf{X}} \\ \gamma^{a}_{\mathbf{Y}} & \mathbf{e}^{-\mathcal{C}[\tau]} + \mathbf{c}_{\mathbf{Y}} \\ \gamma^{a}_{\mathbf{X}} & \mathbf{e}^{-\mathcal{C}[\tau]} + \mathbf{c}_{\mathbf{Y}} \end{split}$$

 $SF_{TX} = Scale-factor error in X gyro$  $<math>SF_{TY} = Scale-factor error in Y gyro$  $<math>SF_{TX} = Scale-factor error in Z gyro$ .

Gyro-Scale-Factor Error

Gyro Gross-Coupling Rates

 $\mathbf{g}_{1}(\vec{\omega})_{\mathbf{X}} = -\omega_{\mathbf{Y}\mathbf{T}}\phi_{\mathbf{Z}} + \omega_{\mathbf{Z}\mathbf{T}}\phi_{\mathbf{Y}}$  $\mathbf{g}_{1}(\vec{\omega})_{\mathbf{Y}} = -\omega_{\mathbf{Z}\mathbf{T}}\phi_{\mathbf{X}} + \omega_{\mathbf{X}\mathbf{T}}\phi_{\mathbf{Z}}$  $\mathbf{g}_{1}(\vec{\omega})_{\mathbf{Z}} = -\omega_{\mathbf{X}\mathbf{T}}\phi_{\mathbf{Y}} + \omega_{\mathbf{Y}\mathbf{T}}\phi_{\mathbf{X}}$ 



Gyro Non-orthogonalities

$$\begin{split} \mathbf{h}_{1}(\vec{\mathbf{R}})_{\mathbf{X}} &= \mathbf{k}_{\mathbf{u}}\mathbf{R}_{\mathbf{Y}} - \mathbf{k}_{\mathbf{s}}\mathbf{R}_{\mathbf{g}} \\ \mathbf{h}_{1}(\vec{\mathbf{R}})_{\mathbf{Y}} &= \mathbf{k}_{1}\mathbf{R}_{\mathbf{g}} - \mathbf{k}_{\mathbf{s}}\mathbf{R}_{\mathbf{X}} \\ \mathbf{h}_{1}(\vec{\mathbf{R}})_{\mathbf{g}} &= \mathbf{k}_{\mathbf{s}}\mathbf{R}_{\mathbf{X}} - \mathbf{k}_{\mathbf{s}}\mathbf{R}_{\mathbf{Y}} \,. \end{split}$$

Gyro Acceleration-Sensitive Errors

$$f_{3}(\overline{A})_{\chi} = -A_{Y}C_{3} + A_{\chi}C_{4} - A_{\chi}A_{Y}C_{22}$$

$$f_{3}(\overline{A})_{Y} = A_{Y}O_{1} - A_{\chi}C_{2} + A_{\chi}A_{Y}C_{22}$$

$$f_{4}(\overline{A})_{\chi} = -A_{V}C_{4} + A_{\chi}C_{4} + A_{\chi}A_{4}C_{33}$$

#### 3.3 Environmental Models

Observations made by the airborne Doppler sensor are relative to the sea, which contaminates the observation with three basic error components. These components fall in three categories: white noise, short-term and long-term correlated noise with subcourvelation functions of the form described in Section 3.1 (Figure 3). The white-noise component results from white caps. The short-term correlated sea current can be characterized by correlation distances of 18-20 nautical miles with nominal amplitudes of 0.4 to 1.0 knots. The long-term correlated noise has nominal amplitudes of 3 knots with correlation distances on the order of 500 nautical miles.

#### 4. STATE VARIABLE-SELECTION-AND-COVARIANCE-MATRIX PROPAGATION

#### 4.1 State Variable-Selection - and -Dofining Equations

Due to physical constraints, such as computer program size and processing time, a fully optimized Kalman filter mechanization is virtually impossible for an airborne computer. Hence, several trade-offs must be made in avlecting a suboptimal filter which will offer the advantages available from an optimal filter and, yet minimize the drawbacks. The selection of the state variables which adequately describe the system is the most prominent area for trade studies with computer restrictions in mind. The state variables should include all the quantities reguired to describe the condition of the system and, when possible. its significant error contributors, since these error contributors can be reduced with the aid of the optimal filter. The degree of optimality is dependent on several factors such as accuracy in the description of the error model, processing techniques and selection of the state variables. The states of the inertial system which improve the performance accuracy when accurately modeled are: position, velocity, platform tilts, and platform heading. System performance accuracy is improved considerably, also, if the principal contributors to the errors in each of the above states can be modeled. The inertial system as presently modeled for the alignment and navigation modes has 41 dominant-error sources not including the effects due to mid sensors, computer or environment. Because many of these errors contribute only second-order effects (even in a conventionally-damped system), it is meaningless to include these as state variables. This reduces the number of significant error sources to basically the bias and random error components of the syros and accelerometers.

Prior to the final selection of the states, one must also examine the aid sensors available to the system. If the errors induced by utilization of the aid sensors are tignificant as compared to the state variable candidates mentioned, then the aid-sensor contributions must also be considered as possible candidates. For the present application, the Doppler performance specifications are assumed sufficiently large, such that the Doppler error is one of the primary error contributors to the system performance accuracy and must be considered along with the gyros and accelerometers.

In performing the final system-trade studies for state variable selection, the inertial sensor and aid-sensorerror sources were weighed against each other. Although the level-accelerometer bias and random errors are primary contributors in system accuracy, the level accelerometers were eliminated as state variables since the reference velocity error was so large in comparison to the inertial system error attributable to the accelerometers. In fact, in the present mechanization, the inertial system is used to trim the Doppler. As a result of the trade studies, the level-gyro drifts and two Doppler-velocity states were added.

Several other errors were considered as possible state variables. These were vertical gyro drift, gyro torquer scale factors, accelerometer scale factors, and vertical accelerometer bias. In the vertical channel, the accelerometer is not compensated by the filtering te hnique; however, the vertical-channel error is minimized by the baro-inertial-loop mechanization. The azimuth-gyro drift is a large contributor to the system position error; however, to explicitly model this as a state did not improve system accuracy to the extent to warrant mechanizing an additional state. Instead, the azimuth gyro is modeled as a "pseudo-state" by incorporating its approximate effects as a noise on the platform heading error equation. Similar rationale was employed in eliminating the lavel accelerometer errors as states. In the intrest of maintaining the computational-load minimal and in the trade-off of computer capacity required versus improvement in navigation accuracy, the remainder of the error sources were eliminated.







### linnin Orlandur Olan

han a Barring Barra

Formally, Litton has selected elsven state variables to model the system:

 $\phi$ ,  $\delta\lambda \cos \phi_{\rm T}$  latitude and east position errors, respectively  $\delta V_{\rm H}$ ,  $\delta V_{\rm H}$  platform morth and east axis velocity errors, respectively  $(V_{\rm X}, V_{\rm Y})$   $\phi_{\rm H}$ ,  $\phi_{\rm H}$ ,  $\phi_{\rm H}$  angular errors in platform tilt ( $\phi_{\rm H}$  and  $\phi_{\rm H}$ ) and azimuth angle, respectively  $\epsilon_{\rm H}$ ,  $\epsilon_{\rm H}$  platform level-gyro-drift rates, east and north ( $\epsilon_{\rm X}$  and  $\epsilon_{\rm Y}$ ).

The state-variable-defining equations corresponding to these states for the north-seeking coordinate system are:

$$\begin{aligned} \delta p_{V} &= \delta V_{W} R_{W} \\ \delta \lambda \cos \phi_{V} &= \frac{\delta V_{W}}{R_{W}} + \frac{V_{W}}{R_{W}} \tan \phi_{V} \delta \phi - \frac{V_{W}}{R_{W}} \tan \phi \quad (\delta \lambda \cos \phi_{V}) \\ \delta V_{W} &= \frac{W}{R} \phi_{W} - \frac{A_{W}}{R_{W}} \phi_{W} - (\rho_{V} + 2\Omega_{V}) \quad \frac{\delta V_{W}}{R_{W}} - \left(\frac{V_{W}}{R_{W}} \tan \phi\right) \quad \frac{\delta V_{W}}{R_{W}} - \left(\frac{2\Omega \frac{V_{W}}{R_{W}} \cos \phi_{V} + \frac{V_{W}}{R_{W}^{2}} \sec^{0} \phi_{V}}{R_{W}}\right) \quad \delta \phi + OK6 \times \delta \tilde{V}_{W} \\ \delta \tilde{V}_{W} &= \frac{A_{W}}{R_{W}} \phi_{W} - \frac{g}{R} \phi_{W} + (\rho_{V} + 2\Omega_{V}) \quad \frac{\delta V_{W}}{R_{W}} + \frac{V_{W}}{R_{W}} \tan \phi_{V} \frac{\delta V_{W}}{R_{W}} + \\ &+ \frac{V_{W}}{R_{W}} \left(\frac{V_{W}}{R_{W}} \sec^{0} \phi_{V} + 2\Omega \cos \phi_{V}\right) \quad \delta \phi + OK6 \times \delta \tilde{V}_{W} \\ \phi_{W} &= \sigma_{W} + \omega_{X} \phi_{W} - \omega_{V} \phi_{W} - \Omega_{V} \delta \phi + \frac{\delta V_{W}}{R_{W}} \\ \phi_{W} &= \sigma_{W} + \omega_{X} \phi_{W} - \omega_{V} \phi_{W} - \Omega_{V} \delta \phi + \frac{\delta V_{W}}{R_{W}} \\ \phi_{W} &= \sigma_{W} + \omega_{X} \phi_{W} - \omega_{V} \phi_{W} - \frac{\delta V_{W}}{R_{W}} \\ \phi_{W} &= \omega_{V} \phi_{W} - \omega_{X} \phi_{W} + \Omega_{L} \quad \delta \phi + \frac{V_{W}}{R_{W}} \sec^{0} \phi_{V} \delta \phi + \tan \phi_{T} \quad \frac{\delta V_{W}}{R_{W}} + EN77 \times \delta \phi_{W} \\ \phi_{W}^{L} &= -\phi D_{V} \\ dD_{X}^{R} &= -\phi D_{V} \\ dD_{X}^{R} &= -\phi D_{X} \\ dD_{X}^{R} &= 0 D_{X} - K_{V} D_{X}^{R} = OKA \times \delta_{DX}^{T} + BB \times D_{X}^{R} \\ \frac{dD_{Y}^{R}}{dt} &= w_{DY} - K_{V} D_{X}^{R} = OKA \times \delta_{DY}^{T} + BB \times D_{Y}^{R} \\ \epsilon_{X} &= 0 D \times \epsilon_{W} + OKD \times \delta_{T}^{T} \\ \epsilon_{W} &= 0 D \times \epsilon_{W} + OKD \times \delta_{T}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{T}^{T} \\ \epsilon_{W} &= 0 D \times \epsilon_{W} + OKD \times \delta_{T}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{T}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{T}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{T}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{T}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{T}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{T}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{W}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{W}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{W}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{W}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{W}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{W}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{W}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{W}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{W}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{W}^{T} \\ \epsilon_{W} &= 0 D \times \delta_{W} + OKD \times \delta_{W}^{T$$

A DANSING DA

where

, ω<sub>y</sub>, ω<sub>y</sub> = platform spatial rates

x gravity

 $\Omega_{\rm L}$  = level component of earth rate (=  $\Omega \cos \phi_{\rm T}$ )

 $\Omega_{\rm e}$ ,  $\Omega_{\rm e}$  = vertical component of earth rate (=  $\Omega \sin \phi_{\rm e}$ )

 $I_{\rm H}$  ,  $V_{\rm g}$  = platform component of oraft velocity

 $A_{\rm H}$  ,  $A_{\rm g}$  = north and east components of platform acceleration, respectively.

Note, this can also be computed

by 
$$\frac{\partial A_{\rm H} dt}{\Delta t} = \frac{V_{\rm H}}{\Delta t}$$
, for small  $\Delta t$ .)

 $R_{N}$ ,  $R_{M}$  = radii from center of earth.

Superscripts H and S refer to hardware and sea components, respectively, and the  $w_1$  terms are the elements of the internally generated noise matrix. The last four defining equations define the Doppler and gyro states. Note that a first order lag simulation model is used. From the above equations the  $\underline{F}$ ,  $\underline{O}$  and  $\underline{w}$  matrices referred to in Kalman's works can be written.

The  $\underline{X}$ ,  $\underline{\Gamma}$ ,  $\underline{G}$  and  $\underline{w}$  matrices are shown in Table II. The state variable matrix is comprised of the eleven states and  $\underline{\Gamma}$ ,  $\underline{G}$  and  $\underline{w}$  matrices can be written by inspection of the state variable defining equations.

#### TABLE II

#### X. E. G. and w Matrices

,	δφ <sub>1</sub>	δλ. com φt	δV <sub>N</sub> /R <sub>N</sub>	δv <sub>b</sub> 'R <sub>N</sub>	$\phi_{\rm N}$	$\phi_{\rm E}$	$\phi_{\rm g}$	D <sub>X</sub>	D,	<,	«x		
ě¢,	0	D	1	0	0	0	0	0	0	0	ø	8.9	0
$\delta\lambda\cos\phi_{ m T}$	$\frac{V_{\rm H}}{R_{\rm H}} \tan \phi_{\rm Y}$	-V <sub>N</sub> tan dy R <sub>M</sub>	0	1	0	D	0	0	0	0	0	δλ cos φ <sub>τ</sub>	0
۶۴ <sub>N</sub>	$= 2\Omega \frac{V_{\rm B}}{R_{\rm N}} \cos \phi_{\rm T}$ $= \frac{V_{\rm B}^2}{R_{\rm N}^2} \sec^2 \phi_{\rm T}$	0	D	$= (\rho_{v} + 2\Omega_{v})$ $= \frac{V_{H}}{R_{H}} \tan \phi_{T}$	0	g/R	- <u>A</u> n R <sub>N</sub>	0	0	0	0	δV <sub>N</sub> B <sub>M</sub>	CK6 x $\delta_{V_g}^T$
ðů <sub>e</sub>	$\frac{\frac{V_{H}}{R_{H}}}{\frac{V_{E}}{R_{H}}} \frac{sec^{2}\phi_{\gamma}}{score}$ $+ 2\Omega \cos\phi_{\gamma}$	U	(py - 20y)	V <sub>N</sub> tan ¢ <sub>p</sub> N <sub>M</sub>	-g/R	0	Ayi Bu	U	0	0	0	δv <sub>B</sub> R <sub>N</sub>	CK6 × δ <sup>7</sup> <sub>νy</sub>
¢,	-0,	0	0	1	0		ω	0	0	0	1	¢ <sub>N</sub>	0
¢,	0	υ	-1	0	ω <sub>s</sub>	0	-w;	0	0	1	0	φ	0
\$_1	$\Omega_{\rm L} + \frac{V_{\rm H}}{R_{\rm H}} \sec^4 \phi_{\rm g}$	0	0	· tan φ <sub>y</sub>	-w <sub>z</sub>	4	0	0	0	0	0	$\phi_{I\!I}$	EN 77 × 87
Ď,	0	0	0	0	0	0	0	88	-0	0	0	Dx	CKA × 5 <sup>7</sup> <sub>D<sub>1</sub></sub>
D <sub>y</sub>	0	0	0	0	J	0	٥	θ	80	0	0	Dy	CKA × 87
¢,	. 0	0	0	0	0	0	0	0	0	DD	0	¢ <sub>x</sub>	CKD x 5"
4,	0	0	0	0	0	0	0	0	0	0	DD	¢y	CKD × 87
\$			ļ	r X									a ¥

200 - 14.8.8.8.8.8.8.8.

Note that Doppler-error components are chosen rather than ground-speed and drift-angle errors. This was done so that the cross-channel-correlation computation formulas could be used to full advantage. Also, X- and Y-sxis position errors are state variables but system latitude and longitude are updated in the inertial corrections.







tering Officiation Official



300.000 100

#### 4.3 Covariants Matrix Computation

The of \_riance matrix and Kalman gain computations form the bulk of the computer program to be mechanized. This section middresses the computations required to propagate the covariance matrix. As previously stated, the covariance matrix is a running estimate of the system performance and hence the ability to predict and correct the system is totally dependent on the exactnoss of this matrix. Regarding one aspect in programming the covariance matrix, the technique employed in propagating the matrix over time and computational exactness sust be investigated.

Litton has chosen to use direct integration of the covariance matrix which was derived on a continuous basis. In keeping with the moduracy requirements, a higher-order-integration algorithm was found necessary. Litton is currently using a product of variables-type algorithm. This algorithm is presented in more detail in the following section.

In the interest of minimizing computer requirements and maintaining good performance accuracy, trade-offs performed on the various elements of the covariance matrix resulted in segmenting the matrix into two major computational parts. These major subdivisions were based on the rate of change of the covariance matrix elements. As a result, the covariance matrix is composed of "fast" loops and "slow" loops. The significance of these loops are that the fast loop components vary quite rapidly and possess a wide dynamic range compared to the slow loop computations. Hence, direct integration of the slow loops at the same rate as the fust loops would result in a more coarse estimation in that round-off effects would be more pronounced. Therefore, to maintain computational precision, the slow loop processing interval is approximately 80 times the fast loop processing. interval.

Figure 6 illustrates the notational convention and the members of the covariance matrix that are computed. Where blanks appear, the matrix elements are computed by their closed form approximations if these elements are needed. The notation used is a capital letter A, R or B to indicate that the element is a member of that submatrix. The subscripts on these elements define the state variables which are statistically correlated by that function, e.g.,  $B_{k,\psi y}$  is the S-matrix covariance of x-axis velocity (x) and y-axis platform tilt  $(\phi_y)$ .

The covariance-matrix elements which are left blank in Figure 6 are not computed explicitly. If they are needed in propagating other elements, they are computed from closed form, cross-channel correlation formulas<sup>†</sup>. This allows computing of only 42 of the 66 elements of the covariance matrix.

#### 4.3 Integration Algorithm (for Filter)

The programmed integration algorithm is one designed to correctly integrate a product of two variables with constant rates of change.

The algorithm should compute, from t = 0 to  $t = \Delta t$ ,

$$\Delta J = \int_0^{\Delta t} y z dt = \int_0^{\Delta t} (y_0 + \dot{y}_0 t) (z_0 + \dot{z}_0 t) dt$$
$$= [y_0 z_0 + (\dot{y}_0 z_0 + \dot{z}_0) (\Delta t/2) + \dot{y}_0 \dot{z}_0 (\Delta t^2/3)] \Delta t.$$

Consider an algorithm which utilizes the computer value for the preceding increment in the computation of the present increment. Assume the algorithm is correct for variables with constant rate, so that the computed value of the preceding increment is equal to

$$\Delta J_{old} = [y_0 z_0 - (\dot{y}_0 \dot{z}_0 + \dot{z}_0 y_0) (\Delta t/2) + \dot{y}_0 \dot{z}_0 (\Delta t^2/3)] \Delta t .$$

Let the algorithm hy. ? the form

$$\Delta \mathbf{J} = \eta \Delta \mathbf{J}_{n1d} + (1 - \eta) (\mathbf{y}_{n1d} + \mu \Delta \mathbf{y}) (\mathbf{z}_{n1d} + \mu \Delta \mathbf{z}) \Delta \mathbf{t},$$

where  $\gamma_1$ ,  $(1-\gamma)$  weights are implied by the necessity of computing correctly for zero rate variables expressing  $\Delta J$  in terms of  $y_0$ ,  $z_0$ ,  $\dot{y}_0$ ,  $\dot{z}_0$ .

$$\Delta \mathbf{J} = \eta [\mathbf{y}_0 \mathbf{z}_0 - (\mathbf{\dot{y}}_0 \mathbf{z}_0 + \mathbf{\dot{z}}_0 \mathbf{y}_0) (\Delta t/2) + \dot{\mathbf{y}}_0 \mathbf{\dot{z}}_0 (\Delta t^2/3) ] \Delta t + (1 - \eta) (\mathbf{y}_0 + \mu \dot{\mathbf{y}}_0 \Delta t) (\mathbf{z}_0 + \mu \dot{\mathbf{z}}_0 \Delta t) \Delta t$$

$$= \left[ \mathbf{y}_0 \mathbf{z}_0 + (\mu(1 - \eta) - \eta/2) (\mathbf{\dot{y}}_0 \mathbf{z}_0 + \mathbf{\dot{z}}_0 \mathbf{y}_0) + \left( \frac{\eta}{3} + (1 - \eta) \mu^2 \right) \mathbf{\dot{y}}_0 \mathbf{\dot{z}}_0 \Delta t^2 \right] \Delta t$$

$$= \left[ \mathbf{y}_0 \mathbf{z}_0 + \mathbf{0}, (\mathbf{\dot{y}}_0 \mathbf{z}_0 + \mathbf{\dot{z}}_0 \mathbf{y}_0) + \mathbf{0}, \mathbf{\dot{y}}_0 \mathbf{\dot{z}}_0 \Delta t^2 \right] \Delta t ,$$

<sup>†</sup> The closed form relationships utilize invariant products.











For correctness, the linear and quadratic coefficients  $C_1$  and  $C_2$  must be 1/2 and 1/3, respectively. Hence take

$$\mu(1-\eta) - \eta/2 = 1/2$$
  
$$\eta/3 + (1-\eta)\mu^2 = 1/3 ,$$

which gives

$$\mu = 1/\sqrt{3} = 0.877$$

$$\eta = \frac{2-\sqrt{3}}{2+\sqrt{3}} = 0.0718.$$

An approximate solution

$$\eta = 2^{-4}, \quad \mu = (2^{-1} + 2^{-4})$$

implies

Į,

$$C_1 = \frac{1}{2} \left( 1 - 2^{-7} \right), \qquad C_2 = \frac{1}{3} \left[ 1 - \frac{3}{64} \left( 1 + \frac{1}{64} \right) \right],$$

which might be adequate if desired.

The general form of error-propagation equations, including the invariant form, involves the summation of disparate terms to obtain each integral increment of a basic variable during a program cycle.

The computation scheme as stated implies the storage of 42 quantities each for old values of integral increments, for interpolated values, and for integrated state variables. This totals 126 words of storage for the integration algorithm. This is 54 words more than the simplest natural algorithm.

Consider algorithm computational requirements for initialization and inertial system correction where the old estimates of the integral increments may not be available. One approach is to add a cycle of the increment computation where there is no accumulation of the integral, and thereafter return to the normal program for the first iteration, using  $\mu = 1/4$ ,  $\eta = 2/3$ . System corrections with the Doppler radar are programmed as a quasi-continuous process in which their effects in error estimation are part of the steady-state integration process. The position first, wariables. This may be accomplianted with the pseudo-cycle computation with initialization after dead reckoning with Doppler can be tolerated without a pseudo-cycle, the pseudo-cycle can be obviated altogether provided the steps in a pseudo-old-increment of A, R for position fix are computed. These increments are computed by formulas derived by incrementing the difference equations for error growth of A, B, for increments of A on fixing. The additional computations involve 15 multiplications.

#### 4.4 Implementation of Invariants-Scaling Technique

In implementing a scaling algorithm, "real-world" constraints dictate several of the limiting conditions mechanized. For instance, separate scales are needed for various mission phases; however, to fix a distinct set of scale factors obviously imposes system constraints and these constraints lead to the development of an adaptive rescaling module. The rescaling module would be exercised at the beginning of each filter-updating cycle. The successful implementation of this rescaling technique relias on the regular behavior of the elements of the covariance matrix. A large change could cour in these elements if, for instance, a position observation is made. If the incremental change in these elements is too large to be accommodated in one filter cycle, a "segmented" fix can be easily mechanized so as to avoid register overflow or constrain maximum information by left-shifting if the quantity being updated is rapidly decreasing.

In implementing the rescaling module, the philosophy is to test the disgonal elements of the covariance matrix and to constrain the scaled quantities to lie within a specified range of the register by controlling the scale factors. With the chosen constraints mechanised, the rescaling loop shifts left or right by one bit if the rescaling tests fail.

#### 4.4.1 Initialization Scaling

For the scaling of the program parameters, all of the scale factors are formed from basically six factors, AP(1) through EF(6). The numerical value in parentheses refere to the type of parameter being scaled, e.g.,

MF (1) position
MF (2) velocity
MF (3) tilt
AF (4) gyro drift
MF (5) Doppler
OF (6) aximuth.



# Arryni 💭 ny sa yang 🛄 sa bag

in the Rough's states

At time mero, values for each of the scale factors are selected.

The elements of the covariance matrix can be scaled by implementing the equivalent to the following FORTHAN DO loop.

$$\begin{array}{rcl} D & 4 & 1 & = & 1, & 5 \\ D & 4 & J & = & 1, & 5 \\ BFP(1,J) & \simeq & 1/SF'(1)/SF'(J) \\ X(I,J) & = & SFP(I,J)^*X(I,J) \\ 44Y(I,J) & = & SFP(I,J)^*Y(I,J) \end{array}$$

Scaling of special functions at time zero is performed separately.

All parameters must have an assigned scaling before their insertion into the program; however, this is not a drawback, since the chosen parameters in the filter program can be rescaled at will before the actual filter initialization by implementing the "litten technique for aduptive rescaling, e.g. the quantities BP(J), J = 1, 6 are rescaling factors for changes in the covariance matrix elements X(T,J), Y(T,J) and for the quantities DX(T,J) and DY(T,J) and DY(T,J) and for the quantities X(T,J) and Y(T,J), where DX(T,J) and DY(T,J) are the propagation increments to covariance matrix elements X(T,J) and Y(T,J), respectively. Note: indexing is used in programming the ovariance matrix in order to minimize computer instructions. This does result in some processing time penalty.

#### 4.4.2 Rescaling Module

The Litton adaptive rescaling module is to be tested at the beginning of wach complete pass through the filter. The philosophy is to scale both channels identically; therefore, the maximum values of the corresponding diagonal elements (designated by AMM(J)) of the dowariance matrix must be determined. Easing constraints on the elements are set so that all state variables lie between 1/16 and 1/4 of full scale except for asimuth, which is mosted between 1/4 and 1/2, since division by azimuth is necessary in the closed form cross-channel correlations. With these constraints mechanismd, the remainder of the scaling loop is necessary to scale the quantities indicated by whifting either left or right one bit, depending on the whifting parameter. Figure 7 is a flow chart of the rescaling program.

1 set

#### 8. SIMULATION REBULTS

This section contains simulation results for a test mission. The assumptions are that the system was groundaligned using conventional tochniques and the filter was initialized at the instant the vehicle began moving. In order to illustrate the potential accuracy improvement by using a filter, the system and filter were run "open-loop", i.e., the filter was run as an estimator, not a controller, such that normalized rather than dependent errors could be compared.

In the system-performance curves (Figures 8 through 11) the net resulting error is the difference between the estimated and sotual mystem errors. The curves presented in Figures 8 through 11 are based on a profile which contains several velocity fluctuations; starting and stopping; during the profile; 45° latitude; and error sources similar to an LN-15 type platform (the system errors are larger than that expected from an actual LN-15 for a similar spplication; however, for the simulated mission, the full knowledge of error-source correlations was not exployed).

#### REFERENCES

1. Kalman, A.M.

New Methods and Results in Linear Prediction and Filtering Theory, WIAM Technical -Report 61-1.

2. Manbrook, H. et al. LORAN D/Inertial Navigation System Integration. Litton Publication AQ22009M65 (Air Force contract No.AF-33(887)-14389). January 1986.



the second states and the second s

in the second 1100

.





Fig.3 Doppler-error model



Fig.4 Velocity-difference-signal formation





Fig.5 IMU error model block diagram

਼ਿ













A LANDAL FROM

fer tant and the set of the set





A-ELEMENTS: POSITION ELEMENTS WITH POSITION ELEMENTS R-ELEMENTS: NON-POSITION ELEMENTS WITH POSITION ELEMENTS S-ELEMENTS: NON-POSITION ELEMENTS WITH NON-POSITION ELEMENTS

Fig.6 Notation convention used in partitioned covariance matrix





STATES STATES



State and Ott

Fig.8 Longitude error versus time



Fig.9 Latitude error versus time





à





449

17

C HATTY ALD

KOY.

#### CHAPTER 17 - MARINE APPLICATIONS OF KALMAN FILTERING

by

Sector Sector

1.12 1.12 Parts &

BARRANKA INTERNET

A. 18 1.

10.0

J. Holdsworth and J. Stolz

Advanced Marine Technical Division Litton Industries 9920 W.Jefferson Blvd Culver City, California, UBA

## PRECEPTING PAGE BLANK

451

THE WAY Distant Sont

۶,


STRUCTURE STRUCTURE STRUCTURE



# CHAPTER 17 - MARINE APPLICATIONS OF KALMAN FILTERING J.Holdsworth'and J.Stols

## 1. INTRODUCTION AND DESCRIPTIONS OF THE PROBLEM

This chapter is concerned with the application of Kalman filtering techniques to estimating the equations of motion of submerged objects in the ocean. Applications can be made to monitor the movement of submarines as well as the movement of schools of large fish or submarine mammals. Another interesting application is the optimal control of cabled communication, wire-guided submarine devices. Here the estimation of the remote control device's track would be useful to evaluate if a device has sufficiently searched a particular topographical area of interest. Also of interest is the situation where a single command and control ship monitors and controls the operation of some activity in the ocean area of its immediate vicinity. The Kalman filter can be used to track the sotivity of all objects in a predetermined area. Thus the Kalman filter can be used to track of scuba divers in underses exploration or in the harvesting of food on the ocean floor.

The ocean is the medium in which the Kalman filter's estimation process has to operate. Communications between two objects separated by several thousand yards is obtained by sound. The speed of sound in water is nominally 4800 ft/sec and it is primarily a function of water temperature, pressure, and salinity and is given by the empirical formula

 $C = 4422 + 11.251 - 0.0451^2 + 0.01820 + 4.3 (s - 34)$ 

where

× N

- C = velocity of sound (ft/sec)
- d = water depth (ft)
- t = temperature of water at d ( $^{O}$ F)
- s = salinity (parts per thousand).

The propagation path of sound in water is quite different from the paths of electromagnetic radiations used by radar tracking systems. The propagation path of sound is quite non-linear and its general nature can be described by ray theory<sup>1</sup>. The different modes of propagation are various combinations of direct path, surface channel, bottom bounce and convergence zone. The direct path mode is where the sound path that connects two objects is not reflected by any accustical boundary. The surface channel mode is where the sound path consecting two objects is reflected at least once by the physical boundary between the water and the atmosphere. The bottom bounce mode is where the sound path connecting two objects is reflected at least once by the boundary between the water and the ocean floor. The convergence sone mode is where the sound path connecting two objects is refracted due to the sound ruy "vertexing". This occurs when the ray path attains a horizontal tangency at a certain ocean dapth<sup>1</sup>.

Thus, the total transit time of sound from one object to another is a function of the propagation path and the velocity of sound in the water. The different propagation modes are illustrated in Figure 1.

The underwater systems that either communicate by sound or monitor sound sources are called sonars. The sonar system of interest to us is the echo-ranging sonars, which operate by monitoring sound reflected by an underwater object that was originally an input to the water by the receiving ship. The measurements taken by the echo-ranging sonar are range, bearing and sometizes range-rate. The range information (R) is calculated by

$$\mathbf{R} = \frac{\mathbf{t}}{2} \mathbf{V}_{\mathbf{s}} , \qquad (1.1)$$

where t is the total travel time of sound and  $V_{\mu}$  is the average speed of sound in water. The bearing information (B) is obtained by measuring the direction of the reflecting underwater object. The range-rate (R) informatice is obtained from the relationship

 $f = 0.69 f_{\rm g} R$ , (1.2)

where  $f_{g}$  is the frequency of the transmitting source (kc/s), f (c/s) is the observed change in frequency when the sound signal is returned and 0.69 is a Doppler constant for echo transmission (c/kt kc). The measured observations (B, K, Å) are corrupted by several types of noise sources. The structure of the noise is very complex<sup>1</sup> and will not be dealt with here. The accuraties to which the observations B, R, Å can be measured are a function of beam-width, pulse-length, frequency discrimination, and signal-to-noise ratio measuring characteristics of each sonar. For the purposes of this analysis, the additive noise on bearing information can be thought of as a stationary mero mean, gaussian process where the standard deviation of the error is about 0.5 degrees. The range information can also be thought of as a gaussian process where the variance changes linearly with relative range (15) (distance between one object and another). The time difference between consecutive samples is long enough

# **PRECEDING PAGE BLANK**

100.00

such that autocorrelation of bearing and ranges is insignificant. Also the orosa correlation between ranges and bearings is insighificant. The accuracies of the range-rate information are about 6 knots.

3. APPLICATION OF KALMAN FILTERING TO THE MARINE Tracking problem

## 2,1 General

We shall now briefly describe the basic equations which are required to perform the tracking process. The tracking process is basically a Kalman filter operating on the observations of range, bearing, and Doppler. The above observational set corresponds to the active sonar mode of operation. The equations for the passive sonar mode are different and will not be considered here. The observations have first to be corrected to give true bearing, horizontal range and range rate. This involves pre-processing the observations to account for the sound propagation mode. The observation pre-processing can be accounted for directly in the Kalman filter itself, but this would greatly complicate the formulation. Our truatment will be brief and descriptive instead of detailed and rigorows. Derivations of the equations for the subitoms mean square prediction error oriterion are well known<sup>2</sup> and are detailed in previous sections of this publication.

## 2.2 Kalman Filter Equations

The basic component of the estimation process is a state vector whose value at some time ty is given by

$$\dot{X}(k) = \begin{bmatrix} \dot{X} \\ X(k) \\ \dot{Y} \\ Y(k) \end{bmatrix}$$
 (2.1)

The components of the target's state vector are the components of the relative position with respect to our tracking vessel and the absolute components of the velocity of the target. The state equation which describes the dynamic development of the target state vector in time may be written

$$\mathbf{X}(\mathbf{k}) = \phi(\mathbf{k}) \overline{\mathbf{X}}(\mathbf{k}-1) , \qquad (2,2)$$

where, for our case, the state transition matrix is given by

$$\phi(\mathbf{k}) = \begin{pmatrix} 1 & 0 & 0 & 0 \\ \Delta t_{\mathbf{k}} & 1 & 0 & 0 \\ 0 & 0 & 1 & 0 \\ 0 & 0 & \Delta t_{\mathbf{k}} & 1 \end{pmatrix}.$$
 (2.3)

The quantity  $\Delta t_k$  is the time between receipt of the  $(k-1)^{th}$  and the  $k^{th}$  piece of data (report). If the date rate is constant, then  $\Delta t_k = \Delta t$  is constant and the transition matrix  $\phi(k)$  described by Equation (2.3) does not depend upon time. The value of  $\Delta t$  is determined a prior by a range setting that is well beyond the range of the target. The solual processing of the data will be divided into two parts. The first involves range  $(\tilde{R}_k)$  and bearing  $(\tilde{B}_k)$  only. The second involves, in addition to range  $(\tilde{R}_k)$  and bearing  $(\tilde{B}_k)$ , range-rate  $(\tilde{R})$  as well. The data vector  $Z'_k$  for the first set of observations may be written as follows:

$$Z'_{k} = \begin{pmatrix} \tilde{R}_{k} \sin \tilde{B}_{k} \\ \tilde{R}_{k} \cos \tilde{B}_{k} \end{pmatrix} = H'(k)\tilde{X}(k) + \epsilon'_{k}, \qquad (2,4)$$

In Equation (2.4), M'(k) is called the measurement matrix, which relates the observed data vector  $Z'_k$  to the target state vector  $\overline{X}(k)$ . In our case the measurement matrix does not depend upon the time  $t_k$  and may be explicitly written out as

 $\mathbf{M}^{t}(\mathbf{k}) = \mathbf{M}^{t} = \begin{pmatrix} 0 & 1 & 0 & 0 \\ 0 & 0 & 0 & 1 \end{pmatrix}.$  (2.5)

The term  $\epsilon'_k$  denotes the 2×1 noise vector due to the noise perturbations on the measured bearing angle  $\vec{B}_k$ and range  $\vec{R}_k$ . Assuming that the measurement noise on the range and bearing is uncorrelated and mero mean, it follows that the covariance matrix of our noise vector is approximately given by

$$\mathbf{W}_{k}' = \mathbf{g}\left[\mathbf{e}_{k}'\mathbf{e}_{k}'^{T}\right] = \begin{pmatrix} \sin^{2}\tilde{\mathbf{s}}_{k}\sigma_{k}^{2} + \tilde{\mathbf{s}}_{k}^{2}\cos^{2}\tilde{\mathbf{s}}_{k}\sigma_{k}^{2} + \tilde{\mathbf{s}}_{k}'\cos^{2}\mathbf{s}_{k} + \frac{\sin\mathbf{s}_{k}\cos\mathbf{s}_{k}}{\sin\mathbf{s}_{k}\cos\mathbf{s}_{k}} + \frac{\cos\mathbf{s}_{k}}{\mathbf{s}}\sigma_{k}^{2} + \tilde{\mathbf{s}}_{k}'\sin^{2}\mathbf{s}_{k}\sigma_{k}^{2} \end{pmatrix}.$$
(2.6)

The target state vector  $\tilde{X}(k)$  is unknown to us and must be estimated from the observed data vector  $Z'_{k}$  and whatever knowledge we have of the measurement noise added to the observed ranges and bearings. Let  $\tilde{X}(k-1)$  denote the estimate of the target state vector based upon whatever a priors knowledge we have, together with all the observed data obtained up to and including time  $t_{k-1}$ . At time  $t_k$  a new data vector  $Z'_k$  is observed.





4 5 5

In the Kalman filter model, the new updated estimate of the target state vector is recursively determined according to the following equation:

$$\hat{X}(k) = \phi(k)\hat{X}(k-1) + K(k)\left[Z_{k}' - M'\phi(k)\hat{X}(k-1)\right]. \qquad (2.7)$$

In Equation (2.7), the term  $\phi(k)\hat{X}(k-1)$  is the updated prediction of the target state from the dynamic Equation (2.2), based just on the data available at stage k-1 and before. Likewise the term

 $M'\phi(k)\vec{X}(k-1)$  (2.8)

is our prediction of the measurement vector  $Z'_k$  based on the  $(k-1)^{\text{th}}$  stage data, so that  $Z'_k = M' \psi(k) \hat{X}(k-1)$  (2.9)

is the observed discrepancy between the measured data vector at stage k and the predicted data vector based upon prior information.

K(k) in Equation (2.7) is the Kalman gain matrix which weights the observed measurement discrepancy. It is chosen to minimize the mean square error in the estimates of the components of the target state vector. If we let P(k) denote the covariance matrix of the estimate of the target state vector at time  $t_k$ , i.e.,

$$P(k) = E[(\hat{X}(k) - \hat{X}(k))(\hat{X}(k) - \hat{X}(k))^T],$$
 (2.10)

where E[] denotes the expected value operation, then it may be shown<sup>4</sup> that the Kalman gain matrix may be written as

$$K(k) = \phi(k)P(k-1)\phi^{T}(k)M^{T}[M'\phi(k)P(k-1)\phi^{T}(k)M^{T} + W_{k}^{\prime}]^{+1}, \qquad (2,11)$$

Beginning with an initial dovariance matrix P(0) of an estimated state vector, it may be shown that the development of the target state dovariance matrix , a function of  $t^{4}$ me is obtained from the following resurgive equation:

$$P(k) = \phi(k)P(k-1) \int (k) - K(k)M'\phi(k)P(k-1)\phi^{T}(k), \qquad (2, 12)$$

When Doppier data is available the procedure, wha combined with the foregoing process, is as follows. The computational equations are quite similar for the second set of observations. The state transition matrix  $\phi(\mathbf{k})$  becomes

$$\begin{pmatrix} 1 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 \\ 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 1 \end{pmatrix}$$
(2.13)

and the data vector may be written as

$$Z_{k}^{\prime\prime} = \begin{pmatrix} \bar{R}_{k} \sin \bar{S}_{k} + \bar{K}_{k} \cos \bar{S}_{k} \bar{S}_{k} \\ \bar{R}_{k} \cos \bar{S}_{k} - \bar{R}_{k} \sin \bar{S}_{k} \bar{S}_{k} \end{pmatrix} = M^{\prime\prime}(k) \bar{X}(k) + \epsilon_{k}^{\prime\prime} . \qquad (3.14)$$

In Equation (2.14), M"(k) is given by

or by

$$\mathbf{M}''(\mathbf{k}) = \mathbf{M}'' = \begin{pmatrix} 1 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 \end{pmatrix}$$
(2.15)

and, similar to Equation (2.4), the term  $\epsilon_k''$  in Equation (2.14) is a 2 × 1 noise vector due to the noise perturbations on the measured bearing angle  $\tilde{B}_k$ , range  $\tilde{R}_k$ , range-rate  $\tilde{R}_k$  and the support observation  $\tilde{B}_k$ . Since  $\tilde{B}_k$  is not an explicit observation, it can be synthesized by either of two methods. The term  $\tilde{B}_k$  can be culculated by

$$\mathbf{i}_{\mathbf{k}} = (\mathbf{i}_{\mathbf{k}} - \mathbf{i}_{\mathbf{k}-1})/\Delta t \tag{2.16}$$

$$\vec{B}_{k} = \frac{Y(k)\dot{X}_{k} - X(k)\dot{Y}_{k}}{Y(k)^{2} + X^{2}(k)}, \qquad (2.17)$$

The components Y(k) and X(k) in Equation (2.17) are obtained from Equation (2.1). The terms  $X_p$  and  $Y_p$  are obtained by

$$\begin{array}{c} \mathbf{x}_{\mathbf{r}} = \mathbf{x} - \mathbf{x}_{\mathbf{0}} \\ \mathbf{t}_{\mathbf{r}} = \mathbf{y} - \mathbf{y}_{\mathbf{0}} \end{array}$$

$$(2.18)$$

where  $\hat{X}_0$  is the motion of the observation vessel in the X-direction and  $\hat{X}_0$  is the motion of the observation vessel in the Y-direction. Also, the range-rate information in the data vector given by Equation (2.14) is relative range-rate, which was obtained by

$$\tilde{R}_{\mu} = \tilde{R}_{\mu} - \dot{R}_{0}$$
 (2.19)

The term  $\tilde{R}_{\mu}$  is the actual observation obtained from the sonar. The term  $\tilde{R}_{\mu}$  is obtained from the observational vassel's mensors. The covariance matrix of the noise vector given by Equation (2.14) may be written us

WALL DISLOATS 1 1

The same process is repeated for Equations (2.7), (2.11) and (2.12), replacing  $Z'_k$ , M', and W'\_k by  $Z''_k$ , M' and M''\_k.

The beginning of the Kalman filtering process is given by

$$\mathbf{F}(0) = \begin{pmatrix} 5000 & 0 & 0 \\ 0 & 5000 & 0 & 0 \\ 0 & 0 & 5000 & 0 \\ 0 & 0 & 5000 & 0 \\ 0 & 0 & 0 & 5000 \end{pmatrix}, \qquad (2.21)$$

where the quantity 5000 is somewhat arbitrary, the important thing being that the diagonal elements of P(0) be large so that the filtor will forget the initial values as more data arrive, and

$$X(0) = \begin{bmatrix} 0 \\ 0 \\ 0 \\ 0 \end{bmatrix}$$
 (2.22)

The process can be carried out iteratively for each new set of observational reports.

The variables  $\sigma_{g}$ ,  $\sigma_{h}$ ,  $\sigma_{h}$  in Equations (2.6) and (2.20) are determined a priori and can be readjusted adaptively if desired. The variable  $\sigma_{h}$  in Equation (2.20) is calculated by

$$\sigma_{\pm} = \sqrt{2\sigma_{\pm}/\Delta T} \qquad (2.23)$$

## 3. THE MANEUVERING TARGET PROBLEM

## 3.1 General

In the typical tracking problem the target (submarine object of interest) will not, in general, maintain a policy of a constant course and opeed. If the target is a submarine, it will usually perform a sequence of short straight-line maneuvers. If the target is a cabled communication, wire-guided submarine device, the movements are constrained by the topography of the ocean floor. Since the mission usually requires that the target operates in close proximity to ocean floor the motion could be highly nonlinear. In this case, the classical estimation of the target's motion will be corrupted by the nonlinear movement of the target. Thus the problem indicates a need to model directly the uncertainties of the target's motion in the estimation process. The Kalman fliter can be adapted to reflect explicitly the uncertainty of the target's state estimates at any time due to the random unalinearity of the target. This has led to the development of an uncertainty gain matrix  $U_{\rm g}$ .

3.3 The Derivation of the Uncertainty Gain Matrix  $(U_{\alpha})$ 

The state vector of a target with random maneuvaring capability is given by

where X ,  $X_{g}$  ,  $K_{g}$  are 4 x 1 column vectors.

In Equation (3.1),  $\overline{X}_{\mu}$  is given by



(3. 2)

(3.1)

TRANT PRANT

and N<sub>a</sub> is given by

 $\vec{R}_{\underline{u}} = \begin{bmatrix} \vec{u}_{\underline{d}} \\ \vec{u}_{\underline{d}} \\ \vec{u}_{\underline{d}} \\ \vec{v}_{\underline{d}} \end{bmatrix}, \quad (3, 3)$ 

In Equation (3.2) the components represent the true motion of the target for the non-managevering case and the expected motion for the random managevering case. In Equation (3.3) the components represent the deviations about the expected state vector for the managevering case. Thus the error covariance matrix P is given by

$$\mathbb{E}\left(\left(\hat{X} - \bar{X}\right)\left(\hat{X} - \bar{X}\right)^{\dagger}\right) = \mathbb{E}\left(\left(\hat{X} - \bar{X}_{n}\right)\left(\hat{X}\bar{X}_{n}\right)^{\dagger}\right) + \mathbb{E}\left(\mathbb{E}_{n}\mathbb{R}_{n}^{\dagger}\right)$$

$$(3.4)$$

$$\mathbf{P} = \mathbf{P}^{T} + \mathbf{U}_{\mathbf{G}} \,. \tag{3.6}$$

In the non-maneuvering case, the uncertainty gain matrix ( $U_0$ ) is zero in Equation (3.6) and  $P=P^2$  .

The components are derived as follows:

$$\epsilon_{\pm} = \epsilon_{\pm} \sin \theta + \epsilon_{\pm} \otimes \cos \theta$$
 (3.7)

 $I_{\mu} = \left\{ e_{\mu} \cos \theta - e_{\mu} \otimes \sin \theta \right\}$  (3.9)

$$s = \epsilon_{s} \Delta t$$
 (0.11)

where  $\mathbf{s} = \mathbf{target}$  speed,  $\mathbf{C} = \mathbf{target}$  course and  $\Delta \mathbf{t} = \mathbf{times}$  between consecutive data reports.

All the components of  $N_{g}$  are assumed to be uncorrelated, as well as the < quantities. Thus  $U_{d}$  is a diagonal matrix obtained by calculating the variances of Equations (3.7), (3.9), (3.10) and (3.11).

 $\overline{U}_{q} = \begin{bmatrix}
\sigma_{q}^{g} & 0 & 0 & 0 \\
0 & \sigma_{q}^{g} & 0 & 0 \\
0 & 0 & \sigma_{p}^{g} & 0 \\
0 & 0 & 0 & \sigma_{p}^{g}
\end{bmatrix}$ (3.12)

The components of Equation (3.12) are

$$\sigma_{\pm}^{2} = \sigma_{\pm}^{2} \sin^{2}\theta + \sigma_{0}^{2} A^{2} \cos^{2}\theta \qquad (3.13)$$
$$\sigma_{\pm}^{2} = \Delta t^{2} \sigma_{\pm}^{2}$$

 $\sigma_{g}^{2} = \sigma_{g}^{2} \cos^{2} 0 + \sigma_{0}^{2} \, S^{2} \, \sin^{2} 0 \qquad (3.14)$ 

 $\sigma_y^4 = \Delta t^2 \sigma_y^4.$ 

In Equations (3.13), (3.14),  $\sigma_{\rm g}$  is the speed change capability in the time interval  $\Delta t$  and  $\sigma_{\rm g}$  is the course change capability in the time interval  $\Delta t$ . The variables  $\sigma_{\rm g}$  and  $\sigma_{\rm g}$  are obtained by

$$\sigma_0 \approx 0\Delta t$$
, (3.16)

where **A** in Equation (3.15) and O in Equation (3.18) are input quantities that describe the target's speed and course change capability.

## 4. ADAPTIVE DETERMINATION OF NEASUREMENT NOISE ONARACTERINTICS

Thus far in our treatment we have assumed that we possessed total statistical knowledge of the noise which is assumed to additively corrupt our range and bearing measurements. As oun be seen from several of the foregoing equations, for example, Equation (3.6), the values of the bearing and range error measurement noise variances are necessary to calculate the gain matrix. A perfect knowledge of the defining parameters of the noise statistics would permit us to utilize these quantities in the most efficient manner in the design of the filter and to construct a theoretically optimal filter.

Litter**O**rrichten**O**reite

and the little state of the second

۰.

and the second second

(UNICE) ADDIDE OUTDES AND IN

Unfortunately one rarely has complete knowledge of the statistical structure of the measurement noise. Moreover, in marine applications the actual noise is likely to possess a very complicated structure so that the assignment of fixed numerical values to such quantities as measurement variances is likely to be presumptuous at best. This practice of assuming values for measurement moles variances and building them into the filter could result in the construction of the best filter for the wrong problem.

intuitively, what one would like is to design a filter which had a minimal dependence upon numerical values of parameters which were known with imperfect confidence. Some knowledge of measurement noise variances is necessary, however, in order to allow the filter to apply the appropriate weighting to the input data. This suggests the pensibility of designing the filter so that it can estimate, or at least monitor, the value of the measurement noise variances to ensure that the proper weighting was being given to the data. If this could be done, it would release the filter designer from the necessity of supplying numerical values for measurement noise variances where the validity of these values was quite uncertain.

One manner in which this adaptive variance estimation might be performed is as follows. Let us denote

$$\bar{\chi}^{+}(k) = -\psi_{-}(k)\hat{\chi}(k-1) = \begin{pmatrix} \hat{\chi}^{+} \\ \chi^{+}(k) \\ \hat{\chi}^{+} \\ \Upsilon^{+}(k) \end{pmatrix},$$
 (4.1)

Then based upon our  $(k-1)^{th}$  stage estimate, we may predict the value of the bearing to be observed at time  $t_k$  by writing

$$J_{\rm R}^{\rm 3} = \tan^{-1} \frac{\chi^{\rm 4}(k)}{\chi^{\rm 4}(k)}$$
, (4.2)

At stage n , the bearing error variance may be adaptively estimated from the equation

$$\partial \overline{J}_{\mathrm{H}} = \frac{1}{n} \sum_{i=1}^{n} (\alpha_i \gamma_i^* - \overline{\beta}_i)^{T_i}$$
(4.8)

where  $\mathcal{I}_{i}^{2}$  is the actual measured value of the bearing at time  $t_{i}$ . For real time processing Equation (4.3) may be written in the following very simple reduces form:

$$\hat{\sigma}_{\beta_{n}}^{\sharp} = \left(1 - \frac{1}{n}\right) \hat{\varphi}_{n-1}^{\sharp} + \frac{1}{n} \left(\hat{\omega}_{n}^{\dagger} - \hat{\beta}_{n}\right)^{\sharp} . \tag{4.4}$$

Similarly, if we define our predicted range  $R_i^{\bigstar}$  to be given by

$$R_{1}^{+} = \sqrt{(X^{+}(1)^{2} + Y^{+}(1)^{2})}, \qquad (4.5)$$

then equations analogous to Equations (4,3) and (4,4) may be written to adaptively estimate the variance on the range measurement errors.

One of the holes sources might cohostually be non-stationary. In particular, in many situations it seems that the range measurement error variance is itself dependent upon range. In this event we could account for the non-stationarity in the noise variance by limiting the "memory" of our adaptive range variance estimator to the most recent m pieces of data. In this case the "smoothed" estimated range variance at time  $t_{\rm H}$  would be given by

$$\partial_{\mathbf{R}_{1/\mathbf{m}}}^{\mathbf{r}} \approx \frac{1}{\mathbf{m}_{1}} \sum_{n=1}^{\mathbf{n}_{1}} (\mathbf{R}_{k}^{\mathbf{r}} - \mathbf{R}_{k})^{\mathbf{r}}.$$
 (4.6)

The trick in using Equation (4.6) is to choose m to be sufficiently large so that sampling fluctuations are smoothed out in the computation, but sufficiently small so that the actual variation of the true variance with the range is reflected. Equation (4.6) can be easily manipulated and written in the following recursive form for real time computation.

$$\hat{\sigma}^{a}_{\bar{R}_{n/m}} = \hat{\sigma}^{a}_{\bar{R}_{(n-1)/m}} + \frac{1}{m} \left[ (\bar{R}^{a}_{\bar{n}} - \bar{R}_{n})^{a} - (\bar{R}^{a}_{\bar{n}+1,m} - \bar{R}_{n+1-m})^{a} \right].$$
(4.7)

We have applied the adaptive techniques just described to the case of a linearly moving and a maneuvering target. The resulting root mesh square speed errors, based upon 10 Monte Carlo runs are shown in Figure 3. This shows that the results of the adaptive variance estimation gave considerable improvement over the non-sclaptive estimates, where, for the non-sclaptive case, the input measurement standard deviations were in error by a factor of 3. Errors in knowledge of input measurement variances can often be this large so that we feel that this figure represents a registive situation and that erforts to extend such adaptive techniques should be pursued.



المناجع والمعادية والمعالية والمعالية والمحاج و

The potentiality for an adaptive measurement noise variance estimation capability to be included within the filter is especially promising for tracking situations where the tracking extends over an appreciable period of time. For very short term tracking engagements the times involved are likely to be too short for the adaptive variance estimates to stabilize. Thus, for short tracking encounters, one will probably have to be content with utilization of whatever a priori knowledge he has of the measurement noise variances in the data weighting. However, for long time period tracking, this technique looks very promising.

#### 5, EXAMPLES

We will consider four typical steering patterns for submarines; constant course and speed (Fig. 3), fast 90<sup>6</sup> turn (Fig. 4), slow 90° turn (Fig. 5), and a sequence of maneuvers that will maintain an average speed in a preferred direction (Fig. 6). The constant course and speed geometry will test how well the Kalman filter can track a cooperative submarine. For this case the Kalman filter will have a zero uncertainty gain function and this will demonstrate its heavy smoothing capability. The fast 90° turn maneuver will test two important factors. The first factor is the maneuver recognition time and the second is the tracker recovery time. The maneuver recognition time is the time from the beginning of a sharp manouver to the time the Kalman filter is definitely correcting to the new heading. The tracker recovery time is the time from the termination of the sharp muneuver to the time that the Kalman filter is estimating the new course and speed within some predetermined error bounds. The slow 90° turn manouver will test the Kalman filter's steady-state response to a constant rate course change The sequence of fast 90° maneuvers will test the Kalman filter's sbility to track a rapidly maneuvering submarine The Kalman filter will never be able to estimate, to a high degree of accuracy, the target course and speed, but it will be able to given an indication of the agility of the tracker for different uncertainty gain functions. In this case the position errors and velocity errors become the important evaluation factors. For the experiments we will consider a North-East coordinate system, where North is the Y-axis and East is the X-axis. We will also have the tracking platform (our ship) standing still at X = 0 and Y = 0.

## Geometry I. Constant Course and Speed

The geometry is illustrated in Figure 3. Initially the target starts at X = 2000 yards and Y = 10,000 yards. The target will maintain a constant speed of 25 knots on a course of 135 degrees.

## Geometry II. Fast 30-Degree Turn

ł

The geometry is illustrated in Figure 4. Initially the target starts at X = 2000 yards and Y = 10,000 yards on a course of 180 degrees maintaining a speed of 25 knots. After 204 seconds the submarine performs a 45-second maneuver to a new course of 90 degrees and a speed of 15 knots. At the time of the maneuver the target was approximately 7000 yards from our own ship.

## Geometry III. Slow 90-Degree Turn

The geometry is illustrated in Figure 5. Initially the target starts at X = 2000 yards and Y = 10,000 yards. on a course of 180 degrees maintaining a speed of 25 knots. After 10 seconds the submarine performs a 450-second maneuver to a new course of 90 degrees at the same speed of 25 knots.

## Geometry IV. Sequence of Fast 90-Degree Turns

The geometry is illustrated in Figure 6. Initially the target starts at X = 3000 yards and Y = 10,000 yards on a course of 180 degrees maintaining a speed of 25 knots. The submarine will perform a sequence of 90-degree turns so that it /ill still have a preferred direction of 180 degrees. The speed in the preferred direction was to be 90% of the actual speed of the submarine. The time duration of each maneuver was 80 seconds, so that the final range will be approximately 4000 yards.

## 6. ANALYSIS OF RESULTS

The purpose of this mection is to present a description of the graphical results. All the computer runs were made with 10 different Monte Carlo runs to obtain the root mean square (r.m.s.) error statistics associated with each geometry. The results presented are for r.m.s. course, speed, velocity and position errors. The error in course is found by calculating the r.m.s. difference between the estimated and true course. The error in speed is found by taking the r.m.s. difference between the true and estimated speed. The velocity error is the r.m.s. magnitude of the velocity error vector. The position error is the r.m.s. magnitude of the position error vectors.

The result of the adaptive estimation process is presented by Figure 4. In the weighting of the observational dita it was assumed (guessed) that the noise sigma on the range information was one percent of range and one degree on the bearing observations. Actually the noise sigma was three percent of range and three degrees on the bearing observation. Two effects are immediately apparent. First the adaptive noise estimation process is worthwhile in that it has smaller associated error statistics. Second it gives a significantly more stable estimate of the target's state vector. Both techniques used the same bad guess signas values for the first nine data reports. From the tenth report on, the adaptive Kalman filter used its own estimate of the noise sigmas, while the other continued on with the bad guess.

Salt 🕒 marun stil

------

## The results of the constant course and speed target are presented in Figure 3.

This geometry was run with the Kalman filter processing range and bearing information (KF) and with the Kalman filter with Doppler processing range, bearing and range-rate information (KFWD). The only significant difference is the smaller initial errors of the KFWD tracking system.

The results of the fast 90-degree turn target are presented in Figure 4. As in the previous case, both tracking systems (KF, KFWD) were employed. Due to the rapid closing rate of the target, the initial convergence of the KFWD tracker is very significant. Also the use of Doppler information greatly aided the response of the Kalman filter during the maneuver that occurred at 204 seconds into the run. The time difference between the two trackers to return to the new track is significant.

The results of the slow 90-degrees turn target are presented in Figure 5. The geometry indicates the difference between the steady-state response of the KF and KFWD trackers. The steady-state response to the constant turn rate (0.2 degrees/second) of the KF tracker was 2.5 knots speed error, 9.0 degrees course error and 80 yards position error. The steady-state response of the KFWD tracker was 1.7 knots speed error, 4.5 degrees course error and 70 yards position error.

The results of the sequence of the fast 90-degree maneuvering target are presented in Figure 6. This geometry indicates the Kalman filter's response to a sequence of consecutive 90-degree turns of the target. It is apparent that the KFWD tracker benefits from the Doppler information. The speed error of the KFWD tracker, throughout the entire sequence of maneuver, is one-half the KF tracker's speed errors.

## ACKNOWLEDGEMENT

The authors are indebted to many who have facilitated the preparation of this chapter. In particular, the authors acknowledge the support given them by the Advanced Marine Technology Division of Litton Industries, and especially Mrs.Estelle Hasson, who typed and otherwise prepared the manuscript.

## REFERENCES

1. Tucker, D.G. Galey, B.K. Applied Underwater Acoustics. Pergamon Press, Oxford, 1966.

2. Sorenson, H.W.

Theoretical and Computational Considerations of Linear Estimation Theory in Navigational Guidance. AC Electronics Division, General Motors, Memorandum-LA-3005, November 1965.

















Line The Art Tithin



Tradation to be

an Distant Old

lin an a'

i de la

i Dete



0



Fig.2 Adaptive estimation process

S. 1.



Fig.3 Constant course and speed geometry

HAC NAMES AND A MORE Carrier and MIL INCOME The statistical 1

13**0**1

462

1.6223333

1241212121212

A PRESS NO DECEMBER NO. 49-554

見り見

10 X X X



Pig.4 Fast 90-degree turn geometry





4

. "A

و ا





.



have the second state



Same Maria

ŝ

Ē

Scores Sec

REALISER POTRONS

and the second second and the second s



· · · · · · · · ·





Manual Discondinate Distance

101111111111111111111















the Statestar North



# CHAPTER 18 - OPTIMAL USE OF REDUNDANT INFORMATION IN AN INERTIAL NAVIGATION SYSTEM

F.G.Unger and N.Ott

Ъy

## TELDIX, Heidelberg, Germany

PRECEDING PAGE BLANK

## NOTATION

		1.24
Vectors:	Lower case 1stters (Latin or Greek, with arrows)	
Matrices:	Capital letters (Latin or Greek, underlined)	
Functions and Constan	nts: Capital or lower case letters (Latin or Greek)	
7	thrust acceleration vector	
ชี	total acceleration vector	
ř	position vector	
7	velocity vector	
ដ	rate vector of navigational coordinate system with respect to inertial space	
ವ್ಯ	earth rate vector with respect to inertial space	
7 •	$\vec{\omega} = \vec{\omega}_{\rm g}$ , rate vector of the navigational coordinate frame with respect to an earth-fixed frame	
ř.	vector of mass attraction	
<b>ء</b> ' •	$\vec{z}_{n} = \vec{\omega}_{n} \times (\vec{\omega}_{n} \times \vec{r})$ , gravity vector	
<b>E</b> 0	value of the gravity vector at the surface of the earth	
r	length of the radius of the spherical earth	
♥ .	Accelerometer bias vector	
7	gyro drift vsotor	
φ.	error angle vector between navigational and platform coordinate frame	
alpha	error angle vector between computer and platform coordinate frame	
Δř	position error vector of the inertial system	
<b>⊼</b> v =	$\Delta r$ , velocity error vector of the institial system	
ð <del>v</del>	reference velocity error vector	2.2.1
<b>∀</b> _	velocity vector computed in the inertial system	
₹,	reference velocity vector	
$\mathbf{C}_{\boldsymbol{\nu}}\left(\boldsymbol{\nu=1,2,3}\right)$	control signals in updating mechanisations	
K <sub>v</sub> (∨≈ 1,2,3)	gain constants	
ω	Schuler frequency	1 .3
T <sub>s</sub>	Schuler period	
t, <i>T</i>	time variables	
•	Laplace operator	
I(t)	input signal of a linear system .	
y(t)	output signal of a linear system	i bigan bigan
a,t	Variance	
B(7 <sup>-</sup> )	impulse response of a linear system	
G(#)	transfer function of a linear system	
T <sub>z</sub>	correlation time of the input signal	

		460	
δ(τ)	Dirmo 8-distribution		
$\rho(\tau)$	autocorrelation function		
φ(ω)	power spectral donaity		
j =	<pre>v'(-1) , imaginary unit vector</pre>		
( <u>N</u> )	direction comine matrix	•	and the second
$\overline{x(t)}$ , $\overline{x}_{K}^{+}$	state vector		
	controllable part of the state vector		
xx	uncontrollable part of the state vector		
$\overline{y(t)}$ , $\overline{y_{K}}$	nensurement vector		
<u>A(t)</u>	state coefficient matrix		
$\underline{\mathbf{M}(\mathbf{t})}, \underline{\mathbf{M}_{\mathbf{K}}}$	measurement matrix		
n(t), hg	system white noise vectors		ار در از می از مار از می
$\overline{\mathbf{h}_{\mathbf{x}}}$ , $\overline{\mathbf{h}_{\mathbf{x}}}$	white noise driving functions of the controllable and uncontrollable states		
$\overline{q(t)}, \overline{q_{K}}$	measurement white noise vector		202222222222222
u(t) , u <sub>k</sub>	control vector		
x <sub>K</sub>	a posteriori estimate of the state vector		
xi , xi	the controllable and uncontrollable states of $\hat{\mathbf{x}}_{\mathbf{k}}$		
귝	a priori estimate of the state vector		
$\overline{x_{k}^{1+}}$ , $\overline{x_{k}^{2+}}$	controllable and uncontrollable states of $\overline{x_{k}^{*}}$		(, * 4 <sup>°</sup> , * <sup>1</sup> , * 4 <sup>°</sup> , * 4 <sup></sup>
7	a priori estimate of the measurement		
<u>L</u>	weighting matrix (gain matrix)		
22	covariance matrix of the a priori estimation error.		
<u>Px</u>	covariance matrix of the a posteriori estimation error		
\$ +1/4	state transition matrix		
$\phi_{k+1/k}^{11}, \phi_{k+1/k}^{12}, \phi_{k+1/k}^{22}$	appropriate parts of $\frac{\phi_{R+1}/\kappa}{2}$		ر می و در می و می می این می و می و این می و این می و در می و در می و می و می و در می و در می و در می و در می و می و می و در می و می و در می و د در می و در می و
I	identity matrix		
u <u>k</u>	covariance matrix of the system white noise vector' $\widetilde{h}_K^{\bullet}$	·	
ب	covariance matrix of the measurement white noise vector $\vec{\mathbf{q}}_{k}^{*}$		
FR+1/R	distribution matrix of the control vector $\vec{u}_{K}^{*}$		(* * 4 * * * * * * * * * * *   * * * *
Tit 1/K	appropriate part of $\frac{\sum_{k+1/K}}{k}$ for the controllable states		terre Vienteria's Chief
w(t)	a unity variance white noise source		
1/ß	correlation time		
f(t)	output function of a shaping filter		د میر میر میر میرد. مربخ میر میر میر میرد می
7 <del>48</del>	random walk parameter	•	
ω . 	rrequency in rms/sec .		
astJ N(t).N.	covariance matrix of the white noise vector n(t)		
δ <sub>xL</sub>	Kronecker symbol		

1 .

The state

## CHAPTER IN - OPTIMAL UNE OF REDUNDANT INFORMATION IN AN INERTIAL NAVIGATION SYSTEM

## F.G. Unger and N. Ott

## 1. INTRODUCTION

Accurate and reliable navigation is a very stringent requirement, especially for military missions. Desired outputs from a navigation system are position, velocity and orientation of the sonwors and the vehicle with respect to a navigational coordinate system. A pure inortial navigation system can provide all this information, but the errors of an inertial system are not bounded, i.e. they increase with time. Therefore, with large mission times, the errors can become intolerably large. Other methods of navigation, like Doppler dead reckoning, radio, stellar etc., can provide only part of the information required, but with bounded errors in velocity or position. A navigation system that fulfils a wide variety of requirements for relatively long mission times will therefore be a combination of an inertial system with other navigational side. Such a system can provide all the information required and, at the sume time, control the system errors. But in order to use all the available information in the hest possible way the statistics of the error sources must be known. In addition a heavy load is placed on the digital computer.

This chapter presents the different methods (i.e. conventional versus modern, or deterministic versus statistical) of aiding a pure instital system with additional or redundant navigation information. But, before the basic problem of updating is treated, the available sensor outputs are categorised and a general block diagram for aided institation systems is derived. The pure instital system in a Schuler-tuned mode is included in this block diagram as a special case.

## 3. BLOCK DINGRMA OF ADDED THERTIAL NAVIGATION SYSTEMS

The central unit of a navigation system is the navigation computer. Its inputs are available data from sensors, such as inertial acceleration, ground speed, air speed, heading, position information, stellar angles, etc. These data may be continuous in analogue or digital form, or intermittent like ground fix points. The computer has to calculate the desired navigation outputs: position, velocity and orientation of the sensors and the vehicle with respect to a navigational coordinate system. In addition the computer must size compensate for sensor biases and calculate torquing commands for platform gyros or positional serves for stellar or redar equipment.

Figure 1 shows the possible inputs and required outputs of a navigation system.

 $\left(\frac{d^{\dagger}\vec{x}}{dt^{\dagger}}\right) = \vec{a} + \vec{a}_{n}$ 

 $\mathcal{J} \to \mathcal{I} \to \mathcal{I}$ 

The available sensor information, as well as the computed output data, dan be classified into three levels:

Level 1: Acceleration Level 2: Velocity Level 3: Position and orientation.

Orientation belongs to the same level as powition, because angular measurements lead to position information and both data complement each other. Furthermore, in a rotating coordinate frame orientation angles can be derived by integration of the angular rate vector  $\vec{\omega}$ , just as position is derived by integration of the velocity vector  $\vec{\nabla}$ .

Figure 2 shows the three levels and the equivalent inputs.

Information for a higher level can be derived from a lower level by integration; but it is generally not possible to derive lower level information from higher levels by differentiation.

The information in each level is a vector in the navigation coordinate frame, except for orientation, where the information is in the form of three angles or the direction cosine matrix (M) for possible coordinate transformations,

If the navigation frame is rotating with respect to inertial space, corrections for acceleration must be included. This can be recognized by writing the basic equations

PRECEDING PAGE BLANK

(2.1)

$$\left(\frac{d\vec{r}}{dt}\right)_{t} = \vec{v} + \vec{\omega} \times \vec{r}$$
(2.2)

$$\begin{pmatrix} d^{\theta}\vec{r} \\ d\vec{t}^{2} \\ r \end{pmatrix} = \begin{pmatrix} d\vec{v} \\ d\vec{t} \\ N \end{pmatrix} + \vec{\omega} \times \vec{r} + 2\vec{\omega} \times \vec{v} + \vec{\omega} \times (\vec{\omega} \times \vec{r}) .$$
 (2.3)

where

 $\vec{r}$  = position vector

 $(d\vec{r}/dt)_{1} = time derivative of <math>\vec{r}$  with respect to inertial space

 $(d\vec{r}/dt)_N = \vec{v} \neq derivative of \vec{r}$  with respect to the mavigation coordinate frame

 $\vec{\omega}$  = rate of change of navigational coordinate system with respect to inertial space

a = thrust acceleration vector

 $\vec{s}_m =$ vector of mass attraction

 $\vec{b} = (\vec{uv}/dt)_N = \text{total acceleration in navigation coordinates.}$ 

Substitution of Equation (2.3) into Equation (2.1) yields

B = 〒+ 〒 - 芯 × 〒 - 2芯 × マ - 芯 × (芯 × F) 、

Equation (2.4) shows that the following corrections are required to the acceleration level, in addition to measurable thrust acceleration:

(1) Acceleration due to mass attraction:  $\vec{s}_{ai}$ 

(11) Coriolis acceleration:  $2\vec{\omega} \times \vec{v}$ 

(iii) Tangential acceleration: 芯×干

(iv) Centripetal acceleration: ፚ × (ፚ×ア),

If the sensitive axes of the sensors do not coincide with the axes of the navigational coordinate system, the sensor outputs must be transformed with the orientation matrix  $(\underline{N})$ .

Figure 3 shows a general block diagram of navigation systems, including the corrections just mentioned for a rotating frame. Redundant data may enter the system at levels 2, 3a and 3b. These input data cohsist of the correct values and errors. The errors can be bias yalues or stochastic processes or a combination of both. Consequently the mavigation information is erroneous too.

The following two problems must be solved in order to use the available redundant information in the best possible way:

- (a) Determine the desired output information with minimum errors (minimum variation). This is a filtering or estimation problem.
- (b) Control the system error states. This task is a system control problem.

Both tasks are accomplished with a minimum variance or Kalman filter. Task (a) requires valid mathematical models of the different error sources. Task (b) is accomplished by feedback loops (inside the computer) as indicated in Figure 2 by dotted lines.

Before these two tasks are discussed in more detail in Section 6, the error behaviour of pure inertial aystems and the conventional methods of updating are considered.

## 3. DETERMINISTIC ERROR ANALYSIS AND ERROR BEHAVIOUR OF FURE INERTIAL SYSTEMS

In a pure inertial system thrust acceleration represents the only input formation measured at the first level. In most cases the input axes of the sensors are parallel to the axes of the navigational coordinate frame. If this is not the case, a coordinate transformation is necessary.

Figure 4 shows the block diagram of a pure inertial mystem used for navigation over the surface of the carth. In this case the velocity  $\vec{v}$  of a vehicle is defined as the rate of change of the vector  $\vec{r}$  with respect to the earth ( $v = (d\vec{r}/dt)_g$ ). Therefore, instead of Equation (2.4), we now have the following relation for acceleration corrections:

$$\vec{\mathbf{b}} = \left(\frac{d\vec{\mathbf{v}}}{dt}\right)_{\mathbf{H}} = \vec{\mathbf{a}} + \vec{\mathbf{c}}_{\mathbf{g}} = \vec{\mathbf{a}}_{\mathbf{g}} \times (\vec{\mathbf{a}}_{\mathbf{g}} \times \vec{\mathbf{f}}) - (2\vec{\mathbf{a}} \times \vec{\boldsymbol{\rho}}) \times \vec{\mathbf{v}}; \qquad (3.1)$$

where

₩ earth rate vector

 $\vec{p} = \vec{a} - \vec{a}_{\mu}$  = rate vector of the navigational coordinate frame with respect to an earth-fixed frame.

472

(2.4)





As shown in Figure 4, a feedback loop exists owing to the gravity vector  $\vec{B}$ , i.e. the system is capable of oscillating. The oscillation occurs with the Schuler period

$$T_{\rm B} = 2\pi \sqrt{\left(\frac{r_{\rm A}}{s_{\rm O}}\right)},$$

The error behaviour of a pure inertial system in a Schuler-tuned mode will now be discussed 1.

Figure 5 shows the error block diagram for one horizontal channel of such a system. The errors considered are accelerometer bias, gyro drift, and azimuth misalignment.

The following notation is used in Figure 5:

マンドンド

- $\Delta \mathbf{r}_{\mathbf{x}}$  = computed velocity error in x-direction
- $\Delta \mathbf{r}_{\mathbf{x}} = \text{computed position error in x-direction}$
- $\nabla_{\mathbf{x}} = \mathbf{x}$ -accelerometer bias
- e\_ = fixed drift of the y-gyro -
- $\phi_y$  = y-component of the error angle vector between the navigational and platform coordinate frame.
- $\psi_{\rm g}$  = z-component of the error angle vector between the platform and computer coordinate frame (azimuth misalignment).

The velocity and position errors, as derived from Figure 5, are

$$\Delta \mathbf{r}_{\mathbf{x}} = \frac{\mathbf{s}\nabla_{\mathbf{x}} - \mathbf{s}_{\mathbf{y}}(\mathbf{s}_{\mathbf{y}} + \omega_{\mathbf{x}}\psi_{\mathbf{z}})}{\mathbf{s}^{3} + \omega_{\mathbf{z}}^{2}}$$
(3.2)

$$\Delta \mathbf{r}_{\mathbf{x}} = \frac{\mathbf{s} \nabla_{\mathbf{x}} - \mathbf{g}_0(\mathbf{e}_{\mathbf{y}} + \omega_{\mathbf{x}} \psi_{\mathbf{y}})}{\mathbf{s}(\mathbf{s}^4 + \omega_{\mathbf{x}}^2)}, \qquad (3,3)$$

The sharacteristic equation (denominator of Equation (3.2)) is that of an undamped oscillation with Schuler frequency. For constant sensor errors the steady-state value of the velocity error is

$$\Delta \mathbf{r}_{\mathbf{x}}^{*}|_{\mathbf{x}=0} = -\frac{\mathbf{s}_{0}}{\omega_{\mathbf{s}}^{2}} \left( \boldsymbol{\epsilon}_{\mathbf{y}} + \omega_{\mathbf{x}} \boldsymbol{\psi}_{\mathbf{s}} \right) , \qquad (3.4)$$

The position error is constantly increasing with time,

$$\Delta r_{\mathbf{x}} = 0 \quad (3, 5)$$

and it is unbounded using to gyro drift and aziruth misalignment; accelerometer bias does not cause an unbounded position error.

## 4. DETERMINISTIC UPDATING METHODS

Although Schuler tuning is of great importance in navigation over the surface of the earth, it has been shown that for longer mission times the position errors can become intolerably large, especially because of gyro drift and azimuth error. To avoid this, the pure inertial system can be aided with additional or redundant navigation information from other sources. Available continuous or discrete data for the navigation computer may be of levels 2, 3n and 3b. No additional information on level 1 is available, because we do not know any other method of measuring socieleration except by inertial means.

Accordingly we distinguish different kinds of sided systems:

- (i) Velocity aided systems (redundant data at level 2).
- (11) Position mided systems (redundant data at level 3m).
- (iii) Stellar monitored systems (redundant data at level 3b).
- (iv) Stellar monitored and velocity damped systems (redundant data at lavels 2 and 3b).

The additional or redundant data can be intermittent and used to replace the computed data at certain time periods. If it is always available, it can be continuously compared with the computed information; the difference signal, multiplied with various gains  $K_{\mu}$ , can be added to the excelerometer output signals or the gyro torquing commands either directly or over additional integrators. If the difference or error signal is introduced at the next lower level it is used to damp the oscillations of the system. If it is introduced at the same level inside the Schuler loop, then the frequency and amplitude can be changed. If additional integrators are used, sensor biases can be compensated.



Figure 6 shows the x-channel of a Schuler-tuned system and the mechanisation of the different updating methods,

These methods are now discussed in more detail, where Doppler velocity is available as redundant information. The Doppler velocity signal cannot very well be used to react the inertial velocity signal directly because of its high noise level. The direct resotting method is not very useful and is only applicable if angular measurements or position fixes are available.

As Figure 6 indicates, the difference signal can be used to provide three different correction signals,  $c_1$ ,  $c_2$ , and  $c_3$ , into the Schuler loop.

The first signal  $C_1$ , the difference signal multiplied with the gain factor  $K_1$ , is added to the acceleroseter output. This provides damping of the Bohuler oscillation. The second signal  $C_2$ , the difference signal multiplied with the gain factor  $K_2/r_0$ , affects the oscillation frequency by adding to it the gyre input. The third input  $C_2$  provides compensation for some steady-state errors of velocity by increducing an additional integrator with gain  $K_1/r_0$ .

Assuming that both the computed inertial and the measured reference volocity are subjected to errors, the three corrections corresponding to the x-channel error block diagram can be written as follows:

$$O_{1x} = -K_1(v_{0x} - v_{rx}) = -K_1(\Delta r_x - \delta v_A)$$
(4.1)

$$O_{BE} = \frac{K_g}{r_0} \left( v_{0E} - v_{FE} \right) = \frac{K_g}{r_0} \left( \Delta \dot{r}_{E} - 5 v_{E} \right)$$
(4.2)

$$\sigma_{3x} = \frac{\kappa_3}{r_0} \int (v_{0x} - v_{1x}) dt = \frac{\kappa_3}{r_0} \int (\Delta r_x - \delta v_x) dt , \qquad (4.3)$$

where  $\Delta r_{x}$  and  $\delta v_{x}$  represent the errors of the equputed inertial and the measured Doppler reference velocity, respectively.

Figure 7 shows the x-channel error block diagram of the Doppler inertial system described.

The transfer functions of the velocity and position errors can be derived from Figure 7 as follows:

$$\Delta \mathbf{r}_{x} = \frac{\mathbf{s}^{2}\nabla_{x} - \mathbf{s}\mathbf{g}_{0}(\mathbf{c}_{y} + \omega_{x}\psi_{y}) + [\omega_{0}^{2}(\mathbf{K}_{y}\mathbf{s} + \mathbf{K}_{y}) + \mathbf{s}^{4}\mathbf{K}_{1}]\mathbf{\delta}\mathbf{v}_{x}}{\mathbf{s}^{3} + \mathbf{K}_{1}\mathbf{s}^{2} + (\mathbf{1} + \mathbf{K}_{1})\omega_{0}^{4}\mathbf{s} + \mathbf{K}_{1}\omega_{1}^{2}}$$
(4.4)

$$\Delta Y_{\rm g} = \frac{{\rm g}^2 \nabla_{\rm g} - {\rm ag}_0 \left( {\rm e}_{\rm g} + \omega_{\rm g} \psi_{\rm g} \right) + \left[ \omega_{\rm g}^2 \left( {\rm K}_{\rm g} {\rm e} + {\rm K}_{\rm g} \right) + {\rm a}^2 {\rm K}_{\rm g} \right] \delta v_{\rm g}}{{\rm a} \left[ {\rm a}^2 + {\rm K}_{\rm g} {\rm a}^2 + \left( {\rm i} + {\rm K}_{\rm g} \right) \omega_{\rm g}^2 + {\rm K}_{\rm g} {\rm a}^2 \right]}$$
(4.5)

$$\Delta \mathbf{r}_{\mathbf{x}}|_{\mathbf{s}=\mathbf{0}} = \delta \mathbf{v}_{\mathbf{x}}$$
 (4.5)

Equations (4.4) and (4.5) can be used to evaluate the velocity and position errors for different values of  $K_1$ ,  $K_3$  and  $K_3$ , or to obtain optimal values of these three gain factors. The steady-state velocity error now contains the reference velocity error  $\delta v_{\chi}$  only (Equation (4.8)). The fixed gyro drift and asimuth error are compensated by the additional integrator.

Figure 3 shows, as an example, the time functions of the velocity error due to gyro drift for the pure inertial system  $(K_1 = K_g = K_g = 0)$  and for the different methods of updating  $(K_1 \neq 0; K_1 \text{ and } K_g \neq 0 \text{ and } K_1, K_g, K_g \neq 0)$ . In this example only the influence of the various gain factors  $K_{i}$  on the error behaviour  $\Delta r_g$  are of interest, and not the values of these factors themselves. This means that the chosen values of the gain fautors cannot be considered as optimal values. Furthermore, an exact definition of "optimal" has not been given so far, because only one error source is taken into account. The problem of optimal gains is discussed in Section 6.

## 5. STATISTICAL ERROR ANALYSIS AND ERROR BEHAVIOUR

So far constant errors (binnes) have been discussed, but in many cases the errors are not constant, but rather stochastic processes characterised by their autocorrelation functions  $\rho_{\mu}(\tau)$  or their spectral density functions  $\phi_{\mu}(\omega)$ .

For linear, non-time-varying systems the statistics of the output signal y(t) follows directly from the time-independent statistics of the input signal x(t). The mean value of the output signal is zero if the mean value of the input signal is zero. The average value of the square of the output signal, the time-dependent variance  $\sigma_{y(t)}^2$ , can be computed by the expression<sup>2</sup>

$$\sigma_{y(t)}^{d} = \int_{0}^{t} \int_{0}^{t} g(\tau_{1}) g(\tau_{2}) \rho_{x}(|\tau_{1} - \tau_{2}|) d\tau_{1} d\tau_{2}$$

(5.1)

Anten of Frich

· 475

(5.2)

where

$$g(\tau) = \frac{1}{2\pi j} \int_{\alpha - j\infty}^{\alpha + j\infty} G(s) e^{s\tau} ds$$
, system impulse response

Q(s) = system transfer function

$$\rho_{\mathbf{x}}(|\tau|)$$
 = input signal autocorrelation function.

If the input signal is white noise, the autocorrelation function is given by

$$\rho_{\mathbf{x}}(|\tau_1 - \tau_2|) = A\delta(|\tau_1 - \tau_2|); \quad \mathbf{A} = \text{constant},$$

Equation (5.2) substituted into Equation (5.1) yields

$$\begin{aligned} \tau_{\mathbf{y}(t)}^{2} &= \int_{0}^{t} \int_{0}^{t} \mathbf{g}(\tau_{1}) \mathbf{g}(\tau_{2}) \mathbf{A} \delta(|\tau_{1} - \tau_{2}|) d\tau_{1} d\tau_{2} \\ &= \mathbf{A} \int_{0}^{t} \mathbf{g}(\tau_{1}) d\tau_{1} \int_{0}^{t} \mathbf{g}(\tau_{2}) \delta(|\tau_{1} - \tau_{2}|) d\tau_{2} . \end{aligned}$$

The inter integral can be evaluated by using the distribution property of the  $\delta$ -function

$$\int_{0}^{t} \mathbf{g}(\tau_{2}) \delta(|\tau_{1} - \tau_{2}| \cdot d\tau_{2}) = \begin{cases} \mathbf{g}(\tau_{1}) & \text{for } \tau_{1} & \text{in } [0, t] \\ 0 & \text{for } \tau_{1} & \text{not in } [0, t] \end{cases} ,$$
 (5.3)

With Equation (5.3) the expression for  $\sigma^2_{y(t)}$  becomes

$$\sigma_{y(t)}^{2} = \Lambda \int_{0}^{t} \mathbf{g}^{2}(\tau_{1}) d\tau_{1} , \qquad (5.4)$$

If the input signal is bendwidth limited white noise with autocorrelation function

$$\rho_{\mathbf{x}}(|\tau|) = \sigma_{\mathbf{x}}^2 \bullet^{\frac{|\tau|}{\tau_{\mathbf{x}}}}, \qquad (8.5)$$

Equation (5.4) can also be used if  $\tau_{\rm g}$  is small compared with the system time constants. Equation (5.5) can then be approximated by

$$\rho_{\mathbf{x}}(|\tau|) \simeq 2\tau_{\mathbf{x}} \sigma_{\mathbf{x}}^2 \delta(\tau) \tag{5.6}$$

and the cutput variance yields, in this crase.

$$\sigma_{y(t)}^{2} = 2r_{x}\sigma_{x}^{2}\int_{0}^{t} g^{2}(\tau) d\tau . \qquad (5.7)$$

Another method of determining the steady-state value of the variance  $\sigma_{y(t)}^2$ , if it is convergent, is given by<sup>2</sup>

$$\lim_{t \to +\infty} t \sigma_y^2(t) = \frac{1}{2\pi} \int_{-\infty}^{+\infty} \phi_y(\omega) d\omega . \qquad (5.8)$$

The output spectral density function is given by

$$\phi_{\mathbf{y}}(\omega) = |\mathbf{Q}(\mathbf{j}\omega)|^2 \phi_{\mathbf{x}}(\omega) , \qquad (5.9)$$

For bandwidth limited white noise,  $\phi_{\mathbf{x}}$  can be evaluated from the approximated autocorrelation function  $\rho_{\mathbf{x}}(|\tau|)$  of Equation (5.6).

$$\phi_{\mathbf{X}}(\omega) = \int_{-\infty}^{+\infty} \rho_{\mathbf{X}}(|\tau|) e^{-j\omega\tau} d\tau = 2\tau_{\mathbf{X}}\sigma_{\mathbf{X}}^2 . \qquad (5.10)$$

Equations (5.9) and (5.10) substituted into Equation (5.8) yield

$$\lim_{t\to+\infty} \frac{1}{\sigma_{\mathbf{y}}(t)} = (2\tau_{\mathbf{x}}\sigma_{\mathbf{x}}^2) \frac{1}{2\tau} \int_{-\infty}^{+\infty} |G(j\omega)|^2 d\omega .$$
 (5.11)

It should be noted that Equations (5.8) and (5.11) can only be used if G(s) has no poles on the imaginary axis or in the right half-plane. A pure inertial system in a Schuler-tuned mode is undamped, i.e. it has poles on the imaginary axis. This means that the effect of noise inputs must be calculated according to Equations (5.1) or (5.7).

As an example the variance of the velocity and position errors due to a white noise Doppler input error  $\delta v_{\chi}$  for a damped system are now calculated.



州市市市

in the second

Using Equations (4.4) and (4.5) the transfer functions for  $K_{g} = K_{g} = 0$  are

$$G_{\Delta_{\Gamma_{R}}^{+}}(s) = \frac{K_{1}s}{s^{2} + K_{1}s + \omega_{R}^{2}}$$
 (5.12)

$$G_{\Delta r_{\chi}}(s) = \frac{K_{1}}{s^{2} + K_{1}s + \omega_{s}^{2}}$$
 (5.13)

At first Equation (5.11) is used:

$$\sigma_{\Delta \mathbf{r}_{\mathbf{x}}}^{2} = 2\tau_{\mathbf{s}_{\mathbf{y}_{\mathbf{x}}}} \sigma_{\mathbf{s}_{\mathbf{y}_{\mathbf{x}}}}^{2} \frac{1}{2\pi} \int_{-\infty}^{+\infty} |\mathbf{0}_{\Delta \mathbf{r}_{\mathbf{x}}}(j\omega)|^{2} d\omega$$
  
$$= 2\tau_{\mathbf{s}_{\mathbf{y}_{\mathbf{x}}}} \sigma_{\mathbf{s}_{\mathbf{y}_{\mathbf{x}}}}^{2} \frac{\mathbf{K}_{1}^{2}}{\pi} \cdot \int_{0}^{\infty} \frac{\omega^{2}}{(\omega_{\mathbf{s}}^{2} - \omega^{2})^{2} + \mathbf{K}_{1}^{2} \omega^{2}} d\omega$$
  
$$= 2\tau_{\mathbf{s}_{\mathbf{y}_{\mathbf{x}}}} \sigma_{\mathbf{s}_{\mathbf{y}_{\mathbf{x}}}}^{2} \frac{\mathbf{K}_{1} \omega_{\mathbf{s}}}{(\omega_{\mathbf{s}}^{2} - \omega^{2})^{2} + \mathbf{K}_{1}^{2} \omega^{2}} d\omega$$

$$= \tau_{\mathbf{S}_{\mathbf{Y}_{\mathbf{X}}}} \sigma_{\mathbf{S}_{\mathbf{Y}_{\mathbf{X}}}} \mathbf{K}_{1}$$
 (5.14)

$$\sigma_{\Delta r_{\chi}}^{a} = 2\tau_{\delta v_{\chi}} \sigma_{\delta v_{\chi}}^{a} \frac{K_{1}^{a}}{\pi} \int_{0}^{\infty} \frac{d\omega}{(\omega_{\mu}^{2} - \omega^{2})^{2} + K_{1}^{2} \omega^{2}}$$

$$=\frac{\tau_{3\sqrt{2}}\sigma_{3\sqrt{2}}^{2}K_{1}}{\omega_{s}^{2}},\qquad(5.15)$$

Now Equation (5.7) is used to determine the time-dependent variances.

The system impulse responses become

$$\mathbf{z}_{\Delta \mathbf{r}_{\mathbf{x}}}(t) = \mathbf{K}_{1} \mathbf{e}^{-\frac{1}{2}\mathbf{K}_{1}t} \left( \cos \omega_{0} t - \frac{\mathbf{K}_{1}}{2\omega_{0}} \sin \omega_{0} t \right) = \dot{\mathbf{z}}_{\Delta \mathbf{r}_{\mathbf{x}}}(t)$$
 (5.16)

$$\mathbf{E}_{\Delta \mathbf{r}_{\mathbf{x}}}(\mathbf{t}) = \frac{\mathbf{K}_{\mathbf{1}}}{\omega_{0}} \mathbf{e}^{-\frac{1}{2}\mathbf{K}_{\mathbf{1}}\mathbf{t}} \sin \omega_{0} \mathbf{t}, \qquad (5.17)$$

where

$$= \frac{1}{2} \sqrt{4\omega_{\rm B}^2 - K_1^2} \,, \qquad (5.18)$$

 $\omega_0 ~=~ \frac{1}{4} v' (4 \omega_g^2 - K_1^2) ~.$  The desired time-dependent variances are obtained as follows:

$$\sigma_{\Delta \mathbf{r}_{\mathbf{x}}}^{2}(\mathbf{t}) = 2\tau_{\mathbf{i}\mathbf{v}_{\mathbf{x}}}\sigma_{\mathbf{\delta}\mathbf{v}_{\mathbf{x}}}^{2} \int_{0}^{\mathbf{t}} \mathbf{g}_{\Delta \mathbf{r}_{\mathbf{x}}}^{2}(\tau) d\tau$$

$$= \frac{2\tau_{\mathbf{i}\mathbf{v}_{\mathbf{x}}}\sigma_{\mathbf{\delta}\mathbf{v}_{\mathbf{x}}}^{2}\mathbf{K}_{1}^{2}}{\omega_{0}(4\omega_{0}^{2} + \mathbf{K}_{1}^{2})} \left[\frac{\omega_{0}(4\omega_{0}^{2} + \mathbf{K}_{1}^{2})}{2\mathbf{K}_{1}} - \mathbf{e}^{-\mathbf{K}_{\mathbf{i}}\mathbf{t}} \left(\frac{8\omega_{0}^{2}\mathbf{K}_{1}^{2} + 16\omega_{0}^{4} + \mathbf{K}_{1}^{4}}{8\omega_{0}\mathbf{K}_{1}} - \frac{4\omega_{0}^{2}\mathbf{e}^{2}\mathbf{K}_{1}^{2}}{\mathbf{k}_{1}} \sin 2\omega_{0}\mathbf{t} - \frac{\mathbf{K}_{\mathbf{i}}\mathbf{t}}{8\omega_{0}\mathbf{K}_{1}}\cos 2\omega_{0}\mathbf{t}}{8\omega_{0}\mathbf{K}_{1}}\right]$$

$$(5.19)$$

$$\sigma_{\Delta \mathbf{F}_{\mathbf{X}}}^{3}(t) = 2\tau_{\mathbf{S}_{\mathbf{Y}_{\mathbf{X}}}}\sigma_{\mathbf{S}_{\mathbf{Y}_{\mathbf{X}}}}^{3}\int_{0}^{t} \mathbf{g}_{\Delta \mathbf{F}_{\mathbf{X}}}^{2}(\tau) d\tau$$
$$= \frac{2\tau_{\mathbf{S}_{\mathbf{Y}_{\mathbf{X}}}}\sigma_{\mathbf{S}_{\mathbf{X}}}^{3}\mathbf{k}_{\mathbf{X}}^{4}}{\omega_{0}(4\omega_{0}^{2} + \mathbf{K}_{1}^{2})} \left[\frac{2\omega_{0}}{\mathbf{K}_{1}} - e^{-\mathbf{K}_{1}t} \left(\frac{4\omega_{0}^{2} + \mathbf{K}_{1}^{2}}{2\omega_{0}\mathbf{K}_{1}} + \sin 2\omega_{0}t - \frac{\mathbf{K}_{1}}{2\omega_{0}}\cos 2\omega_{0}t\right)\right].$$
(5.20)

If  $K_i$  is small compared with  $\omega_g$ , the following approximations for  $\sigma^q_{\Delta r_g}(t)$  and  $\sigma^q_{\Delta r_g}(t)$  result:

٠t

$$\sigma_{\Delta \mathbf{r}_{\mathbf{x}}}^{2}(\mathbf{t}) \simeq \tau_{\delta \mathbf{v}_{\mathbf{x}}} \sigma_{\delta \mathbf{v}_{\mathbf{x}}} \mathbf{K}_{1} (1 - \mathbf{e}^{-\mathbf{K}_{1} \mathbf{t}})$$
(5.21)

$$\sigma_{\Delta \mathbf{r}_{\mathbf{x}}}^{2}(\mathbf{t}) \simeq \frac{\tau_{\mathbf{i}\mathbf{y}_{\mathbf{x}}}\sigma_{\mathbf{j}\mathbf{y}_{\mathbf{x}}}^{2}\mathbf{k}_{\mathbf{i}}}{\omega_{\mathbf{a}}^{2}} (1 - e^{-\mathbf{K}_{\mathbf{i}}\mathbf{t}}) , \qquad (5.22)$$

Equations (5.21) and (5.22) show that the "filtering" of the Doppler noise through the inertial system depends largely on the gain factor  $K_i$ . Also, the time constant for the variance of the velocity and the position



error are inversely proportional to  $K_1$ . These facts give additional oriteria for  $K_1$ . If  $K_2$  and  $K_3$  are also  $\neq 0$ , it is difficult to obtain a closed-form solution of Equation (5.7). In this case a computer program may be necessary to evaluate the integral and to obtain criteria for the values of  $K_2$  and  $K_3$ .

## 6. STATISTICAL FILTERING AND UPDATING

6d

535757755 57575755

The updating mechanism described in Section 4 used a deterministic approach. The advent of very powerful navigation computers made it possible to retrieve more of the information contained in the navigation measurements by applying statistical filtering techniques.

In this section the problem of optimal estimation and control of aided inertial system state variables is discussed. Although the problem of optimal estimation and filtering is actually a fairly old one, it has only recently yielded a decisive breakthrough by Kalman and others. The classical theory of Wiener and Kolmogoroff has, as an essential disadvantage, operation with the difficult formalism of integral equations. The Kalman discrete filter equations are purely algebraic; hence they are particularly well auited for processing on modern digital computers. Common to both theories is the fact that they are applicable only to linear systems and that they make use of the same criterion of being optimal, namely the smallest mean square error. Moreover, there exists an interesting connection between the two theories: if the Wiener-Hopf equation is extended to non-stationary processes and presented in matrix form, it is an exact analogy to the orthogonality theorem of the optimal Kalman filter equations<sup>1</sup>. This orthogonality theorem states that the mean square estimation error becomes a sininum if, and only if, the measurement error and the estimation error of the system state are orthogonal to each other.

Inertial navigation systems can be expressed mathematically by nonlinear differential equations. But the navigation errors are well describable by linear differential equations using only the first terms in the series expansion shout the nominal values. The error sources, for example gyro drift, accelerometer bias etc., appear as forcing functions. If these error sources, as well as the error sources of the reference measurements, are considered as statistical variables with zero means and known Jecond-order moments, then the modern Kalman theory of optimal filtering applies very well to aided inertial systems. In this way optimum use is made of the external measurements as additional information and of the known statistics of the error sources, to obtain the optimal system gain factors for every time instant and to keep the navigation errors as small as possible. It is important to recognise that, contrary to the conventional updating methods described in Section 4, the gain factors are not constant. They are recomputed in every Kalman cycle in order to use the redundant information in the optimal way within the system.

The Kalman filtering theory and the derivation of the equations is fully discussed elsewhere (References 8, -13, 4, 14, etc.) and can therefore be deleted.

The most important properties of a Kalman filter can be summarized as follows:

The filter estimates are all variables of the state vector in the least mean square error sense. The estimation is based upon statistical data of all error sources and is completely carried

out in the time domain.

The filter formulae satisfy minisum variance criteria for all problem parameters.

The formulae implemented are recursive. This means that the optimum estimate for the time being can be computed from the previous estimate and the current observation without recourse to earlier estimates or observations.

The recursive formulae are well suited to digital computers.

The basic method of using a linear optimal filter within an aided inertial system is shown in Figure 9.

The two upper blocks in Figure 9 represent a pure inertial system. The navigation equations are mechanised in the navigation computer in the conventional way. Compensation for sensor biases may be included in the navigation equations.

The information available in the mavigation computer is compared with redundant external data. The differences are the measurement data used in the filter to improve the estimate of the system states.

These error estimates of the filter may be used either for updating the navigation information, and thus for correction of IMU errors (closed loop operation), or only for improvement of the error states (open loop operation)<sup>2</sup>. The block "controller" regulates this operation.

If an inertial mavigation system is mechanised in the open loop operation, all errors and their external measurements can be represented by a set of two linear vector equations

x(t)	=	$\underline{A(t)}\overline{x(t)} + \overline{n(t)}$	(6.1)
y(t)	=	$\underline{W(t)x(t)} + \overline{q(t)}$ .	(6.2)















Equation (6.1) is the system equation. It describes the propagation of all errors occurring in an aided navigation system. The elements of the  $n \times n$  matrix  $\underline{A(t)}$  are known functions of time. The components of the white noise vector n(t) represent the forcing functions. Equation (6.2) is the measurement equation. It relates the values of the error state vector to the measurement data. The vector q(t) represents the disturbing white noise of the external measurements.

It is an essential supposition of the Kalman filter theory that the vector differential equation (6.1) can be solved by the same method used for deterministic differential equations, namely the conventional Picard iteration<sup>15</sup>. The so-called transition matrix  $\phi_{t/t_0}$ , which plays a fundamental role in the whole theory, appears in this solution. Using this fact and dividing the time axis into equally spaced discrete intervals (the sampling times of the computer) the dynamics of the linear system under consideration can be described in the form

$$\overline{\mathbf{x}_{\mathbf{K}+1}} = \frac{\phi_{\mathbf{K}+1}/\mathbf{k}}{\mathbf{x}_{\mathbf{K}}} + \overline{\mathbf{h}_{\mathbf{K}}}$$
(0.3)  
$$\overline{\mathbf{y}_{\mathbf{K}}} = \underline{\mathbf{h}_{\mathbf{K}}} \mathbf{x}_{\mathbf{K}} + \overline{\mathbf{q}_{\mathbf{K}}} .$$
(0.4)

where

## $\vec{x}_{k}$ = system error state vector at time $t_{k}$

 $\frac{\phi_{K+1/K}}{\text{in the noise-free case}} = n \times n \text{ state transition matrix, relating the state vectors } \vec{x_{K+1}} \text{ and } \vec{x_K}$ 

 $\vec{h_K}$  = Additive white noise sequence representing the forcing function in the dynamic system

(Note: If Equation (6.3) were presented in the continuous form, then the vector

$$\overline{\mathbf{n}(\mathbf{t})} = \int_{\mathbf{t}_0}^{\mathbf{t}} \underline{\phi(\mathbf{t},\tau)} \, \overline{\mathbf{n}(\tau)} \, \mathrm{d}\tau \, ,$$

as the response of the system to the white noise input n(t), would be a correlated random process. But in the discrete form the values of the integrals

$$\overline{\mathbf{h}_{\mathbf{K}}} = \int_{\mathbf{t}_{\mathbf{K}}}^{\mathbf{t}_{\mathbf{K}+1}} \frac{\phi(\mathbf{t}_{\mathbf{K}+1},\tau)}{\mathbf{n}(\tau)} \, \mathbf{n}(\tau) \, \mathrm{d}\tau$$

for different K are independent of each other; therefore the vector  $\overline{h_g}$  can be represented as a white noise sequence with the property

$$\left[\frac{\mathbf{h}_{\mathbf{K}}\mathbf{h}_{\mathbf{L}}^{\mathrm{T}}\right] = H_{\mathbf{K}}\delta_{\mathbf{KL}}.$$

 $\vec{x_{g}}$  = observation vector of measurement data at time  $t_{g}$  linearly related to  $\vec{x_{g}}$ 

 $M_{K} = m \times n$  measurement matrix at time  $t_{K}$  relating  $\vec{y}_{K}$  to  $\vec{x}_{K}$  in the noise-free case

 $q_{\rm g}$  = additive white noise sequence corrupting measurement data at time t<sub>g</sub> .

The following five basic equations represent the Kalman filter:

$$\widehat{\vec{x}_{K}} = \overline{\vec{x}_{K}} + \underline{B_{K}}(\overline{\vec{y}_{K}} - \overline{\vec{y}_{K}})$$
(6.5)

$$\vec{\mathbf{x}}_{\mathbf{k}+1}^{T} = \phi_{\mathbf{k}+1/\mathbf{k}} \vec{\mathbf{x}}_{\mathbf{k}}^{T}$$
(6.6)

$$\mathbf{B}_{\mathbf{K}} = \mathbf{P}_{\mathbf{K}}^{\star} \mathbf{M}_{\mathbf{K}}^{T} (\mathbf{H}_{\mathbf{K}} \mathbf{P}_{\mathbf{K}}^{\star} \mathbf{M}_{\mathbf{K}}^{T} + \mathbf{Q}_{\mathbf{K}})^{-1}$$
(6.7)

$$\frac{P_{x}}{P_{x}} = (\frac{1}{B_{x}} \frac{B_{x}}{B_{x}}) \frac{P_{x}^{+}}{B_{x}}$$
(6.8)

$$\mathbf{P}_{\mathbf{K}+\mathbf{j}}^{\dagger} \simeq \phi_{\mathbf{K}+\mathbf{j}/\mathbf{K}} \mathbf{P}_{\mathbf{K}} \phi_{\mathbf{K}+\mathbf{j}/\mathbf{K}}^{\mathsf{T}} + \mathbf{H}_{\mathbf{K}}$$
(6.9)

where

 $\vec{y_{k}} = \underline{M_{k}}\vec{x_{k}}$  $\vec{x_{k}} = a \text{ priori}$  estimate of the system state vector predicted at time  $t_{K}$ before using redundant measurement data

 $\vec{x}_{R}$  = a posteriori estimate of the system state vector at time  $t_{R}$  after using redundant measured data

 $\mathbf{P}_{\mathbf{K}} = \mathbf{n} \times \mathbf{n}$  covariance matrix of the *a posteriori* estimation error  $\mathbf{x}_{\mathbf{K}} = \hat{\mathbf{x}}_{\mathbf{K}}$ 

 $\mathbf{F}_{\mathbf{x}}^{\mathbf{r}}$  =  $\mathbf{x} \times \mathbf{n}$  covariance matrix of the *a priori* estimation error  $\mathbf{x}_{\mathbf{x}}^{\mathbf{r}} - \mathbf{x}_{\mathbf{x}}^{\mathbf{r}}$ 

 $H_g = E[h_g h_g^{\to \to T}] = n \times n$  covariance matrix of the system white noise vector

 $Q_{R} = \mathbb{E}[\overrightarrow{q_{R}q_{R}}] = n \times n$  covariance matrix of the measurement white noise vector

 $B_{g}$  = weighting (gain) matrix at time  $t_{g}$  .

These equations are processed on an iterative basis to determine consecutively the optimum estimate  $\overline{x}_{k}^{2}$  of the system state at time  $t_{k}$ . The mechanisation of an aided inertial navigation system in an open loop operation as just described is not satisfactory, especially for long missions. The outputs of the inertial navigation system could be corrected by subtracting the estimated errors; but, since some errors in the inertial system are not bounded, the linear error codel of the navigator would become invalid. The requirement of small errors and linear error propagation can only be schieved by controlling the state vector in the closed loop operation. For such a controlled system some of the above equations must be modified slightly. Instead of Equations (6.1) and (6.3) the following equations are valid<sup>2</sup>:

「いっていていた」

$$\bar{x}(t) = A(t)\bar{x}(t) + p(t)\bar{u}(t) + n(t)$$
 (6.10)

$$\overline{\mathbf{x}_{K+1}} \approx \frac{\phi_{K+1/K}}{\kappa_{K}} \mathbf{x}_{K} + \frac{\Gamma_{K+1/K}}{\kappa_{K}} \mathbf{u}_{K} + \mathbf{h}_{K} \,. \tag{6.11}$$

The terms  $\underline{D(t)}u(t)$  and  $\underline{\Gamma}_{K+1/K}u_{K}$  represent the above mentioned correction signals entering the system to be controlled; in the ideal case of a completely controllable system the matrices  $\underline{D(t)}$  and  $\underline{\Gamma}_{K+1/K}$  are of dimension  $n \times n$ .

The modification of the filter Equations (6.5) to (6.9) results in a slight change of Equation (6.6). This result is noticeable, since the original equations given for the uncontrolled system are derived in the literature under the important assumption that the mean values of the input forcing functions are zero, i.e. that always the condition is valid  $\mathbb{E}[\overline{n(t)}] = 0$  and  $\mathbb{E}[\overline{h_R}] = 0$ , respectively. In the controlled system the condition of zero mean value inputs is not achieved, because now the forcing functions consist not only of the white noise vectors  $\overline{n(t)}$  or  $\overline{h_R}$ , but also of control signals which are treated as known deterministic functions, based upon the computed estimate of the state vector. In most cases the condition of  $\mathbb{E}[\underline{D(t)}u(t)] = 0$  or  $\mathbb{E}[\frac{1}{K_{+1}/K}\overline{u_R}] = 0$  is not valid for the control signals, because the estimator  $\overline{x_R}$  is unbiased and the initial value  $\mathbb{E}[x_0]$  does not need to be zero (Section D) of Reference 13). Nevertheless it is obvious (and can be shown by minimising the mean square estimation error (Section C of Reference 13), that the modified Equation (6.6) now becomes

$$\mathbf{x}_{\mathbf{k}+1}^{\mathbf{x}} \xrightarrow{\phi_{\mathbf{k}+1/\mathbf{k}}} \mathbf{x}_{\mathbf{k}}^{\mathbf{x}} \xrightarrow{\Gamma_{\mathbf{k}+1/\mathbf{k}}} \mathbf{u}_{\mathbf{k}}^{\mathbf{x}} . \tag{6.12}$$

Equations (6.10) to (6.12) would be valid if all components of the state vector were controllable. Unfortunately the ideal case of complete controllability is not possible in an inertial navigation system. The controllable states are in most cases restricted to errors in position, velocity, and platform orientation. The effects of the other variables can also be removed by including compensation signals in the control signals, based upon the Kalman filter estimates, but nevertheless these variables are not controllable in a strict sense, because they are not accessible to direct measurements and corrections.

In an inertial navigation system the state variables are usually separated into two parts, the first one containing the controllable states, the other one the uncontrollable states. Thus the dynamics of such a system can be described in the following form:

$$\begin{bmatrix} \overline{\mathbf{x}_{k+1}^{\dagger}} \\ \overline{\mathbf{x}_{k+1}^{\dagger}} \end{bmatrix} = \begin{bmatrix} \frac{\phi_{k+1/k}^{\dagger}}{\mathbf{x}_{k+1/k}} & \frac{\phi_{k+1/k}^{\dagger}}{\mathbf{x}_{k}^{\dagger}} \\ \mathbf{\Omega} & \frac{\phi_{k+1/k}^{\dagger}}{\mathbf{x}_{k}^{\dagger}} \end{bmatrix} \cdot \begin{bmatrix} \overline{\mathbf{x}_{k}^{\dagger}} \\ \overline{\mathbf{x}_{k}^{\dagger}} \end{bmatrix} + \begin{bmatrix} \overline{\mathbf{x}_{k+1/k}^{\dagger}} \\ \mathbf{\Omega} \end{bmatrix} \overline{\mathbf{u}_{k}^{\dagger}} + \begin{bmatrix} \overline{\mathbf{h}_{k}^{\dagger}} \\ \overline{\mathbf{h}_{k}^{\dagger}} \\ \overline{\mathbf{h}_{k}^{\dagger}} \end{bmatrix} .$$
(6.13)

The control signals for the time interval from  $t_{K}$  to  $t_{K+1}$  should be based upon the best estimate of the state vector  $\vec{x_{K}}$  available at time  $t_{K}$ . But, since at this point in time a new measurement is made and the computation of the weighting matrix  $\mathbf{B}_{K}$  by Equations (6.7) to (8.9) requires some time, the a posteriori estimate  $\vec{x}_{K}$  cannot be available at the time instant  $t_{K}$ . Therefore, the control signals  $\frac{\sum_{k+1/K}^{1} u_{K}}{\sum_{k+1/K}^{1} u_{K}}$  must be computed from the a priori estimate  $\vec{x}_{K}^{2}$ .

Considering the fact mentioned above and the so-called separation theorem<sup>9,6,6</sup> the following relation can be derived:

$$\frac{\Gamma_{k+1/K}^{i}}{K+1/K} \overline{U}_{K}^{n} = \left[\frac{\phi_{k+1/K}^{i}}{\phi_{k+1/K}^{i}}\right] \cdot \left[\frac{x_{k}^{i}}{x_{k}^{i}}\right] \cdot \left[\frac{x_{k}^{i}}{x_{k}^{i}}\right] \cdot \left[\frac{\phi_{k+1/K}^{i}}{x_{k}^{i}}\right] \cdot \left[\frac$$

Substituting Equation (6.14) into Equation (6.12) the a priori and a posteriori estimates of the separated state vector in the Kalman equations are represented by the following equations:

$$\frac{\mathbf{x}_{k+1}^{i,*}}{\mathbf{x}_{k+1}^{*}} = \left[ \phi_{k+1/K}^{i,1} \left| \frac{\phi_{k+1/K}^{i,2}}{\mathbf{x}_{k}^{*}} \right| \frac{\mathbf{B}_{K}(\vec{\mathbf{y}}_{K} - \vec{\mathbf{y}}_{K}^{*})}{\mathbf{x}_{k}^{2}} \right]$$

$$(0.15)$$

Note: It is permissible, in order to make the transition from Equation (6.10) to (6.11), to cornider the vector  $\overline{u(t)}$  as a constant  $\overline{u(t_K)} = \overline{u_K}$  over the Kalman cycle interval. The term  $\frac{v_{K+1/K}u_K}{u_K}$  is then given by the expression

$$\frac{\Gamma_{K+1/K}}{u_{K}} = \int_{t_{K}}^{t_{K+1}} \frac{\phi(t_{K+1},\tau)}{D(\tau)} \frac{D(\tau)}{d\tau} \cdot \overline{u(t_{K})} .$$





Some remarks are now made with regard to the state vector, the computation of the transition matrix  $\phi$  , and the computation of the matrix  $H_{\rm g}$  .

The state vector  $\vec{x}_{k}^{\bullet}$  must include all important errors and error sources in the system and the associated sensors which cannot be represented by white noise. It contains the navigation errors, i.e. errors in position, velocity and platform orientation, errors of the inertial sensors and their respective misalignment on the platform, and errors of the redundant sensors. The sensor errors may originate in the sensor itself, in the environment or in the measurement principle. It is important to note that all errors which can be represented by white noise must be contained in  $\vec{h}_{k}^{\bullet}$  and  $\vec{q}_{k}^{\bullet}$ , while all the other errors must appear in the state vector  $\vec{x}_{k}^{\bullet}$ . The time behaviour of the latter ones must be describable either by linear homogeneous or by linear inhomogeneous differential equations, so that they can be included in the system of Equations (6.1) or (6.10). Many of the uncontrollable state variables are supplements to the state vector, necessary when the respective error sources are not describable by white noise sequences.

The most important error models are Markov processes, random walks and constant random biases. The correlated random processes can be generated from white noise by linear systems (shaping filters).

The Markov process, whose power spectral density has the form

$$\phi(\omega) = \frac{2\sigma^2 \beta}{\omega^2 + \beta^2} \tag{6.17}$$

can be described in the time domain by the linear differential equation

$$f(t) + \beta f(t) = \sqrt{(2\sigma^2 \beta)} w(t)$$
, (6.18)

where w(t), as the driving function, represents a unity variance white noise source.

A random walk process is obtained as the output of an integrator whose input is white noise. Its differential equation is of the form

$$f(t) = \sqrt{(r_{r_{T}})} w(t)$$
. (6.19)

A constant random bias can also be described by a linear differential equation, in fact by the most simple one which exists, namely,

$$f(t) = 0.$$
 (6.20)

Second Second

Its solution, supplied with a random initial condition, represents the desired constant bias error.

It can be seen that this practice of supplementing the state vector is a converient method if the error source forcing functions do not represent uncorrelated, zero-mean random processes. But this procedure may lead to a considerable number of state variables. Since the computational extent grows excessively with the number of variables (approximately by the third power), one is often constrained to consider only the most important error sources; this means to "suboptimise" the system. Another method of suboptimising, which results in a procedure similar to that described above for the controllable and uncontrollable states, is the partitioning of the state vector by placing the variables to be eliminated in one subsystem and the remaining variables in one or more additional subsystems. By this means the correlation between the variables in different subsystems is, in fact, ignored and the estimate is somewhat deteriorated as compared to the optimal case, but this reduction of the matrix dimensions results in some savings in computing time for the approximate optimal weighting matrix<sup>10</sup>. In every case of designing such a suboptimal system, the performance deterioration due to elimination or partitioning of some state variables should be predetermined by digital computer simulations and compared with optimal results. For example, using the method of partitioning the state vector, the "beat" partitioning depends also on the system conditions.

The computation of the transition matrix  $\frac{\dot{\phi}_{t/t_n}}{\phi_{t/t_n}}$  will now be discussed. It is known that  $\frac{\phi_{t/t_n}}{\phi_{t/t_n}}$  can be obtained from the coefficient matrix  $\underline{A(t)}$  by the differential equation<sup>12,15</sup>

$$\frac{d\phi_t/t_0}{dt} = \underline{A(t)} \frac{\phi_t/t_0}{t}, \qquad (6.21)$$

where

$$\frac{\phi_{t_0/t_0}}{I} = I \quad (identity matrix)$$

is the initial condition.

In the general case an explicit closed solution of this differential equation is not possible, but an infinite series can be derived, which converges uniformly in t for every matrix  $\underline{A(t)}$ . This solution is given as follows<sup>18</sup>:

والمراقبة والمراجب والراجان والمراجل والمراجل والمراجل والمراجل والمراجل والمراجل والمراجل والمراجل والمرجل

$$\frac{\phi_{0/t_0}}{f_0} = \underline{I} + \int_{t_0}^t \frac{A(\tau_1)}{t_0} d\tau_1 + \int_{t_0}^t \frac{A(\tau_1)}{t_0} \int_{t_0}^{\tau_1} \frac{A(\tau_2)}{t_0} d\tau_2 d\tau_1 + \int_{t_0}^t \frac{A(\tau_1)}{t_0} \int_{t_0}^{\tau_1} \frac{A(\tau_2)}{t_0} \int_{t_0}^{\tau_2} \frac{A(\tau_2)}{t_0} d\tau_2 d\tau_1 + \dots$$
(6.22)

In the case of constant coefficients, A = constant, Equation (6.22) can be transformed into the form

$$\underline{\phi}_{t/t_0} = \mathbf{I} + \underline{A}(t-t_0) + \underline{A}^2 \frac{(t-t_0)^2}{2!} + \dots = \mathbf{A}^{(t-t_0)}. \quad (6.23)$$

This result can also be derived directly from the differential Equation (0.21). However, in most cases the matrix A is not constant, at least some of the elements  $\mathbf{s}_{1k}$  are functions of time. In these cases the computation of the terms in Equation (0.22) is laborious, and it is better to subdivide the whole interval  $(t-t_0)$  and to consider A as a constant matrix during these partial intervals. The time-dependent elements of the matrix A are now approximated by a staircase function. By this means Equation (0.23) can be used to compute the partial transient matrices  $\frac{\phi_{t_p/t_{p-1}}}{\phi_{t_p/t_{p-1}}}$  and it can be shown<sup>18</sup> that the matrix  $\frac{\phi_{t_p/t_0}}{\phi_{t_p/t_{p-1}}}$  over the whole interval  $(t-t_0)$  is simply the product of the partial matrices  $\frac{\phi_{t_p/t_{p-1}}}{\phi_{t_p/t_{p-1}}}$ .

$$\frac{\phi_{t/t_0}}{t_{z}} = \frac{\phi_{t=t_p/t_{p-1}}}{t_{p-1}} \cdot \frac{\phi_{t_{p-1}/t_{p-p}}}{t_{p-2}} \cdots \frac{\phi_{t_g/t_1}}{t_{g}} \cdot \frac{\phi_{t_1/t_0}}{t_{g}}.$$
(6.24)

It is important to note that by this method the iteration time for the computation of the Kalman equations does not need to agree with the iteration time for computing the partial matrices  $\frac{\phi_{t_p}/t_{p-1}}{\phi_{t_p/t_{p-1}}}$ . Thus essentially longer cycle times for the computation of the Kalman equations can be chosen, depending on the speed of the available computer and the times when additional redundant measurements are available. The matrix  $\phi_{t_{t_1/K}}$  over the whole Kalman cycle interval must in these cases be computed using Equation (6.24). It is of great importance to pay attention to the number of partial intervals into which the Kalman cycle interval is divided. If it is sub-divided too finely, the linear term in the series expansion (6.23) is sufficient to compute the partial matrix  $\frac{\phi_{t_p/t_{p-1}}}{\phi_{t_p/t_{p-1}}}$ , but then the computation of the total matrix  $\frac{\phi_{t_p/t_{p-1}}}{\phi_{t_p/t_{p-1}}}$  may consume a too great portion of the available computer time because of the great number of sufficients into variables. It is often better, with regard to computing time and also with regard to accuracy, to subdivide the Kalman cycle for computation of the partial matrices  $\frac{\phi_{t_p/t_{p-1}}}{\phi_{t_p/t_{p-1}}}$  less finely and, instead, to take the quadratic term of Equation (6.23) into account. The matrix  $\frac{\phi_{t_p/t_{p-1}}}{\phi_{t_p/t_{p-1}}}$  less finely and, instead, to take the quadratic term of Equation (6.23) into account. The matrix  $\frac{\phi_{t_p/t_{p-1}}}{\phi_{t_p/t_{p-1}}}$  less finely and, instead, to take the subdivide the Kalman cycle for computation. The matrix  $\frac{\phi_{t_p/t_{p-1}}}{\phi_{t_p/t_{p-1}}}$  less finely and, instead, to take the quadratic term of Equation (6.23) into account. The matrix  $\frac{\phi_{t_p/t_{p-1}}}{\phi_{t_p/t_{p-1}}}$  less finely and, instead, to take the approximated by the equation

$$\frac{\phi_{t_p/t_{p-1}}}{2!} \simeq \frac{1}{2} + \frac{\Lambda(t_{p-1})}{2!} (t_p - t_{p-1}) + \frac{\Lambda^2(t_{p-1})}{2!} \frac{(t_p - t_{p-1})^2}{2!} .$$
(6.25)

For every system it is necessary to determine which of the two methods is better in a practical case. Probably the first method (finer subdividing and consideration of only the linear term) is to be preferred in a real system, in which the matrix  $\underline{A(t)}$  may change very quickly, whereas the second method is to be preferred for system performance error analysis, where comparatively slow changes of the matrix  $\underline{A(t)}$  occur.

The computation of the matrix  $H_g$  can be derived from the vector  $\overline{h_g}$  as follows:

$$\vec{\mathbf{h}}_{\mathbf{K}} = \int_{\mathbf{t}_{\mathbf{K}}}^{\mathbf{t}_{\mathbf{K}+1}} \frac{\phi(\mathbf{t}_{\mathbf{K}+1},\tau)}{\mathbf{n}(\tau)} \, \mathbf{n}(\tau) \, d\tau \qquad (6,26)$$

$$\underline{\mathbf{H}}_{\mathbf{K}} = \mathbf{E} \left[ \int_{\mathbf{t}_{\mathbf{K}}}^{\mathbf{t}_{\mathbf{K}+1}} \frac{\phi(\mathbf{t}_{\mathbf{K}+1},\tau_{1})}{\mathbf{n}(\tau)} \, \mathbf{n}(\tau_{1}) \, \mathbf{n}(\tau_{1}) \, \mathbf{n}(\tau_{1}) \, \mathbf{n}(\tau_{2})^{T} \, \frac{\phi(\mathbf{t}_{\mathbf{K}+1},\tau_{2})^{T}}{\mathbf{n}(\tau_{2})^{T}} \, \mathbf{n}(\tau_{2})^{T} \, \mathbf{n}(\tau_$$

Interchanging the order of averaging and integration, and defining the covariance matrix of the white noise vector n(t) as

$$\mathbb{K}[n(t_1)n(t_2)^T] = N(t_1) \delta(t_1 - t_2), \qquad (6.27)$$

one can write further

$$\begin{aligned} \mathbf{t}_{\mathbf{k}} &= \int_{\mathbf{t}_{\mathbf{k}}}^{\mathbf{t}_{\mathbf{k}+1}} \frac{\phi(\mathbf{t}_{\mathbf{k}+1},\tau_{1})}{\phi(\mathbf{t}_{\mathbf{k}+1},\tau_{1})} \frac{\mathbf{N}(\tau_{1})}{\mathbf{t}_{\mathbf{k}}} d\tau_{1} \int_{\mathbf{t}_{\mathbf{k}}}^{\mathbf{t}_{\mathbf{k}+1}} \delta(\tau_{1}-\tau_{2}) \frac{\phi(\mathbf{t}_{\mathbf{k}+1},\tau_{2})^{\mathrm{T}}}{\phi(\mathbf{t}_{\mathbf{k}+1},\tau_{2})^{\mathrm{T}}} d\tau_{2} \\ &= \int_{\mathbf{t}_{\mathbf{k}}}^{\mathbf{t}_{\mathbf{k}+1}} \frac{\phi(\mathbf{t}_{\mathbf{k}+1},\tau)}{\phi(\mathbf{t}_{\mathbf{k}+1},\tau)} \frac{\mathbf{N}(\tau)}{\phi(\mathbf{t}_{\mathbf{k}+1},\tau)^{\mathrm{T}}} d\tau \quad . \end{aligned}$$
(6.28)

In general, Equation (6.28) must be approximated by numerical methods. For example, if N(t) is considered as a constant matrix  $N_{\rm g}$  over the whole Kalman interval  $(t_{\rm g+1}-t_{\rm g})$  and if, according to Equation (6.24), this interval is divided into an even number p of sub-intervals of equal length  $\Delta t = t_{\rm g+1} - t_{\rm g}$ , then Equation (6.28) can very well be approximated by Simpson's rule:

ただと述べてい

ababababababat ing seberah tahun ing seberah seberah seberah seberah seberah seberah seberah seberah seberah se

$$H_{R} \simeq \frac{\Delta t}{3} \left( \frac{\phi_{t_{k+1} = t_{p}/t_{p-1}} \cdots \phi_{t_{1}/t_{0} = t_{k} \cdot k} \phi_{t_{1}/t_{0} = t_{k}}^{T} \cdots \phi_{t_{k+1} = t_{p}/t_{p-1}}^{T} + \frac{4\phi_{t_{k+1} = t_{p}/t_{p-1}} \cdots \phi_{t_{q}/t_{1} \cdot k} \phi_{t_{q}/t_{1}}^{T} \cdots \phi_{t_{k+1} = t_{p}/t_{p-1}}^{T} + \frac{2\phi_{t_{k+1} = t_{p}/t_{p-1}}}{\cdots \phi_{t_{q}/t_{p}}} \cdots \frac{\phi_{t_{q}/t_{q} \cdot k} \otimes k}{\varepsilon_{q}/\varepsilon_{q}} \cdots \phi_{t_{k+1} = t_{p}/t_{p-1}}^{T} + \frac{2\phi_{t_{k+1} = t_{p}/t_{p-1}}}{\cdots \varepsilon_{q}/\varepsilon_{q}} + \frac{2\phi_{t_{k+1} = t_{p}/t_{p-1}}}{\varepsilon_{q}} + \frac{1}{\varepsilon_{q}} + \frac{2\phi_{t_{k+1} = t_{p}/t_{p-1}}}{\varepsilon_{q}} + \frac{1}{\varepsilon_{q}} + \frac{2\phi_{t_{k+1} = t_{p}/t_{p-1}}}{\varepsilon_{q}} + \frac{1}{\varepsilon_{q}} + \frac{1}{$$

Of course, this solution can only be of theoretical interest if p > 2 , because the computation of Equation (5.29) becomes too laborious. In most cases, it will be sufficient to approximate Equation (6.24) by the simple expression

$$\frac{H_{\rm R}}{2} \simeq \frac{t_{\rm R+1} - t_{\rm R}}{2} \left( \frac{\phi_{\rm R+1/8} N_{\rm R} \phi_{\rm R+1/8}^{\rm T} + N_{\rm R}}{2} \right) \,. \tag{6.30}$$

No general expression can be derived for the measurement matrix  $\frac{M_{\rm H}}{M_{\rm H}}$  , because it depends upon the mechanized system and the mode of operation.

In Section 5 of this chapter the conventional statistical error analysis of aided inertial systems was described. With the sid of the Kalwan filter equations a statistical performance analysis with minimum error variances can be accomplished very effectively. For such an analysis only the last three of the five Kalman equations, namely (5.7) to (5.9), are required. It is an advantage that, in an ideal controlled system, the matrix P gives the variances of the states themselves, either directly or after simple transformations, for in such a system the error state variances will be reduced to a level equal to the variance with which each state is estimated. Therefore the following definitions are valid for the matrix  $P_{\mu}^{\phi}$  :

> Uncontrolled system (open loop operation)

# Ideal controlled system

(closed 1co

$$H_{R}^{2} = H[(\overline{x_{R}} - \overline{x_{R}^{2}})(\overline{x_{R}} - \overline{x_{R}^{2}})]$$

As pointed out earlier, no ideal controllable system exists in can also be derived for the covariances of the controllable states of a real system, in which the state vector is separated into con-trollable and uncontrollable states, as shown in Equation (6.13). By substituting Equation (6.14) into Equation (6.13) with negative sign (dorresponding to the way in which the control signals enter the system) the following relation is yielded:

$$\frac{1}{x_{k+1}^{2}} = \left[\frac{\phi_{k+1}^{1}}{x_{k}^{2}} + \frac{\phi_{k+1}^{1}}{x_{k}^{2}}\right] \cdot \left[\frac{x_{k}^{2}}{x_{k}^{2}} - \frac{x_{k}^{2}}{x_{k}^{2}}\right] + \frac{1}{h_{k}^{1}}, \qquad (6.32)$$

From Equation (6.32), finally,

والمراجع والمحافظ والمتحافظ والمحوط والمراجع والمحافظ والمحافظ والمحافظ والمحافظ والمحافظ والمحافظ والمحاف

$$\mathbf{E}[\overline{\mathbf{x}_{k+1}^{*}}, \overline{\mathbf{x}_{k+1}^{*}}] = [\phi_{k+1/k}^{*}] \phi_{k+1/k}^{*}] \mathbf{F}_{k}^{*}[\phi_{k+1/k}^{*}] (\phi_{k+1/k}^{*}) + \pi[\mathbf{h}_{k}^{*}\mathbf{h}_{k}^{*}].$$
(6.33)

If a short Kalman cycle time is considered, as compared to the time constants of the navigator and the associated error processes, the matrix  $\phi_{k+1/k}^{ii}$  does not differ such from the identity matrix and the matrix  $\phi_{k+1/k}^{ii}$  is mean zero. Besides, the vector  $h_k^{ii}$  is zero in all practical cases, because the random forcing  $\phi_{Kel/E}^{in}$  is near zero. Besides, the vector  $h_{k}^{in}$  is zero in all practical cases, because the random forcing Tunctions of the controllable states consist of correlated error sources. Therefore the appropriate part of T can be used as a good description of the actual covariances of the controlled states. Using these facts It is not necessary to carry out the solution of Equation (6.33), and the computational burden during performance analysis can be reduced substantially.

The error behaviour of an aided navigation system will now be investigated, using the method described for system performance analysis. It should be noted that, by this method, optimal values for the elements of the weighting matrix Bg are automatically computed. In contrast to the conventional, deterministic method described in Section 4, the effects of these gain factors will certainly be more complex, wince they do not enter the system at a single definite place, but because they influence in an optimal way every system state Accordingly they are, of course, not exactly equivalent to the formerly used deterministic gain factors  $K_1$ ,  $K_2$ or  $K_{0}$ , for it is easily possible to include these factors also in the matrix A(t) of the system Equation (6.1) or (6.10), respectively, and, in addition, to carry out a Kalman filtering. The meaning of the matrix  $B_{\rm g}$  is more comprehensive and more profound. Mathematically  $B_{\rm g}$  has the task of weighting the estimates with the redundant measurements and, in the optimal case, of establishing an orthogonal relation between the linear independent random vectors  $\overline{X_{\rm g}} - \overline{X_{\rm g}}$  (estimation error) and  $\overline{y_{\rm g}} - \overline{y_{\rm g}}$  (measurement error), and thus to accomplish the orthogonality theorem mentioned earlier.

$$\frac{K}{R} = \frac{K}{R} + \frac{K}$$

Figures 10 and 11 show the propagation of the position and velocity errors of an inertial navigation system in different modes used to navigate over the surface of the earth. It can be seen that here also considerable improvement of the accuracy can be obtained by additional redundant informations, as compared to pure inertial navigation. The velocity error in the pure inertial mode shows the characteristic Schuler period and the 24 hour period. In contrast to Figure 8, not only the influence of a single error source but the entire influence of all error sources is considered. The influence of any single error source could also be investigated by a change of the mothed described for the system performance error analysis, but, since this is an objective of suboptimal studies, it will not be discussed further.

In this example the state vector  $\overline{X_{K}}$  consists of 24 variables. The first 7 states are the well-known controllable inortial navigution errors whose propagation equations are derived in several textbooks (References 7, 11, etc.). The other variables are the uncontrollable error sources which cannot be represented by white noise.

## REFERENCES

1,	Branner, K.	Zur optimalen linearen Filterung und Vorhersage instationärer Zufallsprozesse in diskreter Zeit. Regelungstechnik, Heft 3, 16 Jahrg., 1968, pp.105–110.	
2.	Brown, R.G., Friest, D.T.	Optimization of a Hybrid Inertial Solar-Tracker Navigation System. Institute of Electrical and Electronic Engineers, International Convention Record, New York, 1964, Part 7, March 23-26, 1964, pp.121-135.	
3,	Davenport, W.B., Jr, Root, W.L.	An Introduction to the Theory of Random Signals and Noise. International Student Edition, McGraw-Hill, New York, 1958.	
4,	Fagin, S.L.	Recursive Linear Regression Theory, Optimal Filter Theory, and Error Analysis of Optimal Systems. Institute of Electrical and Electronic Engineers, International Convention Record, Vol.12, Part 1, New York, March 1964, pp.216-245.	
. <b>B</b> ,	Gunokel, T.F. Franklin, G.F.	A General Solution for Linear Sampled Data Control Systems. Transactions, American Society of Mechanical Engineers, Journal of Basic Engineering, June 1963, pp.197-203.	
6.	Joseph, P., Tou, J.T.	On Linear Control Theory. American Institute of Electrical Engineers, Transactions (Applications and Industry), Vol.80, 1961, pp.193-196.	
7.	Kachikus, G.A.	Error Analysis for Gruiss Systems. Chapter 7 of "Inertial Guidance", edited by G.R.Pitman Jr. University of California Engineering and Physical Sciences Extension Meries, John Wilay, New York, 1962.	
8,	Kalman, R.E.	A New Approach to Linear Filtering and Prediction Problems. Transactions of the American Society of Mechanical Engineers, Series D, Volume 82D, 1930, pp.35-45.	
Đ.	Kalman, N.E.	Contributions to the Theory of Optimal Control. Bulletin of the Mexican Mathemati- cal Scriety, Hecond Beries, Volume 5, 1960, pp.102-119.	0
10,	Pentacost, E.E., Stubberud, A.R.	Synthesis of Computationally Efficient Sequential Linear Estimators. Institute of Electrical and Electronic Engineers, Transactions, Vol.AFS-3, No. 2, March 1987.	
11.	Pinson, J.C.	Inertial Guidance for Cruise Vehicles. Part 2 of "Guidance and Control of Aero- space Vehicles", edited by C.T.Leondes, University of California Engineering and Sciences Extension Series, McGraw-Hill, New York, 1963.	
12,	Nohmuidler, W.	Vorträge über Determinanten und Matrisen mit Anwendungen in Physik und Technik. Berlia, 1949, pp.137-140.	an an the second se
13	. Sorenson, H.W.	Kalman Filtering Techniques. Contribution to the book "Advances in Control Systems, Theory and Applications". Volume 3, edited by C.T.Leondes, Academic Press, New York, 1966.	
14.	. Tou, J.T.	Optimum Design of Digital Control Systems. Academic Press, New York, 1963.	
15	. Zarmühl, R.	Matrizen. Springer, Berlin-Göttingen-Heidslberg, 1964, pp.442-446.	



AL .......



Contesti and

Bin William Market

Fig.1 Inputs and outputs of navigation systems



Fig. 2 Levels of navigation information

٠١.

484











Ś



XC A

In Manual Million

may - shop

487



Ì.

## Fig.5 Error block diagram for the x-channel of a Schuler-tuned system



Fig.6 Updating mechanisation



الأنور ويورد المؤاهلة

Well And the A Line

State What Hands





Fig.8 Velocity error due to fixed gyro drift ( $\epsilon_y = 0.1^{\circ}/h$ ) for different systems

- 1. Pure inertial system  $(K_1 = 0; K_2 = 0; K_3 = 0)$
- 2. Doppler damped inertial system  $(K_1 = 0.6\omega_0; K_2 = 0; K_3 = 0)$
- 3. Doppler inertial system ( $K_1 = 0.6\omega_8$ ;  $K_2 = 0.236$ ;  $K_3 = 0$ )
- 4. Doppler inertial system with an additional integrator  $(K_1 = 0.0\omega_g; K_g = 0.236; K_g = 0.2312\omega_g)$

Velocity Increments  $|\Delta_v|$ **Navigation** IMU  $\Delta \phi$ Computer Torquing Commands -**Corrections** of Navigation Information Controller Optimum Filter Measurement Data (Linear Error Model)

## Fig.9 Block diagram of an inertial system with optimum filter

489

a set para





والمرجع والمناجع والمتعادية والمرجع والمتعادية والمتعادية والمتعادية

External Data

Corrections of the Error Model

and the state of the second second second second

ŀ



الم المراجع في المراجع عنه المراجع عنه المراجع المراجع المراجع المراجع المراجع المراجع المراجع المراجع المراجع والمراجع المراجع المراجع

· · · · ·

Anna Carner Pal
# CHAPTER 19 - APPLICATION OF KALMAN FILTERING TECHNIQUES TO STRAPDOWN SYSTEM INITIAL ALIGNMENT

**KO**X: it's

Di Mannaha

-

by

Earle B. Crocker and Leonard Rabins

Inertial Guidance Engineering Ordnance Systems General Electric Company 100 Plastics Avenue Pittsfield, Massachusetts, USA



# 

# STAR MANY MARK

# ante anter anter anter



CHAPTER 19 - APPLICATION OF KALMAN FILTERING TECHNIQUES To Strapdown System Initial Alignment

### Earle B. Crocker and Leonard Rabins

# 1. INTRODUCTION

155757578 1

Â,

Initial attitude alignment of systems involving inertial instruments is a requirement caused by the fact that sensors commonly used in inertial guidance, navigation, or attitude reference systems measure angular rates or integrals of angular rates (gyro outputs), or linear acceleration or integrals of linear accelerations (accelerometer outputs). When the meed exists for continuous knowledge of absolute angular orientation of the frame or platform to which the sensors are mounted, then (just as initial values for integrals) the initial attitude must be established. This is the alignment problem which is considered in this chapter<sup>a</sup>. "The basic objective is the investigation of digital methods for the determination of the initial alignment of a "tradown inertial navigation system from vibration and away corrupted data on the launch pad.

The inertial components may be mounted on a stabilized platform or directly strupped down to an uncontrolled swaying body. In either case the essential mathematics of establishing alignment from outputs of the inertial components themselves is identical. In a strapdown inertial system, the measurements are made in the vehicle or body coordinate system. Angular information obtained from three single-axis platforms or three pulse-torqued gyros is used to update a coordinate transformation matrix relating a set of snalytical or computing axes to the body axes. This boordinate transformation matrix is, then used to transform the accelerometer outputs from the body axes for the computing axes. The initial alignment problem is concerned with determining the proper initial walves for the elements of this coordinate transformation matrix.

If the initial computing axes are earth-fixed at the launch site, the problem is reduced to one of determining the orientation of these axes with respect to local vertical and an animuth reference.

The azimuth angle can be determined from an optical measurement, as shown in Reference 1. The determination of vertical involves using the accelerometer outputs which have been resolved through the coordinate transformation matrix into a reference coordinate system. This system is maintained nominally earth-fixed at the launch site by using the gyro outputs to compensate for rotations caused by vehicle away. This reference system will deviate from an earth-fixed system due to gyro drift because proper compensation for the earth's rate cannot be provided until the orientation of the reference system with respect to the earth is known.

If vehicle motions are sero, the determination of the orientation of the reference system with respect to vertical is trivial since the accelerometer outputs in the reference system are a simple function of the orientation and the acceleration of gravity. When there is vehicle motion, the output of the accelerometers due to gravity must be separated from the output due to vehicle motion by filtering.

Filtering methods investigated under this ctudy were all based on the criterion of a minimum mean square error and differ mainly in the mechanization technique and the amount of a priori knowledge of the statistics which is assumed. A prime effort of the study is to investigate the accuracy of a Kalman filter in which both the state equations and noise are modeled differently from the "true" characteristics. A useful filter must exhibit little sensitivity to these modeling errors since the "true" characteristics, particularly the noise Sources, are never known.

Methods considered included an optimal filter of the Kalman type, a simplified Kalman filter in which the effects of gyro drift and improper earth's rate compensation were neglected, a maximum likelihood approach, and a leastsquares curve fit technique with and without preconditioning of the data. This chapter will emphasize the results and analysis of the Kalman filter, since the recommended approach was use of the simplified Kalman filter. The details of the analysis of the other techniques can be found in Reference 2.

# 2. ANALYSIS

# 3.1 Coordinate System Definition

والمستوية والمروح والمروح والمراجع والمراجع والمروح والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع

Alignment of a coordinate system defining a strapdown or platform system implies knowledge of a basic reference system relative to which the alignment is performed. The reference system used here is earth fixed, with one axis along the local vertical corresponding to the average position of the swaying strapdown system. One horizontal

PRECEDING PAGE BLANK

\* This need obviously occurs for instital guidance and other systems. See the book Inertial Guidance by U.R.Pitsan.

axis is assigned an arbitrary azimuth. This coordinate system is designated CS(0). The strapdown reference analytic system is CS(1) and is the coordinate system in which all measurements by the strapdown instrument system CS(2) are referred. Alignment establishes the relationship between CS(0) and CS(1). The strapdown system computer further contains information expressing the relation between CS(1) and CS(2). In the following analysis CS(1) can be considered to coincide with CS(2) initially.

At the initial time CS(1) is missligned from CS(0) by the small angle  $(\alpha, \beta, \gamma) = \theta$ , as shown in Figure 1.

Nhile CS(0) rotates exactly at the earth's rotational rates, CS(1) cannot be correctly "torqued" since the misalignment angles  $\alpha$ ,  $\beta$ ,  $\gamma$  are not known. This causes these angles to vary with time due to "earth's rate coupling". The angles, additionally, change due to error drift rate of the gyros.

Since  $x_0$  is defined as locally vertical, a neasurement of acceleration, A, along axes  $z_1$  and  $y_1$  is proportional to the misalignment angles; thus,

$$A_{\mathbf{z}_1} = \mathbf{z}^{\mathcal{A}} \tag{1}$$

 $A_{\nu} = s\gamma$  (2)

In the strapdown system, the acceleration sensing instruments are actually located along axes of CS(2) (or some other axes whose relationship to CS(2) is known by physical alignment), but the measurements are transformed into CS(1) in the strapdown computer. The acceleration measured includes the effect of random horizontal accelerations caused by the missile away. This is the principal noise input to which the filtering methods are applied.

# 2.2 Direction Comine Relations Between Rotating Coordinate Systems

Certain basic direction cosine matrix equations are of importance in understanding relationships between rotating coordinate systems. Let  ${}^{10}c_{g}$  denote the direction cosine matrix which transforms components  $(V_{xx}, V_{xy}, V_{xy})$  of a vector V in coordinate system 2 to its components in coordinate system 1. Then,

$$V_{1} = {}^{1}C_{\mu}V_{\mu}$$
(3.)

or, in detail,

$$\begin{pmatrix} V_{1x} \\ V_{1y} \\ V_{1y} \\ V_{1z} \\ \end{pmatrix} = \begin{pmatrix} C_{x_1, x_2} & C_{x_1, y_2} & C_{x_1, x_2} \\ C_{y_1, x_2} & C_{y_1, y_2} & C_{y_1, x_2} \\ C_{z_1, x_2} & C_{z_1, x_2} & C_{z_1, x_2} \\ \end{pmatrix} \begin{pmatrix} V_{zx} \\ V_{zy} \\ V_{zy} \\ V_{zz} \end{pmatrix} , \qquad (3b)$$

The individual components of the "C, matrix denote the cosine of the angle between the two subscripted axes.

Let coordinate system 1 be fixed in inertial space while system 2 rotates at a rate  $\omega_{g}$ , where the angular rotation rate is measured or given as components along the axes of coordinate system 2. Then\*

$$\frac{d^{1}C_{g}}{dt} = {}^{1}C_{g}\omega_{g}, \qquad (4)$$

where  $\omega_s$  is the skew-symmetric matrix

 $\omega_{g} \approx \begin{bmatrix} 0 & -\omega_{\mathbf{x}_{g}} & \omega_{\mathbf{y}_{g}} \\ \omega_{\mathbf{x}_{g}} & 0 & -\omega_{\mathbf{x}_{g}} \\ -\omega_{\mathbf{y}_{g}} & \omega_{\mathbf{x}_{g}} & 0 \end{bmatrix} .$ (5)

Finally, let coordinate system 0 be considered inertial, with systems 1 and 2 both rotating relative to the inertial system. Then, to establish the relation between systems 1 and 2, we note from Equation (4)

 $\frac{d}{dt} = {}^{0}C_{1} \omega_{1}$   $\frac{d}{dt} = {}^{0}C_{2} \omega_{2} .$   ${}^{0}C_{2} = {}^{0}C_{1} {}^{1}C_{2} .$   $\frac{d}{dt} {}^{1}C_{2} = {}^{1}C_{2} \omega_{2} - \omega_{1} {}^{1}C_{2} .$ 

it follows that

and

Since

• Matrix Methods for Engineering by L.A.Pipes, Prentice-Hall, 1963, p. 167.





(6)

### 1.3 System Dynamics

The equations describing the strapdown erection process are derived in this section. These include the differential equations governing the angles describing the misslignment between the earth-fixed launch-site coordinate system (CS) and the strapdown inertial reference CS, the transformed accelerometer outputs, and the misslie sway velocity. The differential equations are then converted to a set of first-order difference equations, since this is the form spyroprinte for the application of the discrete-time Kalman filter equations. Certain assumptions made in this analysis are as follows:

- (i) The initial misalignment angles are small enough so that the small angle approximations (sin  $\theta = \theta$ , cos  $\theta = 1$ ) are valid.
- (ii) Inertial component sensing axes are orthogonal to each other.
- (111) The gyro drift rates are constant.
- (iv) Gyro torquing errors and accelerometer instrument errors are small and not considered.

(v) The sway velocity is adequately represented by a narrow-band noise process having zero mean value.

# 2.3.1 Transformed Accelerometer Outputs

The inertial instruments which measure linear acceleration are integrating devices, so that their outputs are proportional to the integral of acceleration over fixed time intervals. In particular, the outputs along  $y_1$  and  $z_1$  axes are given by

$$\mathbf{P}_{1}(\mathbf{t}_{1}) = \int_{\mathbf{t}_{1-1}}^{\mathbf{t}_{1}} \mathbf{s}\beta(\tau) \, d\tau + \mathbf{V}_{1}(\mathbf{t}_{1}) - \mathbf{V}_{1}(\mathbf{t}_{1-1}) \tag{7}$$

$$P_{g}(t_{i}) = \int_{t_{i-1}}^{t_{i}} -g\gamma(\tau) d\tau + V_{g}(t_{i}) - V_{g}(t_{i-1}) , \qquad (8)$$

where  $P_i(t)$  is the output along  $z_i$  at time t

- $P_{e}(t)$  is the output along  $y_{1}$  at time' t
- V<sub>1</sub>(t) is the sway velocity along B<sub>1</sub>
- V.(t) is the sway velocity slong y,
- and g is the value of gravity (s/sec2).

The horizontal accelerations caused by missile sway are seen to be included in the system measurements. While the missile sway causes spurious accelerations (noise) to be measured, the sway is limited to finite values whose average acceleration is zero. This allows the noise to be modeled as a zero mean process. In certain practical situations (ships, sirplanes) in which the net average motion is not zero, additional inputs are required to describe the motion.

2.3.2 Equations Coverning the Angles α, β, γ Describing the Alignment Between the Earth-Fixed Launch-Site Coordinate System and the Strapdown Reference Coordinate System

The three coordinate systems of concern in this analysis are

# CB(0) - Earth-fixed launch-site CS

- CB(1) Strapdown reference CS
- CB(2) Strapdown instrument CS.

The transformation equation solved by the strapdown computing system relates CB(2) to CS(1). This equation follows from Equation (6),

$$\frac{d}{dt} = {}^{1}C_{g}\omega_{g} - \omega_{0}{}^{1}C_{g}, \qquad (9)$$

5-160-16-1-1

where  $\omega_{a}$  is the matrix of angular rates measured by the gyros.

 $\omega_n$  is the matrix of earth's rate in CS(0).

In Equation (9), the value of earth's rate in CS(0) is used rather than the proper values corresponding to the rate in CS(1), since there is as yet no information relating the alignment of CS(1) to CS(0).

The gyro measured rates are in error by the gyro drifts. Thus

 $\omega_s = \dot{\theta} + d_s$ 

495







fett me Beifere firm Dreiber





(10)

where  $\hat{\theta}$  represents the true angular rates of the awaying body and d<sub>2</sub> represents the gyro drift rates in the instrument axes.

The inverse equation to Equation (9) is

$$\frac{d^{2}C_{1}}{dt} = {}^{2}C_{1}\omega_{0} - \omega_{2} {}^{2}C_{1} .$$
(11)

Subsituting for the measured rate  $\omega_2$  from Equation (10),

 $\frac{d^{2}C_{1}}{dt} = {}^{2}C_{1}\omega_{0} - \dot{\theta} {}^{2}C_{1} - d_{2} {}^{2}C_{1}. \qquad (12)$ 

In the same manner as for Equation (9),

$$\frac{d}{dt} = {}^{0}C_{g}\dot{\theta} - \omega_{0} {}^{0}C_{g}$$
(13)

and, with

°C, = °C, <sup>2</sup>C, ,

we find, using Equations (12) and (13) that

$$\frac{d}{dt} {}^{0}C_{1} = {}^{0}C_{1} \omega_{0} - \omega_{0} {}^{0}O_{1} - {}^{0}O_{2} d_{2} {}^{0}O_{1} . \qquad (14)$$

This is the basic equation which relates the sizapdown reference CB(1) to the earth-fixed system CB(0).

The equation is simplified by first noting that the drift term can be written as

$${}^{0}C_{g} d_{g} {}^{2}C_{i} = {}^{0}C_{i} {}^{1}C_{g} d_{g} {}^{2}C_{i} = {}^{0}C_{i} d_{i} .$$
 (15)

Components of the drift will be assumed constant in CB(1) rather than CB(2).

Since CS(1) differs from CS(0) by small angles, as shown in Figure 1, <sup>6</sup>C, can be written as

$${}^{0}C_{1} = \begin{bmatrix} 1 & -\gamma & \beta \\ \gamma & 1 & -\alpha \\ -\beta & \alpha & 1 \end{bmatrix}, \quad (16)$$

Substituting Equation (16) in Equation (14) then causes Equation (14) to simplify to

$$\begin{bmatrix} \mathbf{d} & \boldsymbol{\alpha} \\ \mathbf{d} \mathbf{t} \\ \mathbf{d} \\$$

The equation for the full metrix d  ${}^{0}C_{1}/dt$  reduces to only three equations, since the angles  $\alpha$ ,  $\beta$ ,  $\gamma$  are small. The  $\omega_{0}$  matrix contains the values of earth's rate in CS(0), and the unknown drifts appear in the d matrix. Filtering will produce estimates for both the mixulignment angles and the drifts.

# 2.3.3 Equations Describing the Noise Inputs to the System Velocity Measurements

The velocity equations (7) and (8) indicate the presence of noise terms. This noise is predominantly chused by missile sway. The noise cannot be represented as white, but rather a correlated velocity noise model is a better representation, due to the dynamics of the missile sway. The correlation function to be used is

$$\mathbf{R}_{\mathbf{v}_{1}}(\Delta) = \sigma_{\mathbf{v}}^{2} \in [\Delta]/\tau \cos \omega_{\mathbf{g}} \Delta .$$
 (18)

To generate this random variable from white noise inputs, let"

$$\mathbf{v}_{1}(t) = \mathbf{n}_{1}(t) \cos \omega_{n} t + \mathbf{n}_{s}(t) \sin \omega_{n} t .$$
 (19)

• Probability, Rundom Variables, and Stochastic Processes by A. Papoulis, Modraw-Hill, 1965, p. 373

Antheniter A (W)



where  $n_i(t)$  and  $n_g(t)$  are independent random variables with the autocorrelation function

$$\mathbf{R}_{\mathbf{n}_{1}}(\Delta) = \mathbf{R}_{\mathbf{n}_{2}}(\Delta) = \sigma_{\mathbf{v}}^{\mathbf{g}} e^{-|\Delta|/\tau}, \qquad (20)$$

Noise with this autocorrelation function can be considered as being the solution to the stochastic differential equations

$$h_{i}(t) = (-1/\tau)n_{i}(t) + w_{i}(t)$$
 (21)

$$\dot{u}_{g}(t) = (-1/\tau) n_{g}(t) + w_{g}(t) ,$$
 (22)

where  $w_i(t)$  and  $w_g(t)$  are white noise random variables with autocorrelation function

$$\mathbb{E}[w_{1}(t_{1})w_{1}(t_{g})] = \mathbb{E}[w_{g}(t_{1})w_{g}(t_{g})] = \frac{2\sigma_{y}^{2}}{\tau} \delta(t_{1}-t_{g}); \qquad (23)$$

 $\delta(t_1 - t_2)$  is the Dirac delta function.

 $v_1(t)$  is the away velocity along the H axis. For the sway velocity along the y axis,  $v_2(t)$ , we have similar equations:

$$r_{g}(t) = n_{g}(t) \cos \omega_{g} t + n_{g}(t) \sin \omega_{g} t$$
 (24)

$$\dot{n}_{s}(t) = (-i/r)n_{s}(t) + w_{s}(t)$$
 (25)

$$\hat{n}_{u}(t) = (-1/\tau)n_{u}(t) + w_{u}(t) , \quad (26)$$

where

. .

$$\mathbf{E}[\mathbf{w}_{s}(t_{i})\mathbf{w}_{s}(t_{s})] = \frac{2\sigma_{v}^{2}}{\tau} \delta(t_{i} - t_{s}) , \qquad (27)$$

# 2.3.4 Conversion of the System Equations to the Kalman State Equation Form

The system equations are described by Equations (7), (8), (17), (19), (21), (22), (24), (35) and (26). These will be changed to a difference or state equation form directly recognizable as the common form for application of the Kalman filter equations. The process is fairly direct though tedious and intermediate results will not be shown\*. All the differential equations in the set are linear; they are of the form

$$\frac{dx}{dt} = Fx(t) + \omega(t)$$
(28)

and have the general solution

۰.

$$\mathbf{x}(t_{1}) = \phi(t_{1}, t_{1-1}) \mathbf{x}(t_{1-1}) + \int_{t_{1-1}}^{t_{1}} \phi(t_{1}, \tau) \omega(\tau) d\tau$$
(30)

which is of the proper state equation form. In Equation (30) the state transition matrix is the matrix exponential

$$\phi(t_{i}, t_{i+1}) = e^{P(t_{i}-t_{i+1})} . \tag{31}$$

The full state equation is finally obtained in the following form:

$$\begin{bmatrix} \mathbf{a}(\mathbf{t}_{1}) \\ \beta(\mathbf{t}_{1}) \\ \gamma(\mathbf{t}_{1}) \\ \mathbf{d}_{\mathbf{x}}(\mathbf{t}_{1}) \\ \mathbf{d}_{\mathbf{x}}(\mathbf{t}_{1}-\mathbf{1}) \\ \mathbf{d}_{\mathbf{x}}(\mathbf{t}_{1-1}) \\ \mathbf{d}_{\mathbf{x}}(\mathbf{t}$$

• For a detailed derivation see Reference 2.

(32=)

的一种对于

 $x(t_i) = \phi(t_i, t_{i-1})x(t_{i-1}) + Q(t_i)u(t_i)$ ,

where

$$\phi_{11} = I - \frac{\sin \omega (t_1 - t_{1-1})}{\omega} \omega_0 + \frac{1 - \cos \omega (t_1 - t_{1-1})}{\omega^2} \omega_0^2$$
(33)  
$$\omega = (\omega_{\pi_0}^2 + \omega_{\pi_0}^2 + \omega_{\pi_0}^2)^{1/2}$$

# $\omega_{\rm o}$ is the $\omega$ matrix of Equation (17) J is the 3 x3 unit matrix,

# $\phi_{i2} = I(t_i - t_{i-1}) + \frac{1 - \cos \omega(t_i - t_{i-1})}{\omega^2} \omega_0 - \left[\frac{t_i - t_{i-1}}{\omega^2} - \frac{\sin \omega(t_i - t_{i-1})}{\omega^3}\right] \omega_0^2$ (34)

$$\phi_{31} = \dots \begin{bmatrix} 0 & \mathbf{g} & 0 \\ 0 & 0 & -\mathbf{g} \end{bmatrix} \phi_{12}$$
(35)

$$\phi_{st} = \begin{bmatrix} 0 & s & 0 \\ 0 & 0 & -s \end{bmatrix} \left\{ -\frac{(t_1 - t_{1-1})^2}{2} \mathbf{I} + \left( \frac{t_1 - t_{1-1}}{\omega^2} - \frac{\sin(t_1 - t_{1-1})}{\omega^3} \right) \omega_0 - - \left( \frac{(t_1 - t_{1-1})^2}{2\omega^3} - \frac{1 - \cos \omega(t_1 - t_{1-1})}{\omega^3} \right) \omega_0 \right\}$$

$$(36)$$

$$\phi_{au} = O(h_1) \phi_{uu} - C(t_{i-1})$$
(37)
$$\phi_{uu} = e^{-(t_1 - t_{i-1})/\tau} I$$
(38)

I is the 4×4 unit matrix

$$\mathbf{U}(\mathbf{t}) = \begin{bmatrix} \cos \omega_{\mathbf{k}} \mathbf{t} & \sin \omega_{\mathbf{k}} \mathbf{t} & \mathbf{0} & \mathbf{0} \\ \mathbf{0} & \mathbf{0} & \cos \omega_{\mathbf{k}} \mathbf{t} & \sin \omega_{\mathbf{k}} \mathbf{t} \end{bmatrix},$$
(39)

The noise terms which appear in the state equations are related to the white noise of Equations (21) to (28)

$$\begin{bmatrix} u_1(t_1) \\ u_2(t_1) \\ u_3(t_1) \\ u_4(t_1) \end{bmatrix} = \int_{t_{1-1}}^{t_1} \phi_{au}(t_1,\tau) \begin{bmatrix} u_1(\tau) \\ u_2(\tau) \\ u_3(\tau) \\ u_4(\tau) \end{bmatrix} d\tau$$

and have the variance

$$Q(t_{*}) = B(uu') = \sigma_{*}^{2}(1 - \epsilon^{-2(t_{1}-t_{1})/\tau}) I. \qquad (40)$$

The data measurement consists of the two transformed accelerometer outputs. There is additional noise of the measurement due to quantization of the accelerometer outputs. The measurement equation is given by

$$y(t_i) = H(t_i)X(t_i) + r(t_i)$$
, (41)

where

r(t<sub>1</sub>) is

 $\mathbf{H}(\mathbf{t_{i}}) = \begin{bmatrix} \mathbf{0} & \mathbf{0} \\ \mathbf{0} & \mathbf{0} \end{bmatrix}.$ 

$$\mathbf{R}(t_{1}) = \mathbf{R}[r(t_{1})r'(t_{1})] = \frac{\Delta^{2}}{12} \begin{bmatrix} 1 & 0 \\ 0 & 1 \end{bmatrix}.$$
(43)

where  $\Delta$  is the quantization level.

When a theodolite measurement is taken and processed, the resulting quantity is the azimuth misalignment plus an error which is equal to the isveling error about the projection of the LOB in the horizontal plane multiplied by the cangent of the elevation angle. Thus, when a theodolite measurement is taken, the  $H(t_i)$  matrix is

$$H(t_{i}) = [1, \tan \gamma_{\alpha} \sin \zeta, \tan \gamma_{\gamma} \cos \zeta, 0, 0, 0, 0, 0, 0, 0, 0], \qquad (44)$$

where  $\zeta$  is the angle between the projection of the LOS in the horizontal plane and the  $x_1$  axis.

(325)

(42)

# 3. OPTIMAL FILTER EQUATIONS

In Section 2, the system equations were derived. They are repeated here for convenience.

$$\mathbf{x}(t_i) = \phi(t_i, t_{i-1}) \mathbf{x}(t_{i-1}) + \mathbf{G}(t_i) \mathbf{u}(t_i)$$
(45)

$$y(t_1) = H(t_1) x(t_1) + r(t_1)$$
. (46)

The well-known Kalman filter Equations can be applied to systems with this description. These equations give an estimate of the state vector  $x(t_i)$ , based on the measurements  $\{y(t_i) \dots y(t_i)\}$ , which is optimal in the mean-square sense. The filter is described by the following difference equations:

$$\hat{\mathbf{x}}(t_1) = \hat{\mathbf{x}}(t_1, t_{1-1}) + \mathbf{x}(t_1)[\mathbf{y}(t_1) - \mathbf{H}(t_1)\hat{\mathbf{x}}(t_1, t_{1-1})], \qquad (48)$$

The optimal filter gains  $K(t_i)$  are calculated from the following set of difference equations:

$$P(t_{i},t_{i+1}) = \phi(t_{i},t_{i+1})P(t_{i+1})\phi(t_{i},t_{i+1}) + u(t_{i})Q(t_{i})u'(t_{i})$$
(42)

$$K(t_{i}) = P(t_{i}, t_{i-i})H'(t_{i})[H(t_{i})P(t_{i}, t_{i-i})H'(t_{i}) + R(t_{i})]^{-1}$$
(50)

 $P(t_i) = [I - K(t_i)H(t_i)] P(t_i, t_{i-i}) , \qquad (51)$ 

where  $\hat{\mathbf{x}}(t_1)$  is the optimal estimate of  $\mathbf{x}(t_1)$ , given the measurements  $\mathbf{y}(t_1)$ ,  $\mathbf{y}(t_2)$ ,  $\dots$   $\mathbf{y}(t_2)$ 

- $\hat{x}(t_i, t_{i-1})$  is the optimal estimate of  $x(t_i)$ , given the measurements  $(y(t_i), y(t_j), \dots, y(t_{i-1}))$ 
  - $P(t_i)$  is the covariance matrix of the error in the estimate  $\hat{x}(t_i)$  .
- $P(t_i, t_{i+1})$  is the covariance matrix of the error in the estimate  $\hat{x}(t_i, t_{i+1})$  .

These equations describe the optimal filter. It is desirable, however, to simplify and decouple the equations to reduce the computational requirements. The performance of the optimal filter is described in Section 5. It is also demonstrated that the degradation caused by certain simplifications is negligible.

These simplifications will be investigated in the following section.

# 4. SUBOPTIMAL FILTER EQUATIONS

### 4.1 Introduction

The filter equations presented in the previous section are rather complex. They involve 12 simultaneous difference equations and hence multiplication of 12 × 12 matrices. It is desirable, then, to investigate the possibility of simplifying these equations. In order to determine if the simplifications are acceptable, one must calculate the degradation in filter performance caused by them. In addition, the sensitivity of the filter to errors in the noise should be investigated. The equations required to evaluate the suboptimal filter performance are derived in this section. The approximations to the system dynamics which are used in the suboptimal filter.

The suboptimal filter is obtained by simplifying the state equation such that the state transition matrix  $\phi(t_i, t_{i-1})$  of Equation (45) is replaced by an approximate and simpler form  $\phi_g(t_i, t_{i-1})$ .

A fairly general model for the suboptimal filter equations is then given by

$$\hat{\mathbf{x}}_{-}(t_{1},t_{1-1}) = \phi_{-}(t_{1},t_{1-1})\hat{\mathbf{x}}_{-}(t_{1-1})$$
(52)

$$\hat{X}_{0}(t_{1}) = \hat{X}_{0}(t_{1}, t_{1-1}) + K_{0}(t_{1})[y(t_{1}) - H(t_{1})\hat{X}_{0}(t_{1}, t_{1-1})]$$
. (53)

The subscript s denotes suboptimal quantities. The gain  $f_{\mu}(t_1)$  is obtained from

 $P_{n}(t_{1},t_{1-1}) = \phi_{n}(t_{1},t_{1-1})P_{n}(t_{1-1})\phi_{n}'(t_{1},t_{1-1}) + O_{n}(t_{1})Q_{n}(t_{1})O'(t_{1})$ (54)

$$\mathbf{K}_{\mathbf{g}}(t_{1}) = \mathbf{P}_{\mathbf{g}}(t_{1}, t_{1-1})\mathbf{H}'(t_{1})[\mathbf{H}(t_{1})\mathbf{P}_{\mathbf{g}}(t_{1}, t_{1-1})\mathbf{H}'(t_{1}) + \mathbf{R}(t_{1})]^{-1}$$
(85)

$$P_{n}(t_{i}) = [I - K_{n}(t_{i})H(t_{i})]P_{n}(t_{i}, t_{i-1})$$
 (36)

These equations are identical with the usual Kalman set except that they involve suboptimal quantities. To determine the true covariance matrix of the estimate errors, we first define the quantities





V. A Landarstard A. de

 $P_1(t_1) = E[x(t_1)x'(t_1)]$ 

$$P_{\mathbf{I}}(\mathbf{e}_{\mathbf{i}},\mathbf{e}_{\mathbf{i}-\mathbf{i}}) = K[\mathbf{x}_{\mathbf{H}}(\mathbf{e}_{\mathbf{i}},\mathbf{e}_{\mathbf{i}-\mathbf{i}}),\mathbf{x}'(\mathbf{e}_{\mathbf{i}})]$$

$$P_{\underline{x}}(t_{\underline{i}}) = E[\underline{x}_{\underline{n}}(t_{\underline{i}}) \mathbf{x}'(t_{\underline{i}})]$$

$$P_a(t_i, t_{i-i}) = \mathbb{E}[\hat{x}_n(t_i, t_{i+1})\hat{x}_n(t_i, t_{i-1})]$$

$$P_{\mathbf{s}}(\mathbf{t}_{1}) = \mathbf{E}[\hat{\mathbf{x}}_{\mathbf{s}}(\mathbf{t}_{1})\hat{\mathbf{x}}_{\mathbf{s}}'(\mathbf{t}_{1})] ,$$

The covariance matrix of estimate errors for the suboptimal filter is given by

$$\Gamma(t_{i}) = \mathbb{E}\left[\left[x(t_{i}) - \hat{x}_{i}(t_{i})\right]\left[x(t_{i}) - \hat{x}_{i}(t_{i})\right]^{2}\right], \quad (57)$$

Equation (57) can be written, in terms of the covariance matrices defined above,

$$\Gamma(t_{i}) = P_{i}(t_{i}) - P_{i}(t_{i}) - P_{i}(t_{i}) + P_{i}(t_{i}), \qquad (50)$$

By multiplying Equation (48) by its transpose and taking expected values of both sides, one immediately obtains

$$P_{i}(t_{1}) = \phi(t_{1}, t_{1,1})P_{i}(t_{1-1})\phi'(t_{1}, t_{1-1}) + 0(t_{1})Q(t_{1})O'(t_{1})$$
(59)

Multiplying Equation (52) by the transpose of Equation (45) and taking expected values yields

$$P_{a}(t_{1},t_{1-1}) = \phi_{a}(t_{1},t_{1-1})P_{a}(t_{1-1})\phi'(t_{1},t_{1-1}), \qquad (60)$$

The equation for  $P_g(t_i)$  is obtained by multiplying Equation (63) by  $x'(t_i)$ . It is given by

$$P_{g}(t_{j}) = [I - K_{g}H(t_{j})]P_{g}(t_{j}, t_{j-1}) + K_{g}H(t_{j})P_{j}(t_{j}) .$$
(61)

Multiplying Equation (52) by its transpose and taking expected values gives

$$P_{a}(t_{i}, t_{i-1}) = \phi_{a}(t_{i}, t_{i-1})P_{a}(t_{i+1})\phi_{a}'(t_{i}, t_{i-1}) , \qquad (62)$$

The equation for  $P_{q}(t_{1})$  is obtained by multiplying Equation (53) by its transpose and taking expected values. After some significant manipulation, one obtains

$$P_{3}(t_{1}) = P_{3}(t_{1}, t_{1-1}) + K_{8}(t_{1})H(t_{1})LP_{4}(t_{1}) - P_{9}(t_{1}, t_{1-1})J + + [P_{3}(t_{1}) - P_{3}(t_{1}, t_{1-1})]H'(t_{1})K_{8}'(t_{1}) + + K_{8}(t_{1})H(t_{1})[P_{3}(t_{1}, t_{1-1}) - P_{1}(t_{1})]H'(t_{1})K_{8}'(t_{1}) + + K_{8}(t_{1})K(t_{1})K_{8}'(t_{1}) .$$
(63)

This completes the derivation of the equations necessary to compute the covariance matrix of estimation errors.

# 4.1 Approximations to Mystem Dynamics

If earth's rotation rate is neglected ( $\omega = 0$ ), the system dynamics are considerably simplified.

The state transition matrix becomes

The result of this simplification is that the equations for  $\alpha$ ,  $\beta$ , and  $\gamma$  become uncoupled. If the drifts are neglected, a further simplification results. The equations describing  $\beta$  are given by



$$\begin{bmatrix} \beta(\mathbf{t}_{1}) \\ \mathbf{p}_{1}(\mathbf{t}_{1}) \\ \mathbf{n}_{1}(\mathbf{t}_{1}) \\ \mathbf{n}_{2}(\mathbf{t}_{1}) \end{bmatrix} = \begin{bmatrix} 1 & 0 & 0 & 0 \\ \mathbf{g}(\mathbf{t}_{1} - \mathbf{t}_{1-1}) & 1 & \phi_{2,3} & \phi_{1,1} \\ 0 & 0 & \phi_{2,3} & 0 \\ 0 & 0 & 0 & \phi_{1,1} \end{bmatrix} \begin{bmatrix} \beta(\mathbf{t}_{1-1}) \\ \mathbf{p}(\mathbf{t}_{1-1}) \\ \mathbf{n}_{1}(\mathbf{t}_{1-1}) \\ \mathbf{n}_{2}(\mathbf{t}_{1-1}) \end{bmatrix} + \begin{bmatrix} 0 & 0 \\ \mathbf{0}_{2,1} & \mathbf{0}_{2,2} \\ 1 & 0 \\ 0 & 1 \end{bmatrix} \begin{bmatrix} \mathbf{u}_{1}(\mathbf{t}_{1}) \\ \mathbf{u}_{2}(\mathbf{t}_{1}) \\ \mathbf{u}_{2}(\mathbf{t}_{1}) \end{bmatrix} ,$$
(65)

where  $\phi_{22} = \phi_{uu} = \exp[-(t_{i-1})/7]$ 

 $\phi_{ii} = \phi_{ii} \cos \omega_i t_i - \cos \omega_i t_{i-1}$ 

 $\phi_{iu} = \phi_{33} \quad \sin \omega_{b} t_{i} - \sin \omega_{b} t_{i-1}$  $G_{ii} = \cos \omega_{b} t_{i}$ 

 $a_{gg} = \sin \omega_n t_i$ .

The equations describing  $\gamma(t_i)$  are of almost the identical form with a sign difference.

The equation for  $\alpha$  is simply

$$\alpha(t_i) = \alpha(t_{i+1}) , \qquad (66)$$

The filter corresponding to these uncoupled dynamics is slap uncoupled. A summary of the suboptimal fourstate filter equations follows.

$$\begin{aligned} \hat{\pi}_{i}(t_{i}, t_{i-1}) &= \phi^{*}(t_{i}, t_{i-1}) \hat{\pi}_{i}(t_{i-1}) & (67) \\ \hat{\pi}_{i}(t_{i}, t_{i-1}) &= \phi^{*}(t_{i}, t_{i-1}) \hat{\pi}_{i}(t_{i-1}) & (68) \\ \hat{\pi}_{i}(t_{1}) &= \hat{\pi}_{i}(t_{i}, t_{i-1}) + K^{*}(t_{i}) [p_{i}(t_{i}) - H^{*}(t_{i}) \hat{\pi}_{i}(t_{i}, t_{i-1})] & (69) \\ \hat{\pi}_{p}(t_{i}) &= \hat{\pi}_{i}(t_{i}, t_{i-1}) + K^{*}(t_{i}) [-p_{i}(t_{i}) - H^{*}(t_{i}) \hat{\pi}_{i}(t_{i}, t_{i-1})] & (70) \\ P^{*}(t_{i}, t_{i-1}) &= \phi^{*}(t_{i}, t_{i-1}) P^{*}(t_{i-1}) \phi^{*}(t_{i}, t_{i-1}) + a^{*}(t_{i}) a^{*}(t_{i}) & (71) \end{aligned}$$

$$\mathbf{P}^{*}(\mathbf{t}_{i}) = \mathbf{P}^{*}(\mathbf{t}_{i}, \mathbf{t}_{i-1})\mathbf{H}^{*}(\mathbf{t}_{i})[\mathbf{H}^{*}(\mathbf{t}_{i})\mathbf{P}^{*}(\mathbf{t}_{i}, \mathbf{t}_{i-1})\mathbf{H}^{*}(\mathbf{t}_{i}) + \mathbf{R}^{*}(\mathbf{t}_{i})]^{-1}$$
(72)

$$P^{+}(t_{i}) = [I - K^{+}(t_{i})H^{+}(t_{i})]P^{+}(t_{i}, t_{i-1}), \qquad (73)$$

where

$$b^{*}(t_{1}, t_{1-1}) = \begin{bmatrix} 0 & 0 & 0 & 0 \\ g(t_{1} - t_{1-1}) & 1 & \phi_{12} & \phi_{14} \\ 0 & 0 & \phi_{12} & 0 \\ 0 & 0 & 0 & \phi_{14} \end{bmatrix}$$

$$a^{*}(t_{1}) = \begin{bmatrix} 0 & 0 \\ a_{31} & a_{32} \\ 1 & 0 \\ 0 & 1 \end{bmatrix}, \qquad q^{*}(t_{1}) = \sigma_{V}^{g}[1 - e^{-s(t_{1} - t_{1-1})/r}] I$$

$$\hat{g}^{1}(t_{1}) = \begin{bmatrix} \hat{\beta}(t_{1}) \\ \hat{\beta}_{1}(t_{1}) \\ \hat{\beta}_{1}(t_{1}) \\ \hat{\beta}_{2}(t_{1}) \end{bmatrix}, \qquad z_{2}(t_{1}) = \begin{bmatrix} \hat{\gamma}(t_{1}) \\ -\hat{\beta}_{2}(t_{1}) \\ \hat{\beta}_{3}(t_{1}) \\ \hat{\beta}_{4}(t_{2}) \end{bmatrix}$$

$$H^{*}(t_{1}) = [0 - 1 - 0 - 0], \qquad B^{*}(t_{1}) = \phi^{2}/13,$$

# 4.3 Sine Errors in the Suboptiuml Filter

The approximations to the system dynamics can cause a bias to appear in the suboptimal estimate, and no direct method has yet appeared to determine this bias as readily as the determination of the covariance satimate of the suboptimal errors given by Equation (58). In general, a bias will occur if the non-noise associated part of the dynamic equations is approximated, or if the measurement equations are changed.

1.2.4.4.4.4

۲. من المراجع مع المراجع المراج المراجع A LINANA AND

In the simplifications discussed in Section 4.2, the variation in time of the angles  $\alpha$ ,  $\beta$ ,  $\gamma$  caused by the earth's cross-coupling is neglected. This causes a bias to occur in the estimates made by the simplified filter.

An approximate expression for the bias can be calculated by neglecting sway noise. In this case, the angle is varying essentially as a linear function of time due to earth's rate cross-couplings. However, the filter neglects time variation of the angle and assumes the angle to be a constant. It can be shown that the error in the estimate which minimizes the mean-aquare error is equal to one-half the change in angle over the measurement time. This fact can be used to write expressions for the measured angles in terms of earth's rate, the geometry of the optical measurement of azimuth and the actual angles. Then equations can be solved to yield a better estimate of the actual angles. Details are shown in Reference 2.

# **B. RESULTS SUMMARY**

Figures 2 through 9 show the results of the analysis of the Kalman filter. These results are obtained from computer simulation of the appropriate equations. Figure 2 is the 10 section error for the optimal 12-state filter. This filter is described by Equations (47)-(51). The conditions used for this baseline case are given by

Initial integrated velocity uncertainty	σ	×	0
Initial eraction uncertainty	o,	2	1/2 degree
Gyro drift uncertainty	$\sigma_{d}$	×	0.1 meru (1 meru = 0.015 <sup>0</sup> /hr)
R.M.S. sway velocity	σy	=	0.5 m/s
Accelerometer quantization level	Δ	z	0.02 m/s
Center frequency of sway velocity power spectrum	fn	=	0.25 Hz
Correlation time (reciprocal of bandwidth)			
of away velocity power spectrum	7	z	200 =
Eampling rate	1 1	ي ا	ple/m .

Figure 3 shows the performance of the suboptimal 4-state filter for various initial erection angles. This filter is described by Equations (67)-(73). The main simplification employed in reducing the 12-state optimal filter to two identical 4-state filters was the neglect of earth's rate cross-coupling terms in the system dynamics used in the filter equations. In addition, gyro drifts were neglected in the simplified dynamics. The difference in errors for various initial angles is attributable to the earth's rate coupling. For initial angles less than 0.5 degrees the increase in error over the optimal filter is less than 1 second of arc at 60 seconds time. If the initial angle uncertainty is large, the 12-state optimal filter performance can be approached by updating the computer coordinate system after about 20 seconds to reduce the cross-coupling effects. Another approach to removing the cross-coupling terms is to use the correction matrix derived in Reference 2.

Figure 4 shows the degradation in performance due to the neglect of gyro drift. In Figure 4, the gyro drift numbers shown as parameters are the 1 $\sigma$  uncertainties in the true gyro drift. The filter equations assume zero gyro drift. It is seen that the drift uncertainty must be quite large (20 meru = 0.3 deg/hr) before the filter performance is degraded appreciably.

Figure 5 shows the erection error versus time for various Lampling times. A sampling time of 1 second seems to provide a good trade-off between computational requirements and erection accouracy. The worst case conurs when the sampling time is 4 seconds. This is because the center frequency of the away velocity power spectrum is 0.25 berts. Since the power spectrum is very marrow, the away velocity is almost sinusoidal, with a period of 4 seconds. The sampling is thus at the same part of the sine wave, and it becomes quite difficult to filter out the noise.

The correlation time (reciprocal of the bandwidth) of the narrow band noise has a significant effect on the estimation accuracy. This is demonstrated in Figure 6, where the correlation time assumed in the filter equations matches the correlation time of the actual noise. It was found that the filter was insensitive to errors in the assumed correlation time. When the assumed correlation time was 20 seconds rather than the degradation in filter performance was negligible. The important parameter is the actual noise of the assumed for the filter calculations.

Since the frequency of the missile away may not be known exactly, it is of interest to determine the sensitivity of the filter to that parameter. This is shown in Figure 7. In this case, the actual frequency is 0.25 hertz, while the number used in the filter calculations is 0.2 hertz. It is meen that the accuracy of the estimate is somewhat poorer, but still acceptable. The performance for an assumed frequency of 0.3 hertz is almost identical to the curve shown in Figure 7 and, therefore, has not been presented.

An alternative method of erecting the analytic coordinate system is to use a least-squares filter. The computational requirements of this method are much less than those of the optimal filtering technique. The estimate, of gourse, takes longer to converge to within acceptable limits. Figure 8 compares the performance of the least squares and the optimal 4-state filters of the case when  $\tau = 200$  s. The optimal filter is significantly better than the least-squares filter for this case. Figures 8 and 10 show the same comparison for  $\tau = 20$  s and 5 s respectively. It is seen that, as  $\tau$  decreases, the least-squares filter performance approaches that of the optimal filter. This is to be expected since, for white noise  $(\tau = 0)$ , the least-squares filter is optimal.



ς.,

Ę,















The preceding curves were calculated by determining the covariance matrices for the estimatic: errors. In addition, a simulation of the filter equations with input noise derived from a random number generator was performed. Figure 11 shows the results of one run of this simulation with the sway velocity noise spectrum lying in a narrow band around a center frequency. The  $i\sigma$  value obtained from the covariance matrix calculation is shown for comparison. Figure 12 shows the results of a run in which the input noise is assumed to consist of three narrow band uncess at different center frequencies. The filter still assumes that the noise is centered in a narrow band around a single frequency. The three-frequency modes is considered to be a more realistic representation of the velocity of a swaying minsile with several bending modes on the launch pad. The center frequencies used were 1.82 rad/s, 2.14 rad/s, and 2.32 rad/s, with a correlation time of 200 seconds for each band. The total runs, away velocity was 0.5 m/s.

Certain "rules of thumh" which have been developed to allow extrapolation of the results given in this chapter to different noise characteristics and filtering times are as follows:

- (a) The t.m.s. filtering error is proportional to the r.m.s. value of the sway velocity noise.
- (b) If earth's rate cross-couplings are removed, by updating or by using the correction matrix, the filtering error is not a strong function of the initial angle error.
- (c) The filtering error is approximately inversely proportional to the noise center frequency.
- (d) For short correlation times, the filtering error for both the optimal 4-state and least-squares filter is approximately inversely proportional to the three-halves power of the filtering time (Error = K/T<sup>3/8</sup>). This is shown by plotting the results of Figure 9 on log-log paper in Figure 13. For longer correlation times, the least-squares filter has a filtering error which is more nearly inversely proportional to the square of the filtering time, while the 4-state optimal remains inversely proportional to the square of the filtering time. This is shown by plotting the results of Figure 9 on log-log paper in Figure 13. For longer correlation times, the least-squares filter has a filtering error which is more nearly inversely proportional to the square of the filtering time. This is shown by plotting the results of Figure 8 on log-log paper in Figure 14. It must be resedered, however, that the curves for the two filters can never intersect, so that the least-squares filter error must approach being inversely proportional to the three-halves power of filtering time as the time increases.
- (e) The filtering error for the optimal 4-state filter is approximately inversely proportional to the square root of the sway velocity correlation time (Error =  $K/\tau^{1/2}$ ). The least-squares filter error only approaches being inversely proportional to the square root of the correlation time at low correlation times, however. For higher correlation time the improvement in performance with increasing correlation time is much leas then that of the optimal 4-state filter. This is shown by plotting the results of Figures 8, 9 and 10 at 60 seconds time on log-log paper in Figure 15. This rule obviously cannot be used as  $\tau$  approaches mero, but it does hold at least down to  $\tau = 1$  s.

# ACKNOWLEDGEMENT

This chapter is based on work performed under Contract No.NAM-8-21273 to the NASA George C.Marshall Space Flight Center, July 1968. Principal investigators were Earle B.Crocker, Leonard Rabins, and Kovin A.Clements.

### REFERENCES

1. Kennel, Hans F.

Initial Alignment of a Strapdown Inertial Reference and Navigation System. AIAA Paper 67-555. NASA Marshall Space Flight Center, August 1967.

2. -

Final Report - Optimum Digital Filter Study for Straudown Inertial Navigation System Initial Alignment. Ordnance Systems, General Electric Company, Pittsfield, Massachusetts, July 8, 1968.



-











al lancourse because traded weeker without bearing





States S



104

All Mary

Longon L

AND PROVIDENT

Sinte a line





Fig. 12 Error when sway velocity noise has 3 narrow band components (4-state optimal filter)



inter an and the second



li }



Fig.14 Dependence of filtering error on filtering time (7 = 200 econds)



K

12

U







1

E an the Windowski Street

# A KALMAN FILTER AUGMENTED MARINE NAVIGATION SYSTEM

# by

H. Halamandaris and D. Ozdes

# Estellite Positioning Corporation 18033 Ventura Blvd Encino, California 91316, USA

•

.

513









arrows O the second Oracles



and the second second second

**See 1** 

Ű

# NOTATION

P <sub>1</sub> , T <sub>1</sub>	satellite positions and time at the i <sup>th</sup> interval
t <sub>s</sub>	ground referance oscillator frequency
ft	stable frequency transmitted from satellite
t <sub>r</sub>	user received frequency
Δ£	frequency difference (satellite)
N <sub>i,j</sub>	Doppler count for satellite interval i , j
t	time
ti	time at satellite i <sup>th</sup> interval
Δt	incremental time
s <sub>i</sub>	slant range at 1 <sup>th</sup> interval
∆t <sub>i,j</sub>	incremental slant range in i , j interval
C	mpeed of light
ć,	speed of sound in water
J <sub>1.1</sub>	gravity harmonic (i) and order (j)
k	frequency scale constant
δ <sub>L</sub> , δ <sub>H</sub>	150 Miz and 400 MHz frequency distortions
γ	sonst Doppler beam transmitting angle
∆ <b>∂</b>	steady state attitude error
$\Delta \phi$	dynamic attitude error
٥v	velocity error
T	temperature (water)
5	salinity
D	depth of transducer
X	state variable vector
A	system dynamics matrix
α.	state poise vector
<b>•</b> .	state transition matrix
Ŧ	system noise
¥:	velocity components along ith axis
τ <sub>i</sub>	coordinate transformation matrix
R <sub>N</sub> , R <sub>R</sub>	earth radii
81	perturbation quantities
A <u>i</u>	ith axis acceleration
Ω	earth rate
λ	latitude
•	



10.0

٨	longitude
6	gravity
¢	gyro drift
ψ <sub>α</sub> , η, α	internal gyro angles
τ <sub>α</sub>	gyro time constant
K D	gyro damping gain
$\Delta t_i$	sonar Doppler frequency difference along i <sup>th</sup> axis
fi	i <sup>th</sup> axis frequency for monar Doppler
θ	ship's pitch angle
φ	ahip's roll angle
$\psi$	ship's heading
δ <sub>ij</sub>	Kronecker delta
Q(1)	variance of white noise
σi	variance of stochastic process
$\beta_{i}$	reciprocal of stochastic process correlation time
P	covariance matrix
K <sub>k</sub>	Kalman gain
R'	noise covariance matrix
Ŷ	estimate of state vector
Z, 2	observation and estimate of observation matrix
H	mensurement matrix
л, р	range and range-rate vectors
r. 3. £	unit vectors slong $\mathbf{x}$ , $\mathbf{y}$ and $\mathbf{z}$ axes respectively
τ <sub>i</sub>	noordinate transformation metrix
S <sub>1.1</sub> S <sub>2,1</sub>	satellite clock errors

Subscripts	
ŧ	porth
6	east
)	down
1	μφ. ·
έτ.	sem track
ia.	ses current
1	starboard axis
2	fore/aft axis
•	aft alis
•	port axis -

D): 

H+ 11 1

# Superscripts

T	transpose
1 4	reference coordinate frames
E	Past
n	north
•	time differentiation

• •

.

. .

.

• •







Laura Charman Char

Personal Will a section William

# A KALWAN FILTER AUGMENTED MARINE NAVIGATION SYSTEM

### H. Halamandaris and D. Ozdes

# 1. INTRODUCTION

With the technical horizons expanding very rapidly in recent years, new demands have been imposed on the performance indices of marine navigation systems. Typical indices are performance accuracy, reliability, maintainability, cost, etc. Reliability and maintainability are enhanced by utilizing proven technology and hardware while retaining a simple system physically. Performance accuracy of hardware commensurate with the reliability and muintainability constraints can be improved by completely integrating the navigation sensors via a digital interface with primary control in a general purpose computer with the controller being a Kalman filter.

The system described and analyzed herein employs a classical dead-reckoning navigation system to provide instantaneous position. Characteristic of all dead-reckoning systems is the monotonically growing error function; bence a navigation sid is required to constrain the long-term positional stability of the system. The Navy Navigation Satellite System (NNSS) provides an independent measure of positional information, and this information is used to optimally reset and calibrate the dead-reckoning system which contains

(i) A Doppler sonar to measure ship's velocity.

(ii) A gyrocompass to provide a measure of ship's heading.

(iii) A velocimeter to provide a measure of the speed of sound in water.

(iv) An inclinemeter to provide a measure of ship's pitch and roll.

A general purpose digital computer in which all the mechanization equations are implemented is the central controller.

Section 2 contains a description of the operation for each of the four basic sensors and the NNSS. A generalization from the basic implementation to the format compatible with a Kalman filter implementation is presented in Section 3. Also contained is a development of the pertinent filter mechanization equations and real-time mechanization constraints. A performance summary for various environmental and dynamic conditions is presented in Section 4, which illustrates the femsibility of utilizing a Kalman filter to improve the integrated mystem performance accuracy.

# 2. SYSTEM DESCRIPTION

### 2.1 Integrated System Concepts

The primary objective for the integrated system as configured is to automatically provide continuous, accurate, real-time geodetic position while in no way compromising system operation. As a result, an improved selfcontained, shipborne navigation system is synthesized, employing several discrete navigation subsystems. Figure 1 is a functional block diagram of the system which contains the basic constituents in solid blocks and optional mavigational systems are shown in the dashed blocks. Instantaneous position velocity and heading are derived from the outputs of the Doppler sonar, gyrocompass and velocimeter with long-term position accuracy stability preserved via the satellite fixes. The nucleus of the system is the interface unit/computer complex. Within the computer/interface all signal transmission, conditioning, sixing, and optimal filtering is implemented.

The computer receives ship's fore/aft and port/starboard velocity (frequency) components from the Doppler sonar and corrects them for speed of sound anomalies derived from the velocimeter data. The component velocities are corrected for the ship's pitch and roll by the vertical reference unit and are resolved about the gyrocompass heading input to produce ship's North and East velocity components. A numerical integration is continuously performed to compute ship's latitude and longitude. An independent absolute position is calculated at the end of every satellite pass, and a system position update is automatically obtained. This update is derived using the optimal (Kalman) filter and the satellite position observation data.

The optional features which can be incorporated in the system to allow complete system automation are as follows:

Ship's steering commands can be generated in the computer and sent to the autopilot.

Point-to-point navigation options can be provided.

X

1

<u>ا</u>

Automatic timing sequence controls as a function of distance traveled along course or olapsed time are generated for-data acquisition systems.

Real-time data logging for post-mission analysis can be implemented through the use of a magnetic tape recorder.

Ship's track may be plotted through the use of an incremental plotter.

Each primary sensor's operation and theory is delineated below.

### 2.2 System Description and Theory of Operation

# 2.2.1 Navy Navigation Satellite System

The Navy Navigation Satellite (NNSS) is a world-wide, all-weather system from which accurate navigational position files can be obtained from the data transmitted from the orbiting satellite. The NNSS consists of four near-earth orbiting satellites, four tracking stations, two injection stations, the US Naval Observatory, and a computing center. Any number of shipboard navigational installations, such as that aummarized above, can exist.

The navigation satellites are placed in a circular polar orbit at an altitude of approximately 600 nautical miles. The orbital places of the satellites have a common point along the warth's rotational axis. The geometrical placement of the orbiting matellites allows an earth bound observer to cross directly under the satellite twice daily. The observer receives data from the satellite twice each time he is near the orbit because the satellites appear to traverse longitudinally as the earth rotates. A maximum of 16 fixes is possible at the equator. Realistically, about 12 fixes daily can be realized. The number of fixes available varies as a function of user latitude, as can be seen in Figure 2.

Each satellite orbits the earth in approximately 108 minutes, which is 90 minutes between passes plus roughly 18 minutes of tracking. Through its useful life, each satellite continuously transmits the following phasemodulated data as two-minute messages on two radio frequency carriers: two-minute mark synchronization signals, a 400 Hz reference signal, fixed and variable parameters describing the satellite's polar orbit.

The classical utilization of the satellite data to compute a position fix is similar in concept to any hyperbolic positioning system where the satellite simulates the sultiple transmitting stations by its inherent motion relative to the user. Figure 3 illustrates the relative geometry of the user and the satellite.  $P_1$ ,  $P_2$  and  $P_3$  where the distinct positions of the satellite during one orbital pass. If each of the  $P_1$ 's differ by two minutes in time, the distance between points is about 960 km. This separation is comparable to baselines on the hyperbolic radio positioning systems. The earthbound navigator lies on the surface of a hyperboloid generated from the satellite information from any two successive points. The intersection of two or more hyperboloids uniquely locates the earthbound user's position.

Additional data points can be used to establish the user's position more accurately. A maximum of 9 data points ( $P_1$ ) are available during a satellite pass; however, many factors such as geometry (low elevation), radio interference, etc., can cause data to be missed or be invalid.

In determining the  $P_i$ 's and establishing the user's position, sutellite orbital data is required, as well as some measure of relative distance between user and satellite. The satellite transmits both fixed and variable orbital parameters.

The fixed parameters describe the satellite's nominal orbits and are accurate only for a 12 to 16 hour interval. The variable parameters describe the fine structure in the satellite's nominal orbit as a function of time and are correct only for the time at which they are transmitted by the satellite. Thus the satellite's memory stores sufficient variable parameters to describe its orbit at two-minute intervals between subsequent injections of data into its memory. All data transmitted which do not change are permanently wired into the satellite's memory.

The satellite transmits a stable fraquency  $f_{\rm c}$  which is received by the user and designated as  $f_{\rm p}$ . The user in turn compares the received frequency to a stable oscillator output frequency  $f_{\rm g}$  to produce a frequency difference

# $\Delta \mathbf{f} = \mathbf{f}_{\mathbf{g}} - \mathbf{f}_{\mathbf{r}} \ .$

Figure 4 illustrates the relationship between the frequencies. Over each two-minute time interval, the number of beat cycles are counted or, essentially, an integration of the beat cycles is accomplished.

This information can be directly related to changes in slant range from the satellite to the user in that one frequency count is equivalent to one wavelength of distance traveled by the satellite. Reception of three twominute satellite messages during an orbital pass, along with the related Doppler shift measured by the user's receiver, defines hyperbolic comes of differential slant range, intersecting at the user's ship's position. Taking date over a minimum of three intervals yields two equations in the two unknowns, and hence a position solution can be realized.



















From Figure 4, and using the first two intervals,

$$\mathbf{R}_{1,2} = \int_{t_1+\Delta t_1}^{t_2+\Delta t_2} (f_2 - f_1) \, dt = (f_2 - f_1)(t_2 - t_1) + f_2(\Delta t_2 - \Delta t_1), \qquad (2.1)$$

using the assumption that  $f_t = f_r$ . Also  $\Delta t_g - \Delta t_1$  is related to the slant range between satellite and receiver at  $P_1$  and  $P_g$ .

$$\Delta t_{g} - \Delta t_{1} = \frac{1}{C} \langle \mathbf{s}_{g} - \mathbf{s}_{1} \rangle , \qquad (2.2)$$

where C is the speed of light, and  $S_2$  and  $S_1$  are the slant ranges from the user to the satellite at  $P_2$ and  $P_1$ , respectively. Substituting Equation (2.2) into Equation (2.1) and solving for the difference in distance from the earth-bound observer to two satellite positions, in effect, determines the baseline, e.g.,

$$\Delta \mathbf{S}_{1,2} = \mathbf{S}_2 - \mathbf{S}_1 = \frac{C}{f_0} N_{1,2} - \left(1 - \frac{f_1}{f_0}\right) C \left(t_2 - t_1\right) . \qquad (2.3)$$

Similarly,

the second

Sec. Same

1.11.1.1.1.1.1

$$\Delta \mathbf{S}_{2,3} = \mathbf{S}_3 - \mathbf{S}_2 = \frac{C}{f_0} \mathbf{N}_{2,3} - \left(1 - \frac{f_1}{f_0}\right) C (t_3 - t_2) . \qquad (2.4)$$

Noting that,  $t_{i+1} - t_i$ , where  $i = i, i, \dots n$ , multiplied by C is the distance traveled by a particle of light in two minutes allows some simplifying assumptions to be made. However,  $f_{\rm g} = f_{\rm t} + \Delta f$  and  $\Delta f$  is very small compared to  $f_{\rm t}$ ; thus, to preserve computational accuracy, the above equations must be rearranged to avoid computations of the form  $\Delta f/f_{\rm g}$ . This can be easily accomplianed by using an additional point. Manipulating a third equation which is similar to the above yields

$$-S_{1} + 2S_{2} - S_{3} = \frac{C}{f_{0}} (N_{1,2} - N_{2,3})$$
 (2.5)

$$-\delta_{g} + 2\delta_{g} - \delta_{u} = \frac{\sigma}{r_{0}} \left( N_{g, g} - N_{g, u} \right) , \qquad (2.6)$$

where the  $S_1$  are functions of the satellite position at each point i, where i = 1, 2, ... n, and the observer position. Hence, two squations in two unknowns allow the determination of the user's two unknown coordinates, latitude and longitude. An alternate implementation is to use a direct ranging approach which obviates the requirement to compute position from the slant range data. This approach tends to preserve accuracy, as the equivalent of GDOP (Geometric Dilution of Positica) is not experienced.

The position determination accuracy was seen to be greatly affected by the earth's atmosphere and gravity. Gravity model refinements have continued since 1960. In 1964 the eighth harmonic and order term  $(J_{0,0})$  in the gravity expansion model was incorporated, and on 8 January 1968, the terms  $J_{15,13}$ ,  $J_{14,14}$ ,  $J_{15,14}$  and  $J_{15,13}$ ware added to the mechanization; a substantial reduction in position detormination innecuracy resulted. The information carried in the mavigation measages, which are phase-modulated, is distorted by the troposphere and ionosphere. Thus, the earth's atmospheric effects were determined to be functions of frequency. The tropospheric refraction is proportional to frequency, as is the Doppler shift which contains the usable information, and can not be isolated. However, the ionospheric refraction was observed to be inversely proportional to frequency and could easily be determined. To enhance the system performance, the satellites transmit two coherent carrier frequencies (150 MHz and 400 MHz). These two frequencies are controlled by a reference oacilator and one frequency is an exact multiple (3/8) of the other. The actual transmission frequencies are 400 MHz offset by 32 KHz (399.968 MHz) and 150 KHz offset by 12 KHz (189.888 MHz). This allows a more precise Doppler count. Let  $f_{\rm H}$  and  $f_{\rm L}$  be the respective high and low frequencies transmitted by the satellite; then

$$t_{\mu} = k t_{L}$$
 (2.7)

where k = 5/3. The frequencies received on the high and low channels respectively are

$$f_{\mathbf{R},\mathbf{H}} = f_{\mathbf{H}} + \Delta f_{\mathbf{H}} + \delta_{\mathbf{H}} \tag{2.8}$$

$$f_{\mathbf{R},\mathbf{L}} = f_{\mathbf{L}} + \Delta f_{\mathbf{L}} + \sigma_{\mathbf{L}} . \tag{2.9}$$

Imploying the observed results of frequency distortion allows

$$\mathbf{f}_{\mathbf{R},\mathbf{R}} = \mathbf{k}\mathbf{f}_{\mathbf{L}} + \mathbf{k}\Delta\mathbf{f}_{\mathbf{L}} + \frac{1}{\mathbf{k}\delta_{\mathbf{L}}}.$$
 (2.10)









Solving for  $\bar{a}_{\mu}$  from Equations (2.8) and (2.9) yields

$$\delta_{\rm H} = \frac{1}{1-k^2} \left( f_{\rm R,\,H} - k f_{\rm R,\,L} \right) \,, \qquad (2.$$

### which is the Doppler refraction correction.

Since the satellite transmits circularly polarized mignals, either circularly or linearly, polarized receiving antennas can be used. However, if highly directional, narrow-band antennas are used, two antennas are required. one for each operating frequency, or a single broadband antenna designed for dual frequency operation may be used. The antennas used are larger and more difficult to handle than the simplest antenna arrungement which is composed of two whip antennas, one for each frequency; however, this should be avoided, if possible, since the whip has an overhead null, and satellite data would, therefore, be lost for satellite passes of more than 70° elevation angle. The elevation angle limits on satellite coverage do exceed this value.

The satellite signals are received by the user from the time the approaching satellite rises above the radio horimon until the time it sets. The receiving antenna patterns should, therefore, provide complete overhead COVERES.

# 2.3 Velocity Measuring Subsystems

# 2.3.1 Somar Doppler

The monar Duppler provides a precise measurement of ship's fore/aft, port/starboard, and up/down velocity relative to the earth for water depths up to 600 feet. The sonar Doppler is a two-axis, bottom referenced speedometer which derives its information from high froquency somer signals emitted from a transmitter on the boat. The basic operation is delineated below.

The Doppler system employs beams of ultrasonic energy, directed obliquely at the ocean floor at an angle of y degrees relative to the vertical, to obtain true velocity measurements in the fore-aft and port-starboard directions and the up-down axis. Because the pitch, roll and heave of the ship add other apparent motions relative to the ocean bottom, the Doppler system should employ pairs of bears (one pair angled fore and sit, the other angled to port and starboard, see Figure 5) to componente for these motions. By averaging the frequency difference between fore and aft, and port and starboard, the true velocities are determined. The concept of four beams is an application of the Janus configuration which is common in airborne Doppler radars. A primary peagon for using this configuration is for the cancellation of errors which results from averaging the return Signals in the signal processing.

Typical sonar Doppler sensors utilize a 300 KHz sonar signal in a pulsed transmission mode of operation. Signal feedthrough from transmitter to receiver is eliminated and excellent signal-to-noise ratio characteristics are obtained with receiving circuit gating and time varied gain. The pulsed operating mode provides a signal amplitude sufficient to supply accurate signal data at vessel over-the-bottom operating heights up to 600 feet. Beyond 600 to 650 feet the system automatically locks on to the water solatter return; the sonar energy is reflected from small scatterers present in all oceans. The electronic gating is set to observe this reverberation return only from the undisturbed water below the ship, thus excluding the effects of turbulence and boundary layers. The result is a highly accurate and reliable indication of the ship's true velocity relative to the ocean bottom up to 600 feet and relative to the water mass at all greater depths.

Thus, the water mass return represents webicle motion relative to the sea and the expected accuracy is degraded by the unknown sea currents which are typically on the order of 0.5 to 1.0 knots for short-term correlated noise. The sonar accuracies are tabulated below.

Water Depth (ft)	Mode	Typical (10) Accuracy (%)
1 - 400	Bottom	. 0. 2
<b>400 - 60</b> 0	Bottom	0.5
Greater than 600	Volume reverberation	. 9. <b>5</b> *

An error source which is additive to the one above is the receiver/transmitter electronics noise, which is approximately 0.02 knots. The primary contributor is the crystal controlled oscillator which is stable to within 0.01%. The biggest concern in sonar performance is that in the reverberation mode the sea current motion must be statistically added to the sonar error. Tyrical random sea current drifts are on the order of 0.5 knots to 1.0 mots, depending on the sies, with typical correlation distances of 18-20 neutical miles, assuming a Gaussian distribution with exponential autocorrelation function.

Measured relative to sea currents.



11)

Major contributing error sources in a conventional unaugmented Doppler mechanization, which are eliminated in the integrated mechanization, are the errors induced by the vehicle pitch and roll dynamics. The pitch and roll angular resolution will be mechanized and will yield negligible errors. If the resolution is not implemented, the following velocity errors per axis result:

(i) Due to stendy state pitch (trim) and soll (list),

 $\delta V \simeq \Delta \theta^2 / 6570$ ,

where  $\Delta \theta$  is the pitch or roll angle in degrees.

(ii) Due to dynamic pitch and roll (sinusoidal)

# $\delta V \simeq 0.101 \Delta \phi^2/6570$ ,

where  $\Delta \phi$  is the amplitude of the pitching or rolling motion in degrees.

# 2.3.2 Vertical Reference Unit

The vertical reference unit measures ship's pitch and roll through which the sonar Doppler outputs are resolved to essentially create a stabilized, locally-level coordinate frame in which the system velocities are expressed. The basic sensors in the vertical reference unit are two single-axis fluid-damped pendulums (inclinemeters) which output the deflection of the ship from the local vertical. Their outputs are whole angle indications of ship's pitch and roll. These angles are then used to generate the Kuler transformation matrix consistent with the rotational dynamics of the ship.

# 2.3.3 Velocimeter

DEDENT RECENCE RELEVE BREAKS BUILDE

The velocimeter is a precision instrument for measuring the velocity of sound in water ... an accuracy of one part in ten thousand. Its output is used to weight the sonar Doppler output pulses.

The operation of the velocimeter is based upon the sing-around principle. A pair of piezoelectric ceramic transducers and two reflectors are mounted to form a sound path of fixed length in the water. In operation, a pulse of acoustic energy is transmitted through the water, received, amplified and used to generate another pulse of acoustic energy. The repetition frequency of this regenerative action is dependent upon the transit time of the signal pulse and is therefore a measure of the propagation velocity. Errors resulting from the flow of water along the sound path length are winimized by folding the sound path.

The basic parameters which cause a variation in the speed of sound are water temperature, pressure and salinity. A typical expression for C. is

$$C_{1} = 4422 + 11.257 - 9.0457^{2} + 0.01524 + 4.3(5 - 34)$$

where  $G_{y}$  is in ft/sec. T is temperature in degrees Fahrenheit, d is depth in feet and S is salinity in parts per thousand. The primary items of concern in this expression are as follows:

(a) Temperature - a 20 - 25 degree F variation can be expected between low latitudes to polar areas.

(b) Salinity - large variations can occur if the vehicle is in combining salt water currents, or near river outlets.

### 2.3.4 Gyrocompany

The gyrocompass, with its integral gyroscope, is controlled such that it continually aligns itself with the meridian and tracks true north. Hence, its natural output is ship's heading. Sonar Doppler velocity components in the stabilized, locally-level plane are resolved about heading to provide ship's velocity components in north and east components. Figure 1 indicates a feedback torquing signal from the computer to compensate for the ship's instantaneous position and spatial rate. In the gyrocompass error model which is presented in the following section, all gyro drifts are assumed to be either deterministic or to possess exponentially autocorrelated noise functions.

# 3. SYSTEM ANALYTICAL MODEL

### 3.1 Introduction

The integrated multi-sensor marine navigation system defined in Section 2 employs optimal signal processing and multiply-redundant navigation dats, in the form of range and/or range-rate or range-difference and/or range-rate-difference, in synthesising minimum-variance state vector estimates.

The state variables of the state vector consist of position, velocity, heading and the stochastic models of the sensor error sources. The optimal estimate of the navigation variables are employed in updating (i.e., servoing or correcting) the navigation system; the prediction of the sensor error characteristics leads to cross-calibration of sensors; thus the mavigation performance is further enhanced.



(2.12)

The states of the second

ant Keng mangang tan

The physical dynamics involved in optimally implementing the multisensor navigation system require a precise definition of the propagation of linearized error differential equations characteristic to the Doppler-Sonar gyrocompass system augmented by the satellite data processor.

In the optimum linear continuous estimators the system error state vector is defined by the relationship

$$\dot{\mathbf{X}} = \mathbf{A}\mathbf{\ddot{X}} + \mathbf{U}, \qquad (3, 1)$$

where X is the n-dimensional error state vector, A is an  $n \times n$  coefficient matrix and U is state noise vector. The corresponding discrete equation is

$$\tilde{X}_{k+1} = [\Phi]_{k+1,k} [X]_{k} + [W]_{k},$$
 (3.2)

where  $\Phi$  is the well known state transition matrix characterizing the error transition from  $k^{th}$  to  $(k+1)^{th}$  iteration.

# 3.2 System Sensor Error Models

The primary system sensor (i.e., Doppler-Sonar (D/S), gyrocompass and satellite receiver) error mode's are usually obtained by a first-order perturbation of mechanisation equations.

The (D/S) mechanisation equations are

$$[\overline{\mathbf{v}}]_{\mathbf{z}} = [\mathbf{T}_{\mathbf{z}}][\mathbf{v}''] , \qquad (3.3)$$

where  $(\nabla_g)^T = [\nabla_g \nabla_g \nabla_g]$  is the velocity vector defined in the ship's static coordinate system,  $(\nabla_g')^T = [\nabla_g' \nabla_g' \nabla_g]$  is the velocity vector defined in the ship's dynamic (i.e., pitched and rolled) coordinate system<sup>\*</sup> and the matrix  $[T_i]$  provides transformation from ship's dynamic to state coordinate systems.

The ship's velocities in the navigation coordinate system are obtained by

ſ

$$\left[\nabla\right]_{\mathbf{n}\mathbf{T}} = \left[\mathbf{T}_{\mathbf{n}}\right]\left[\nabla_{\mathbf{n}}\right], \qquad (3.4)$$

where  $[V]_{g_T}^{*} = [v_g v_h v_u]$  is the ship's sea-track velocity vector defined in a (east, north, up) coordinate system sharing the same origin of the ship's coordinate systems. The matrix  $[T_g]$  provides transformation from ship's static to navigation coordinate systems. The ship's velocities are resolved through the sea-track heading sample  $\psi_{g_T}$  to mortherly and easterly components.

Combining Equations (3.3) and (3.4) yields

$$[V_{a1}] = [T_{3}][V_{a}'']$$
, (3.5)

$$[T_{*}] = [T_{*}][T_{*}],$$
 (3.6)

However, if the D/S system is operated in the volume reverberation mode, a sea current vector,  $[V]_{do}^T = [V_{Bd}^H v_{Bd}^H 0]$ , must' be subtracted from Equation (3.5). Hence

$$[\overline{\mathbf{V}}]_{\mathbf{a}\mathbf{T}} = [\mathbf{T}_{\mathbf{a}}][\mathbf{V}_{\mathbf{a}}^{\prime\prime}] - [\overline{\mathbf{V}}]_{\mathbf{a}\mathbf{a}}\mathbf{K}$$
(8.7)

where K = i (i=0, i) in the bottom and sea volume reverberation modes, respectively.

δλ = <u>δV</u>H

The latitude and longitude rates of the system are obtained by dividing sea-tilck velocity components by the radii of curvature of the earth. First-order perturbation of these rates, neglecting second-order terms

$$\frac{\delta R_{\rm H}}{R_{\rm H}^2} \simeq \frac{\delta R_{\rm H}}{R_{\rm H}^2} \simeq 0$$
$$\delta R_{\rm H} \simeq \delta R_{\rm H} \simeq 0$$

yields

where

$$\delta \Lambda = \frac{\delta V_{\rm H}}{R_{\rm H}} + \frac{\delta \lambda V_{\rm H} \sin \lambda}{R_{\rm H} \cos^2 \lambda}$$
(3.9)

where R. (i=E,N) are the radii of curvature of the earth<sup>1</sup>.

f See current velocity vector is defined in the mavigation coordinate system and is resolved through the sea current heading eagle to the northerly and easterly components.









(3.8)

All coordinate systems used are Cartesian, right hand oriented.

The velocity error rate equations obtained from Equation (3.7) in scalar form are

$$\delta \hat{\mathbf{V}}_{\mathbf{N}} = \{\delta \hat{\mathbf{V}}_{\mathbf{g}} \cos \psi_{\mathbf{q}} - \delta \mathbf{V}_{\mathbf{g}} \psi_{\mathbf{q}} \sin \psi_{\mathbf{q}} - \delta \psi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \sin \psi_{\mathbf{q}} - \delta \psi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \sin \psi_{\mathbf{q}} - \delta \psi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \psi_{\mathbf{q}} - \delta \psi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \psi_{\mathbf{q}} \cos \psi_{\mathbf{q}} - \delta \psi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \phi_{\mathbf{q}} \cos \psi_{\mathbf{q}} - \delta \psi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \phi_{\mathbf{q}} \cos \psi_{\mathbf{q}} - \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \phi_{\mathbf{q}} \sin \psi_{\mathbf{q}} - \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \phi_{\mathbf{q}} \cos \phi_{\mathbf{q}} - \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \phi_{\mathbf{q}} \cos \phi_{\mathbf{q}} + \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} - \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} + \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} - \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} + \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} - \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} + \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} - \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} + \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} - \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} + \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} - \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} + \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} - \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} + \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} - \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} + \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} - \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} + \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} - \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} + \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} + \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} - \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} + \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} + \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} + \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} - \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} + \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} + \delta \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \mathbf{V}_{\mathbf{g}} \cos \phi_{\mathbf{q}} \mathbf{V}_{\mathbf{g}} \mathbf{V}_{\mathbf{g}} \mathbf{V}_{\mathbf{g}} \mathbf{V}_{\mathbf{g}} \mathbf{V}_{\mathbf{g}} \mathbf{V}_{\mathbf{g}} \mathbf{V}_{\mathbf{g}}$$

$$- S \dot{\Psi}_{g} V_{g} \sin \psi_{g} - \delta \dot{\Psi}_{g} \dot{V}_{g} \sin \psi_{g} - \delta \dot{\Psi}_{g} V_{g} \dot{\Psi}_{g} \cos \psi_{g} - \delta \dot{\Psi}_{g} V_{g} \dot{\Psi}_{g} \cos \psi_{g} + \delta \dot{\Psi}_{g} \dot{\Psi}_{g} \cos \psi_{g} \sin \psi_{g} + \delta \dot{\Psi}_{g} \dot{\Psi}_{g} \sin \psi_{g} \sin \psi_{g} + \delta \dot{\Psi}_{g} \dot{\Psi}_{g} \sin \psi_{g} \sin \psi$$

$$+ \delta \psi_{ac} \dot{\Psi}_{ac} \cos \psi_{ac} - \delta \psi_{ac} \Psi_{ac} \dot{\Psi}_{ac} \sin \psi_{ac} ] \qquad (3.11)$$

$$\delta \hat{\mathbf{V}}_{\mathbf{u}} = -\delta \hat{\mathbf{V}}_{\mathbf{D}} , \qquad (3.12)$$

 $\delta \psi_{g_0}$  and  $\delta V_{g_0}$  are modeled as time-correlated noise sources. The  $\delta \dot{\psi}_{g_0}$  and  $\delta \dot{V}_{g_0}$  terms have second-order effects which are neglected. The  $\delta \dot{V}_i$  (i = s, f, D) are modeled as time-correlated noise sources and  $\delta V_i$  (i = s, f, D) terms are presented in Section 3.2.2.

# · 3.2.1 Gyrocompass Rate Error Equations

3232232

(Kanada and

A typical two-degree-of-freedom gyrocompass is shown in Figure 6. The rates sensed by the gyrocompass are contained in the bracketed terms. In theory the compensation of the bracketed terms is implemented by applying signals of the same magnitude but of opposite sign to the corresponding summing functions. In practice, however, the innecessibility of the summing functions requires compensating terms to be applied to the level and manuth or to level (only) torquers of the gyrocompass. In the analysis below a theoretical compensation method is assumed; this deviation from the practice simplifies the computer simulation of error equations and, most importantly, yields an universal error model compatible with most gyrocompasses available.

The error performance in each case, with few exceptions, yields the same results.

Referring to Figure 6, the first-order perturbation of the mechanisation equations, and assuming  $\delta R_1$ (i=E,N)  $\simeq 0$ ,  $\delta K_n \simeq 0$ ,  $\delta \Omega \simeq 0$ ,  $\delta K_m \simeq 0$ ,  $\delta \tau \simeq 0$ ,  $A_n \delta \alpha > \alpha \delta A_n$ ,  $g_s \theta \gg k_n \delta A_n$ , yields

$$b\dot{\psi} = -\left\{ \left[ \delta \lambda \left( \Omega \cos \lambda + \frac{V_{B}}{R_{B}} \sec^{2} \lambda \right) + \frac{\delta V_{B}}{R_{B}} \tan \lambda \right] + g \delta \alpha + \epsilon_{\psi} \right\}$$
(3.13)

$$\delta \dot{\eta} = \left\{ -\delta \psi \left( \Omega \cos \lambda + \frac{V_{\rm H}}{R_{\rm H}} \right) - K_{\rm D} g \delta \eta + A_{\rm H} \delta \alpha - \frac{\delta V_{\rm H}}{R_{\rm H}} - \delta \dot{\psi}_{\rm T} + \epsilon_{\eta} \right\}$$
(3.14)

$$\delta \alpha = \left\{ \frac{\mathbf{g} \mathbf{k}_{\mathbf{g}}}{\tau_{\mathbf{g}}} \delta \eta - \frac{1}{\tau_{\mathbf{g}}} \delta \alpha \right\}.$$
(3.15)

The  $\partial \mu_{\tau}$  term in Equation (3.14) is the residual torquing error. The  $\epsilon_1$  ( $i = \psi, \eta$ ) are the total instrument drift errors due to biam, noise (short and/or long correlated), warm-up noise, scale factor error, non-orthogonality, mass unbalance (g-sensitive), anisoelasticity ( $g^2$ -sensitive). The orthogonality error angles and the scale factor errors are multiplied by the corresponding total rates (i.e., vehicle and earth rate). The Equations (3.13) and (3.14) are valid whether the gynocompass consists of two single-degree-of-freedom gyros or a two-degree-of-freedom gyro. The number of elements to be implemented in the on-board computer is dictated by the sources requirements, memory allocates and the type of compensation used.

# 3.2.2 Doppler/Sonar Error Equation

The Doppler/Sonar output is the calibrated velocity vector defined in the ship's dynamic coordinate frame,

$$(\overline{\mathbf{v}}'')^{\mathsf{T}} = (\mathbf{v}_{\mathsf{f}}'' \mathbf{v}_{\mathsf{ss}}'' \mathbf{v}_{\mathsf{D}}'') .$$

# Referring" to Section 2,

In Equations (3.14) through (3.48) the sonar beam angles are assumed to be 30°, measured from the vertical (down) direction.



usons highly helder was

$$V_{g}^{\mu} = \frac{\Delta f_{g} \sigma_{\psi}}{2f_{g}} \quad (\Delta f_{g} = f_{g} - f_{h})$$
(3.16)

$$V_{\mathbf{s}}^{\prime\prime} = \frac{\Delta \mathbf{f}_{\mathbf{s}} \mathbf{o}_{\mathbf{v}}}{2 \mathbf{f}_{\mathbf{t}}} \quad (\Delta \mathbf{f}_{\mathbf{s}} = \mathbf{f}_{\mathbf{s}} - \mathbf{f}_{\mathbf{p}}) \tag{3.17}$$

$$V_{D}^{\prime\prime} = \frac{\sigma_{V}}{\sqrt{3}} \left( \frac{\Sigma f_{n}}{2t_{e}} - 1 \right) = \frac{\sigma_{V}}{\sqrt{3}} \left( \frac{\Sigma f_{e}}{2f_{e}} - 1 \right)$$
 (3.18a)

$$\frac{\sigma_v}{4f_v/3} \left(\Sigma f_f + \Sigma f_g - 4f_b\right) , \left(\Sigma f_f = f_f + f_h\right) , \left(\Sigma f_f = f_g + f_p\right) , \qquad (3.16b)$$

where  $f_f$ ,  $f_a$ ,  $f_a$  and  $\iota_p$  are the fore, alt, starboard and port transducer frequency outputs of the transducer;  $v_q$  is the velocity of propagation of sound in the sea medium. The  $f_t$  is the transmitting frequency.

Combining Equations (3.16) through (3.18b) with (3.3) and first-order perturbation of the result, with the assumptions  $\delta(\Delta f_{\ell}) \simeq 0$ ,  $\delta(\Delta f_{g}) \simeq 0$ , yields

$$\delta V_{g.} = \left\{ \left( \frac{\sigma_{v}}{2f_{v}} \right) \left( \frac{\delta \sigma_{v}}{\sigma_{v}} - \frac{\delta f_{v}}{f_{v}} \right) \left[ \Delta t_{g} \cos \theta + \frac{\sin \theta}{2\sqrt{3}} \left( \Sigma t_{g} + \Sigma f_{u} - 4f_{v} \right) - \left( \delta \theta \right) \left[ \Delta t_{g} \sin \theta - \frac{\cos \theta}{2\sqrt{3}} \left( \Sigma f_{g} + \Sigma f_{u} - 4f_{v} \right) \right] + \left[ \delta \Sigma f_{g} + \delta \Sigma f_{u} - 4\delta t_{v} \right] \frac{\sin \theta}{2\sqrt{3}} \right\}$$

$$(3.19)$$

$$\delta V_{g} = \left(\frac{\delta_{v}}{2f_{v}}\right) \left[ \left(\frac{\delta \sigma_{v}}{\sigma_{v}} - \frac{\delta f_{v}}{f_{v}}\right) \left[ \Delta f_{f} \sin \theta \sin \varphi + \Delta f_{g} \cos \varphi - \frac{1}{2\sqrt{3}} \left(\Sigma f_{f} + \frac{1}{2\sqrt{3}} + \Sigma f_{g} - 4f_{v}\right) \sin \varphi \cos \theta \right] + \delta \theta \left[ \Delta f_{f} \cos \theta \sin \varphi + \sin \varphi \sin \theta + \frac{1}{2\sqrt{3}} \left( \frac{\Sigma f_{f} + \Sigma f_{g} - 4f_{v}}{2\sqrt{3}} \right) \right] + \delta \varphi \left[ \Delta f_{f} \cos \varphi \sin \theta - \Delta f_{g} \sin \varphi - \frac{1}{2\sqrt{3}} - \frac{1}{2\sqrt{3}} \left( \frac{\Sigma f_{f} + \Sigma f_{g} - 4f_{v}}{2\sqrt{3}} \right) \right] - \left( \frac{\sin \varphi \cos \theta}{2\sqrt{3}} \right) \left[ \delta \Sigma f_{f} + \frac{1}{2} \delta \Sigma f_{g} - 4\delta f_{g} \right], \qquad (3.20)$$

Equations (3.19) and (3.20) contain the  $\delta V_0''$  term.  $\varphi$  and  $\theta$ , of the trigonometric terms shown in the equations, are the woll and pitch angles, respectively. The corresponding pitch and roll errors (i.e.,  $\delta\theta$  and  $\delta\varphi$ ) are due to attitude readout or inclinemeter resolut errors and pitch and roll amplitude<sup>\*</sup> and frequency errors, i.e.,

$$\delta\theta = \delta\theta_0 + \delta\theta_1 \sin \omega_0 t + \delta\omega_0 \theta_1 \cos \omega_0 t + n(\theta)$$
(3, 21)

$$\delta \varphi = \delta \varphi_0 + \delta \varphi_s \sin (\omega_0 t + \gamma) + \delta \omega_0 \varphi_s \cos (\omega_0 + \gamma) + n(\varphi) , \qquad (3.22)$$

The  $[\delta \omega_i (i = \theta, \varphi) = 2\pi \delta f_i]$  is a function of the ship's dynamics;  $\gamma_i$  is an arbitrary phase angle. The noise terms  $a(i) \quad (i = \theta, \varphi)$  are defined by

$$\mathbf{E}\{n_{k}(i)[n_{j}(i)]^{T}\} = \delta_{kj}Q(i), \qquad (3.23)$$

where  $\delta_{kj}$  is the Krouecker delta and Q(i) is the so-called white-poise variance<sup>†</sup> .

3.2.3 Velocimeter Error Equations

Referring to Section 2.3.3, the velocimeter output is defined by

 $C_{v} = \left\{ 4422 + 11.257 - 0.0457^{2} + 0.01820 + 4.3(8-34) \right\}, \qquad (3.24)$ 

" A constant plus simusoid is assumed.

<sup>†</sup> Tyrical values for  $f_i = [(6 \text{ to } 10) \text{ sec}]^{-1}$ .

Barri Barris I I 

(3.25)

where

Services.

- $v_{\psi}$  = speed of sound in water (ft/sec)
- T = water temperature ( <sup>Op</sup>)
- D = depth of transducer (ft)
- salinity of water (parts per thousand).

The first-order perturbation of Equation (3.24) yields

SO<sub>w</sub> = 11.258T - 0.09T8T + 0.01528D + 4.388 .

The errors due to measurement of the depth of the transducer and salinity of water are small and hence may be neglected. Hence

$$\delta C_{\psi} \simeq (11.25 - 0.09T) \, 8T$$
 (3.26)

The rate of change of the temperature error is modeled as an exponentially correlated noise process.

3.2.4 Nodels for Exponentially Correlated Noise Sources

The exponentially correlated noise sources  $k_{j}$  , stated in previous sections, are modeled by

$$\mathbf{E}[\mathbf{k}_{1}\mathbf{k}_{1}^{T}] \stackrel{i}{=} \sigma_{1}^{t} \mathbf{e}^{-\beta_{1}[\tau]}, \qquad (3.27)$$

where  $\sigma_1^2$  is variance of the process and  $\beta_1$  is the inverse correlation time. Note that Equation (3.27) is obtained by passing a white-noise process through a first-order filter.

. . . .

3.2.5 The System State Transition Matrix

The equations presented in Sections 3.1 and 3.1.1 through 3.1.3 are in the form of

₹ = AŽ + ₩ , ' (3.28)

The state transition matrix  $\Phi$  is defined by<sup>2</sup>

$$\exp\left(\int_{0}^{\Delta t} A(\tau) \ d\tau\right) = \left\{I + \int_{0}^{T} A(\tau) \ d\tau + \frac{1}{2!} \left(\int_{0}^{T} A(\tau) \ d\tau\right)^{2} + \frac{1}{2!} \left(\int_{0}^{T} A(\tau) \ d\tau\right)^{2} \dots (3.29)\right\}$$

Expansion of each term yields

$$A(\tau) = \left\{ A_0 + \tau \dot{A}_0 + \frac{\tau^2}{2i} \dot{A}_0 \dots \right\}$$
 (3.30)

$$\int_{0}^{\Delta C} A(\tau) d\tau = \left\{ A_{0} \Delta t + \dot{A}_{0} \frac{\Delta t^{2}}{2!} + \ddot{A}_{0} \frac{\Delta t^{2}}{3!} \dots \right\}$$
(3.31)

$$\left(\int_{0}^{\Delta t} A(\tau) d\tau\right)^{2} = \left\{A_{0}^{2} (\Delta t)^{2} + (A_{0}\dot{A}_{0} + \dot{A}_{0}A_{0})\frac{\Delta t^{2}}{21}\right\}$$
(3.32)

$$\left(\int_{0}^{\Delta t} A(\tau) d\tau\right)^{3} = \left\{A_{0}\Delta t^{2} \dots\right\}.$$
(3.33)

Hence

$$\exp\left(\int_{0}^{\Delta t} A(\tau) d\tau\right) = \left\{I + \Delta t A_{0} + \frac{\Delta t^{2}}{2I} (\dot{A}_{0} + A_{0}^{2}) + \frac{\Delta t^{2}}{3I} \left[X_{0} + \frac{3}{2} (AA_{0} + \dot{A}_{0}A_{0}) + A_{0}^{2}\right] + \ldots\right\}$$
(3.24)

Such expansion, however, has mome restrictions<sup>2</sup>. Assuming  $\Delta t$  to be small and A(t) constant during the interval. Equation (3.34) reduces to a simply Taylor series expansion. The computation of Equation (3.34) contains only a sufficient number of terms so that additional terms are negligible by comparison with the partial sum to that point<sup>3</sup>. For this marine navigation system, the state transition matrix is computed by truncating the series after the third term<sup>4</sup>.

Using the above approximations allows writing the discrete form

 $\hat{x}_{k+1} = \Phi X_k \,, \qquad (3.35)$ 

\* It is noted, however, that truncation beyond two terms for up to 4 hours of mavigation, with  $\Delta t = 80$  sec, yields less than is error. Such a method is equivalent to simple rectangular integration.

The state transition matrix obtained is in the form of

$$\Phi = \begin{bmatrix} \Phi_{11} & | \Phi_{12} \\ 0 & | \Phi_{22} \end{bmatrix}, \qquad (3.36)$$

where  $\Phi_{i1}$  and  $\Phi_{i2}$  are derived by forming the A matrix from the equations given in previous sections. The  $\Phi_{22}$  matrix contains exponential terms  $e^{-A_1\Delta t}$ . The simulation state vector is presented in Table I.

The ship-borne mechanization of a large state vector, as presented in Table I, is constrained by memory and the computational speed of the computer. Judicious selection of state variables through the use of a sophisticated simulation program will (and indeed does) reduce the size of the state vector to less than one half of the size shown.

# 3.3 Kalwan Mechanization

The Doppler/Sonar-gyrocompass navigation system is sugmented by the NNSS generated Doppler signals received and processed by the receiver. Optimal mixing of sensor signals are implemented by the Kalman filter (Ref.4, Vol.III, pp. 223-280). The Kalman equations are

(a) Extrapolate

$$P_{k} = \Phi P_{k-1} \Phi^{T} + Q_{k-1}$$
 (3.37)

(b) Filter

$$K_{k} = P_{k} H_{k}^{T} (H_{k} P_{k} H_{k}^{T} + R_{k})^{-1}$$
 (3.38)

$$P_k^0 = (I - R_k H_k) P_k \qquad (3.39)$$

$$\hat{\mathbf{x}}_{\mathbf{r}}^{\mathbf{r}} = \hat{\mathbf{x}} \hat{\mathbf{x}}_{\mathbf{r}} + \mathbf{K}_{\mathbf{r}} (\mathbf{n}_{\mathbf{r}} - \hat{\mathbf{z}}_{\mathbf{r}}) \quad (3.40)$$

Equations (3.37) and (3.39) are the covariance of errors in the estimate before and after the updating process, respectively". The estimate vector is set to zero following a correction after each update.

### .3.3.1 The State Noise Covariance Matrix Q

The state noise covariance matrix Q is in the form of

$$Q = \begin{bmatrix} 0 & | & 0 \\ 0 & | & Q_{2} \end{bmatrix}$$
(3.41)

The Qgg matrix consists of diagonal elements

$$q_i = \sigma_i^2 (1 - e^{-2\beta_i \Delta t}),$$
 (3.42)

where  $\sigma_i^2$  is the variance of the process and  $\beta_i$  is the inverse correlation time. If the correlation is defined apatially, it is normalised by the ship's sea-track velocity.

# 3.3.2 The Kalman Extraction Matrix II

The satellite dynamic signal processing will yield range and range-rate measurements. The processing of the signals is a characteristic of the receiver used. Correspondingly, the user has the options of available optimal data-averaging techniques and implementing range and/or range-rate or range-difference and/or range-rate-difference sethods.

Let Z = f(x) define the actual observation, where x is the dynamic state of the satellite and the navigator. If Z = f(x) is expanded in a Taylor series and higher order terms are neglected,

where s is the measurement error vector and  $\mathbf{v}_i$  is the observation noise. The H matrix shows how the observed errors are distributed in the state space. The H matrix is defined by

 $z = \begin{bmatrix} \rho \\ \rho \end{bmatrix}$ 

$$H = \frac{\partial z}{\partial x} = \begin{bmatrix} H_{\rho} \\ H_{\bar{\rho}} \end{bmatrix}$$
(3.44)

Let the dynamic measurement consist of range and range-rate vector, i.e.,



(3.45)

(3.56)

412

$$\hat{P} = (\hat{X}_{ai} - \hat{x})\hat{1} + (\hat{Y}_{ai} - \hat{y})\hat{j} + (\hat{Z}_{ai} - \hat{x})\hat{k}$$
(3.46)  
$$\vec{p} = (\hat{X}_{ai} - \hat{z})\hat{1} + (\hat{Y}_{ai} - \hat{y})\hat{j} + (\hat{Z}_{ai} - \hat{x})\hat{k}$$
(3.47)

The 1, 1, 2 are unit vectors in the geocentric, right-handed Cartesian coordinate frame\*.

Correspondingly, the absolute range is

$$|\mathcal{P}| = \rho = \sqrt{\langle \Delta x^2 + \Delta y^2 + \Delta x^2 \rangle}, \qquad (3.48)$$

where

$$\Delta i (i F x, y, z) = (i_{g} - i) , \qquad (3.49)$$

and the absolute range rate is

$$|\tilde{\rho}| = \dot{\rho} = \frac{p \cdot \tilde{\rho}}{|p|}. \qquad (3.50)$$

Let the state vector consist of the ship's and satellite's position and velocity errors, i.e., a total of 12 state variables. Hence

$$\frac{\partial \rho}{\partial(\mathbf{x}, \mathbf{y}, \mathbf{z})} = \left[ \left( \frac{\mathbf{x} - \mathbf{X}_{g1}}{\rho} \right) \left( \frac{\mathbf{y} - \mathbf{Y}_{g1}}{\rho} \right) \left( \frac{\mathbf{z} - \mathbf{Z}_{g1}}{\rho} \right) \right] = \mathbf{H}_{\lambda \lambda}, \qquad (3.51)$$

where i is the index of the observed satellite.

$$\frac{\partial \rho}{\partial (\dot{x}, \dot{y}, \dot{s})} = H_{12} = 0 \qquad (3.52)$$

$$\frac{1}{\partial (X_{ni}, X_{ni}, Z_{ni})} = -H_{11} = H_{13}$$
(3.53)

$$\overline{\partial(\hat{X}_{ui}, \hat{Y}_{ui}, \hat{Z}_{ui})} = H_{1u} = -H_{1z} = 0.$$
 (3.54)

Hence

$$I_{\mu} = \begin{bmatrix} H_{11} \\ H_{12} \end{bmatrix} = 0 \begin{bmatrix} H_{13} \\ H_{13} \end{bmatrix} = - \begin{bmatrix} H_{11} \\ H_{14} \end{bmatrix} = 0 \begin{bmatrix} 1 \\ H_{14} \end{bmatrix} .$$
(2.55)

Similarly,

$$\frac{\partial \rho}{\partial (\mathbf{x}, \mathbf{y}, \mathbf{z})} = \left[ \left( \frac{\mathbf{\hat{x}} - \dot{\mathbf{x}}_{\mathbf{z}1}}{\rho} - \frac{\mathbf{x} - \mathbf{x}_{\mathbf{z}1}}{\rho^{\mathbf{z}}} \dot{\rho} \right) \left( \frac{\dot{\mathbf{y}} - \dot{\mathbf{x}}_{\mathbf{z}1}}{\rho} - \frac{\mathbf{y} - \mathbf{y}_{\mathbf{z}1}}{\rho^{\mathbf{z}}} \dot{\rho} \right) \left( \frac{\dot{\mathbf{x}} - \dot{\mathbf{z}}_{\mathbf{z}1}}{\rho} - \frac{\mathbf{z} - \mathbf{z}_{\mathbf{z}1}}{\rho^{\mathbf{z}}} \dot{\rho} \right) \right]$$

$$\frac{\partial \phi}{\partial(\underline{s},\underline{y},\underline{s})} = \left[ \left( \frac{\underline{x} - \underline{x}_{a\,\underline{s}}}{\rho} \right) \left( \frac{\underline{y} - \underline{y}_{a\,\underline{s}}}{\rho} \right) \left( \frac{\underline{x} - \underline{z}_{a\,\underline{s}}}{\rho} \right) \right] = H_{a\,\underline{s}} = H_{11}$$
(3.57)

$$\frac{\partial \rho}{\partial (\mathbf{X}_{s1}, \mathbf{Y}_{s1}, \mathbf{Z}_{s1})} = \mathbf{H}_{s3} = -\mathbf{H}_{s1}$$
(3.58)

$$\frac{\partial \dot{\partial}}{\partial (\dot{x}_{a1}, \dot{x}_{a1}, \dot{x}_{a1})} = H_{a_{a}} = -H_{a_{a}}.$$
 (3.59)

Hence

$$H_{j} = [H_{21}| H_{22}| H_{23} = -H_{21}| H_{24}].$$
 (3.60)

" The satellite and navigator position (velocity) are defined in terms of orbital and geodetic parameters, respectively. Inorder to compute range and range-rate vectors, all positions and velocit/es are transformed to the geocentric doordinate frame.


These universal models are applicable to any navigation satellite system. For example, when the NNSS is used, only one satellite per observation is available; thus the user may prefer to process range only, or rangerate only or range and range-rate models. If a constellation of matellites is available, as in the TRW-Navatar system, the  $H_{\rho}$  and  $H_{b}$  matrices are computed one per satellite; for example, if three satellites are simultaneously available.

$$H^{T} = [H_{\mu}^{1}H_{\mu}^{3}H_{\mu}^{3}] , \qquad (3.61)$$

where each  $H_{\rho}^{i}$  (i=1,2,3) and  $H_{\rho}^{i}$  (i=1,2,3) are in the form of Equations (3.55) and (3.60), respectively. Note that the number of satellites determines the number of rows of the extraction matrix. If a differencing scheme is employed in a constellation type system,

$$\mathbf{H}^{T} = \left[ (\mathbf{H}_{j}^{1} - \mathbf{H}_{j}^{2}) \right] \left[ (\mathbf{H}_{j}^{1} - \mathbf{H}_{j}^{3}) \right] \left[ (\mathbf{H}_{j}^{1} - \mathbf{H}_{j}^{2}) \right] \left[ (\mathbf{H}_{j}^{1} - \mathbf{H}_{j}^{3}) \right] .$$
(3.62)

The differencing scheme, with slight modifications in the transition matrix, is also applicable to the NNSS (single satellite per observation), i.e.,

$$H = \begin{bmatrix} H_{\rho}(t = t_{a}) - H_{\rho}(t = t_{b}) \\ H_{\rho}(t = t_{a}) - H_{\rho}(t = t_{b}) \end{bmatrix}.$$
 (3.63)

Equation (3.55), i.e.,  $H_{12}$ , must be computed to evaluate its numerical significance: for example, for high altitude satellites it may be neglected.

For the NNSS, the satellite errors are not carried in the extraction matrix H ; hence  $H_{13}$  ,  $H_{14}$  ,  $H_{23}$  , and  $H_{34}$  are deleted, i.e.,

$$H = \begin{bmatrix} \frac{H_{11}}{H_{21}} & 0 \\ H_{21} & H_{22} = H_{11} \end{bmatrix}$$
(3.64)

The satellite errors which are not lumped to the observation noise, v , are added to the observation vector, i.e.,

$$=$$
 H8X + L(mi) +  $v$ , (3.65)

where L(si) is defined by two time-dependent bias and noise processes:

$$L(si) = S_{1,i}t + S_{2,i}t^{2} + n_{si}$$
, (3.66)

The n<sub>si</sub> are modeled as time-correlated noise sequences.

$$\mathbf{E}[\mathbf{n}_{ei}\mathbf{n}_{ei}^{\mathrm{T}}] = \sigma_{ei}^{2} \mathbf{e}^{-\beta} \mathbf{e}^{\Delta t}, \qquad (3.67)$$

where  $B_{i,i}$ ,  $B_{2,i}$  and n are approximated identically for all NNSS satellites<sup>6</sup> and are added as state variables 46, 47, 48 to the state vector presented in Table I.

Recall that the state transition matrix was derived in terms of the geodetic coordinate parameters; thus the covariance of errors and the corresponding error vector are defined in the geodetic coordinate system. On the other hand, the extraction matrix H is derived in terms of geocentric coordinates; consequently, the observation vector for a six-element system is modified as

$$\mathbf{z} = \mathbf{H}^{\pm} \delta \mathbf{\bar{X}} + \mathbf{L}(\mathbf{z}\mathbf{i}) + \mathbf{v} \qquad (3, 68)$$

$$H^{\bullet} = H \begin{bmatrix} T_u \\ 0 \end{bmatrix} T_u \end{bmatrix}$$
(3.69)

where T<sub>a</sub> is a geodetic to geocentric transformation matrix and is defined by

$$T_{u} = \begin{bmatrix} -\sin \Lambda & (-\sin \lambda \cos \Lambda) & (\cos \lambda \cos \Lambda) \\ \cos \Lambda & (-\sin \lambda \sin \Lambda) & (\cos \lambda \sin \Lambda) \\ 0 & \cos \lambda & \sin \lambda \end{bmatrix} = \begin{bmatrix} C_{xe} & C_{xn} & C_{xu} \\ C_{ye} & C_{yn} & C_{yu} \\ 0_{me} & C_{mn} & C_{mu} \end{bmatrix}.$$
 (3.70)

To this point all H elements corresponding to the position and velocities are defined. The partial derivatives of the range and range-rate with respect to the remaining variables are zero. Referring to Table 1, the x, y position errors are defined in terms of longitude and latitude, and in units of radians, respectively. After converting the pertinent variables to feet, the system H matrix is defined by

 $H_{1,e} = [h_{g}h_{b}h_{c}] \begin{bmatrix} \bar{C}_{ge} \\ C_{ge} \\ C_{ge} \end{bmatrix} \cos \lambda \qquad (3.71)$ 

In the sea-borne modelization,  $S_{1,1}$  and the noise process are modeled.







(3, 75)

(3. 78)

(3.77)

(3.78)

(3.79)

(3, 80)

(3,81)

(3.82)

(3, 83)

(3, 36)

(3.89)

x**o**nd

$$H_{i, ur} = \frac{t^2}{R_k}$$

$$H_{k,ve} = \frac{v}{R_k}$$

H1, 86 H

 $H_{1,s} = (h_{a}b_{b}h_{c}) \begin{bmatrix} C_{1a} \\ C_{ya} \\ C_{za} \end{bmatrix}$ 

$$H_{B_{1}1} = H_{1,B}$$

$$I_{g_1,i} = (h_{ij}h_{ij}h_{j}) \begin{bmatrix} C_{xe} \\ C_{ye} \\ C_{ge} \end{bmatrix} = 0$$

$$H_{s,s} = [h_{d}h_{s}h_{f}] \begin{bmatrix} 0_{y_{1}} \\ 0_{y_{2}} \end{bmatrix} = \begin{bmatrix} 0_{y_{2}} \\ 0_{s_{2}} \end{bmatrix}$$

$$H_{2, 27} = H_{1, 45}$$

$$\mathbf{f}_{s, ss} = [\mathbf{h}_{d}\mathbf{h}_{s}\mathbf{h}_{r}] \begin{bmatrix} \mathbf{O}_{xu} \\ \mathbf{O}_{yu} \end{bmatrix} \mathbf{R}_{r}$$
$$\mathbf{O}_{su} \begin{bmatrix} \mathbf{O}_{xu} \\ \mathbf{O}_{su} \end{bmatrix}$$

1

$$H_{2,x7} = 2t$$
 (3.64)

and

$$[\mathbf{h}_{a}\mathbf{h}_{b}\mathbf{h}_{c}] = \left[\left(\frac{\mathbf{i}-\mathbf{x}_{a}}{\rho}\right)\left(\frac{\mathbf{y}-\mathbf{y}_{a}}{\rho}\right)\left(\frac{\mathbf{z}-\mathbf{z}_{a}}{\rho}\right)\right]$$
(3.87)

$$\begin{bmatrix} h_{d}h_{e}h_{f} \end{bmatrix} = \left[ \left( \frac{\dot{x} - \dot{x}_{ei}}{\rho} - \frac{x - x_{ai}}{\rho^{2}} \dot{\rho} \right) \left( \frac{\dot{y} - \dot{x}_{ai}}{\rho} - \frac{y - Y_{ai}}{\rho^{2}} \dot{\rho} \right) \left( \frac{\ddot{x} - \dot{x}_{ai}}{\rho} - \frac{x - Z_{ai}}{\rho^{2}} \dot{\rho} \right) \right]$$
(3.88)  
$$\dot{\rho} = \frac{1}{\rho} \left[ (x_{ai} - x) (\dot{x}_{ai} - \dot{x}) + (Y_{ai} - y) (\dot{y}_{ai} - \dot{y}) + (Z_{ai} - x) (\dot{z}_{ai} - \dot{x}) \right]$$
(3.89)

$$\rho = \sqrt{[(X_{ai} - x)^2 + (Y_{ai} - y)^2 + (Z_{ai} - x)^2]} .$$
 (3.90)

12- 1 1' in the state of th Hin Hing

and the second second

#### 3.3.3 The Observation Error Covariance Matrix R

The  $V_j$  term presented in Equation (3.43), the observation noise vector, is an additive random sequence with known statistics, 1.9.,

$$\mathbf{E}[\mathbf{v}_{\mathbf{i}}] = \mathbf{0} \tag{3.91}$$

$$\mathbf{E}[\mathbf{v}_{i\mathbf{t}_{k}}\mathbf{v}_{i\mathbf{t}_{k+1}}] = 0 \tag{3.92}$$

$$\mathbb{E}\left[\Psi_{j}\overline{X}\right] = 0 \tag{3.93}$$

$$\mathbf{E}[\nabla_{i}\nabla_{i}^{T}] = \delta_{ij}R_{ij}, \qquad (3.94)$$

where  $\delta_{ij}$  is the Kronecker delta and  $R_i = \sigma^2_{wi}$  is a non-negative definite matrix. The R matrix, for nondifferencing cases, is a diagonal matrix. If, however, the satellite system provides more than one satellite and a differencing scheme is used, the R matrix contains off-diagonal elements as well.

In our model the R matrix is defined by

$$\mathbf{R} = \begin{bmatrix} \left( \sigma_{\mathbf{x}\rho} / \mathbf{R}_{\mathbf{k}} \right)^2 & \mathbf{0} \\ \mathbf{0} & \sigma_{\mathbf{x}\rho}^2 \end{bmatrix} . \tag{3.95}$$

## 3.4 Real Time Processing Considerations

Since Kalman mechanizations consist of sets of matrix operations, it can readily be seen that if the number of state variables in X is large, the matrix manipulations can become unwieldy. Hence, by judicious choice of state variables, the system dimension can be minimized; however, the degree of suboptimality can be constrained within limits. Also certain pseudo-states can be implemented to approximate the effects of noise in the system rather than explicitly modeling all noise functions as states; for example, in the form of

$$P_{\mathbf{k}} \odot = \Phi_{\mathbf{k}, \mathbf{k}-1} P_{\mathbf{k}-1} \Phi_{\mathbf{k}, \mathbf{k}-1}^{T} + B_{\mathbf{k}, \mathbf{k}-1} Q_{\mathbf{k}-1} B_{\mathbf{k}, \mathbf{k}-1}^{T}$$
(3.96)

extrapolation of the noise terms is incorporated in the Q matrix through a B coefficient matrix. Obviously, such modeling eliminates the capability of estimating the subject noise processes. The inversion required in the Kalman gain matrix can be simplified if the R matrix is a diagonal matrix. Under such circumstances the inversion is performed serially (i.e., row by row); this reduces each row to a scalar (i×i) value or simply to a constant or fraction. If, however, the Z matrix has more than 3 rows, the serial processing becomes a time consuming process. The extrapolation routine requires n×n matrix multiply operations, which consumes large computer memory. The collapsible-matrix-multiply routine!, based on element indexing, however, reduces memory requirements more than one half.

Each of the above simplifications aids in reducing the digital computer memory requirements for real time operation.

Still mother scheme to mid in preserving accuracy in a fixed-point mechanization is a real time rescaling module. This can be used to keep the covariance and related computations optimally scaled by periodically checking the elements of the covariance matrix and shifting if necessary. A typical situation where a rescaling module would be useful is postulated below. Assume that unfavorable geometric conditions prevail and three successive satellite passes are missed; near the equator this would mean about six to eight hours without a position fix. If positions (latitude error and longitude error) are state variables and assuming a linear degradation of position, a ratio of 64 to 1 on the respective covariance matrix elements would result if the fix occurred at 8 hours instead of one hour. If the covariance matrix were scaled to accommodate the worst case condition, then, when the system is operating at optimum accuracy, significance is lost since the data are shifted to the right in the "register". Essentially the scheme described above is a "pseudo-floating-point" rechanization. A true floating point mechanization could be implemented if system processing time and memory constraints were not severe. Computational precision can still be a problem in floating point if sufficient data bits are not carried.

Another concept to consider is the effect of the finite word length of the on-board computer on the Kalman filter-estimator mechanization.

All these problems may be evaluated through use of a sophisticated simulation program which includes the following:

Simulation of filter/sensor mismatches and their effects on system navigation performance.

Simulation of the suboptimal filter and of its effects on mystem navigation performance (i.e., the effects of neglecting some of the state variables).

The auto-invert routine is the optimal solution for up to 3×3 matrices. The concept may be extended to multiplem of 3 or modified for 2×2 and its multiples.

<sup>†</sup> The collapsible-matrix-multiply routine has been simulated and used in numerous occasions in air/seaborne Kalman filter mechanizations.

a Dia

Lan bar (me ) an Berne Birge (B) feine b

Simulation of the finite word length of the on-board computer and its effects on system navigation performance.

The problems presented showe may be rigorously analyzed through the use of the REAL-WORLD-COVARIANCE concept\* and addition of word length masking instructions to each arithmetical statement.

### 4. DYNAMIC SIMULATION RESULTS

In order to evaluate the system performance, several configurations were simulated. The mission profiles and performance results are presented below.

#### 4.1 Mission Profile

.

a bar a b

Two missions were simulated. The first mission assumed is described by a course originating near the equator which is consistent with about minutes between satellite passes. The ship's speed is 10 knots at a heading of  $45^{\circ}$ . Pitching and rolling motions of the ship were assumed to be  $3^{\circ}$  and  $5^{\circ}$ , respectively.

A variable water profile was assumed to illustrate the degradation in water depths of 0-400 ft and 400-600 ft. Depths greater than 600 ft were not considered in the error analyses described in the following section because the sea current drifts swamp the remaining system error sources. However, under such conditions, a local sea current biasing can be realized by implementing corresponding states in the state vector.

The second mission originated at 30 degrees North latitude and 118 degrees West longitude. Ship's speed was initially zero and it accelerated to 5 knots while simultaneously turning from a heading of 309 degrees to 315 degrees. For the remainder of the five and oue half hour mission, the ship remained on a straight course. The ship's dynamics are as described above.

#### 4.2 Error Analysis Results

Basic assumptions relating to the system mechanization in the first mission simulated are as follows:

- (i) Ship's attitude compensation is implemented. Inclinometer errors are about 0.3 degrees per axis.
- (ii) Gyro motion compensation is included. A standard heading error of 0.5 degrees for the non-Kalman filtered mechanizations is shown. The same gyro damped with Kalman filter corrections results in a heading error of less than 0.15 degrees.
- (iii) A one sigma position error of 300 ft/axis is assumed for the static satellite receiver subsystem error.
- (iv) The sonar Doppler errors are a function of water depth and are described in Section 2.
- (v) Satellite elevation angles at closest approach are between 20 and 60 degrees.

Figure 7 is a plot illustrating the radial position time history for three configurations in the first mission. The initial error is assumed to be zero. This is justifiable since the satellite yields position fixes which are zero-mean processes. Hence, by taking many fixes at the same static location, the position error in the limit approaches zero. In the first ninety minutes the radial error grows monotonically, at which point the satellite is assumed overhead. At this point the system is reset. Note that the Kalman filtered mechanization is reset to a lower value and, in the subsequent minety minutes, the error growth is at a reduced rate relative to the non-filtered systems. This results from the cross-calibration of the mavigation geneors upon receipt of satellite fix. As shown in Figure 7, the residual position error at each fix gets progressively better, from the increased knowledge of the system errors.

The error profile for the second mission is shown in Figure 8. These curves were generated using a dynamic simulation with the 48 state variables described in Section 3. The observation vectors employed were range and range-rate, as indicated on the figure. The range error is 170 feet per observation and the range-rate error is 0.1 ff/sec per observation. A gradual update is indicated, since a range and/or range-rate observation is made every one minute. Passes of variable duration and spacing were simulated to approximate a real-world environment. As cited above, the general improvements in accuracy are also present in this plot with one explainable exception -- the fix at five hours. The residual for this fix is higher than the previous fix residual, due to the satellite/

The dead reckoning error is larger for the second mission, due to the additional errors relating to the equipment that were implemented. For instance, frequency offset errors were assumed for the sonar Doppler and . velocimeter. Scale factor errors, non-orthogonalities, mass unbalance and anisoelasticity were initialized to describe the gyro compass as well as gyro drifts. The initial heading error was set at 0.75 degrees and was corrected down to 0.08 degrees at the fix.

"The concept is to simulate an optimal extrapolator in parallel with the suboptimal one. The Kalman gain computed in the suboptimal filter is applied to the real-world (full-size) covariance matrix. Hence the scientist conducting the study has at his disposal two (i.e., real-world and system) dovariance matrices. Obviously, the real-world dovariance matrix computed to inside the sole indication of the suboptimally computed Kalman gain effects in real-world.













L'and ther ten O letter

ant di sa disarta 🖌

اللم



## REPERENCES

ι.	Pitman, G.R.	Inertial Guidance.	John Wiley.	New York,	1962.	pp. 455-460.

2. DeRusso, P.M. State Variables for Engineers. John Wiley, New York, 1966, pp.364-366.

3. Ogata, K. State Space Analysis of Control Systems. Prentice-Hall, New Jersey, 1987, p.315.

 Leondes, C.T. Advances in Control Systems, Academic Press, 1966. et al.

 Chernoff, J. Utilization of Satellite Navigation Techniques in Oceanographic Operations. Marine Sciences Instrumentation, Vol. 4, Plenum Press, 1968.

6. Kalman, R.E. A New Approach to Linear Filtering and Prediction Problems. Journal of Basic Engineering, Vol. 82, No. 1, March 1980.

7. Kershner, R. B. Present State of Nuvigation by Doppler Measurement from Near Earth Satellites. APL Technical Digest, November-December, 1965.

 Newton, R. R. The Navy Navigation Satellite System, Space Research VII, North-Holland Publishing Company, Amsterdam, 1965.

9. Stansell, T.A.

The Navy Navigation Satellite System: Description and Status. Journal of the Institute of Navigation, Washington, DC, Vol. 15, No.3, Fall, 1968.



the set of the set of the set

and the second second

TABLE I The State Vector

Number	Symbol	Name of Error State Variable
1	8V.,	North velocity
2	٥٧ <sub>2</sub>	East velocity
3	8ψα	Gyro heading
4	δλ	Longitude
5	δΛ	Latitude
6	δη	Gyro tilt
7	δα	Gyro gravity reference
8*	ða <sub>v</sub>	Velocimeter
9	နံထို	Roll biss
10*	δφ <sub>a</sub>	Roll sinusoidal
11	80 <sub>0</sub>	Fitch bias
12*	80	Pitch sinosoidal
13*	õa,	Pitch frequency
14*	مسح	Roll frequency
15	δVao	Sea current velocity
16	$\delta \psi_{eo}$	Sea current heading
17	δψ <sub>τ</sub>	Gyro azimuth loop residual torquing
18	€ýs	Cyro bias A
19*	<sup>∉</sup> ¥¥1	Gyro noise No. 1A
20*	¢∳n ∎	Cyro naise No.2A
21*	k <sub>N</sub>	Mcale factor A
22*	77 <sub>NT</sub>	Orthogonality A
23*	77 <sub>NS</sub>	Orthogonality A
24*	γ <sub>ri</sub>	Mass unbalance A
25*	$\gamma_{NS}$	Mass unbalance A
26*	SN .	Animoelasticity A
37*	¢13	Gyro bias, T
28*	Con 1	Gyro noise No.1, T
29*	€øµz	dyro noise No.2, T
30*	77 <b>78</b>	Orthogonality, T
81.	771 	MARE UNDELENCE, T
04*	778	Anima under anima
. 60*	57	Anisoeinsticity, T
34.	14 1 20	Bownand welcotter wate
-00-	400 a	Forward velocity rate
40°	ave tu	Newtoni velocity rate
101	0 <sup>4</sup> 0 02	Vertical velocity
30-	3.0	
40	85.0	Cross svis fraguency sum
<b>4</b> 1	854	Along avis framency sum
420	u.	Warm-un bias A
410		
44	à.	Gyra tilt loop residuel torouine
45	δh	altitude (monidal)
•••		

Notes: 1. Element 44 is added for gyro model modification.

STUDE STUDE

<u> CARANY AN</u>

1.1.1

2. Asterisked numbers are not used in the sea-borne computer.

3. State variables due to satellite errors are added and are shown in Section 3.3.3.

HIC MADE DAY

533

Carls III Provide International

<u>}\_\_\_\_\_</u>

terine when the first

hand Other and the Other Station









÷.,

¢

с.

\* ++













534

ŝ

Ņ

ſ



计关注 的现在分词 医大会结核 的复数分组

1-1 A BITE Prestor









ι







# NATIONAL DISTRIBUTION CENTRES FOR AGARD PUBLICATIONS

AGAND publications are distributed to NATO Heater Nations through the National Distribution Control listed below.

		General J. MELHATE	
		Coordinatour AGAND - V.S.L. Frat Maior Porces Africanos	
		Caserue Princ' Baudouis	
	•	Place Dailly	
		Bruzelles 3	
	CANADA	Dis ctor of Brientific Information Services	
		Defence weensch Board	
		Department of Nutional Defence - 'A' Building	
		Uttema, Uncario	
	DERIARK	Dagish Defence Acsesses Goard	
		Østerbrogsdes kaberne	
		Copeningen D	
	PRANCE .	O.H.E.R.A. (Direction)	
		29, Avenue de la Division Leclere	
		92. Châtillo - Jous-Bagneus	
		Zentralatelle für läftfebröcknum tets	
		ved laforetion	
		maria-Theirain Str. 21	
		8 Nüchen 27	
		Atta: Dr Ing. H.J.RAUTERBRDG	
	GREECE	Greek Matimal Defence Staff	
		D, Jas	
		Athens	
•			
• •	ICELAND	F TOCLO: OF AVIAL ION	
		k diavir	
•			
	ITALY	Efficio del Delegato Mazional all' AGARP	
		Nimistero Difesa - Aeromántica Roma	
	LUXIMBOURG	Obtainable through MELGIUM	
		Katharlanda Dularation to article	
		Sational Aeroanece Laboratory W.R.	
		Atta: Hr A.H.GELLENER	
		P.O. Box 128	
·		delft	
	NORWAY	Notwerian Defense Research Establishment	
		Hais Library, c/o Hr P.L. EXEN	
		P.C. Box 25	
		N-2007 Kjeller	
	FORTUGAL	Directed do Servico de Material da Porca Aerea	
		Rus da Escois Politecnics 42	
		Listos	
		Attm. Brig. Generol Jose de Souse OLIVEIRA	
	TURKEY	Turbish General Staff / 18(2)	
		Aghara	
		· · · · · · · · · · · · · · · · · · ·	
	CHITED KINGDOM	Ministry of Tecanology	
		Block A Station forece line A Best Available Cor	<b>VC</b>
		St. Hary Cray	- J
		Orpington, Tect	
	VALL V GIATES	NELIGNAL AFTONAULICS AND POSCE Administration j	
		Langley Field, Virginia	
		Atta: Report Distribution and Storage Unit	

.

,

This AGARC publication has burn issued with a limited number of copies, and distribution has been made through the AGARD distribution centres set up in the NATO countries, a list of which is given in the inside cover.

If spare copies are not available at these custres, microfiche or mard copy (printed facesimile, or r production from Licroropy) may be purchased from

 Clearlushouse for Pederal
 ESRO/ELDO Space

 Scientific and Technical
 Documentation Service

 Isfo.mation (CPSTI)
 Suropean Space kesterch

 Springfield,
 Organization

 Virginia 22151, USA
 11°, Aveue de Neully

 92, Noully-sur-Seine, Prance

The request for microfishe or hard copy of an AGARD document chould include the AGARD surial number hen symilable, title, author (or editor) and publication dute; if known, the NASA Agression Number should also be quoted.

The unit price for a microfiche g ovided by CP5/I is \$0.65, and for a hard copy \$3.00. Payment should be made by check and must accompany order. Resittances from foreign countries should be ande by international money order or draft on an American bank, payable to CP571.

The unit cost for a microficht ,rovided by DERU//ELDO is French france 2.50 up to 60 pages, Lad for hard copy French france 0.50 per page.

Full billiographical references and abstracts of the newly isound AGARD publications are given in the following bi-monthly abstract journals with indexes:

> Scientific and Technical Aerospace Reports (SIAR) published by NASA. Scientific and Technical information Facility P O Box 33, College Perk, Maryland 20740, USa.

and

an anna an anna an

National Accounties and Space Administration WASHING TURE D. C. 2004 PUSTALE AND FEES PAID NATIONAL AFPOMALTICS AND SPACE AMMINISTRATION

OFFICIAL BUSINESS 100 001/1 00 365 70099 02672 DEFENSE MACHAE MATIL & SCATER SCAULTIFIC A M. AFCINICAL CAMERING DATION SLOVE STA ALYXAMORTATION (LEUDING) 22314 UMM14451114 KHAN TT COD RELEASEN LESRARY IVI B stel' :21

Harford House, 7-9 Charlotte St. London, NUP 1HD